

# Randomness, structure preservation, and low-regularity numerical approximations for nonlinear dispersive equations

Yvonne Alama Bronsard

# ► To cite this version:

Yvonne Alama Bronsard. Randomness, structure preservation, and low-regularity numerical approximations for nonlinear dispersive equations. Mathematics [math]. Sorbonne Université, 2024. English. NNT: 2024SORUS065 . tel-04615051

# HAL Id: tel-04615051 https://theses.hal.science/tel-04615051

Submitted on 17 Jun2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





# Sorbonne Université LJLL

École doctorale Sciences Mathématiques de Paris Centre Unité de recherche Laboratoire Jacques-Louis Lions

Thèse présentée par **Yvonne Alama Bronsard** Soutenue le **3 mai 2024** 

En vue de l'obtention du grade de docteur de Sorbonne Université

Discipline Mathématiques appliquées Spécialité Analyse numérique

# Schémas numériques pour les équations dispersives non linéaires : analyse à faible régularité, cadre aléatoire et préservation de symétries

Randomness, structure preservation, and low-regularity numerical approximations for nonlinear dispersive equations

Thèse dirigée par Katharina SCHRATZ

#### Composition du jury

Rapporteurs	Christophe Besse	professeur à l'Université Toulouse III Paul Sabatier	
	Ilya Chevyrev	MCF à l'Université d'Édimbourg	
Examinateurs	Albert Cohen Frédéric Lagoutière	professeur à Sorbonne Université professeur à l'Université Claude Bernard Lyon 1	
	Gigliola Staffilani	professeure au Massachusetts Insti- tute of Technology	
	Jérémie Szeftel	professeur à Sorbonne Université	président du jury
Directeur de thèse	Katharina Schratz	professeure à Sorbonne Université	

 $\grave{A}$  Lia, Stan, Sylvie, et Camille

iv

# Table des matières

Т	able	des matières	v
R	eme	erciements	ix
R	ésui	né	xvi
1	Int	roduction	1
	1.1	A scheme at first order : classical versus low-regularity approaches	2
		1.1.1 Illustration on the nonlinear Schrödinger equation	3
	1.2	Explicit convergence rates	7
		1.2.1 Comparing convergence results for the nonlinear Schrödinger equation	8
		1.2.2 Comparing convergence results for the Korteweg-de Vries equation	8
	1.3	Branching out in new directions	10
	1.4	A brief introduction to Part I	10
		1.4.1 Beyond periodic boundary conditions	10
		1.4.2 Higher order approximations	11
	1.5	A brief introduction to Part II	15
	1.6	A brief introduction to Part III	17
		1.6.1 A symmetric scheme for the nonlinear Schrödinger equation, suited for non-smooth solutions	18
		1.6.2 Symmetric higher-order extensions	20
Ι	в	eyond periodic boundary conditions and higher-order extensions	<b>23</b>
2	Err	or analysis of a class of semi-discrete schemes for the Gross-Pitaevskii equation	25
_	21	Introduction	25
	2.1	2.1.1 First-order low regularity integrator	26
		2.1.2 Second-order low regularity integrator	27
		2.1.3 Outline of the paper	
	2.2	Notation and nonlinear estimates	28
	2.3	First order scheme and analysis	29
	-	2.3.1 Case of periodic boundary conditions: $\Omega = \mathbb{T}$	29
		2.3.2 General boundary conditions: $\Omega \subset \mathbb{R}^d$	30
		2.3.3 Local error estimates	31
		2.3.4 Global error estimates	33
	2.4	Second order scheme and analysis	34
		2.4.1 A first approach to guarantee stability	35
		2.4.2 A second approach to guarantee stability	36
		2.4.3 Local error estimates	39
		2.4.4 Global error estimate	41
	2.5	Numerical Experiments	41
			10

~	-		
3	Lov	v regularity integrators via decorated trees	45
	3.1	Introduction	45
		3.1.1 Linearization and decorated trees	46
		3.1.2 Main strategy of the general numerical scheme	51
		3.1.3 Outline of the paper	52
	3.2	Decorated trees	52
		3.2.1 Definitions and dominant operators	52
		3.2.2 Approximated decorated trees	56
	22	Approximations of iterated integrals	57
	0.0	3.2.1 Characters on descripted forests	57
		2.2.2. Level some enderie for engenering to diterrate distance.	07 C1
	0.4	3.3.2 Local error analysis for approximated iterated integrals	01
	3.4	Low regularity numerical scheme	64
		3.4.1 Exact solution up to order $r$	64
		3.4.2 Numerical scheme and local error analysis	66
	3.5	Examples	67
		3.5.1 The Gross–Pitaevskii equation	68
		3.5.2 Filter functions, commutators and stabilisation	78
		3.5.3 Klein– and Sine–Gordon equations	79
Π	[ ]	The setting of a randomized initial condition	<b>85</b>
4	Ap	proximations of dispersive PDEs in the presence of low-regularity randomness	87
	4.1	Introduction	87
		4.1.1 Example	91
		4.1.2 Outline of the paper	94
	4.2	Decorated tree based structures for oscillatory iterated integrals	95
		4.2.1 Decorated trees	95
		4.2.2 Paired decorated forests	97
	4.3	Discretizing oscillatory iterated integrals	98
	4.4	Low regularity scheme	102
	4.5	Applications	102
	4.0		105
		4.5.1 Nonlinear Schrödinger	105
		4.5.2 KdV	110
IJ	Ι	Structure preserving schemes suited for low-regularity approximations	119
<b>5</b>	$\mathbf{A} \mathbf{s}$	ymmetric low-regularity integrator for the nonlinear Schrödinger equation	121
	5.1	Introduction	121
		5.1.1 Derivation of the scheme	125
		5.1.2 Result	127
		5.1.3 Outline of the paper	128
	5.2	Norms, spaces, and nonlinear estimates	128
	J.2	5.2.1 The case of periodic boundary conditions	120
		5.2.1 The case of periodic boundary conditions	129
	EЭ	The implicit potume of the scheme	129
	ى 5.3 7-4	The implicit nature of the scheme	131
	5.4		132
		5.4.1 Local error analysis	133
		5.4.2 Stability	142

6	Symmetric resonance based integrators and forest formulae				
	6.1	Introduction	145		
	6.2	Main ideas of the derivation of symmetric resonance based schemes	150		
	6.3	Decorated trees and generalised resonance based schemes	154		
		6.3.1 Dominant part and polynomial interpolation	156		
		6.3.2 A forest formula for resonance based schemes	160		
		6.3.3 Midpoint general resonance based schemes	163		
	6.4	Symmetric schemes	168		
		6.4.1 Symmetric interpolation	168		
		6.4.2 Conditions for symmetry	170		
		6.4.3 Examples	173		
	6.5	Numerical Experiments	183		
		6.5.1 The Korteweg–de Vries equation	183		
		6.5.2 The Nonlinear Schrödinger equation	183		
D	:hl:a	omen hy	101		
D.	IDHO	igi apiny	191		

# Bibliography

vii

# Remerciements

J'aimerais débuter en remerciant grandement Katharina Schratz, qui m'a ouvert les yeux sur un grand nombre de sujets très intéressants à l'interface entre le numérique et les EDPs. Merci de m'avoir initiée à tous ces projets captivants, en me laissant le champ libre sur certains, et des collaborateurs passionnés pour les autres; les deux expériences m'ont beaucoup apporté. Merci de m'avoir introduit à tous tes collègues qui passaient par le LJLL, ce qui m'a permis de m'insérer dans ton milieu académique, de m'avoir fait une confiance totale dans tout ce que je faisais, de m'avoir donné l'occasion de te représenter lors de conférences (à Nice, Augsburg, et La Thuile), et de m'avoir incitée à exposer mes travaux de nombreuses fois (20!). Pour tout cela je te suis extrêmement reconnaissante, et j'ai déjà hâte aux prochaines conférences que nous ferons ensemble.

J'adresse ma profonde gratitude à Christophe Besse et Ilya Chevyrev pour avoir accepté de rapporter ma thèse, un sincère merci pour votre relecture appliquée de mon manuscrit. Je souhaite également grandement remercier Albert Cohen, Frédéric Lagoutière, Gigliola Staffilani, et Jérémie Szeftel pour avoir accepté de faire partie de mon jury de soutenance, un tel jury me rend fière.

J'aimerais maintenant reculer dans le temps et remercier la FSMP qui m'a financé pendant le M1 et le M2, et qui m'a permis de suivre les cours d'excellent niveau du master à Jussieu. Merci à Fabrice Bethuel d'avoir été mon mentor pédagogique pendant ces deux ans de master, et d'avoir été présent lors de mes premier pas à Paris. Je tiens à remercier les permanents qui m'ont enseigné de superbes cours lors de ces deux années, qui m'ont grandement préparé et incité à poursuivre en recherche : merci beaucoup à Albert Cohen, Frédéric Hecht, Diane Peurichard, Benoît Perthame, Nastassia Pouradier Duteil, Sébastien Martineau, Hervé Le Dret, Fabrice Bethuel, Bruno Després, Ayman Moussa, Anne-Laure Dalibard, Anne de Bouard, Didier Smets, Xavier Claeys, Muriel Boulakia, et Laurent Boudin.

Merci aux autres permanents qui, par leur présence en salle café et dans les couloirs, contribuent à la convivialité et à l'atmosphère chaleureuse du laboratoire. Merci tout particulièrement à Valeria Banica pour toutes nos discussions, c'est toujours un grand plaisir de parler avec toi, de même au voisin tant adoré (avec raison) de Katharina : merci Antoine Gloria; merci François Murat pour les picnics saucisson et l'odeur unique de ta machine à café (et désolée du nombre de fois où vous souhaitiez trouver Katharina à son bureau, et qu'il n'y avait que moi!), merci Frédéric Hecht pour ta bonne humeur dans notre couloir 15-25!

Merci à Luis Almeida et à Albert Cohen d'avoir assuré mon comité de suivi de thèse (un peu dernière minute au mois d'août, mea culpa). Merci aussi à Malika Larcher pour avoir traité toutes mes missions (en principe il ne m'en reste plus beaucoup), merci à Emmanuel Trélat pour tout ce que tu fais pour assurer le bon fonctionnement du laboratoire.

Merci aussi aux personnes que j'ai vu lors de mes passages à Orsay. Merci beaucoup Frédéric Rousset, pour toutes nos discussions mathématiques, le temps que tu m'accordes à parler des espaces de Bourgain et à répondre à toutes mes questions avec grand intérêt, ce que j'apprécie beaucoup. Merci Patrick Gérard, ce fût à l'occasion de votre 60ème que j'ai mis pied pour la première fois à Orsay, et finalement merci aux doctorants et chercheurs que j'ai croisés lors de ces visites et qui les ont rendus toutes très plaisantes : Élise, Nicolas, Philippe, Louise, Rana, Nhi, Clément, Chenmin, Jiao, et Annalaura.

Je passe maintenant aux post doctorants avec qui j'ai eu le plaisir de discuter : David Lee, Nikola Stoilov, Franco Zivcovich, Yue Feng, et Georg Maierhofer avec qui j'ai beaucoup apprécié collaborer. Georg, je suis très contente d'avoir passé une bonne et intense année et demi ensemble au LJLL, et de tous nos échanges mathématiques fructueux. L'année scolaire prochaine tu pourras nous visiter à Rennes avec Erwan... toi qui raffoles de mille-feuilles et de mousses au chocolat lors de tes séjours parisiens, à Rennes on t'introduira au kouign-amann et au far breton!

Justement, ayant terminé le plat principal, passons aux desserts : les doctorants du LJLL. Je commence avec les anciens doctorants, les babas aux rhum.

Jules merci pour ton investissement incandescent au labo et chez les doctorants (et oui je vais prendre des vacances après ma soutenance), Lucas E j'ai hâte qu'on se retrouve dans une prochaine conf et en attendant j'espère que tu profites des puddings anglais, Antoine je te confie le fondant à la patate douce -toi qui me faisait des burgers exquis à la patate douce- merci pour tes soirées et tes devinettes que tu me laissais sur le tableau. Anatole, connais-tu l'omelette norvégienne (qu'on appelle au Canada, Baked Alaska) tu pourrais la flamber lors de tes spectacles. Pierre, sais-tu qu'il existe un « gâteau des roux », on t'en ferra un avec Matthieu le jour où tu nous invites chez toi, ou vice versa. Juliette, on t'apportera un Bratapfelkuchen la prochaine fois qu'on se voit à Aachen! Lucas J et Rémi c'est certainement le mont-blanc qui vous allèche le plus, bonne chance Lucas J pour en trouver un à Seoul. Lise, merci pour ton dynamisme, ta persévérance et tes conseils à Matthieu, à toi l'éclair au chocolat. Chourouk, tant de bonnes choses à dire de toi, je te donne le mille-feuilles, et j'espère que pleins de superbes choses sortiront de toutes les études que tu entreprends. Pauline, bon courage pour l'écriture du livre de documentation de ton code; María merci pour ton accueil au LJLL et pour avoir partagé ta gourmandise avec moi à la boulangerie de la Tour d'Argent; Ramòn, je me rappelle encore de tes excellents turròn que tu avais à ton pot, je vais essayer d'en trouver pour le mien. Nicolas, je suis encore impressionnée que tu aies réussi à avoir la nationalité française, moi je galère, un gros bravo. Rui, clémentine confite, un dessert exquis, bravo pour avoir trouvé dans l'industrie!

Puis je passe au plus jeunes, tout premièrement un grand merci à Liangying et Nicolaï, notre bureau était le meilleur des Saint-Honoré (désolé Liangying je sais que tu n'aimes pas trop le sucré). Aux doctorants de mon nouveau bureau, Thomas — just ripe — continue ton heureuse humeur et un jour tu me feras gouter à ta tarte tatin (et merci pour la tarte aux poires!); Elena, le limoncello du bureau 15-25-320, grazie a te; Ludovic merci d'être fidèle au GTT et bonne chance avec les calculs très longs que tu laisses au tableau; Manon et Tianyu bon courage pour la suite! Agustin mon alfajor, fidèle à son maté, tu étais un voisin de bureau absolument génial, et tu le seras encore l'année prochaine, toi à Nantes, moi à Rennes. Charles, en postdoc à Lyon tu te régaleras de tartes pralines. Toai, on s'est connu en M1 je suis bien contente qu'on ait fait le chemin jusqu'ici ensemble. Nga depuis le M2, ça fait quand même 4 ans! Mingyue merci encore pour les petits cadeaux; Eleanor et Ruikang pour les biscuits les jeudis; Zhe merci pour le GTT; Zhengping pour ton français impeccable; Robin pour les petites pauses café au long de la journée (imagine-toi un très grand monsieur arriver avec une très petite tasse en salle café, c'est pas mal); Yipeng et Christobal pour les coucous au long du couloir 15-25; Valentin pour avoir fait très bonne usage de ma carte crous; Fabrice pour nos discussions sur le monde quantique et le Québec; Roxane pour ton engagement dans la vie du labo et au-delà; Lucas P et Guillaume, pour avoir gardé le GTT actif avec Charles; Jean-Guillaume pour nos joyeuses discussions en 15-25-324 et au crous; Alessandro à nos discussion dans la cuisine de la MEC. Siguang, Aloïs, Federica, et Aleksandra, merci d'avoir pris le chapeau et d'être respos resto doctorants, vous faites un super boulot. Crisse qu'il y en a des doctorants! Je vous remercie tous pour l'ambiance chaleureuse et joyeuse que vous créez au labo!

J'en profite pour remercier l'ICERM pour les 6 semaines géniales de workshop en 2021, ainsi que Benoît Pausader, et les doctorants et post doctorants que j'ai rencontrés à cette occasion : Susi, Daniel, Joel, Patrick, Nicolas, Jaemin, Annalaura, Stefan, Louise, Chenmin, Eduardo, Jiaqi et Ana. Merci à Katie pour tes séjours à Paris, j'ai hâte à nos prochaines discussions sur les espaces de Bourgain. Merci à France Mainville et à la MEC, pour les quatre ans passés à la Cité Universitaire. Merci Florence et Damien Rocourt pour le champagne qui arrose ce pot. Finalement, à ma chère amie d'enfance Juliana, merci pour tous nos restos gourmands et nos pièces de théâtre! Je passe maintenant au digestif et au café gourmand : la famille.

Camille, le millésime, tu as l'art d'être pépére (deux accents aigus s.v.p., l'auteur n'est pas avare), économiste, poète, ardent indépendantiste québécois et défenseur de la langue française, fervent adorateur de la Belgique et de la France, toi qui m'as accueilli au 3750 rue de Frère André à bras grands ouverts lors de mon entrée à Montréal à 17 ans, qui m'a accompagnée (avec des langues de veaux et une confiance en moi inébranlable) pendant mes quatre ans d'études, et m'a incitée à poursuivre mes études dans un milieu francophone.

Lia, matheuse avec une vivacité incroyable, aussi grande indépendantiste québécoise, et

Stan, matheux, musicien et -même si américain- indépendantiste québécois par adoption.

Sylvie, grande soeur fidèle à son rôle, prête à tout pour moi, en particulier lorsqu'il est question de chaises musicales, indépendantiste québécoise par évidence.

Matthieu, matheux enthousiasmé par la vie, avec une curiosité mathématique étincelante à en avoir le sourire aux lèvres, toujours prêt à parler de maths à l'infini, petit nouveau, et indépendantiste québécois en formation.

Sylvie et Jean, couple dynamique en tabarouette, merci du temps passé avec vous, et pour avoir crée le petit nouveau.

À vous tous je lève mon verre, et vous remercie pour tout,... et aux prochaines aventures.

Gens du pays, c'est votre tour De vous laisser parler d'amour *Gilles Vignault, 1975*  nous te ferons, Terre de Québec lit des résurrections et des mille fulgurances de nos métamorphoses de nos levains où lève le futur de nos volontés sans concessions les hommes entendront ton pouls dans l'histoire c'est nous ondulant dans l'automne d'octobre c'est le bruit roux de chevreuils d'automne l'avenir dégagé l'avenir engagé

Gaston Miron, 1970, l'Hexagone

Remerciements

## Contraste

Et ronronnait le réfrigérateur Dans le silence du matin

Les deux petits enfants Et leur grand-père Ouvrirent sa porte

Il y avait là Dans le réfrigérateur La moitié d'un Saint-Honoré Ce gâteau à la crème Entouré de choux à la crème Et recouvert de miel

Il y avait là des confitures De la confiture de rhubarbe De la confiture de cassis De la confiture de framboise Et même de la gelée de coings Mère de toutes les confitures

Il y avait là Deux morceaux de fromage Une motte de beurre Des œufs Du beurre d'arachide Et la moitié D'un gigot d'agneau

Les deux petits enfants Qui étaient ravissants Étaient ravis

Le vieux grand-père Était un vieux grognon Il n'était ni ravi ni ravissant

Il referma la porte du réfrigérateur « Qu'allons-nous manger ? » Dit le vieux grand-père

> Camille Bronsard, Au jour J du mois M de l'an A, Les carnets de Dame Plume, 2011

# 1976

Tu es au carrefour des Cultures anciennes

Tu as vu Le tombeau de Khéops Celui d'Agamemnon celui de Clytemnestre

Tu as vu Les pierres du défi de maintes acropoles Les marbres raffinés de maintes agoras

Et du mont Palatin

tu as vu Les pierres du défi de Rome l'Éternelle

> Séduit par Andrinople Ébloui par Grenade Tu as vu Les étoiles mourir Sur les blancs minarets des mosquées de Sinan Et le soleil mourir Sur les murs cramoisis du divin Alhambra

L'Histoire est passée par ici Elle qui ne passe pas chez toi Elle qui ne veut pas de toi

Terrassier sous Khéops Potier sous Périclès Eunuque sous Soliman

Toi tu veux un pays que chacun te refuse Tu habites aujourd'hui une terre sans promesse Tu voudrais un destin

pour ton peuple et pour toi

Esclave Esclave d'âge en âge Tu appelles ta Revanche

Tu recherches l'Esprit qui vibrait en ces terres

Tu ne peux rien sauver de ce peuple qui meurt Au choc le plus léger ton empreinte s'efface Tu ne peux rien sauver de ta vie qui s'envole Par ce peuple qui meurt tu ne peux te survivre

Antigone mourante en un désert futur Tu ne peux qu'amuser la galerie française Funambule immobile entre des pics sublimes Tu ne peux qu'épater ceux qui ont réussi

#### Remerciements

Renié par tes dieux dont triomphe Athéna Tu ne peux plus changer ton échec en victoire Condamné au succès par tous les dieux anciens Tu ne peux réussir sans être mercenaire

Seul et désemparé tu reviens en ces lieux Et promènes sans fin ta lamentable errance Ton désespoir est laid tu maudis tous les dieux

Tu recherches l'Esprit qui vibrait en ces terres

Il ne peut rien pour toi Ta quête est sans objet

Tu es seul

ironiquement apatride comiquement cénobite superbement citoyen du monde Tu réalises tous les rêves fous

Ceux qui ont réussi les ont rêvés pour toi

Tu es seul

et tu penses à ton coin de terre À ton peuple lassé pourtant si près du but Doucement Tristement Bêtement Dans une euthanasie finement programmée

Tu es seul et tu sais que tu ne peux déchoir

Condamné au succès il te faut comploter Contre les dieux contre les faits contre l'Histoire Condamné au succès tu ne peux que lutter Avec des plans nouveaux et des trames nouvelles

Tu maudissais l'Histoire Trouves-y une faille

Tu dois vaincre l'Esprit qui vibrait en ces terres

Camille Bronsard, Au jour J du mois M de l'an A, Les carnets de Dame Plume, 2011

Remerciements

Schémas numériques pour les équations dispersives non linéaires : analyse à faible régularité, cadre aléatoire et préservation de symétries

Randomness, structure preservation, and low-regularity numerical approximations for nonlinear dispersive equations

#### Résumé

Le travail présenté dans cette thèse relève du domaine de l'analyse numérique et s'appuie sur des outils issus de l'étude des équations aux dérivées partielles (EDP). Nous nous concentrons sur les discrétisations temporelles des équations dispersives non linéaires. L'objectif est de réduire les hypothèses de régularité nécessaires lors de la conception et de l'analyse des méthodes numériques, afin de traiter les dynamiques à faible régularité.

La partie I de la thèse introduit de nouveaux schémas à faible régularité, adaptés à des domaines bornés génériques. Le chapitre 2 présente des résultats de convergence au premier et au second ordre pour l'approximation de l'équation de Gross-Pitaevskii, lorsque la donnée initiale et le potentiel sont peu réguliers. Le chapitre 3 généralise la construction de ces schémas aux ordres supérieurs, et pour une classe générale d'équations d'évolution non linéaires.

La partie II est constituée du chapitre 4, qui génère des constructions d'ordre élevé dans le cadre de conditions initiales aléatoires.

Finalement, la partie III se consacre à l'étude en temps long d'équations dispersives, et de leurs invariants, en considérant des schémas préservant leur structure. Elle débute avec le chapitre 5, qui introduit un nouvel intégrateur symétrique pour l'équation de Schrödinger non linéaire, et démontre des résultats de convergence à des taux fractionnaires, en fonction de la régularité de Sobolev de la donnée initiale. Par la suite, le chapitre 6 étend cette construction symétrique aux ordres supérieurs et pour la résolution numérique d'une classe générale d'équations dispersives. Des simulations numériques montrent que ces nouveaux schémas symétriques présentent d'excellentes propriétés de préservation de la structure. Les extensions aux ordres supérieurs développées aux chapitres 3, 4, et 6 se fondent sur de nouvelles techniques d'arbres

décorés, inspirées par le champ des EDP stochastiques singulières, via la théorie des structures de régularité.

Mots clés : équations dispersives nonlinéaires, faible régularité, schémas préservant la structure, arbres décorés

#### Abstract

The work presented in this thesis belongs to the field of numerical analysis, and builds on tools stemming from the study of partial differential equations (PDEs). We focus on time discretizations to nonlinear dispersive equations. The aim is to reduce the *smoothness* assumptions on the design and analysis of numerical methods, in order to treat *low-regularity* dynamics.

Part I of the thesis develops novel low-regularity schemes, suited for general bounded domains. Chapter 2 presents first and second order convergence results for the Gross-Pitaevskii equation, when both the initial data and the potential are non-smooth. Chapter 3 generalizes the construction of these schemes to higher order and to a general class of nonlinear evolution equations with potentials.

Part II of the thesis consists of Chapter 4, which considers higher-order constructions for *randomized* initial conditions. Part III of the thesis considers the long-time properties and invariants of the equation, and deals with *structure-preserving* schemes. We first introduce in Chapter 5 a novel symmetric time integrator for the nonlinear Schrödinger equation. We give fractional convergence rates as a function of the Sobolev regularity of the initial data. Chapter 6 extends the latter work by constructing higher order symmetric integrators for a general class of dispersive equations. All these new symmetric schemes exhibit excellent structure preservation and convergence properties, which are witnessed in numerical experiments.

The higher order extensions of Chapters 3, 4, 6 follow new techniques based on decorated tree series, inspired by singular stochastic PDEs via the theory of Regularity Structures.

Keywords: nonlinear dispersive equations, low-regularity, structure preserving schemes, decorated trees

Résumé

# Chapter 1

# Introduction

The study of nonlinear partial differential equations (PDEs), and more specifically those modeling nonsmooth phenomena, arising in nonlinear optics, shallow water waves, and wave turbulence, is at the heart of an intense international activity. While linear problems or smooth solutions are generally well understood, a reliable description of *nonlinear* and *non-smooth* phenomena remains an important challenge in computational mathematics today. This is especially the case when studying nonlinear dispersive equations, since one cannot expect the smoothing effects that are present in parabolic equations. The lack of smoothness in the solution leads to large errors in numerical simulations, massive computational costs and ultimately causes the failure of classical schemes. Yet many interesting physical phenomena exhibit non-smooth behavior. An essential task is thus to develop and study suitable numerical schemes for approximating such non-smooth behavior. To this end, we will be interested in constructing and analyzing numerical approximations to the following class of dispersive equations

$$\begin{cases} \partial_t u(t,x) - \mathcal{L}u(t,x) = p(u(t,x)), \quad (t,x) \in \mathbb{R} \times \mathbb{T}^d, \ d \in \{1,2,3\}\\ u(0,x) = u_0(x), \end{cases}$$
(1.1)

where p(u(t, x)) is a polynomial in u(t, x), its conjugate  $\bar{u}(t, x)$ , and their spatial derivatives, see (6.5) for more detail. We ask  $\mathcal{L}$  to be a linear operator which generates a strongly continuous group  $\{e^{t\mathcal{L}}\}_{t\in\mathbb{R}}$  of contractions on  $L^2$ , and require  $\mathcal{L}$  to involve only polynomials in the frequencies when mapped in Fourier space. Namely, we denote P(k) the Fourier multiplier of  $\mathcal{L}$  and require P(k) to be polynomial in k. We give two prototypical examples of nonlinear dispersive equations belonging to the above class (1.1).

**Example 1** A classical model in quantum mechanics and a natural starting point for the study of dispersive equations is the (cubic) nonlinear Schrödinger equation (NLS), which corresponds to taking  $\mathcal{L} = i\Delta$ ,  $P(k) = -i|k|^2$ , and  $p(u) = -i|u|^2u$ :

$$\partial_t u(t,x) - i\Delta u(t,x) = -i|u(t,x)|^2 u(t,x).$$

$$(1.2)$$

**Example 2** Another example is the Korteweg–de Vries equation (KdV), which serves as a model for waves in shallow waters, and corresponds to the case  $\mathcal{L} = -\partial_x^3$ ,  $P(k) = ik^3$ , and  $p(u) = \frac{1}{2}\partial_x(u^2)$ :

$$\partial_t u(t,x) + \partial_x^3 u(t,x) = \frac{1}{2} \partial_x \left( u^2(t,x) \right).$$
(1.3)

Throughout this manuscript I have been interested in developing new techniques to obtain computational schemes for approximating *in time* low regularity solutions to such equations, with an emphasis on what can be rigorously proven to guarantee their reliability. Numerically approximating low regularity solutions is a challenge since classical approximation techniques require smoothness. As an illustration, in the case of the NLS equation (1.2), both classical finite differences or splitting methods require at least  $H^2$ -solutions for  $L^2$ -convergence at first order, and  $H^4$ -solutions for  $L^2$ -convergence at second order [Lub08, BBD02, SS84, BDDLV21, HP17]. On the other hand, the equation is locally well-posed (LWP) for solutions in  $H^{\alpha}(\mathbb{T}^d)$ , with  $\alpha \ge 0$ ,  $\alpha > 0$ , and  $\alpha > 1/2$ for d = 1, 2 and 3 respectively. A question I have investigated is how to reduce the gap between the degree of regularity required by the numerical approximation and the regularity needed for the local well-posedness theory. The numerical challenges of discretizing such equations are in part governed by the theoretical challenges in proving their well-posedness. We quote Daniel Tataru [Tat04] introducing well-posedness results for the wave map equation :

The initial work in this field was mostly devoted to the local study of problems with reasonably smooth initial data. Often this reduces to analyzing linear problems. The well-posedness questions becomes harder to answer when one considers either problems with less regular data or solutions which exist for a larger time. Then one needs to gain a better and better understanding of the nonlinear wave interactions.

The same can be said in the discrete setting. The goal of this manuscript is to deepen our understanding on both aspects, by constructing and analyzing new schemes which converge under *less regular data* and which have a good *long-time* behavior. Parts I and II of this manuscript focus on obtaining schemes that converge under less regular data, in deterministic and randomized settings. Part III deals with long-time behavior of the schemes.

Remark 1.0.1 (From semi-discrete to fully discrete approximations) Throughout this manuscript we will focus on building and analyzing novel approximations in *time*. Going from the semi-discretization in time to the fully discrete problem is an important step. We refer for instance to the works [LW21, OY22, CLR20, LMS21, LW21, BMW23, BW23b] which explore fully discrete low-regularity settings.

We begin with a comparison between *classical* approximation techniques and novel *low-regularity* approaches.

### 1.1 A scheme at first order: classical versus low-regularity approaches

A large toolbox of numerical schemes for dispersive equations has been developed, based on different discretization techniques. *Exponential integrators* and *Lawson methods* rely on discretizing Duhamel's formula, otherwise called the variation-of-constants formula. *Finite difference* schemes directly discretize the underlying equation (1.1) via explicit or implicit Taylor approximations. *Splitting methods* are based upon splitting the full equation into a series of simpler subproblems. The above classical schemes often allow for a precise and efficient approximation, but are designed for *smooth* solutions, requiring high regularity assumptions. To illustrate this, we will construct and compare first order time-approximations to the class of dispersive equations (1.1). We start by writing *Duhamel's formula* for u(t):

$$u(t) = e^{t\mathcal{L}}u_0 + \int_0^t e^{(t-s)\mathcal{L}}p(u(s))ds$$
(1.4)

and iterate Duhamel's formula for u(s) inside p(u(s)) to obtain the next iterate

$$u(t) = e^{t\mathcal{L}}u_0 + \int_0^t e^{(t-s)\mathcal{L}}p(e^{s\mathcal{L}}u_0)ds + \int_0^t \int_0^s \dots ds_1 ds.$$
 (1.5)

Namely, the above equality (1.5) follows by Taylor expanding the nonlinearity p appearing in (1.4) around the linear flow:  $p(u(s)) = p(e^{s\mathcal{L}}u_0) + \int_0^s \dots ds_1$ , with an integral remainder. We search an approximation over one time step  $t = \tau$ . In order to obtain a first order approximation in *time*, we neglect the double integral in (1.5), which corresponds to the second order term:

$$u(\tau) = e^{\tau \mathcal{L}} u_0 + \int_0^\tau e^{(\tau - s)\mathcal{L}} p(e^{s\mathcal{L}} u_0) ds + O(\tau^2).$$
(1.6)

If we suppose that p(u) is locally Lipschitz, as defined in (2.11), neglecting the  $O(\tau^2)$ -term does not require any additional regularity, see for example (2.25). As we are considering time approximations, the main challenge is to replace the integrand

$$e^{-s\mathcal{L}}p(e^{s\mathcal{L}}u_0) \tag{1.7}$$

by an approximation which can be integrated explicitly on  $[0, \tau]$ . Classical methods are based upon Taylor expanding in time the exponentials appearing in the above equation (1.7) as follows:  $e^{s\mathcal{L}}u = 1 + O(s\mathcal{L}u)$ . Hence, they are based upon linear frequency approximations. We give the example of a first order exponential integrator.

**Example 3 (The Euler exponential method)** A first order exponential integrator consists in Taylor expanding the exponential appearing inside the nonlinearity :

$$e^{-s\mathcal{L}}p(e^{s\mathcal{L}}u_0) \approx e^{-s\mathcal{L}}p(u_0).$$
(1.8)

The scheme after one time-step, which expresses  $u^{n+1}$  in terms of  $u^n$ , is then obtained by integrating exactly the remaining term  $e^{(\tau-s)\mathcal{L}}$ :

$$u^{n+1} = e^{\tau \mathcal{L}} u^n + \tau \varphi_1(\tau \mathcal{L}) p(u^n) \quad \text{with} \quad \varphi_1(z) = \frac{e^z - 1}{z}.$$
(1.9)

It remains to quantify the local error induced by the approximation (1.8). For the NLS equation (1.2), following the reference [HO10], the local error has the following explicit form:

$$|u|^{4}u - 2|u|^{2}\Delta u + u^{2}\overline{\Delta u}.$$
(1.10)

First order convergence of this method thus requires two additional derivatives on the solution, and more specifically requires  $H^2$ -solutions when measuring the error in  $L^2$ . If the solution is however not sufficiently smooth, the local error becomes unbounded and the exponential integrator fails to converge at first order, see Figure 1.1.

A new discretization technique was introduced in the works [HS17b, OS18] in the case of the KdV equation (1.3) and the NLS equation (1.2). The idea is based on a new nonlinear approach, and a refined analysis made in Fourier space. Instead of linearizing the frequency interactions in the central oscillatory term (1.7), the nonlinear frequency interactions between the linear flow  $e^{-s\mathcal{L}}$  and the nonlinear factor  $p(e^{s\mathcal{L}}u_0)$  are incorporated into the numerical discretization. This allows for the convergence of novel time discretizations under lower regularity assumptions on the solution.

In the following section we illustrate the new discretization technique of [OS18] for the NLS equation (1.2), and compare it with classical approaches.

### 1.1.1 Illustration on the nonlinear Schrödinger equation

We propose in this section a method of construction introduced in [AB23b] for the NLS equation (1.2), which offers a framework for the derivation of different classes of schemes: the previous scheme of [OS18] built at low-regularity, newly designed ones presented in this manuscript, and classical integrators. We recall that  $p(u) = -i|u|^2 u$  and let  $t_n = n\tau$ , with  $\tau$  the time step.

We start by making a change of variables. Instead of working with the equation (1.2) in u, we twist the variables with respect to the leading differential operator  $\mathcal{L} = i\Delta$ :

$$e^{it\Delta}v = u,\tag{1.11}$$

and work with the equation in v:

$$\partial_t v = e^{-it\Delta} p(e^{it\Delta} v), \quad v(0) = u_0, \tag{1.12}$$

as well as its integral form (obtained by integrating on  $[t_n, t_n + s]$ ):

$$v(t_n + s) = v(t_n) + e^{-it_n\Delta} \int_0^s e^{-is_1\Delta} p(e^{i(t_n + s_1)\Delta} v(t_n + s_1)) ds_1.$$
(1.13)

This change of variables is a well-known approach in the theoretical analysis of dispersive equations in low regularity spaces [Bou93], in rough path theory [Gub12], as well as in numerical analysis for constructing Lawson methods [Law67, HLO20] and resonance-based schemes [HS17b, OS18].

Remark 1.1.1 (The advantage of twisting the variable) The equation (1.2) for u is driven by the differential operator  $i\Delta$ , an unbounded operator on  $L^2$ . This is not the case for the equation (1.12) in v, which is written in terms of the nonlinearity and the free Schrödinger group, a bounded operator in  $L^2$ . The numerical advantage thus lies in the fact that the approximation

$$v(t_n + s) \approx v(t_n), \quad |s| \le \tau,$$

$$(1.14)$$

does not require additional regularity on the solution v. Indeed, it follows from equation (1.13) that for  $\sigma > d/2$ 

$$\|v(t_n+s)-v(t_n)\|_{H^{\sigma}} \le \int_0^s \|p(e^{i(t_n+s_1)\Delta}v(t_n+s_1))\|_{H^{\sigma}} ds_1 \le \left(\sup_{0\le s_1\le s} \|v(t_n+s_1)\|_{H^{\sigma}}^3\right) s.$$

The same analysis would yield in the untwisted variable the approximation  $u(t_n + s) \approx e^{is\Delta}u(t_n)$ ,  $|s| \leq \tau$ . Obtaining the approximation  $u(t_n + s) \approx u(t_n)$  requires Taylor expanding at first order the linear flow  $e^{is\Delta}$ , and hence two additional derivative on the solution.

Next, we map Duhamel's formula (1.13) in Fourier space, with  $s = \tau$  the time step. Expanding v in Fourier series  $v(t,x) = \sum_{k \in \mathbb{Z}} v_k(t)e^{ikx}$ , allows to express the action of the linear Schrödinger flow on v,  $e^{\pm it\Delta}v(t,x) = \sum_{k \in \mathbb{Z}} v_k(t)e^{\mp itk^2}e^{ikx}$ . We note that the upcoming derivation can analogously be obtained in higher dimensions with  $\mathbf{k} = (k^1, \dots, k^d) \in \mathbb{Z}^d$ . The k-th Fourier coefficient of  $v(t_{n+1})$  reads

$$v_k(t_{n+1}) = v_k(t_n) - i \sum_{k=-k_1+k_2+k_3} e^{it_n R(k)} \int_0^\tau e^{isR(k)} \overline{v}_{k_1}(t_n+s) v_{k_2}(t_n+s) v_{k_3}(t_n+s) ds,$$
(1.15)

where we collected all the frequency interactions into the resonance structure R(k):

$$R(k) = k^{2} + k_{1}^{2} - k_{2}^{2} - k_{3}^{2} = 2k_{1}^{2} - 2k_{1}(k_{2} + k_{3}) + 2k_{2}k_{3}, \qquad k = -k_{1} + k_{2} + k_{3}.$$
(1.16)

The goal is to approximate in time the integrand in equation (1.15). The idea behind the novel nonlinear approach in [OS18], is to base the approximation on the specific form of the resonance structure (1.16). We illustrate this approach with the general decomposition presented in [AB23b], which improves on the scheme of [OS18]. Namely, we decompose the integrand in (1.15) into three factors:

$$I_{\tau} = \int_0^{\tau} e^{isR_{\text{dom}}(k)} e^{isR_{\text{low}}(k)} g(s) ds$$
(1.17)

with

$$R(k) = R_{\text{dom}}(k) + R_{\text{low}}(k), \qquad g(s) = \overline{v}_{k_1}(t_n + s)v_{k_2}(t_n + s)v_{k_3}(t_n + s), \tag{1.18}$$

$$R_{\text{dom}} = 2k_1^2, \ R_{\text{low}} = -2k_1(k_2 + k_3) + 2k_2k_3.$$

The choice of the above decomposition (1.17) is dictated by the regularity assumptions needed to approximate each of the factors. Using Duhamel's formula (1.13) we can approximate g(s) at its left end point:  $g(s) \approx g(0)$  with a local error in O(s) without additional regularity requirements, see equation (1.14). It remains to treat the oscillatory term:

$$e^{isR_{\text{dom}}(k)}e^{isR_{\text{low}}(k)}g(0), \tag{1.19}$$

which corresponds to the term (1.7), when mapped in Fourier space. A second approximation step is necessary to obtain a practical implementation of the scheme, see Remark (1.1.2). We observe that approximating the factor  $e^{isR_{\text{low}}(k)}$  asks for *less* regularity than approximating  $e^{isR_{\text{dom}}(k)}$ . Indeed, the terms  $k_lk_j$  (for  $l \neq j$ ) appearing in  $R_{\text{low}}(k)$  correspond to products of first order derivatives. Hence, by denoting  $v = v(t_n)$ , a formal first-order Taylor expansion yields

$$e^{isR_{\text{low}}} = 1 + O(sR_{\text{low}}), \quad \text{with} \quad R_{\text{low}}g(0) \sim -2(|\nabla v|^2\bar{v} + 2v\nabla v \cdot \nabla\bar{v}), \tag{1.20}$$

after mapping back from frequency to spatial variables. If we tried to approximate the dominant factor  $e^{isR_{dom}(k)}$ , as  $2k_1^2$  corresponds to second order derivatives in Fourier space, we would require more regularity:

$$e^{isR_{\text{dom}}} = 1 + O(sR_{\text{dom}}), \text{ with } R_{\text{dom}}g(0) \sim 2v^2 \Delta \bar{v}.$$
 (1.21)

Therefore, we choose to Taylor expand the lower term (1.20), making the approximation  $R(k) \approx R_{\text{dom}}(k)$ , and *integrate* exactly the dominant term:

$$I_{\tau} \approx \int_{0}^{\tau} e^{i s R_{\text{dom}}(k)} ds \, g(0) = \frac{e^{2i\tau k_{1}^{2}} - 1}{2ik_{1}^{2}} g(0),$$

in order to obtain a scheme whose local error requires the least regularity. We finally map back in physical space, and in the untwisted variables. We thereby obtain the so-called first-order *resonance-based* integrator

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau e^{i\tau\Delta} \left( (u^n)^2 \varphi_1(-2i\tau\Delta)\overline{u}^n \right), \quad \varphi_1(z) = \frac{e^z - 1}{z}.$$
(1.22)

The local error of the scheme is governed by the approximation made in (1.20) and is of the form:

$$-2(|\nabla v|^2\bar{v} + 2v\nabla v \cdot \nabla \bar{v}), \tag{1.23}$$

which only requires the boundedness of one, instead of two, additional derivatives on the solution. For a discussion on the practical implementation of the above scheme we again refer to Remark 1.1.2, and for the terminology *resonance-based* integrator to Remark 1.1.3.

Remark 1.1.2 (An interplay between efficient implementation and proximity to the solution) Ideally, once

the first approximation step  $g(s) \approx g(0)$  is performed, one would hope to integrate the remaining oscillations (1.19) exactly in time (on  $[0, \tau]$ ). While this can be done analytically:

$$\int_{0}^{\tau} e^{iR(k)s} ds = \frac{e^{i\tau R(k)} - 1}{iR(k)}, \quad R(k) = 2k_1^2 - 2k_1(k_2 + k_3) + 2k_2k_3, \tag{1.24}$$

it does not yield a practical implementation, as it results in a generalized convolution (of Coifman-Meyer type [CM75]) which cannot be mapped back to physical space, due to the factor  $(2k_1^2 - 2k_1(k_2 + k_3) + 2k_2k_3)^{-1}\bar{v}_{k_1}v_{k_2}v_{k_3}$ . Hence, the computations would need to be fully made in Fourier space. Carrying this out in higher spatial dimensions d would cause large memory and computational efforts of order  $O(|K|^{d\cdot\ell})$ , where K denotes the highest frequency in the discretization and  $\ell$  is the number of factors in the nonlinearity. For practical computations, we want to be able to express the discretization also in physical space in order to use the Fast Fourier Transform (FFT) which is of computational effort of order  $O(|K|^{d}\log|K|^d)$ . Therefore, we choose an approximation of the integrand (1.19) which allows for a practical implementation (by not performing exact integration), while optimizing the local error to require the least regularity. The latter dictates the choice of the decomposition (1.18). If one was not concerned with the lack of regularity of the solution, but for example the conservation of a certain quantity of interest, then the choice of the frequency decomposition (1.18) could be taken differently.

**Remark 1.1.3 (The scheme entitled "resonance-based integrator")** The terminology *resonance-based* integrator comes from the fact that the scheme is based upon the resonance structure (1.16) of the underlying PDE. Another name used is *Fourier* integrator. Further on, we will discuss extensions to these results which are suited on more general domains. These integrators will then be termed *low regularity* integrators.

We would like to address a common misconception. The construction of these resonance-based integrators are not based on the resonant case R(k) = 0. From a numerical point of view, the resonant case R(k) = 0 is trivial given that there are no oscillations to approximate; the discretization of Duhamel's integral would simply amount to  $\tau p(u^n)$ . However, the case  $R(k) \neq 0$  is more subtle as one needs to capture the oscillations  $e^{isR(k)}$  also in the numerical approximation. We note that interestingly, the difficulties are reversed when working in the theoretical analysis of dispersive equations, since in the resonant case R(k) = 0 we do not gain the decay  $\frac{1}{R(k)}$  in the frequencies when integrating (see equation (1.24)).

The above scheme (1.22) was first obtained by [OS18], using slightly different steps. The authors of [OS18] do not pass directly in Fourier space – as in (1.15) – and first approximate  $v(t_n + s_1) \approx v(t_n)$  in Duhamel's integral (1.13), as this does not require additional regularity (see (1.14)). However, making this first approximation ( $v(t_n + s) \approx v(t_n)$  or in Fourier space  $g(s) \approx g(0)$ ) breaks the symmetric structure of the underlying PDE (1.2) and leads to loss of structure preservation. We will see in Section 1.6 that to obtain schemes which have good structure-preserving properties we needs to make a different approximation to  $v(t_n + s)$  (or in Fourier g(s)) and should consider the general form (1.17). This approach was first taken in [AB23b].

We refer to Figure 1.1a where first order convergence of the resonance-based scheme (1.22) in  $L^2$  norm and for  $H^1$  solutions is observed in numerical experiments. In Section 1.2.1 rigorous convergence results of the scheme are discussed. We next compare this resonance-based discretization with classical time integrators.

Comparison with classical approximation techniques. The Euler exponential method corresponds to making the approximation  $R(k) \approx k^2$  recovering the scheme (1.9) with local error (1.10).

Another quite popular method of approximation are splitting methods ([MQ02, Fao12, BBD02, Lub08]). We recall that splitting methods consists in separating the linear  $\mathcal{L} = i\Delta$  and nonlinear part  $p(u) = -i|u|^2 u$  and composing the exact flows of the differential equations:

$$\partial_t u = i\Delta u, \quad \text{and} \quad \partial_t u = -i|u|^2 u.$$
 (1.25)

The first-order Lie splitting scheme is then given by:

$$u^{n+1/2} = e^{-i\tau |u^n|^2} u^n,$$
  

$$u^{n+1} = e^{i\tau\Delta} u^{n+1/2},$$
(1.26)

with the explicit local error

$$2(2|\nabla u|^2 u + \bar{u}(\nabla u \cdot \nabla u) + u^2 \Delta \bar{u}),$$

see for example [Lub08, Section 4.2]. Because of the last term  $u^2 \Delta \bar{u}$  in the above formula, the Lie splitting method requires two additional derivatives on the solution, as does the Euler exponential method.



D) II uata.

Figure 1.1 – Convergence plots taken from [AB23b], for classical integrators (green, pink, purple) versus integrators suited for low-regularity solutions (light blue, dark blue, yellow). The Euler exponential method (1.9) in purple, and the first order Lie splitting (1.26) in pink do not converge at first order in Plot 1.1a. The second order Strang splitting in green oscillates erratically in both plots, due to the lack of regularity. The first order resonance-based scheme in light blue (Osterman & Schratz o1) is given by (1.22) and was derived in [OS18]. Its extension to second order (Res based o2, dark blue) is derived for instance in [BS22]. The symmetric scheme in yellow (Alama Bronsard) is given by (1.49) and introduced in Section 1.6.1.

To understand the difference behind this splitting approximation 1.26 and the resonance-based approach (1.17), we formally Taylor expand the nonlinear part in (1.26):  $e^{-i\tau|u|^2} = 1 - i\tau|u|^2 + O(\tau^2)$ . This does not require additional regularity assumptions since the error term in  $O(\tau^2)$  only involves a polynomial of u. The local error analysis is then dominated by the remainder

$$u(t_{n+1}) - \left(e^{i\tau\Delta}u(t_n) - i\tau e^{i\tau\Delta} (|u(t_n)|^2 u(t_n))\right).$$
(1.27)

In the frequency variables, the above corresponds to taking  $R(k) \approx 0$  (and  $g(s) \approx g(0)$ ), thereby neglecting all the oscillations in (1.19). Equivalently, by recalling Duhamel's integral (1.6) written in the spatial variables, the splitting method corresponds to the approximation

$$e^{-s\mathcal{L}}p(e^{s\mathcal{L}}u_0) \approx p(u_0),\tag{1.28}$$

which treats the problem as a linear one. Such linearized frequency approximations are computational very convenient,

since the oscillations in (1.28) no longer appear in the approximation. However, the method does not resolve the nonlinear frequency interactions of the system and treats the resolution of the linear  $i\Delta$  and the nonlinear term p(u) independently.

In conclusion, these classical approximation techniques do not take into account the interactions between the frequencies from the linear and nonlinear term. The idea behind the resonance-based schemes is to do so, while maintaining a scheme that can be efficiently implemented at the same computational cost as classical schemes. In the course of this manuscript, we will build on this nonlinear approach to obtain both novel low-regularity approximations and better structure preserving schemes.

The next section is devoted to a discussion on prior rigorous convergence results and on the spaces in which they are obtained, both for time integrators as well as finite difference methods in time.

### **1.2** Explicit convergence rates

An important part of this manuscript (i.e. Chapters 2 and 5) deals with obtaining explicit convergence rates of order  $\tau^{\nu}$ , which are quantified as a function of the Sobolev regularity of the solution,  $u \in C([0,T], H^{\alpha})$ , and the Sobolev norm  $H^{\sigma}$  in which we measure the error. We let  $\nu > 0$  and  $\alpha \ge \sigma \ge 0$  and consider the error:

$$\|u(t_n) - u^n\|_{H^{\sigma}} \le C_T \left( \sup_{[0,T]} \|u(t)\|_{H^{\alpha}} \right) \tau^{\nu}, \quad 0 \le t_n = n\tau \le T.$$
(1.29)

Throughout this thesis we denote  $C_T(\cdot)$  a generic constant which depends continuously on its arguments, and exponentially on the final time T. These error analysis results are thereby *local in time*, see Remark 1.6.2. Unless stated otherwise we set ourselves on  $\mathbb{T}^d$ , with  $d \in \{1, 2, 3\}$ . We present two different perspectives when exploring the interplay between  $\nu$ ,  $\alpha$ and  $\sigma$ :

- 1. Full order of convergence. We fix the desired order of convergence  $\nu = m \in \mathbb{N}$ , and search for the optimal regularity  $\alpha = \alpha(d, \sigma, \nu)$  which is needed for the convergence of the method to order  $\nu$ . The required regularity  $\alpha(d, \sigma, \nu)$  depends on the spatial dimension d, the norm  $H^{\sigma}$ , and the order of convergence  $\nu$ . This approach is the most common one in the literature, and Chapter 2 presents results of this type.
- 2. Fractional order of convergence. We fix the  $H^{\sigma}$ -norm, and assume the regularity  $u \in C([0, T], H^{\alpha})$ . One cannot hope to obtain full order of convergence when the boundedness of the local error term requires higher regularity assumptions than what is assumed on the solution u(t). Fractional convergence rates are then expected, with the rate as a function of the regularity  $\alpha$ . This approach is more general, since we recover the full-order convergence result, when enough regularity is assumed. The results of Chapter 5 go in this direction.

The choice of the regularity parameters  $\alpha$  and  $\sigma$  strongly influences the complexity of the analysis to obtain a convergence result of type (1.29). We distinguish three categories, from easiest to most challenging :

- Case  $\sigma > d/2$ . We need  $\alpha \geq \sigma$  and hence also  $\alpha > d/2$ . In this case we have that  $H^{\sigma} \hookrightarrow L^{\infty}$ , and by using the Kato-Ponce estimate it follows that  $H^{\sigma}$  forms an algebra:  $||uv||_{H^{\sigma}} \leq C_{\sigma} ||u||_{H^{\sigma}} ||v||_{H^{\sigma}}$ . Both the local error and stability analysis follow easily for a polynomial nonlinearity, and applying Lady Windermere's fan argument allows to conclude global convergence, see Chapter 2.
- Case  $0 \le \sigma \le d/2$  and  $\alpha > d/2$ . We have  $u(t) \in L^{\infty}$ , however the space  $H^{\sigma}$  does not form an algebra, and hence one needs refined bilinear estimates (see 2.10, 5.14, 5.15). In particular, for the stability of the scheme the following estimate is crucial for bounding the nonlinear terms:

$$||wz||_{H^{\sigma}} \le C_{\sigma,d} ||w||_{\frac{d}{d}+\epsilon} ||z||_{H^{\sigma}}, \quad 0 \le \sigma \le d/2, \quad \epsilon > 0.$$

$$(1.30)$$

Using a refined global error analysis, one can still push the error analysis down to  $H^{\sigma}$  through a Lady-Windermere's fan argument, by first proving fractional convergence of the scheme in a suitable higher order Sobolev space. This is presented in Chapters 2 and 5.

- Case  $0 \le \sigma \le d/2$  and  $0 < \alpha \le d/2$ . Given that an estimate of the form (1.30) is out of reach in this rough setting, we cannot obtain the global convergence by combining a local error and stability analysis through Lady Windermere's fan argument. As in the theoretical analysis of PDEs at low regularity, in order to consider very rough solutions  $u(t) \in H^{\alpha}$  with  $\alpha \le d/2$ , one needs to call upon more refined tools such as Strichartz estimates and Bourgain spaces [Bou93, BGT04, Tao06].

The above cases are particularly relevant when the nonlinearity is a polynomial in u and  $\bar{u}$ , such as for the NLS equation (1.2), which is discussed next.

In the following Sections 1.2.1 and 1.2.2, we discuss the state of the art regarding convergence results for the NLS and KdV equations, before introducing new research directions in Section 1.3, which are explored in the rest of the manuscript.

#### 1.2.1 Comparing convergence results for the nonlinear Schrödinger equation

Resonance-based schemes. We expect resonance-based schemes to have the following convergence rates:

Convergence in  $H^{\sigma}$  at order  $\tau^{\nu}$  for  $u_0 \in H^{\alpha}$ , with  $\alpha = \sigma + \nu$  and  $\sigma > d/2$ .

We refer to [OS18] in the case  $0 < \nu \le 1$ , and [OWY22] in the case  $\nu = 2$ . In these results the analysis is restrained to the torus using Fourier series and the frequency decomposition (1.19).

In the case  $0 \le \sigma \le d/2$  more regularity assumptions are needed and we refer to the work [AB23a] presented in Chapter 2, which treats the cases  $\sigma \ge 0$ ,  $\nu = 1, 2$  and  $\alpha > d/2$ .

Finally, the authors of [ORS22b] introduce discrete Bourgain spaces allowing them to consider the rough case of  $0 < \alpha \leq 1/2$  on the one-dimensional torus  $\mathbb{T}$ . They show fractional convergence in  $L^2$  of order  $\tau^{\nu}$  for  $u_0 \in H^{\alpha}(\mathbb{T})$  with  $0 < \alpha \leq 1$  and  $\nu < 1$ .

When set on the one-dimensional torus  $\mathbb{T}$ , several recent results build on the approach of making a decomposition in Fourier space and use the resonance structure (1.16), either for coupling the time integrator with a spectral spatial discretization, or for designing schemes under even lower regularity assumptions, see [WY22, LW21, OY22, CLL23].

Splitting and exponential integrators. In general, the order of convergence of these integrators behaves as follows:

Convergence in  $H^{\sigma}$  of order  $\tau^{\nu}$  for  $u_0 \in H^{\alpha}$ , with  $\alpha = \sigma + 2\nu$  and  $\sigma \ge 0$ .

In the case  $u_0 \in H^{\alpha}$  with  $\alpha > d/2$ , we refer to the works of [BBD02, Lub08, Gau11, Fao12, Tha12, ESS16] for convergence of Lie and Strang splitting methods, and to [Duj09, HO10, HLO20] for exponential integrators. In the more subtle case  $\alpha \leq d/2$ , we refer to [ORS22a] and [JO23], which show fractional rates up to first order with  $\sigma = 0$ ,  $\nu = \frac{\alpha}{2}$ , and  $0 < \alpha \leq 2$ , of a filtered version of the Lie splitting method.

*Finite difference approximations in time.* For an overview on these methods (with a comparison to splitting techniques) we refer to [ABB13]. The convergence of these methods requires at least as much regularity as splitting or exponential methods. For example, second-order convergence of a leap-frog and relaxation type finite difference schemes are obtained in [SS84] and [BDDLV21], respectively. They require at least four additional derivatives on the solution, as is the case of exponential and splitting methods.

Type	Reference	Domain	Order $\nu$	Error norm $\sigma$	Relation $\frac{\alpha - \sigma}{\nu}$
	[OS18]	$\mathbb{T}^d$	$\in (0,1]$	> d/2	1
Resonance based	[OWY22]	$\mathbb{T}^d$	2	> d/2	1
and low-regularity	Chapter 2	$\Omega \subset \mathbb{R}^d$	1 & 2	> d/2	1
schemes	Chapter 2	$\Omega \subset \mathbb{R}^d$	1 & 2	$\in [0, d/2)$	$1 + (d/2 - \sigma)/2\nu$
Splitting or	[BBD02]	$\mathbb{R}^2$	1 & 2	0	2
Exponential	[Lub08]	$\mathbb{R}^3$	2	0	2
Integrators	[Tha12], etc.	$\mathbb{R}^{d}$	$\in \mathbb{N}$	0	2
Finite differences	[BDDLV21]	$\mathbb{R}^{d}$	2	> d/2	2

Table 1.1 – Convergence results of the form (1.29) for the NLS equation (1.2). The interplay between the error norm  $\sigma$ , the convergence rate  $\gamma$ , and the regularity assumption  $\alpha$ , is expressed by the value of  $\frac{\alpha - \sigma}{\nu}$ , that is, by the additional regularity requirement per order of convergence.

To give some perspective, we take a step back from the NLS equation and compare the regularity requirements for the approximation to the KdV equation (1.3), which is posed in dimension d = 1.

#### **1.2.2** Comparing convergence results for the Korteweg-de Vries equation

Two effects are present in the KdV equation: Burger's nonlinearity  $\partial_x(u^2)$ , which set alone forms shocks in finite time, and Airy's dispersive term  $\partial_x^3$ , whose linear evolution  $\partial_t u + \partial_x^3 u = 0$  preserves all Sobolev norms. In the error analysis, the challenge lays in the stability bound due to the derivative appearing in the nonlinearity. Indeed, given that the nonlinearity is not locally Lipschitz, the bilinear estimates of the form (1.30) do not suffice to obtain stability.

Type	Reference	Order $\nu$	Error norm $\sigma$	Relation $\frac{\alpha-\sigma}{\nu}$
Resonance	[HS17b]	1	1	2
based	[WZ21]	2	$\geq 0$	2
schemes or	[WZ22b]	1	> 1/2	1
Fourier	[WZ22b]	2	$\geq 0$	3/2
integrators	[LW22]	$\in (0,1]$	0	1
Splitting or	[RS22]	$\in [0,1]$	0	3
Exponential	[HKRT11]	2	$\geq 8$	9/2
Integrators	[HLR13]	2	$\geq 1$	5/2
Fully discrete	[CLR20]	$\in [1/14, 1]$	0	$6/\min(1, \nu + 1/2)$
schemes	[OS20]	1	0	3

Table 1.2 – Convergence results of the form (1.29) for the KdV equation (1.3).

Resonance-based schemes: For the KdV equation, mapping Duhamel's iterate in Fourier space as in (1.15), we obtain

$$R(k) = k_1^3 + k_2^3 - k^3 = -3kk_1k_2, \quad k = k_1 + k_2.$$
(1.31)

The first resonance-based scheme was introduced for the KdV equation in the work [HS17b], where the design of first and second order schemes is presented. A convergence analysis using energy estimates allows to show first-order convergence in  $H^1$  for solutions in  $H^3$ , thereby requiring two additional derivatives. Constructing the first-order resonance-based scheme for the KdV equation is simpler than for the NLS equation, as one can integrate exactly all of the resonance structure R(k) into the numerical discretisation, as opposed to the NLS equation, see Remark 1.1.2. On the other hand, the stability analysis, and hence the global error analysis, is more intricate. The authors of [WZ21] offer a convergence result for the second-order resonance based scheme of [HS17b], in the case  $\nu = 2$ ,  $\alpha = \sigma + 4$ , and  $\sigma \ge 0$ .

A further improvement on these results is given in [WZ22b] which introduces new first and second order integrators converging in the case  $\nu = 1$ ,  $\alpha = \sigma + 1$ ,  $\sigma > 1/2$  and  $\nu = 2$ ,  $\alpha = \sigma + 3$ ,  $\sigma \ge 0$ , respectively. A first order (unfiltered) integrator requiring even less regularity is then analyzed in [LW22] with  $\nu = \alpha$ ,  $0 < \alpha \le 1$ ,  $\sigma = 0$ .

Splitting and exponential integrators: For filtered versions of the Lie splitting and Euler exponential methods the authors of [RS22] shows the following fractional convergence result:

Convergence in 
$$H^{\sigma}$$
 of order  $\tau^{\nu}$ , for  $u_0 \in H^{\alpha}$ , with  $\alpha = \sigma + 3\nu$  and  $\nu \leq 1$ .

For an analysis at second order of the Strang splitting method we cite [HKRT11] and [HLR13], which show convergence in the case  $\nu = 2$ ,  $\alpha = \sigma + 5$ ,  $\sigma \ge 1$ , under the assumption that the nonlinear part, i.e., Burgers equation  $\partial_t u = -\frac{1}{2}\partial_x u^2$ , is solved exactly. Applying the same analysis on the Lie splitting method would yield first-order convergence ( $\nu = 1$ ) for  $\alpha = \sigma + 3$ , coinciding with the result of [RS22].

Fully-discrete schemes when coupled with a finite difference method in space. We refer to [CLR20], where a fully discrete convergence analysis of a finite difference (FD) scheme for the KdV equation with explicit rates up to first order is introduces for the first time. The authors discretize this equation with a Rusanov scheme for the nonlinear part, which corresponds to introducing a linear and explicit diffusion effect. This allows them to handle the derivative in the Burger's nonlinearity and to show first-order convergence in  $L^2(\mathbb{R})$  for  $u_0 \in H^6(\mathbb{R})$ , together with fractional convergence rates up to first order in the case  $\nu \leq 1$ ,  $3/4 \leq \alpha \leq 6, \sigma = 0$  (when the dispersive term  $\partial_x^3$  is treated in an implicit manner). They work on the real line  $\mathbb{R}$  and obtain convergence rates for non-smooth initial data (down to  $\alpha = 3/4$ ) by making use of the dispersive smoothing effects of the underlying equation on  $\mathbb{R}$ .

We also mention the work of [OS20] which couples the space-discretization used in [CLR20], based on the Rusanov correction for the Burgers nonlinearity, with an exponential integrator in time. They show first order convergence of the explicit scheme in  $L^2$  for  $u_0 \in H^3$ , and with the CFL condition  $\tau = O(h)$ . We notice that coupling the FD Rusanov discretization in space with an exponential integrator in time ([OS20]) requires less regularity than if a FD time approximation is considered ([CLR20]).

# 1.3 Branching out in new directions

Following this introduction to resonance-based integrators, many questions remain.

- 1. General domains. The construction of these resonance-based schemes entirely relies on decomposing the solution on its Fourier basis, and thereby on imposing periodic boundary conditions. A first natural question is whether one can generalize the construction of these schemes from the torus  $\mathbb{T}^d$  to more general domains, such as smooth bounded domains  $\Omega \subset \mathbb{R}^d$  with homogeneous Dirichlet boundary conditions.
- 2. **Higher order.** Up to now we have only constructed the first order resonance-based scheme (1.22), and discussed results of order at most two. A next question is whether we can push the design up to (arbitrarily) high order.
- 3. Unifying approach. These resonance-based schemes are built on the resonance structure of the underlying equation, see (1.16) for NLS and (1.31) for KdV. Can one build an algorithmic procedure to construct these schemes for a class of equations, such as for the class (1.1), without the need of analyzing one equation at a time?
- 4. Structure preservation. Finally, we are interested in obtaining schemes which exhibit good long-time behavior, preserving up to a certain extent the conserved quantities of the equation on the discrete level. The resonance-based scheme (1.22) unfortunately does not have good *structure-preserving* properties. Is it possible to construct schemes at low-regularity which also preserves the geometric structure of the equation?

This manuscript is divided into three parts, that address the above topics.

Part I answers questions regarding topics 1, 2 and 3, by offering in Chapter 3, an algorithmic way to construct higher-order schemes to a class of nonlinear evolution equations (with potentials) set on general domains. This is based on the article [ABBS22b]. This latter work generalizes the construction of the schemes made in the article [AB23a] presented in Chapter 2, which designs and analyses schemes for the Gross-Pitaevskii equation in low-regularity regimes.

*Part II* pursues in the direction of topics 2 and 3 in a randomized setting, inspired by the works in probabilistic well-posedness and in Wave Turbulence theory. The single work [ABBS22a] is presented in Chapter 4.

Part III focusses on topic 4 and introduces schemes which are advantageous both in non-smooth and smooth settings. We first introduce in Chapter 5 the work [AB23b] which presents a novel time integrator for the NLS equation (1.2). The integrator is symmetric, provably converges at low-regularity, and exhibits in numerical experiments remarkable long-time near-preservation of the mass and energy, without any CFL type conditions. This chapter also adresses topic 1. We then present the work [ABBMS23] which combines the topics 2, 3, and 4, by generalizing the latter work to the class of dispersive equations (1.1) up to high orders. In numerical simulations, we witness better structure preserving properties of these new symmetric integrators compared to classical splitting schemes, even in the smooth setting of  $C^{\infty}$ -solutions.

We begin with the state of the art regarding Part I of this manuscript, and more specifically topic 1.

# 1.4 A brief introduction to Part I

#### 1.4.1 Beyond periodic boundary conditions

Our starting point is the work [RS21], which overcomes the necessity of Fourier-based expansions, and hence of periodic boundary conditions, by introducing new filtering techniques and a functional setting based on semi-group theory. The authors introduce a framework allowing for the construction and analysis of first and second order *low-regularity integrators* on more general domains and to the following class of nonlinear evolution equations:

$$\begin{cases} \partial_t u - \mathcal{L}u = f(u, \overline{u}), & (t, x) \in \mathbb{R} \times \Omega, \\ u(0, x) = u_0(x), & x \in \Omega, \end{cases}$$
(1.32)

with  $\Omega$  either equal to the full space  $\mathbb{R}^d$ , the torus  $\mathbb{T}^d$ , or a smooth bounded domain  $\Omega \subset \mathbb{R}^d$ . In the case of a smooth bounded domain, the problem is equipped with homogeneous boundary conditions, which are encoded in the choice of the domain of the operator  $\mathcal{L}$ . The main assumptions on  $\mathcal{L}$  are the following:

- i)  $\mathcal{L}$  is a linear operator defined on a Hilbert space X of complex valued functions  $u \in \mathbb{C}$ ;
- ii)  $\mathcal{L}$  generates a strongly continuous semigroup  $\{e^{t\mathcal{L}}\}_{t\geq 0}$  of contractions on X;
- iii)  $\mathcal{A} = -\mathcal{L} + \overline{\mathcal{L}}$  generates a group  $\{e^{t\mathcal{A}}\}_{t\in\mathbb{R}}$  of unitary operators on X;
- iv)  $\mathcal{L}$  and  $\overline{\mathcal{L}}$  commute:  $[\mathcal{L}, \overline{\mathcal{L}}] = 0$ .

**Example 4** Admissible operators  $\mathcal{L}$  include the heat operator  $\Delta$ , the Schrödinger operator  $i\Delta$ , the half wave operator  $i|\nabla|$ , and the Klein-Gordon and wave type operators  $i\sqrt{-\Delta+m^2}$ , with mass  $m \geq 0$ .

In addition, it is assumed that the nonlinearity f is tensorized under the form

$$f(v,w) = \mathcal{B}(F(v) \cdot G(w)), \quad F, G : \mathbb{C} \to \mathbb{C}^J,$$
(1.33)

where  $\mathcal{B}$  is a linear operator. It is also assumed that f is locally Lipschitz and maps a ball into itself, as defined for example in equations (5.18).

The class of evolution equations (1.32) encompasses the class of dispersive equations (1.1). Indeed, with the above framework, the authors of [RS21] can deal in a unified way with parabolic, hyperbolic, dispersive as well as mixed type equations. However, an important class of equations is missing, namely equations with a *potential*. The goal of the work [AB23a] presented in Chapter 2, is to close this gap and propose a novel low regularity integrator for the Gross Pitaevskii (GP) equation,

$$i\partial_t u(t,x) = -\Delta u(t,x) + V(x)u(t,x) + |u(t,x)|^2 u(t,x), \quad (t,x) \in \mathbb{R} \times \Omega,$$
(1.34)

with a non-smooth potential V(x). In this work first and second order integrators are constructed and are proven to converge at integer order  $\nu = m \in \{1, 2\}$ , under low-regularity assumptions both on the solution u(t) and on the potential V. We denote by  $u^n = \Phi_m^{\tau}(u^{n-1})$  the low-regularity integrators of order m, given at equation (2.5) and (2.7) respectively. Set on the d-dimensional torus  $\mathbb{T}^d$ , the convergence result reads as follows:

**Theorem 1.4.1** (*Theorems 2.1.1 and 2.1.2*) Let  $d \in \{1, 2, 3\}$ ,  $\sigma \ge 0$ , T > 0,  $m \in \{1, 2\}$  and

$$\alpha := \begin{cases} m + \sigma, & \text{if } \sigma > \frac{d}{2}, \\ m + \frac{\sigma}{2} + \frac{d}{4}, & \text{if } 0 \le \sigma < \frac{d}{2}. \end{cases} \qquad \qquad m + \frac{d}{2}$$

For every  $u_0 \in H^{\alpha}$  and  $V \in H^{\alpha}$ , there exists  $\tau_{\min} > 0$  and  $C_T > 0$  such that for every  $\tau \leq \tau_{\min}$ .

 $||u(t_n) - u^n||_{H^{\sigma}} \le C_T \tau^m, \qquad 0 \le t_n = n\tau \le T.$ 

$$\leq \tau_{\min} \colon \begin{array}{c} m + \frac{d}{2} \\ m + \frac{d}{4} \\ 0 \\ d/2 \end{array} \sigma$$

 $\alpha_{\blacktriangle}$ 

For simplicity, the error analysis was made on the torus  $\mathbb{T}^d$ . Nevertheless, the tools involved are inspired by the filtering techniques of [RS21], which do not use Fourier-based expansions and hence can be extended to more general domains. We refer to the work [AB23b], presented in Part III of this manuscript, which introduces a functional framework yielding convergence result both on the torus and on smooth bounded domains with homogeneous Dirichlet boundary conditions. Finally, for an explanation on the filtering techniques used, we refer to equation (5.31) and Section 2.3.2.

The work of Chapter 2 motivated the second result of this manuscript, presented in Chapter 3, which generalizes the construction of low-regularity schemes *up to arbitrary order* and for a class of nonlinear evolution equations with potentials.

In the next section we discuss topics 2 and 3, and present the results of Chapter 3.

#### **1.4.2** Higher order approximations

In order to obtain more precise approximations, which are also computationally efficient, one needs to construct higher order methods. Developing such methods requires involved nontrivial calculations, especially when made at low regularity or in high dimensions, and is very laborious when the schemes are constructed one equation at a time.

Based on iterating Duhamel's formula (1.4) once, we constructed in Section 1.1 the *first* order resonance-based integrator (1.22) as well as the Euler exponential method (1.9). In order to obtain higher-order schemes one needs to include the approximation to higher-order Duhamel iterates:

$$u(t) = e^{t\mathcal{L}}u_0 + \int_0^t e^{(t-s_1)\mathcal{L}} p(e^{s_1\mathcal{L}}u_0) ds_1 + \int_0^t e^{(t-s_1)\mathcal{L}} \left[ p'(e^{s_1\mathcal{L}}u_0) \int_0^{s_1} e^{(s_1-s_2)\mathcal{L}} p(e^{s_2\mathcal{L}}u_0) ds_2 \right] ds_1 + \dots$$
(1.35)

In the above we approximated p at  $u(s_i)$  by its Taylor expansion about the linear flow  $e^{s_j \mathcal{L}} u_0, j \in \mathbb{N}$ :

$$p(u(s_j)) = p(e^{s_j \mathcal{L}} u_0) + p'(e^{s_j \mathcal{L}} u_0) \int_0^{s_j} e^{(s_j - s_{j+1})\mathcal{L}} p(e^{s_{j+1} \mathcal{L}} u_0) ds_{j+1} + \dots$$

We note that Taylor expanding  $p(u(s_j))$  in the time variable was avoided, as this would require additional regularity on the solution.

We let r + 1 be the desired order of the scheme. To obtain a scheme of order r + 1, we need to consider the terms in the above formula (1.35) that contain up to r + 1 integrals in time, and approximate them suitably.

We start by discussing higher-order approximations to the class of dispersive equations (1.1). For an approximation at low-regularity using resonance-based schemes, the idea is to map the Duhamel iterates into Fourier space. The k-th Fourier coefficient of u(t) is denote by:

$$u_{k}(t) = e^{tP(k)}u_{k}(0) + e^{tP(k)}\int_{0}^{t} e^{-s_{1}P(k)}p_{k}(e^{s_{1}P(k)}u_{k}(0))ds_{1} + e^{tP(k)}\int_{0}^{t} e^{-s_{1}P(k)}\left[p_{k}'(e^{s_{1}P(k)}u_{k}(0))e^{s_{1}P(k)}\int_{0}^{s_{1}} e^{-s_{2}P(k)}p_{k}(e^{s_{2}P(k)}u_{k}(0))ds_{2}\right]ds_{1} + \dots$$

$$(1.36)$$

where the Fourier multiplier of  $\mathcal{L}$  is denoted by P(k). For instance, for the Schrödinger equation we have  $\mathcal{L} = i\Delta$  and  $P(k) = -ik^2$ . With a slight abuse of notation, we denoted by k all of the frequencies in Fourier, instead of  $k_1, k_2, k_3$ , etc.

At higher order, the resonance structure of the iterated integrals becomes very intricate, and it is a complicated task to choose a suitable resonance-based discretization. In the following, we discuss new techniques allowing for higher order constructions.

#### **Decorated trees**

We first present the work [BS22], which introduces an algorithmic approach to building resonance-based schemes up to arbitrary order and to the class of dispersive equations (1.1) on the *d*-dimensional torus. Generally speaking, the work [BS22] develops an algebraic framework – inspired by those in SPDEs via regularity structures [CK99, Hai14] – to systematize a new analytical idea. Namely, they encode the scheme using combinatorial structure called *decorated trees*, allowing to systematize the approximation of the Duhamel iterates. We divide the idea behind the work [BS22] into two steps:

a.) On the continuous level. We encode each of the Duhamel iterates in (1.36) by a decorated tree. In the case of the NLS equation (1.2), we have:

$$\begin{split} u_{k}(t) &= \underbrace{e^{tP(k)}}_{\Pi\left(\overset{k}{\bigcup}\right)(t)} u_{k}(0) + e^{tP(k)} \underbrace{\int_{0}^{t} e^{-s_{1}P(k)} p_{k}(e^{s_{1}P(k)} u_{k}(0)) ds_{1}}_{\Pi\left(\overset{k}{\bigcup}\right)(t)} \\ &+ e^{tP(k)} \underbrace{\int_{0}^{t} e^{-s_{1}P(k)} \left[ p_{k}'(e^{s_{1}P(k)} u_{k}(0)) e^{s_{1}P(k)} \int_{0}^{s_{1}} e^{-s_{2}P(k)} p_{k}(e^{s_{2}P(k)} u_{k}(0)) ds_{2} \right] ds_{1}}_{\Pi\left(\overset{k}{\bigcup}\right)} + \dots \\ &= \underbrace{\prod_{k=1}^{k} \frac{1}{2^{k}} \left[ \frac{1}{2^{k}$$

where  $\Pi$  maps decorated trees to Duhamel iterates. The edges encode the operators in Duhamel's formula. Namely, the brown edge encodes the linear flow  $e^{tP(k)}$ , and the blue edge encodes the time integral  $\int_0^t e^{-s_1P(k)} \dots ds_1$ . A dotted edge corresponds to taking the complex conjugate of the expressions. For example, we have

$$\Pi(\overset{(k)}{\downarrow})(t) = e^{tP(k)}, \quad \Pi(\overset{(k)}{\overset{(k)}{\bullet}})(t) = e^{-tP(k)}.$$

Lastly, the node decoration encodes the frequencies. This approach of encoding Duhamel's iterates in Fourier using decorated trees is also widely used in the theoretical study of dispersive equations. We discuss this further in Section 1.5.

The character  $\Pi$ , called a *pre-model* or *Feynman rule*, separates the algebraic level from the analytical one, and allows to expand the solution as a tree series expansion. We let r + 1 denote the order of the scheme, and  $\mathcal{V}_k^r$  the set of trees of size r + 1, encoding an iterated integral with up to r + 1 integrals. We can expand the solution (truncated at order r + 1) as the following Butcher-type tree series:

$$u_k^r(t) = \sum_{T \in \mathcal{V}_k^r} \frac{\Upsilon^p(T)}{S(T)} \Pi(T)(t), \qquad (1.37)$$

where  $S(T) \in \mathbb{N}$  is the symmetry factor associated to the tree T,  $\Upsilon^p(T)$  is the elementary differential which depends on  $u_k(0)$  and encodes the coefficient appearing in the iteration of Duhamel's formulation, and  $\Pi(T)(t)$  represents a Fourier iterated integral. In their article, the authors denote the above expansion (1.37) by  $U_k^r(t, u_0)$ . Lastly, we give an example of the set  $\mathcal{V}_k^r$ : for a first order scheme we fix r = 0 and obtain

 $(k_2)$ 

The above tree based formalism is inspired by the works of [BCCH20] and [Hai14]. There, this formalism is used to derive a general framework for solving a large class of singular SPDEs in subcritical-regimes, via the theory of Regularity Structures.

b.) On the discrete level. The second step is to approximate each Duhamel integral  $\Pi(T)(\tau)$ , where  $\tau$  denotes the time step. The approximation of each Duhamel integral is made using the resonance-based approach detailed previously (see Section 1.1.1). The discretization is automated via a notion of high and lower parts (1.18) dictated by the resonance structure R(k) of the underlying equation. This automatization is led by the approximation operator

#### $\Pi^{r,n}: \mathcal{V}_k^r \rightarrow approximate \ iterated \ integral,$

which is a resonance-based approximation of  $\Pi$ . Namely, given a decorated tree T,  $\Pi^{r,n}(T)$  is a resonance-based approximation of the iterated integral encoded by the tree T. The index n corresponds to a formal a priori assumption on the regularity required on the initial data, and allows to make simple Taylor expansions when sufficient regularity is assumed. The mapping  $\Pi^{r,n}$  expresses in Fourier space the numerical scheme after one time step, denoted by  $U_k^{n,r}(\tau)$ , as the following tree series:

$$U_k^{n,r}(\tau) = \sum_{T \in \mathcal{V}_k^r} \frac{\Upsilon^p(T)}{S(T)} \Pi^{n,r}(T)(\tau).$$
(1.39)

The main theorem of the work [BS22] is the following formal local error expansion, which sums all the local error made by the approximation of each iterated integral:

$$U_k^{n,r}(\tau) - u_k^r(\tau) = \sum_{T \in \mathcal{V}_k^r} \mathcal{O}_{\parallel \cdot \parallel} \left( \tau^{r+2} \mathcal{L}_{\text{low}}^r(T, n) \Upsilon^p(T) \right),$$
(1.40)

where  $\mathcal{L}_{low}^r$  involves all lower order frequency interactions neglected during the resonance-based approximation. The global order of the scheme is r + 1, and hence the local error is of order r + 2. We now give the author's definition of the big-O notation in (1.40) to express the local error term.

**Definition 1.4.1** Let  $\Phi^{\tau}(u_0) = u^1 \approx u(\tau)$  denote the numerical solution at time  $t = \tau$ . We write

$$u(\tau) - \Phi^{\tau}(u_0) = \mathcal{O}_{\parallel \cdot \parallel}(\tau^m \tilde{\mathcal{L}} u_0)$$
(1.41)

if in a suitable norm  $\|\cdot\|$ , it holds that

$$\|u(\tau) - \Phi^{\tau}(u_0)\| \le C(T, d)\tau^m \sup_{0 \le t \le \tau} \|q\Big(\tilde{\mathcal{L}}u(t)\Big)\|,$$
(1.42)

for some polynomial q, differential operator  $\tilde{\mathcal{L}}$ , and constant C independent of  $\tau$ .

If (1.41) holds, it is said that the numerical solution  $u^1$  approximates the exact solution u(t) at time  $t = \tau$  with a local error of order  $\mathcal{O}_{\|\cdot\|}(\tau^m \tilde{\mathcal{L}} u_0)$ .

**Example 5 (First order scheme for NLS)** For the first order low-regularity approximation to the NLS equation, we fix r = 0 and n = 1 in equation (1.39). The scheme after one time step is given by

$$U_k^{1,0}(\tau) = e^{-i\tau k^2} u_k(0) - i\tau e^{-i\tau k^2} \sum_{k=-k_1+k_2+k_3} \bar{u}_{k_1}(0) u_{k_2}(0) u_{k_3}(0) \varphi_1(2i\tau k_1^2),$$

and coincides with the first order resonance-based scheme (1.22). The only approximation made is on the integral  $\Pi(T_1)$ , where  $T_1$  is given in (1.38). Moreover,  $\mathcal{L}_{low}^0(T_1, n = 1)$  is given in Fourier by  $R_{low} = -2k_1(k_2 + k_3) + 2k_2k_3$ , and can be mapped in physical space to (1.23). In [BS22, Corollary 5.1] the authors write that the local error is of order  $\mathcal{O}_{\parallel,\parallel}(\tau^2|\nabla|u)$ , meaning that one obtains a local error of order  $\tau^2$  in a suitable norm  $\parallel \cdot \parallel$ , if one can bound the term (1.23) in this norm. The choice of the norm  $\|\cdot\|$  dictates the regularity assumptions needed on the solution to obtain global convergence, see Remark 1.4.2. Fixing the regularity index n to one, allows to construct a scheme whose local error only involves differential operators of order one. By taking n = 2, one recovers the classical Euler exponential method (1.9), whose local error is of order  $\mathcal{O}_{\|\cdot\|}(\tau^2 \Delta u)$ , involving differential operators of order two.

Obtaining the formal local error term (1.40) is an important first step towards obtaining convergence of the scheme. However, to obtain a global convergence result, one needs to fix the space and norm  $\|\cdot\|$  in which to work with, rigorously bound the local error term in this norm (as in (1.42)) and couple this with a stability argument to conclude. Deducing the global error from the local error is in general non trivial for PDE problems. The two articles [AB23a, AB23b] deal with this rigorous analysis, and are presented in Chapters 2 and 5 respectively.

**Remark 1.4.2 (Rigorous regularity assumptions)** The choice of norm taken in Definition 1.4.1 is important and determines the regularity needed to bound the local error term (1.42), and hence for the convergence of the scheme. We illustrate this on the first order approximation to the NLS equation, by taking two different choices of norms and quantifying the regularity assumptions thereby required.

When measuring in  $L^2$ -norm: one needs to bound the local error term (1.23) in  $L^2$ , which by using classical Sobolev embedding require  $u_0 \in H^{1+\frac{d}{4}}$ . Indeed, for each of the terms in (1.23), we have that for  $d \leq 3$ :

$$\|u\nabla u \cdot \nabla u\|_{L^{2}} \le C_{d} \|u\|_{L^{\infty}} \|\nabla u\|_{L^{4}}^{2} \le C_{d} \|u\|_{H^{\frac{d}{2}+\epsilon}} \|\nabla u\|_{H^{\frac{d}{4}}}^{2} \le C_{d} \|u\|_{H^{1+\frac{d}{4}}}^{3},$$

where we used the Sobolev embeddings  $H^{\frac{d}{4}} \hookrightarrow L^4$  and  $H^{\sigma} \hookrightarrow L^{\infty}$ ,  $\sigma > d/2$ .

We make the important point that even though the formal local error term in [BS22, Corollary 5.1] reads  $\mathcal{O}_{\|\cdot\|}(\tau^2 |\nabla|u)$ , one cannot bound this local error term in  $L^2$  while only asking for  $H^1$ -solutions. This is due to the polynomial q appearing in (1.42).

When measuring in  $H^{\sigma}$ ,  $\sigma > d/2$ : using the fact that the space is an algebra we can bound the local error term (1.23) for  $u_0 \in H^{\sigma+1}$ :

$$\|u\nabla u\cdot\nabla u\|_{H^{\sigma}} \leq C_d \|u\|_{H^{\sigma}} \|\nabla u\|_{H^{\sigma}}^2 \leq C_d \|u\|_{H^{\sigma+1}}^3.$$

This analysis is made more generally in  $H^{\sigma}$ -norm for  $\sigma \ge 0$  in Chapter 2, where first and second order convergence results are obtained.

We finish by discussing a limitation of the work [BS22]. Given that their algorithm heavily relies on expansions made in Fourier space, the method is restricted to spatial domains which are periodic and to dispersive equations with polynomial nonlinearities. We overcome this limitation in the work [ABBS22b], presented next. We introduce a nested commutator structure to pull out the oscillatory phases, which replaces and generalizes the Fourier-based constructions. This work extends the tools presented in Section 1.4.1 to higher-order.

#### **Introducing Chapter 3**

In this chapter, we use an algebraic framework, to systematically derive low-regularity integrators on the general domains of Section 1.4.1. We consider the approximation, up to arbitrary order, to the following class of evolution equations:

$$\begin{cases} \partial_t u(t,x) - \mathcal{L}u(t,x) = \sum_{\mathfrak{l}} f_{\mathfrak{l}} \big( u(t,x), \overline{u}(t,x) \big) V_{\mathfrak{l}}(x), \quad (t,x) \in \mathbb{R} \times \Omega, \ \Omega \subseteq \mathbb{R}^d, \\ u(0,x) = u_0(x). \end{cases}$$
(1.43)

Here,  $\mathcal{L}$  and  $f_{\mathfrak{l}}$  satisfy the same hypotheses as  $\mathcal{L}$  and f in equation (1.32) of Section 1.4.1. We reconize the GP equation (1.34) in the case where  $V_0 = 1, V_1 = V, f_0(u, \bar{u}) = \bar{u}u^2, f_1(u, \bar{u}) = u$ . The above class of evolution equations is more general than both (1.1) and (1.32).

In order to obtain higher-order approximations we consider the Duhamel's iterates (1.35) where we replace the nonlinearity p(u) by  $\sum_{i} f_{i}(u, \overline{u}) V_{i}$ . We obtain a scheme by discretizing each iterated integral, as detailed below.

a.) On the continuous level. This time we encode by decorated trees the Duhamel iterates (1.35) in the physical variables. We generalize the decorated trees in [BS22] which relied on Fourier-based expansions, to consider more general domains and equations. We let the brown edge encode the integral of the linear flow :

$$\int_0^t e^{(t-s)\mathcal{L}} \cdots ds = \mathbf{I}.$$

Each node of the tree encodes a term in the nonlinearity, and the decoration on the I-th node encodes the potential  $V_{I}$  (and possible monomials). For example, for the GP equation there are two nodes corresponding to the two nonlinear terms  $(f_{I}, I = \{0, 1\})$ :

$$\circ \equiv (e^{is\Delta}u_0^2)(e^{-is\Delta}\overline{u_0}), \quad \bullet \equiv (e^{is\Delta}u_0)V.$$

Incoming edges to a node induce derivatives on  $f_1$ , allowing to encode Duhamel iterates with more than one integral in time. This allows to iteratively define a set of trees  $\mathcal{T}^r$  encoding – through the mapping  $\Pi$  – each Duhamel iterate consisting of r + 1 integrals in time. We can thus express u(t) truncated at order r + 1, as the tree series expansion:

$$u^{r}(t) = e^{t\mathcal{L}}u_0 + \sum_{T \in \mathcal{T}^{r}} \Pi(T)(t).$$

We note that the symmetry factor S(T) and elementary differential  $\Upsilon^p(T)$  previously appearing in (1.37), are directly included in  $\Pi(T)$ , see Remark 3.3.1.

b.) On the discrete level. We define a new operator  $\Pi_A^r$  mapping a decorated tree in  $\mathcal{T}^r$  to an approximate Duhamel iterate. The map  $\Pi_A^r$  is a discretized version of  $\Pi$ . The desired order of the scheme is r + 1 and A is a given domain of regularity of  $u_0$  and  $(V_1)_{\mathfrak{l}}$ , reflecting the a priori regularity assumed on the initial data and the potentials. The role of A was played by the index n in the periodic setting, see (1.39). The operator  $\Pi_A^r$  discretizes the Duhamel iterates at low-regularity by filtering out the dominant parts, integrating them exactly, and approximating the lower order parts through commutator based expansions. The numerical scheme  $U_A^r$ , after one time step, is given by:

$$U_A^r(\tau) = e^{\tau \mathcal{L}} u_0 + \sum_{T \in \mathcal{T}^r} \Pi_A^r(T)(\tau),$$

For the sake of clarity, we have not included in the above discussion a technical preliminary step, which extracts the linear flow  $e^{t\mathcal{L}}u_0$  from inside the nonlinearity  $p(e^{t\mathcal{L}}u_0)$ . This step gives a unified form of the Duhamel iterates, allowing to encode them using decorated trees, see Section 3.4. A comparison between the decorated trees used in [BS22] and those from the present work is given in Remark 3.2.4.

The main theorem is a formal local error expansion, measuring the error made when approximating each Duhamel iterate over one time step.

**Theorem 1.4.2** (Theorem 3.4.3 in a simplified notation)

$$U_A^r(\tau) - u^r(\tau) = \sum_{T \in \mathcal{T}^r} \mathcal{O}_{\|\cdot\|} \left( \tau^{r+2} \mathcal{L}_{low}^r(T, A) \right)$$

where the operator  $\mathcal{L}_{low}^{r}(T, A)$ , embeds the necessary regularity on the solution.

The above accounts for the local error made by the approximation in step b. In the end, we need to sum the above error terms with the error from the truncation made in step a. If the nonlinearity considered does not involve derivatives, step b is the dominant one and dictates the needed regularity, for a discussion on the subject see Remark 1.5.1.

We stress the fact that these higher-order techniques only guarantee a formal local error bound as above, where the big-O notation is to be understood in the sense of Definition 1.4.1. In order to obtain convergence of the scheme, a separate and careful analysis needs to be made: first for estimating the local error in an appropriate functional space, and secondly to obtain stability of the scheme.

For an example of this algorithmic approach, we refer to Section 3.5.1 where classical and low-regularity first and second order integrators for the GP equation are obtained, together with their formal local error expansion. To obtain a rigorous convergence result of these schemes, we once again refer to the results of Chapter 2.

We finish this section with a short historic note. The idea of using a tree-series based expansions to express the numerical scheme was heavily studied in the context of ODEs in the 70s by the pioneering works of [But72, HW74]. More precisely, it was used for the characterization of Runge-Kutta methods via Butcher series (B-series), see the reviews [But16, HLW10]. The algebraic framework used at this time was a foundation first for Gubinelli's approach to Rough Path Theory [Gub04, Gub10], and afterwards to Hairer's theory of regularity structures [Hai14], a generalization of Rough Paths. Nowadays, these ideas are making their way back to the study of numerical schemes, but for PDEs instead of ODEs.

### 1.5 A brief introduction to Part II

In the previous Section 1.4.2 we discussed the design to higher order of new families of numerical schemes for *deterministic* PDEs, suited for *low-regularity* initial data. Typical examples of low regularity functions are given by realizations of random fields. In the work [ABBS22a] presented in Chapter 4, we enter a probabilistic setting by considering a randomized initial condition:

$$u(0,x) = v^{\eta}(x) = \sum_{k \in \mathbf{Z}^d} c_k \eta_k e^{ikx},$$
(1.44)

where the  $\eta_k$  are independent standard complex Gaussian random variables. The above expression is a Gaussian randomization of a deterministic function  $\sum_k c_k e^{ikx}$ . The choice of how one measures the numerical discretization in a stochastic setting can take a probabilistic form and be formulated in different ways. In Chapter 4, we choose to quantify the convergence rate of approximations of the second moment  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$  of the k-th Fourier coefficient  $u_k$  of the solution u to the class of dispersive equations (1.1). We construct approximations of arbitrarily high order to the second moment  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$ , based upon the tree series expansions discussed in Section 1.4.2. Working with gaussian random variables and integer moments of polynomial nonlinearities of these random variables is a crucial choice that allows us to use Wick formulas and their rich algebraic structure to set up the framework. We now outline the main steps involved in Chapter 4.

Following equation (1.37), one can express the time continuous solution, truncated at order r + 1, as a tree series expansion:

$$u_{k}^{r}(\tau, v^{\eta}) = e^{tP(k)}\eta_{k}c_{k} + e^{tP(k)}\int_{0}^{\tau} e^{-s_{1}P(k)}p_{k}(e^{s_{1}P(k)}\eta_{k}c_{k})ds_{1} + \dots = \sum_{T\in\mathcal{V}_{k}^{r}}\frac{\Upsilon^{p}(T)}{S(T)}\Pi(T)(\tau),$$

where we have taken  $t = \tau$ , the time step. We can thus make a first approximation, with local error in  $O(\tau^{r+2})$ , of the second moment  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  using the truncated solution  $u_k^r$ :

$$\mathbb{E}(|u_k(\tau, v^{\eta})|^2) \approx \mathcal{Q}_{\leq r+1} \mathbb{E}(|u_k^r(\tau, v^{\eta})|^2), \qquad (1.45)$$

where  $\mathcal{Q}_{\leq r+1}$  keeps only the terms of order at most r+1, i.e. in  $O(\tau^m)$ , with  $m \leq r+1$ . When computing the product  $|u_k^r(\tau, v^\eta)|^2$ , a number of terms involve products of an odd number of noise variables, that have null expectation. The other terms are even polynomials in the noise variables, whose expectation is described by Wick's formula, which states that for centered gaussian variables the expectation of the product of an even number of gaussians is given by the product of the expectations of their two-by-two product, summed over all possible pairings. For example, if u satisfies the NLS equation (1.2), one term we would want to approximate is the expectation  $\mathbb{E}\left(\overline{\Upsilon^p(T_0)\Pi(T_0)}\Upsilon^p(T_1)\Pi(T_1)\right)$ , where  $T_0$  and  $T_1$  correspond to the terms of order zero and one in Duhamel's formula, see (1.38). Given that the trees  $T_0$  and  $T_1$  have in total four leaves, the expectation contains terms of the form:

$$\mathbb{E}(\bar{\eta}_k \bar{\eta}_{k_1} \eta_{k_2} \eta_{k_3}) = \mathbb{E}(\bar{\eta}_k \bar{\eta}_{k_1}) \mathbb{E}(\eta_{k_2} \eta_{k_3}) + \mathbb{E}(\bar{\eta}_k \eta_{k_2}) \mathbb{E}(\bar{\eta}_{k_1} \eta_{k_3}) + \mathbb{E}(\bar{\eta}_k \eta_{k_3}) \mathbb{E}(\bar{\eta}_{k_1} \eta_{k_2}) \\ = \delta_{k, -k_1} \delta_{k_2, -k_3} + \delta_{k, k_2} \delta_{k_1, k_3} + \delta_{k, k_3} \delta_{k_1, k_2}.$$

The first line in the above uses Wick's formula, while the second line exploits the properties of the random variables  $(\eta_k)_k$ . We see that the above yields a nontrivial value under the following conditions on the frequencies:  $(k, k_2) = (-k_1, -k_3)$  or  $(k, k_1) = (k_2, k_3)$  or  $(k, k_1) = (k_3, k_2)$ . We can capture these conditions diagrammatically through Feynman-type diagrams:



We see that computing the expectation amounts, through Wick's formula, to pairing the leaves of the trees. We introduce for this purpose a new combinatorial structure called *paired decorated forests*, which are composed of two decorated trees whose decorations on the leaves come in pair. The expansion for  $\mathbb{E}(|u_k(t, v^{\eta})|^2)$ , following the first approximation (1.45), can then be rewritten in terms of these paired decorated forests:

$$\mathbb{E}\left(\left|u_{k}(\tau, v^{\eta})\right|^{2}\right) \approx \mathcal{Q}_{\leq r+1}\mathbb{E}\left(\left|u_{k}^{r}(\tau, v^{\eta})\right|^{2}\right) = \sum_{T_{1}: T_{2}\in\mathcal{G}_{k}^{r}} \frac{\Upsilon^{p}(T_{1})\Upsilon^{p}(T_{2})}{S(T_{1})S(T_{2})} \bar{\Pi}(T_{1})(\tau) \Pi(T_{2})(\tau).$$

The set  $\mathcal{G}_k^r$  consists of all paired forests  $F = T_1 \cdot T_2$  whose encoding as Duhamel iterates  $\overline{\Pi}(T_1)(\tau) \Pi(T_2)(\tau)$  contains at most r+1 integrals in time.

As discussed in steps b.) of Section 1.4.2, by replacing the continuous mapping  $\Pi$  by a discrete analogue  $\Pi^{n,r}$  we obtain a tree-series expansion of our numerical approximation after one time step, which we denote by  $V_k^{n,r}(\tau)$ :

$$V_k^{n,r}(\tau) = \sum_{F=T_1 \cdot T_2 \in \mathcal{G}_k^r} \frac{\Upsilon^p(T_1) \,\Upsilon^p(T_2)}{S(T_1) S(T_2)} \mathcal{Q}_{\le r+1} \big( \bar{\Pi}^{n,r} T_1 \Pi^{n,r} T_2 \big)(\tau).$$
(1.46)

The operator  $\Pi^{n,r}$  is the same as the one used in the expansion (1.39) and allows for a resonance-based approximation to the Duhamel iterates. Together with the design (1.46) of higher-order approximation to the second moment  $V_k(\tau) = \mathbb{E}(|u_k(\tau, v^{\eta})|^2)$ , we derive the following formal local error:

$$V_k^{n,r}(\tau) - V_k(\tau) = \sum_{T_1 \cdot T_2 \in \mathcal{G}_k^r} \mathcal{O}_{\|\cdot\|} \left( \tau^{r+2} \mathcal{L}_{\text{low}}^r(T_1 \cdot T_2, n) \bar{\Upsilon}^p(T_1) \Upsilon^p(T_2) \right),$$
(1.47)

where the operator  $\mathcal{L}_{low}^{r}(T_1 \cdot T_2, n)$  embeds the necessary regularity on the solution.

Remark 1.5.1 (An additional term in the local error) We note that the above equation (1.47) takes into account the second approximation step (1.46), by summing up the local errors made by  $\Pi^{n,r}$  to  $\Pi$ , during the resonance-based approximation to each Duhamel iterate. The first approximation step made at equation (1.45) can also require additional regularity on the initial data. In most cases, the second approximation step made by  $\Pi^{n,r}$  requires the most regularity. This is however not the case for instance for the KdV equation (1.3), where the leading error term stems from the first approximation (1.45). In this case one needs to look at the general form of the local error, by taking into account both approximation steps as follows:

$$V_k^{n,r}(\tau) - V_k(\tau) = \sum_{T_1 \cdot T_2 \in \mathcal{G}_k^r} \mathcal{O}_{\|\cdot\|} \left( \tau^{r+2} \mathcal{L}_{\text{low}}^r(T_1 \cdot T_2, n) \bar{\Upsilon}^p(T_1) \Upsilon^p(T_2) \right) \\ + \mathcal{O}_{\|\cdot\|} \left( \tau^{r+2} |\nabla|^{\alpha(r+2)}(k) \mathbb{E}(\tilde{p}_k(u(t), v^\eta)) \right),$$

for some polynomial  $\tilde{p}$ . We refer to the proof of Theorem 4.4.3 for details.

While the derivation of the formal local error term (1.47) is a first important step, the next step is to rigorously prove convergence of the method through global error bounds. Contrary to the deterministic setting, no convergence results have been obtained in the probabilistic setting (1.44) with i.i.d. gaussian random initial conditions. In the theoretical analysis of PDEs, many interesting results have been established in this probabilistic setting in the last decade, see for instance [CO12, NS15] for NLS type equations, [BT08] for the cubic wave equation, [Bri20] for the derivative nonlinear wave equation, [NPS13] for the periodic Navier-Stokes equations, and more generally [BOP19] for a survey on recent developments in *probabilistic Cauchy theory*. These papers study nonlinear PDEs with random initial data in singular regimes, and establish probabilistic well-posedness results at much lower regularity regimes than what can be attained in the deterministic setting, reaching down to scaling-supercritical regularities. A question of personal interest for the future is whether one can devise discrete counterparts to these probabilistic PDE techniques to obtain improved convergence results.

Finally, this numerical probabilistic work uses tools which are similar to the ones used in the theory of wave turbulence for the rigorous derivation of wave kinetic equations, where the quantity  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$  plays an important role. These recent works in wave turbulence theory also use decorated trees and construct Feynman diagrams through Wick's formula to express the second moment  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$  as a tree series expansion. We refer for instance to [BGHS21, ACG21, DH23, DNY22, LS11] in the context of NLSEs and to [HRST22, ST21] for KdV-type systems. This active field has numerous applications in physics and applied sciences (e.g. oceanography and atmospheric sciences) and seeks a statistical description of the out-of-equilibrium dynamics of equations, ranging from the NLSE to water waves or Vlasov-Maxwell, see [ZF67]. The fundamental equation of wave turbulence theory, called the wave kinetic equation or the Kolmogorov-Zakharov equation, is used on a daily basis for wave forecasting.

The author would like to mention that the main motivation in considering the randomized initial condition (1.44) and the approximation to the second moment  $\mathbb{E}(|u_k(t, v^{\eta})|^2)$ , came from listening to talks at a semester workshop at the ICERM and through the Simons Collaboration on Wave Turbulence, regrouping in part the probabilistic Cauchy theory and wave turbulence theory communities.

### 1.6 A brief introduction to Part III

Algorithms preserving the structure of the underlying equation constitute an important part of computational mathematics, see [HLW10, Fao12]. However, in general they heavily rely on highly regular solutions. For instance, for the NLS equation (1.2), we mentioned in Section 1.2.1 finite difference schemes [Bes04, BDDLV21, SS84] which preserve the mass and energy, as well as the Strang splitting methods [Lub08, BBD02] which preserves the mass and nearly preserves the energy, see Remark 1.6.1. Yet, all of these schemes require at least  $H^4$  solutions for second order convergence in  $L^2$ .

The goal of Part III is to extend the novel nonlinear approach and low-regularity approximation techniques presented in Part I, to the construction of symmetric schemes for dispersive equations. The schemes we obtain show very good long-time behavior in numerical experiments both in the smooth and non-smooth setting. Surprisingly, even in the smooth setting, our schemes surpass the long time energy preservation properties of classical splitting methods.
#### 1.6.1 A symmetric scheme for the nonlinear Schrödinger equation, suited for nonsmooth solutions

We start by presenting the work [AB23b] on the NLS equation (1.2) contained in Chapter 5. The NLS equation is time reversible, meaning that  $\overline{u(-t,x)}$  is again solution of (1.2), and preserves the mass

$$||u(t)||_{L^2}^2 = \int |u(t,x)|^2 dx = ||u_0||_{L^2}^2$$

and the energy

$$E(t) = \frac{1}{2} \int |\nabla u(t,x)|^2 dx + \frac{1}{4} \int |u(t,x)|^4 dx = E(0)$$

of the solution over its interval of existence in time.

Part III builds on the observation that low regularity schemes discussed in earlier sections do not preserve the invariants of the underlying equation, such as the mass and energy for the NLS equation, see Figures 5.2 and 5.3. In Chapter 5, we construct and rigorously analyze a novel scheme with strong geometric structure at low regularity, allowing for a reliable symmetric approximation to the NLS equation. Namely, the scheme preserves the time-reversibility of the equation and numerical simulations show good near-preservation of the mass and energy over long times. Moreover, we witness experimentally that the new symmetric low-regularity scheme has better long time near-preservation of the energy than classical splitting schemes. We refer to Figure 1.2 where we plot the error in the energy over time, first for a smooth initial condition  $u_0 \in C^{\infty}$ , then in the case of a rougher initial condition  $u_0 \in H^2$ . We notice that the first order resonance-based scheme (blue) does not preserve the energy, even over short times. In Figure 1.2a the Lie and Strang splitting (pink and green) approximatively preserves the energy up until a point ( $t \sim 30$ ), while in the rougher case of Figure 1.2b, the energy explodes after much shorter times. This behavior is rigorously studied in the work of [Fao12] where the author shows that a CFL condition is necessary to guarantee long-time approximate energy preservation for splitting methods, see also Remark 1.6.1. Namely, if we fix the time step  $\tau$  and increase the highest Fourier frequency K, thereby going from an ODE to the (desired) PDE setting, the energy ceases to be preserved at some point. In comparison, the new symmetric low-regularity scheme (yellow triangle) does not suffer from a comparable CFL condition, and performs well for both smooth and low-regularity solutions. We refer to Figures 6.5a and 6.6a for simulations over longer times (t = 4000).

#### Remark 1.6.1 (Near energy preservation of splitting methods for the NLS equation under a CFL condition)

Splitting methods preserve the symplectic structure and  $L^2$  norm of the solution, however they do not preserve exactly the energy on the discrete level. Indeed, under suitable assumptions, the numerical approximation can be shown to solve (almost) exactly a *modified* PDE at each time step. This results in the existence of a *modified* energy which is preserved along the numerical flow, where – among other assumptions – for smooth solutions and under a CFL condition, it can be shown that the modified energy is close to the exact energy. More specifically, the CFL condition imposes that the time step size  $\tau$  has to be chosen such that  $\tau \leq K^{-2}$  where K denotes the highest frequency in the discretization, see [Fao12]. This step size restriction is not only a theoretical technicality, but also observed in numerical experiments, see Figure 1.2. We also refer to a number of important works where the near-preservation of the energy over exponentially long times is shown using modulated Fourier expansions (see [CHL08, GL10b, GL10a]) or normal form techniques ([FGP10a, FGP10b, FG11]).

We now portray the idea behind the construction of this symmetric scheme. We recall that a scheme is said to be symmetric or time reversible if it remains unchanged under the transformations  $\tau \to -\tau$  and  $n \leftrightarrow n + 1$ .

We once again start from Duhamel's integral mapped in Fourier space (1.15), whose integrand is decomposed in (1.17) into the three factors:  $e^{isR_{\text{dom}}(k)}e^{isR_{\text{low}}(k)}g(s)$ , with  $R(k) = R_{\text{dom}}(k) + R_{\text{low}}(k)$  given in (1.18). To construct the first-order resonance-based scheme (1.22) we simply Taylor expanded both g(s) and  $e^{isR_{\text{low}}(k)}$  about the left end point:  $e^{isR_{\text{low}}(k)}g(s) \approx g(0)$ . This however breaks the symmetric structure of the underlying PDE and leads to schemes which are not structure preserving. We thus look for a symmetric approximation of both the non-oscillatory g(s) and lower part  $e^{isR_{\text{low}}(k)}$ . In Chapter 5 we choose the following symmetric approximation:  $h(s) \approx h(0)\mathbb{1}_{[0,\tau/2]} + h(\tau)\mathbb{1}_{(\tau/2,\tau]}$ , which in our context yields

$$I_{\tau} = \int_{0}^{\tau} e^{isR_{\text{dom}}(k)} \underbrace{e^{isR_{\text{low}}(k)}g(s)}_{\approx g(0)^{1}_{[0,\tau/2]} + e^{i\tau R_{\text{low}}(k)}g(\tau)^{1}_{[\tau/2,\tau]}} ds.$$
(1.48)

By integrating exactly the dominant part  $e^{isR_{\text{dom}}(k)}$ , mapping back to physical space and in the untwisted variable, we obtain the symmetric scheme:

$$u^{n+1} = \Phi_{-\tau/2}^{-1} \circ \Phi_{\tau/2}(u^n) = e^{i\tau\Delta}u^n - i\frac{\tau}{2}e^{i\tau\Delta}((u^n)^2\varphi_1(-i\tau\Delta)\overline{u^n}) - i\frac{\tau}{2}\Big((u^{n+1})^2\varphi_1(i\tau\Delta)\overline{u^{n+1}}\Big),$$
(1.49)

where  $\Phi_{\tau}$  is the first order resonance-based scheme (1.22) constructed in the introduction. In fact, the symmetric approximation (1.48) is chosen because it coincides with the symmetrization of the first order resonance-based scheme  $\Phi_{\tau}$ . For other choices of symmetric approximations, we refer to Section 1.6.2 and to Chapter 6.



Figure 1.2 – Error in the Hamiltonian for smooth and non-smooth initial data with  $t = n\tau \in [0, 50]$ ,  $\tau = 0.02$  and K = 1024. The plots are taken from the work [ABBMS23], see Section 6.5.2 for further numerical experiments.

A notable difference between this symmetric scheme and previous low-regularity schemes is its *implicit* nature. We refer to Figure 5.4 which shows that the computational cost of the symmetric scheme is comparable to that of the asymmetric second order resonance-based scheme. Namely, its improved convergence properties make up for the extra cost of solving the implicit system (1.49) at each time step.

A desired property of symmetric schemes is that they are of *even order*. This however is only true under sufficient regularity assumptions of the solution. Chapter 5 focuses on obtaining the minimal regularity assumptions needed for convergence of the scheme from first to second order. The error analysis presented is suited both for periodic domains  $\Omega = \mathbb{T}^d$  and smooth bounded domains  $\Omega \subset \mathbb{R}^d$  with homogeneous Dirichlet boundary conditions. We choose to work in the fractional space

 $X^{\sigma}(\Omega) = \mathcal{D}((-\Delta)^{\sigma/2}), \quad ||u||_{X^{\sigma}}^{2} = ||u||_{L^{2}}^{2} + ||(-\Delta)^{\sigma/2}u||_{L^{2}}^{2}.$ 

In particular  $X^{\sigma}(\mathbb{T}^d) = H^{\sigma}(\mathbb{T}^d)$ , and in the more subtle case of a smooth bounded domain, the space  $X^{\sigma}(\Omega)$  embeds both the regularity and the necessary boundary conditions. In this case, one is lead to characterize domains of fractional powers of the Dirichlet Laplacian, via complex interpolation spaces, see Section 5.2. We obtain the following convergence result:



where the regularity  $\alpha$  is minimized through well-chosen bilinear estimates. The admissible values of  $\alpha$  and  $\gamma$  are represented by the blue zone on the above right picture. The detailed theorems are given in Section 5.1.2, and the bilinear estimates in Section 5.2.2. We refer to Figure 1.1 where convergence plots are made for  $H^1$  and  $H^2$  initial data. We observe that the symmetric scheme (yellow) has a better error constant than the previous resonance-based schemes (light and dark blue), the exponential integrator (purple), and the Lie splitting method (pink).

We give a brief overview of current *local in time* results that build new bridges between structure preserving methods and low-regularity approaches. In [WY22], a first-order Fourier integrator is introduced for the NLS equation (1.2) set on the 1-dimensional torus  $\mathbb{T}$ , which almost conserves the mass. For the cubic Klein-Gordon equation set on  $\mathbb{T}$ , an explicit symmetric three time-step low-regularity integrator was obtained by [WZ22a], using Gautschi-type methods. The authors of [MS23] introduce a framework for building novel symplectic resonance-based integrators, encompassing both the KdV equation and the 1-dimensional NLS equation. A symmetrized resonance-based integrator for the Schrödinger map (SM) is obtained in [BMS22], where they relate the SM flow to the NLS equation set on 1-*d* torus  $\mathbb{T}$  via the Hasimoto transform. Finally, the recent work [BW23a] establishes a second-order explicit and symmetric three-time step method for the GP equation (1.34) with a low-regularity potential *V*, using Gautchi-type exponential wave integrators.

Remark 1.6.2 (From short to long time scales) All of these convergence results allow to rigorously prove the reliability of a numerical discretization over short times [0, T], as the error constant  $C_T$  usually depends exponentially on the final time T. Pushing the analysis to obtain error estimates over longer times, or even *uniformly in time*, is thus a highly nontrivial problem.

A first step was made by the authors of [CS22b]. Using scattering theory in order to obtain quantitative time decay estimates, they show uniform in time error estimates for the NLS equation on the full space  $\mathbb{R}^d$ , for a Lie splitting discretization. Their convergence analysis is, however, limited to  $\mathbb{R}^d$  as it heavily relies on dispersive effects, which do not hold on the torus  $\mathbb{T}^d$  or more generally on compact domains. The step from the full space to the compact setting is – as in the continuous setting – nontrivial since in the latter dispersion does not translate into decay. Nevertheless, compact domains are computationally very interesting as spatial discretizations of nonlinear PDEs are in general performed in truncated domains. Hence, obtaining results uniformly in time which can also be implemented on the discrete level is a very interesting open problem.

To prove well-posedness on the theoretical level, a common approach is to either start by showing LWP over a *small* time-interval [0, T], or global well-posedness with *small* initial data  $||u_0|| < \epsilon$ . On the numerical level, we have previously discussed the first case; the second case is explored at low-regularity in [FMS23].

The work of Chapter 5 initiated the fruitful collaboration [ABBMS23] presented in Chapter 6, which we discuss next.

#### **1.6.2** Symmetric higher-order extensions

Chapter 6 extends the construction of the symmetric low-regularity integrator (1.49) to higher order and for the class of dispersive equations (1.1), when set on the torus  $\mathbb{T}^d$ .

We first recall Duhamel's formula (1.4), which, in the twisted variables (1.11) and mapped in Fourier, corresponds to

$$v_k(t) = v_k(0) + \int_0^t e^{-s_1 P(k)} p_k \left( e^{s_1 P(k)} v_k(s_1) \right) ds_1.$$
(1.50)

Previously, to obtain higher-order schemes we iterated Duhamel's formula about the linear flow  $e^{s_j \mathcal{L}} u_0$ , yielding (1.35) and (1.36) in Fourier space. In the twisted variable, this amounts to iterating the above equation (1.50) about the left-end point  $v_k(0)$ . Namely, we substitute  $v_k(s_1) = v_k(0) + \int_0^{s_1} \cdots ds_2$  in the above equation, then use a Taylor expansion in  $p_k$  about  $e^{s_1 P(k)} v_k(0)$ .

In the previous Section 1.6.1, we saw that making this asymmetric approximation about the left end point  $v_k(s) \approx v_k(0)$ (or equivalently  $g(s) \approx g(0)$  in (1.48)) leads to schemes which are not structure preserving. The solution we propose is to make a symmetric approximation of the non-oscillatory part g(s) (and hence  $v_k(s)$ ), such as the approximation  $v_k(s) \approx v_k(0) \mathbb{1}_{[0,\tau/2]} + v_k(\tau) \mathbb{1}_{(\tau/2,\tau]}$ , which puts equal weights on both end points s = 0 and  $s = \tau$ . Hence, the first step towards obtaining higher order symmetric schemes is to iterate Duhamel's formula in  $v_k(s_j)$  in a symmetric fashion, by putting equal weights on Duhamel's formulas centered about 0 and  $\tau$ :

$$v_k(s_j) = v_k(0) + \int_0^{s_j} e^{-s_{j+1}P(k)} p_k \Big( e^{s_{j+1}P(k)} v_k(s_{j+1}) \Big) ds_{j+1},$$
  
$$v_k(s_j) = v_k(\tau) + \int_{s_j}^{\tau} e^{-s_{j+1}P(k)} p_k \Big( e^{s_{j+1}P(k)} v_k(s_{j+1}) \Big) ds_{j+1}.$$

In Chapter 6, we first take the mid-point rule as our symmetric approximation:

$$v_k(s_j) = \frac{v_k(0) + v_k(\tau)}{2} + O(\tau), \quad s_j \in [0, \tau].$$
(1.51)

We let r + 1 denote the order of the scheme. We iterate the above equation (1.51) inside Duhamel's formula (1.50) and obtain an expansion of the continuous solution (written in the untwisted variables) and up to order r + 1 as a tree series

$$u_{\mathrm{mid},k}^{r}(t) = \sum_{T \in \mathcal{V}_{k}^{r}} \frac{\Upsilon_{\mathrm{mid}}^{p}(T)}{S(T)} \Pi_{\mathrm{mid}}(T)(t).$$

$$(1.52)$$

The above series is of the same form as the tree-series expansion (1.37), the only difference lies in the definition of the character  $\Pi$  and coefficient  $\Upsilon^p$ . Here, we encode the symmetric iterations of Duhamel's formula in  $\Pi_{\text{mid}}$  and  $\Upsilon^p_{\text{mid}}$ , whereas  $\Pi$  and  $\Upsilon^p$  encoded Duhamel's iterates centered about zero (see (1.36) or (1.50)). Note that in Chapter 6 everything is directly written in the untwisted variables. It is however much more intuitive to start by working in the twisted variables, and to make the decomposition (1.15) and (1.17), as presented above.

The next step is to discretize at low-regularity each Duhamel iterate  $\Pi_{\text{mid}}(T)(\tau)$  in a symmetric fashion. This is to be compared with Section 1.6.1: up to this point we have approximated g(s) in (1.48), and we are now ready to approximate the lower part  $e^{isR_{\text{low}}(k)}$ . We introduce the operator  $\Pi_{\text{mid}}^{n,r}$ , which based on the frequency decomposition R(k) (and on the a priori degree of regularity of the solution n), gives a symmetric low-regularity approximation of the remaining oscillatory factors, based on polynomial interpolation. After one time step the scheme takes the form

$$U_{\mathrm{mid},k}^{n,r}(\tau) = \sum_{T \in \mathcal{V}_k^r} \frac{\Upsilon_{\mathrm{mid}}^p(T)}{S(T)} \Pi_{\mathrm{mid}}^{n,r}(T)(\tau),$$
(1.53)

together with the formal local error expansion:

$$U^{n,r}_{\mathrm{mid},k}(\tau) - u^r_{\mathrm{mid},k}(\tau) = \sum_{T \in \mathcal{V}^r_k} \mathcal{O}_{\|\cdot\|} \left( \tau^{r+2} \mathcal{L}^r_{\mathrm{low}}(T,n) \Upsilon^p_{\mathrm{mid}}(T) \right).$$
(1.54)

This is to be compared with the tree series (1.39) and local error (1.40).

Throughout this iterative procedure of building the scheme, we choose the mid-point rule as our symmetric approximation, and obtain the class (1.53). We are also interested in obtaining a more general class of symmetric approximations encompassing the class (1.53), the previously built symmetric integrators (such as (1.49)), as well as new symmetric integrators. In Section 6.3.2 we derive a *forest formula*, which characterizes a large class of symmetric schemes. We derive a general formula (6.14) parametrizing resonance-based and classical integrators and deduce conditions on the coefficients of this formula to obtain a symmetric scheme, see Proposition 6.52 and Section 6.4.2. While the forest formula (6.14) comprises a large class of integrators, it is not defined in a recursive manner, hence obtaining a closed formula for the local error terms (such as (1.54)) is not obvious. The advantage of fixing a symmetric approximation (such as the mid-point rule (1.51)) and constructing the iterative procedure (1.52) on each tree T, is that one has a derivation of the local error terms (1.54) at hand.

We refer to Figure 1.2 where new symmetric integrators (light yellow and purple) are tested numerically, and show good near-preservation of the energy without CFL conditions.

Remark 1.6.3 (Symmetric versus symplectic schemes) The idea behind the chapters 5 and 6 is heavily based upon the decompositions (1.17) and (1.18). We first iterate Duhamel's formula in the nonlinearity in a symmetric fashion, give a symmetric approximation to the lower order parts and integrate the dominant part exactly. This yields symmetric schemes. To obtain symplectic resonance-based schemes, a different decomposition of the oscillatory terms  $e^{isR(k)}$  must be done to ensure that the approximation originates in another Hamiltonian system with the same symplectic structure and conservation of normalization. A suitable decomposition of the frequencies R(k) was recently obtained by the authors of [MS23] in the case of the KdV equation and the 1-dimensional NLS equation (see also Remark 6.2.5).

*Prospects:* Part III makes a first step towards obtaining reliable long-time approximations to dispersive equations, both in smooth and non-smooth regimes. An interesting next step is to see what can be rigorously proven on the

near-preservation of the mass and the energy over long times, as observed in the numerical experiments of Section 6.5.2. Further developing these low-regularity and structure preserving techniques should enhance our understanding of the global behavior of dispersive equations, such as the existence of global solutions or quantization effects.

In contrast, for PDEs whose solution is not defined globally in time, finding reliable descriptions of blow-up phenomena, and hence describing the formation of singularities faithfully, remains a grand challenge. We refer to the major theoretical breakthroughs in this direction in the context of the NLS equation [MRS10, MRRS22]. Another interesting application is the detection of dispersive blow-ups [BS93, BS10, BPSS14], which are point-wise blow-ups preserving the global well-posedness of the solution. These mathematical phenomena may be relevant in explaining the formation of rogue waves in shallow and deep waters [KPS08, CHA11].

# Part I

Beyond periodic boundary conditions and higher-order extensions

## Chapter 2

# Error analysis of a class of semi-discrete schemes for solving the Gross-Pitaevskii equation at low regularity

This chapter is based on the article [AB23a] published in the Journal of Computational and Applied Mathematics.

Abstract. We analyze a class of time discretizations for solving the nonlinear Schrödinger equation with non-smooth potential and at low-regularity on an arbitrary Lipschitz domain  $\Omega \subset \mathbb{R}^d$ ,  $d \leq 3$ . We show that these schemes, together with their optimal local error structure, allow for convergence under lower regularity assumptions on both the solution and the potential than is required by classical methods, such as splitting or exponential integrator methods. Moreover, we show first and second order convergence in the case of periodic boundary conditions, in any fractional positive Sobolev space  $H^r$ ,  $r \geq 0$ , beyond the more typical  $L^2$  or  $H^{\sigma}(\sigma > \frac{d}{2})$ -error analysis. Numerical experiments illustrate our results.

## 2.1 Introduction

We consider the Gross-Pitaevskii (GP) equation

$$i\partial_t u(t,x) = -\Delta u(t,x) + V(x)u(t,x) + |u(t,x)|^2 u(t,x), \quad (t,x) \in \mathbb{R} \times \Omega$$

$$(2.1)$$

with  $\Omega \subset \mathbb{R}^d$ ,  $d \leq 3$ , and an initial condition

$$u_{|t=0} = u_0. (2.2)$$

When  $\partial \Omega \neq \emptyset$ , we assign boundary conditions which will be encoded in the choice of the domain of the operator  $\mathcal{L} = i\Delta$ . We recall that the linear operator  $\mathcal{L} = i\Delta$  generates a group  $\{e^{t\mathcal{L}}\}_{t\in\mathbb{R}}$  of unitary operators on  $L^2(\Omega)$ . We will deal with mild solutions of the initial value problem (2.1) and (2.2) which are given by Duhamel's formula;

$$u(t) = e^{it\Delta}u_0 + \int_0^t e^{i(t-\zeta)\Delta} f(u,\bar{u},V)(\zeta,x)d\zeta$$
(2.3)

where we denote the nonlinearity by

$$f(u,\bar{u},V)(t,x) = -i(V(x)u(t,x) + u^{2}(t,x)\bar{u}(t,x)).$$
(2.4)

Throughout this article we will be interested in studying numerical schemes which approximate the time dynamics of (2.1) at low-regularity, by means of appropriate approximations of Duhamel's formula. Namely, we are interested in providing a reliable approximation of (2.1) (or equivalently of (2.3)) when the initial data  $u_0$  and the potential V are non-smooth, in the sense that they belong to Sobolev spaces of low order.

One setting for the Gross-Pitaevskii equation is to describe the dynamics of Bose-Einstein condensates in a potential trap. In many physically relevant situations the potential is assumed to be rough or disordered, and hence the study of equation (2.1) in this non-smooth or low-regularity framework is of physical interest ([NBP13], [WWW<sup>+</sup>98]).

Recently much progress has been made in the development of low-regularity approximations to nonlinear evolution equations. First, in the case of periodic boundary conditions a class of schemes called *Fourier integrators* [ORS21] or *resonance based schemes* [BS22] were introduced to approximate the time dynamics of dispersive equations such as NLS, KdV, and Klein-Gordon (see [HS17b], [OS18], [CS22a]). Recently, higher order extensions of these *resonance based* 

schemes were introduced in [BS22] for approximating in a unified fashion a large class of dispersive equations with periodic boundary conditions. These resonance based schemes were shown to converge in a more general setting, namely under lower regularity assumptions, than classical methods required (see [OS18], [ORS21] and references therein for a comparative analysis). The name of these schemes is due to their construction which revolves around Fourier based expansions of the solution and of the resonance structure of the equation. We explain the idea behind these resonance based schemes in detail in Section 2.3.1. These ideas were then extended in [RS21] to treat more general domains  $\Omega \subset \mathbb{R}^d$ and boundary conditions and allow to deal with a class of parabolic, hyperbolic and dispersive equations in a unified fashion. The resulting schemes were termed low-regularity integrators, or Duhamel's integrators (see [RS21]). A next natural step in this study of low-regularity approximations to nonlinear PDEs is to introduce a potential term uV with minimal regularity assumptions on the solution u and the potential V. The goal of this article is to study first and second low-regularity schemes for equation (2.1), with an emphasis on the error analysis. In Chapter 3 we present a general framework for deriving low-regularity schemes up to arbitrary order, using new techniques based on decorated trees series analysis to extend the construction of the schemes presented in this article.

In this article we study a class of low-regularity integrators to solve the Gross-Pitaevskii equation (2.1) on an arbitrary domain  $\Omega \subset \mathbb{R}^d$ . In the case where the domain is a torus  $\mathbb{T}^d$ , we state and prove first and second order convergence in any fractional positive Sobolev space  $H^r$ , under moderate regularity assumptions on both the solution u and the potential V. These are stronger convergence results than the more typical  $L^2$  or  $H^\sigma$  ( $\sigma > \frac{d}{2}$ ) -convergence analysis, and apply to the nonlinear Schrödinger equation as an immediate consequence.

#### 2.1.1 First-order low regularity integrator

In Section 2.3.2 we construct the following first order low-regularity integrator on  $\Omega$ . For  $n \geq 0$ , we define,

$$u^{n+1} = \Phi_1^{\tau}(u^n) := e^{i\tau\Delta} \left( u^n - i\tau (u^n \varphi_1(-i\tau\Delta)V + (u^n)^2 \varphi_1(-2i\tau\Delta)\bar{u}^n) \right), \quad \text{where } u^0 = u_0, \tag{2.5}$$

and  $\varphi_1(z) = \frac{e^z - 1}{z}$  is a bounded operator on  $i\mathbb{R}$ . The construction of this scheme does not rely on Fourier based techniques, and hence one can couple the above time discretization not only with spectral methods but with more general types of spatial discretizations. Indeed, on general domains one can call upon Krylov space methods for the approximation of the matrix exponential  $e^{it\Delta}$ , and the action of the  $\varphi_1(\cdot)$  functions (see [GRT18] and [HO10]). The fully discrete analysis on a smooth bounded domain with homogeneous Dirichlet conditions and with a finite elements space discretization, is the objective of future work.

We prove in Section 2.3.4 the following first order convergence result for the scheme (2.5), in the case of periodic boundary conditions. For the local-wellposedness result of (2.1) and (2.2) we refer to Theorem 2.2.1 given in Section 2.2.

**Theorem 2.1.1** Let  $T > 0, r \ge 0, and$ 

$$r_1 := \begin{cases} 1+r, & \text{if } r > \frac{d}{2}, \\ 1+\frac{r}{2}+\frac{d}{4}, & \text{if } 0 \le r < \frac{d}{2}. \end{cases}$$
(2.6)

For every  $u_0 \in H^{r_1}(\mathbb{T}^d)$  and  $V \in H^{r_1}(\mathbb{T}^d)$ , let  $u \in \mathcal{C}([0,T], H^{r_1}(\mathbb{T}^d))$  be the unique solution of (2.1). Then there exists  $\tau_{\min} > 0$  and  $C_T > 0$  such that for every time step size  $\tau \leq \tau_{\min}$  the numerical solution  $u^n$  given in equation (2.5) has the following error bound:

$$||u^n - u(n\tau)||_{H^r} \le C_T \tau, \quad 0 \le n\tau \le T.$$

Before moving on to the second order scheme and its convergence result we make a few remarks on the regularity assumptions made in the above theorem. A consequence of Theorem 2.1.1 is that for any initial data and potential in  $H^{r+1}(\Omega)$  where  $r > \frac{d}{2}$  and  $\Omega = \mathbb{T}^d$  (or on the full space  $\Omega = \mathbb{R}^d$ ) we have the following global error estimate:

$$\max_{1 \le n\tau \le T} \|u^n - u(n\tau)\|_{H^r} \le C(\sup_{[0,T]} ||u(t)||_{H^{r+1}}, ||V||_{H^{r+1}})\tau.$$

Namely we only ask one additional Sobolev derivative on the initial data  $u_0$  and the potential V in order to obtain first-order convergence of our low-regularity scheme (2.5). This is due to the favorable local error structures that these low-regularity schemes inherit. See ([OS18], [BS22], [RS21]), and references therein for an in depth comparative analysis of these low-regularity schemes with classical methods such as splitting methods, or exponential integrator methods.

Secondly, in the case  $0 < r < \frac{d}{2}$  the convergence analysis in  $H^r$ -norm of a time discretization of equation (2.1) has not to our knowledge previously been studied, and these are the first convergence results in this regime. A direct consequence of the above theorem is that in the critical case  $r = \frac{d}{2}$ , we obtain first order convergence in  $H^{\frac{d}{2}}$  with  $r_1 = 1 + \frac{d}{2} + \epsilon$ , for arbitrarily small  $\epsilon > 0$ .

Finally, when r = 0, we consider the regularity assumptions required for an  $L^2$ -error analysis, and compare them with the existing  $L^2$  convergence results for the Gross-Pitaevskii equation (2.1). When r = 0, Theorem 2.1.1 states,

$$||u^{n} - u(n\tau)||_{L^{2}} \le C(\sup_{[0,T]} ||u(t)||_{H^{1+\frac{d}{4}}}, ||V||_{H^{1+\frac{d}{4}}})\tau, \quad 0 \le n\tau \le T.$$

To our knowledge, this is the first convergence result of this type with low-regularity assumptions on both the solution u(t)and the potential V. Indeed, in the literature  $L^2$ -convergence results have been established for smooth potentials. See, for example [JL00], where the authors showed first-order convergence of a Lie splitting scheme for the linear Schrödinger equation with a potential term uV where they require V to be a  $C^5$ -smooth potential. The authors of [HP17] were able to show first-order convergence to (2.1) of a Crank-Nicholson scheme for a rough, discontinuous potential V, which is of physical relevance in the context of Bose-Einstein condensates. Namely, for  $V = V_d + V_s$ , where  $V_s \in C_0^{\infty}(\Omega)$  is smooth perturbation of a disordered potential  $V_d \in L^{\infty}(\Omega)$ , the authors obtained first order convergence of their scheme under -among other assumptions-  $u_t \in L^2(0,T; H^2(\Omega))$ , [HP17, Theorem 4.1]. Further, as detailed in [HP17, Appendix A], due to the roughness of  $V_d$  the highest regularity assumption one can hope for on the solution is  $u(t) \in H^2(\Omega)$ , and this regularity is required for their error analysis. In contrast to these results Theorem 2.1.1 permits low-regularity assumptions simultaneously on both u(t) and V. Finally, we refer to [RS21, Corollary 20] where the authors show first order convergence in  $L^2(\Omega)$  of a low-regularity scheme for the nonlinear Schrödinger equation (NLS) while analogously asking for  $1 + \frac{d}{4}$  Sobolev regularity on the initial data.

#### 2.1.2 Second-order low regularity integrator

In [Section 2.4, Corollary 2.4.3] we derive the following second order low-regularity integrator on  $\Omega$ . For  $n \ge 0$ , we define,

$$u^{n+1} = \Phi_{2}^{\tau}(u^{n}) := e^{i\tau\Delta}u^{n} - i\tau e^{i\tau\Delta} \left( u^{n}\varphi_{1}(-i\tau\Delta)V + (u^{n})^{2}\varphi_{1}(-2i\tau\Delta)\bar{u}^{n} \right)$$

$$- i\tau \left( (e^{i\tau\Delta}u^{n})\varphi_{2}(-i\tau\Delta)(e^{i\tau\Delta}V) + (e^{i\tau\Delta}u^{n})^{2}\varphi_{2}(-2i\tau\Delta)e^{i\tau\Delta}\bar{u}^{n} \right)$$

$$+ i\tau e^{i\tau\Delta} \left( u^{n}\varphi_{2}(-i\tau\Delta)V + (u^{n})^{2}\varphi_{2}(-2i\tau\Delta)\bar{u}^{n} \right)$$

$$- \frac{\tau^{2}}{2} e^{i\tau\Delta} (|u^{n}|^{4}u^{n} + 3u^{n}|u^{n}|^{2}V - |u^{n}|^{2}u^{n}\bar{V} + u^{n}V^{2}),$$
(2.7)

where  $\varphi_2(z) = \frac{e^z - \varphi_1(z)}{z}$  is a bounded operator on  $i\mathbb{R}$ . We present in [Section 2.4.2, Corollary 2.4.8] yet another derivation of a low-regularity second order scheme for (2.1). We also offer in Chapter 3 a similar low-regularity scheme as above, using a different construction based on tree series analysis. As for the first order scheme, the above time discretization can be coupled with various spatial discretizations such as with finite elements.

We prove in Section 2.4.4 the following second order convergence result for the scheme (2.7), in the case of periodic boundary conditions.

**Theorem 2.1.2** Let T > 0,  $r \ge 0$ , and

$$r_2 := \begin{cases} 2+r, & \text{if } r > \frac{d}{2}, \\ 2+\frac{r}{2}+\frac{d}{4}, & \text{if } 0 \le r < \frac{d}{2}. \end{cases}$$
(2.8)

For every  $u_0 \in H^{r_2}(\mathbb{T}^d)$  and  $V \in H^{r_2}(\mathbb{T}^d)$ , let  $u \in \mathcal{C}([0,T], H^{r_2}(\mathbb{T}^d))$  be the unique solution of (2.1). Then there exists  $\tau_{\min} > 0$  and  $C_T > 0$  such that for every time step size  $\tau \leq \tau_{\min}$  the numerical solution given in equation (2.7) has the following error bound:

$$\max_{1 \le n \le N} \|u^n - u(n\tau)\|_{H^r} \le C_T \tau^2.$$
(2.9)

We comment on the regularity assumptions made in the above theorem. First, in the (smooth) regime  $r > \frac{d}{2}$ , we ask only for two additional derivatives on the initial data  $u_0$  and the potential V. A preceding second order convergence result has been established for the NLS equation by [OWY22] in this regime. Indeed, using a resonance-based approach the authors [OWY22] showed second order convergence in  $H^r$ , for  $r > \frac{d}{2}$ , and  $u_0 \in H^{r+2}$ , of a low-regularity scheme for NLS (given by equation (2.7) with V = 0). Secondly, to our knowledge this is the first convergence result in  $H^r$  of a second order time discretization of equation (2.1) in the intermediate regime  $0 < r < \frac{d}{2}$ . In the critical case  $r = \frac{d}{2}$ , second order convergence directly follows from the above theorem with  $r_2 = 2 + \frac{d}{2} + \epsilon$  and  $\epsilon > 0$  arbitrarily small. Finally, we compare once again our result to the  $L^2$ -convergence results obtained in the literature; we mention the authors [JL00] who show second order convergence of a Strang splitting scheme for a  $C^5$ -smooth potential. Whereas, the authors [HP17] obtain second order convergence of a Crank-Nicholson scheme for a smooth potential V and -among other assumptionsfor  $u_{tt} \in L^2(0,T; H^2(\Omega))$ . As mentioned previously, we emphasize that in contrast to previous results we establish convergence under low-regularity assumptions on both u and V.

#### 2.1.3 Outline of the paper

We motivate the construction of the first order low-regularity integrator in Section 2.3.1 by first deriving the scheme in the periodic setting. We then generalize to the construction of the low-regularity scheme for an arbitrary domain  $\Omega \subset \mathbb{R}^d$ . In Section 2.4 we introduce the second order low-regularity integrator and discuss stability issues which arise. We then propose two different approaches to guarantee the stability of our proposed scheme. Section 2.3 and 2.4 also include the local and global error analysis of the first and second low-regularity integrators. Finally, in Section 2.5 we present numerical experiments underlining our theoretical findings. In the next section we briefly introduce some notation and nonlinear estimates which are crucial for the local and global error analysis.

## 2.2 Notation and nonlinear estimates

We begin by establishing some notation used in the paper, starting by the definition of the norm used throughout the error analysis sections. Our analysis will be made in the periodic fractional Sobolev space,

$$H^{r}(\mathbb{T}^{d}) := \{ u = \sum_{k \in \mathbb{Z}^{d}} u_{k} \frac{e^{ikx}}{\sqrt{(2\pi)^{d}}} \in L^{2}(\mathbb{T}^{d}) : |u|_{r}^{2} \triangleq \sum_{k \in \mathbb{Z}^{d}} |k|^{2r} |u_{k}|^{2} < +\infty \}$$

which is endowed with the norm

$$\begin{aligned} ||u||_{H^r}^2 &= ||u||_{L^2(\mathbb{T}^d)}^2 + ||(-\Delta)^{r/2}u||_{L^2(\mathbb{T}^d)}^2 \\ &= \sum_{k \in \mathbb{Z}^d} (1+|k|^{2r})|u_k|^2, \end{aligned}$$

where  $u_k = \frac{1}{\sqrt{(2\pi)^d}} \int_{\mathbb{T}^d} u e^{-ikx} dx$ .

Throughout the remainder of this section we fix  $\sigma > \frac{d}{2}$ , and we restrict the class of initial data and potential to belong to the Sobolev space  $H^{\sigma}$ . We now present some nonlinear estimates which will be fundamental for our analysis. We separate our  $H^r$ -error analysis into two cases:  $0 \le r \le \frac{d}{2}$ , and  $r > \frac{d}{2}$ . In the case where  $0 \le r \le \frac{d}{2}$  we have,

$$||vw||_{H^r} \lesssim ||v||_{H^\sigma} ||w||_{H^r},$$
(2.10)

while in the regime  $r > \frac{d}{2}$  the above holds with  $\sigma = r$  (see for example [HS17a, equation (2.49)]). For completeness, we provide a proof of the above inequality (2.10) in the Appendix for both regimes of r. These estimates will be used frequently throughout the error analysis sections (see Sections 2.3.3, 2.3.4, 2.4.3, 2.4.4).

One can easily deduce from the inequality (2.10) the following estimates on the nonlinearity (2.4);

$$||f(w,\bar{w},V)||_{H^{r}} \leq C_{r,\sigma}(||w||_{H^{\sigma}},||V||_{H^{\sigma}})||w||_{H^{r}} ||f(w,\bar{w},V) - f(v,\bar{v},V)||_{H^{r}} \leq C_{r,\sigma}(||w||_{H^{\sigma}},||v||_{H^{\sigma}},||V||_{H^{\sigma}})||w - v||_{H^{r}},$$
(2.11)

where  $C_{r,\sigma}(||u||, ||v||, ||w||)$  denotes a generic constant which depends on the bounded arguments ||u||, ||v||, and ||w||. In the regime  $r > \frac{d}{2}$  the above holds with  $\sigma = r$ .

**Remark 2.2.1** In order to deal with less smooth initial data  $u_0 \in H^r$ ,  $r \leq \frac{d}{2}$  one cannot make use of the bilinear estimate (2.10) and one would need to call upon more subtle tools to show appropriate fractional convergence of the scheme. Several works have been made when working on  $\Omega = \mathbb{T}$  or  $\Omega = \mathbb{R}^d$  where low-regularity estimates for very rough data  $u_0 \in H^r$ ,  $r \leq \frac{d}{2}$  could be obtained by using tools from dispersive PDE such as discrete Strichartz estimates, or Bourgain spaces, see [ORS21], [ORS22b]. This refined error analysis is out of scope for this paper.

We finish this subsection by stating the following local well-posedness result of a solution to (2.1) and (2.2) of the form (2.3). Indeed, using the estimates (2.11), one obtains from a classical Banach fixed point argument the following result:

**Theorem 2.2.1** Let  $\sigma_0 > \frac{d}{2}$ . Given any  $u_0 \in H^{\sigma_0}(\mathbb{T}^d)$ , and  $V \in H^{\sigma_0}(\mathbb{T}^d)$  there exists T > 0 and a unique solution  $u \in C([0,T], H^{\sigma_0}(\mathbb{T}^d))$  to (2.1).

### 2.3 First order scheme and analysis

In this section we start by giving the main ideas behind the construction of the first order low-regularity scheme (see [RS21]). We propose a novel low-regularity integrator for the approximation of Duhamel's formula (2.3). We will approximate equation (2.3) at the time step  $t_n + \tau$ , where  $\tau$  is the time step size. By iterating Duhamel's formula (2.3), we obtain the first order iteration

$$u(t_n + \tau) = e^{i\tau\Delta} [u(t_n) - i\mathcal{J}_1(\tau, \Delta, u(t_n))] + R_{1,0}(\tau, u)$$
(2.12)

where the principal oscillatory integral (at first order) is given by

$$\mathcal{J}_1(\tau, \Delta, v) = \int_0^\tau e^{-i\zeta\Delta} [V(x)(e^{i\zeta\Delta}v) + (e^{i\zeta\Delta}v)^2(e^{-i\zeta\Delta}\bar{v})]d\zeta$$
(2.13)

and the remainder

$$R_{1,0}(\tau,u) = \int_0^\tau e^{i(\tau-\zeta)\Delta} [f(u(t_n+\zeta),\bar{u}(t_n+\zeta),V) - f(e^{i\zeta\Delta}u(t_n),e^{-i\zeta\Delta}\bar{u}(t_n),V)]d\zeta$$

We will construct a suitable discretization of the integral (2.13) to allow for a low-regularity approximation to the first order Duhamel iterate (2.12). The idea is to filter out the dominant parts, which we denote by  $\mathcal{L}_{dom}$ , of the nonlinear frequency interactions within the integral (2.13) and embed them in the discretization. The lower-order parts will be approximated and incorporated in the local error analysis.

First, to illustrate the underlying idea and to provide intuition behind the construction of these low-regularity integrators we start by analyzing the case of periodic boundary conditions  $\Omega = \mathbb{T}$ , with V a periodic potential. The ideas presented in the next section were first introduced by the authors [OS18] for solving a class of semilinear Schrödinger equations. After presenting the periodic case in a formal way, we rigorously detail in Section 2.3.2 the construction of the first order scheme in the more general case of an arbitrary domain  $\Omega \subset \mathbb{R}^d$ .

#### **2.3.1** Case of periodic boundary conditions: $\Omega = \mathbb{T}$

Assuming that  $v \in L^2$ , we can expand v in Fourier series  $v = \sum_{k \in \mathbb{Z}} \hat{v}_k e^{ikx}$ . This allows us to express the action of the Schrödinger flow on v,  $e^{\pm it\Delta}v(x) = \sum_{k \in \mathbb{Z}} \hat{v}_k e^{\mp itk^2} e^{ikx}$ . Similarly assuming  $V \in L^2$  we have  $V(x) = \sum_{l \in \mathbb{Z}} \hat{V}_l e^{ilx}$ . In Fourier space, the oscillatory integral (2.13) is then given by,

$$\mathcal{J}_1(\tau, \Delta, v) = \sum_{l_1+l_2=l} \hat{V}_{l_1} \hat{v}_{l_2} e^{ilx} \int_0^\tau e^{i\zeta R_2(l)} d\zeta + \sum_{-k_1+k_2+k_3=k} \bar{\hat{v}}_{k_1} \hat{v}_{k_2} \hat{v}_{k_3} e^{ikx} \int_0^\tau e^{i\zeta R_1(k)} d\zeta$$
(2.14)

with the resonance structure,

$$R_1(k) = 2k_1^2 - 2k_1(k_2 + k_3) + 2k_2k_3$$
, and  $R_2(l) = l_1^2 + 2l_1l_2$ . (2.15)

Ideally we would like to integrate all the nonlinear frequency interactions (2.15) exactly and embed them in the discretization. This, however, would result in a generalized convolution (of Coifman-Meyer type [CM75]), which cannot be rewritten in physical space. Hence, the computations would need to be fully made in Fourier space. Carrying this out in higher spatial dimensions d would cause large memory and computational efforts of order  $O(|K|^{d \cdot \ell})$ , where K denotes the highest frequency in the discretization and  $\ell$  is the number of factors in the nonlinearity. For practical computations, we want to be able to express the discretization also in physical space in order to use the Fast Fourier Transform (FFT) which is of computational effort of order  $O(|K|^d \log |K|^d)$ . Therefore, we choose in the following an approximation of the integral (2.14) which allows for a practical implementation (by not performing exact integration), while optimizing the local error in the sense of regularity. We detail this procedure below.

We can extract the dominant and lower-order parts from the resonance structures (2.15) by recalling that  $2k_1^2$  and  $l_1^2$  correspond to second order derivatives in Fourier while the terms  $k_m k_j$  (for  $m \neq j$ ) correspond to product of first order derivatives. We choose,

$$R_1(k) = \mathcal{L}_{dom,1}(k_1) + \mathcal{L}_{low,1}(k_1, k_2, k_3), \quad R_2(l) = \mathcal{L}_{dom,2}(l_1) + \mathcal{L}_{low,2}(l_1, l_2)$$

with

$$\mathcal{L}_{dom,1}(k_1) = 2k_1^2, \ \mathcal{L}_{low,1}(k_1, k_2, k_3) = -2k_1(k_2 + k_3) + 2k_2k_3, \text{ and}$$

 $\mathcal{L}_{dom,2}(l_1) = l_1^2, \ \mathcal{L}_{low,2}(l_1, l_2) = 2l_1 l_2.$ 

From the above and from equation (2.14), by a simple Taylor's expansion on the lower-order parts we (formally) allow for the following approximation of the oscillatory integral in Fourier space,

$$\begin{split} \mathcal{J}_{1}(\tau,\Delta,v) &= \sum_{l_{1}+l_{2}=l} \hat{V}_{l_{1}} \hat{v}_{l_{2}} e^{ilx} \int_{0}^{\tau} e^{i\zeta \mathcal{L}_{dom,2}(l_{1})} e^{i\zeta \mathcal{L}_{low,2}(l_{1},l_{2})} d\zeta \\ &+ \sum_{-k_{1}+k_{2}+k_{3}=k} \bar{\hat{v}}_{k_{1}} \hat{v}_{k_{2}} \hat{v}_{k_{3}} e^{ikx} \int_{0}^{\tau} e^{i\zeta \mathcal{L}_{dom,1}(k_{1})} e^{i\zeta \mathcal{L}_{low,1}(k_{1},k_{2},k_{3})} d\zeta \\ &= \sum_{l_{1}+l_{2}=l} \hat{V}_{l_{1}} \hat{v}_{l_{2}} e^{ilx} \int_{0}^{\tau} e^{i\zeta \mathcal{L}_{dom,2}(l_{1})} \left(1 + O(\zeta \mathcal{L}_{low,2}(l_{1},l_{2}))\right) d\zeta \\ &+ \sum_{-k_{1}+k_{2}+k_{3}=k} \bar{\hat{v}}_{k_{1}} \hat{v}_{k_{2}} \hat{v}_{k_{3}} e^{ikx} \int_{0}^{\tau} e^{i\zeta \mathcal{L}_{dom,1}(k_{1})} \left(1 + O(\zeta \mathcal{L}_{low,1}(k_{1},k_{2},k_{3}))\right) d\zeta. \end{split}$$

Mapping back into physical space we thus have

$$\mathcal{L}_{dom,1}(v) = -2\Delta v, \quad \mathcal{L}_{low,1}(v) = 2(2|\nabla v|^2 v - |\nabla v|^2 \bar{v}),$$
  
$$\mathcal{L}_{dom,2}(v) = -\Delta v, \quad \mathcal{L}_{low,2}(v,V) = -2\nabla V \nabla v,$$

and

$$\mathcal{J}_{1}(\tau, \Delta, v) = \int_{0}^{\tau} [e^{i\zeta \mathcal{L}_{dom,2}} V] v + [e^{i\zeta \mathcal{L}_{dom,1}} \bar{v}] v^{2} + O(\zeta(\mathcal{L}_{low,2}(v, V) + \mathcal{L}_{low,1}(v))) d\zeta$$

$$= \tau [v\varphi_{1}(i\tau \mathcal{L}_{dom,2})V + v^{2}\varphi_{1}(i\tau \mathcal{L}_{dom,1})\bar{v}] + O(\tau^{2}(\mathcal{L}_{low,2}(v, V) + \mathcal{L}_{low,1}(v))).$$
(2.16)

Hence, for a small time step  $\tau$ , by plugging the above expression of  $\mathcal{J}_1$  in the iterate (2.12) and ignoring the lower-order terms yields the first-order resonance based discretization

$$u^{n+1} = e^{i\tau\Delta} [u^n - i\tau (u^n \varphi_1(-i\tau\Delta)V + (u^n)^2 \varphi_1(-2i\tau\Delta)\bar{u}^n)].$$
(2.17)

The above scheme (2.17) has a favorable local error structure; namely from equation (2.16) we see that (formally) this discretization only ask for *first order* derivatives on the initial data and potential.

We now place ourselves in the general framework  $\Omega \subset \mathbb{R}^d$ , and make use of filtering techniques to recover the first order low-regularity approximation (2.17) in this general setting. The ideas presented in the next section are inspired by the work of [RS21].

### **2.3.2** General boundary conditions: $\Omega \subset \mathbb{R}^d$

The goal of this section is to construct a first order discretization of the oscillatory integral (2.13) when working on a general domain  $\Omega$ , and which allows for the improved local error structure (2.16) established in the preceding section. This is achieved by introducing a properly chosen filtered function which will filter out the dominant oscillatory terms  $\mathcal{L}_{dom,1}, \mathcal{L}_{dom,2}$  explicitly found in the preceding section.

First, we recall the definition of the commutator-type term C[H, L] for  $H(v_1, \dots, v_n)$ ,  $n \ge 1$ , a function and L a linear operator:

$$\mathcal{C}[H,L](v_1,\cdots,v_n) = -L(H(v_1,\cdots,v_n)) + \sum_{i=1}^n D_i H(v_1,\cdots,v_n) \cdot Lv_i.$$

We make an important note that the above differs from the well known Lie commutator used for the error analysis of classical methods, such as for splitting methods (see for example [HLW10], [JL00]).

We define the filtered function by

$$\mathcal{N}(\tau, s, \zeta, \Delta, v) = e^{-is\Delta} [e^{is\Delta} e^{-i\zeta\Delta} V(e^{is\Delta} v) + (e^{is\Delta} v)^2 (e^{is\Delta} e^{-2i\zeta\Delta} \bar{v})].$$
(2.18)

The principal oscillations (2.13) can be expressed with the aid of the filter function  $\mathcal{N}$  as

$$\mathcal{J}_1(\tau, \Delta, v) = \int_0^\tau \mathcal{N}(\tau, \zeta, \zeta, \Delta, v) d\zeta$$

By the fundamental theorem of calculus we have

$$\mathcal{J}_1(\tau, \Delta, v) = \int_0^\tau \mathcal{N}(\tau, 0, \zeta, v) d\zeta + \int_0^\tau \int_0^\zeta \partial_s \mathcal{N}(\tau, s, \zeta, v) ds d\zeta$$
(2.19)

where

$$\mathcal{N}(\tau, 0, \zeta, v) = [e^{i\zeta \mathcal{L}_{dom,2}}V]v + [e^{i\zeta \mathcal{L}_{dom,1}}\bar{v}]v^2$$
(2.20)

and

$$\partial_s \mathcal{N}(\tau, s, \zeta, v) = e^{-is\Delta} \mathcal{C}[f, i\Delta] (e^{is\Delta} v, e^{is\Delta} e^{-2i\zeta\Delta} \bar{v}, e^{is\Delta} e^{-i\zeta\Delta} V),$$
$$\mathcal{C}[f, i\Delta](u, v, w) = -2i(\nabla w \cdot \nabla u + |\nabla u|^2 v + \nabla(u^2) \cdot \nabla v).$$

Hence, we recover the discretization of the oscillatory integral (2.13) together with an improved local error structure of the form (2.16);

$$\mathcal{J}_1(\tau, \Delta, v) = \tau [v\varphi_1(i\tau \mathcal{L}_{dom,2})V + v^2 \varphi_1(i\tau \mathcal{L}_{dom,1})\bar{v}] + R_{1,1}(\tau)$$

where

$$R_{1,1}(\tau) = \int_0^\tau \int_0^\zeta e^{-is\Delta} \mathcal{C}[f, i\Delta](e^{is\Delta}v, e^{is\Delta}e^{-2i\zeta\Delta}\bar{v}, e^{is\Delta}e^{-i\zeta\Delta}V) dsd\zeta$$

**Corollary 2.3.1** The exact solution u of (2.1) can be expanded as

$$u(t_n + \tau) = e^{i\tau\Delta} [u(t_n) - i\tau(u(t_n)\varphi_1(-i\tau\Delta)V + (u(t_n))^2\varphi_1(-2i\tau\Delta)\bar{u}(t_n))] + \mathcal{R}_1(\tau, t_n)$$

where the remainder is given by

$$\mathcal{R}_{1}(\tau, t_{n}) = \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} [f(u(t_{n}+\zeta), \bar{u}(t_{n}+\zeta), V) - f(e^{i\zeta\Delta}u(t_{n}), e^{-i\zeta\Delta}\bar{u}(t_{n}), V)] d\zeta + \int_{0}^{\tau} \int_{0}^{\zeta} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta] (e^{is\Delta}u(t_{n}), e^{is\Delta}e^{-2i\zeta\Delta}\bar{u}(t_{n}), e^{is\Delta}e^{-i\zeta\Delta}V) dsd\zeta.$$
(2.22)

The first order low-regularity scheme (2.5) follows from the above Corollary 2.3.1 by neglecting the remainder  $\mathcal{R}_1(\tau, t_n)$ . We next show the first-order error estimates for the scheme (2.5) by first estimating it's favorable commutator-type local error structure.

#### 2.3.3 Local error estimates

**Proposition 2.3.2** Let T > 0,  $r \ge 0$ , and  $r_1$  as in Theorem 2.1.1, namely

$$r_1 = \begin{cases} 1+r, & \text{if } r > \frac{d}{2}, \\ 1+\frac{r}{2}+\frac{d}{4}, & \text{if } 0 \le r \le \frac{d}{2} \end{cases}$$

Assume there exists  $C_T > 0$  such that

$$\sup_{[0,T]} ||u(t)||_{H^{r_1}} \le C_T, \text{ and } ||V||_{H^{r_1}} \le C_T,$$
(2.23)

then there exists  $M_T > 0$  such that for every  $\tau \in (0, 1]$ ,

$$||\mathcal{R}_1(\tau, t_n)||_{H^r} \le M_T \tau^2, \qquad 0 \le t_n \le T,$$
 (2.24)

where  $t_n = n\tau$  and  $\mathcal{R}_1(\tau, t_n)$  is given in equation (2.22).

*Proof.* We write the error term  $\mathcal{R}_1(\tau, t_n)$ , defined in equation (2.22), as the sum of two terms,  $\mathcal{R}_1(\tau, t_n) = \mathcal{G}^1(\tau, t_n) + \mathcal{G}^2(\tau, t_n)$ . We begin by estimating the first term  $\mathcal{G}^1(\tau, t_n)$ . Using the inequalities (2.11) on f, and the boundedness of

(2.21)

 $e^{it\Delta}$  on Sobolev spaces we have that for all  $r \ge 0$  and some  $\sigma > \frac{d}{2}$ ,

$$\begin{aligned} ||\mathcal{G}^{1}(\tau,t_{n})||_{H^{r}} &\leq \tau \sup_{\zeta \in [0,\tau]} ||f(u(t_{n}+\zeta),\bar{u}(t_{n}+\zeta),V) - f(e^{i\zeta\Delta}u(t_{n}),e^{-i\zeta\Delta}\bar{u}(t_{n}),V)||_{H^{r}} \\ &\leq \tau C_{r,\sigma}(\sup_{[0,T]}||u(t)||_{H^{\sigma}},||V||_{H^{\sigma}}) \sup_{\zeta \in [0,\tau]} ||u(t_{n}+\zeta) - e^{i\zeta\Delta}u(t_{n})||_{H^{r}} \\ &\leq \tau C_{r,\sigma}(\sup_{[0,T]}||u(t)||_{H^{\sigma}},||V||_{H^{\sigma}}) \sup_{\zeta \in [0,\tau]} ||\int_{0}^{\zeta} e^{i(\zeta-s)\Delta}f(u(t_{n}+s),\bar{u}(t_{n}+s),V)ds||_{H^{r}} \\ &\leq \tau^{2} C_{r,\sigma}(\sup_{[0,T]}||u(t)||_{H^{\sigma}},||V||_{H^{\sigma}}) \sup_{s \in [0,\tau]} ||f(u(t_{n}+s),\bar{u}(t_{n}+s),V)||_{H^{r}} \\ &\leq C_{r,\sigma}(\sup_{[0,T]}||u(t)||_{H^{\sigma}},\sup_{[0,T]}||u(t)||_{H^{r}},||V||_{H^{\sigma}})\tau^{2}, \end{aligned}$$

where we use Duhamel's formula to go from the third to the fourth line. First, in the regime  $r > \frac{d}{2}$ , the above holds with  $\sigma = r$ . Using the fact that  $r_1 > r$  yields the desired bound  $||\mathcal{G}^1(\tau, t_n)||_{H^r} \leq C_T \tau^2$  in this regime. When  $r < \frac{d}{2}$ , we will construct an appropriate  $\sigma > \frac{d}{2}$  which will be used throughout the remainder of the proof when making the analysis in this non-smooth regime.

Let  $0 < \epsilon < \frac{1}{4}$ , and let

$$\sigma_0 = \frac{d}{2} + \frac{\epsilon}{2}.\tag{2.26}$$

For  $d \leq 3$ , we have that  $\frac{d}{2} + \epsilon < 1 + \frac{d}{4}$ , and hence  $\sigma_0$  satisfies

$$\frac{d}{2} < \sigma_0 < \frac{d}{2} + \epsilon < 1 + \frac{d}{4}.$$
(2.27)

Consequently, by recalling the definition (2.6) of  $r_1$ , we have that  $r_1 > \sigma_0$  when  $r < \frac{d}{2}$ . Hence, in the regime  $r < \frac{d}{2}$ , since  $r_1 > \sigma_0$  we obtain the desired bound  $||\mathcal{G}^1(\tau, t_n)||_{H^r} \leq C_T \tau^2$  by taking  $\sigma = \sigma_0$  in equation (2.25).

We now estimate the second term  $\mathcal{G}^2(\tau, t_n)$  in the remainder (2.22), using the explicit expression of the commutator (2.21) and the nonlinear estimate (2.10). In the regime  $r > \frac{d}{2}$ , we have  $r_1 = r + 1$  and

$$\begin{aligned} ||\mathcal{C}[f, i\Delta](u, v, w)||_{H^r} &\leq C_r(||\nabla w \cdot \nabla u||_{H^r} + |||\nabla u|^2 v||_{H^r} + 2||u\nabla u \cdot \nabla v||_{H^r}) \\ &\leq C_r(||u||_{r+1}, ||v||_{r+1}, ||w||_{r+1}) \\ &\leq C_r(||u||_{r_1}, ||v||_{r_1}, ||w||_{r_1}). \end{aligned}$$

When  $0 \le r < \frac{d}{2}$ , we will make use of the bilinear estimate

$$\|uv\|_{H^r} \le C_{r,d} \|u\|_{H^{\frac{r}{2}+\frac{d}{4}}} \|v\|_{H^{\frac{r}{2}+\frac{d}{4}}}, \tag{2.28}$$

which is a particular case of [Hör97, Theorem 8.3.1]. It follows from the above estimate (2.28) and the inequality (2.10) with  $\sigma = \sigma_0$  as defined in (2.26) that

$$\begin{aligned} ||\mathcal{C}[f, i\Delta](u, v, w)||_{H^{r}} &\leq C_{r, d}(||\nabla w||_{H^{\frac{r}{2} + \frac{d}{4}}}, ||\nabla u||_{H^{\frac{r}{2} + \frac{d}{4}}}, ||v||_{H^{\sigma_{0}}}, ||u||_{H^{\sigma_{0}}}, ||\nabla v||_{H^{\frac{r}{2} + \frac{d}{4}}}) \\ &\leq C_{r, d}(||w||_{H^{1 + \frac{r}{2} + \frac{d}{4}}}, ||u||_{H^{1 + \frac{r}{2} + \frac{d}{4}}}, ||v||_{H^{1 + \frac{r}{2} + \frac{d}{4}}}). \end{aligned}$$

Hence, by definition (2.6) of  $r_1$ , given any  $r \in \mathbb{R}^+ \setminus \{\frac{d}{2}\}$ , we have shown the following bound,

$$||\mathcal{C}[f, i\Delta](u, v, w)||_{H^r} \le C_{r,d}(||u||_{r_1}, ||v||_{r_1}, ||w||_{r_1}).$$
(2.29)

Further, since  $e^{is\Delta}$  is an isometry on Sobolev spaces we obtain the following estimate of  $\mathcal{G}^2$  in  $H^r$  norm,

$$||\mathcal{G}^{2}(\tau, t_{n})||_{H^{r}} \leq C_{r,d}(\sup_{[0,T]} ||u(t)||_{H^{r_{1}}}, ||V||_{H^{r_{1}}})\tau^{2}, \quad r \in \mathbb{R}^{+} \setminus \{\frac{d}{2}\}$$

The local error estimate is hence demonstrated.

**Remark 2.3.3** By following the proof of Proposition 2.3.2 one can ask for less Sobolev regularity on the potential V, while asking for more regularity on the solution u(t). Indeed, for example, by making the analysis in the  $L^2$ -norm, one can ask for  $V \in H^1$ , and  $u(t) \in H^{1+\frac{d}{2}+\epsilon}$ , with  $\epsilon > 0$  arbitrarily small.

#### 2.3.4 Global error estimates

Using the local error estimates established in the preceding section we show global first order convergence of our scheme (2.5) under the favorable regularity assumptions on the initial condition and the potential established previously.

Proof of Theorem 2.1.1. Let  $e^n = u^n - u(t_n)$ , where  $u^n = \Phi_1^{\tau}(u^{n-1})$  is given in equation (2.5). We begin by decomposing the error term as follows,

$$||e^{n+1}||_{H^r} \le ||\Phi_1^{\tau}(u(t_n)) - u(t_{n+1})||_{H^r} + ||\Phi_1^{\tau}(u^n) - \Phi_1^{\tau}(u(t_n))||_{H^r}.$$
(2.30)

The first term of the above expression is given by the local error  $\mathcal{R}_1(\tau, t_n)$  defined at equation (2.22), and which is of order  $\tau^2$  by Proposition 2.3.2. We wish to establish a stability estimate of the numerical flow  $\Phi_1^{\tau}$  to bound the second term in equation (2.30), and to conclude by a *Lady Windermere's fan* argument ([HLW10]).

By using the estimate and (2.10), together with the fact that  $e^{i\xi\Delta}$  and  $\varphi_1(i\xi\Delta)$  are bounded on Sobolev spaces (for all  $\xi \in \mathbb{R}$ ), it easily follows from the definition of our scheme (2.5) that for all  $r \geq 0$ ,

$$||\Phi_1^{\tau}(u(t_n)) - \Phi_1^{\tau}(u^n)||_{H^r} \le e^{L_n \tau} ||e^n||_{H^r}, \quad L_n := C(||u^n||_{H^{\sigma}}, ||u(t_n)||_{H^{\sigma}}, ||V||_{H^{\sigma}}),$$
(2.31)

where for  $r > \frac{d}{2}$  we have  $\sigma = r$ , and for  $r \le \frac{d}{2}$ , we have  $\sigma > \frac{d}{2}$ .

Using Proposition 2.3.2, and the estimate (2.31) a bound of the error term (2.30) is given by,

$$||e^{n+1}||_{H^r} \le M_T \tau^2 + e^{L_n \tau} ||e^n||_{H^r}, \quad e^0 = 0, \quad r \ne \frac{d}{2}.$$
(2.32)

The global error estimate then easily follows by induction on the above inequality (2.32) once the following uniform bound is obtained:

$$\sup_{n\tau \le T} ||u^n||_{H^{\sigma}} < +\infty, \tag{2.33}$$

for sufficiently small time step  $\tau$ , which depends on the final time T and the norm of the initial condition  $||u_0||_{H^{\sigma}}$ . In the remainder of the proof we establish the bound (2.33) for appropriate choices of  $\sigma$  depending on the  $H^r$ -norm considered.

In the regime  $r > \frac{d}{2}$  we take  $\sigma = r$ , and the result follows by the classical Lady Windermere's fan argument ([HLW10]). Indeed, the uniform bound (2.33) easily follows for sufficiently small  $\tau$  by a bootstrap argument on the estimate (2.32).

Using a refined global error analysis one can push down the error analysis to the  $H^r$ -norm for  $0 \le r < \frac{d}{2}$ . We take  $\sigma = \sigma_0$  where  $\sigma_0$  is given in equation (2.26). In order to show the uniform bound (2.33) we establish fractional convergence of the scheme (2.5) in the higher order Sobolev space  $H^{\sigma_0}$ . Namely, we show that there exists  $\delta > 0$  such that the following estimate holds

$$||e^{n+1}||_{H^{\sigma_0}} \le M_T \tau^{1+\delta} + e^{L_n \tau} ||e^n||_{H^{\sigma_0}}.$$
(2.34)

where  $L_n = C(||u^n||_{H^{\sigma_0}}, ||u(t_n)||_{H^{\sigma_0}}, ||V||_{H^{\sigma_0}})$ . Using the decomposition (2.30), and the bound (2.31) with  $r = \sigma_0$ , we are left to show the following local error estimate,

$$||\mathcal{R}_1(\tau, t_n)||_{H^{\sigma_0}} \le M_T \tau^{1+\delta},$$
 (2.35)

in order to obtain the bound (2.34). We obtain the bound (2.35) by an interpolation argument. We first show a bound on the remainder  $\mathcal{R}_1(\tau, t_n)$  in  $H^{r_1}$ -norm. By using Duhamel's formula and by construction of our numerical scheme (2.5) we have

$$\mathcal{R}_1(\tau, t_n) = \int_0^\tau e^{i(\tau-s)\Delta} f(u(t_n) + s, \overline{u}(t_n) + s, V) ds - i\tau e^{i\tau\Delta} \big( u(t_n)\varphi_1(-i\tau\Delta)V + (u(t_n))^2 \varphi_1(-2i\tau\Delta)\overline{u}(t_n) \big).$$

One can estimate each of the above terms separately using the first estimate in equation (2.11) with  $r = \sigma = r_1$ . Indeed, this yields

$$\begin{aligned} \|\mathcal{R}_{1}(\tau, t_{n})\|_{H^{r_{1}}} &\leq C_{r_{1}}(\|u(t_{n})\|_{H^{r_{1}}}, \|V\|_{H^{r_{1}}})\tau \\ &\leq C_{r_{1},T}\tau, \end{aligned}$$
(2.36)

Finally, since  $r_1 > \sigma_0 > r$  there exists  $\theta \in (0, 1)$  such that

 $||\mathcal{R}_1(\tau, t_n)||_{H^{\sigma_0}} \le ||\mathcal{R}_1(\tau, t_n)||_{H^{r_1}}^{\theta} ||\mathcal{R}_1(\tau, t_n)||_{H^r}^{1-\theta}.$ 

Using the local error estimates (2.24) and (2.36) we have

$$||\mathcal{R}_1(\tau, t_n)||_{H^{\sigma_0}} \le M_T \tau^{2-\theta}$$

where  $2 - \theta > 0$ . Hence we have shown the bounds (2.35) and (2.34) with  $\delta = 1 - \theta$ . This yields the desired bound

(2.33) with  $\sigma = \sigma_0$ , and for sufficiently small  $\tau$ , by a classical bootstrap argument on equation (2.34). The first order convergence of the scheme (2.5) follows by induction using the global bound (2.32), with  $\sigma = \sigma_0$ .

## 2.4 Second order scheme and analysis

The idea to derive a higher order scheme is to iterate Duhamel's formula (2.3), and to Taylor expand f around  $e^{i\tau\Delta}v$ , where we let  $v = u_0$ . For a second order scheme this yields the following expansion,

$$u(\tau) = e^{i\tau\Delta} \bigg[ v - i\mathcal{J}_1(\tau, \Delta, v) - i\int_0^\tau e^{-i\zeta_1\Delta} [D_1 f(e^{i\zeta_1\Delta}v, e^{-i\zeta_1\Delta}\bar{v}, V) \cdot e^{i\zeta_1\Delta}\mathcal{J}_1(\zeta_1, \Delta, v)] d\zeta_1$$

$$+ i\int_0^\tau e^{-i\zeta_1\Delta} [D_2 f(e^{i\zeta_1\Delta}v, e^{-i\zeta_1\Delta}\bar{v}, V) \cdot e^{-i\zeta_1\Delta}\overline{\mathcal{J}_1(\zeta_1, \Delta, v)}] d\zeta_1 \bigg] + O(\tau^3)$$
(2.37)

where

$$D_1 f(v, \bar{v}, V) = -i(V + 2v\bar{v}), \text{ and } D_2 f(v, \bar{v}, V) = -iv^2,$$
(2.38)

and where we Taylor expanded f around  $e^{i\tau\Delta}v$  up to second order in order to obtain a remainder of order three. Next, the aim is to derive a second order approximation to the integrals appearing in the above expansion (2.37).

First, we treat the iterated integrals appearing in the above expression, namely the third and fourth term in equation (2.37). By a standard Taylor expansion we linearize the exponentials appearing in these iterated integrals. For both  $v, V \in H^2$ , this yields

$$-ie^{i\zeta_1\Delta}\mathcal{J}_1(\zeta_1,\Delta,v) = \zeta_1 f(v,\bar{v},V) + O(\zeta_1^2(\Delta v + \Delta V))$$

Using the above we make the following second order approximation of the iterated integrals in equation (2.37);

$$-i\int_{0}^{\tau} e^{-i\zeta_{1}\Delta} [D_{1}f(e^{i\zeta_{1}\Delta}v, e^{-i\zeta_{1}\Delta}\bar{v}, V) \cdot e^{i\zeta_{1}\Delta}\mathcal{J}_{1}(\zeta_{1}, \Delta, v)]d\zeta_{1} = \int_{0}^{\tau} \zeta_{1}D_{1}f(v, \bar{v}, V) \cdot f(v, \bar{v}, V)d\zeta_{1}$$

$$+ O(\tau^{3}(\Delta v + \Delta V)),$$

$$i\int_{0}^{\tau} e^{-i\zeta_{1}\Delta} [D_{2}f(e^{i\zeta_{1}\Delta}v, e^{-i\zeta_{1}\Delta}\bar{v}, V) \cdot e^{-i\zeta_{1}\Delta}\overline{\mathcal{J}_{1}(\zeta_{1}, \Delta, v)}]d\zeta_{1} = \int_{0}^{\tau} \zeta_{1}D_{2}f(v, \bar{v}, V) \cdot \overline{f(v, \bar{v}, V)}d\zeta_{1}$$

$$+ O(\tau^{3}(\Delta v + \Delta V)).$$

$$(2.39)$$

The above calculations motivate the choice of the expansion for u stated in the following lemma.

**Lemma 2.4.1** Let  $v = u^0$ . At second order u can be expanded as

$$u(\tau) = u_2(\tau) + R_{2,0}(\tau)$$

with

$$u_{2}(\tau) = e^{i\tau\Delta}v - ie^{i\tau\Delta}\mathcal{J}_{1}(\tau,\Delta,v) - \frac{\tau^{2}}{2}e^{i\tau\Delta}(|v|^{4}v + 3v|v|^{2}V - |v|^{2}v\bar{V} + vV^{2}),$$

and

$$\begin{aligned} R_{2,0}(\tau) &= \int_0^\tau e^{i(\tau-\zeta_1)\Delta} [f(u(\zeta_1),\bar{u}(\zeta_1),V) - f(e^{i\zeta_1}v,e^{-i\zeta_1}\bar{v},V)] d\zeta_1 \\ &+ e^{i\tau\Delta} \int_0^\tau \zeta_1 (|v|^4v + 3v|v|^2V - |v|^2v\bar{V} + vV^2) d\zeta_1. \end{aligned}$$

*Proof.* The result immediately follows by recalling the definition of the principal oscillations (2.13), and Duhamel's formula (2.3). Moreover, we note that by simple calculations one has

$$D_1 f(v, \bar{v}, V) \cdot f(v, \bar{v}, V) = -(V^2 v + 3v^2 \bar{v} V + 2v^3 \bar{v}^2), \quad D_2 f(v, \bar{v}, V) \cdot \overline{f(v, \bar{v}, V)} = \bar{V} \bar{v} v^2 + v^3 \bar{v}^2,$$

and hence by equation (2.38) we have,

$$D_1 f(v, \bar{v}, V) \cdot f(v, \bar{v}, V) + D_2 f(v, \bar{v}, V) \cdot \overline{f(v, \bar{v}, V)} = -(|v|^4 v + 3v|v|^2 V - |v|^2 v \bar{V} + v V^2).$$
(2.40)

The above calculations together with equations (2.37) and (2.39) motivates the inclusion of the last term in the expansion of  $u_2(\tau)$ .

It remains to establish a low-regularity second order approximation to the principal oscillatory integral (2.13). We first recall from Section 2.3.2 that in order to derive a low-regularity approximation of  $\mathcal{J}_1$  at first order we used the filtered

function (2.18) and its first order Taylor expansion (2.19). Analogously, to obtain a second order approximation of  $\mathcal{J}_1$  we Taylor expand equation (2.18) around s = 0 up to second order, and include the first two terms of this expansion into our scheme. This yields,

$$\mathcal{N}(\tau,\zeta,\zeta,\Delta,v) = \mathcal{N}(\tau,0,\zeta,\Delta,v) + \zeta \partial_s \mathcal{N}(\tau,0,\zeta,\Delta,v) + \int_0^\zeta \int_0^s \partial_{s_1}^2 \mathcal{N}(\tau,s_1,\zeta,\Delta,v) ds_1 ds.$$
(2.41)

Equivalently, using the filtered function (2.18), the above expression is written as

$$e^{-i\zeta\Delta}f(e^{i\zeta\Delta}v,e^{-i\zeta\Delta}\bar{v},V) = f(v,e^{-2i\zeta\Delta}\bar{v},e^{-i\zeta\Delta}V) + \zeta \mathcal{C}[f,i\Delta](v,e^{-2i\zeta\Delta}\bar{v},e^{-i\zeta\Delta}V)$$

$$+ \int_0^\zeta \int_0^s e^{-is_1\Delta} \mathcal{C}^2[f,i\Delta](e^{is_1\Delta}v,e^{is_1\Delta}e^{-2i\zeta\Delta}\bar{v},e^{is_1\Delta}e^{-i\zeta\Delta}V) ds_1 ds$$
(2.42)

where the local error structure is governed by the second-order commutator type term

$$\mathcal{C}^{2}[f, i\Delta](u, v, w) := \mathcal{C}[\mathcal{C}[f, i\Delta], i\Delta](u, v, w)$$

$$= \sum_{l=1}^{d} (\partial_{l}^{2} u \partial_{l}^{2} w) + (2u+v)(\partial_{l}^{2} u \partial_{l}^{2} v) + (\partial_{l} v)^{2} \partial_{l}^{2} u + 2(\partial_{l} u)^{2} \partial_{l}^{2} v + \partial_{l} v \partial_{l} u(2\partial_{l}^{2} u + \partial_{l}^{2} v).$$

$$(2.43)$$

In practical computations we need to address the stability issues caused by including into the scheme the second term  $C[f, i\Delta](v, e^{-2i\zeta\Delta}\bar{v}e^{-i\zeta\Delta}V)$  which has the form (2.21), since it involves spatial derivatives. Different approaches can be made to treat this issue and guarantee the stability of the scheme and in what follows we offer two different approaches. The first approach is based on [RS21] and consists in first introducing a stabilization in the Taylor series expansion (2.41) based on finite difference approximations. The second approach relies on directly embedding the commutator term appearing in equation (2.42) into the discretization, and then stabilizes the scheme a posteriori by the use of a properly chosen filter function.

#### 2.4.1 A first approach to guarantee stability

A first approach consists in stabilizing the second term of equation (2.41). This may be done by introducing the following finite difference approximation of  $\partial_s \mathcal{N}(\tau, 0, \zeta, \Delta, v)$ :

$$\partial_s \mathcal{N}(\tau, 0, \zeta, \Delta, v) = \frac{1}{\tau} (\mathcal{N}(\tau, \tau, \zeta, \Delta, v) - \mathcal{N}(\tau, 0, \zeta, \Delta, v)) + O(\tau \partial_s^2 \mathcal{N}(\tau, \eta, \zeta, \Delta, v))$$
(2.44)

for some  $\eta \in [0, \tau]$ , and where

$$\partial_s^2 \mathcal{N}(\tau, \eta, \zeta, \Delta, v) = e^{-i\eta\Delta} \mathcal{C}^2[f, i\Delta](e^{i\eta\Delta}v, e^{i\eta\Delta}e^{-2i\zeta\Delta}\bar{v}, e^{i\eta\Delta}e^{-i\zeta\Delta}V).$$
(2.45)

The above expansion comes into play in the following lemma, where we obtain a stable second order approximation of the principal oscillation  $\mathcal{J}_1$ .

Lemma 2.4.2 At second order the principal oscillations can be expanded by

$$\begin{aligned} \mathcal{J}_1(\tau,\Delta,v) &= \tau \left( v \varphi_1(-i\tau\Delta) V + v^2 \varphi_1(-2i\tau\Delta) \bar{v} \right) \\ &+ \tau e^{-i\tau\Delta} \left( (e^{i\tau\Delta} v) \varphi_2(-i\tau\Delta) (e^{i\tau\Delta} V) + (e^{i\tau\Delta} v)^2 \varphi_2(-2i\tau\Delta) (e^{i\tau\Delta} \bar{v}) \right) \\ &- \tau \left( v \varphi_2(-i\tau\Delta) V + v^2 \varphi_2(-2i\tau\Delta) \bar{v} \right) + i R^1_{2,1}(\tau) \end{aligned}$$

where  $\varphi_2(z) = \frac{1}{z}(e^z - \varphi_1(z))$  and the remainder is given by

$$R_{2,1}^{1}(\tau) = \int_{0}^{\tau} \int_{0}^{\zeta} \int_{0}^{s} e^{-is_{1}\Delta} \mathcal{C}^{2}[f, i\Delta] (e^{is_{1}\Delta}v, e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{v}, e^{is_{1}\Delta}e^{-i\zeta\Delta}V) ds_{1} ds d\zeta$$

$$+ \int_{0}^{\tau} \zeta \int_{0}^{1} \int_{0}^{\tau s} e^{-is_{1}\Delta} \mathcal{C}^{2}[f, i\Delta] (e^{is_{1}\Delta}v, e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{v}, e^{is_{1}\Delta}e^{-i\zeta\Delta}V) ds_{1} ds d\zeta.$$

$$(2.46)$$

*Proof.* Using the filtered function (2.18), and plugging the finite difference (2.44) into the Taylor expansion (2.41) we

obtain

$$\begin{aligned} \mathcal{J}_{1}(\tau,\Delta,v) &= \int_{0}^{\tau} \mathcal{N}(\tau,\zeta,\zeta,\Delta,v) d\zeta \\ &= \int_{0}^{\tau} \mathcal{N}(\tau,0,\zeta,\Delta,v) d\zeta + \frac{1}{\tau} \int_{0}^{\tau} \zeta (\mathcal{N}(\tau,\tau,\zeta,\Delta,v) - \mathcal{N}(\tau,0,\zeta,\Delta,v)) d\zeta \\ &+ \int_{0}^{\tau} \zeta \int_{0}^{1} \int_{0}^{\tau s} \partial_{s_{1}}^{2} \mathcal{N}(\tau,s_{1},\zeta,\Delta,v) ds_{1} ds d\zeta + \int_{0}^{\tau} \int_{0}^{\zeta} \int_{0}^{s} \partial_{s_{1}}^{2} \mathcal{N}(\tau,s_{1},\zeta,\Delta,v) ds_{1} ds d\zeta \end{aligned}$$

Using the filtered function (2.18), equation (2.45) and the definition (2.46) of  $R_{2,1}^1(\tau)$  it follows from the above that,

$$\begin{split} \mathcal{J}_{1}(\tau,\Delta,v) &= \int_{0}^{\tau} \left( [e^{-i\zeta\Delta}V]v + [e^{-2i\zeta\Delta}\bar{v}]v^{2} \right) d\zeta \\ &+ \frac{1}{\tau} e^{-i\tau\Delta} \int_{0}^{\tau} \zeta ([e^{-i\zeta\Delta}e^{i\tau\Delta}V](e^{i\tau\Delta}v) + [e^{-2i\zeta\Delta}e^{i\tau\Delta}\bar{v}](e^{i\tau\Delta}v)^{2} ) d\zeta \\ &- \frac{1}{\tau} \int_{0}^{\tau} \zeta ([e^{-i\zeta\Delta}V]v + [e^{-2i\zeta\Delta}\bar{v}]v^{2} ) d\zeta + iR_{2,1}^{1}(\tau) \\ &= \tau \big( v\varphi_{1}(-i\tau\Delta)V + v^{2}\varphi_{1}(-2i\tau\Delta)\bar{v} \big) \\ &+ \tau e^{-i\tau\Delta} \Big( (e^{i\tau\Delta}v)\varphi_{2}(-i\tau\Delta)(e^{i\tau\Delta}V) + (e^{i\tau\Delta}v)^{2}\varphi_{2}(-2i\tau\Delta)(e^{i\tau\Delta}\bar{v}) \Big) \\ &- \tau \big( v\varphi_{2}(-i\tau\Delta)V + v^{2}\varphi_{2}(-2i\tau\Delta)\bar{v} \big) + iR_{2,1}^{1}(\tau) \end{split}$$

which concludes the proof.

By merging the two preceding lemmas 2.4.1 and 2.4.2, we obtain the following second order low-regularity scheme for (2.1).

**Corollary 2.4.3** The exact solution u of (2.1) can be expanded as

$$u(t_{n} + \tau) = e^{i\tau\Delta}u(t_{n}) - i\tau e^{i\tau\Delta} \left( u(t_{n})\varphi_{1}(-i\tau\Delta)V + u(t_{n})^{2}\varphi_{1}(-2i\tau\Delta)\bar{u}(t_{n}) \right)$$

$$- i\tau \left( (e^{i\tau\Delta}u(t_{n}))\varphi_{2}(-i\tau\Delta)(e^{i\tau\Delta}V) + (e^{i\tau\Delta}u(t_{n}))^{2}\varphi_{2}(-2i\tau\Delta)e^{i\tau\Delta}\bar{u}(t_{n}) \right)$$

$$+ i\tau e^{i\tau\Delta} \left( u(t_{n})\varphi_{2}(-i\tau\Delta)V + u(t_{n})^{2}\varphi_{2}(-2i\tau\Delta)\bar{u}(t_{n}) \right)$$

$$- \frac{\tau^{2}}{2} e^{i\tau\Delta} \left( |u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2} \right)$$

$$+ \mathcal{R}_{2}^{1}(\tau, t_{n})$$
(2.47)

where the remainder is given by

$$\mathcal{R}_{2}^{1}(\tau,t_{n}) = \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} [f(u(t_{n}+\zeta),\bar{u}(t_{n}+\zeta),V) - f(e^{i\zeta}u(t_{n}),e^{-i\zeta}\bar{u}(t_{n}),V)]d\zeta + e^{i\tau\Delta} \int_{0}^{\tau} \zeta (|u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2})d\zeta + \int_{0}^{\tau} \int_{0}^{\zeta} \int_{0}^{s} e^{i(\tau-s_{1})\Delta} \mathcal{C}^{2}[f,i\Delta](e^{is_{1}\Delta}u(t_{n}),e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{u}(t_{n}),e^{is_{1}\Delta}e^{-i\zeta\Delta}V)ds_{1}dsd\zeta + \int_{0}^{\tau} \zeta \int_{0}^{1} \int_{0}^{\tau s} e^{i(\tau-s_{1})\Delta} \mathcal{C}^{2}[f,i\Delta](e^{is_{1}\Delta}u(t_{n}),e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{u}(t_{n}),e^{is_{1}\Delta}e^{-i\zeta\Delta}V)ds_{1}dsd\zeta.$$
(2.48)

and where the explicit expression for the commutator is given in equation (2.43).

#### 2.4.2 A second approach to guarantee stability

We next present a second approach to the second order approximation of the principal oscillations  $\mathcal{J}_1$ . In contrast to the preceding section we will first write the approximation in terms of the commutator, then stabilize the scheme by the use of a properly chosen filter function.

Lemma 2.4.4 To second order the principal oscillations can be expanded by

$$\Delta, v) = \tau \left( v \varphi_1(-i\tau \Delta) V + v^2 \varphi_1(-2i\tau \Delta) \bar{v} \right) + \tau^2 \mathcal{C}[f, i\Delta](v, \varphi_2(-2i\tau \Delta) \bar{v}, \varphi_2(-i\tau \Delta) V) + i R_{2,1}^2(\tau)$$
(2.49)

where the remainder is given by,

 $\mathcal{J}_1(\tau,$ 

$$R_{2,1}^2(\tau) = \int_0^\tau \int_0^\zeta \int_0^s e^{-is_1\Delta} \mathcal{C}^2[f, i\Delta](e^{is_1\Delta}v, e^{is_1\Delta}e^{-2i\zeta\Delta}\bar{v}, e^{is_1\Delta}e^{-i\zeta\Delta}V) ds_1 ds d\zeta.$$
(2.50)

*Proof.* Using the definition of the principal oscillations (2.13) and equation (2.42) one has the following expansion,

$$\begin{aligned} \mathcal{J}_{1}(\tau,\Delta,v) &= \int_{0}^{\tau} f(v,e^{-2i\zeta\Delta}\bar{v},e^{-i\zeta\Delta}V)d\zeta + \int_{0}^{\tau} \zeta \mathcal{C}[f,i\Delta](v,e^{-2i\zeta\Delta}\bar{v},e^{-i\zeta\Delta}V)d\zeta \\ &+ \int_{0}^{\tau} \int_{0}^{\zeta} \int_{0}^{s} e^{-is_{1}\Delta} \mathcal{C}^{2}[f,i\Delta](e^{is_{1}\Delta}v,e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{v},e^{is_{1}\Delta}e^{-i\zeta\Delta}V)ds_{1}dsd\zeta \\ &= \tau \big(v\varphi_{1}(-i\tau\Delta)V + v^{2}\varphi_{1}(-2i\tau\Delta)\bar{v}\big) + \tau^{2}\mathcal{C}[f,i\Delta](v,\varphi_{2}(-2i\tau\Delta)\bar{v},\varphi_{2}(-i\tau\Delta)V) \\ &+ iR_{2,1}^{2}(\tau). \end{aligned}$$

where the second term could be integrated exactly using the structure of the commutator (2.21) and of the nonlinearity (2.4)

The following lemma provides the second order low regularity integrator up to this step.

**Lemma 2.4.5** The exact solution u of (2.1) can be expanded as

$$u(t_{n} + \tau) = e^{i\tau\Delta}u(t_{n}) - i\tau e^{i\tau\Delta} (u(t_{n})\varphi_{1}(-i\tau\Delta)V + u(t_{n})^{2}\varphi_{1}(-2i\tau\Delta)\bar{u}(t_{n}))$$

$$- i\tau^{2}e^{i\tau\Delta}C[f, i\Delta](u(t_{n}), \varphi_{2}(-2i\tau\Delta)\bar{u}(t_{n}), \varphi_{2}(-i\tau\Delta)V)$$

$$- \frac{\tau^{2}}{2}e^{i\tau\Delta}(|u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2})$$

$$+ R_{2,2}^{2}(\tau, t_{n})$$

$$(2.51)$$

where the remainder is given by

$$\begin{split} R_{2,2}^{2}(\tau) &= \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} [f(u(t_{n}+\zeta),\bar{u}(t_{n}+\zeta),V) - f(e^{i\zeta}u(t_{n}),e^{-i\zeta}\bar{u}(t_{n}),V)]d\zeta \\ &+ e^{i\tau\Delta} \int_{0}^{\tau} \zeta (|u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2})d\zeta \\ &+ \int_{0}^{\tau} \int_{0}^{\zeta} \int_{0}^{s} e^{i(\tau-s_{1})\Delta} \mathcal{C}^{2}[f,i\Delta](e^{is_{1}\Delta}u(t_{n}),e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{u}(t_{n}),e^{is_{1}\Delta}e^{-i\zeta\Delta}V)ds_{1}dsd\zeta. \end{split}$$

To stabilize the term appearing in the second line of equation (2.51), which is of the form  $\tau^2 C[f, i\Delta](v, \bar{v}, V)$ , we introduce an appropriate filter operator which we denote by  $\Psi$ . More precisely, we will construct a filter operator of the form

$$\Psi = \psi(i\tau |\nabla|),$$

where  $\psi$  is a suitably chosen filter function which allows to stabilize the scheme while introducing an error term which only requires  $H^2$ -regularity on the initial data and potential. Namely, we require the filter function  $\psi$  to introduce the same optimal local error of  $O(\tau^3 \Delta(v+V))$  as is introduced by the low-regularity second order scheme up to this step (see equations (2.39), (2.43) and Section 2.4.3 for the thorough analysis). We refer to [HLW10] for an introduction to filter functions in the ODE setting. We now present two sufficient assumptions on the filter operator which once established, guarantees the stability of the low-regularity scheme (2.51).

Assumption 1. The filter operator  $\Psi = \psi(i\tau |\nabla|)$ , satisfies the following bound

$$||\tau \Psi[\mathcal{C}[f, i\Delta](v, \bar{v}, V)]||_{r} \le C_{r, d, V} ||v||_{r}^{m}$$
(2.52)

for some  $m = m(f) \in \mathbb{N}$  and  $r = r(d) \ge 0$ .

**Assumption 2**. The filter operator 
$$\Psi = \psi(i\tau |\nabla|)$$
 satisfies the following expansion

$$\Psi[\mathcal{C}[f, i\Delta](v, \bar{v}, V)] = \mathcal{C}[f, i\Delta](v, \bar{v}, V) + O(\tau |\nabla|^2 (v + V)).$$

$$(2.53)$$

The condition in Assumption 1 guarantees the stability of the scheme in the  $H^r$ -norm, while the condition in Assumption 2 preserves the optimal local error structure of  $O(\tau^3 |\nabla|^2 (v+V))$  with the inclusion of the filter function  $\psi$ . This is an essential ingredient for the local and global error analysis of the scheme.

**Remark 2.4.6** The stability estimate (2.52) relies on the algebraic structure of the underlying space. In the following stability analysis we will restrict our attention to sufficiently smooth Sobolev spaces  $H^r$  with  $r > \frac{d}{2} + 1$ . This allows us to exploit the following classical bilinear estimate,

$$||vw||_{s_0} \leq C||v||_{s_0}||w||_{s_0},$$

where  $s_0 = r - 1$ . An analysis in a lower order Sobolev space would require the use of more refined estimates on the commutator term using the generalized Leibniz rule (see [BCD11, Chapter 2]). This analysis is not detailed here, since we tackle the error analysis of the second order scheme (2.47), based upon the first approach (see Section 2.4.1).

A choice of filter operator which is well adapted for the second order scheme (2.51) is the following.

Lemma 2.4.7 The filter operator

$$\Psi = \varphi_1(i\tau|\nabla|) := \frac{e^{i\tau|\nabla|} - 1}{i\tau|\nabla|}$$
(2.54)

satisfies Assumption 1 and 2 with  $r > \frac{d}{2} + 1$ .

*Proof.* We first show how the filter function (2.54) satisfies Assumption 1 and hence guarantees the stability of the second order low-regularity scheme. By definition of the  $\varphi_1$  function and using the explicit form of the commutator (2.21) together with the bilinear estimate we have,

$$\begin{aligned} ||\tau\varphi_{1}(i\tau|\nabla|)\mathcal{C}[f,i\Delta](v,\bar{v},V)||_{r} &\leq ||(e^{i\tau|\nabla|}-1)|\nabla|^{-1}\mathcal{C}[f,i\Delta](v,\bar{v},V)||_{r} \\ &\leq 4||\nabla V \cdot \nabla v + |\nabla v|^{2}\bar{v} + 2v\nabla v \cdot \nabla \bar{v}||_{r-1} \\ &\leq C_{r,d}(||\nabla V||_{r-1}||\nabla v||_{r-1} + ||\nabla v||_{r-1}^{2}||v||_{r-1} \\ &\quad + 2||v||_{r-1}||\nabla v||_{r-1}^{2}) \\ &\leq C_{r,d}(||V||_{r}||v||_{r} + 3||v||_{r}^{3}) \\ &\leq C_{r,d,V}(||v||_{r} + ||v||_{r}^{3}). \end{aligned}$$

Furthermore, by a simple Taylor's expansion we have that,

$$\varphi_1(i\tau|\nabla|) = 1 + O(\tau|\nabla|). \tag{2.55}$$

It then follows by the form of the commutator (2.21) that the filter function (2.54) satisfies Assumption 2. Indeed, from the above equation we have,

$$\varphi_1(i\tau\nabla)\mathcal{C}[f,i\Delta](v,\bar{v},V) = \mathcal{C}[f,i\Delta](v,\bar{v},V) + O(\tau|\nabla|\mathcal{C}[f,i\Delta](v,\bar{v},V)),$$

where by equation (2.21) we have that formally  $O(\tau |\nabla| \mathcal{C}[f, i\Delta](v, \bar{v}, V)) = O(\tau |\nabla|^2 (v + V))$ . Hence, formally we see that the inclusion of the filter function (2.54) preserves the optimal local error, by only requiring two additional derivatives on the initial datum and the potential (see Proposition 2.4.9, in the regime  $r > \frac{d}{2}$ ).

The following Corollary provides a *stable* second order low-regularity scheme for (2.1), by using the filter function (2.54).

**Corollary 2.4.8** The exact solution u of (2.1) can be expanded as

$$u(t_{n} + \tau) = e^{i\tau\Delta}u(t_{n}) - i\tau e^{i\tau\Delta} \left( u(t_{n})\varphi_{1}(-i\tau\Delta)V + u(t_{n})^{2}\varphi_{1}(-2i\tau\Delta)\bar{u}(t_{n}) \right) - i\tau^{2}e^{i\tau\Delta}\varphi_{1}(i\tau\nabla)[\mathcal{C}[f,i\Delta](u(t_{n}),\varphi_{2}(-2i\tau\Delta)\bar{u}(t_{n}),\varphi_{2}(-i\tau\Delta)V)] - \frac{\tau^{2}}{2}e^{i\tau\Delta} (|u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2}) + \mathcal{R}_{2}^{2}(\tau,t_{n}).$$

$$(2.56)$$

where

$$\begin{aligned} \mathcal{R}_{2}^{2}(\tau,t_{n}) &= \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} [f(u(t_{n}+\zeta),\bar{u}(t_{n}+\zeta),V) - f(e^{i\zeta}u(t_{n}),e^{-i\zeta}\bar{u}(t_{n}),V)]d\zeta \\ &+ e^{i\tau\Delta}\int_{0}^{\tau} \zeta(|u(t_{n})|^{4}u(t_{n}) + 3u(t_{n})|u(t_{n})|^{2}V - |u(t_{n})|^{2}u(t_{n})\bar{V} + u(t_{n})V^{2})d\zeta \\ &+ \int_{0}^{\tau}\int_{0}^{\zeta}\int_{0}^{s} e^{i(\tau-s_{1})\Delta}\mathcal{C}^{2}[f,i\Delta](e^{is_{1}\Delta}u(t_{n}),e^{is_{1}\Delta}e^{-2i\zeta\Delta}\bar{u}(t_{n}),e^{is_{1}\Delta}e^{-i\zeta\Delta}V)ds_{1}dsd\zeta \\ &- i\tau^{2}e^{i\tau\Delta}(I - \varphi_{1}(i\tau\nabla))[\mathcal{C}[f,i\Delta](u(t_{n}),\varphi_{2}(-2i\tau\Delta)\bar{u}(t_{n}),\varphi_{2}(-i\tau\Delta)V)]. \end{aligned}$$

#### 2.4.3 Local error estimates

In this section we prove that the second order scheme (2.47) introduces a local error of order three under favorable regularity assumptions of the initial datum and potential. As was the case for the error analysis of the first-order scheme (Section 2.3.3), we make the analysis in  $H^r$ -norm where the regularity assumptions on v and V will depend on the regime of r considered.

**Proposition 2.4.9** Let T > 0,  $r \ge 0$ , and  $r_2$  as in Theorem 2.1.2, namely

$$r_{2} = \begin{cases} 2+r, & \text{if } r > \frac{d}{2}, \\ 2+\frac{r}{2}+\frac{d}{4}, & \text{if } 0 \le r < \frac{d}{2}. \end{cases}$$

Assume there exists  $C_T > 0$  such that

$$\sup_{[0,T]} ||u(t)||_{H^{r_2}} \le C_T, \text{ and } ||V||_{H^{r_2}} \le C_T,$$

then there exists  $M_T > 0$  such that for every  $\tau \in (0, 1]$ 

$$||\mathcal{R}_{2}^{1}(\tau, t_{n})||_{H^{r}} \leq M_{T}\tau^{3}, \quad 0 \leq t_{n} \leq T_{n}$$

where  $t_n = n\tau$ , and  $\mathcal{R}_2^1(\tau, t_n)$  is given in equation (2.48).

*Proof.* We write the error term  $\mathcal{R}_2^1(\tau, t_n)$ , as the sum of four terms,  $\mathcal{R}_2^1(\tau, t_n) = \mathcal{E}^1(\tau, t_n) + \mathcal{E}^2(\tau, t_n) + \mathcal{E}^3(\tau, t_n) + \mathcal{E}^4(\tau, t_n)$ . We start by establishing the third order estimate for the two last terms  $\mathcal{E}^3$  and  $\mathcal{E}^4$ . These bounds are obtained using the same arguments as those made to bound the term  $\mathcal{G}^2$  in the proof of Proposition 2.3.2, by noticing that  $r_2 = r_1 + 1$ . Indeed, from the explicit expression of the second-order commutator (2.43) and using the inequality (2.10), we obtain that for  $r > \frac{d}{2}$ ,

$$||\mathcal{C}^{2}[f, i\Delta](u, v, w)||_{H^{r}} \leq C_{r}(||w||_{H^{r+2}}, ||u||_{H^{r+2}}, ||v||_{H^{r+2}}),$$
(2.57)

and for  $0 \le r < \frac{d}{2}$  and l = 1, ..., d,

$$\begin{aligned} \|\mathcal{C}^{2}[f,i\Delta](u,v,w)\|_{H^{r}} &\leq C_{r,d}(\|\partial_{l}^{2}w\|_{H^{\frac{r}{2}}+\frac{d}{4}}, \|\partial_{l}^{2}u\|_{H^{\frac{r}{2}}+\frac{d}{4}}, \|\partial_{l}^{2}v\|_{H^{\frac{r}{2}}+\frac{d}{4}}, \|u\|_{H^{\sigma_{0}}}, \|v\|_{H^{\sigma_{0}}}, \|\partial_{l}^{2}u\|_{H^{r}}, \|\partial_{l}^{2}v\|_{H^{r}}, \\ &\|\partial_{l}v\|_{H^{\sigma_{0}}}, \|\partial_{l}u\|_{H^{\sigma_{0}}}) \\ &\leq C_{r,d}(\|w\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}, \|u\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}, \|v\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}, \|u\|_{H^{2+r}}, \|v\|_{H^{2+r}}, \|v\|_{H^{1+\sigma_{0}}}, \|v\|_{H^{1+\sigma_{0}}}) \\ &\leq C_{r,d}(\|w\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}, \|u\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}, \|v\|_{H^{2+\frac{r}{2}}+\frac{d}{4}}), \end{aligned}$$

where we once again used the bilinear estimates (2.28) and (2.10) with  $\sigma = \sigma_0$  in the first line, and in the third line the definition (2.26) of  $\sigma_0$  and the fact that  $r < \frac{r}{2} + \frac{d}{4}$  when  $r < \frac{d}{2}$ .

Next, we show that the sum of the remaining terms,  $(\mathcal{E}^1 + \mathcal{E}^2)(\tau, t_n)$ , of equation (2.48) is of third order. We have that  $r_2 > 2$ , and hence  $u(t), V \in H^2$ . Thereby, by Taylor expanding the exponential appearing inside the Duhamel's integral (2.3) we obtain the following expansion :  $u(t_n + \zeta) = e^{i\zeta\Delta}u(t_n) + \zeta f^n + \tilde{R}(\zeta, t_n)$  where  $f^n = f(u(t_n), \bar{u}(t_n), V)$  and

$$\tilde{R}(\zeta, t_n) = \int_0^{\zeta} e^{i(\zeta - s)\Delta} f(u(t_n + s), \bar{u}(t_n + s), V) ds - \zeta f^n.$$
(2.58)

Using the above expansion for u we rewrite the error term  $(\mathcal{E}^1 + \mathcal{E}^2)(\tau, t_n)$  as,

$$(\mathcal{E}^{1} + \mathcal{E}^{2})(\tau, t_{n}) = \int_{0}^{\tau} e^{i(\tau - \zeta)\Delta} [f\left(e^{i\zeta\Delta}u(t_{n}) + \zeta f^{n} + \tilde{R}(\zeta, t_{n}), e^{-i\zeta\Delta}\bar{u}(t_{n}) + \zeta\overline{f^{n}} + \overline{\tilde{R}}(\zeta, t_{n}), V\right)$$

$$- f(e^{i\zeta\Delta}u(t_{n}), e^{-i\zeta\Delta}\bar{u}(t_{n}), V)]d\zeta$$

$$- e^{i\tau\Delta} \int_{0}^{\tau} \zeta(D_{1}f^{n} \cdot f^{n} + D_{2}f^{n} \cdot \overline{f^{n}})d\zeta,$$

$$(2.59)$$

where  $D_1 f^n \cdot f^n + D_2 f^n \cdot \overline{f^n}$  is given by equation (2.40). For notational convenience we let  $a_1 := e^{i\zeta\Delta}u(t_n) + \zeta f^n$ . The idea in order to show that the above error term (2.59) is of third-order revolves around making three suitable Taylor expansions. By Taylor expanding f around  $(a_1, \overline{a}_1, V)$  and  $(e^{i\zeta\Delta}u(t_n), e^{-i\zeta\Delta}\overline{u}(t_n), V)$  respectively we obtain,

$$f\left(a_{1}+\tilde{R}(\zeta,t_{n}),\bar{a}_{1}+\overline{\tilde{R}(\zeta,t_{n})},V\right) = f(a_{1},\bar{a}_{1},V) + E_{1}(\zeta),$$

$$f(a_{1},\bar{a}_{1},V) = f\left(e^{i\zeta\Delta}u(t_{n}),e^{-i\zeta\Delta}\bar{u}(t_{n}),V\right) + E_{2}(\zeta),$$
(2.60)

where

$$E_{1}(\zeta) = \int_{0}^{1} D_{1}f\left(a_{1} + \theta\tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta\bar{\tilde{R}}(\zeta, t_{n}), V\right) \cdot \tilde{R}(\zeta, t_{n})$$

$$+ D_{2}f\left(a_{1} + \theta\tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta\bar{\tilde{R}}(\zeta, t_{n}), V\right) \cdot \overline{\tilde{R}}(\zeta, t_{n}) d\theta,$$

$$E_{2}(\zeta) = \zeta \int_{0}^{1} [D_{1}f\left(e^{i\zeta\Delta}u(t_{n}) + \theta\zeta f^{n}, e^{-i\zeta\Delta}\bar{u}(t_{n}) + \theta\zeta \overline{f^{n}}, V\right) \cdot f^{n}$$

$$+ D_{2}f\left(e^{i\zeta\Delta}u(t_{n}) + \theta\zeta f^{n}, e^{-i\zeta\Delta}\bar{u}(t_{n}) + \theta\zeta \overline{f^{n}}, V\right) \cdot \overline{f^{n}}] d\theta.$$

$$(2.61)$$

Hence, plugging equation (2.60) into equation (2.59) yields,

$$(\mathcal{E}^1 + \mathcal{E}^2)(\tau, t_n) = \int_0^\tau e^{i(\tau - \zeta)\Delta} E_1(\zeta) d\zeta + e^{i\tau\Delta} \int_0^\tau e^{-i\zeta\Delta} E_2(\zeta) d\zeta - e^{i\tau\Delta} \int_0^\tau \zeta (D_1 f^n \cdot f^n + D_2 f^n \cdot \overline{f^n}) d\zeta.$$
(2.63)

In order to show that the first term in the above equation is of third-order we show the bound  $||E_1(\zeta)||_r \leq C_T \zeta^2$ . By equation (2.61) and by using the bilinear inequality (2.10) we have that for all  $r \geq 0$  and some  $\sigma > \frac{d}{2}$ ,

$$||E_{1}(\zeta)||_{r} \leq \sup_{\theta \in ]0,1[} \left( ||D_{1}f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}(\zeta, t_{n})}, V\right)||_{\sigma} + ||D_{2}f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}(\zeta, t_{n})}, V\right)||_{\sigma} \right)||\tilde{R}(\zeta, t_{n})||_{r}$$

$$\leq C_{r} \left( ||V||_{\sigma}, \sup_{t \in [0,T]} ||u(t)||_{\sigma}, \sup_{(\zeta,t) \in [0,\tau] \times [0,T]} ||\tilde{R}(\zeta, t)||_{\sigma} \right)||\tilde{R}(\zeta, t_{n})||_{r}$$

$$(2.64)$$

where the last inequality follows by using the explicit form of the derivatives (2.38), the first estimation of equation (2.11), and the fact that  $e^{i\zeta\Delta}$  is an isometry on Sobolev spaces. When  $r > \frac{d}{2}$ , the above holds for  $\sigma = r$ . Next, we show that,

$$\sup_{(\zeta,t)\in[0,\tau]\times[0,T]} ||\tilde{R}(\zeta,t)||_{\sigma_0} < +\infty, \quad \text{and} \quad ||\tilde{R}(\zeta,t_n)||_r \le C_T \zeta^2,$$
(2.65)

where  $\sigma_0$  is given by equation (2.26). We obtain the first bound by using the first estimate of equation (2.11) on f with  $r = \sigma_0$ ,

$$\sup_{(\zeta,t)\in[0,\tau]\times[0,T]} ||\tilde{R}(\zeta,t)||_{\sigma_0} \le \tau C_{\sigma_0}(\sup_{t\in[0,T]} ||u(t)||_{\sigma_0}, ||V||_{\sigma_0}) < +\infty.$$

Next, we obtain the second estimate in equation (2.65) using the Fundamental Theorem of Calculus. By letting

 $u = u(t_n + s_1)$  we have,

$$\tilde{R}(\zeta, t_n) = \int_0^{\zeta} \int_0^s \partial_{s_1} \left( e^{i(\zeta - s_1)\Delta} f(u, \bar{u}, V) \right) ds_1 ds + \zeta e^{i\zeta\Delta} f^n - \zeta f^n$$

$$= \int_0^{\zeta} \int_0^s e^{i(\zeta - s_1)\Delta} \left( i\Delta f(u, \bar{u}, V) + D_1 f(u, \bar{u}, V) (i\Delta u + f(u, \bar{u}, V)) + D_2 f(u, \bar{u}, V) (-i\overline{\Delta u} + \overline{f(u, \bar{u}, V)}) \right) ds_1 ds + i\zeta^2 \varphi_1 (i\zeta\Delta) \Delta f(u, \bar{u}, V),$$
(2.66)

where to obtain the second line we used equation (2.1), and for the last term we used the definition of the  $\varphi_1$  operator. The second estimate of equation (2.65) then follows immediately from equation (2.66) by observing that,

$$||\Delta f(v, \bar{v}, V)||_{r} + ||D_{i}f(v, \bar{v}, V)\Delta u||_{r} \le C_{r}(||v||_{r_{2}}, ||V||_{r_{2}}), \quad i = 1, 2.$$

It remains to show that the difference of the second and third term in equation (2.63) is of third-order. Formally, this directly follows by making a Taylor expansion of  $e^{i\zeta\Delta}u(t_n) + \theta\zeta f^n$  around  $\zeta = 0$ :  $e^{i\zeta\Delta}u(t_n) + \theta\zeta f^n = u(t_n) + O(\zeta\Delta u(t_n))$ . By using the same application of the Fundamental Theorem of Calculus as done in equation (2.66) one obtains this third order bound. This concludes the proof.

#### 2.4.4 Global error estimate

Using the local error estimates established in the preceding section together with a stability argument we show global second order convergence of our low regularity integrator under the regularity assumptions established in Proposition 2.4.9.

Proof of Theorem 2.1.2. We let  $u^{n+1} = \Phi_2^{\tau}(u^n)$  be the numerical scheme defined in equation (2.7). The outline of the proof of this second-order convergence result follows exactly the same lines as the first order convergence result given in Theorem 2.1.1. Indeed, in the case where  $r > \frac{d}{2}$  the global error estimate follows by a classical Lady Windermere's argument ([HLW10]). In the regime  $r < \frac{d}{2}$ , by exploiting the same interpolation argument as made in the proof of Theorem 2.1.1, and by using definition (2.48) of  $\mathcal{R}_2^1$  we have that there exists an  $\delta > 0$  such that

$$||\mathcal{R}_2^1(\tau, t_n)||_{H^{\sigma_0}} \le M_T \tau^{1+\delta}$$

where  $\sigma_0$  is defined in equation (2.26). From the above we obtain the bound (2.34) and hence the apriori  $H^{\sigma_0}$ -bound on the iterates:  $\sup_{n\tau \leq T} ||u^n||_{H^{\sigma_0}} < +\infty$ , for  $\tau$  sufficiently small. Using the local error analysis given in Proposition 2.4.9 we obtain second order convergence of the scheme (2.7) by performing an inductive argument with

$$||e^{n+1}||_{H^r} \le M_T \tau^3 + e^{L_n \tau} ||e^n||_{H^r}, \quad e^0 = 0, \tag{2.67}$$

since  $\sup_{n\tau \leq T} L_n \leq C_{T,r}(||u^n||_{H^{\sigma_0}}) < +\infty$ . This concludes the proof.

## 2.5 Numerical Experiments

In this section we provide some numerical experiments to support our theoretical convergence results. We consider the Gross-Pitaevskii equation (2.1) with an initial data of the form

$$u_0(x) = \sum_{k \in \mathbb{Z}} (1+|k|)^{-\vartheta - \frac{1}{2}} a_k e^{ikx}, \qquad (2.68)$$

where the coefficients  $(a_k)_{k\in\mathbb{Z}}$  are chosen as uniformly distributed random complex numbers in [0, 1] + i[0, 1], using the mathlab function rand. The parameter  $\vartheta \ge 0$  dictates the regularity assumption on the above function (2.68), namely it insures that  $u_0 \in H^{\vartheta - \epsilon}$ . We choose the potential V to have the same form (2.68), as the initial condition.

For the space discretization, we couple the first and second order low-regularity time integrators (2.5) and (2.7) with a standard Fourier pseudo-spectral method. We take the largest Fourier mode as  $K = 2^{10}$ , yielding a spatial mesh size of  $\Delta x = 0.0061$ .

In order to test our convergence result, we choose to measure the error in the (discrete)  $L^2$ ,  $H^{\frac{1}{2}}$ , and  $H^1$  norms. For each of these three norms we plot the first and second order low regularity integrators given in equations (2.5) and (2.7) for  $u_0, V \in H^{r_1}$  and  $u_0, V \in H^{r_2}$  respectively (see equations (2.6) and (2.8)). The error at time T = 1 are given in Figure (2.1). The results of our numerical experiments agree with the corresponding theoretical convergence results: we observed first and second order convergence for the regularity assumptions given in Theorem 2.1.1 and 2.1.2. We recall that in the



Figure 2.1 – Convergence plots for three different norms, taken from the article [AB23a]. The slopes of the continuous lines are one and two, respectively. Plot (a) : first and second order scheme with  $\vartheta = 5/4$  (pink and yellow), and for the second-order scheme with  $\vartheta = 9/4$  (green). Plot (b): first and second order scheme with  $\vartheta = \frac{3}{2}$  (red and yellow), and for the second order scheme with  $\vartheta = 5/2$  (blue). Plot (c): first and resp. second order schemes with  $\vartheta = 2$  (orange and yellow) and second order with  $\vartheta = 3$  (purple).

critical case  $r = \frac{1}{2}$ , we have  $r_j = j + \frac{1}{2} + \epsilon$ , j = 1, 2, where  $\epsilon > 0$  can be taken arbitrarily small. Moreover, as expected, the yellow lines in Figure 2.1 show how the second order scheme exhibits order reduction for the less regular data and potential  $u_0, V \in H^{r_1}$ . Nevertheless, it successfully converges to second order for  $u_0, V \in H^{r_2}$ . We lastly note that the observed convergence is slightly better than predicted by Theorem 2.1.1 and 2.1.2 (see as well the work of [OS18]).

Acknowledgements. This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 850941). The author would like to express her thanks to Katharina Schratz for helpful discussions, and to the anonymous referees for their insightful comments and suggestions.

## Appendix

In this section we derive the bilinear  $H^r$ -estimates, for r > 0, as stated in Section 2.2. Namely we show the estimate (2.10) in both regimes  $r > \frac{d}{2}$ , and  $0 < r \le \frac{d}{2}$ . We note that the case r = 0 of the  $L^2$ -norm follows easily by the Sobolev embedding  $H^{\sigma} \hookrightarrow L^{\infty}$ ,  $\sigma > \frac{d}{2}$ .

The notation and tools we use are based upon Littlewood-Paley Theory (see [BCD11, Chapter 2]). We will apply this machinery to our study on the torus  $\mathbb{T}^d$ . Given any tempered distribution u, the Littlewood-Paley theory provides the following decomposition,

$$u = \sum_{k \ge -1} \Delta_k u, \quad \mathcal{F}(\Delta_k u)(\xi) = \varphi_k(\xi)\hat{u}(\xi),$$

where  $\varphi_k(\xi) = \varphi(\xi/2^k)$ , for  $k \ge 0$ ,  $\varphi_{-1} = \chi$ , and  $\varphi, \chi$  satisfy the assertions in [BCD11, Proposition 2.10], namely they form a dyadic partition of unity. In the above and in the remainder of this section we use the fact that, as is the case when working on  $\mathbb{R}^d$ , one can make use of the Fourier transform on  $\mathbb{T}^d$ , where in the periodic case we have that  $\xi \in \mathbb{Z}^d$ .

Using the above Littlewood-Paley decomposition, we introduce Bony's decomposition in order to express the product of two tempered distributions uv as the following sum of three terms,

$$uv = T_u(v) + T_v(u) + R(u, v)$$
(2.69)

where 
$$T_u(v) = \sum_j S_{j-1} u \Delta_j v$$
,  $S_{j-1} u = \sum_{i \le j-2} \Delta_i u$ , and the remainder  $R(u, v) = \sum_{|k-j| \le 1} \Delta_k u \Delta_j v$ 

In what follows we shall make use of the embedding  $H^r \hookrightarrow B_{\infty,\infty}^{r-\frac{\mu}{2}}$ , for non homogeneous Besov spaces, (see [BCD11, Proposition 2.71]).

We are now ready to demonstrate the estimate (2.10) in the regime  $0 < r \leq \frac{d}{2}$ . Let  $\epsilon > 0$ . First by [BCD11, Proposition 2.85] together with the embedding  $H^{\frac{d}{2}} \hookrightarrow B^0_{\infty,\infty}$ , we have the following estimate of the remainder,

$$||R(u,v)||_{H^r} \le C_{r,d} ||v||_{H^r} ||u||_{H^{\frac{d}{2}}}, \quad r > 0.$$

Next, by using the first estimate of [BCD11, Theorem 2.82] together with the embedding  $H^{\frac{d}{2}+\epsilon} \hookrightarrow L^{\infty}$  we have the following estimate on the paraproduct of v by u,

$$||T_u v||_{H^r} \le ||u||_{H^{\frac{d}{2}+\epsilon}} ||v||_{H^r}, \quad r > 0.$$

Using the second estimate of [BCD11, Theorem 2.82] we obtain that for any  $0 < r \leq \frac{d}{2}$  the following estimate of the paraproduct of u by v holds,

$$||T_v u||_{H^r} \le \begin{cases} ||u||_{H^{\frac{d}{2}}} ||v||_{H^r}, & \text{for } 0 < r < \frac{d}{2}, \\ ||u||_{H^{\frac{d}{2}+\epsilon}} ||v||_{H^{\frac{d}{2}}}, & \text{if } r = \frac{d}{2}, \end{cases}$$

where we used the embedding  $H^r \hookrightarrow B^{r-d/2}_{\infty,\infty}$ , and respectively  $H^{\frac{d}{2}-\epsilon} \hookrightarrow B^{-\epsilon}_{\infty,\infty}$  in the case  $r = \frac{d}{2}$ . Therefore, when  $0 < r \leq \frac{d}{2}$ , by collecting the above bounds, together with the decomposition (2.69), we recover the estimate (2.10):

$$||uv||_{H^r} \leq C_{r,d}||u||_{\frac{d}{2}+\epsilon}||v||_{H^r}.$$

Finally, we show the estimate (2.10) in the regime  $r > \frac{d}{2}$ , with  $\sigma = r$ . By [BCD11, Corollary 2.86] we have,

$$||uv||_{H^r} \le \frac{C^{r+1}}{r} (||u||_{L^{\infty}} ||v||_{H^r} + ||u||_{H^r} ||v||_{L^{\infty}}), \quad r > 0$$

Hence, given any  $r > \frac{d}{2}$ , using the embedding  $H^r \hookrightarrow L^{\infty}$ , we obtain the claimed estimate (2.10), with  $\sigma = r$ .

## Chapter 3

# Low regularity integrators via decorated trees

This chapter is based on the article [ABBS22b].

Abstract. We introduce a general framework of low regularity integrators which allows us to approximate the time dynamics of a large class of equations, including parabolic and hyperbolic problems, as well as dispersive equations, up to arbitrary high order on general domains. The structure of the local error of the new schemes is driven by nested commutators which in general require (much) lower regularity assumptions than classical methods do. Our main idea lies in embedding the central oscillations of the nonlinear PDE into the numerical discretisation. The latter is achieved by a novel decorated tree formalism inspired by singular SPDEs with Regularity Structures and allows us to control the nonlinear interactions in the system up to arbitrary high order on the infinite dimensional (continuous) as well as finite dimensional (discrete) level.

## 3.1 Introduction

We consider a general class of evolution equations under the form

$$\partial_t u_o - \mathcal{L}_o u_o = \sum_{\mathfrak{l} \in \mathfrak{L}_-} \Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}}) V_{\mathfrak{l}}(x), \quad (t, x) \in \mathbf{R} \times \Omega, \quad o \in \mathfrak{L}_+,$$
(3.1)

where  $\mathfrak{L}_{-}, \mathfrak{L}_{+}$  are finite sets,  $\Omega \subseteq \mathbf{R}^{d}$ ,  $\mathbf{u}_{o}^{\mathfrak{l}} = (u_{l})_{l \in \mathfrak{L}_{+}^{o,\mathfrak{l}}}$  and  $\mathfrak{L}_{+}^{o,\mathfrak{l}} \subset \mathfrak{L}_{+}$ . For every  $(o,\mathfrak{l}) \in \mathfrak{L}_{+} \times \mathfrak{L}_{-}$ ,  $\mathcal{L}_{o}$  denotes a linear (possibly) unbounded operator,  $\Psi_{o}^{\mathfrak{l}}$  represents the nonlinearity and  $V_{\mathfrak{l}}$  the potential or noise. The precise assumptions on the operator  $\mathcal{L}_{o}$ , the nonlinearity  $\Psi_{o}^{\mathfrak{l}}$  and the potential  $V_{\mathfrak{l}}$  are stated in Section 3.1.1 below. We add initial conditions  $u_{o}(0) = v_{o}$  and when  $\partial \Omega \neq \emptyset$  suitable boundary conditions which will be encoded in the domain of the operator  $\mathcal{L}_{o}$ .

Evolution equations of type (3.1) are meanwhile extensively studied in numerical analysis literature and a large variety of discretisation techniques for their time resolution was proposed, reaching, e.g., from splitting methods over exponential integrators up to Runge–Kutta and Lawson type schemes [Fao12, HNW93, HLW10, HO10, HLO20, Hol10, LR04, MQ02, SSC18]. While such *classical* discretisation techniques provide a good approximation to smooth solutions, they often drastically break down at low regularity: Rough data and high oscillations in general cause severe loss of convergence which leads to huge computational costs hindering in many situations reliable approximations. Nonlinear partial differential equations (PDEs) at low regularity are at large an ongoing challenge in numerical analysis.

In this work we introduce a general framework of low regularity integrators which allows us to approximate the time dynamics of (3.1) up to arbitrary high order under lower regularity assumptions than classical methods, such as Runge–Kutta, splitting, exponential integrators or Lawson schemes, require. Our new framework greatly enhances the low regularity framework recently introduced in [BS22] which is restricted to dispersive equations with periodic boundary conditions and polynomial nonlinearities  $u^{\ell}\overline{u}^{m}$ . Within our new framework we in particular overcome periodic boundary conditions and cover a much larger class of equations, including for instance parabolic, and hyperbolic problems, as well as dispersive equations. Furthermore, our model (3.1) allows for noise or potentials and non polynomial nonlinearities.

In order to bypass the limitations of [BS22] we introduce novel *commutator structures*. Such commutators were previously used in [RS21] for the derivation of first- and second-order low regularity integrators for a simplified version of (3.1) (in particular without potential, i.e.,  $V_{\rm I} \equiv 1$ ). At low order the central oscillations in (3.1) can quite easily be computed and their dominant parts can be extracted "by hand" (see [RS21]). This is however no longer the case at higher order: The control of the underlying oscillations of (3.1) up to arbitrary high order involves a scale of nested, iterated oscillatory integrals depending on the nonlinear interaction between the leading differential operator  $\mathcal{L}_o$  and the nonlinearity  $\sum_{l \in \mathcal{L}_-} \Psi_o^l(\mathbf{u}_o^l) V_l(x)$  which is in particular challenging on the finite dimensional (discrete) level. In order to overcome this, our key idea lies in developing a new class of decorated trees. This formalism is crucial as it allows us to encode the oscillatory systems of iterated integrals on the continuous as well as on the discrete level. We control the local error at each order with suitable nested commutators. The decorated trees used in this work are inspired by [BS22]. We develop a new framework overcoming the limitations of the formalism developed in [BS22] which heavily relied on Fourier expansion techniques, periodic boundary conditions and polynomial nonlinearities. Our decorated trees are close in spirit to the ones used in singular SPDEs in [BCCH20] where decorations on the nodes encode the drivers/potentials of the equation (3.1). This work is the second example after [BS22] where Regularity Structures decorated trees introduced in [Hai14, BHZ19] appear in context of numerical analysis which stresses their robustness. Another key difference with [BS22] lies in the fact that our setting does not allow for a Hopf algebra to conduct the local error analysis. The latter seems limited to Fourier space and the control of the corresponding Fourier modes. The lack of a Hopf algebra in this work is, however, not surprising as we push forward low regularity integrators to the much more general setting (3.1). As we are no longer in the Fourier case, the resonance analysis has to be replaced by a new splitting on the operators directly in Physical space given in Definition 3.3.2. This new idea is crucial for implementing a low regularity scheme and allows to bypass the limitation of the Fourier analysis when one can only consider operators that are polynomial in the frequencies in Fourier space.

Our main result is the new general low regularity scheme presented in Definition 3.4.2 with its error structure given in Theorem 3.4.3. The approximation relies crucially on the formulae given in Definition 3.3.2. We will illustrate our framework on the concrete examples of the Gross–Pitaevskii and the Sine–Gordon equation in full detail in Section 3.5. In the next subsection, we illustre the main ideas of our scheme on a parabolic equation.

#### 3.1.1 Linearization and decorated trees

Before introducing our linearization techniques based on decorated trees, let us first specify the assumptions on the nonlinearity  $\Psi$  and on the leading operator  $\mathcal{L}_o$  appearing in (3.1).

Assumption 1 (nonlinearity  $\Psi$ ) In the following we assume that the nonlinearity  $\Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}})$  is tensorized under the form

$$\Psi_{o}^{\mathfrak{l}}(\mathbf{u}_{o}^{\mathfrak{l}}) = \mathcal{B}_{o}^{\mathfrak{l}}\left(\prod_{\mathfrak{l}\in\mathfrak{L}_{+}^{\mathfrak{l},o}} f_{o,\mathfrak{l}}^{\mathfrak{l}}(u_{\mathfrak{l}})\right), \quad f_{o,\mathfrak{l}}^{\mathfrak{l}}: \mathbf{C} \to \mathbf{C}^{J},$$
(3.2)

where  $\left(\mathcal{B}_{o}^{\mathfrak{l}}\right)_{(o,\mathfrak{l})\in\mathfrak{L}_{+}\times\mathfrak{L}_{-}}$  is a family of (linear) operators and we use the notation  $\prod_{i} X^{(i)} = \sum_{k \leq J} \prod_{i} X_{k}^{(i)}$  where  $X^{(i)} \in \mathbf{C}^{J}$ .

Assumption 2 (leading operator  $\mathcal{L}_o$ ) The linear operator  $\mathcal{L}_o$  shall be defined on a Hilbert space X of complex valued functions  $u \in \mathbf{C}$  with norm denoted by  $\|\cdot\|$  and domain  $D(\mathcal{L}_o)$ . To make sense of the exponential functions, we in addition assume that  $\mathcal{L}_o$  generates a strongly continuous semigroup  $\{e^{t\mathcal{L}_o}\}_{t\geq 0}$  of contractions on X.

We illustrate the main idea of our general scheme via a simple example that contains already some important challenges which could not be overcome in previous works [BS22, RS21]. Let us consider the abstract parabolic equation:

$$\partial_t u - \Delta u = f(u)V, \quad u_0 = v, \quad (t, x) \in \mathbf{R} \times \Omega$$
(3.3)

with  $\Omega \subseteq \mathbf{R}^d$  sufficiently smooth. In the case of  $\partial \Omega \neq \emptyset$  we assume suitable boundary conditions encoded in  $D(\Delta)$  and shall denote by V a potential or noise.

The starting point of our new class of schemes is based on Duhamel's formula for (3.3) which is given by

$$u(t) = e^{t\Delta}v + \int_0^t e^{(t-\xi)\Delta} f(u(\xi)) V d\xi.$$
 (3.4)

Classical methods are in general based on Taylor series expansions of  $u(\xi)$  (the solution of (3.3) at time  $t = \xi$ ) around the initial value  $u_0 = v$  in the sense that for small  $\xi$  we have at first order that

$$u(\xi) = v + \mathcal{O}(\xi u'). \tag{3.5}$$

This classical Taylor series expansion, however, requires regularity in the solution since from the PDE (3.3) we have

$$\mathcal{O}(\xi u') = \mathcal{O}(\xi \Delta u).$$

Hence, the numerical scheme will only converge to order one for sufficiently smooth solutions

 $u \in D(\Delta).$ 

The situation is even worse for higher order methods which are in general based on higher order Taylor series expansions of the solution

$$u(\xi) = v + \xi u'(0) + \frac{\xi^2}{2} u''(0) + \ldots + \frac{\xi^r}{r!} u^{(r)}(0) + \mathcal{O}(\xi^{r+1} u^{(r+1)}).$$
(3.6)

By using the PDE (3.3) we have at leading order

$$\mathcal{O}(\xi^{r+1}u^{(r+1)}) = \mathcal{O}(\xi^{r+1}\Delta^{r+1}u)$$

Hence, it follows that at order r + 1 the necessary regularity assumptions on the solutions is increased to

$$u \in D(\Delta^{r+1}). \tag{3.7}$$

**Remark 3.1.1** Note that (3.7) does not only require sufficiently smooth solutions (i.e., that the  $\partial^{2(r+1)}$ th derivative is bounded), but also additional compatibility conditions on the boundary whenever  $\partial \Omega \neq \emptyset$ . For example, for a second order approximation, namely when r = 1, the classical Taylor expansion (3.6) requires

$$u \in D(\Delta^2) = \{ f \in D(\Delta) \, ; \, \Delta f \in D(\Delta) \}.$$

In case of  $\Omega$  a bounded smooth open set of  $\mathbf{R}^d$  equipped with homogeneous Dirichlet boundary conditions this means that

$$u \in D(\Delta^2) = \left\{ f \in (H^2 \cap H^1_0)(\Omega) \, ; \, \Delta f \in (H^2 \cap H^1_0)(\Omega) \right\}.$$
(3.8)

Hence, zero trace of the solution u and  $\Delta u$  on the boundary is imposed.

In this manuscript we want to overcome these high regularity assumptions by developing a new class of schemes with improved local error structures. It is essential to note that if we want to develop such a class of (high order) methods under low(er) regularity assumptions

$$u \in Y \supset D(\Delta^{r+1})$$

we cannot base our new schemes on classical Taylor series expansions of type (3.6).

Our idea to overcome the regularity assumptions (3.7) lies in embedding the central oscillations of the PDE into the numerical discretisation, while only approximating the lower order terms. More precisely, iterating Duhamel's formula (see also [BS22, RS21]) we have thanks to (3.4) at time  $t = \xi$  that

$$u(\xi) = e^{\xi \mathcal{L}} v + \mathcal{O}(\xi)$$

where the central oscillations are captured in the term

$$e^{\xi \mathcal{L}} v$$

and the remainder  $\mathcal{O}(\xi)$  does not involve any derivatives on the solution. In detail one has for  $u(\xi) = e^{\xi \Delta} v + A(\xi)$  that

$$f(u(\xi)) = \sum_{k \le r} \frac{A(\xi)^k}{k!} f^{(k)}(e^{\xi \Delta} v) + \mathcal{O}(A(\xi)^{r+1})$$
(3.9)

where we truncate the expansion at r which is associated to the order of the scheme one wants to achieve. The remainder A(t) will correspond in practice (after a suitable approximation) to a finite sum of iterated integrals only involving bounded operators.

Plugging (3.9) into (3.4), we obtain the following expansion of the exact solution

$$u(t) = e^{t\Delta}v + \sum_{k \le r} \int_0^t e^{(t-\xi)\Delta} \frac{A(\xi)^k}{k!} f^{(k)}(e^{\xi\Delta}v) V d\xi + \mathcal{O}(t^{r+2}),$$
(3.10)

where the central oscillations are embedded in the interaction of

$$e^{(t-\xi)\Delta}$$
 and  $f^{(k)}(e^{\xi\Delta}v)$ 

and the remainder  $\mathcal{O}(t^{r+2})$  does, in contrast to classical approximation techniques (3.6), not involve any derivatives on the solution.

The main challenge lies in controlling the oscillatory integrals in (3.10)

$$\sum_{k \le r} \int_0^t e^{(t-\xi)\Delta} \frac{A(\xi)^k}{k!} f^{(k)}(e^{\xi\Delta}v) V d\xi$$
(3.11)

on the discrete level. Note that a classical Taylor series expansion of the oscillations  $e^{\xi\Delta}$  would again lead to the high regularity assumptions (3.7). Our key idea to overcome this lies in embedding the central oscillations in (3.11) explicitly into our numerical discretization. In order to do so, in a first step we have to linearise the terms  $f^{(k)}(e^{\xi\Delta}v)$ . This will allow us to filter out the dominant oscillations while controlling the approximation error by commutators  $C[\cdot, \cdot]$  with an improved error structure introduced as follows:

**Definition 3.1.2** For a function  $H(v_1, ..., v_n)$ ,  $n \ge 1$  and a linear operator  $\mathcal{L}$  we set

$$\mathcal{C}[H,\mathcal{L}](v_1,...,v_n) = -\mathcal{L}(H(v_1,...,v_n)) + \sum_{i=1}^n D_i H(v_1,...,v_n) \cdot \mathcal{L}v_i$$

where  $D_i H$  stands for the partial differential of H with respect to the variable  $v_i$ . We define the iterated commutator  $\mathcal{C}^k[H,\mathcal{L}]$  recursively by

$$\mathcal{C}^{k+1}[H,\mathcal{L}] = \mathcal{C}^{k}[\mathcal{C}[H,\mathcal{L}],\mathcal{L}], \quad \mathcal{C}^{0}[H,\mathcal{L}] = H.$$

Let us illustrate this definition with our example (3.3). With the above notation (and  $\mathcal{L} = \Delta$ ) one has for sufficiently smooth f:

$$f(e^{\xi\Delta}v) = \sum_{\ell=0}^{r} \frac{\xi^{\ell}}{\ell!} e^{\xi\Delta} \mathcal{C}^{\ell}[f,\Delta](v) + \mathcal{O}(\xi^{r+1}\mathcal{C}^{r+1}[f,\Delta](v)).$$
(3.12)

This linearisation produces an error of the form

$$\mathcal{C}^{r+1}[f,\Delta](v)$$

which in general requires less regularity and less compatibility conditions on the boundary than does a classical Taylor-series expansion of the propagator  $e^{\xi \Delta}$ . Indeed, up to order r we have

$$e^{\xi\Delta} = \sum_{k \le r} \frac{\xi^k}{k!} \Delta^k + \mathcal{O}(\xi^{r+1} \Delta^{r+1}), \qquad (3.13)$$

whereby

$$D(\mathcal{C}^{r+1}[f,\Delta]) \supset D(\Delta^{k+1}).$$

Hence, the commutator expansion (3.12) requires less regularity assumptions and compatibility conditions than classical Taylor expansion of the the propagator  $e^{\xi\Delta}$ . We illustrate this through an example in the following two remarks. This observation also holds true in the general setting (3.1).

Remark 3.1.3 In general it holds that

$$D(\mathcal{C}[f,\mathcal{L}]) \supset D(\mathcal{L}).$$

Let us for instance recall the example given in [RS21, Sec. 2.1] where  $\mathcal{L} = \Delta$  on the torus  $\mathbf{T}^d$  and  $f(u) = u^2$ . Then one has  $D(\mathcal{L}) = H^2$  (the classical Sobolev space) and

$$\mathcal{C}[f,\mathcal{L}](v) = -\sum_{k=1}^{d} (\partial_k v)^2.$$

If we choose to measure the error in  $L^2$ , in order to bound the above expression we need  $(\partial_k v)^2 \in L^2$ . This follows for  $v \in H^{1+\frac{d}{4}}$  by using the injection  $H^{1+\frac{d}{4}} \hookrightarrow L^4$ . Hence we have  $v \in H^{1+\frac{d}{4}} \subset D(\mathcal{C}[f,\mathcal{L}])$ . For an analysis in  $H^s$ -norm with  $s \ge 0$  we refer to [AB23a].

We note that the above also holds true when working on a bounded domain of  $\mathbf{R}^d$ , equipped with homogeneous Dirichlet boundary conditions. Indeed, the same argument as above holds with  $D(\Delta) = (H^2 \cap H_0^1)(\Omega)$ , and where  $v \in (H^{1+\frac{d}{4}} \cap H_0^1)(\Omega) \subset D(\mathcal{C}[f, \Delta]).$ 

**Remark 3.1.4** The commutator term  $C^{r+1}[f, \Delta](v)$  in general asks for less regularity and compatibility conditions at the boundary than the classical Taylor's expansion (3.13). Indeed, in case of homogeneous Dirichlet boundary conditions

for example when r = 1, the commutator term  $C^2[f, \Delta](v)$  involves derivatives of at most second order. Namely, formally we have  $C^2[f, \Delta](v) = O(\Delta v)$ , which requires  $v \in D(\Delta) = (H^2 \cap H_0^1)(\Omega)$ . A. classical second-order Taylor's expansion on the other hand asks for  $v \in D(\Delta^2)$ , which does not only requires v to have zero trace on the boundary, but also its second derivatives (see also Remark 3.1.1).

Let us now turn back to presenting the main idea of our new schemes on the example (3.3): Plugging (3.12) into (3.10), we get

$$u(t) = e^{t\Delta}v + \sum_{k,\ell \le r} \int_0^t e^{(t-\xi)\Delta} \frac{A(\xi)^k}{k!} \frac{\xi^\ell}{\ell!} \Big( e^{\xi\Delta} \mathcal{C}^\ell[f^{(k)}, \Delta](v) \Big) V d\xi + \mathcal{O}(t^{r+2} \mathcal{C}^{r+1}[f^{(k)}, \Delta](v))$$
(3.14)

where the leading error term is driven by the commutator

$$\mathcal{O}(t^{r+2}\mathcal{C}^{r+1}[f^{(k)},\Delta](v))$$

In general, the latter is more important (i.e., worse in terms of regularity assumptions) than the ones introduce by the approximation of A(t) (cf. (3.9)) which is a polynomial in the initial condition v involving only bounded operators. Also, one can mention that in general  $\mathcal{L}_o$  is more singular than the  $\mathcal{B}_o^{\mathfrak{l}}$ , where the (linear) operators  $\mathcal{B}_o^{\mathfrak{l}}$  are given by the structure of the nonlinearity (3.2).

From the approximation (3.14), we can collect iterated integrals for building an approximate solution  $w^{r}(v, t)$  up to order r. For example, when r = 2, we get

$$w^{2}(v,t) = e^{t\Delta}v + \int_{0}^{t} e^{(t-\xi)\Delta} \left( e^{\xi\Delta}f(v) \right) V d\xi$$

$$+ \int_{0}^{t} \xi e^{(t-\xi)\Delta} \left( e^{\xi\Delta}\mathcal{C}[f,\Delta](v) \right) V d\xi$$

$$+ \int_{0}^{t} e^{(t-\xi)\Delta} \left( \int_{0}^{\xi} e^{(\xi-\xi_{1})\Delta} \left( e^{\xi_{1}\Delta}f(v) \right) V d\xi_{1} \right) \left( e^{\xi\Delta}f^{(1)}(v) \right) V d\xi.$$
(3.15)

**Remark 3.1.5** The remaining challenge lies in embedding the central oscillations triggered by

$$e^{(t-\xi)\Delta} \left( e^{\xi\Delta} f(v) \right) V, \quad e^{(t-\xi)\Delta} \left( e^{\xi\Delta} \mathcal{C}[f,\Delta](v) \right) V,$$
$$e^{(t-\xi)\Delta} \left( \int_0^{\xi} e^{(\xi-\xi_1)\Delta} \left( e^{\xi_1\Delta} f(v) \right) V d\xi_1 \right) \left( e^{\xi\Delta} f^{(1)}(v) \right) V$$

into the numerical discretisation.

For this purpose we will introduce a recursive map  $\Pi$  defined on a set of trees  $\mathcal{T}$  to encode these iterated integrals. The set  $\mathcal{T}$  will be decorated trees whose formalism has been introduced in [BHZ19] and has been extensively used in [BCCH20] for giving wellposedness results for a large class of Stochastic partial differential equations (SPDEs). After the recent work [BS22], it is the second time that one uses this formalism also in context of Numerical Analysis. The iterated integrals for  $w^2$  are given by trees of size two (having at most two edges)  $\mathcal{T}^2$ :

$$\mathcal{T}^2 = \left\{ \stackrel{\circ}{\underset{\bullet}{\downarrow}}, \stackrel{\circ}{\underset{\bullet}{\downarrow}}, \stackrel{\circ}{\underset{\bullet}{\downarrow}} \right\}.$$

Brown edges encode integrals of the form  $\int_0^t e^{(t-\xi)\Delta} \cdots d\xi$ , white nodes the nonlinearity  $e^{\xi\Delta}f(v)V$  and the nodes marked with a cross correspond to  $\xi(e^{\xi\Delta}\mathcal{C}[f,\Delta](v))V$ . Incoming edges to a grey dot induce derivatives on f which will allow us to encode the third term in (3.15).

Our decorated trees strongly differ from the ones introduced in [BS22]: The scheme introduced in [BS22] heavily relies on Fourier series expansion techniques and is therefore restricted to periodic boundary conditions. In particular, the tree structure in [BS22] is based on Fourier decorations. The latter allowed us to easily filter out the dominant frequency interactions in the system based on the underlying structure of resonances of the PDE. In order to handle the general setting (3.1), and in particular deal with non periodic boundary conditions and non polynomial nonlinearities, we can, however, not make use of the previously proposed resonance analysis. Instead, we have to introduce new tools based on the analysis of the operator interactions and an extensive use of the commutator based approximation (3.12). Let us illustrate this analysis on the first iterated integral that we denote by  $(\Pi \overset{\circ}{\downarrow})(t)$ . It takes the following form

$$\left(\Pi \bigcup_{i=1}^{0} \right)(t) = \int_{0}^{t} e^{(t-\xi)\Delta} \prod_{i=1}^{2} e^{\xi \mathcal{L}_{i}} u_{i} d\xi,$$
$$\mathcal{L}_{1} = \Delta, \quad u_{1} = f(v), \quad \mathcal{L}_{2} = 0, \quad u_{2} = V.$$

The first step is to distribute  $e^{(t-\xi)\Delta}$  on the product  $\prod_{i=1}^2 e^{\xi \mathcal{L}_i} u_i$ :

$$e^{(t-\xi)\Delta} \left( \prod_{i=1}^{2} e^{\xi \mathcal{L}_{i}} u_{i} \right) = \prod_{i=1}^{2} e^{\xi \mathcal{L}_{i} + (t-\xi)\Delta} u_{i}$$

$$+ \sum_{1 \le k \le r} \frac{(t-\xi)^{k}}{k!} \mathcal{C}^{k} [(e^{t\Delta} \mathcal{M}_{\{1\}})(e^{(t-\xi)\Delta} \mathcal{M}_{\{2\}}), \Delta](u_{1}, u_{2})$$

$$+ \mathcal{O} (t^{r+1} \mathcal{C}^{r+1} [\mathcal{M}_{\{1,2\}}, \Delta](u_{1}, u_{2}))$$
(3.16)

where we have used the following notations: Let  $(u_i)_{i \in I}$  and  $(v_j)_{j \in J}$  be two finite sets of functions and  $\mathcal{L}_1$  and  $\mathcal{L}_2$  two linear operators. Then we set

$$(\mathcal{L}_1 \mathcal{M}_I \mathcal{L}_2 \mathcal{M}_J)(u_{i \in I}, v_{j \in J}) := \mathcal{L}_1 \left(\prod_{i \in I} u_i\right) \mathcal{L}_2 \left(\prod_{j \in J} u_j\right)$$

and

$$\mathcal{C}[\mathcal{M}_I, \mathcal{L}_1](u_{i \in I}) := -\mathcal{L}_1 \mathcal{M}_I(u_{i \in I}) + \sum_{j \in I} \mathcal{M}_I(..., \mathcal{L}_1 u_j, ...).$$

**Remark 3.1.6** We can notice that the term  $e^{-\xi\Delta}$  does, however, not make sense (recall that  $\xi \ge 0$ ). In the dispersive setting of [BS22], one did not have such an issue (as for instance  $e^{i\xi\Delta}$  is well defined for all  $\xi \in \mathbf{R}$ ) and hence one could perform the resonance analysis only on the terms depending on  $\xi$ . In our general setting (3.1) we have to be more careful and take the oscillations of the full operator  $e^{(t-\xi)\Delta}$  into account.

The second step is to identify the dominant part in the operators  $\mathcal{L}_i - \Delta$  in (3.16). For this purpose we set  $\mathcal{A}_i = \mathcal{L}_i - \Delta$ and introduce the splitting

$$\mathcal{A}_i = a_i \mathcal{L}_{\text{dom}} + \mathcal{A}_{\text{low}}^i, \qquad D(\mathcal{L}_{\text{dom}}) \subset D(\mathcal{A}_{\text{low}}^i)$$

where  $a_i \in \{0, 1\}$ . The purpose of this splitting lies in the fact that in the construction of our schemes we will separate the dominant from the lower order oscillations and only embed the dominant parts exactly into our numerical discretization, while Taylor expanding the lower order parts. Note that in the above example one for instance has that

$$\mathcal{A}_1 = 0, \quad \mathcal{A}_2 = -\Delta,$$

and hence,  $a_1 = 0$ ,  $a_2 = 1$ ,  $\mathcal{A}_{low}^i = 0$  for  $i \in \{1, 2\}$  and  $\mathcal{L}_{dom} = -\Delta$ . In this simple case one does not need any further approximation and one can write the following approximation at order r:

$$\left(\Pi^{r}\right)(t) = \sum_{0 \le k \le r} \int_{0}^{t} \frac{(t-\xi)^{k}}{k!} \mathcal{C}^{k}[(e^{t\Delta}\mathcal{M}_{\{1\}})(e^{(t-\xi)\Delta}\mathcal{M}_{\{2\}}), \Delta](u_{1}, u_{2})d\xi$$

**Remark 3.1.7** In most applications, one does not have to go through the second step of the approximation like the example shown above. The second step is needed for linear combination of operators like  $\partial_x^2 + \partial_x$ . One may have to approximate the following iterated integral (with notation  $\Delta = \partial_x^2$ ):

$$\int_{0}^{t} e^{(t-\xi)\Delta} \left( e^{\xi(\Delta+\partial_{x})} v_{2} \right) V d\xi$$

This integral is coming from a system of the form:

$$\partial_t u_1 - \Delta u_1 = u_2 V, \quad u_1(0) = v_1,$$
(3.17)

$$\partial_t u_2 - \Delta u_2 - \partial_x u_2 = 0, \quad u_2(0) = v_2.$$

Then, from (3.16), we get

$$\sum_{1\leq k\leq r}\frac{(t-\xi)^k}{k!}\mathcal{C}^k[(e^{t\Delta+\xi\partial_x}\mathcal{M}_{\{1\}})(e^{(t-\xi)\Delta}\mathcal{M}_{\{2\}}),\Delta](v_2,V).$$

Now,  $\mathcal{A}^1_{\text{low}} = \partial_x$  and one has to Taylor-expand the term  $e^{\xi \partial_x}$ . This leads to the following approximation:

$$\int_0^t e^{(t-\xi)\Delta} \Big( e^{\xi(\Delta+\partial_x)} v_2 \Big) V d\xi = \int_0^t \sum_{0 \le k+\ell \le r} \frac{(t-\xi)^k}{k!} \frac{\xi^\ell}{\ell!}$$
$$\mathcal{C}^k[(e^{t\Delta}\mathcal{M}_{\{1\}})(e^{(t-\xi)\Delta}\mathcal{M}_{\{2\}}), \Delta](\partial_x^\ell v_2, V) d\xi$$
$$+ \sum_{k+\ell=r+1} \mathcal{O}\Big(t^{r+2}\mathcal{C}^k[\mathcal{M}_{\{1,2\}}, \Delta](\partial_x^\ell v_2, V)\Big).$$

In the next section, we will present our main strategy in the general setting (3.1).

#### 3.1.2 Main strategy of the general numerical scheme

Duhamel's formulation of (3.1) is given by

$$u_o(\tau) = e^{\tau \mathcal{L}_o} v_o + \int_0^\tau e^{(\tau - \xi)\mathcal{L}_o} \sum_{\mathfrak{l} \in \mathfrak{L}_-} \Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}}) V_{\mathfrak{l}} d\xi.$$
(3.18)

By iterating this formulation together with (3.12) and (3.9), we get iterated integrals that can be represented via decorated trees. Let r + 1 be the order of the scheme, we first truncate this expansion and get the following approximation of the solution:

$$w_o^r(\hat{\mathbf{v}},\tau) = e^{\tau \mathcal{L}_o} v_o + \sum_{T \in \mathcal{V}^r} (\Pi T)(\hat{\mathbf{v}},\tau)$$
(3.19)

where  $\hat{\mathbf{v}} = (\mathbf{v}, \mathbf{V})$ ,  $\mathbf{v} = (v_o)_{o \in \mathfrak{L}_+}$ ,  $\mathbf{V} = (V_l)_{l \in \mathfrak{L}_-}$ ,  $\mathcal{V}^r$  is a finite set of decorated trees, and  $(\Pi T)(\hat{\mathbf{v}}, \tau)$  represents the iterated integral associated to T. The exponent r in  $\mathcal{V}^r$  means that we consider only trees of size r which are the trees producing an iterated integral with r integrals. Proposition 3.4.1 shows that  $w_o^r$  solves (3.18) with an error involving  $\mathcal{C}^{r+1}[f, \mathcal{L}_o]$  with f a function coming from the coefficients  $\Psi_o$ . The main improvement lies in the error structure: The error  $\mathcal{C}^{r+1}[f, \mathcal{L}_o]$  is in general better than  $\mathcal{L}_o^r$  in the sense of requiring lower regularity assumptions on the solution.

The main difficulty boils down to developing for every  $T \in \mathcal{V}^r$  a suitable approximation to the iterated integrals  $(\Pi T)(\hat{\mathbf{v}}, \tau)$  with the aim of minimising the local error structure (in the sense of regularity), see also Remark 3.1.5. In order to achieve this, the key idea lies in commutator based approximations (3.12) which replace the resonance analysis given in [BS22]. The approximation of  $(\Pi T)$  is given by a new map on decorated trees denoted by  $\Pi_A^r$  where r + 1 is the order of the scheme and A is a domain corresponding to the a priori assumed regularity of  $\hat{\mathbf{v}}$ . Our general scheme takes the form

$$w_{o,A}^{r}(\hat{\mathbf{v}},\tau) = e^{\tau \mathcal{L}_{o}} v_{o} + \sum_{T \in \mathcal{V}^{r}} (\Pi_{A}^{r} T)(\hat{\mathbf{v}},\tau)$$
(3.20)

where the map  $\Pi_A^r T$  is a low regularity approximation of order r of the map  $\Pi T$  in (3.19) in the sense that

$$(\Pi T - \Pi_A^r T)(\hat{\mathbf{v}}, \tau) = \mathcal{O}\big(\tau^{r+2} \mathcal{L}^r_{\text{low}}(T, \hat{\mathbf{v}}, A)\big).$$
(3.21)

Here  $\mathcal{L}_{low}^r(T, \hat{\mathbf{v}}, A)$  involves nested commutators that require in general less regularity than powers of the full operator  $\mathcal{L}_o^r$ (see also Remark 3.1.3). The scheme (3.20) and the local error approximations (3.21) are the main results of this work (see Theorem 3.4.3). The approximation  $\Pi_A^r$  is constructed from a character  $\Pi_A$  defined on the vector space  $\mathcal{H}$  spanned by decorated forests taking values in a space  $\mathcal{C}$  that will depend on the operators  $\mathcal{L}_o$  and  $\hat{\mathbf{v}}$ . We add a decoration r at the root with the meaning that we will consider an approximation of order r. This is performed by the symbol  $\mathcal{D}^r$  that gives  $\Pi_A \mathcal{D}^r(T) = \Pi_A^r T$ .

As in [BS22], we define the map  $\Pi_A$  recursively from an operator  $\mathcal{K}$ . This operator computes a suitable approximation (matching the regularity of the solution) of the integrals introduced by the iteration of Duhamel's formula. Here the main difference is that we perform the analysis directly on the operator without going into Fourier space. We single out the dominant parts of the operators we are dealing with. These dominant parts are integrated exactly and only the lower order terms are approximated which allows for an improved local error structure compared to classical approaches. For the local error analysis, a structure of Hopf algebra and comodule was used in [BS22] for singling out oscillations. It was based on a variant of the Butcher-Connes-Kreimer coproduct [But72, CK99, CK00] inspired by [BHZ19]. This structure can be seen as a deformation of the Butcher-Connes-Kreimer coproduct (see [BM23]). The central object used as in Quantum Field Theory for the local error analysis in [BS22] was a Birkhoff factorisation similar to the one presenting in [BHZ19] for recentering iterated integrals. This is put into perspective in the work [BEF20]. The context of commutators is much more involved than the one in Fourier space. We therefore do not use such a construction but rather the recursive formulae for computing the local error. It is given in Definition 3.3.8. Via Proposition 3.3.7, we illustrate what are the dominant operators involved in the computation of the local error.

#### 3.1.3 Outline of the paper

Let us give a short review of the content of this paper. In Section 3.2, we introduce the general algebraic framework by first defining a suitable vector space of decorated trees  $\hat{\mathcal{T}}$  and decorated forests  $\hat{\mathcal{H}}$ . Next, we set how to compute the dominant part of a set of operators (see Definition 3.2.1). We define then the dominant operators associated to a decorated forest (see Definition 3.2.3). Then, we introduced new spaces of decorated trees  $\mathcal{T}$  that we call approximated decorated trees. They carry an extra decoration r at the root and they represent approximation at order r of the corresponding iterated integrals. We also introduced new decorated forests based on the same construction.

In Section 3.3, we construct the approximation of the iterated integrals given by the character  $\Pi : \hat{\mathcal{H}} \to \mathcal{C}$  (see (3.38)) through the character  $\Pi_A : \mathcal{H} \to \mathcal{C}$  (see (3.42)) where A is a domain that is the regularity of  $\hat{\mathbf{v}}$ . This is the regularity assumed a priori before writing the scheme. The approximation  $\Pi_A$  relies on a recursive construction where the operator  $\mathcal{K}$  given in Definition 3.3.2 is heavily used. The local error analysis which is the error estimate on the difference between  $\Pi$ and its approximation  $\Pi_A$  is given in Theorem 3.3.9. It is built upon a recursive definition (see Definition 3.3.8) involving Taylor remainders of  $\Pi_A$  with commutators. They are given in Lemma 3.3.6. Proposition 3.3.7 gives the structures of the dominant operators appearing in the local error analysis. In Section 3.4, we introduce truncated series of decorated trees that solves up to order r equation 3.1 (see Proposition 3.4.1). Then, from this series built upon the character  $\Pi$ , one can write the general scheme (see Definition 3.57) which boils down to replace  $\Pi$  by its approximation  $\Pi_A$ . In the end, we compute its local error structure (see Theorem 3.4.3) based on the local error between  $\Pi$  and  $\Pi_A$  for each decorated tree that appears in the expansion of the scheme.

In Section 3.5, we illustrate the general framework on various applications. The main example is the Gross-Pitaevskii (GP) equation that includes both a potential and a rough initial data. With our general framework we derive a first and second order scheme for GP and carry out precisely its local error analysis. We also discuss adapted filter operators for the stability of the low regularity scheme in Section 3.5.2. The last example of this section is the Klein and Sine-Gordon equations which illustrated our framework for non-polynomial nonlinearities.

Acknowledgements. Y. A. B. and K. S. have received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 850941). Y. B. gratefully acknowledges funding support from the European Research Council (ERC) through the ERC Starting Grant Low Regularity Dynamics via Decorated Trees (LoRDeT), grant agreement No. 101075208. Y. B. thanks the Max Planck Institute for Mathematics in the Sciences (MiS) in Leipzig for supporting his research via a long stay in Leipzig from January to June 2022 where part of this work was written.

## 3.2 Decorated trees

In this section, we introduce the formalism of decorated trees that is used for describing the iterated integrals stemming from the iteration of the Duhamel's formula (3.18) of our main equation (3.1). Decorations on the nodes will encode monomials  $\xi^k$  and potentials  $V_l$  that appear in the equation whereas decorations on the edges encode the various integrals in time. The two spaces of importance are  $\hat{\mathcal{T}}$  (resp.  $\hat{\mathcal{H}}$ ) space of decorated trees (resp. forests). They correspond to iterated integrals without approximations. We define on these combinatorial objects the dominant operators that are used for the discretisation (see Definition 3.2.3). In the end, we consider the spaces  $\mathcal{T}$  and  $\mathcal{H}$  obtained by adding decorations at the root. This decoration gives the order at which we want to approximate these iterated integrals.

#### 3.2.1 Definitions and dominant operators

We assume a finite set  $\mathfrak{L}$  which has the following splitting  $\mathfrak{L} = \mathfrak{L}_+ \sqcup \mathfrak{L}_-$ . We consider a family of differential operators  $(\mathcal{L}_o)_{o \in \mathfrak{L}_+}$  indexed by  $\mathfrak{L}_+$  and a family of potentials  $(V_{\mathfrak{l}})_{\mathfrak{l} \in \mathfrak{L}_-}$  indexed by  $\mathfrak{L}_-$ . We suppose given  $(\mathfrak{L}_+^{o,\mathfrak{l}})_{(o,\mathfrak{l}) \in \mathfrak{L}_+ \times \mathfrak{L}_-}$  a collection of subsets of  $\mathfrak{L}_+$ . We define the set of decorated trees  $\hat{\mathcal{T}}$  as elements of the form  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{f}} = (T,\mathfrak{n},\mathfrak{f},\mathfrak{e})$  where

- T is a non-planar rooted tree with root  $\rho_T$ , node set  $N_T$  and edge set  $E_T$ . We denote the leaves of T by  $L_T$ .

- the map  $\mathfrak{e}: E_T \to \mathfrak{L}_+$  are edge decorations. An edge decorated by  $o \in \mathfrak{L}_+$  will encode an integral of the form  $\int_0^t e^{(t-s)\mathcal{L}_o}...ds.$
- the map  $\mathbf{n}: N_T \to \mathbf{N}$  are node decorations. They correspond to monomials of the form  $\xi^k, k \in \mathbf{N}$  that appear in iterated integrals.
- the map  $f: N_T \to \mathfrak{L}_-$  are node decorations encoding the various potentials  $V_{\mathfrak{l}}$ .

When the node decoration  $\mathfrak{n}$  is omitted, we will denote the decorated trees as  $T_{\mathfrak{e}}^{\mathfrak{f}}$ . This set of decorated trees is denoted by  $\hat{\mathcal{T}}_0$ . We say that  $\bar{T}_{\mathfrak{e}}^{\mathfrak{f}}$  is a decorated subtree of  $T_{\mathfrak{e}}^{\mathfrak{f}} \in \hat{\mathcal{T}}_0$  if  $\bar{T}$  is a subtree of T and the restriction of the decorations  $\mathfrak{e}$  and  $\mathfrak{f}$  of T to  $\bar{T}$  are given by  $\bar{\mathfrak{e}}$  and  $\bar{\mathfrak{f}}$ . We denote by  $\hat{\mathcal{P}}$  the set of planted trees  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{f}}$  that are decorated trees with only one edge connected to the root and with no decorations at the root. Similarly when the node decoration  $\mathfrak{n}$  is omitted we denote this set  $\hat{\mathcal{P}}_0$ . We denote by  $\hat{H}$  (and resp.  $\hat{H}_0$ ) the (unordered) forests composed of trees in  $\hat{\mathcal{P}}$  (and resp.  $\hat{\mathcal{P}}_0$ ) (including the empty forest denoted by  $\mathfrak{1}$ ). Their linear spans are denoted by  $\hat{\mathcal{H}}$  (and resp.  $\hat{\mathcal{H}}_0$ ).

In order to represent these decorated trees, we introduce a symbolic notation. An edge decorated by  $o \in \mathfrak{L}_+$  is denoted by  $\mathcal{I}_o$ . The symbol  $\mathcal{I}_o(\cdot) : \hat{\mathcal{T}} \to \hat{\mathcal{H}}$  is viewed as the operation that connects the root of a decorated tree to a new root with no decorations via an edge decorated by o. Any decorate tree T admits the following decomposition:

$$T = \lambda_{\mathfrak{l}}^{\ell} \prod_{i=1}^{m} \mathcal{I}_{o_i}(T_i) \tag{3.22}$$

where the  $T_i$  are decorated trees,  $\lambda_{\mathfrak{l}}^{\ell}$  corresponds to the decorations at the root with  $\ell \in \mathbf{N}$  and  $\mathfrak{l} \in \mathfrak{L}_-$ . The product in (3.22) is viewed as the forest product. This decomposition means that every decorated tree can be identified with a forest  $\prod_{i=1}^{m} \mathcal{I}_{o_i}(T_i)$  and a decoration  $\lambda_{\mathfrak{l}}^{\ell}$ . When  $\ell = 0$ , we will use the shorthand notation:  $\lambda_{\mathfrak{l}}^0 = \lambda_{\mathfrak{l}}$ . Below, we provide an example of such decorated trees:

$$\lambda_{l_{0}}^{\ell_{0}} \mathcal{I}_{o_{1}}(\lambda_{l_{1}}^{\ell_{1}}) \mathcal{I}_{o_{2}}(\lambda_{l_{0}}^{\ell_{2}}) \mathcal{I}_{o_{3}}(\lambda_{l_{0}}^{\ell_{3}}) = \underbrace{(\ell_{0}, \iota_{0})}^{(\ell_{1}, \iota_{1})} (\ell_{0}, \iota_{0}) = \underbrace{(\ell_{0}, \iota_{0})}^{(\ell_{0}, \iota_{0})} (\ell_{0}, \iota_{0})} (\ell_{0$$

Iterated integrals will be associated to these decorated trees. In order to approximate them numerically we will have to resolve the underlying oscillations by extracting the dominant parts of the leading operators. In the next definition, we introduce these dominant parts.

**Definition 3.2.1** Let  $\mathcal{A}_i, i \in \{1, ..., m\}$  be a finite set of operators with domains  $D(\mathcal{A}_i)$  in the sense that  $\mathcal{A}_i : D(\mathcal{A}_i) \subseteq X \to X$ . We define  $\mathcal{P}_{\text{dom}}(\{\mathcal{A}_1, ..., \mathcal{A}_m\})$  as follows: We first consider  $I \subset \{1, ..., m\}$  such that  $\mathcal{A}_i, i \in I$  are of smaller domain in the sense that:

 $D(\mathcal{A}_i) = D(\mathcal{A}_j), i, j \in I, \quad D(\mathcal{A}_i) \subsetneq D(\mathcal{A}_\ell), \ \ell \in \{1, ..., m\} \setminus I, i \in I.$ 

If there exists an operator  $\mathcal{L}_{dom}$  such that for every  $i \in I$ 

$$\mathcal{A}_i = \mathcal{L}_{\rm dom} + \mathcal{A}^i_{\rm low}. \tag{3.24}$$

where  $\mathcal{L}_{\text{dom}}$  satisfies  $D(\mathcal{L}_{\text{dom}}) \subsetneq D(\mathcal{A}_{\text{low}}^i)$ , then we set

$$\mathcal{P}_{\mathrm{dom}}(\{\mathcal{A}_1,...,\mathcal{A}_m\})=\mathcal{L}_{\mathrm{dom}}.$$

Otherwise, it is equal to 0.

**Remark 3.2.2** The identity (3.24) has to be understood as a decomposition between lower and upper part of the operators  $\mathcal{A}_i$ . What is important in order to have a non-zero dominant part is to have a factorisation with some operator  $\mathcal{L}_{dom}$ . It is similar to the approach in Fourier space presented in [BS22, Def. 2.2] when one was looking at the form of the higher monomials of polynomials in the frequencies.

**Example 6** We illustrate the previous definition with  $\{0, \mathcal{L}\}$ , where  $\mathcal{L}$  has the domain  $D(\mathcal{L}) \neq X$ . Then

$$\mathcal{P}_{dom}(\{0,\mathcal{L}\}) = \mathcal{L}$$

and for  $\{0, -\mathcal{L}, \mathcal{L}\}$ , one has
The latter follows as  $D(-\mathcal{L}) = D(\mathcal{L})$  and the coefficients in front of  $\mathcal{L}$  are different (1 and -1).

Given a decorated tree, we want to compute its dominant part and lower part using the map  $\mathcal{P}_{dom}$ . We suppose given a space of operators  $\mathcal{O}_{\sqrt{1}}$  in X and we denote by  $\mathcal{OS}$  the collection of finite sets of operators belonging to  $\mathcal{O}_{\sqrt{1}}$ . Given two sets  $\mathcal{A}_1$  and  $\mathcal{A}_2$ , we denote their union by  $\mathcal{A}_1 \cup \mathcal{A}_2$ . Given a finite set of operators  $\mathcal{A}$  and an operator  $\mathcal{L}$ , we define  $\mathcal{L} \oplus \mathcal{A}$ as the set where  $\mathcal{L}$  is added to each component of  $\mathcal{A}$ :

$$\mathcal{L} \oplus \mathcal{A} = \cup_{\mathcal{L}_o \in \mathcal{A}} \{\mathcal{L} + \mathcal{L}_o\}$$

The term  $\mathcal{A} \ominus \mathcal{L}$  is computed as follows for each component  $\mathcal{L}_o$  of  $\mathcal{A}$ . If the dominant part of  $\mathcal{L}_o$  is equal to  $\mathcal{L}$ , in the sense that one has the following decomposition:

$$\mathcal{L}_o = \mathcal{L} + \bar{\mathcal{L}} \tag{3.25}$$

with  $D(\mathcal{L}) \subsetneq D(\overline{\mathcal{L}})$ , then we replace  $\mathcal{L}_o$  by  $\overline{\mathcal{L}}$ . Otherwise, we keep  $\mathcal{L}_o$ . We define  $\max(\mathcal{L}, \mathcal{A})$  as the operation that replaces each element  $\mathcal{L}_o$  satisfying (3.25) by  $\mathcal{L}$  otherwise we replace it by 0. Equipped with these notations, we are able to state the main definition for computing the dominant part associated to a decorated tree:

**Definition 3.2.3** We recursively define  $\mathcal{R}_{dom}^{o}: \hat{\mathcal{T}}_{0} \to \mathcal{OS}, \mathcal{R}_{dom}: \hat{\mathcal{P}}_{0} \to \mathcal{OS}, \mathcal{R}_{low}: \hat{\mathcal{P}}_{0} \to \mathcal{OS}$  and  $\mathcal{L}_{dom}: \hat{\mathcal{P}}_{0} \to \mathcal{O}_{s}$  as:

$$\begin{aligned} \mathcal{R}^{o}_{\mathrm{dom}}(\lambda_{\mathrm{f}}\prod_{i}\mathcal{I}_{o_{i}}(T_{i})) &= \{\mathcal{L}_{\bar{o}}, \, \bar{o} \in \mathfrak{L}^{o,\mathrm{f}}_{+}\} \cup \bigcup_{i}\mathcal{R}_{\mathrm{dom}}(\mathcal{I}_{o_{i}}(T_{i})), \\ \mathcal{L}_{\mathrm{dom}}(\mathcal{I}_{o}(T)) &= \mathcal{P}_{\mathrm{dom}}(-\mathcal{L}_{o} \oplus \mathcal{R}^{o}_{\mathrm{dom}}(T)), \\ \mathcal{R}_{\mathrm{dom}}(\mathcal{I}_{o}(T)) &= \mathcal{L}_{o} \oplus \max(\mathcal{L}_{\mathrm{dom}}, -\mathcal{L}_{o} \oplus \mathcal{R}^{o}_{\mathrm{dom}}(T)), \\ \mathcal{R}_{\mathrm{low}}(\mathcal{I}_{o}(T)) &= (\mathrm{id} \ominus \mathcal{P}_{\mathrm{dom}})(-\mathcal{L}_{o} \oplus \mathcal{R}^{o}_{\mathrm{dom}}(T)). \end{aligned}$$

We extend these maps to  $\hat{\mathcal{T}}$  and  $\hat{\mathcal{P}}$  by ignoring the node decorations  $\mathfrak{n}$ .

**Example 7** We illustrate the previous abstract definitions on some trees stemming from the cubic nonlinear Schrödinger equation

$$i\partial_t u + \Delta u = |u|^2 u, \quad u_0 = v \tag{3.26}$$

set on  $\mathbf{R}^d$ ,  $1 \le d \le 3$ . For the equation (3.26), we have two variables which are u and  $\bar{u}$ . Thus, we rewrite (3.26) in the following form:

$$\partial_t u_o - \mathcal{L}_o u_o = -i(u_o)^2 u_{\bar{o}}, \quad u_o(0) = v_o$$
  
$$\partial_t u_{\bar{o}} - \mathcal{L}_{\bar{o}} u_{\bar{o}} = i(u_{\bar{o}})^2 u_o, \quad u_{\bar{o}}(0) = v_{\bar{o}}$$
(3.27)

where one has  $\mathcal{L}_o = i\Delta$ ,  $\mathfrak{L}_+ = \{o, \bar{o}\}$ ,  $\mathfrak{L}_- = \{0\}$ ,  $\mathfrak{L}_+^{o,0} = \mathfrak{L}_+^{\bar{o},0} = \{o, \bar{o}\}$  with  $V_0 = 1$ ,  $\mathcal{B}_o^0 = \mathcal{B}_{\bar{o}}^0 = \mathrm{id}$ ,  $\mathcal{L}_o = \mathcal{L}$ ,  $\mathcal{L}_{\bar{o}} = -\mathcal{L}$  and  $X = L^2(\mathbf{R}^d)$ ,  $D(\mathcal{L}) = H^2$ ,  $u_o = u$ ,  $u_{\bar{o}} = \bar{u}$ ,  $v_o = v$ ,  $v_{\bar{o}} = \bar{v}$ . The nonlinearities are given by:  $f_{o,o}^0(u) = -iu^2$ ,  $f_{o,\bar{o}}^0(\bar{u}) = \bar{u}$ ,  $f_{\bar{o},\bar{o}}^0(\bar{u}) = +i\bar{u}^2$  and  $f_{\bar{o},o}^0(u) = u$ . Let us recall the meaning of the subscripts in  $f_{o,\bar{o}}^0$ . The 0 corresponds to the driver  $V_0$ ,  $\bar{o}$  that it appears in the equation for  $\bar{u}$  and the last subscript says that  $f_{o,\bar{o}}^0$  depends only on the variable u. Next we consider the following decorated tree

$$T = \mathcal{I}_o(\lambda_0) = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$
(3.28)

which encodes the iterated integral:

$$-i\int_{0}^{t} e^{(t-s)\mathcal{L}} \left( \left( e^{s\mathcal{L}}v^{2} \right) \left( e^{-s\mathcal{L}}\bar{v} \right) \right) ds.$$
(3.29)

The brown edge stands for the integral in time  $\int_0^t e^{(t-s)\mathcal{L}}...ds$ . The white dot associated to the potential  $V_0$  encodes the product  $-i(e^{s\mathcal{L}}v^2)(e^{-s\mathcal{L}}\bar{v})$ . Indeed, it is connected via an edge associated to  $\mathcal{L}$ , so we know that we have to consider the nonlinearity associated to  $V_0$  in the equation for u. One gets from the definition of  $\mathcal{L}^{o,0}_+$ 

$$\mathcal{R}^o_{\mathrm{dom}}(\mathbf{0}) = \{-\mathcal{L}, \mathcal{L}\}.$$

Then applying the previous definitions (see Definition 3.2.3), one gets

$$egin{aligned} -\mathcal{L}\oplus\mathcal{R}^o_{ ext{dom}}(\circ) &= \{-2\mathcal{L},0\}, \quad \mathcal{L}_{ ext{dom}}(T) = \mathcal{P}_{ ext{dom}}(\{-2\mathcal{L},0\}) = -2\mathcal{L}, \ \mathcal{R}_{ ext{dom}}(T) &= \mathcal{L}\oplus\max(\mathcal{L}_{ ext{dom}},-\mathcal{L}\oplus\{-\mathcal{L},\mathcal{L}\}) = \{-\mathcal{L},\mathcal{L}\} \end{aligned}$$

$$\mathcal{R}_{\text{low}}(T) = (\text{id} \ominus \mathcal{P}_{\text{dom}})(\{-2\mathcal{L}, 0\}) = \{-2\mathcal{L}, 0\} \ominus -2\mathcal{L} = \{0\}$$

**Remark 3.2.4** One can notice that some of the notations that we have introduced are similar to the one used in [BS22] which took a direct inspiration from the algebraic structures developed for singular SPDEs in [BHZ19, BCCH20]. Our decorations are however different to [BS22], because we focus on a more general set up no longer restricted to periodic dispersive equations:

- The conjugate operator was encoded in [BS22] with edge decorations in  $\{-1, 1\}$  which allows the computation of the frequency interactions. Now, we have a larger set of operators by having a bigger set  $\mathcal{L}_+$  with elements such as  $\overline{o}$  like in the NLS example. One has  $\mathcal{L}_{\overline{o}} = \overline{\mathcal{L}} = \mathcal{L}_{(o,-1)}$  where the last identity corresponds to the old notation coming from [BS22].
- Another change on the edge decorations is that in [BS22] certain edges correspond to some integrals in time others not. This excludes the parabolic case. If we rewrite an integral of the form  $\int_0^t e^{(t-s)\mathcal{L}} \cdots ds$  into  $e^{t\mathcal{L}} \int_0^t e^{-s\mathcal{L}} \cdots ds$  then it does not make sense when  $\mathcal{L} = \Delta$ . We recall one decorated tree coming from [BS22]:

$$= -ie^{-itk^2} \int_0^t e^{isk^2} \left( e^{isk_1^2} e^{-isk_2^2} e^{-isk_3^2} \right) ds,$$
(3.30)

where  $k = -k_1 + k_2 + k_3$ , the leaves are decorated by the frequencies  $k_1, k_2, k_3$  and the inner nodes are decorated by k. The blue edge encodes an integral in time  $-i \int_0^t e^{isk^2} \cdots ds$ , the brown edges are used for a factor  $e^{-isk^2}$  and the dashed brown edges are for  $e^{isk^2}$ . If we consider the same integral not in Fourier mode, one can rewrite it as

$$= -i \int_0^t e^{(t-s)\mathcal{L}} \left( \left( e^{s\mathcal{L}} v \right)^2 \left( e^{-s\mathcal{L}} \overline{v} \right) \right) ds$$

$$(3.31)$$

where  $\mathcal{L} = i\Delta$ , the blue dot corresponds to  $\bar{v}$  and the blue edge now encodes an integral of the form  $\int_0^t e^{(t-s)\mathcal{L}} \cdots ds$ . In (3.31), we have to incorporate the initial data v and  $\bar{v}$  in the definition while they are implicit when the integral is written in Fourier space in (3.30) (one has just to multiply the integral with  $\bar{v}_{k_1}\hat{v}_{k_2}\hat{v}_{k_3}$ ). In fact, the coding given by (3.31) is sufficient for cubic NLS, but not for the general case we have in mind that contains potentials and non-polynomial nonlinearities.

— We add new decorations at the nodes stemming from the drivers  $V_{\mathfrak{l}}$  of (3.1). The initial conditions are associated to the drivers and the equations we are using via the sets  $\mathfrak{L}^{o,\mathfrak{l}}_+$ . Let us take for example a nonlinearity of the form

 $f(u)g(\overline{u})V_{\mathfrak{l}}.$ 

Then, after inserting the approximation  $u(s) = e^{s\mathcal{L}}v + A(s)$ , one gets thanks to (3.12)

$$\left(e^{s\mathcal{L}}f(v)\right)\left(e^{s\overline{\mathcal{L}}}g(\overline{v})\right)V_0 \equiv \circ = \lambda_0.$$
(3.32)

Next if we integrate by  $\int_0^t e^{(t-s)\mathcal{L}} \cdots ds$  encoded by a brown edge, we obtain the following integral when  $f(u) = u^2$  and  $g(\bar{u}) = \bar{u}$ :

$$\mathcal{I}_{o}(\lambda_{0}) = \int_{0}^{\tau} e^{i(\tau-s)\Delta} \left( \left( e^{is\Delta} v^{2} \right) \left( e^{-is\Delta} \bar{v} \right) \right) ds$$
(3.33)

The main difference between (3.31) and (3.33) is  $(e^{is\Delta}v)^2$  which is replaced by  $e^{is\Delta}v^2$ . We are dealing in (3.33) with integrals that contain less factors of the form  $e^{is\Delta}v$ .

**Example 8** As a second example, let us consider the Gross–Pitaevskii equation. The main difference with the cubic NLS equation (3.26) is the adjunction of a potential V. The equation takes the form:

$$i\partial_t u + \Delta u = |u|^2 u + uV, \quad u_0 = v \tag{3.34}$$

set on  $\mathbf{R}^d$ ,  $d \leq 3$ . Now, one has  $\mathfrak{L}_- = \{0, 1\}$ ,  $\mathfrak{L}_+^{o,1} = \{o\}$ ,  $\mathfrak{L}_+^{\overline{o},1} = \{\overline{o}\}$  with  $V_1 = V$ . The other sets remain the same, and are defined in Example 7. The new nonlinearities coming from the added potential term are given by:  $f_{o,o}^1(u) = -iu$  and  $f_{\overline{o},\overline{o}}^1(\overline{u}) = i\overline{u}$ . Hence, together with the nonlinearities defined in (3.26), we indeed recover the nonlinear terms of (3.34)

from the general form (3.1):

$$\sum_{\mathfrak{l}\in\mathfrak{L}_{-}} \Psi_{o}^{\mathfrak{l}}(\mathbf{u}_{o}^{\mathfrak{l}})V_{\mathfrak{l}}(x) = \Psi_{o}^{0}(\mathbf{u}_{o}^{0})V_{0}(x) + \Psi_{o}^{1}(\mathbf{u}_{o}^{1})V_{1}(x)$$
$$= f_{o,o}^{0}(u_{o})f_{o,\bar{o}}^{0}(u_{\bar{o}}) + f_{o,o}^{1}(u_{o})V$$
$$= -i(u_{o})^{2}u_{\bar{o}} - iu_{o}V,$$

where we recall that we set  $\mathcal{B}_{o}^{1} = \mathcal{B}_{\bar{o}}^{1} = \text{id.}$  We rewrite equation (3.34) into

$$\frac{\partial_t u_o - \mathcal{L}_o u_o}{\partial_t u_{\bar{o}} - \mathcal{L}_{\bar{o}} u_{\bar{o}}} = -i(u_o)^2 u_{\bar{o}} - iu_o V, \quad u_o(0) = v_o \\
\frac{\partial_t u_{\bar{o}} - \mathcal{L}_{\bar{o}} u_{\bar{o}}}{\partial_t u_{\bar{o}} - \mathcal{L}_{\bar{o}} u_{\bar{o}}} = i(u_{\bar{o}})^2 u_o + iu_{\bar{o}} V, \quad u_{\bar{o}}(0) = v_{\bar{o}}$$
(3.35)

where for simplicity we have assumed a real potential  $\overline{V} = V$ . The central iterated integrals up to second order then take the form

$$\begin{split} \mathcal{I}_{o}(\lambda_{1}) &= \int_{0}^{t} e^{(t-s)\mathcal{L}} \left( \left( e^{s\mathcal{L}}v \right) V \right) ds \\ \mathcal{I}_{\overline{o}}(\lambda_{1}) &= \int_{0}^{t} e^{(s-t)\mathcal{L}} \left( \left( e^{-s\mathcal{L}}\overline{v} \right) V \right) ds \\ \mathcal{I}_{o}(\lambda_{0}\mathcal{I}_{o}(\lambda_{1})) &= \int_{0}^{t} e^{(t-s)\mathcal{L}} \left( \left( \int_{0}^{s} e^{(s-r)\mathcal{L}} \left( \left( e^{r\mathcal{L}}v \right) V \right) dr \right) \left( e^{s\mathcal{L}}v \right) \left( e^{-s\mathcal{L}}\overline{v} \right) \right) ds \\ \mathcal{I}_{o}(\lambda_{0}\mathcal{I}_{\overline{o}}(\lambda_{1})) &= \int_{0}^{t} e^{(t-s)\mathcal{L}} \left( \left( \int_{0}^{s} e^{(r-s)\mathcal{L}} \left( \left( e^{-r\mathcal{L}}\overline{v} \right) V \right) dr \right) \left( e^{s\mathcal{L}}v^{2} \right) \right) ds, \end{split}$$

where we have used a blue dot for encoding the potential V. A full list of these trees is given in the Section 3.5.1.

#### 3.2.2 Approximated decorated trees

We denote by  $\mathcal{T}$  the set of decorated trees  $T_{\mathfrak{e},r}^{\mathfrak{n},\mathfrak{f}} = (T,\mathfrak{n},\mathfrak{f},\mathfrak{e},r)$  where

 $- T^{\mathfrak{n},\mathfrak{f}}_{\mathfrak{e}} \in \hat{\mathcal{T}}.$ 

— The decoration of the root is given by  $r \in \mathbf{Z}, r \geq -1$  such that

$$r+1 \ge \deg(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{f}}) \tag{3.36}$$

where deg is defined recursively by

$$\deg(\mathbf{1}) = 0, \quad \deg(\lambda_{\mathsf{I}}^{\ell} \prod_{i=1}^{m} \mathcal{I}_{o_i}(T_i)) = \ell + 1 + \max(\deg(T_1), ..., \deg(T_m)).$$

We call  $\mathcal{T}$  approximated decorated trees following the terminology introduced in [BS22]. Indeed, the decoration r at the root means that we consider an approximation at order r of the iterated integrals associated to the same trees without this decoration. The quantity deg $(T_{\mathfrak{e}}^{\mathfrak{n}})$  is the maximum number of edges and node decorations  $\mathfrak{n}$  lying on the same path from one leaf to the root.

We denote by  $\mathcal{P}$  the planted trees satisfying the same condition as  $\mathcal{T}$ . The forests formed of these trees are given by H and their linear span by  $\mathcal{H}$ . They are of the form  $\mathcal{I}_o^r(T)$ . The map  $\mathcal{I}_o^r(\cdot) : \hat{\mathcal{T}} \to \mathcal{H}$  is defined as the same as for  $\mathcal{I}_o(\cdot)$  except now that the root is decorated by r and it could be zero if the inequality (3.36) is not satisfied. It can be naturally extended into a map from  $\mathcal{T}$  into  $\mathcal{H}$  by removing the r decoration at the root of the tree that is grafted onto a new root. We define the map  $\mathcal{D}_r : \hat{\mathcal{P}} \to \mathcal{H}$  which adds the decoration r and performs the projection along the identity (3.36). It is given by

$$\mathcal{D}_r(\mathbf{1}) = \mathbf{1}_{\{0 < r+1\}}, \quad \mathcal{D}_r(\mathcal{I}_o(T)) = \mathcal{I}_o^r(T)$$
(3.37)

We extend this map to  $\mathcal{T}$  by:

$$\mathcal{D}_r(\lambda_{\mathfrak{l}}^{\ell}\prod_{i=1}^m \mathcal{I}_{o_i}(T_i)) = \lambda_{\mathfrak{l}}^{\ell}\prod_{i=1}^m \mathcal{I}_{o_i}^{r-\ell}(T_i)$$

where the term on the right hand-side is identified with  $\lambda_{\mathfrak{l}}^{\ell} \prod_{i=1}^{m} \mathcal{I}_{o_i}(T_i)$  where the decoration r is added to its root with the projection given by (3.36).

# 3.3 Approximations of iterated integrals

In this section, we introduce the iterated integrals associated to decorated trees via a character  $\Pi : \hat{\mathcal{H}} \to \mathcal{C}$ , a multiplicative map for the forest product. We approximate this map via a new character  $\Pi_A : \mathcal{H} \to \mathcal{C}$  defined on approximated decorated forests with A a domain to which  $\hat{\mathbf{v}}$  belongs to. The crucial part of the recursive definition of  $\Pi_A$  is given by an operator  $\mathcal{K}$  that filter out the dominant part of the operators. It expands only the lower order parts into a Taylor series expansion while integrating exactly the dominant part (see Definition 3.3.2). The definition of these operators is similar to the one in Fourier given in [BS22]. Then, in the second part of this section, we conduct the local error analysis by comparing  $\Pi$  with  $\Pi_A$  with the main result given in Theorem 3.3.9. The main argument for establishing this bound is a recursive definition (see Definition 3.3.8) which involves several Taylor remainders (see Lemma 3.3.6). These remainders are stemming from the various approximations performed by the map  $\mathcal{K}$ . Such a description is analogue to the Fourier case that can provide a further expansion of the local error analysis via a Birkhoff factorisation involving a Butcher-Connes-Kreimer coproduct. We do not have such a characterisation for our scheme. However in Proposition 3.3.7, we are able to show how dominant operators are involved in the local error analysis. This was previously observed in the Fourier case.

# 3.3.1 Characters on decorated forests

We first introduce the linear space  $\mathcal{C}$  in which the iterated integrals associated to Duhamel's formula (3.18) live. Any element  $f \in \mathcal{C}$  is such that for every  $\tau \in \mathbf{R}_+$ , one has that  $f(\cdot, \tau)$  is an operator on  $\hat{\mathbf{v}}$ . We define a character  $\Pi$  on  $\hat{\mathcal{H}}$ taking values in  $\mathcal{C}$ . It is given for  $T \in \hat{\mathcal{T}}$  and  $F_1, F_2 \in \hat{\mathcal{H}}$  by

$$(\Pi \mathcal{I}_o(T))(\hat{\mathbf{v}}, \tau) = \int_0^\tau e^{(\tau - \xi)\mathcal{L}_o}(\tilde{\Pi}_o T)(\hat{\mathbf{v}}, \xi)d\xi, \qquad (3.38)$$
$$(\Pi F_1 F_2)(\hat{\mathbf{v}}, \xi) = (\Pi F_1)(\hat{\mathbf{v}}, \xi) (\Pi F_2)(\hat{\mathbf{v}}, \xi)$$

where  $F_1F_2$  is the forest product and  $\tilde{\Pi}_o$  is given on a decorated tree  $T = \lambda_{\mathfrak{l}}^k \prod_{i=1}^n \mathcal{I}_{o_i}(T_i)$  by

$$\left(\tilde{\Pi}_{o}\lambda_{\mathfrak{l}}^{k}\prod_{i=1}^{n}\mathcal{I}_{o_{i}}(T_{i})\right)(\hat{\mathbf{v}},\xi) = \mathcal{B}_{o}^{\mathfrak{l}}\left(\frac{\Upsilon_{\mathrm{root}}^{\Psi_{o}}[T]}{S_{\mathrm{root}}(T)}V_{\mathfrak{l}}\xi^{k}\Pi\left(\prod_{i=1}^{n}\mathcal{I}_{o_{i}}(T_{i})\right)\right)(\hat{\mathbf{v}},\xi)$$
(3.39)

where  $S_{\text{root}}(T)$  is the symmetry factor associated to the root of the decorated tree T and the (linear) operators  $\mathcal{B}_o^{\mathfrak{l}}$  are given by the structure of the nonlinearity (3.2). The symmetry factor is defined by setting  $S_{\text{root}}(1) = 1$  and

$$S_{\text{root}}\left(\lambda_{\mathfrak{l}}^{k}\prod_{i,j}(\mathcal{I}_{o_{i}}(T_{i,j}))^{\beta_{i,j}}\right) = k!\prod_{i,j}\beta_{i,j}!$$
(3.40)

with  $T_{i,j} \neq T_{i,\ell}$  for  $j \neq \ell$ . The coefficient  $\Upsilon^{\Psi_o}_{\text{root}}[T]$  in (3.38) is given by

$$\Upsilon^{\Psi_o}_{\text{root}}[T](\mathbf{v},\xi) = (\partial_k D_{o_1} \cdots D_{o_n} \hat{\Psi}^{\mathbb{I}}_o)(\mathbf{v},\xi)$$
(3.41)

where for  $\Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}}) = \mathcal{B}_o^{\mathfrak{l}}\left(\prod_{\mathfrak{l}\in\mathfrak{L}_+^{\mathfrak{l},o}} f_{o,\mathfrak{l}}^{\mathfrak{l}}(u_{\mathfrak{l}})\right)$  we have

$$\hat{\Psi}_o^{\mathfrak{l}}(\mathbf{v},\xi) = \prod_{\boldsymbol{\lambda}\in\mathfrak{L}_+^{\mathfrak{l},o}} e^{\xi\mathcal{L}_{\boldsymbol{\lambda}}} f_{o,\boldsymbol{\lambda}}^{\mathfrak{l}}(v_{\boldsymbol{\lambda}}).$$

The various derivatives  $\partial$ ,  $D_o$  follow the Leibniz rule and are given by:

$$D_{o_i}\mathcal{C}^\ell \Big[ D_{o_i}^k f_{o,o_i}^{\mathfrak{l}}, \mathcal{L}_{o_i} \Big] = \mathcal{C}^\ell \Big[ D_{o_i}^{k+1} f_{o,o_i}^{\mathfrak{l}}, \mathcal{L}_{o_i} \Big],$$

$$\partial \mathcal{C}^{\ell} \Big[ D_{o_i}^k f_{o,o_i}^{\mathfrak{l}}, \mathcal{L}_{o_i} \Big] = \mathcal{C}^{\ell+1} \Big[ D_{o_i}^k f_{o,o_i}^{\mathfrak{l}}, \mathcal{L}_{o_i} \Big],$$

where  $D_{o_i}$  stands for the derivative with respect to  $v_{o_i}$ . One has  $D_{o_i}f(v_{o_j}) = 0$  for  $o_i \neq o_j$  and by convention we set  $D_{o_i}(e^{\xi \mathcal{L}_{o_i}}v_{o_i}) = 1$ . The derivative  $\partial_o$  adds a commutator depending on  $\mathcal{L}_{o_i}$  to  $f_{o_i,o_i}^{\mathfrak{l}}$ .

**Remark 3.3.1** The subscript o in  $\Pi_o$  is needed in the definition for identifying which nonlinearity has to be used in the coefficient  $\Upsilon_{\text{root}}^{\Psi_o}[T]$ . This is also a crucial difference with [BS22] where the coefficients  $\Upsilon_{\text{root}}^{\Psi_o}[T]$  are detached from the definition of the iterated integral. This is possible due to the fact that one only works in Fourier space in [BS22]. We have used the subscript root to stress that  $\Upsilon_{\text{root}}^{\Psi_o}[T]$  computes a coefficient of T taking into account only its edges connected to its root. This is in contrast with the classical elementary differential for B-series where all the nodes of the tree are considered for computing it. In fact, the rest of the coefficient is computed recursively via an iteration of  $\Pi$  in (3.39).

**Example 9** We continue Example 8 on the Gross-Pitaevskii equation and illustrate the definition of  $\Pi, \tilde{\Pi}_o$ . One has

$$\begin{pmatrix} \tilde{\Pi}_o \lambda_0 \end{pmatrix} (\hat{\mathbf{v}}, \xi) = \left( -ie^{\xi \mathcal{L}} v^2 \right) \left( e^{-\xi \mathcal{L}} \bar{v} \right), \quad \left( \tilde{\Pi}_{\bar{o}} \lambda_0 \right) (\hat{\mathbf{v}}, \xi) = \left( ie^{-\xi \mathcal{L}} \bar{v}^2 \right) \left( e^{\xi \mathcal{L}} v \right), \\ \left( \tilde{\Pi}_o \lambda_1 \right) (\hat{\mathbf{v}}, \xi) = \left( -ie^{\xi \mathcal{L}} v \right) V, \quad \left( \tilde{\Pi}_{\bar{o}} \lambda_1 \right) (\hat{\mathbf{v}}, \xi) = \left( ie^{-\xi \mathcal{L}} \bar{v} \right) V,$$

since

$$S_{\text{root}}(\lambda_0) = S_{\text{root}}(\lambda_1) = 1, \ V_0 = 1, \ V_1 = V, \ \mathcal{B}_o^o = \mathcal{B}_o^1 = \text{id},$$
  
$$\Upsilon^{\Psi_o}_{\text{root}}[\lambda_0](\mathbf{v},\xi) = \left(-ie^{\xi\mathcal{L}}v^2\right) \left(e^{-\xi\mathcal{L}}\bar{v}\right), \quad \Upsilon^{\Psi_{\bar{o}}}_{\text{root}}[\lambda_0](\mathbf{v},\xi) = \left(ie^{-\xi\mathcal{L}}\bar{v}^2\right) \left(e^{\xi\mathcal{L}}v\right),$$
  
$$\Upsilon^{\Psi_o}_{\text{root}}[\lambda_1](\mathbf{v},\xi) = -ie^{\xi\mathcal{L}}v, \quad \Upsilon^{\Psi_{\bar{o}}}_{\text{root}}[\lambda_1](\mathbf{v},\xi) = ie^{-\xi\mathcal{L}}\bar{v}.$$

Hence,

$$(\Pi \mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}},\tau) = \int_0^\tau e^{(\tau-\xi)\mathcal{L}_o}(\tilde{\Pi}_o\lambda_0)(\hat{\mathbf{v}},\xi)d\xi$$
$$= -i\int_0^\tau e^{(\tau-\xi)\mathcal{L}_o} \left(e^{\xi\mathcal{L}}v^2\right) \left(e^{-\xi\mathcal{L}}\bar{v}\right)d\xi.$$

Then again, using the definition of the characters  $\Pi$ , and  $\Pi_o$  in (3.38), we obtain

$$\begin{split} \left(\tilde{\Pi}_o \lambda_0^1\right) & (\hat{\mathbf{v}}, \xi) = \frac{\Upsilon_{root}^{\Psi_o}[\lambda_0^1]}{S_{root}(\lambda_0^1)} (\mathbf{v}, \xi) V_0 \xi \\ &= -i\xi \Big( e^{\xi \mathcal{L}} \mathcal{C}[u^2, \mathcal{L}](v) \Big) \Big( e^{-\xi \mathcal{L}} \bar{v} \Big), \end{split}$$

since,

$$\begin{split} \Upsilon^{\Psi_o}_{\text{root}}[\lambda_0^1](\mathbf{v},\xi) &= (\partial_1 \hat{\Psi}_o^0)(\mathbf{v},\xi) \\ &= \partial_1 [(e^{\xi \mathcal{L}} f_{o,o}^0(v_o))(e^{-\xi \mathcal{L}} f_{o,\bar{o}}^0(v_{\bar{o}}))] \\ &= -i \Big( e^{\xi \mathcal{L}} \mathcal{C}[u^2,\mathcal{L}](v) \Big) \Big( e^{-\xi \mathcal{L}} \bar{v} \Big) \end{split}$$

and  $S_{\text{root}}(\lambda_0^1) = 1$ .

Let A be a given domain of regularity of  $\hat{\mathbf{v}}$  that corresponds to the regularity we assume a priori on the initial data and the potentials. We introduce an approximation of  $\Pi$  via a new character  $\Pi_A$  defined on  $\mathcal{H}$ , by:

$$(\Pi_{A}\mathcal{I}_{o}^{r}(T))(\hat{\mathbf{v}},\xi) = \mathcal{K}_{o,A}^{r}((\Pi_{o,A}\mathcal{D}_{r-1}(T))(\hat{\mathbf{v}},\cdot))(\xi)$$

$$\left(\tilde{\Pi}_{o,A}\lambda_{\mathfrak{l}}^{k}\prod_{i=1}^{n}\mathcal{I}_{o_{i}}^{r}(T_{i})\right)(\hat{\mathbf{v}},\xi) = \mathcal{B}_{o}^{\mathfrak{l}}\left(\frac{\Upsilon_{\text{rot}}^{\Psi_{o}}[T]}{S_{\text{rot}}(T)}V_{\mathfrak{l}}\xi^{k}\prod_{i=1}^{n}\left(\Pi_{A}\mathcal{I}_{o_{i}}^{r}(T_{i})\right)\right)(\hat{\mathbf{v}},\xi)$$

$$(3.42)$$

where  $T = \lambda_{\Gamma}^k \prod_{i=1}^n \mathcal{I}_{o_i}(T_i)$  and the operator  $\mathcal{K}_{o,A}^r$  is given in Definition 3.3.2. As one can notice the approximation follows the recursive definition of  $\Pi$  and  $\tilde{\Pi}_o$ . The main difference occurs when one replaces the integral in time given in (3.38) by

59

#### 3.3. Approximations of iterated integrals

the operator  $\mathcal{K}_{r,A}^{r}$ . This operator is used to give an approximation of order r+1 to the following integral:

$$\int_{0}^{t} e^{(t-s)\mathcal{L}_{o}} \mathcal{B}\left(s^{\ell} \prod_{i=1}^{m} e^{s\mathcal{L}_{i}} u_{i}\right) ds$$
(3.43)

where  $\mathcal{B}$ ,  $\mathcal{L}_i$  are some operators and the  $u_i$  depend on  $\hat{\mathbf{v}}$ . The approximation should take into account the various interactions between the operators  $\mathcal{L}_o$  and  $\mathcal{L}_i$ . In the sequel, we will use the following shorthand notations:

$$\Pi_A^r = \Pi_A \mathcal{D}_r, \quad \tilde{\Pi}_{o,A}^r = \tilde{\Pi}_{o,A} \mathcal{D}_r$$

where now  $\Pi^r_A$  and  $\tilde{\Pi}_{o,A}$  are defined on  $\hat{\mathcal{H}}$  and  $\hat{\mathcal{P}}$ .

**Definition 3.3.2** Let us consider a function F of the form

$$F(\hat{\mathbf{v}},\xi) = \xi^{\ell} \mathcal{B}\left(\prod_{i \in J} e^{\xi \mathcal{L}_i} u_i(\hat{\mathbf{v}})\right)$$

where  $\mathcal{B}$  is a linear operator, the operators  $\mathcal{L}_i$  satisfy Assumption (2), and J is a finite set. We suppose that the  $u_i$  are smooth functions of  $\hat{\mathbf{v}}$ , and we assume a given domain A for which we have  $\hat{\mathbf{v}} \in A$ . Let r > 0,  $o \in \mathfrak{L}_+$  and the operator  $\mathcal{L}_o$  associated to the decoration o. We assume that  $\mathcal{B}$  commutes with  $\mathcal{L}_o$ . Given  $(k_i)_{i \in J}$  with  $k_i \in \mathbf{N}$ , we define

$$G_{n,(k_i)_{i\in J}}: \hat{\mathbf{v}} \to \mathcal{L}_o^n \mathcal{B}\left(\prod_{i\in J} \mathcal{L}_i^{k_i} u_i(\hat{\mathbf{v}})\right)$$

Note that the domain of G depends on the space X and it's associated norm. Now we distinguish two cases:

— If  $A \subseteq D(G_{n,(k_i)_{i \in J}})$  for all  $(k_i)_{i \in J}$  and n be such that,  $\sum_i k_i + n \leq r - \ell + 1$  we can carry out a Taylor-series expansion of all the operators and set:

$$\mathcal{K}_{o,A}^{r}(F(\hat{\mathbf{v}},\cdot))(\tau)$$

$$= \sum_{\sum_{i}k_{i}+n \leq r-\ell} \int_{0}^{\tau} \frac{(\tau-\xi)^{n} \xi^{\sum_{i \in J} k_{i}+\ell}}{n! \prod_{i \in J} k_{i}!} \mathcal{L}_{o}^{n} \mathcal{B}\left(\prod_{i \in J} \mathcal{L}_{i}^{k_{i}} u_{i}(\hat{\mathbf{v}})\right) d\xi.$$
(3.44)

— Otherwise, we set

$$\mathcal{A}_i = \mathcal{L}_i - \mathcal{L}_o, \quad \mathcal{L}_{dom} = \mathcal{P}_{dom}(\{\mathcal{A}_1, \cdots, \mathcal{A}_m\})$$

Then, if  $\mathcal{L}_{\text{dom}} \neq 0$ , let  $I \subset J$  such that for every  $i \in I$ , one has

$$\mathcal{A}_i = \mathcal{L}_{\mathrm{dom}} + \mathcal{A}^i_{\mathrm{low}}, \quad D(\mathcal{L}_{\mathrm{dom}}) \subset D(\mathcal{A}^i_{\mathrm{low}}),$$

and for  $i \in J \setminus I$ 

$$\mathcal{A}_i = \mathcal{A}_{\text{low}}^i, \quad D(\mathcal{L}_{\text{dom}}) \subset D(\mathcal{A}_i).$$

One can rewrite F as

$$F(\cdot,\xi) = \xi^{\ell} \mathcal{B}\left(\left(\prod_{i\in I} e^{\xi(\mathcal{L}_o + \mathcal{L}_{\rm dom} + \mathcal{A}^i_{\rm low})} u_i\right) \left(\prod_{i\in J\setminus I} e^{\xi(\mathcal{L}_o + \mathcal{A}^i_{\rm low})} u_i\right)\right).$$
(3.45)

We assume that the operators that appear in (3.45) commute and generate a continuous semigroup. If this is not the case, one needs to apply the approximation (3.44). We have to distinguish two cases:

(i) If  $\mathcal{L}_{dom} \neq 0$  then

$$\mathcal{K}_{o,A}^{r}(F)(\tau) = \int_{0}^{\tau} \sum_{q \leq r-\ell} \sum_{q=n+m+p+\sum_{i \in J} k_{i}} \frac{(\tau-\xi)^{m}\xi^{\ell+\sum_{i \in J} k_{i}}}{p!m!n!\prod_{i \in J} k_{i}!} \qquad (3.46)$$

$$\mathcal{B}\Big(\mathcal{C}^{m}[\Big(e^{\tau\mathcal{L}_{o}}\mathcal{C}^{n}[\mathcal{M}_{J\setminus I},\tau\mathcal{L}_{o}]\Big)\Big(e^{\xi\mathcal{L}_{dom}+\tau\mathcal{L}_{o}}\mathcal{C}^{p}[\mathcal{M}_{I},\xi\mathcal{L}_{dom}+\tau\mathcal{L}_{o}]\Big),\mathcal{L}_{o}]$$

$$(((\mathcal{A}_{low}^{i})^{k_{i}}u_{i})_{i \in J})\Big)d\xi.$$

(ii) If  $\mathcal{L}_{dom} = 0$  then

$$\mathcal{K}_{o,A}^{r}(F)(\tau) = \int_{0}^{\tau} \sum_{q \leq r-\ell} \sum_{q=n+m+\sum_{i \in J} k_{i}} \frac{(\tau-\xi)^{m} \xi^{\sum_{i \in J} k_{i}}}{m! n! \prod_{i \in J} k_{i}!} \qquad (3.47)$$

$$\mathcal{B}\Big(\mathcal{C}^{m}[\Big(e^{\tau \mathcal{L}_{o}} \mathcal{C}^{n}[\mathcal{M}_{J}, \tau \mathcal{L}_{o}]\Big), \mathcal{L}_{o}](((\mathcal{A}_{i})^{k_{i}} u_{i})_{i \in J})\Big) d\xi.$$

Remark 3.3.3 In general, one has to face more complicated products to approximate. They are of the form:

$$F(\hat{\mathbf{v}},\xi) = \xi^{\ell} \mathcal{B}\left(\prod_{i \in K} \mathcal{B}_i\left(\prod_{j \in J_i} e^{\xi \mathcal{L}_j} u_{i,j}(\hat{\mathbf{v}})\right)\right)$$

Such a product will not appear in the Gross-Pitaevski equation neither for the first order of Sine-Gordon. In that case, one can perform the same analysis computing dominant and lower parts. To be in the cases (i) and (ii), there must exist some  $I \subset K$  such that one has to get the following factorisation:

$$F(\hat{\mathbf{v}},\xi) = \xi^{\ell} \mathcal{B}\left(\prod_{i \in I} \mathcal{B}_{i}\left(\prod_{j \in J_{i}} e^{\xi(\mathcal{L}_{o} + \mathcal{L}_{\mathrm{dom}} + \mathcal{A}_{\mathrm{low}}^{j})} u_{i,j}\right)\right)$$
$$\left(\prod_{i \in K \setminus I} \mathcal{B}_{i}\left(\prod_{j \in J_{i}} e^{\xi(\mathcal{L}_{o} + \mathcal{A}_{\mathrm{low}}^{j})} u_{i,j}\right)\right)$$

Then, one has just to replace the products  $\mathcal{M}$  by new ones taking into account the operators  $\mathcal{B}_i$ :

$$\prod_{i\in I} \mathcal{B}_i(\mathcal{M}_{J_i}), \quad \prod_{i\in K\setminus I} \mathcal{B}_i(\mathcal{M}_{J_i})$$

instead of  $\mathcal{M}_I$  and  $\mathcal{M}_{K\setminus I}$ . If F happens not to have this form one performs the full Taylor expansion as in (3.44).

**Remark 3.3.4** The  $\{k_i\}_{i \in J}$  appear after having Taylor expanded the lower order parts of the operators. Let us mention that this step is necessary in a coupled system with two differential operators  $\mathcal{L}_1$  and  $\mathcal{L}_2$  that differ from a lower differential operator see equation (3.17).

**Remark 3.3.5** The previous approximation is optimized according to an a priori given domain A. For solutions and potentials regular enough, for example when the following inclusions hold:  $A \subset D(\mathcal{L}_l)$  or/and  $A \subset D(\mathcal{L}_{dom})$ , then in these cases, one can perform Taylor expansions (see (3.44)) in order to simplify the scheme. This approach is deeply used in Section 3.5.1 for the construction of the second order low-regularity scheme for Gross-Pitaevskii, and in Sections 3.5.1 and 3.5.3 to construct a simplified first-order scheme for the Gross-Pitaevskii and Sine-Gordon equation when we assume more regularity on the solution and/or potential. Moreover, given that the boundary conditions are encoded in the domain of the operator  $\mathcal{L}_o$  and  $\mathcal{L}_{dom}$ , we also define the domain A with the prescribed boundary conditions.

**Example 10** We illustrate Definition 3.3.2 on a tree coming from the NLS equation with  $A = (H^2(\Omega))^2$ ,  $\hat{\mathbf{v}} = (v, \bar{v})$ , where for simplicity we assume that  $\Omega = \mathbf{R}^d$ .

$$(\Pi_A \mathcal{I}_o^r(\lambda_0))(\hat{\mathbf{v}}, \xi) = \mathcal{K}_{o,A}^r((\Pi_{o,A} \mathcal{D}_{r-1}(\lambda_0))(\hat{\mathbf{v}}, \cdot))(\xi)$$
$$(\Pi_{o,A} \mathcal{D}_{r-1}(\lambda_0))(\hat{\mathbf{v}}, \xi) = (-ie^{\xi \mathcal{L}}v^2)(e^{-\xi \mathcal{L}}\bar{v}).$$

Then using the notations in Definition 3.3.2, one gets

$$F(\hat{\mathbf{v}}, \xi) = \prod_{i \in \{1,2\}} e^{\xi \mathcal{L}_i} u_i(\hat{\mathbf{v}}), \quad u_1(\hat{\mathbf{v}}) = -iv^2, \quad u_2(\hat{\mathbf{v}}) = \bar{v},$$
$$\mathcal{L}_1 = \mathcal{L}, \quad \mathcal{L}_2 = -\mathcal{L}.$$

We can now compute the operator interactions as follows

$$\mathcal{A}_1 = \mathcal{L}_1 - \mathcal{L} = 0, \quad \mathcal{A}_2 = \mathcal{L}_2 - \mathcal{L} = -2\mathcal{L}$$
$$\mathcal{L}_{\text{dom}} = \mathcal{P}_{\text{dom}}(\{\mathcal{A}_1, \mathcal{A}_2\}) = \mathcal{P}_{\text{dom}}(\{0, -2\mathcal{L}\}) = -2\mathcal{L}, \quad \mathcal{A}_{\text{low}}^1 = \mathcal{A}_{\text{low}}^2 = 0.$$

We have  $\mathcal{L}_{dom} = -2\mathcal{L} \neq 0$ , therefore we apply the first point (i) of Definition 3.3.2:

$$\mathcal{K}_{o,A}^{r}(F)(\tau) = \int_{0}^{\tau} \sum_{q \leq r-\ell} \sum_{q=n+m+p} \frac{(\tau-\xi)^{m}}{p!m!n! \prod_{i \in J} k_{i}!}$$
$$\mathcal{C}^{m}[\left(e^{\tau \mathcal{L}_{o}} \mathcal{C}^{n}[\mathcal{M}_{J\setminus I}, \tau \mathcal{L}_{o}]\right) \left(e^{\xi \mathcal{L}_{dom} + \tau \mathcal{L}_{o}} \mathcal{C}^{p}[\mathcal{M}_{I}, \xi \mathcal{L}_{dom} + \tau \mathcal{L}_{o}]\right), \mathcal{L}_{o}]$$
$$((u_{i})_{i \in J}) d\xi.$$

We have  $I = \{2\}$  and  $J = \{1, 2\}$ , therefore for every  $n, p \neq 0$ 

$$\mathcal{C}^{n}[\mathcal{M}_{J\setminus I}, \tau \mathcal{L}_{o}] = \mathcal{C}^{p}[\mathcal{M}_{I}, \xi \mathcal{L}_{dom} + \tau \mathcal{L}_{o}] = 0.$$

Hence, from the above computations and by considering the case where r = 1 (i.e., a second order scheme) we obtain the following,

$$\begin{aligned} \mathcal{K}^{1}_{o,A}(F)(\tau) &= B + C \\ B &= \int_{0}^{\tau} (\tau - \xi) \mathcal{C}[\left(e^{\tau \mathcal{L}_{o}} \mathcal{M}_{\{1\}}\right) \left(e^{\xi \mathcal{L}_{\mathrm{dom}} + \tau \mathcal{L}_{o}} \mathcal{M}_{\{2\}}\right), \mathcal{L}_{o}](u_{1}, u_{2}) \, d\xi \\ C &= \int_{0}^{\tau} \left(e^{\tau \mathcal{L}_{o}} u_{1}\right) \left(e^{\xi \mathcal{L}_{\mathrm{dom}} + \tau \mathcal{L}_{o}} u_{2}\right) d\xi. \end{aligned}$$

In the end, one has

$$(\Pi_{A}\mathcal{I}_{o}^{1}(\lambda_{0}))(\hat{\mathbf{v}},\tau) = -i\int_{0}^{\tau} \left(e^{\tau\mathcal{L}}v^{2}\right) \left(e^{(\tau-2\xi)\mathcal{L}}\overline{v}\right) d\xi + \int_{0}^{\tau} (\tau-\xi)\mathcal{C}[\left(e^{\tau\mathcal{L}}\mathcal{M}_{\{1\}}\right) \left(e^{(\tau-2\xi)\mathcal{L}}\mathcal{M}_{\{2\}}\right), \mathcal{L}](-iv^{2},\overline{v}) d\xi$$

## 3.3.2 Local error analysis for approximated iterated integrals

Before comparing the character  $\Pi$  with  $\Pi_A$ , one has to understand the error introduced by the approximation operator  $\mathcal{K}$ . This depends on the various cases in Definition 3.3.2.

**Lemma 3.3.6** We keep the notations of Definition 3.3.2. We suppose that  $r \ge \ell$  then one has

$$\int_{0}^{\tau} e^{(\tau-\xi)\mathcal{L}_{o}}F(\hat{\mathbf{v}},\xi)d\xi - \mathcal{K}_{o,A}^{r}(F(\hat{\mathbf{v}},\cdot))(\tau) = \mathcal{O}(\tau^{r+2}R_{o,A}^{r}(F)(\hat{\mathbf{v}},\cdot)(\tau))$$
(3.48)

where  $R_{o,A}^r(F)$  takes the following values:

(i) If we are using the approximation (3.44), we get the following error:

$$R_{o,A}^{r}(F) = \sum_{\sum_{i} k_{i}+n=r-\ell+1} \mathcal{L}_{o}^{n} \mathcal{B}\left(\prod_{i \in J} \mathcal{L}_{i}^{k_{i}} u_{i}\right).$$

(ii) If  $\mathcal{L}_{dom} \neq 0$  then

$$R_{o,A}^{r}(F) = \sum_{\sum_{i \in J} k_{i}+m+n+p=r-\ell+1} \mathcal{B}\big(\mathcal{C}^{m}[\big(\mathcal{C}^{n}[\mathcal{M}_{J\setminus I}, \mathcal{L}_{o}]\big) \\ (\mathcal{C}^{p}[\mathcal{M}_{I}, \mathcal{L}_{dom} + \mathcal{L}_{o}]), \mathcal{L}_{o}](((\mathcal{A}_{low}^{i})^{k_{i}}u_{i})_{i \in J})\big).$$

(iii) If  $\mathcal{L}_{dom} = 0$  then

$$R_{o,A}^{r}(F) = \sum_{\sum_{i \in J} k_i + m + n + p = r - \ell + 1} \mathcal{B}\Big(\mathcal{C}^{m}[\mathcal{C}^{n}[\mathcal{M}_J, \mathcal{L}_o], \mathcal{L}_o](((\mathcal{A}_i)^{k_i}u_i)_{i \in J})\Big).$$

Proof. The proof reduces to applying several times Taylor expansions and using identity (3.12).

**Example 11** We want to compute the second order error term for the tree  $\mathcal{I}_o(\lambda_0)$  illustrating point (*ii*) of Lemma 3.3.6. From Example 10, we are in the case where  $D(\mathcal{L}_{dom}) = D(\mathcal{L}_o)$  and we have

$$R_{o,A}^{1}(\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}}) = \sum_{m+n+p=2} \mathcal{C}^{m}[\left(\mathcal{C}^{n}[\mathcal{M}_{J\setminus I},\mathcal{L}_{o}]\right)$$
$$\left(\mathcal{C}^{p}[\mathcal{M}_{I},\mathcal{L}_{\mathrm{dom}}+\mathcal{L}_{o}]\right),\mathcal{L}_{o}](-iv^{2},\bar{v})$$

where  $I = \{2\}$  and  $J = \{1, 2\}$  and where the only non-zero term in the above is when (m, n, p) = (2, 0, 0). Hence, we have,

$$R^1_{o,A}(\tilde{\Pi}_{o,A}\lambda_0)(\hat{\mathbf{v}}) = \mathcal{C}^2[\mathcal{M}_{\{1,2\}}, \mathcal{L}_o](-iv^2, \bar{v}).$$

The next proposition follows the steps of [BS22, Prop. 3.9]. It singles out the dominant parts of the oscillations and shows that approximated iterated integrals are connected with Definition 3.2.3. This decomposition was also connected to the Birkoff factorisaton given in [BS22]. It provides information about the dominant operators that will be involved in the local error analysis. In the next proposition, we assume that  $\mathcal{B}_o^{\mathfrak{l}} = \mathfrak{id}$ . In the general case, one can write a similar statement using Remark 3.3.3.

**Proposition 3.3.7** For every tree  $\mathcal{I}_o^r(T) \in \mathcal{H}$ , one has the following decomposition:

$$(\Pi_{A}\mathcal{I}_{o}^{r}(T))(\hat{\mathbf{v}},\xi) = \sum_{\bar{T}\subset\mathcal{I}_{o}(T)}\prod_{\mathcal{L}\in\mathcal{R}_{dom}(\bar{T})} \left(e^{i\xi\mathcal{L}}B^{r}(T,\bar{T},\mathcal{L})(\hat{\mathbf{v}},\xi)\right) + \left(e^{i\xi\mathcal{L}_{o}}B^{r}(T,\mathbf{1},\mathcal{L}_{o})(\hat{\mathbf{v}},\xi)\right) \left(e^{i\xi\mathcal{L}_{o}}C^{r}(T,\mathbf{1},\mathcal{L}_{o})(\hat{\mathbf{v}},\xi)\right)$$
(3.49)

where the sum is over all subtrees  $\overline{T} \neq \mathbf{1}$  that have at least one edge and that contain the root of  $\mathcal{I}_o(T)$ . The  $B^r(T, \overline{T}, \mathcal{L})(\hat{\mathbf{v}}, \xi)$ ,  $B^r(T, \mathbf{1}, \mathcal{L}_o)(\hat{\mathbf{v}}, \xi)$  and  $C^r(T, \mathbf{1}, \mathcal{L}_o)(\hat{\mathbf{v}}, \xi)$  are polynomials in  $\xi$  and the  $\mathcal{R}_{dom}(\overline{T})$  are given in Definition 3.2.3.

*Proof.* We proceed by induction on the size of T and we suppose that  $T = \lambda_{\mathfrak{l}}^{\ell} \prod_{i} \mathcal{I}_{o_{i}}(T_{i})$  where one can apply the induction hypothesis on each  $\mathcal{I}_{o_{i}}(T_{i})$ . One has from (3.42)

$$(\Pi_A \mathcal{I}_o^r(T))(\hat{\mathbf{v}}, \xi) = \mathcal{K}_{o,A}^r \Big( \tilde{\Pi}_{o,A} \big( \mathcal{D}_{(r-1)}(T) \big)(\hat{\mathbf{v}}, \cdot) \Big)(\xi)$$

and

$$\tilde{\Pi}_{o,A}\big(\mathcal{D}_{(r-1)}(T)\big)(\hat{\mathbf{v}},\xi) = \frac{\Upsilon^{\Psi_o}_{\text{root}}[T]}{S_{\text{root}}(T)}(\mathbf{v},\xi)V_{\mathfrak{l}}\xi^{\ell}\prod_i \Big(\Pi_A \mathcal{I}_{o_i}^{(r-1-\ell)}(T_i)\Big)(\hat{\mathbf{v}},\xi).$$

By applying the induction hypothesis on the  $\mathcal{I}_{o_i}(T_i)$ , one has the following decomposition

$$\begin{split} \tilde{\Pi}_{o,A} \big( \mathcal{D}_{(r-1)}(T) \big) (\hat{\mathbf{v}}, \xi) &= \frac{\Upsilon^{\Psi_o}_{\text{root}}[T]}{S_{\text{root}}(T)} (\mathbf{v}, \xi) V_{\mathfrak{l}} \prod_i \sum_{\bar{T}_i \subset \mathcal{I}_{o_i}(T_i)} \\ &\prod_{\mathcal{L} \in \mathcal{R}_{\text{dom}}(\bar{T}_i)} \Big( e^{i\xi\mathcal{L}} B^{r-1-\ell} \big( T_i, \bar{T}_i, \mathcal{L} \big) (\hat{\mathbf{v}}, \xi) \Big) \\ &+ \Big( e^{i\xi\mathcal{L}_{o_i}} B^r(T_i, \mathbf{1}, \mathcal{L}_{o_i}) (\hat{\mathbf{v}}, \xi) \Big) \Big( e^{i\xi\mathcal{L}_{o_i}} C^r(T_i, \mathbf{1}, \mathcal{L}_{o_i}) (\hat{\mathbf{v}}, \xi) \Big) \end{split}$$

where the  $B^r(T_i, \bar{T}_i, \mathcal{L})(\hat{\mathbf{v}}, \xi)$ ,  $B^r(T_i, \mathbf{1}, \mathcal{L}_{o_i})(\hat{\mathbf{v}}, \xi)$  and  $C^r(T_i, \mathbf{1}, \mathcal{L}_{o_i})(\hat{\mathbf{v}}, \xi)$  are polynomials in  $\xi$ . As a consequence of Definition 3.3.2, the map  $\mathcal{K}_{o,A}^r$  applied to the term coming from a subforest  $\bar{T} = \lambda_{\mathfrak{l}}^{\ell} \prod_i \bar{T}_i$ , integrates exactly in the end a term of the form

$$\int_0^{\xi} \left( e^{\xi \mathcal{L}_o} P(\hat{\mathbf{v}}, s) \right) \left( e^{s \mathcal{L}_{\text{dom}}(\mathcal{I}_o(\bar{T})) + \xi \mathcal{L}_o} Q(\hat{\mathbf{v}}, s) \right) ds$$

where  $P(\hat{\mathbf{v}}, s)$  and  $Q(\hat{\mathbf{v}}, s)$  are polynomials in s. Then, by computing this integral, we obtain two terms of the form

$$\left(e^{s\mathcal{L}_o}P_1(\hat{\mathbf{v}},s)\right)\left(e^{is\mathcal{L}_{\rm dom}(\mathcal{I}_o(\bar{T}))}Q_1(\hat{\mathbf{v}},s)\right), \quad \left(e^{s\mathcal{L}_o}P_2(\hat{\mathbf{v}},s)\right)\left(e^{s\mathcal{L}_o}Q_2(\hat{\mathbf{v}},s)\right)$$

where  $P_1, Q_1$  and  $P_2$  are also polynomials in s. The first term corresponds to the subtree  $\mathcal{I}_o(\bar{T})$  and the second term corresponds to 1. Therefore, the structure of (3.49) is preserved.

**Definition 3.3.8** Let  $r \in \mathbf{Z}$  and a domain A of  $\hat{\mathbf{v}}$ . We recursively define  $\mathcal{L}^{r}_{low}(\cdot, \hat{\mathbf{v}}, A)$  and  $\mathcal{L}^{r,o}_{low}(\cdot, \hat{\mathbf{v}}, A)$ ,  $o \in \mathfrak{L}_{+}$  as

$$\mathcal{L}_{low}^{r}(T, \hat{\mathbf{v}}, A) = 1, \quad r < 0, \quad \mathcal{L}_{low}^{r}(\mathbf{1}, \hat{\mathbf{v}}, A) = 1.$$

Else for  $T = \lambda_{\mathfrak{l}}^{\ell} \prod_{i} \mathcal{I}_{a_i}(T_i)$ , one has

$$\mathcal{L}_{low}^{r,o}(\lambda_{l}^{\ell}\prod_{i}\mathcal{I}_{a_{i}}(T_{i}),\hat{\mathbf{v}},A) =$$

$$\mathcal{B}_{o}^{\mathfrak{l}}\left(\frac{\Upsilon_{root}^{\Psi_{o}}[T]}{S_{root}(T)}(\mathbf{v},0)V_{\mathfrak{l}}\left(\sum_{i}\mathcal{L}_{low}^{r-\ell}(\mathcal{I}_{a_{i}}(T_{i}),\hat{\mathbf{v}},A)\right)\right).$$
(3.50)

And

$$\mathcal{L}^{r}_{\text{low}}(\mathcal{I}_{o}(T), \hat{\mathbf{v}}, A) = \mathcal{L}^{r-1,o}_{\text{low}}(T, \hat{\mathbf{v}}, A) + R^{r}_{o,A}\Big(\Big(\tilde{\Pi}^{r-1}_{o,A}T\Big)(\hat{\mathbf{v}}, \cdot\Big)\Big),$$
(3.51)

where  $R_{o,A}^r$  is defined in Lemma 3.3.6

**Example 12** We continue Example 11. For the decorated tree  $T = \mathcal{I}_o(\lambda_0)$ , one obtains:

$$\mathcal{L}^{1}_{\text{low}}(T, \hat{\mathbf{v}}, A) = \mathcal{L}^{0}_{\text{low}}(\lambda_{0}, \hat{\mathbf{v}}, A) + R^{1}_{o,A} \left( \left( \tilde{\Pi}^{0}_{o,A} T \right) (\hat{\mathbf{v}}, \cdot) \right)$$
$$= -iv^{2} \bar{v} + \mathcal{C}^{2} [\mathcal{M}_{\{1,2\}}, \mathcal{L}_{o}] (-iv^{2}, \bar{v}).$$

One notices that the first term asks less regularity on v in comparison to the second.

The next theorem shows that the characters  $\Pi_A$  and  $\Pi_{o,A}$  are good approximations of  $\Pi$  and  $\Pi_o$  with a local error given by Definition 3.3.8. We follow the same steps as in the proof of [BS22, Thm 3.17].

**Theorem 3.3.9** For  $T = \lambda_{\mathfrak{l}}^{\ell} \prod_{i} \mathcal{I}_{a_{i}}(T_{i})$  one has,

$$\left(\tilde{\Pi}_{o}T - \tilde{\Pi}_{o,A}\mathcal{D}_{r}(T)\right)(\hat{\mathbf{v}},\tau) = \mathcal{O}\left(\tau^{r+2}\mathcal{L}_{low}^{r,o}(T,\hat{\mathbf{v}},A)\right)$$
(3.52)

and for  $T = \mathcal{I}_o(\bar{T})$ , one gets

$$(\Pi T - \Pi_A \mathcal{D}_r(T))(\hat{\mathbf{v}}, \tau) = \mathcal{O}\big(\tau^{r+2} \mathcal{L}^r_{low}(T, \hat{\mathbf{v}}, A)\big).$$
(3.53)

*Proof.* We proceed by induction by using the recursive definition (3.42) of  $\Pi_A$  and  $\tilde{\Pi}_{o,A}$ . First, one gets

$$(\Pi - \Pi_A)(\mathbf{1})(\hat{\mathbf{v}}, \tau) = 0 = \mathcal{O}\big(\tau^{r+2}\mathcal{L}^r_{\text{low}}(\mathbf{1}, \hat{\mathbf{v}}, A)\big).$$

Then for  $T = \lambda_{\mathfrak{l}}^{\ell} \prod_{i} \mathcal{I}_{a_{i}}(T_{i})$  and every  $o \in \mathfrak{L}_{+}$ , one has

$$\begin{split} \left(\tilde{\Pi}_{o} - \tilde{\Pi}_{o,A}^{r}\right)(T)(\hat{\mathbf{v}},\tau) &= \tau^{\ell} \mathcal{B}_{o}^{\mathfrak{l}} \left(\frac{\Upsilon_{\text{root}}^{\Psi_{o}}[T]}{S_{\text{root}}(T)}(\mathbf{v},\tau) V_{\mathfrak{l}} \sum_{i} \left(\Pi - \Pi_{A}^{r-\ell}\right) (\mathcal{I}_{a_{i}}(T_{i}))(\hat{\mathbf{v}},\tau) \\ &\prod_{j \neq i} \left(\Pi_{A}^{r-\ell} + \Pi\right) (\mathcal{I}_{a_{j}}(T_{j}))(\hat{\mathbf{v}},\tau) \right). \end{split}$$

Then by applying the induction hypothesis (3.53) to each  $\mathcal{I}_{a_i}(T_i)$ , one gets

$$\left( \tilde{\Pi}_{o} - \tilde{\Pi}_{o,A}^{r} \right)(T)(\hat{\mathbf{v}},\tau) = \mathcal{O}\left( \tau^{r+2} \mathcal{B}_{o}^{\mathsf{f}} \left( \frac{\Upsilon_{\text{root}}^{\Psi_{o}}[T]}{S_{\text{root}}(T)}(\mathbf{v},0) V_{\mathfrak{l}} \sum_{i} \mathcal{L}_{\text{low}}^{r-\ell}(\mathcal{I}_{a_{i}}(T_{i}),\hat{\mathbf{v}},A) \right) \right)$$
$$= \mathcal{O}\left( \tau^{r+2} \mathcal{L}_{\text{low}}^{r} (\lambda_{\mathfrak{l}}^{\ell} \prod_{i} \mathcal{I}_{a_{i}}(T_{i}),\hat{\mathbf{v}},A) \right).$$

For  $T = \mathcal{I}_o(\bar{T})$ , one has

$$\begin{aligned} (\Pi - \Pi_A^r)(T)(\hat{\mathbf{v}}, \tau) &= \int_0^\tau e^{\xi \mathcal{L}_o} (\tilde{\Pi}_o - \tilde{\Pi}_{o,A}^{r-1})(\bar{T})(\hat{\mathbf{v}}, \xi) d\xi \\ &+ \int_0^\tau e^{\xi \mathcal{L}_o} (\tilde{\Pi}_{o,A}^{r-1}\bar{T})(\hat{\mathbf{v}}, \xi) d\xi - \mathcal{K}_o^r((\tilde{\Pi}_{o,A}^{r-1}T)(\hat{\mathbf{v}}, \cdot))(\tau) \\ &= \int_0^\tau \mathcal{O}\big(\xi^{r+1} \mathcal{L}_{\text{low}}^{r-1,o}(\bar{T}, \hat{\mathbf{v}}, A)\big) d\xi + \mathcal{O}(\tau^{r+2} R_{o,A}^r \Big(\Big(\tilde{\Pi}_{o,A}^{r-1}\bar{T}\Big)(\hat{\mathbf{v}}, \cdot)\Big)) \\ &= \mathcal{O}\big(\tau^{r+2} \mathcal{L}_{\text{low}}^r(T, \hat{\mathbf{v}}, A)\big). \end{aligned}$$

where the term  $R^r_{o,A}\left(\left(\tilde{\Pi}^{r-1}_{o,A}\bar{T}\right)(\hat{\mathbf{v}},\cdot)\right)$  is obtained by applying Lemma 3.3.6.

# 3.4 Low regularity numerical scheme

The writing of our low regularity numerical scheme follows two steps. The first one is to write a truncated decorated trees series that will solve the Duhamel's formula up to order r (see Proposition 3.4.1). This series is formed of iterated integrals produced by the character  $\Pi$ . Then, one replaces  $\Pi$  by  $\Pi_A$  to get the scheme. Such steps were already used in [BS22]. Let us mention that the first step is more involved here due to the fact that we work with a more general setting.

#### **3.4.1** Exact solution up to order r

Recall the mild solution of (3.1) given by Duhamel's formula

$$u_o(t) = e^{t\mathcal{L}_o} v_o + \sum_{\mathfrak{l}\in\mathfrak{L}_-} \int_0^t e^{(t-\xi)\mathcal{L}_o} \Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}}) V_{\mathfrak{l}} d\xi, \qquad (3.54)$$

where the nonlinearity  $\Psi_{o}^{\mathfrak{l}}(\mathbf{u}_{o}^{\mathfrak{l}})$  is given by

$$\Psi_o^{\mathfrak{l}}(\mathbf{u}_o^{\mathfrak{l}}) = \mathcal{B}_o^{\mathfrak{l}}\left(\prod_{\mathfrak{l}\in\mathfrak{L}_+^{\mathfrak{l},o}} f_{o,\mathfrak{l}}^{\mathfrak{l}}(u_{\mathfrak{l}})\right).$$

In the following we want to construct a scheme with a local error of order

$$\mathcal{O}(\tau^{r+2}).$$

Therefore, before describing our numerical scheme, we need to remove the trees which are already of size  $\mathcal{O}(\tau^{r+2})$ . Indeed, a simple recursion shows that one has for every tree  $\mathcal{I}_o(T)$ 

$$(\Pi \mathcal{I}_o(T))(\hat{\mathbf{v}}, \tau) = \mathcal{O}(p(T, \hat{\mathbf{v}}) \tau^{n_+(T)})$$
(3.55)

where  $p(T, \hat{\mathbf{v}})$  is a polynomial in the variables  $\hat{\mathbf{v}}$  that does not contain any operators  $(\mathcal{L}_o)_{o \in \mathfrak{L}_+}$  and at most  $|N_T|$  operators  $\mathcal{B}$ . The map  $n_+$  is defined on  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{f}}$  as

$$n_+(T^{\mathfrak{n},\mathfrak{f}}_{\mathfrak{e}}) = \sum_{v \in N_T} \mathfrak{n}(v) + |E_T|$$

which corresponds to the number of integrations in time and polynomial decorations. We define the space of decorated trees  $\mathcal{T}_o^r \subset \mathcal{T}_o$  as

$$\mathcal{T}_o^r = \{\mathcal{I}_o(T), T \in \mathcal{T}^r\}, \quad \mathcal{T}^r = \{T \in \mathcal{T}, n_+(T) \le r, (\Pi \mathcal{I}_o(T)) \neq 0\}.$$

The condition  $(\Pi \mathcal{I}_o(T)) \neq 0$  guarantees that we consider only trees which have been generated by iterations of Duhamel's formula 3.54. In the following we consider

$$w_o^r(\hat{\mathbf{v}},\tau) = e^{\tau \mathcal{L}_o} v_o + \sum_{T \in \mathcal{T}^r} (\Pi \mathcal{I}_o(T))(\hat{\mathbf{v}},\tau)$$
(3.56)

which solves (3.54) up to order r + 1 in the following sense:

Proposition 3.4.1 One has that

$$w_o^r(\hat{\mathbf{v}},t) = e^{\tau \mathcal{L}_o} v_o + \sum_{\mathbf{l} \in \mathfrak{L}_-} \int_0^\tau e^{(\tau-\xi)\mathcal{L}_o} \Psi_o^{\mathbf{l}}(\mathbf{w}_o^{\mathbf{l},r})(\hat{\mathbf{v}},\xi) V_{\mathbf{l}} d\xi + \mathcal{O}(\tau^{r+2})$$

where  $\mathbf{w}_{o}^{\mathfrak{l},r}$  denotes the term  $(w_{\bar{o}}^{r})_{\bar{o}\in\mathfrak{L}_{\perp}^{\mathfrak{l},o}}$  and the remainder  $\mathcal{O}(\tau^{r+2})$  involves commutators under the form:

$$\mathcal{C}^{r}[f,\mathcal{L}_{o}]$$

(however, not full powers of  $\mathcal{L}_{o}^{r}$ ) with f a function coming from the coefficients  $\Psi_{o}$ .

*Proof.* Let B be given by

$$B = e^{\tau \mathcal{L}_o} v_o + \sum_{\mathfrak{l} \in \mathfrak{L}_-} \int_0^\tau e^{(\tau - \xi)\mathcal{L}_o} \Psi_o^{\mathfrak{l}}(\mathbf{w}_o^{\mathfrak{l},r})(\hat{\mathbf{v}},\xi) V_{\mathfrak{l}} d\xi$$
$$= e^{\tau \mathcal{L}_o} v_o + \sum_{\mathfrak{l} \in \mathfrak{L}_-} \int_0^\tau e^{(\tau - \xi)\mathcal{L}_o} \mathcal{B}_o^{\mathfrak{l}} \left( \prod_{\bar{o} \in \mathfrak{L}_+^{\mathfrak{l},o}} f_{o,\bar{o}}^{\mathfrak{l}}(w_{\bar{o}}^r(\hat{\mathbf{v}},\xi)) \right) V_{\mathfrak{l}} d\xi.$$

Then, we have

$$w_o^r(\hat{\mathbf{v}},\xi) = e^{\xi \mathcal{L}_o} v_o + R_{r,o}(\hat{\mathbf{v}},\xi), \quad R_{r,o}(\hat{\mathbf{v}},\xi) = \sum_{T \in \mathcal{T}^r} \left(\Pi \mathcal{I}_o(T)\right)(\hat{\mathbf{v}},\xi).$$

By performing Taylor expansions around the point  $e^{\xi \mathcal{L}_o} v_o$ , one gets:

$$B = e^{\tau \mathcal{L}_o} v_o + \sum_{\mathfrak{l} \in \mathfrak{L}_-} \int_0^\tau e^{(\tau-\xi)\mathcal{L}_o} \mathcal{B}_o^{\mathfrak{l}} \sum_{\sum_{\bar{o}} k_{\bar{o}} \leq r} \frac{1}{\prod_{\bar{o}} k_{\bar{o}}!}$$
$$\prod_{\bar{o} \in \mathfrak{L}_+^{\mathfrak{l},o}} R_{r,\bar{o}}^{k_{\bar{o}}}(\hat{\mathbf{v}},\xi) f_{o,\bar{o}}^{\mathfrak{l},(k_{\bar{o}})} (e^{\xi\mathcal{L}_{\bar{o}}} v_{\bar{o}}) V_{\mathfrak{l}} d\xi + \mathcal{O}(\tau^{r+2})$$

where the sum  $\sum_{\bar{o}}$  and the product  $\prod_{\bar{o}}$  run over  $\bar{o} \in \mathfrak{L}^{\mathfrak{l},o}_+$ . The error  $\mathcal{O}(\tau^{r+2})$  depends on the derivatives of  $f_{o,\bar{o}}^{\mathfrak{l}}$  and on a polynomial in  $\hat{\mathbf{v}}$  using the bound (3.55). The next step of the approximation is to pull out the term  $e^{\xi \mathcal{L}_{\bar{o}}} v_{\bar{o}}$ :

$$\begin{split} f_{o,\bar{o}}^{\mathfrak{l},(k_{\bar{o}})}(e^{\xi\mathcal{L}_{\bar{o}}}v_{\bar{o}}) &= e^{\xi\mathcal{L}_{\bar{o}}}f(v_{\bar{o}}) + \sum_{\ell_{\bar{o}}=1}^{r}\frac{\xi^{\ell_{\bar{o}}}}{\ell_{\bar{o}}!}e^{\xi\mathcal{L}_{\bar{o}}}\mathcal{C}^{\ell_{\bar{o}}}[f_{o,\bar{o}}^{\mathfrak{l},(k_{\bar{o}})},\mathcal{L}_{\bar{o}}](v_{\bar{o}}) \\ &+ \mathcal{O}(\xi^{r+1}\mathcal{C}^{r+1}[f_{o,\bar{o}}^{\mathfrak{l},(k_{\bar{o}})},\mathcal{L}_{\bar{o}}](v_{\bar{o}})). \end{split}$$

The remainder coming from this approximation is the leading error. It involves  $\mathcal{L}_o$  with an iterated commutator. For the next computations, we will omit this error and write all the identities up to some terms which are neglectable in comparison to this error. Inserting the previous expansion into A and neglecting the terms which are of order bigger than  $\mathcal{O}(\tau^{r+2})$ , we get:

$$B = e^{\tau \mathcal{L}_{o}} v_{o} + \sum_{\mathfrak{l} \in \mathfrak{L}_{-}} \int_{0}^{\tau} e^{(\tau-\xi)\mathcal{L}_{o}} \sum_{\sum_{\bar{o}} \ell_{\bar{o}} + \sum_{\bar{o},i} n_{+}(T_{\bar{o},i}) \leq r} \frac{\xi^{\sum_{\bar{o}} \ell_{\bar{o}}}}{\prod_{\bar{o}} k_{\bar{o}}! \ell_{\bar{o}}!}$$
$$\mathcal{B}_{o}^{\mathfrak{l}} \left( \prod_{\bar{o} \in \mathfrak{L}_{+}^{\mathfrak{l},o}} \prod_{i=1}^{k_{\bar{o}}} (\Pi \mathcal{I}_{\bar{o}}(T_{\bar{o},i}))(\hat{\mathbf{v}},\xi) \left( e^{\xi\mathcal{L}_{\bar{o}}} \mathcal{C}^{\ell_{\bar{o}}}[f_{o,\bar{o}}^{\mathfrak{l},(k_{\bar{o}})},\mathcal{L}_{\bar{o}}](v_{\bar{o}}) \right) V_{\mathfrak{l}} \right) d\xi$$

where the  $T_{\bar{o},i}$  are decorated trees and  $\sum_{\bar{o},i} n_+(T_{\bar{o},i})$  is a shorthand notation for

$$\sum_{\bar{o}\in\mathfrak{L}^{\mathfrak{l},o}_+}\sum_{k\bar{o}}\sum_{i=1}^{k\bar{o}}n_+(T_{\bar{o},i}).$$

Then, if we fix the product  $\prod_{i=1}^{k_{\bar{o}}} \mathcal{I}_{\bar{o}}(T_{\bar{o},i})$ , one has:

$$\prod_{i=1}^{k_{\bar{o}}} \mathcal{I}_{\bar{o}}(T_{\bar{o},i}) = \prod_{i} \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o},i})^{k_{\bar{o},i}}$$

where the  $\tilde{T}_{\bar{o},i}$  are distincts and  $\sum_{i} k_{\bar{o},i} = k_{\bar{o}}$ . This term appears in B with the following combinatorial coefficient:

$$\frac{k_{\bar{o}}!}{\prod_i k_{\bar{o},i}!}$$

Let us notice

$$S_{\text{root}}\left(\lambda_{\mathbf{I}}^{k}\prod_{\bar{o},i}\mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o},i})^{k_{\bar{o},i}}\right) = k!\prod_{\bar{o},i}k_{\bar{o},i}!$$

and

$$\sum_{\Sigma_{\bar{\sigma}} \ell_{\bar{\sigma}} = \ell} \prod_{\bar{\sigma} \in \mathfrak{L}_{+}^{\mathfrak{l}, \sigma}} \frac{1}{\ell_{\bar{\sigma}}!} \left( e^{\xi \mathcal{L}_{\bar{\sigma}}} \mathcal{C}^{\ell_{\bar{\sigma}}}[f_{o,\bar{\sigma}}^{\mathfrak{l},(k_{\bar{\sigma}})}, \mathcal{L}_{\bar{\sigma}}](v_{\bar{\sigma}}) \right) = \frac{\partial_{\ell}}{\ell!} \prod_{\bar{\sigma} \in \mathfrak{L}_{+}^{\mathfrak{l}, o}} \left( e^{\xi \mathcal{L}_{\bar{\sigma}}} f_{o,\bar{\sigma}}^{\mathfrak{l},(k_{\bar{\sigma}})}(v_{\bar{\sigma}}) \right)$$

In this identity, we exploit the Leibniz rule for the derivative  $\partial$ . It can be understood as a variant of Faa di Bruno formula see [BCCH20, Lem. A.1]. We can rewrite B into

$$\begin{split} B &= e^{\tau \mathcal{L}_{o}} v_{o} + \sum_{\mathbf{l} \in \mathfrak{L}_{-}} \int_{0}^{\tau} e^{(\tau - \xi)\mathcal{L}_{o}} \mathcal{B}_{o}^{\mathsf{I}} \left( \sum_{\ell + \sum_{\bar{o}, i} n_{+}(\tilde{T}_{\bar{o}, i}) \leq r} \frac{\xi^{\ell}}{\ell!} \partial_{\ell} \right) \\ &\prod_{\bar{o} \in \mathfrak{L}_{+}^{o, \mathsf{I}}} \prod_{i} \frac{1}{k_{\bar{o}, i}!} \Big( \Pi \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o}, i})^{k_{\bar{o}, i}} \Big) (\hat{\mathbf{v}}, \xi) \Big( e^{\xi \mathcal{L}_{\bar{o}}} f_{o, \bar{o}}^{\mathsf{I}, (k_{\bar{o}})} (v_{\bar{o}}) \Big) V_{\mathsf{I}} \right) d\xi \\ &= e^{\tau \mathcal{L}_{o}} v_{o} + \sum_{\mathsf{I} \in \mathfrak{L}_{-}} \int_{0}^{\tau} e^{(\tau - \xi)\mathcal{L}_{o}} \mathcal{B}_{o}^{\mathsf{I}} \left( \sum_{\ell + \sum_{\bar{o}, i} n_{+}(\tilde{T}_{\bar{o}, i}) \leq r} \frac{\xi^{\ell}}{S_{\mathrm{root}} \Big( \lambda_{\mathsf{I}}^{\ell} \prod_{\bar{o}, i} \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o}, i})^{k_{\bar{o}, i}} \Big) \\ &\prod_{\bar{o} \in \mathfrak{L}_{+}^{\mathsf{I}, o}} \prod_{i} \Big( \Pi \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o}, i})^{k_{\bar{o}, i}} \Big) (\hat{\mathbf{v}}, \xi) \Upsilon_{\mathrm{root}}^{\Psi_{o}} \left( \lambda_{\mathsf{I}}^{\ell} \prod_{\bar{o}, i} \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o}, i})^{k_{\bar{o}, i}} \right) (\mathbf{v}, \xi) V_{\mathsf{I}} \right) d\xi. \end{split}$$

Then,

$$B = e^{\tau \mathcal{L}_o} v_o + \sum_{\mathfrak{l} \in \mathfrak{L}_{-}} \int_0^{\tau} e^{(\tau - \xi)\mathcal{L}_o} \sum_{\ell + \sum_{\bar{o},i} n_+(\tilde{T}_{\bar{o},i}) \leq r} \left( \tilde{\Pi}_o \lambda_{\mathfrak{l}}^{\ell} \prod_{\bar{o},i} \mathcal{I}_{\bar{o}}(\tilde{T}_{\bar{o},i})^{k_{\bar{o},i}} \right) (\hat{\mathbf{v}}, \xi) d\xi$$
$$= e^{\tau \mathcal{L}_o} v_o + \sum_{T \in \mathcal{T}^r} \int_0^{\tau} e^{(\tau - \xi)\mathcal{L}_o} \left( \tilde{\Pi}_o T \right) (\hat{\mathbf{v}}, \xi) d\xi$$
$$= e^{\tau \mathcal{L}_o} v_o + \sum_{T \in \mathcal{T}^r} (\Pi \mathcal{I}_o(T)) (\hat{\mathbf{v}}, \tau) = w_o^r (\hat{\mathbf{v}}, \tau)$$

which concludes the proof.

# 3.4.2 Numerical scheme and local error analysis

Now, we are able to describe the general numerical scheme:

**Definition 3.4.2 (The general numerical scheme)** For fixed  $r \in \mathbf{N}$  and a domain A, we define the general numerical scheme as:

$$u_{o,A}^{r}(\hat{\mathbf{v}},\tau) = e^{\tau \mathcal{L}_{o}} v_{o} + \sum_{T \in \mathcal{T}^{r+1}} \Pi_{A}^{r}(\mathcal{I}_{o}(T))(\hat{\mathbf{v}},\tau).$$
(3.57)

The numerical scheme (3.57) approximates the exact solution locally up to order r + 2. More precisely, the following theorem holds.

**Theorem 3.4.3 (Local error)** The numerical scheme (3.57) with initial value  $v_o = u_o(0)$  approximates the exact solution  $u_o$  up to a local error of type

$$u_{o,A}^{r}(\hat{\mathbf{v}},\tau) - u_{o}(\hat{\mathbf{v}},\tau) = \sum_{T \in \mathcal{T}^{r+1}} \mathcal{O}\big(\tau^{r+2} \mathcal{L}_{low}^{r}(\mathcal{I}_{o}(T), \hat{\mathbf{v}}, A)\big)$$

where the operator  $\mathcal{L}_{low}^{r}(\mathcal{I}_{o}(T), \hat{\mathbf{v}}, A)$ , given in Definition 3.3.8, embeds the necessary regularity of the solution.

*Proof.* We recall that the exact solution  $u_o$  up to order r is given by

$$w_o^r(\hat{\mathbf{v}},\tau) = e^{\tau \mathcal{L}_o} v_o + \sum_{T \in \mathcal{T}^r} (\Pi \mathcal{I}_o(T))(\hat{\mathbf{v}},\tau)$$

which satisfies from Proposition 3.4.1

$$u_o(\tau) - w_o^r(\tau) = \mathcal{O}\left(\tau^{r+2} p(\hat{\mathbf{v}}, (\mathcal{L}_o)_{o \in \mathcal{L}_+})\right)$$
(3.58)

for some polynomial p such that every  $\mathcal{L}_o$  appear under the form  $\mathcal{C}^{r+1}[f, \mathcal{L}_o]$ . Thanks to Theorem 3.3.9 we furthermore obtain that

$$u_{o,A}^{r}(\hat{\mathbf{v}},\tau) - w_{o}^{r}(\hat{\mathbf{v}},\tau) = \sum_{T \in \mathcal{T}^{r}} (\Pi - \Pi_{A}^{r})(\mathcal{I}_{o}(T))(\hat{\mathbf{v}},\tau)$$

$$= \sum_{T \in \mathcal{T}^{r}} \mathcal{O}\big(\tau^{r+2}\mathcal{L}_{\text{low}}^{r}(\mathcal{I}_{o}(T),\hat{\mathbf{v}},A)\big).$$
(3.59)

Next we write

$$u_{o,A}^r(\hat{\mathbf{v}},\tau) - u_o(\hat{\mathbf{v}},\tau) = u_{o,A}^r(\hat{\mathbf{v}},\tau) - w_o^r(\hat{\mathbf{v}},\tau) + w_o^r(\hat{\mathbf{v}},\tau) - u_o(\hat{\mathbf{v}},\tau)$$

where by the definition of  $\mathcal{L}_{low}^r(T, \hat{\mathbf{v}}, A)$  we easily see that the approximation error (3.59) is dominant compared to (3.58). Indeed, we will have also commutators of the form  $\mathcal{C}^{r+1}[f, \mathcal{L}_o]$  and  $\mathcal{C}^{r+1}[f, \mathcal{L}_{dom}]$  due to Proposition 3.3.7. The extra regularity needed is coming from the Taylor expansion of the lower part in Definition 3.3.2.

**Remark 3.4.4** Theorem 3.4.3 provides a local error estimate (order of consistency) for the low regularity schemes (3.57). With the aid of stability one can easily obtain a global error estimate with the aid of Lady Windamere's fan argument [HLW10]. However, the necessary stability estimates in general rely on the algebraic structure of the underlying space.

In case of parabolic problems the parabolic smoothing property can be used to obtain global error estimates in low regularity spaces. With the aid of semi group theory the regularity assumptions on the solution may be even pushed down further thanks to the point wise smoothing properties. The situation of dispersive PDEs is more involved. In the stability analysis in Sobolev spaces  $H^r$  one classically exploits bilinear estimates of type

$$||vw||_r \le c_{r,d} ||v||_r ||w||_r.$$

The latter only hold for r > d/2 and thus restricts the analysis to sufficiently smooth Sobolev spaces  $H^r$  with r > d/2. To obtain (sharp)  $L^2$  global error estimates one needs to exploit discrete Strichartz estimates and discrete Bourgain spaces in the periodic setting, see, e.g., [IZ09, ORS21, ORS22b]. The main difficulty thereby lies in the fact that their continuous counterparts are not point wise in time.

A general global error analysis in the general setting presented is out of the scope of this paper. Nevertheless, in the case of the Gross-Pitaevskii equation first and second order convergence is obtained in [AB23a].

## 3.5 Examples

In this section we illustrate our general framework (see Definition 3.57) and its local error analysis (see Theorem 3.4.3) on two examples: The Gross–Pitaevskii equation (see Section 3.5.1) and the Sine–Gordon equation (see Section 3.5.3).

# 3.5.1 The Gross–Pitaevskii equation

As a first example let us consider the Gross-Pitaevskii (GP) equation

$$i\partial_t u(t,x) + \Delta u(t,x) = V(x)u(t,x) + |u(t,x)|^2 u(t,x) \quad (t,x) \in \mathbf{R} \times \Omega$$
(3.60)

on a sufficiently smooth domain  $\Omega \subset \mathbf{R}^d$  in dimension  $d \leq 3$ , and an initial condition

$$u_{|t=0} = u_0. (3.61)$$

We prescribe homogeneous Dirichlet boundary conditions,

$$u(t, \cdot)|_{\partial\Omega} = 0, \quad V(\cdot)|_{\partial\Omega} = 0,$$

where  $\Omega$  is a smooth open set with compact boundary. We define the operator  $\mathcal{L} = i\Delta$  on the Hilbert space  $L^2(\Omega)$ . Its domain is given by

$$D(\Delta) = D(\mathcal{L}) = (H^2 \cap H_0^1)(\Omega)$$
(3.62)

where  $H^2(\Omega), H^1_0(\Omega)$  denote the classical Sobolev spaces.

One setting of the Gross-Pitaevskii equation is to describe the dynamics of Bose-Einstein condensates in a potential trap. In many physically relevant situations the potential is rough or disordered ([HP17], [NBP13], [WWW<sup>+</sup>98]) which motivates the study of the low-regularity framework.

#### First order low regularity integrator for Gross-Pitaevskii

**Corollary 3.5.1** At first order our general low regularity scheme (3.57) for the Gross-Pitaevskii equation (3.60) takes the form

$$u^{n+1} = \Phi^{\tau}_{GP}(u^n) = e^{i\tau\Delta}u^n - i\tau \Big[ (e^{i\tau\Delta}u^n)(e^{i\tau\Delta}\varphi_1(-i\tau\Delta)V) + (e^{i\tau\Delta}(u^n)^2)(e^{i\tau\Delta}\varphi_1(-2i\tau\Delta)\overline{u}^n) \Big]$$
(3.63)

where the filter function  $\varphi_1$  is defined as  $\varphi_1(\sigma) = \frac{e^{\sigma}-1}{\sigma}$ . The scheme (3.63) is locally of order  $\mathcal{O}(\tau^2 |\nabla|(u+V))$ .

In case of more regular solutions and potential the above low regularity scheme can be simplified to

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau \left(u^n V + (u^n)^2 \overline{u}^n\right),$$
(3.64)

which is locally of order  $\mathcal{O}(\tau^2 \Delta(u+V))$ .

*Proof.* We choose r = 0 in Definition 3.4.2 in order to obtain a local error of order one. We recall from Example 8 that for the Gross-Pitaevskii equation we have that  $\mathcal{L}_o = i\Delta$ ,  $V_0 = 1$ ,  $V_1 = V$ ,  $\mathfrak{L}_+ = \{o, \bar{o}\}$ ,  $u_o = u$ ,  $u_{\bar{o}} = \bar{u}$ ,  $v_o = v$  and  $v_{\bar{o}} = \bar{v}$ . From equation (3.57) it then follows that the first-order scheme is of the form,

$$u_{o,A}^{0}(\hat{\mathbf{v}},\tau) = e^{i\tau\Delta}v + \sum_{T\in\mathcal{T}^{1}} \Pi_{A}^{0}(\mathcal{I}_{o}(T))(\hat{\mathbf{v}},\tau), \qquad (3.65)$$

where  $\hat{\mathbf{v}} = (v, \bar{v}, V)$  and where on has

$$\mathcal{T}^1 = \{T_0, T_1\}, \quad T_0 = \lambda_0 = \circ, \quad T_1 = \lambda_1 = \bullet.$$

We recall from (3.28) that

$$\mathcal{I}_o(\lambda_0) = \mathbf{I}$$

encodes the iterated integral

$$(\Pi \mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}},\tau) = -i \int_0^\tau e^{i(\tau-s)\Delta} \left( \left( e^{is\Delta} v^2 \right) \left( e^{-is\Delta} \bar{v} \right) \right) ds$$
(3.66)

and that

$$\mathcal{I}_o(\lambda_1) =$$

encodes the iterated integral

$$(\Pi \mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}},\tau) = -i \int_0^\tau e^{i(\tau-s)\Delta} \left( \left( e^{is\Delta} v \right) V \right) ds.$$
(3.67)

In order to compute the approximation (3.65), we refer to equation (3.42) where the approximated character  $\Pi_A^r$  is defined. We have, for i = 0, 1, that

$$(\Pi^0_A \mathcal{I}_o(\lambda_i))(\hat{\mathbf{v}}, \tau) = \mathcal{K}^0_{o,A} (\tilde{\Pi}_{o,A} \mathcal{D}_{-1}(\lambda_i)(\hat{\mathbf{v}}, \cdot))(\tau)$$
$$= \mathcal{K}^0_{o,A} \left( \frac{\Upsilon^{\Psi_o}_{\text{root}}[\lambda_i]}{S_{\text{root}}(\lambda_i)} (\hat{\mathbf{v}}, \cdot) V_i \right)(\tau)$$

where to obtain the second line we used the definition 3.37 of  $\mathcal{D}_{-1}$ , namely that  $\mathcal{D}_{-1}(\lambda_i) = \lambda_i$ . By recalling the computations made in Example 9 one has,

$$S_{\text{root}}(\lambda_0) = S_{\text{root}}(\lambda_1) = 1,$$
  

$$\Upsilon^{\Psi_o}_{\text{root}}[\lambda_0](\mathbf{v},\xi) = \left(-ie^{i\xi\Delta}v^2\right) \left(e^{-i\xi\Delta}\bar{v}\right), \quad \Upsilon^{\Psi_o}_{\text{root}}[\lambda_1](\mathbf{v},\xi) = -ie^{i\xi\Delta}v$$

where  $\mathbf{v} = (v, \bar{v})$  and we recall that we set  $\mathcal{B}_o^{\mathfrak{l}} = \mathrm{id}$  for  $(o, \mathfrak{l}) \in \mathfrak{L}_+ \times \mathfrak{L}_-$  when studying the equation (3.60). Hence, by collecting the above computations one has,

$$\Pi^{0}_{A}(\mathcal{I}_{o}(\lambda_{0}))(\hat{\mathbf{v}},\tau) = \mathcal{K}^{0}_{o,A}(F^{1}(\hat{\mathbf{v}},\cdot)))(\tau), \qquad (3.68)$$
$$\Pi^{0}_{A}(\mathcal{I}_{o}(\lambda_{1}))(\hat{\mathbf{v}},\tau) = \mathcal{K}^{0}_{o,A}(F^{2}(\hat{\mathbf{v}},\cdot))(\tau)$$

where

$$F^{1}(\hat{\mathbf{v}},\xi) := -i(e^{i\xi\Delta}v^{2})(e^{-i\xi\Delta\bar{v}})$$
(3.69)

$$F^2(\hat{\mathbf{v}},\xi) := -i(e^{i\xi\Delta}v)V. \tag{3.70}$$

We are left to apply Definition 3.3.2 of the operator  $\mathcal{K}_{o,A}^0$  on (3.69) and (3.70) which will yield a first-order approximation of the integrals (3.66) and (3.67). The approximation of these integrals, and hence the structure of the scheme, will depend on the regularity assumptions on the initial data v and the potential V. We first show the construction of the first order scheme (3.63), which requires  $H^1$ -regularity on v and V. Hence, by taking into account the boundary conditions, we fix  $A = H_0^1(\Omega)^3$  and construct a first order scheme for  $(v, \bar{v}, V) \in A$  (see also Remark 3.3.5 for further details on the choice of A).

#### The case of $A = (H_0^1(\Omega))^3$ :

We first note that given our regularity assumptions we have that  $v, V \notin D(\mathcal{L}_o) = H^2(\Omega) \cap H^1_0(\Omega)$ . Hence, our approximation cannot consist of merely applying Taylor-expansions (3.44) of all the operators, since it would require  $H^2$  regularity on the initial data and on the potential. Indeed, given that  $r, \ell = 0$  we have that

$$\bigcup_{n+\sum_{i\in J}k_i\leq 1} D(G_{n,(k_i)_{i\in J}}) \subseteq (H^2(\Omega))^{\frac{1}{2}}$$

which implies that we do not have that  $A \subseteq D(G_{n,(k_i)_{i \in J}})$ , for all  $n + \sum_{i \in J} k_i \leq 1$ . In order to make an approximation of order one of the integrals (3.68) while only requiring  $H^1(\Omega)$  regularity on the initial data and potential we will apply the first point (i) of Definition 3.3.2.

We start by dealing with  $F^1$ , namely the first order approximation of the integral (3.66). 1. Computation of  $\Pi^0_A(\mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}}, \tau)$ . Using the notations in Definition 3.3.2 we compute the operator interactions, as is done in example 10 to obtain the following,

$$\mathcal{L}_1 = i\Delta, \ \mathcal{L}_2 = -\mathcal{L}_1, \ J = \{1, 2\}, \text{ and hence, } \mathcal{A}_1 = 0, \ \mathcal{A}_{\in} = -2i\Delta.$$

This implies that

$$\mathcal{L}_{\text{dom}} = \mathcal{A}_{\in} = -2i\Delta, \quad I = \{2\},$$

and that

$$\mathcal{A}_{low}^1 = \mathcal{A}_1 = 0, \quad \mathcal{A}_{low}^2 = \mathcal{A}_2 - \mathcal{L}_{dom} = 0.$$

We note that following (3.45) we indeed recover the same initial expression (3.79) of  $F^1$ :

$$F^{1}(\hat{\mathbf{v}},\xi) = (e^{-i\xi\Delta}u_{2}^{a})(e^{i\xi\Delta}u_{1}^{a}), \quad u_{1}^{a} = -iv^{2}, \quad u_{2}^{a} = \bar{v}.$$

We are now ready to compute  $\mathcal{K}^{0}_{o,A}(F^{1})$  using equation (3.46). Given the form of  $F^{1}$ , and the order of the scheme we enter the case where l = 0 and r = 0 respectively. This implies that q = 0 = n = m = p. Hence,  $\mathcal{K}^{0}_{o,A}(F^{1})$  takes the following simple form,

$$\mathcal{K}^{0}_{o,A}(F^{1}(\hat{\mathbf{v}},\cdot))(\tau) = \int_{0}^{\tau} (e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{-2i\xi\Delta+i\tau\Delta}\mathcal{M}_{\{2\}})(-iv^{2},\bar{v})d\xi \qquad (3.71)$$
$$= -i\int_{0}^{\tau} (e^{i\tau\Delta}v^{2})(e^{-2i\xi\Delta+i\tau\Delta}\bar{v})d\xi.$$

By integrating exactly the above expression using the  $\varphi_1$  function, and by (3.68) we have hence shown that,

$$\Pi^0_A(\mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}},\tau) = -i\tau(e^{i\tau\Delta}v^2)(e^{i\tau\Delta}\varphi_1(-2i\tau\Delta)\bar{v}).$$
(3.72)

2. Computation of  $\Pi^0_A(\mathcal{I}_o(\lambda_1))(\hat{\mathbf{v}},\tau)$ . Similarly, we apply definition 3.3.2 to compute  $\mathcal{K}^0_{o,A}(F^2)$ . Given the expression (3.70) of  $F^2$ , one has

$$\mathcal{L}_1 = i\Delta, \ \mathcal{L}_2 = 0, \ J = \{1, 2\}, u_1^b = -iv, \ u_2^b = V, \text{ and hence}, \ \mathcal{A}_1 = 0, \ \mathcal{A}_{\in} = -i\Delta.$$

This implies that

$$\mathcal{L}_{dom} = \mathcal{A}_{\in} = -i\Delta, \quad I = \{2\}$$

and that

$$\mathcal{A}_{low}^1 = \mathcal{A}_1 = 0, \quad \mathcal{A}_{low}^2 = \mathcal{A}_2 - \mathcal{L}_{dom} = 0$$

Then, again by equation (3.46) we have

$$\begin{aligned} \mathcal{K}^{0}_{o,A}(F^{2})(\tau) &= \int_{0}^{\tau} (e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i(\tau-\xi)\Delta}\mathcal{M}_{\{2\}})(-iv,V)d\xi \\ &= -i\int_{0}^{\tau} (e^{i\tau\Delta}v)(e^{i(\tau-\xi)\Delta}V)d\xi \\ &= -i\tau(e^{i\tau\Delta}v)(e^{i\tau\Delta}\varphi_{1}(-i\tau\Delta)V). \end{aligned}$$

Hence, we have

$$\Pi^0_A(\mathcal{I}_o(\lambda_1))(\hat{\mathbf{v}},\tau) = -i\tau(e^{i\tau\Delta}v)(e^{i\tau\Delta}\varphi_1(-i\tau\Delta)V).$$

Plugging the above computation into (3.65) yields the first order scheme (3.63).

The case of  $A = (H^2 \cap H_0^1)(\Omega)^3$ : Given that  $A = D(\mathcal{L}_o)^3$ , we apply the Taylor expansion (3.44) with  $r = \ell = 0$ , to obtain:

$$\begin{aligned} \mathcal{K}^{0}_{o,A}\Big((\Upsilon^{\Psi_{o}}_{\text{root}}[\lambda_{0}] + \Upsilon^{\Psi_{o}}_{\text{root}}[\lambda_{1}])(\mathbf{v},\cdot)\Big)(\tau) &= \int_{0}^{\tau} \Pi_{i\in\{1,2\}} u_{i}^{a} d\xi + \int_{0}^{\tau} \Pi_{i\in\{1,2\}} u_{i}^{b} d\xi \\ &= -i\tau(v^{2}v + vV), \end{aligned}$$

which thanks to (3.65) yields the first order scheme (3.64). **Local error analysis.** Using the recursive formula in Definition 3.3.8, one gets for  $T \in \{\lambda_0, \lambda_1\}$ 

$$\mathcal{L}^{0}_{low}(\mathcal{I}_{o}(T), \hat{\mathbf{v}}, A) = \mathcal{L}^{-1,o}_{low}(T, \hat{\mathbf{v}}, A) + R^{0}_{o,A}((\Pi_{A}\mathcal{I}^{0}_{o}(T))(\hat{\mathbf{v}}, \cdot)).$$

Then by (3.50), we get for  $l \in \{0, 1\}$ 

$$\mathcal{L}_{\text{low}}^{-1,o}(\lambda_{\mathfrak{l}}, \hat{\mathbf{v}}, A) = \frac{\Upsilon_{\text{root}}^{\Psi_o}[T]}{S_{\text{root}}(\lambda_i)}(\mathbf{v}, 0) V_{\mathfrak{l}}$$

which gives

$$\mathcal{L}_{\text{low}}^{-1,o}(\lambda_1, \hat{\mathbf{v}}, A) = -ivV, \quad \mathcal{L}_{\text{low}}^{-1,o}(\lambda_0, \hat{\mathbf{v}}, A) = -iv^2 \bar{v}.$$

It remains to compute  $R_{o,A}^0((\Pi_A T))$  whose value will depend on A. We start with the case  $A = (H_0^1(\Omega))^3$ . From our previous computations for the scheme, one has to apply (*ii*) from Lemma 3.3.6 that gives using notations of the proof of Corollary 3.5.1

$$\begin{aligned} R_{o,A}^{r}((\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}},\cdot)) &= \sum_{\sum_{i\in J}k_{i}+m+n+p=r-\ell+1} \mathcal{C}^{m}[\left(\mathcal{C}^{n}[\mathcal{M}_{J\setminus I},\mathcal{L}_{o}]\right)\\ (\mathcal{C}^{p}[\mathcal{M}_{I},\mathcal{L}_{\mathrm{dom}}+\mathcal{L}_{o}]),\mathcal{L}_{o}](((\mathcal{A}_{\mathrm{low}}^{i})^{k_{i}}u_{i}^{a})_{i\in J}). \end{aligned}$$

Here,  $r = \ell = k_i = 0$ , so it remains m + n + p = 1. As I and  $J \setminus I$  are singleton, there is only one non zero value given by m = 1. In the end, one obtains

$$R^0_{o,A}((\tilde{\Pi}_{o,A}\lambda_0)(\hat{\mathbf{v}},\cdot)) = \mathcal{C}[\mathcal{M}_{\{1,2\}},i\Delta](-iv^2,\bar{v}).$$

With a similar computation, one gets

$$R^0_{o,A}((\tilde{\Pi}_{o,A}\lambda_1)(\hat{\mathbf{v}},\cdot)) = \mathcal{C}[\mathcal{M}_{\{1,2\}}, i\Delta](-iv, V)]$$

In both cases, one sees that these terms ask first order derivatives on v and V. Indeed using Definition 3.1.2 of the commutator we have,

$$\mathcal{C}[\mathcal{M}_{\{1,2\}}, i\Delta](w, z) = -2i\nabla w \cdot \nabla z.$$
(3.73)

Next, for  $A = (H^2 \cap H_0^1)(\Omega)^3$ , one has to use (i) from Lemma 3.3.6, namely

$$R_{o,A}^{r}((\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}},\cdot)) = \sum_{\sum_{i}k_{i}+n=r-\ell+1} \mathcal{L}_{o}^{n}\left(\prod_{i\in J}\mathcal{L}_{i}^{k_{i}}u_{i}^{a}\right)$$

Here,  $r = \ell = 0$ , there for one obtains three terms:

$$R^0_{o,A}((\tilde{\Pi}_{o,A}\lambda_0)(\hat{\mathbf{v}},\cdot)) = \Delta(v^2\bar{v}) + \Delta(v^2)\bar{v} - v^2\Delta\bar{v}$$

A similar computation shows

$$R_{o,A}^{0}((\Pi_{o,A}\lambda_{1})(\hat{\mathbf{v}},\cdot)) = \Delta(vV) + \Delta(v)V + v\Delta V$$

which allows us to conclude.

Remark 3.5.2 (Error improvement) Classical approximation techniques, such as splitting or exponential integrator methods (see, e.g., [CG12, JL00, Lub08]) introduce a local error structure of type  $\mathcal{O}(\Delta u(t), \Delta V)$  and hence requires the solution and potential  $(u(t), V) \in D(\Delta)^2 = (H^2 \cap H_0^1)(\Omega)^2$  (see also (3.62)). The scheme (3.64) together with its local error coincides with a first order scheme obtained via exponential integrator methods. The local error of the low regularity GP integrator (3.63) on the other hand only requires the boundedness of first instead of second order spatial derivatives of the potential V and solution u. The first-order scheme (3.63) coincides with the first order scheme obtained in [AB23a], where the author proves first order convergence in  $H^s$ -norm while asking for low-regularity assumptions on the solution.

#### Second order Duhamel integrator for Gross-Pitaevskii

We first recall from equation (3.15) in our first example, that by following the Taylor expansion (3.9) and linearization (3.12) steps, we seek to provide a second order low-regularity approximation to the following iterated integrals:

$$w^{2}(\mathbf{v},\tau) = e^{i\tau\Delta}v - i\int_{0}^{\tau} e^{i(\tau-\xi)\Delta} \left( (e^{i\xi\Delta}v^{2})(e^{-i\xi\Delta}\bar{v}) + (e^{i\xi\Delta}v)V \right) d\xi$$

$$-i\int_{0}^{\tau} \xi e^{i(\tau-\xi)\Delta} \left( (e^{i\xi\Delta}\mathcal{C}[u^{2},i\Delta](v))(e^{-i\xi\Delta}\bar{v}) \right) d\xi$$

$$-\int_{0}^{\tau} e^{i(\tau-\xi)\Delta} \left( \left( \int_{0}^{\xi} e^{i(\xi-\xi_{1})\Delta} \left( (e^{i\xi_{1}\Delta}v^{2})(e^{-i\xi_{1}\Delta}\bar{v}) + (e^{i\xi_{1}\Delta}v)V \right) d\xi \right) (2(e^{i\xi\Delta}v)(e^{-i\xi\Delta}\bar{v}) + V) \right) d\xi$$

$$(3.74)$$

$$+ \int_0^\tau e^{i(\tau-\xi)\Delta} \bigg( \bigg( \int_0^\xi e^{-i(\xi-\xi_1)\Delta} \bigg( (e^{-i\xi_1\Delta}\bar{v}^2)(e^{i\xi_1\Delta}v) \\ + (e^{-i\xi_1\Delta}\bar{v})\bar{V} \bigg) d\xi_1 \bigg) (e^{i\xi\Delta}v^2) \bigg) d\xi.$$

**Corollary 3.5.3** At second order our general low regularity scheme (3.57) for the Gross-Pitaevskii equation (3.60) takes the form,

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau \left( (e^{i\tau\Delta}(u^n)^2)(e^{i\tau\Delta}\varphi_1(-2i\tau\Delta)\overline{u}^n) + (e^{i\tau\Delta}u^n)(e^{i\tau\Delta}\varphi_1(-i\tau\Delta)V) \right)$$

$$- i\tau^2 \mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-2i\tau\Delta) - \varphi_2(-2i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta]((u^n)^2, \overline{u}^n)$$

$$- i\tau^2 \mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-i\tau\Delta) - \varphi_2(-i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta](u^n, V)$$

$$- i\tau^2 \left( e^{i\tau\Delta} \mathcal{C}[u^2, i\Delta](u^n) \right) \left( e^{i\tau\Delta}\varphi_2(-2i\tau\Delta)\overline{u}^n \right)$$

$$- \frac{\tau^2}{2} (u^n |u^n|^4 + 3u^n |u^n|^2 V - |u^n|^2 u^n \overline{V} + u^n V^2)$$

$$= \Phi^{\tau}_{GP2}(u^n),$$
(3.75)

where the filter function  $\varphi_2$  is defined as  $\varphi_2(\sigma) = \frac{e^{\sigma} - \varphi_1(\sigma)}{\sigma}$ . The scheme (3.75) is locally of order  $\mathcal{O}(\tau^3 \Delta(u+V))$ .

**Remark 3.5.4** In case of more regular solution and potential the second order scheme (3.75) can be simplified, recovering for sufficiently smooth solutions and potential  $(u(t), V) \in D(\Delta^2)^2$  classical schemes (see also Remark 3.5.5 below).

*Proof.* We choose r = 1 in Definition 3.4.2 in order to obtain a local error of order two. From equation (3.57) it then follows that the second-order scheme is of the form,

$$u^{1}(\hat{\mathbf{v}},\tau) = e^{i\tau\Delta}v + \sum_{T\in\mathcal{T}^{2}}\Pi^{1}_{A}(\mathcal{I}_{o}(T))(\hat{\mathbf{v}},\tau), \qquad (3.76)$$

where  $\hat{\mathbf{v}} = (\mathbf{v}, \mathbf{V})$  and where are interested in the decorated trees  $\mathcal{I}_o(T)$  where T belongs to  $\mathcal{T}^2$  defined by,

$$\mathcal{T}^2 = \{T_0, ..., T_8\}, \quad T_0 = \lambda_0 = \circ, \quad T_1 = \lambda_1 = \bullet, \quad T_2 = \lambda_0^1 = \bullet$$
  
$$T_3 = \lambda_0 \mathcal{I}_o(\lambda_0) = \bigcirc, \quad T_4 = \lambda_0 \mathcal{I}_o(\lambda_1) = \bigcirc, \quad T_5 = \lambda_1 \mathcal{I}_o(\lambda_0) = \bigcirc,$$
  
$$T_6 = \lambda_1 \mathcal{I}_o(\lambda_1) = \bigcirc, \quad T_7 = \lambda_0 \mathcal{I}_{\overline{o}}(\lambda_0) = \bigcirc, \quad T_8 = \lambda_0 \mathcal{I}_{\overline{o}}(\lambda_1) = \bigcirc.$$

As stated previously, the approximation of the above integrals, and hence the structure of the scheme, will depend on the regularity assumptions of the initial data v and the potential V. Here, we show the construction of a second order scheme which requires  $H^2$ -regularity on v and V. Hence, by taking into account the boundary conditions, we fix  $A = (H^2(\Omega) \cap H_0^1(\Omega))^3$  and construct a second order scheme for  $(v, \bar{v}, V) \in A$ . 1. Computation of  $\prod_A^1(\mathcal{I}_o(\lambda_0))(\hat{\mathbf{v}}, \tau)$ .

We recall that  $\mathcal{I}_o(\lambda_0)$  encodes the first integral (3.66). Next, we make the following remark; given our regularity assumptions  $(v, \bar{v}, V) \in A$ , we have that  $v, V \notin D(\mathcal{L}_o^2) \subset H^4(\Omega)$ . Hence, our approximation cannot consist of merely applying Taylor-expansions (3.44) of all the operators, since it would require  $H^4$  regularity on the initial data and on the potential, together with the according boundary conditions. Indeed, given that r = 1 and  $\ell = 0$  we have that

$$\bigcup_{n+\sum_{i\in J}k_i\leq 2} D(G_{n,(k_i)_{i\in J}}) \subseteq (H^4(\Omega))^3$$

which implies that we do not have that  $A \subseteq D(G_{n,(k_i)_{i \in J}})$ , for all  $n + \sum_{i \in J} k_i \leq 2$ . In order to make an approximation of order two of the integrals (3.68) while only requiring  $H^2(\Omega)$  regularity on the initial data we will apply the first point (*i*) of Definition 3.3.2, as done in Example 10. Indeed, following the results established in Example 10 we have:

$$(\Pi_A \mathcal{I}^1_o(\lambda_0))(\hat{\mathbf{v}}, \xi) = -i \int_0^\tau \left( e^{i\tau\Delta} v^2 \right) \left( e^{i(\tau-2\xi)\Delta} \overline{v} \right) d\xi + \int_0^\tau (\tau-\xi) \mathcal{C}[\left( e^{i\tau\Delta} \mathcal{M}_{\{1\}} \right) \left( e^{i(\tau-2\xi)\Delta} \mathcal{M}_{\{2\}} \right), i\Delta](-iv^2, \overline{v}) d\xi = -i[\tau(e^{i\tau\Delta} v^2)(e^{i\tau\Delta} \varphi_1(-2i\tau\Delta)\overline{v})$$

+ 
$$\tau^2 \mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-2i\tau\Delta))$$
  
-  $\varphi_2(-2i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta](v^2, \bar{v})]$ 

#### 2. Computation of $\Pi^1_A(\mathcal{I}_o(\lambda_1))(\hat{\mathbf{v}}, \tau)$ .

We recall that  $\mathcal{I}_o(\lambda_1)$  encodes the second integral (3.67). Using the same arguments as in the above together with the computations of  $\Pi^0_A(\mathcal{I}_o(\lambda_1))(\hat{\mathbf{v}},\tau)$  made in the proof of Corollary 3.5.1, it follows that,

$$(\Pi_A \mathcal{I}_o^1(\lambda_1))(\hat{\mathbf{v}},\xi) = -i \Big( \tau(e^{i\tau\Delta}v)(e^{i\tau\Delta}\varphi_1(-i\tau\Delta)V) + \tau^2 \mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-i\tau\Delta)) - \varphi_2(-i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta](v,V) \Big).$$

3. Computation of  $\Pi^1_A (\mathcal{I}_o(\lambda^1_0))(\hat{\mathbf{v}}, \tau)$ . We have that

$$\mathcal{I}_o(\lambda_0^1) =$$

encodes the iterated integral

$$-i\int_0^\tau \xi e^{i(\tau-\xi)\Delta} \Big( (e^{i\xi\Delta} \mathcal{C}[u^2,i\Delta](v))(e^{-i\xi\Delta}\bar{v}) \Big) d\xi$$

Using the computation made in example (9) we have,

$$\begin{aligned} (\Pi^{1}_{A}\mathcal{I}_{o}(\lambda^{1}_{0}))(\hat{\mathbf{v}},\tau) &= \mathcal{K}^{1}_{o,A}((\tilde{\Pi}_{o,A}\mathcal{D}_{0}(\lambda^{1}_{0}))(\hat{\mathbf{v}},\cdot))(\tau) \\ &= \mathcal{K}^{1}_{o,A}\left(\frac{\Upsilon^{\Psi_{o}}_{\mathrm{root}}[\lambda^{1}_{0}]}{S_{\mathrm{root}}(\lambda^{1}_{0})}(\mathbf{v},\xi)V_{0}\xi\right)(\tau) \\ &= \mathcal{K}^{1}_{o,A}\left(\xi\left(e^{\xi\mathcal{L}}\mathcal{C}[u^{2},\Delta](v)\right)\left(e^{-\xi\mathcal{L}}\bar{v}\right)\right)(\tau). \end{aligned}$$

To apply Definition 3.3.2 we let  $u_1 = C[u^2, \Delta](v)$ , and  $u_2 = \bar{v}$ . The operator interactions are the same as those computed previously. We are in the case where  $r = 1 = \ell$ , hence q = 0 = m = n = p. From Definition 3.3.2 part i., it then follows that,

$$(\Pi^0_A \mathcal{I}_o(\lambda_0^1))(\hat{\mathbf{v}},\tau) = \int_0^\tau \xi[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{-2i\xi\Delta+i\tau\Delta}\mathcal{M}_{\{2\}})](u_1,u_2)d\xi$$
$$= -i\tau^2 \Big(e^{i\tau\Delta}\mathcal{C}[u^2,i\Delta](v)\Big)\Big(e^{i\tau\Delta}\varphi_2(-2i\tau\Delta)\bar{v}\Big)$$

4. Computation of  $\Pi^1_A(\mathcal{I}_o(T_3))(\hat{\mathbf{v}}, \tau)$ . We have that

$$\mathcal{I}_o(T_3) = \bigcup_{i=1}^{N}$$

encodes the iterated integral

$$-\int_0^\tau e^{i(\tau-\xi)\Delta} \left( \left( \int_0^\xi e^{i(\xi-\xi_1)\Delta} \left( (e^{i\xi_1\Delta}v^2)(e^{-i\xi_1\Delta}\bar{v}) \right) d\xi_1 \right) 2(e^{i\xi\Delta}v)(e^{-i\xi\Delta}\bar{v}) \right) d\xi,$$

By definition of the projection operator  $\mathcal{D}_r$  and (3.3.2) we have,

$$\Pi_{A}^{1}(\mathcal{I}_{o}(T_{3}))(\hat{\mathbf{v}},\tau) = \mathcal{K}_{o,A}^{1}(\tilde{\Pi}_{o,A}\mathcal{D}_{0}(\lambda_{0}\mathcal{I}_{o}(\lambda_{0}))(\hat{\mathbf{v}},\cdot))(\tau)$$

$$= \mathcal{K}_{o,A}^{1}(\tilde{\Pi}_{o,A}(\lambda_{0}\mathcal{I}_{o}^{0}(\lambda_{0})))(\hat{\mathbf{v}},\tau)$$

$$= \mathcal{K}_{o,A}^{1}\left(\frac{\Upsilon_{\text{root}}^{\Psi_{o}}[T_{3}]}{S_{\text{root}}(T_{3})}(\mathbf{v},\xi)V_{0}(\Pi_{A}\mathcal{I}_{o}^{0}(\lambda_{0}))(\hat{\mathbf{v}},\xi)\right)(\tau)$$
(3.77)

where  $V_0 = 1$ . Hence, in order to compute the above approximation we are left to compute  $S_{\text{root}}(T_3)$ ,  $\Upsilon^{\Psi_o}_{\text{root}}[T_3]$ , and  $(\Pi_A \mathcal{I}^0_o(\lambda_0))(\hat{\mathbf{v}}, \xi)$ . First, by definition (3.40) of  $S_{\text{root}}(T_3)$  we have that i = 1 = j and  $\beta_{1,1} = 1$ , where  $o_1 = o$ . Hence, we

have that  $S_{\text{root}}(T_3) = 1$ . Second, by definition (3.41) of  $\Upsilon^{\Psi_o}_{\text{root}}[T]$  we have that,

$$\Upsilon^{\Psi_o}_{\text{root}}[T_3](\mathbf{v},\xi) = (D_o\hat{\Psi}^0_o)(\mathbf{v},\xi)$$
$$= D_o\Big((e^{i\xi\Delta}(-iv_o^2))(e^{-i\xi\Delta}v_{\bar{o}})\Big)$$
$$= -2i(e^{i\xi\Delta}v)(e^{-i\xi}\bar{v})$$

since  $v_o = v$ , and  $v_{\bar{o}} = \bar{v}$ . Finally, following the computations made at the beginning of example 10 we have,

$$\left(\Pi_A \mathcal{I}_o^0(\lambda_0)\right)(\hat{\mathbf{v}},\xi) = \mathcal{K}_{o,A}^0(F(\hat{\mathbf{v}},\cdot))(\xi)$$

where

$$F(\mathbf{v},\xi) = \prod_{i \in \{1,2\}} e^{\xi \mathcal{L}_i} u_i, \quad u_1 = -iv^2, \quad u_2 = \bar{v}, \quad \mathcal{L}_1 = i\Delta, \quad \mathcal{L}_2 = -i\Delta$$

In our current setting we have that  $r = 0 = \ell$  and hence  $D(\mathcal{L}_o^{r-\ell+1}) = D(\Delta)$ . Given the regularity assumptions on v, namely that  $v \in H^2 \cap H_0^1$ , it follows that  $v \in D(\Delta)$ . Following (3.44) we can then Taylor expand all the operators appearing in the integral which yields the following approximation,

$$(\Pi_A \mathcal{I}^0_o(\lambda_0))(\hat{\mathbf{v}}, \xi) = \mathcal{K}^0_{o,A}(F(\hat{\mathbf{v}}, \cdot))(\xi) = \int_0^{\xi} \prod_{i \in \{1,2\}} u_i d\xi = -i\xi v^2 \bar{v}$$

Hence, plugging the above computations in (3.77) yields,

$$\Pi^{1}_{A}(\mathcal{I}_{o}(T_{3}))(\mathbf{v},\tau) = -2\mathcal{K}^{1}_{o,A}\left(\xi(e^{i\xi\Delta}v)(e^{-i\xi\Delta}\bar{v})v^{2}\bar{v}\right)(\tau)$$

$$= -2\mathcal{K}^{1}_{o,A}\left(\tilde{F}(\hat{\mathbf{v}},\xi)\right)(\tau)$$
(3.78)

where

$$\tilde{F}(\hat{\mathbf{v}},\xi) = \xi^{\ell} \prod_{i \in \{1,2,3\}} e^{\xi \mathcal{L}_{i}} u_{i}, \qquad \mathcal{L}_{1} = i\Delta, \ \mathcal{L}_{2} = -i\Delta, \ \mathcal{L}_{3} = 0,$$

$$u_{1} = v \ u_{2} = \bar{v}, \ u_{3} = v^{2}\bar{v} \text{ and } \ell = 1.$$
(3.79)

In our current setting we have that  $r = 1 = \ell$  and hence  $D(\mathcal{L}_o^{r-\ell+1}) = D(\Delta)$ . Given that  $v \in D(\Delta)$ , by following (3.44) we can once again Taylor expand the remaining operators inside the integral which yields the following approximation,

$$\mathcal{K}_{o,A}^{1}\Big(\tilde{F}(\hat{\mathbf{v}},\xi)\Big)(\tau) = \int_{0}^{\tau} \xi \prod_{i \in \{1,2,3\}} u_{i} d\xi = \frac{\tau^{2}}{2} v |v|^{4},$$

by definition of the  $(u_i)_{i=1,\dots,4}$ . Hence, from (3.78) it follows that,

$$\Pi^1_A(\mathcal{I}_o(T_3))(\hat{\mathbf{v}},\tau) = -\tau^2 v |v|^4.$$

5. Computation of  $\Pi^1_A(\mathcal{I}_o(T_4))(\hat{\mathbf{v}},\tau)$ ,  $\Pi^1_A(\mathcal{I}_o(T_5))(\hat{\mathbf{v}},\tau)$ , and  $\Pi^1_A(\mathcal{I}_o(T_6))(\hat{\mathbf{v}},\tau)$ . We have that

$$\mathcal{I}_o(T_4) = \left\{, \quad \mathcal{I}_o(T_5) = \left\{, \quad \mathcal{I}_o(T_6) = \right\}.\right\}$$

respectively encode

$$-\int_0^\tau e^{i(\tau-\xi)\Delta} \bigg( \bigg( \int_0^\xi e^{i(\xi-\xi_1)\Delta} \Big( (e^{i\xi_1\Delta}v)V \Big) d\xi_1 \bigg) 2(e^{i\xi\Delta}v)(e^{-i\xi\Delta}\bar{v}) \bigg) d\xi,$$
  
$$-\int_0^\tau e^{i(\tau-\xi)\Delta} \bigg( \bigg( \int_0^\xi e^{i(\xi-\xi_1)\Delta} \Big( (e^{i\xi_1\Delta}v^2)(e^{-i\xi_1\Delta}\bar{v}) \Big) d\xi_1 \bigg) V \bigg) d\xi,$$

$$-\int_0^\tau e^{i(\tau-\xi)\Delta} \bigg( \bigg(\int_0^\xi e^{i(\xi-\xi_1)\Delta} \Big( (e^{i\xi_1\Delta}v)V \Big) d\xi_1 \bigg)V \bigg) d\xi,$$

namely the remaining integrals in the third line of (3.74). These computation follow exactly the same lines as for the computation of  $\Pi^1_A(\mathcal{I}_o(T_3))$ . Namely, given our regularity assumptions we can apply the Taylor based expansion (3.44), to obtain,

$$\begin{split} \left(\Pi^{1}_{A}(\mathcal{I}_{o}(T_{4})) + \Pi^{1}_{A}(\mathcal{I}_{o}(T_{5}))\right)(\hat{\mathbf{v}},\tau) &= -2\int_{0}^{\tau}\xi\prod_{i\in\{1,2,3,4\}}u^{a}_{i}d\xi - \int_{0}^{\tau}\xi\prod_{i\in\{1,2,3\}}u^{b}_{i}\right) \\ &= -\tau^{2}|v|^{2}vV + -\frac{\tau^{2}}{2}Vv^{2}\bar{v}, \\ &= -\frac{3\tau^{2}}{2}|v|^{2}vV, \end{split}$$

and

$$\Pi^1_A(\mathcal{I}_o(T_6))(\hat{\mathbf{v}},\tau) = -\int_0^\tau \xi \prod_{i \in \{1,2,3\}} u_i^c = -\frac{\tau^2}{2} V^2 v,$$

where  $(u_1^a, u_2^a, u_3^a, u_4^a) = (v, V, v, \bar{v}), (u_1^b, u_2^b, u_3^b,) = (V, v^2, \bar{v}), \text{ and } (u_1^c, u_2^c, u_3^c = (V, v, V).$ 

6. Computation of  $\Pi^1_A(\mathcal{I}_o(T_7))(\hat{\mathbf{v}},\tau)$ . We have that,

$$\mathcal{I}_o(T_7) = \bigcup_{i=1}^{\circ}$$

encodes the iterated integral

$$\int_0^\tau e^{i(\tau-\xi)\Delta} \left( \left( \int_0^\xi e^{-i(\xi-\xi_1)\Delta} \left( (e^{-i\xi_1\Delta} \bar{v}^2)(e^{i\xi_1\Delta} v) \right) d\xi_1 \right) (e^{i\xi\Delta} v^2) \right) d\xi$$

Similar to the computations done in (3.77) it follows that,

$$(\Pi_{A}^{1}\mathcal{I}_{o}(T_{7}))(\hat{\mathbf{v}},\tau) = \mathcal{K}_{o,A}^{1}(\tilde{\Pi}_{o,A}\mathcal{D}_{0}(\lambda_{0}\mathcal{I}_{\bar{o}}(\lambda_{0}))(\hat{\mathbf{v}},\cdot))(\tau)$$

$$= \mathcal{K}_{o,A}^{1}(\tilde{\Pi}_{o,A}(\lambda_{0}\mathcal{I}_{\bar{o}}^{0}(\lambda_{0}))(\hat{\mathbf{v}},\cdot))(\tau)$$

$$= \mathcal{K}_{o,A}^{1}\left(\frac{\Upsilon_{\text{root}}^{\Psi_{o}}[T_{7}]}{S_{\text{root}}(T_{7})}(\mathbf{v},\cdot)(\Pi_{A}\mathcal{I}_{\bar{o}}^{0}(\lambda_{0}))(\hat{\mathbf{v}},\cdot)\right)(\tau).$$

$$(3.80)$$

We are left to calculate  $\Upsilon_{\text{root}}^{\Psi_o}[T_7]$ ,  $S_{\text{root}}(T_7)$ , and  $(\Pi_A \mathcal{I}_{\bar{o}}^0(\lambda_0)))(\hat{\mathbf{v}}, \xi)$ . First, by definition (3.40) of  $S_{\text{root}}(T_7)$  we have that i = 1 = j and  $\beta_{1,1} = 1$ , where  $o_1 = \bar{o}$ . Hence, we have that  $S_{\text{root}}(T_7) = 1$ . Secondly, by definition (3.41) of  $\Upsilon_{\text{root}}^{\Psi_o}[T]$  we have that,

$$\begin{split} \Upsilon^{\Psi_o}_{\text{root}}[T_7] &= (D_{\bar{o}} \hat{\Psi}^0_o)(\mathbf{v}, \xi) \\ &= -i D_{\bar{o}} \Big( (e^{i\xi\Delta} (v_o^2)) (e^{-i\xi\Delta} v_{\bar{o}}) \Big) \\ &= -i (e^{i\xi\Delta} v^2) \end{split}$$

since  $v_o = v$ , and  $v_{\bar{o}} = \bar{v}$ . Thirdly, by following definition (3.3.2) we have,

~

$$(\Pi_A \mathcal{I}^0_{\bar{o}} \lambda_0)(\hat{\mathbf{v}}, \xi) = \mathcal{K}^0_{\bar{o}, A}((\tilde{\Pi}_{\bar{o}, A} \lambda_0)(\hat{\mathbf{v}}, \cdot))(\xi) = \mathcal{K}^0_{\bar{o}, A}\left(\frac{\Upsilon^{\Psi_{\bar{o}}}_{\mathrm{root}}[\lambda_0]}{S_{\mathrm{root}}(\lambda_0)}(\hat{\mathbf{v}}, \cdot)\right)(\xi)$$

where by equation (3.40) we have that  $S_{\text{root}}(\lambda_0) = 1$  and by equation (3.41) it follows that,

$$\begin{split} & \Upsilon^{\Psi_{\bar{o}}}_{\text{root}}[\lambda_0](\mathbf{v},\zeta) = \prod_{\boldsymbol{\ell} \in \mathfrak{L}^{0,\bar{o}}_+} e^{\zeta \mathcal{L}_{\boldsymbol{\ell}}} f^0_{\bar{o},\boldsymbol{\ell}}(v_{\boldsymbol{\ell}}) = (e^{i\zeta \Delta} f^0_{\bar{o},o}(v))(e^{-i\zeta \Delta} f^0_{\bar{o},\bar{o}}(\bar{v})) \\ &= (e^{i\zeta \Delta} v)(e^{-i\zeta \Delta}(i\bar{v}^2)). \end{split}$$

Collecting the above computations, we have,

$$\left(\Pi_A \mathcal{I}^0_{\bar{o}}(\lambda_0)\right)(\hat{\mathbf{v}},\xi) = \mathcal{K}^0_{\bar{o},A}(F(\hat{\mathbf{v}},\cdot))(\xi)$$

where

$$F(\hat{\mathbf{v}},\xi) = \prod_{i \in \{1,2\}} e^{\xi \mathcal{L}_i} u_i, \quad u_1 = v, \quad u_2 = i\bar{v}^2, \quad \mathcal{L}_1 = i\Delta, \ \mathcal{L}_2 = -i\Delta,$$

and where we consider the operator  $\mathcal{L}_{\bar{o}} = -i\Delta$  associated to the decoration  $\bar{o} \in \mathfrak{L}_+$ . Given that  $v \in D(\Delta)$ , by following (3.44) we can once again Taylor expand the operators inside the integral which yields the following approximation,

$$\left(\Pi_A \mathcal{I}^0_{\bar{o}}(\lambda_0)\right)(\hat{\mathbf{v}},\xi) = \mathcal{K}^0_{\bar{o},A}(F(\hat{\mathbf{v}},\cdot))(\xi) = \int_0^\xi \prod_{i \in \{1,2\}} u_i d\xi = i\xi v \bar{v}^2.$$

Hence, by collecting the above computations it follows from (3.80) that,

$$\left(\Pi^1_A \mathcal{I}_o(T_7)\right)(\hat{\mathbf{v}},\tau) = \mathcal{K}^1_{o,A}\left(\xi(e^{i\xi\Delta}v^2)v\bar{v}^2\right)(\tau) = \mathcal{K}^1_{o,A}\left(\tilde{F}(\hat{\mathbf{v}},\cdot)\right)(\tau),$$

where

$$\tilde{F}(\hat{\mathbf{v}},\xi) = \xi^{\ell} \prod_{i \in \{1,2,3\}} e^{\xi \mathcal{L}_{i}} u_{i}, \qquad \mathcal{L}_{1} = i\Delta, \ \mathcal{L}_{2} = \mathcal{L}_{3} = 0,$$
$$u_{1} = v^{2}, \ u_{2} = v, \ u_{3} = \bar{v}^{2} \text{ and } \ell = 1$$

Given that  $v \in D(\Delta)$ , by following (3.44) we can once again Taylor expand the remaining operators inside the integral which yields the following approximation,

$$\left(\Pi_{A}^{1}\mathcal{I}_{o}(T_{7})\right)(\hat{\mathbf{v}},\tau) = \mathcal{K}_{o,A}^{1}\left(\tilde{F}(\hat{\mathbf{v}},\xi)\right)(\tau) = \int_{0}^{\tau} \xi \prod_{i \in \{1,2,3\}} u_{i}d\xi = \frac{\tau^{2}}{2}v|v|^{4},$$

7. Computation of  $\Pi^1_A(\mathcal{I}_o(T_8))(\hat{\mathbf{v}}, \tau)$ . We have that,

$$\mathcal{I}_o(T_8) = \bigcup_{i=1}^{n}$$

encodes the iterated integral

$$\int_0^\tau e^{i(\tau-\xi)\Delta} \left( \left( \int_0^\xi e^{-i(\xi-\xi_1)\Delta} \left( (e^{-i\xi_1\Delta}\bar{v})\bar{V} \right) d\xi_1 \right) (e^{i\xi\Delta}v^2) \right) d\xi$$

By using the same arguments as made in the above computation of the iterated integral  $\mathcal{I}_o(T_7)$ , namely by following the Taylor expansion (3.44), we obtain

$$\Pi^1_A(\mathcal{I}_o(T_8))(\hat{\mathbf{v}},\tau) = \int_0^\tau \xi \prod_{i \in \{1,2,3\}} u_i d\xi = \frac{\tau^2}{2} v |v|^2 \bar{V},$$

where  $u_1 = \bar{v}$ ,  $u_2 = \bar{V}$ , and  $u_3 = v^2$ . Collecting the above computation together with equation (3.76) yields the second order low-regularity scheme given in (3.75).

**Local error analysis**. The error for the trees  $\mathcal{I}_o^1(\lambda_0)$  and  $\mathcal{I}_o^1(\lambda_1)$  follows the same computations as for the first order. One has to go one step further in the Taylor approximation:

$$R^{1}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}},\cdot)) = \mathcal{C}^{2}[\mathcal{M}_{\{1,2\}},i\Delta](-iv^{2},\bar{v})$$
  
$$R^{1}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{1})(\hat{\mathbf{v}},\cdot)) = \mathcal{C}^{2}[\mathcal{M}_{\{1,2\}},i\Delta](-iv,V).$$

In the end, we obtain the following contribution to the local error. This produce the following second order commutators:

$$\mathcal{L}^1_{\text{low}}(\mathcal{I}_o(\lambda_0), \hat{\mathbf{v}}, A) = -iv^2 \bar{v} + \mathcal{C}^2[\mathcal{M}_{\{1,2\}}, i\Delta](-iv^2, \bar{v})$$

$$\mathcal{L}^{1}_{\text{low}}(\mathcal{I}_{o}(\lambda_{1}), \hat{\mathbf{v}}, A) = -ivV + \mathcal{C}^{2}[\mathcal{M}_{\{1,2\}}, i\Delta](-iv, V),$$

where using the Leibniz rule it follows from Definition 3.1.2 that these second order commutators ask for two spacial derivatives on v and V.

Next, the decorated tree  $\mathcal{I}_o(\lambda_0^1)$  contains a polynomial decoration meaning that one has  $\xi$  inside the integral. Therefore, applying Definition 3.3.8, one gets

$$\mathcal{L}^{1}_{\text{low}}(\mathcal{I}_{o}(\lambda_{0}^{1}), \hat{\mathbf{v}}, A) = \mathcal{L}^{0,o}_{\text{low}}(\lambda_{0}^{1}, \hat{\mathbf{v}}, A) + R^{r}_{o,A}\Big(\Big(\tilde{\Pi}_{o,A}\lambda_{0}^{1}\Big)(\hat{\mathbf{v}}, \cdot\Big)\Big),$$

where

$$\mathcal{L}_{low}^{r,o}(\lambda_0^1, A) = \frac{\Upsilon_{root}^{\Psi_o}[\lambda_0^1]}{S_{root}(\lambda_0^1)} (\mathbf{v}, 0) V_0 = \mathcal{C}[u^2, i\Delta](-iv)\bar{v}$$
$$R_{o,A}^1 \Big( \Big( \tilde{\Pi}_{o,A} \lambda_0^1 \Big) (\hat{\mathbf{v}}, \cdot \Big) \Big) = \mathcal{C}[\mathcal{C}[\mathcal{M}_{\{1,2\}}, \Delta](\mathcal{M}_{\{1\}}, \mathcal{M}_{\{1\}}) \mathcal{M}_{\{2\}}, i\Delta](-iv, \bar{v})$$

It remains to compute the local error for trees with two edges such as  $T_3 = \mathcal{I}_o(\lambda_0 \mathcal{I}_o(\lambda_0))$ . We proceed with a full derivation for this decorated trees. Two edges mean that we have to go for one extra level of recursivity. One has

$$\mathcal{L}_{\text{low}}^{1}(T_{3}, \hat{\mathbf{v}}, A) = \mathcal{L}_{\text{low}}^{0,o}(\lambda_{0}\mathcal{I}_{o}(\lambda_{0}), \hat{\mathbf{v}}, A) + R_{o,A}^{1}\left(\left(\tilde{\Pi}_{o,A}^{0}\lambda_{0}\mathcal{I}_{o}(\lambda_{0})\right)(\hat{\mathbf{v}}, \cdot)\right).$$
(3.81)

Then, for the first term, one gets

$$\mathcal{L}_{\text{low}}^{0,o}(\lambda_0 \mathcal{I}_o(\lambda_0), \hat{\mathbf{v}}, A) = \frac{\Upsilon_{\text{root}}^{\Psi_o}[\lambda_0 \mathcal{I}_o(\lambda_0)]}{S_{\text{root}}(\lambda_0 \mathcal{I}_o(\lambda_0))}(\mathbf{v}, 0) \, V_0 \, \mathcal{L}_{\text{low}}^0(\mathcal{I}_o(\lambda_0), \hat{\mathbf{v}}, A).$$

Given that  $A = (H^2 \cap H_0^1)(\Omega)^3$ , one has from the local error analysis of the first order scheme:

$$\mathcal{L}^{0}_{\text{low}}(\mathcal{I}_{o}(\lambda_{0}), \hat{\mathbf{v}}, A) = -iv^{2}\bar{v} + \Delta(v^{2}\bar{v}) + \Delta(v^{2})\bar{v} - v^{2}\Delta\bar{v}$$

Therefore, one gets

$$\mathcal{L}_{low}^{0,o}(\lambda_0 \mathcal{I}_o(\lambda_0), \hat{\mathbf{v}}, A) = -v^3 \bar{v}^2 - iv \bar{v} \Delta \left(v^2 \bar{v}\right) - iv \Delta (v^2) \bar{v}^2 + iv^3 \bar{v} \Delta \bar{v}.$$

For the second term of (3.81), we Taylor-expand the operators that give:

$$R^{1}_{o,A}\left(\left(\tilde{\Pi}^{0}_{o,A}\lambda_{0}\mathcal{I}_{o}(\lambda_{0})\right)(\hat{\mathbf{v}},\cdot)\right) = -i\Delta\left(v^{3}\bar{v}^{2}\right) - i(\Delta v)v^{2}\bar{v}^{2} + i(\Delta\bar{v})v^{3}\bar{v}.$$

In fact for the approximation of  $\mathcal{I}_o(\lambda_0 \mathcal{I}_o(\lambda_0))$ , one uses the classical exponential integrators by Taylor-expanding all the operators. We do not ask too much regularity due to the size of the trees which involves two integrations in time. For the other trees  $\mathcal{I}_o(T_i)$ ,  $i \in \{4, 5, 6, 7, 8\}$ , one can proceed with the same computations and get an error of the same order.  $\Box$ 

**Remark 3.5.5 (Error improvement)** Note that classical approximation techniques, such as splitting or exponential integrator methods (see, e.g., [CG12, JL00, Lub08]) introduce a local error of type  $\mathcal{O}(\Delta^2 v, \Delta^2 V)$  and hence require solution and potential in

$$D(\Delta^{2}) = \{ f \in (H^{2} \cap H^{1}_{0})(\Omega) ; \Delta f \in (H^{2} \cap H^{1}_{0})(\Omega) \}.$$

The local error of the second order low regularity GP integrator (3.75) on the other hand only requires the boundedness of second instead of fourth order spatial derivatives of the potential V and solution u, as well as only the classical boundary conditions that the trace vanishes, i.e.,  $H_0^1(\Omega)$ .

Note that the second order low regularity scheme (3.75) involves commutators of the form  $C[f, i\Delta](u^n, \overline{u^n}, V)$  which in general involve spatial derivatives. Indeed, using Definition 3.1.2, one has

$$\mathcal{C}[f, i\Delta](v, \bar{v}, V) = -i\Delta(f(v, \bar{v}, V)) + D_1 f(v, \bar{v}, V) \cdot i\Delta v$$

$$+ D_2 f(v, \bar{v}, V) \cdot i\Delta \bar{v} + D_3 f(v, \bar{v}, V) \cdot i\Delta V.$$
(3.82)

For practical computations we need to address the stability issues caused by the inclusion of these commutator terms in the numerical schemes. Different approaches can be made to treat this issue and guarantee the stability of the scheme. One way to overcome the stability issue lies in imposing a CFL condition on the fully discrete scheme, introducing a restriction on the ratio between time step size  $\tau$  and spatial step size  $\Delta x$ . In this paper we will not impose any CFL type condition, but stabilize the scheme a posteriori by the use of properly chosen filter operators  $\Psi$ . More precisely, instead of considering the second order low regularity integrator (3.75) for the discretisation of the GP equation, we consider its stabilised counterpart

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau\left((e^{i\tau\Delta}(u^n)^2)(e^{i\tau\Delta}\varphi_1(-2i\tau\Delta)\overline{u}^n) + (e^{i\tau\Delta}u^n)(e^{i\tau\Delta}\varphi_1(-i\tau\Delta)V)\right)$$

$$- i\tau^2\Psi\mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-2i\tau\Delta) - \varphi_2(-2i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta]((u^n)^2, \overline{u}^n) - i\tau^2\Psi\mathcal{C}[(e^{i\tau\Delta}\mathcal{M}_{\{1\}})(e^{i\tau\Delta}(\varphi_1(-i\tau\Delta) - \varphi_2(-i\tau\Delta))\mathcal{M}_{\{2\}}), i\Delta](u^n, V) - i\tau^2\left(e^{i\tau\Delta}\Psi\mathcal{C}[u^2, i\Delta](u^n)\right)\left(e^{i\tau\Delta}\varphi_2(-2i\tau\Delta)\overline{u}^n\right)$$

$$- \frac{\tau^2}{2}(u^n|u^n|^4 + 3u^n|u^n|^2V - |u^n|^2u^n\overline{V} + u^nV^2) = \Phi^\tau_{\text{GP2STAB}}(u^n),$$
(3.83)

where we have inserted the filter operator  $\Psi$  in front of all commutator terms. This filter function allows us the stabilization of the scheme. In the following section we briefly discuss the appropriate choice of filter function (which allows us to stabilize the scheme without worsening the local error structure) and outline an example. For a general introduction to filter functions in case of ODEs we refer to [HLW10].

#### 3.5.2 Filter functions, commutators and stabilisation

To illustrate the appropriate choice for the filter operator  $\Psi$  we place ourselves in the following general framework: to construct a stable scheme at order p = r + 1, we wish to stabilize a term of the form

$$\tau^p \mathcal{C}^{p-1}[f, i\Delta](v, \bar{v}, V).$$

In case of the second order scheme for (3.60), the commutator terms are of first order and hence involve first order derivatives (with according boundary conditions), see (3.82), (3.73) and the Remark 3.1.3. This motivates the following choice we make on the filter operator: we denote the filter operator by

$$\Psi = \psi(i\tau |\nabla|),$$

where

 $|\nabla| := (-\Delta)^{1/2}$ 

is self-adjoint on the domain  $D(|\nabla|) = H_0^1(\Omega)$  and  $\psi$  denotes a suitably chosen filter function which allows us to

(a) stabilize the scheme such that the numerical flow is locally Lipschitz continuous (see Assumption 3 below), while still(b) allowing approximations at low regularity (see Assumption 4 below).

Here it is important to note that the main purpose of our new framework lies in reducing the regularity assumptions. Hence, we do not want to introduce a "bad" (classical) error term through the stabilization, otherwise we could use directly classical Taylor series expansion techniques in the construction of our schemes. Thus, we require the filter function  $\psi$  to introduce the same optimal local error  $O(\tau^{p+1}|\nabla|^p(v+V))$  as is introduced by our low-regularity schemes (see Section 3.4.2).

The two necessary assumptions on the filter operator  $\Psi = \psi(i\tau |\nabla|)$  at arbitrary order p reads as follows:

**Assumption 3** Let us denote by X the space endowed with norm  $\|\cdot\|$  in which we carry out the error analysis. Then the filter function  $\Psi$  should satisfy the following estimate

$$\left\|\tau^{p-1}\Psi[\mathcal{C}^{p-1}[f,i\Delta](v,\bar{v},V)]\right\| \le C \|v\|^{c}$$

for some  $\sigma = \sigma(f, p) > 0$ .

Assumption 3 guarantees the stability of the scheme, such that (for the appropriate choice of X) we have local Lipschitz continuity of the numerical flow  $\Phi^{\tau}$ .

**Assumption 4** The filter operator  $\Psi = \psi(i\tau | \nabla |)$  shall satisfy the following low regularity expansion

$$\Psi[\mathcal{C}^{p-1}[f, i\Delta](v, \bar{v}, V)] = \mathcal{C}^{p-1}[f, i\Delta](v, \bar{v}, V) + \mathcal{O}(\tau |\nabla|^p (v + V)).$$
(3.84)

Assumption 4 preserves the optimal local error structure  $O(\tau^{p+1}|\nabla|^p(v+V))$  of our schemes. This is essential as the inclusion of the filter function should not require more regularity than we assumed in the construction of our low regularity framework.

An example of a possible choice of a filter function in order to stabilize the second order scheme (3.75) is given in the following example.

**Example 13 (Filter functions for the GP equation)** A possible choice of filter function for the second order low regularity integrator (3.83) is given by

$$\Psi = \psi(i\tau|\nabla|) = \varphi_1(i\tau|\nabla|) = (i\tau|\nabla|)^{-1}(e^{i\tau|\nabla|} - 1),$$
(3.85)

where we recall that  $|\nabla| = (-\Delta)^{1/2}$  is a self-adjoint operator on the  $D(|\nabla|) = H_0^1(\Omega)$ . For a detailed analysis made on the torus which justifies the above choice of filter operator to stabilize a low-regularity scheme involving a first order commutator see [AB23a].

For a higher p-th order method we will encounter higher order commutators of the form

$$\tau^p \mathcal{C}^{p-1}[f, i\Delta](u^n, \overline{u^n}, V)$$

in our schemes. In order to stabilize the latter we need to introduce a higher order filter operator  $\Psi_p$ . For this purpose, we first note that using the Leibnitz formula, the commutator term  $\mathcal{C}^{p-1}[f, i\Delta](v, \bar{v}, V)$  can be written as a combination of terms involving at most p-1 derivatives of v and V.

As an example, in the case p = 2,  $C[f, i\Delta](v, \bar{v}, V)$  has the explicit form (3.82). Hence the role of  $\Psi_p$  lies in stabilizing terms which involve the differential operator  $|\nabla|^{p-1}$ . This motivates the higher order filter operator

$$\Psi_p = \underbrace{\varphi_1(i\tau|\nabla|)...\varphi_1(i\tau|\nabla|)}_{p-1 \text{ times}}.$$
(3.86)

Note that at first glance another (cheaper) choice of high order filter function would be  $\tilde{\Psi}(i\tau|\nabla|) := \varphi_1(i\tau|\nabla|^{p-1})$ . The latter, however, introduces a local error of order  $\mathcal{O}(\tau|\nabla|^{2(p-1)})$  which at order  $p \geq 3$  is not optimal in the sense of our low regularity approximations. Namely, this (cheaper) choice of filter function does not satisfy Assumption 4.

#### 3.5.3 Klein– and Sine–Gordon equations

As a second example we consider a problem with real non-polynomial nonlinearity: the nonlinear Klein–Gordon (KG) equation

$$\partial_{tt}z - \Delta z + m^2 z = g(z), \qquad (t, x) \in \mathbf{R} \times \Omega, z(0) = u_0, \quad \partial_t z(0) = u_1,$$
(3.87)

where for simplicity we assume non zero mass  $m \neq 0$  and real-valued solutions  $z(t, x) \in \mathbf{R}$ . Nevertheless, our framework can also allow for the complex setting  $z(t, x) \in \mathbf{C}$  and for the case m = 0 for wave equations.

First, in order to apply our abstract framework (3.1) to the nonlinear Klein-Gordon model (3.87) we rewrite (3.87) as a first-order complex system. For this purpose we define

$$\langle \nabla \rangle_m = \sqrt{-\Delta + m^2}$$

and assume that our domain  $\Omega$  (equipped with appropriate boundary conditions) is chosen such that  $\langle \nabla \rangle_m$  is well defined and invertible. For instance in the case of periodic boundary conditions the operator  $\langle \nabla \rangle_m$  can be expressed as the Fourier multiplier

$$(\langle \nabla \rangle_m)_{k \in \mathbb{Z}^d} = \sqrt{(k_1 + \ldots + k_d)^2 + m^2},$$

and is a well defined invertible operator on the space  $H^1(\mathbf{T}^d)$ , given that  $m \neq 0$ . While on a sufficiently smooth bounded domain  $\Omega \subset \mathbf{R}^d$  equipped with homogeneous Dirichlet boundary conditions, using pseudo-spectral methods we have that the operator  $\langle \nabla \rangle_m : (H^2 \cap H^1_0)(\Omega) \to L^2(\Omega)$  is essentially self-adjoint; its closure is self-adjoint on the domain

$$D(\langle \nabla \rangle_m) = H^1_0(\Omega).$$

See [RS80] for a full discussion on the study of unbounded operators, spectral theorems for self-adjoint operators (pg. 263) and of the notion of closable operators (pg. 250). Next we introduce the transformation

$$u = z - i \langle \nabla \rangle_m^{-1} \partial_t z, \tag{3.88}$$

which allows one to rewrite the Klein-Gordon equation (3.87) as the following first order complex system

$$i\partial_t u = -\langle \nabla \rangle_m u + \langle \nabla \rangle_m^{-1} g\big( (\frac{1}{2}(u+\overline{u})), \tag{3.89}\big)$$

and where by (3.88) we have that

$$z = \frac{1}{2}(u + \overline{u}) = \operatorname{Re}(u).$$

A natural space to study wave type equations of the form (3.87) is for  $(z, \partial_t z) \in H^1 \times L^2$ . In terms of u, from the transformation (3.88), we see that a natural space to measure the error in when studying equation (3.89) is  $H^1$ .

In order to illustrate our general theory to the case of a non-polynomial nonlinearity, we choose to study the Sine–Gordon equation. The Sine–Gordon equation corresponds to the following choice of nonlinearity

$$g(z) = -\sin z,$$

which by equation (3.89) yields the first-order complex Sine–Gordon equation:

$$\partial_t u - i \langle \nabla \rangle_m u = i \langle \nabla \rangle_m^{-1} \Big( \sin(\frac{1}{2}u) \cos(\frac{1}{2}\bar{u}) + \cos(\frac{1}{2}u) \sin(\frac{1}{2}\bar{u}) \Big).$$
(3.90)

We now apply our general framework to the above Sine–Gordon equation. The above equation (3.90) is of the form (3.1) with

$$\begin{aligned} \mathfrak{L}_{-} &= \{0,1\}, \quad \mathfrak{L}_{+} = \mathfrak{L}_{+}^{o,0} = \mathfrak{L}_{+}^{o,1} = \mathfrak{L}_{+}^{\bar{o},0} = \mathfrak{L}_{+}^{\bar{o},1} = \{o,\bar{o}\}, \quad V_{0} = V_{1} = 1\\ \mathcal{L}_{o} &= i \langle \nabla \rangle_{m}, \quad \mathcal{L}_{\bar{o}} = -\mathcal{L}_{o}, \quad u_{o} = v, \quad u_{\bar{o}} = \bar{v}\\ f_{o,o}^{0}(u) &= \sin(\frac{1}{2}u), \quad f_{o,\bar{o}}^{0}(\bar{u}) = \cos(\frac{1}{2}\bar{u}), \quad f_{o,o}^{1}(u) = \cos(\frac{1}{2}u), \quad f_{o,\bar{o}}^{1}(\bar{u}) = \sin(\frac{1}{2}\bar{u}),\\ f_{\bar{o},o}^{0}(u) &= \cos(\frac{1}{2}u), \quad f_{\bar{o},\bar{o}}^{0}(u) = \sin(\frac{1}{2}\bar{u}), \quad f_{\bar{o},o}^{1}(u) = \sin(\frac{1}{2}u), \quad f_{\bar{o},\bar{o}}^{1}(\bar{u}) = \cos(\frac{1}{2}\bar{u}). \end{aligned}$$

Using the above we have that for  $(\ell, \mathfrak{l}) \in \{o, \bar{o}\} \times \{0, 1\}$  the nonlinearity is given by

$$\Psi_{\ell}^{\mathfrak{l}}(\mathbf{u}_{\ell}^{\mathfrak{l}}) = i \langle \nabla \rangle_{m}^{-1} \Big( f_{\ell,o}^{\mathfrak{l}}(u) f_{\ell,\bar{o}}^{\mathfrak{l}}(\bar{u}) \Big).$$

$$(3.91)$$

Before introducing the first order analysis for (3.90), we make the following comparative remark. The decorated trees which will be introduced in the following section for the first order analysis of (3.90) will resemble the ones introduced for the analysis of the Gross–Pitaevskii equation (see Section 3.5.1). The main difference is that the nodes will encode different nonlinearities, given that the nonlinearities associated to  $V_0$  and  $V_1$  for the Sine-Gordon equation are different from the ones for the Gross-Pitaevskii equation. In addition, the Sine-Gordon nonlinearity (3.91) includes the linear operator  $\mathcal{B}_o^0 = \mathcal{B}_o^1 = i \langle \nabla \rangle_m^{-1}$ . See Section 3.5.1 where for the study of the Gross-Pitaevskii equation we had the more simple case where  $\mathcal{B}_o^0 = \mathcal{B}_o^1 = \mathrm{id}$ .

#### First order Duhamel integrator for Sine–Gordon

We seek to provide a first order low-regularity approximation to the following oscillatory integral:

$$\begin{split} w_o^0(\hat{\mathbf{v}},\tau) &= e^{i\tau\langle\nabla\rangle_m} v + i\langle\nabla\rangle_m^{-1} \int_0^\tau e^{i(\tau-\xi)\langle\nabla\rangle_m} \\ &\left((e^{i\xi\langle\nabla\rangle_m}\sin(\frac{1}{2}v))(e^{-i\xi\langle\nabla\rangle_m}\cos(\frac{1}{2}\bar{v})) + (e^{i\xi\langle\nabla\rangle_m}\cos(\frac{1}{2}v))(e^{-i\xi\langle\nabla\rangle_m}\sin(\frac{1}{2}\bar{v}))\right) d\xi \end{split}$$

The first order low regularity integrator for the Sine–Gordon equation is given in the following corollary.

**Corollary 3.5.6** At first order our general low regularity scheme (3.57) for the Sine-Gordon equation (3.90) takes the form

$$u^{n+1} = e^{i\tau \langle \nabla \rangle_m} u^n + i\tau \langle \nabla \rangle_m^{-1} [$$

$$\left( e^{i\tau \langle \nabla \rangle_m} \sin(\frac{1}{2}u^n) \right) \left( e^{i\tau \langle \nabla \rangle_m} \left( \varphi_1 \left( -2i\tau \langle \nabla \rangle_m \right) \cos(\frac{1}{2}\bar{u}^n) \right) \right)$$

$$+ \left( e^{i\tau \langle \nabla \rangle_m} \cos(\frac{1}{2}u^n) \right) \left( e^{i\tau \langle \nabla \rangle_m} \left( \varphi_1 \left( -2i\tau \langle \nabla \rangle_m \right) \sin(\frac{1}{2}\bar{u}^n) \right) \right) \right]$$

$$(3.92)$$

with local error

$$\mathcal{O}\big(\tau^2 \mathcal{C}[\langle \nabla \rangle_m^{-1} \mathcal{M}_{\{1,2\}}, i \langle \nabla \rangle_m](u, \bar{u})\big)$$

In case of more regular solutions, we have the following simplified first-order scheme,

$$u^{n+1} = e^{i\tau\langle\nabla\rangle_m} u^n + i\tau\langle\nabla\rangle_m^{-1} \Big(\sin(\frac{1}{2}u^n)\cos(\frac{1}{2}\bar{u}^n) + \cos(\frac{1}{2}u^n)\sin(\frac{1}{2}\bar{u}^n)\Big)$$
(3.93)

with a local error of order  $\mathcal{O}\Big(\tau^2 \langle \nabla \rangle_m^{-1}((\langle \nabla \rangle_m u)\bar{u})\Big).$ 

#### Remark 3.5.7 (Remark on the imposed regularity) In general the local error of type

$$\mathcal{O}\big(\tau^2 \mathcal{C}[\langle \nabla \rangle_m^{-1} \mathcal{M}_{\{1,2\}}, i \langle \nabla \rangle_m](u, \overline{u})\big)$$

asks for less regularity assumptions than the local error  $\mathcal{O}(\tau^2 \langle \nabla \rangle_m^{-1}(\langle \nabla \rangle_m u)\bar{u}))$ , due to the structure of the commutator (see Definition 3.1.2). Indeed, we portray this in the following example, where we measure the error in the space  $H^1(\mathbf{T}^d)$ . First, using the commutator estimate established in [RS21, Lemma 27], we have

$$\begin{aligned} \|\mathcal{C}[\langle \nabla \rangle_m^{-1} \mathcal{M}_{\{1,2\}}, i \langle \nabla \rangle_m](u, \overline{u})\|_{H^1} &\lesssim \|\mathcal{C}[\mathcal{M}_{\{1,2\}}, i \langle \nabla \rangle_m](u, \overline{u})\|_{L^2} \\ &\lesssim \||u\|_{H^{\frac{1}{2} + \frac{d}{4}}}^2. \end{aligned}$$

Namely we have that the commutator term asks for solutions in  $H^{\frac{1}{2}+\frac{d}{4}}(\mathbf{T}^d)$ .

Secondly, using the classical bilinear estimate:

$$||uv||_{L^{2}(\Omega)} \lesssim ||u||_{L^{2}(\Omega)} ||v||_{H^{\frac{d}{2}+\epsilon}(\Omega)}, \ \epsilon > 0.$$

which follows by the Sobolev injection  $H^{\frac{d}{2}+\epsilon} \hookrightarrow L^{\infty}$ , we have,

$$\begin{aligned} ||\langle \nabla \rangle_m^{-1}((\langle \nabla \rangle_m u)\bar{u})||_{H^1(\Omega)} &\lesssim ||\langle \nabla \rangle_m u)\bar{u}||_{L^2(\Omega)} \\ &\lesssim ||\langle \nabla \rangle_m u||_{L^2(\Omega)}||u||_{H^{\frac{d}{2}+\epsilon}(\Omega)} \end{aligned}$$

It follows from the above that the *classical* local error term asks for solutions in  $(H^1 \cap H^{\frac{d}{2}+\epsilon})(\mathbf{T}^d)$  when working on the torus, while when working on a bounded domain equipped with Dirichlet boundary conditions it asks for solutions in  $(H_0^1 \cap H^{\frac{d}{2}+\epsilon})(\Omega)$ . Hence, the commutator-term asks for less regularity than the classical local error term. For example, when d = 1, the low-regularity scheme asks for  $H^{\frac{3}{4}}(\mathbf{T})$ -solutions, whereas the classical scheme asks for  $u(t) \in D(\langle \nabla \rangle_m) = H^1(\mathbf{T})$ .

*Proof.* The trees we are considering are  $\mathcal{T}^1 = \{T_0, T_1\}$ , where following the framework given in (3.32) we have that,

$$\mathcal{B}_{o}\left(\prod_{\iota\in\mathfrak{L}_{+}^{0,o}}e^{\xi\mathcal{L}_{l}}f_{o,\iota}^{0}(v_{\iota})\right)V_{0}=i\langle\nabla\rangle_{m}^{-1}\left(\left(e^{i\xi\langle\nabla\rangle_{m}}u_{1}\right)\left(e^{-i\xi\langle\nabla\rangle_{m}}u_{2}\right)\right)$$
$$\equiv \circ=\lambda_{0}=T_{0},$$
$$\mathcal{B}_{o}\left(\prod_{\iota\in\mathfrak{L}_{+}^{1,o}}e^{\xi\mathcal{L}_{l}}f_{o,\iota}^{1}(v_{\iota})\right)V_{1}=i\langle\nabla\rangle_{m}^{-1}\left(\left(e^{i\xi\langle\nabla\rangle_{m}}u_{3}\right)\left(e^{-i\xi\langle\nabla\rangle_{m}}u_{4}\right)\right)$$
$$\equiv \bullet=\lambda_{1}=T_{1},$$

where  $\mathbf{v} = (v, \overline{v})$  and  $u_1 = \sin(\frac{1}{2}v)$ ,  $u_2 = \cos(\frac{1}{2}\overline{v})$ ,  $u_3 = \cos(\frac{1}{2}v)$ ,  $u_4 = \sin(\frac{1}{2}\overline{v})$ . From equation (3.57) it then follows that the first-order scheme is of the form,

$$u^{0}(\hat{\mathbf{v}},\tau) = e^{i\tau\Delta}v + \sum_{T\in\mathcal{T}^{1}} \Pi^{0}_{A}(\mathcal{I}_{o}(T))(\hat{\mathbf{v}},\tau), \qquad (3.94)$$

where  $\mathbf{v} = (v, \bar{v})$ . We have, for i = 0, 1, that

$$\Pi^0_A \mathcal{I}_o(\lambda_i)(\hat{\mathbf{v}},\tau) = \mathcal{K}^0_{o,A}(\tilde{\Pi}_{o,A}\mathcal{D}_{-1}(\lambda_i)(\hat{\mathbf{v}},\cdot))(\tau) = \mathcal{K}^0_{o,A}\left(\frac{\Upsilon^{\Psi_o}_{\text{root}}[\lambda_i]}{S_{\text{root}}(\lambda_i)}(\mathbf{v},\xi)\right)(\tau)$$

where one has,

$$\begin{split} S_{\text{root}}(\lambda_0) &= S_{\text{root}}(\lambda_1) = 1, \\ \Upsilon^{\Psi_o}_{\text{root}}[\lambda_0](\mathbf{v},\xi) &= i \langle \nabla \rangle_m^{-1} \bigg( \left( e^{i\xi \langle \nabla \rangle_m} \sin(\frac{1}{2}v) \right) \left( e^{-i\xi \langle \nabla \rangle_m} \cos(\frac{1}{2}\overline{v}) \right) \bigg), \\ \Upsilon^{\Psi_o}_{\text{root}}[\lambda_1](\mathbf{v},\xi) &= i \langle \nabla \rangle_m^{-1} \bigg( \left( e^{i\xi \langle \nabla \rangle_m} \cos(\frac{1}{2}v) \right) \left( e^{-i\xi \langle \nabla \rangle_m} \sin(\frac{1}{2}\overline{v}) \right) \bigg). \end{split}$$

1. When  $A \supset (H^1 \cap H^{\frac{d}{2}+\epsilon})(\mathbf{T}^d)^2$  or  $A \supset (H^1_0 \cap H^{\frac{d}{2}+\epsilon})(\Omega)^2$ . Given that  $r, \ell = 0$ , and  $v \notin D(G)$ , we do not have enough regularity to simply apply the Taylor-expansion (3.44). Instead we perform a low-regularity based approximation of the integrals and apply the first point (i) of Definition 3.3.2. From definition (3.3.2) it then follows that

$$\mathcal{L}_1 = \mathcal{L}_o = i \langle \nabla \rangle_m, \ \mathcal{L}_2 = -i \langle \nabla \rangle_m, \ \mathcal{L}_{dom} = -2i \langle \nabla \rangle_m$$

with  $\mathcal{B}_o = \langle \nabla \rangle_m^{-1}$ , and hence we have,

$$\mathcal{K}^{0}_{o,A}\Big(\Upsilon^{\Psi_{o}}_{\mathrm{root}}[\lambda_{0}](\mathbf{v},\cdot)\Big)(\tau) = i\tau\langle\nabla\rangle_{m}^{-1}\Big(e^{i\tau\langle\nabla\rangle_{m}}u_{1}\Big)\Big(e^{i\tau\langle\nabla\rangle_{m}}\big(\varphi_{1}\big(-2i\tau\langle\nabla\rangle_{m}\big)u_{2}\big)\Big)$$
$$\mathcal{K}^{0}_{o,A}\Big(\Upsilon^{\Psi_{o}}_{\mathrm{root}}[\lambda_{1}](\mathbf{v},\cdot)\Big)(\tau) = i\tau\langle\nabla\rangle_{m}^{-1}\Big(e^{i\tau\langle\nabla\rangle_{m}}u_{3}\Big)\Big(e^{i\tau\langle\nabla\rangle_{m}}\big(\varphi_{1}\big(-2i\tau\langle\nabla\rangle_{m}\big)u_{4}\big)\Big).$$

Plugging the above in (3.94) yields the first order low-regularity scheme (3.92) for the Sine-Gordon equation.

2. When  $A = (H^1 \cap H^{\frac{d}{2}+\epsilon})(\mathbf{T}^d)^2$  or  $A = (H^1_0 \cap H^{\frac{d}{2}+\epsilon})(\Omega)^2$ In this case we apply the Taylor expansion (3.44) with  $r = \ell = 0$ , to obtain:

$$\begin{aligned} \mathcal{K}^{0}_{o,A}\Big((\Upsilon^{\Psi_{o}}_{\mathrm{root}}[\lambda_{0}] + \Upsilon^{\Psi_{o}}_{\mathrm{root}}[\lambda_{1}])(\mathbf{v},\cdot)\Big)(\tau) \\ &= i\langle\nabla\rangle_{m}^{-1}\bigg(\int_{0}^{\tau}u_{1}u_{2}d\xi + \int_{0}^{\tau}u_{3}u_{4}d\xi\bigg) \\ &= i\tau\langle\nabla\rangle_{m}^{-1}(\sin(\frac{1}{2}v)\cos(\frac{1}{2}\overline{v}) + \cos(\frac{1}{2}v)\sin(\frac{1}{2}\overline{v}), \end{aligned}$$

which thanks to (3.94) yields the first order scheme (3.93).

Local error analysis: We follow the same steps as for the proof of Corollary 3.5.1. Using the recursive formula in Definition 3.3.8, one gets for  $T \in \{\lambda_0, \lambda_1\}$ 

$$\mathcal{L}^{0}_{\text{low}}(\mathcal{I}_{o}(T), \hat{\mathbf{v}}, A) = \mathcal{L}^{-1,o}_{\text{low}}(T, \hat{\mathbf{v}}, A) + R^{0}_{o,A}\Big(\Big(\tilde{\Pi}_{o,A}T\Big)(\hat{\mathbf{v}}, \cdot\Big)\Big).$$

Then by (3.50), we get for  $l \in \{0, 1\}$ 

$$\mathcal{L}_{low}^{-1,o}(\lambda_{\mathfrak{l}}, \hat{\mathbf{v}}, A) = \frac{\Upsilon_{root}^{\Psi_o}[T]}{S_{root}(T)}(\mathbf{v}, 0)V_{\mathfrak{l}}$$

which gives

$$\mathcal{L}_{low}^{-1,o}(\lambda_0, \hat{\mathbf{v}}, A) = i \langle \nabla \rangle_m^{-1} \left( \left( e^{i\xi \langle \nabla \rangle_m} \sin(\frac{1}{2}v) \right) \left( e^{-i\xi \langle \nabla \rangle_m} \cos(\frac{1}{2}\overline{v}) \right) \right),$$
$$\mathcal{L}_{low}^{-1,o}(\lambda_1, \hat{\mathbf{v}}, A) = i \langle \nabla \rangle_m^{-1} \left( \left( e^{i\xi \langle \nabla \rangle_m} \cos(\frac{1}{2}v) \right) \left( e^{-i\xi \langle \nabla \rangle_m} \sin(\frac{1}{2}\overline{v}) \right) \right).$$

It remains to compute  $R^0_{o,A}((\Pi_A T))$  whose values will depend on A. For  $A \supset (H^1 \cap H^{\frac{d}{2}+\epsilon})(\mathbf{T}^d)^2$  or  $A \supset (H^1_0 \cap H^{\frac{d}{2}+\epsilon})(\Omega)^2$ one gets

$$R^{0}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}},\cdot)) = \mathcal{C}[i\langle\nabla\rangle_{m}^{-1}\mathcal{M}_{\{1,2\}},i\langle\nabla\rangle_{m}](\sin(\frac{1}{2}v),\cos(\frac{1}{2}\overline{v}))$$
  
$$R^{0}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{1})(\hat{\mathbf{v}},\cdot)) = \mathcal{C}[i\langle\nabla\rangle_{m}^{-1}\mathcal{M}_{\{1,2\}},i\langle\nabla\rangle_{m}](\cos(\frac{1}{2}v),\sin(\frac{1}{2}\overline{v})).$$

For  $A = (H^1 \cap H^{\frac{d}{2}+\epsilon})(\mathbf{T}^d)^2$  or  $A = (H^1_0 \cap H^{\frac{d}{2}+\epsilon})(\Omega)^2$ , one Taylor-expands the operators which give:

$$\begin{split} R^{0}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{0})(\hat{\mathbf{v}},\cdot)) &= -\sin(\frac{1}{2}v)\cos(\frac{1}{2}\overline{v}) - \langle \nabla \rangle_{m}^{-1} \Bigg( \left( \langle \nabla \rangle_{m}\sin(\frac{1}{2}v)\right) \left(\cos(\frac{1}{2}\overline{v})\right) \Bigg) \\ &+ \langle \nabla \rangle_{m}^{-1} \Bigg( \sin(\frac{1}{2}v) \left( \langle \nabla \rangle_{m}\cos(\frac{1}{2}\overline{v}) \right) \Bigg) \\ R^{0}_{o,A}((\tilde{\Pi}_{o,A}\lambda_{1})(\hat{\mathbf{v}},\cdot)) &= -\cos(\frac{1}{2}v)\sin(\frac{1}{2}\overline{v}) - \langle \nabla \rangle_{m}^{-1} \Bigg( \left( \langle \nabla \rangle_{m}\cos(\frac{1}{2}v)\right) \left(\sin(\frac{1}{2}\overline{v})\right) \Bigg) \\ &+ \langle \nabla \rangle_{m}^{-1} \Bigg( \cos(\frac{1}{2}v) \left( \langle \nabla \rangle_{m}\sin(\frac{1}{2}\overline{v}) \right) \Bigg) \end{split}$$

,

which concludes the proof.

Remark 3.5.8 Similarly to Corollary 3.5.6 our general framework (3.4.2) also allows for low regularity approximations at higher order for Klein–Gordon equations.

Part II

The setting of a randomized initial condition

# Chapter 4

# Approximations of dispersive PDEs in the presence of low-regularity randomness

This chapter is based on the article [ABBS22a] accepted in Foundations of Computational Mathematics.

Abstract. We introduce a new class of numerical schemes which allow for low regularity approximations to the expectation  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$ , where  $u_k$  denotes the k-th Fourier coefficient of the solution u of the dispersive equation and  $v^{\eta}(x)$  the associated random initial data. This quantity plays an important role in physics, in particular in the study of wave turbulence where one needs to adopt a statistical approach in order to obtain deep insight into the generic long-time behaviour of solutions to dispersive equations. Our new class of schemes is based on Wick's theorem and Feynman diagrams together with a resonance based discretisation [BS22] set in a more general context: we introduce a novel combinatorial structure called paired decorated forests which are two decorated trees whose decorations on the leaves come in pair. The character of the scheme draws its inspiration from the treatment of singular stochastic partial differential equations via Regularity Structures. In contrast to classical approaches, we do not discretize the PDE itself, but rather its expectation. This allows us to heavily exploit the optimal resonance structure and underlying gain in regularity on the finite dimensional (discrete) level.

# 4.1 Introduction

We consider nonlinear dispersive equations

$$i\partial_t u(t,x) + \mathcal{L}(\nabla)u(t,x) = |\nabla|^{\alpha} p(u(t,x), \overline{u}(t,x)), \quad (t,x) \in \mathbf{R} \times \mathbf{T}^d$$

$$\tag{4.1}$$

with random initial data:

$$u(0,x) = v^{\eta}(x) = \sum_{k \in \mathbf{Z}^d} v_k \eta_k e^{ikx}, \qquad (4.2)$$

where  $v_k = \overline{v_{-k}}$  and  $(\eta_k)_{k \in \mathbb{Z}^d}$  is a family of random variables satisfying

$$\mathbb{E}(\eta_k \eta_{-\ell}) = \delta_{k,\ell}.$$
(4.3)

For the condition (4.3) to be satisfied we require  $\eta_0$  to be a real normalized centered Gaussian,  $(\eta_k)_{k>0}$  to be standard complex Gaussians,  $(\eta_k)_{k\geq 0}$  to be independent and to satisfy  $\eta_k = \overline{\eta_{-k}}$ . We first note that the above setting is for real initial conditions. In the complex case, we simply need to take  $(\eta_k)_{k\in\mathbb{Z}^d}$  to be independent standard complex Gaussians which satisfy  $\mathbb{E}(\eta_k \overline{\eta_\ell}) = \delta_{k,\ell}$ , and refer to the Remark 4.1.1. Working with Gaussian random variables allows us to use Wick's formula for computing the second moment of  $u_k(\tau, v^{\eta})$ . Nevertheless, one can go beyond the Gaussian realm. For a discussion on the subject we refer to Remark 4.1.2.

Furthermore, we assume in (4.1) a polynomial nonlinearity p. Moreover, we suppose that there exists a deterministic time T > 0 such that the structure of (4.1) implies at least almost-sure local wellposedness of the problem in an appropriate functional space on the finite time interval [0, T]. This is a non-trivial assumption since naturally the time of existence is random and not necessarily almost surely bounded from below by a positive time T. We refer to the works on probabilistic Cauchy theory given for instance by [BT08, BT13, SX16, CO12, NS15, CG122] in the case of nonlinear Schrödinger and wave equations. With the above assumption we have that the solution u admits up until time T a second moment restricted on the almost surely set of its definition. In order to keep notations simple, we will omit such a set in the sequel.

The differential operators  $\mathcal{L}(\nabla)$  and  $|\nabla|^{\alpha}$  shall cast in Fourier space into the form

$$\mathcal{L}(\nabla)(k) = k^{\sigma} + \sum_{\gamma:|\gamma| < \sigma} a_{\gamma} \prod_{j} k_{j}^{\gamma_{j}}, \qquad |\nabla|^{\alpha}(k) = \sum_{\gamma:|\gamma| \le \alpha} \prod_{j=1}^{d} |k_{j}|^{\gamma_{j}}$$
(4.4)

for some  $\alpha \in \mathbf{R}$ ,  $\sigma \in \mathbf{N}$ ,  $\gamma \in \mathbf{Z}^d$  and  $|\gamma| = \sum_i \gamma_i$ , where for  $k = (k_1, \ldots, k_d) \in \mathbf{Z}^d$  and  $m = (m_1, \ldots, m_d) \in \mathbf{Z}^d$  we set

$$k^{\sigma} = k_1^{\sigma} + \ldots + k_d^{\sigma}, \qquad k \cdot m = k_1 m_1 + \ldots + k_d m_d$$

Concrete examples are discussed in Section 4.5, including the cubic nonlinear Schrödinger (NLS) equation

 $i\partial_t u + \mathcal{L}(\nabla)u = |u|^2 u, \quad \mathcal{L}(\nabla) = \Delta, \quad \alpha = 0, \quad p(u,\bar{u}) = |u|^2 u$ (4.5)

and the Korteweg–de Vries (KdV) equation

$$i\partial_t u + \mathcal{L}(\nabla)u = i\frac{1}{2}\partial_x u^2, \quad \mathcal{L}(\nabla) = i\partial_x^3, \quad \alpha = 1, \quad p(u,\bar{u}) = i\frac{1}{2}u^2.$$
(4.6)

Note that one could deal with non-polynomial nonlinearities, and work on more general domains than the periodic one, by combining the present work with the general framework established in [ABBS22b]. Indeed, by introducing nested commutator structures the work [ABBS22b] overcomes the necessity of periodic boundary conditions and polynomial nonlinearities, see also [RS21, AB23a]. One can use the iterated integrals produced in [ABBS22b] and then proceed in Fourier space as in the present work. Doing all the computations in physical space will require a description of the second moment of the random initial data directly in physical space; however, such a randomisation is in general mostly described in Fourier space.

The aim of this paper is to introduce a new class of schemes which, by denoting by  $\tau$  the time step, allows for a low regularity approximation to

$$\mathbb{E}(|u_k(t,v^{\eta})|^2). \tag{4.7}$$

Here,  $\mathbb{E}$  denotes the expectation and  $u_k$  the k-th Fourier coefficient of the solution u of (4.1) with corresponding initial data  $v^{\eta}$  defined at equation (4.2). The quantity (4.7) plays an important role in physics, in particular in the study of wave turbulence. A fundamental question in the latter is to derive a rigorous justification of the wave kinetic equation (WKE), an equation which describes the effective dynamics of an interacting wave system in the thermodynamic limit. This yields deep insight into the *generic* long-time behaviour of solutions to dispersive equations. In order to derive this equation one starts with a random initial value and work on a large box of size L. Then under suitable scaling laws which depend on the strength  $\mu$  of the non-linearity, and the size L of the box, one can show that in the limit of large L and small  $\mu$  the effective dynamic of  $\mathbb{E}(|u_k(t, v^{\eta})|^2)$  is given by the WKE over sufficiently long time scales.

A rigorous derivation of the WKE for NLS is performed in [DH21, DH23, ACG21] via a diagrammatic expansion. A similar rigorous derivation result for a high dimensional ( $d \ge 14$ ) discrete KdV-type equation was achieved in [ST21] using also Feynman diagrams, in order to derive its associated wave kinetic equation at the kinetic time under a suitable scaling law. All these works are based on the iteration of Duhamel's formula for (4.1):

$$u(\tau, v^{\eta}) = e^{i\tau \mathcal{L}(\nabla)} v^{\eta} - i |\nabla|^{\alpha} e^{i\tau \mathcal{L}(\nabla)} \int_{0}^{\tau} e^{-i\xi \mathcal{L}(\nabla)} p(u(\xi, v^{\eta}), \bar{u}(\xi, v^{\eta})) d\xi$$

$$(4.8)$$

which in Fourier space cast into the form

$$u_{k}(\tau, v^{\eta}) = e^{i\tau \mathcal{L}(\nabla)(k)} \eta_{k} v_{k} - i |\nabla|^{\alpha}(k) e^{i\tau \mathcal{L}(\nabla)(k)}$$

$$\int_{0}^{\tau} e^{-i\xi \mathcal{L}(\nabla)(k)} p_{k}(u(\xi, v^{\eta}), \bar{u}(\xi, v^{\eta})) d\xi,$$

$$(4.9)$$

where  $p_k$  is given for  $p(u, \bar{u}) = |u|^2 u$  by

 $p_k(u,\bar{u}) = \sum_{k=-k_1+k_2+k_3} \bar{u}_{k_1} u_{k_2} u_{k_3}.$ 

Iterations of (4.9) produces a tree series, see also [Chr07, Gub12, GKO13] where the series is made explicit for some specific equations. This expansion can also be described via random tensors introduced in [DNY22]. From this tree series, one can compute  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  using Wick's theorem and Feynman diagrams [DH21, DH23].

The theoretical works mentioned above on the study of wave turbulence, together with the idea of a resonance based discretisation [BS22], set the inspiration for this work. From an algebraic point of view we, however, have to work in a much more general context considering trees with more decorations than introduced in [BS22], with a new combinatorial

structure called *paired decorated forests*. This will allow us to approximate a large class of dispersive PDEs with one general scheme, without the need of treating each and every equation separately.

In contrast to classical approaches, we do not discretize the PDE itself, but rather its expectation. This allows us to heavily exploit the optimal resonance structure and underlying gain in regularity on the discrete level. More precisely, the iteration of Duhamel's formulation (4.8) of (4.1) can be expressed using decorated trees, see [BS22]. Namely, it follows from [BS22, Section 1.2] that the k-th Fourier coefficient of the approximated solution at order r is given by

$$u_k^r(\tau, v^\eta) = \sum_{T \in \mathcal{V}_k^r} \frac{\Upsilon^p(T)(v^\eta)}{S(T)} (\Pi T)(\tau)$$
(4.10)

where  $\mathcal{V}_k^r$  is a set of decorated trees which incorporate the frequency k, S(T) is the symmetry factor associated to the tree T,  $\Upsilon^p(T)$  is the coefficient appearing in the iteration of Duhamel's formulation and  $(\Pi T)(\tau)$  represents a Fourier iterated integral. The exponent p corresponds to the nonlinearity appearing in the right hand side of (4.1). The exponent r in  $\mathcal{V}_k^r$  means that we consider only trees of size at most r+1, these are the trees corresponding to an iterated integral of depth at most r+1. These quantities are described in detail in Section 4.4. Generally speaking, the sum (4.10) is the truncation at order r of the infinite series describing formally the solution of (4.1). The low regularity scheme for (4.1) is then obtained by replacing each oscillatory integral  $(\Pi T)(\tau)$  appearing in the finite sum (4.10) by a low regularity approximation that embeds the resonance structure into the numerical discretization. We will denote the latter low-regularity approximation operator by  $\Pi^{n,r}$ . Here, r corresponds to the order of the discretization and n is a formal a priori regularity assumed on the initial data  $v^{\eta}$ . Namely, we assume that we can differentiate  $v^{\eta}$ , n-times. The general scheme then takes the form:

$$U_{k}^{r}(\tau, v^{\eta}) = \sum_{T \in \mathcal{V}_{k}^{r}} \frac{\Upsilon^{p}(T)(v^{\eta})}{S(T)} (\Pi^{n,r}T)(\tau).$$
(4.11)

The local error structure for each approximated iterated integral is given by

$$(\Pi T - \Pi^{n,r}T)(\tau) = \mathcal{O}\big(\tau^{r+2}\mathcal{L}^r_{\text{low}}(T,n)\big),\tag{4.12}$$

where  $\mathcal{L}_{low}^r$  involves only lower order derivatives. The form of the scheme draws its inspiration from the treatment of singular stochastic partial differential equations (SPDEs) via Regularity Structures in [Hai14, BHZ19, BCCH20, BHZ20]. These decorated trees expansions are generalization of the B-series widely used for ordinary differential equations, we refer to [But72, CCO08, HLW10, MKL13]. In the end, one obtains an approximation of u under much lower regularity assumptions than classical methods (e.g., splitting methods, exponential integrators [CG12, Fao12, HLW10, HO10, Ho110, Law67, Lub08, LR04, MQ02, SSC18]) require, which in general introduce the local error

$$\mathcal{O}\big(\tau^{r+2}\mathcal{L}^r(T,n)\big) \tag{4.13}$$

involving the full high order differential operator  $\mathcal{L}^r$ . Indeed, we have that  $\mathcal{D}(\mathcal{L}_{low}) \supset \mathcal{D}(\mathcal{L})$ , meaning that the local error structure (4.12) allows us to deal with a rougher class of solutions than the classical error bound (4.13). The underlying idea behind these low-regularity approximations was initiated by the work of [OS18], and then generalized by [BS22, ABBS22b] to higher order methods, allowing for approximations to large classes of equations. Although the error bound (4.12) derived here are formal, one can obtain rigorous low-regularity error bounds using a classical Sobolev space setting (see [RS21, AB23a]), as well as sharp  $L^2$  error estimates by using discrete Strichartz estimates and discrete Bourgain spaces (see [IZ09, ORS21, ORS22b, RS22, JORS23]). Let us mention that the local error analysis can be nicely understood via a Birkhoff factorisation of the character  $\Pi^{n,r}$  (see [BS22, BEF20]) that involves a deformed Butcher-Connes-Kreimer coproduct (see [But72, CK99, CK00, BS22, BM23]). In the present work, we give a recursive description with various Taylor remainders in Definition 4.3.5. This formulation is similar to [ABBS22b] where the Birkhoff factorisation is not available. The Birkhoff factorisation can also be used in our work for a more precise description of the Taylor remainders. We also believe that it could have an impact on the general description of symmetries for low regularity schemes. For the moment, only symmetric and symplectic schemes have been work out by hand (see [AB23b, MS23]). Lastly, we note that the scheme (4.11) has been generalized to non-polynomial nonlinearities and to parabolic equations in [ABBS22b] with the use of nested commutators first introduced in [RS21].

A natural route to obtain a low regularity approximation to  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  would be to first compute the low regularity approximation  $U_k^{n,r}(\tau, v^{\eta})$  given in (4.11) to  $u_k(\tau, v^{\eta})$ , and then to evaluate its expectation  $\mathbb{E}(|U_k^{n,r}(\tau, v^{\eta})|^2)$ while removing the terms of higher order. By doing so, we would, however, ignore the enhanced resonance structure and gain in regularity introduced by applying the expectation  $\mathbb{E}$ . In order to exploit the latter on the discrete level, we will directly compute a low regularity approximation to the second moment of  $u_k^r$ , where  $u_k^r$  is defined in (4.10), and is the truncation at order r of the tree series describing  $u_k$ . Namely, we are interested in computing the following series:

$$V_k^r(\tau, v) = \mathcal{Q}_{\leq r+1} \mathbb{E}(|u_k^r(\tau, v^\eta)|^2)$$

$$(4.14)$$
$$= \sum_{F=T_1 \cdot T_2 \in \mathcal{G}_L^r} m_F \frac{\Upsilon^p(T_1)(v) \Upsilon^p(T_2)(v)}{S(T_1)S(T_2)} (\bar{\Pi}T_1)(\tau) (\Pi T_2)(\tau)$$

where  $\mathcal{Q}_{\leq r+1}$  keeps only the terms of order less than or equal to  $\tau^{r+1}$ , and  $m_F$  is a combinatorial coefficient associated to F. Namely,  $m_F$  counts the number of pairings that produce F. It remains to describe the set  $\mathcal{G}_k^r$ . For this purpose we introduce a new combinatorial structure called paired decorated forest which are two decorated trees whose decorations on the leaves come in pair. This pairing among the leaves come from the use of Wick's theorem for products of Gaussian random variables that appear in the computation of  $V_k^r$ . Similar structures have been used in the theoretical analysis of dispersive PDEs [DH21, DH23, ACG21, ST21] as well as in stochastic differential equations and exotic B-series [LV20, Bro22]. The set  $\mathcal{G}_k^r$  consists of all paired forests  $F = T_1 \cdot T_2$  of size at most r + 1, meaning that the term  $(\overline{\Pi}T_1)(\tau)(\Pi T_2)(\tau)$  contains at most r + 1 integrals in time.  $T_1$  and  $T_2$  are the two paired trees and  $\cdot$  denotes the forest product. Further, in the above series we have used the following short hand notation:

$$\overline{\Upsilon}^p(T_1)(v) = \overline{\Upsilon^p(T_1)(v)}, \quad (\overline{\Pi}T_1)(\tau) = \overline{(\Pi T_1)(\tau)},$$

which stands for the complex conjugate, and will be abundantly used throughout the reminder of this article. Notice that the above sum (4.14) is not symmetric, meaning that both  $(\Pi T_1)(\tau)(\Pi T_2)(\tau)$  and  $(\Pi T_2)(\tau)(\Pi T_1)(\tau)$  appear when  $T_1 \neq T_2$ . We then proceed with the discretization in time of (4.14) as is done in [BS22]: we replace  $(\Pi T_i)(\tau)$  by their resonance based approximations  $(\Pi^{n,r}T_i)$  which yields the following tree series:

$$V_{k}^{n,r}(\tau,v) = \sum_{F=T_{1}:T_{2}\in\mathcal{G}_{k}^{r}} m_{F} \frac{\Upsilon^{p}(T_{1})(v)\,\Upsilon^{p}(T_{2})(v)}{S(T_{1})S(T_{2})}$$

$$\mathcal{Q}_{\leq r+1}(\bar{\Pi}^{n,r}T_{1}\Pi^{n,r}T_{2})(\tau).$$
(4.15)

The local error structure is given by

$$\left(\bar{\Pi}T_{1}\Pi T_{2} - \mathcal{Q}_{\leq r+1}\Pi^{n,r}T_{1}\bar{\Pi}^{n,r}T_{2}\right)(\tau) = \mathcal{O}\left(\tau^{r+2}\mathcal{L}^{r}_{low}(T_{1}\cdot T_{2},n)\right).$$
(4.16)

The definition of  $\mathcal{L}_{low}^r(T_1 \cdot T_2, n)$  is, however, much more involved than in the work of [BS22]. Indeed, given that we consider decorated forests  $F = T_1 \cdot T_2$ , whose leaves (which encodes the frequency) come in pairs, we have that the same frequencies can appear in both the decorated tree  $T_1$  and that of  $T_2$ . Hence, unlike in the work of [BS22], the set of frequencies appearing on the leaves of  $T_1$  and  $T_2$  are not disjoint, and in consequence a new framework needs to be considered. Moreover, since our calculations are made with these pairings  $T_1 \cdot T_2$ , cancellations appear when computing the variance (4.7) and the schemes obtained in this work are of simpler form than those obtained in the work of [BS22].

Our main result is the general class of schemes given by (4.15) together with the correct combinatorial structures on paired decorated forests, which yields a low-regularity approximation of (4.7). Our construction is based upon Wick's formula (see Proposition 4.4.1) and is presented in Proposition 4.4.2. The local error structure (4.12) is then given in Theorem 4.4.3. Theorem 4.4.3 is heavily based on Theorem 4.3.6 which introduces a new recursive definition of  $\mathcal{L}^{r}_{\text{low}}(T_1 \cdot T_2, n)$  (see Definition 4.3.5) which differs from the one presented in [BS22].

The new scheme (4.15) is an attempt to implement the low regularity integrators introduced in [BS22] when randomness is introduced at the level of the initial data. We expect that the scheme described in [BS22, ABBS22b] will be able to enhance the error structure of classical methods for (dispersive) stochastic PDEs (see, e.g. [BC23, Deb11, LPS14] and the references therein) for low regularity initial data where randomness is added via a stochastic forcing.

**Remark 4.1.1** Our assumption on the initial data in (4.3) enforces the initial data to be real. Our scheme is fairly general and works for both real or complex initial data. In the case of a complex initial data the authors [DH23, DH21] use the following assumption. Let  $(\eta_k)_{k \in \mathbb{Z}^d}$  be a family of independent identically distributed complex random variables (centered standard Gaussian variables or uniform distributions on the unit circle) satisfying: For all  $k, \ell \in \mathbb{Z}^d$ 

$$\mathbb{E}(|\eta_k|^2) = 1, \quad \mathbb{E}(\eta_k \eta_\ell) = 0$$

In the Gaussian case see also [CG19]. The second identity in the above simplifies a lot the interactions that need to be considered for building up the Feynman diagrams. Indeed, only the terms  $\mathbb{E}(\eta_k \overline{\eta_\ell})$  from Wick's theorem are potentially non-zero. These terms are the easiest to treat requiring no extra regularity, (see Section 4.1.1 and (4.22) where the pairing  $T_0 \cdot T_{1,1}$  is no longer possible, it is the pairing that asks more regularity in the discretisation, see (4.26)).

**Remark 4.1.2** In (4.14), the summation is performed on paired decorated forests of the form  $T_1 \cdot T_2$ . The fact that we have only two trees and not more correspond to the computation of the second moment. For the *p*-th moment, one will get *p* paired decorated trees where the leaves will come in pairs. In both cases second moment or higher moments, one uses Wick's theorem applied to product of Gaussian random variables for computing the pairings. One can potentially

go beyond the Gaussian realm by using cumulants for computing the moments of a given random variable. This was extensively used in the context of singular SPDEs (see [CH16] for a general convergence result of renormalised models and [EH21, BN22] for recent works in the discrete setting). In our context, one can imagine to have trees where the leaves are split into different sets of cardinality bigger than two. The scheme will be robust to this case.

Before introducing our general framework we will in the next section illustrate our main idea on an example taken from NLS.

#### 4.1.1 Example

Let us consider the equation:

$$i\partial_t u + \Delta u = |u|^2 u, \quad u(0,x) = v^{\eta}(x) = \sum_{k \in \mathbf{Z}^d} v_k \eta_k e^{ikx}$$
 (4.17)

set on the *d* dimensional torus. We will discuss the approximation of  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  to first order. We rewrite (4.17) using Duhamel's formula:

$$u(\tau, v^{\eta}) = e^{i\tau\Delta}v^{\eta} - ie^{i\tau\Delta} \int_0^{\tau} e^{-i\xi\Delta} \Big(\overline{u(\xi, v^{\eta})}u(\xi, v^{\eta})^2\Big) d\xi.$$

One obtains the following equation for the Fourier coefficient  $u_k(\tau, v^{\eta})$ :

$$u_{k}(\tau, v^{\eta}) = e^{-i\tau k^{2}} \eta_{k} v_{k} - \sum_{\substack{k_{1}, k_{2}, k_{3} \in \mathbf{Z}^{d} \\ -k_{1}+k_{2}+k_{3}=k}} ie^{-i\tau k^{2}} \int_{0}^{\tau} e^{i\xi k^{2}} \overline{u_{k_{1}}(\xi, v^{\eta})} u_{k_{2}}(\xi, v^{\eta}) u_{k_{3}}(\xi, v^{\eta}) d\xi,$$

$$(4.18)$$

where the linear operator  $e^{i\tau\Delta}$  (resp.  $e^{-i\tau\Delta}$ ) is sent to  $e^{-i\tau k^2}$  (resp.  $e^{i\tau k^2}$ ) in Fourier space. The splitting of k into  $k_1, k_2, k_3$  comes from the fact that moving to Fourier space, products become convolution products on the frequencies. We fix r = 0 and iterate (4.18) by replacing  $u_{k_i}(\tau, v^{\eta})$  by

$$u_{k_j}(\tau, v^{\eta}) = e^{-i\tau k_j^2} \eta_{k_j} v_{k_j} + \mathcal{O}(\tau),$$

with  $j \in \{1, 2, 3\}$ . We obtain the following first order approximation of the k-th Fourier coefficient  $u_k(\tau, v^{\eta})$ :

$$u_{k}(\tau, v^{\eta}) = e^{-i\tau k^{2}} \eta_{k} v_{k} - \sum_{\substack{k_{1}, k_{2}, k_{3} \in \mathbf{Z}^{d} \\ -k_{1} + k_{2} + k_{3} = k}} ie^{-i\tau k^{2}} \int_{0}^{\tau} e^{i\xi k^{2}} (e^{i\xi k_{1}^{2}} \overline{\eta_{k_{1}} v_{k_{1}}}) (e^{-i\xi k_{2}^{2}} \eta_{k_{2}} v_{k_{2}}) (e^{-i\xi k_{3}^{2}} \eta_{k_{3}} v_{k_{3}}) d\xi + \mathcal{O}(\tau^{2}).$$

We encode the above Duhamel iterates using the tree series (4.10). Namely, on the continuous level, from (4.10) we have

$$u_{k}^{0}(\tau, v^{\eta}) = u_{k,1}^{0}(\tau, v^{\eta}) + u_{k,2}^{0}(\tau, v^{\eta}),$$

$$u_{k,1}^{0}(\tau, v^{\eta}) = \frac{\Upsilon^{p}(T_{0})(v^{\eta})}{S(T_{0})}(\Pi T_{0})(\tau),$$

$$u_{k,2}^{1}(\tau, v^{\eta}) = \sum_{\substack{k_{1}, k_{2}, k_{3} \in \mathbf{Z}^{d} \\ -k_{1}+k_{2}+k_{3}=k}} \frac{\Upsilon^{p}(T_{1})(v^{\eta})}{S(T_{1})}(\Pi T_{1})(\tau),$$
(4.19)

where

$$T_{0} = \overbrace{I}^{(k)}, \quad T_{1} = \overbrace{I}^{(k)}, \quad S(T_{0}) = 1, \quad S(T_{1}) = 2, \quad (4.20)$$
$$\Upsilon^{p}(T_{0})(v^{\eta}) = \eta_{k}v_{k}, \quad \Upsilon^{p}(T_{1})(v^{\eta}) = 2\bar{\eta}_{k_{1}}\bar{v}_{k_{1}}\eta_{k_{2}}v_{k_{2}}\eta_{k_{3}}v_{k_{3}},$$

$$(\Pi T_0)(\tau) = e^{-i\tau k^2}, \quad (\Pi T_1)(\tau) = -ie^{-i\tau k^2} \int_0^\tau e^{i\xi k^2} e^{i\xi k_1^2} e^{-i\xi k_2^2} e^{-i\xi k_3^2} d\xi.$$

For the decorated trees written above, we have used the following coding: an edge | (resp.  $\vdots$ ) corresponds to a factor  $e^{-itk^2}$  (resp.  $e^{itk^2}$ ), while an edge | (resp. |) corresponds to an integral  $-i \int_0^\tau e^{i\xi k^2} \cdots d\xi$  (resp.  $-i \int_0^\tau e^{-i\xi k^2} \cdots d\xi$ ). The dash dotted line introduces a minus sign. The decorations on the leaves correspond to the frequencies that add up in the intern node with a minus sign if one faces a dash edge. In  $T_1$ , the two inner nodes are decorated by  $k = -k_1 + k_2 + k_3$  where  $-k_1$  comes from the edge  $\vdots$ . The definition of the symmetry factor S(T) and the coefficients  $\Upsilon^p(T)$  are defined in Section (4.4). The symmetry factor corresponds to the number of internal symmetries of the tree taking the edge decorations into account but not the node decorations, which explains why  $S(T_1) = 2$ .

By construction,  $u_k^0$  is a first order truncated approximation of  $u_k$ , and it follows from (4.19) that

$$\begin{split} \mathbb{E}(|u_{k}^{0}(\tau,v^{\eta})|^{2}) &= \mathbb{E}\Big((u_{k,1}^{0}(\tau,v^{\eta}) + u_{k,2}^{1}(\tau,v^{\eta}))(\overline{u_{k,1}^{0}(\tau,v^{\eta}) + u_{k,2}^{1}(\tau,v^{\eta})})\Big) \\ &= \mathbb{E}(|u_{k,1}^{0}(\tau,v^{\eta})|^{2}) + 2Re\mathbb{E}\Big(\overline{u_{k,1}^{0}(\tau,v^{\eta})}u_{k,2}^{1}(\tau,v^{\eta})\Big) + \mathcal{O}(\tau^{2}). \end{split}$$

The main idea is then that we can express on the continuous level the above expansion using the tree series (4.14), and on the discrete level by the tree series (4.15). We show how this is done by detailing the computation of the term

$$\mathbb{E}\Big(\overline{u_{k,1}^0(\tau,v^\eta)}u_{k,2}^1(\tau,v^\eta)\Big)$$

It follows from (4.20) that we need to consider

$$\mathbb{E}\left(\overline{\Upsilon^p(T_0)(v^\eta)}\Upsilon^p(T_1)(v^\eta)\right) = 2\bar{v}_k\bar{v}_{k_1}v_{k_2}v_{k_3}\mathbb{E}(\bar{\eta}_k\bar{\eta}_{k_1}\eta_{k_2}\eta_{k_3}).$$
(4.21)

Further, we have,

$$\begin{split} \mathbb{E}(\bar{\eta}_{k}\bar{\eta}_{k_{1}}\eta_{k_{2}}\eta_{k_{3}}) &= \mathbb{E}(\bar{\eta}_{k}\bar{\eta}_{k_{1}})\mathbb{E}(\eta_{k_{2}}\eta_{k_{3}}) + \mathbb{E}(\bar{\eta}_{k}\eta_{k_{2}})\mathbb{E}(\bar{\eta}_{k_{1}}\eta_{k_{3}}) \\ &+ \mathbb{E}(\bar{\eta}_{k}\eta_{k_{3}})\mathbb{E}(\bar{\eta}_{k_{1}}\eta_{k_{2}}) \\ &= \delta_{k_{1}\bar{k}_{1}}\delta_{k_{2},\bar{k}_{2}} + \delta_{k_{1}k_{2}}\delta_{k_{1},k_{3}} + \delta_{k_{1}k_{3}}\delta_{k_{1},k_{2}} \end{split}$$

where we have used Wick's formula (see Proposition 4.4.1) for products of Gaussian random variables. The different  $\delta_{\cdot,\cdot}$  correspond to different pairings and fix completely the frequencies on the leaves. We will introduce a new combinatorial object for describing these pairing called *decorated paired forests*. Below, we represent the various pairings coming from (4.21):

$$T_{0} = \bigwedge^{(k)}_{i}, \quad T_{1,1} = \bigwedge^{(k)}_{i}, \quad T_{1,2} = \bigwedge^{(k)}_{i}, \quad T_{1,3} = \bigwedge^{(k)}_{i}$$

where  $\bar{k}_1 = -k_1$  and  $\cdot$  denotes the forest product. The  $F_i$  are decorated forests. One can observe that  $F_2$  and  $F_3$  are similar due to the symmetry of the decorated tree  $T_1$ . Hence, we only need to consider one of the decorated forests during the computations, we choose to consider  $F_2$ . Further, it follows in an analogous fashion from (4.19) and (4.20) that there is a single nontrivial pairing when computing  $\mathbb{E}(|u_{k,1}^0(\tau, v^{\eta})|^2)$ , and that the decorated forest

$$F_0 = T_0 \cdot T_0$$

encodes this pairing and consequently the term  $\mathbb{E}(|u_{k,1}^0(\tau, v^{\eta})|^2)$ . Therefore, the set  $\mathcal{G}_k^0$  is defined as:

$$\mathcal{G}_k^0 = \{ T_0 \cdot T_0, T_0 \cdot T_{1,1}, T_0 \cdot T_{1,2}, k_1 \in \mathbf{Z}^d \}.$$

These decorated forests encode products of iterated integrals, where each iterated integral is encoded by a decorated tree of the form (4.22). Namely, we have

$$(\bar{\Pi}T_0)(\tau) \ (\Pi T_{1,1})(\tau) = e^{i\tau k^2} \left( -ie^{-i\tau k^2} \int_0^\tau e^{i\xi k^2} e^{i\xi k^2} e^{-i\xi k_2^2} e^{-i\xi k_2^2} d\xi \right)$$

$$= -i \int_0^\tau e^{2i\xi(k^2 - k_2^2)} d\xi,$$
  
$$(\bar{\Pi}T_0)(\tau) \ (\Pi T_{1,2})(\tau) = e^{i\tau k^2} \left( -ie^{-i\tau k^2} \int_0^\tau e^{i\xi k^2} e^{i\xi k_1^2} e^{-i\xi k^2} e^{-i\xi k_1^2} d\xi \right)$$
  
$$= -i\tau$$

One observes that for  $F_2$ , due to cancellations in the frequencies, one can perform exact integration in time, and no approximation step is required. The term which requires to be approximated and which asks for some regularity on the initial data is the integral encoded by  $F_1$ . We now discuss how the approximations are made, and how it leads us to the tree series (4.15).

Let us first proceed by describing the idea behind the approximation of  $\Pi T_1$ . The main idea of the general resonance based scheme introduced in [BS22] is to split the frequencies in the integrand into the dominant and lower parts:

$$\mathcal{L} = k^2 + k_1^2 - k_2^2 - k_3^2 = \mathcal{L}_{\text{dom}} + \mathcal{L}_{\text{low}}$$

$$\mathcal{L}_{\text{dom}} = 2k_1^2, \quad \mathcal{L}_{\text{low}} = -k_1(k_2 + k_3) + k_2k_3$$
(4.23)

such that (cf. (4.20))

$$(\Pi T_1)(\tau) = -ie^{-i\tau k^2} \int_0^\tau e^{i\xi\mathcal{L}_{\text{low}}} e^{i\xi\mathcal{L}_{\text{low}}} d\xi.$$
(4.24)

The idea of the splitting (4.23) lies in the observation that  $k_1^2$  asks for two derivatives on the initial data while the cross-terms given in  $\mathcal{L}_{\text{low}}$  require only one. Moreover, the term  $e^{i\xi\mathcal{L}_{\text{dom}}}$  can be integrated out exactly and mapped back into physical space. In order to gain regularity, we will thus only Taylor-expand the lower order part  $e^{i\xi\mathcal{L}_{\text{low}}}$  in (4.24) while integrating the dominant part  $e^{i\xi\mathcal{L}_{\text{dom}}}$  exactly. This provides the following discretisation:

$$(\Pi T_1)(\tau) = -ie^{-i\tau k^2} \int_0^\tau e^{2i\xi k_1^2} (1 + \mathcal{O}(\xi \mathcal{L}_{\text{low}})) d\xi$$
  
=  $(\Pi^{1,0} T_1)(\tau) + \mathcal{O}(\tau^2 \mathcal{L}_{\text{low}})$   
 $\Pi^{1,0} T_1)(\tau) = -ie^{-i\tau k^2} \int_0^\tau e^{2i\xi k_1^2} d\xi$ 

where the index 0 correspond to the order of the approximation inside the time integral and the index 1 embeds that we ask a priori one derivative on the initial data. If we assume more regularity, such as for instance two derivatives, one can perform the full Taylor expansion and obtains:

$$(\Pi T_1)(\tau) = (\Pi^{2,0} T_1)(\tau) + \mathcal{O}(\tau^2 \mathcal{L}), \quad (\Pi^{2,0} T_1)(\tau) = -i\tau e^{-i\tau k^2}.$$

We would like to proceed in the same manner for the approximation of  $(\Pi T_{1,1})(\tau)$ . However, we have to choose the splitting as follows:

$$\mathcal{L} = 2k^2 - 2k_2^2, \quad \mathcal{L}_{\text{low}} = \mathcal{L}, \quad \mathcal{L}_{\text{dom}} = 0.$$
(4.25)

This is due to the fact that  $\frac{1}{k^2-k_1^2}$  cannot be mapped back into Physical space and hence, we would recover a scheme which needs to be computed fully in Fourier space and where we could not make use of the Fast Fourier Transform (FFT). This would not allow for a practical implementation of the scheme and would lead to high computational and memory costs, in particular in higher spatial dimensions  $d \ge 3$ . This motivates our choice of dominant and lower order parts (4.25) and leads to the discretization

$$(\Pi T_{1,1})(\tau) = (\Pi^{1,0} T_{1,1})(\tau) + \mathcal{O}(\tau^2 \mathcal{L}), \quad (\Pi^{1,0} T_{1,1})(\tau) = -i\tau e^{-i\tau k^2}.$$
(4.26)

The discretization (c.f. (4.15)) is then given by:

(

$$(\bar{\Pi}^{1,0}T_0)(\tau) (\Pi^{1,0}T_{1,j})(\tau) = -i\tau.$$

The term

$$\mathbb{E}\left(\overline{u_{k,1}^0(\tau,v^\eta)}u_{k,2}^1(\tau,v^\eta)\right)$$

is then well approximated by

$$\frac{\bar{\Upsilon}^{p}(T_{0})(v^{\eta})}{S(T_{0})} \left( \sum_{k_{1} \in \mathbf{Z}^{d}} \frac{\Upsilon^{p}(T_{1,1})(v^{\eta})}{S(T_{1,1})} \mathcal{Q}_{\leq 1}(\bar{\Pi}T_{0})(\tau)(\Pi T_{1,1})(\tau) \right. \\ \left. + 2 \sum_{k_{1} \in \mathbf{Z}^{d}} \frac{\Upsilon^{p}(T_{1,2})(v^{\eta})}{S(T_{1,2})} \mathcal{Q}_{\leq 1}(\bar{\Pi}T_{0})(\tau)(\Pi T_{1,2})(\tau) \right) \\ \left. = -3i\tau |v_{k}|^{2} \sum_{k_{1} \in \mathbf{Z}^{d}} |v_{k_{1}}|^{2} \right.$$

and introduces the following local error

$$\mathcal{O}(\tau^2 \sum_{k_1 \in \mathbf{Z}^d} (k^2 - k_1^2) |v_k|^2 |v_{k_1}|^2).$$

We notice that a term of the form  $\frac{k^2|v_k|^2}{|v(-x)|^2}$  corresponds (up to a sign) to the Fourier coefficient of  $\nabla v * \nabla \tilde{v}$  where \* is the space convolution, and  $\tilde{v}(x) = \overline{v(-x)}$ . Therefore, in contrast to [BS22], a full Taylor expansion does not ask more than one derivative in space on the initial data. In fact,  $T_0 \cdot T_{1,1}$  is the term in  $\mathcal{G}_k^0$  to compute which requires the most regularity on the initial data, namely a factor of  $k^2$  for a first order approximation, and which corresponds in the end to only one derivative due to the previously mentioned convolution structure. In most of the examples in this paper, we end up by considering only full Taylor expansions, as is made in (4.26). This is however not always the case in a general setting. We give an example of a system of PDEs where a more careful resonance analysis (c.f. (4.23)) is needed to obtain an approximation at low regularity. Let us for instance consider the following system:

$$i\partial_t u + \Delta u = |u|^2 u, \quad u(0) = v_1,$$
  
 $\partial_t v + \partial_x^p v = |u|^2 u, \quad v(0) = v_2.$ 
(4.27)

Then depending on the value of p, we may need a resonance based approach. Iterating Duhamel's formula, we get the following oscillatory integrals:

$$I = \int_0^\tau e^{-\xi(ik)^p} e^{-i\xi k_1^2} e^{i\xi k_2^2} e^{i\xi k_3^2} d\xi, \quad k = -k_1 + k_2 + k_3$$

If we assume that  $p \ge 3$ , then  $k^p$  is clearly the dominant term. Let us compare the local errors depending on the choice of approximation. Using a resonance based discretisation we obtain that

— Resonance scheme:

$$I = \int_0^\tau e^{-\xi(ik)^p} d\xi + \mathcal{O}(\xi^2 P_1), \quad P_1 = -k_1^2 + k_2^2 + k_3^2$$

whereas a classical discretisation leads to

Classical integrators:

$$I = \xi + \mathcal{O}(\xi^2 P_2), \quad P_2 = k^p - k_1^2 + k_2^2 + k_3^2.$$

As soon as  $p \ge 3$ , we ask for at least three derivatives on the initial data for a classical integrator (due to the error term  $\mathcal{O}(\xi^2 P_2)$ ) while only requiring two derivatives for the resonance based scheme (thanks to the improved local error term  $\mathcal{O}(\xi^2 P_1)$ ).

#### 4.1.2 Outline of the paper

Let us give a short review of the content of this paper. In Section 4.2, we introduce the combinatorial structures needed for describing oscillatory integrals coming from the iteration of the Duhamel's formula (4.9). We recall the framework introduced in [BS22] by first defining a suitable vector space of decorated trees  $\hat{\mathcal{T}}$  and decorated forests  $\hat{\mathcal{H}}$ . Then, we consider approximated decorated trees and forest which carry an extra decoration r at the root. It corresponds to the order of the approximation of the oscillatory integrals. The main novelty in this section are paired decorated forests which are specific decorated forests with extra constraints. They reflect the computation of the second moment of oscillatory integrals: pairing among the leaves of two trees correspond to the Wick formula applied to product of Gaussian random variables. In Section 4.3, we construct the approximation of the iterated integrals given by the character  $\Pi : \mathcal{H} \to \mathcal{C}$  (see (4.33)) through the character  $\Pi^n : \mathcal{H} \to \mathcal{C}$  (see (4.34)) where *n* is the a priori regularity assumed on the initial data *v*. The approximation  $\Pi^n$  is given via a recursive construction involving the operator  $\mathcal{K}$  given in Definition 4.3.1. In the definition, we are performing a full Taylor expansion which can be used in the examples mentioned in this paper. It also provides the optimal regularity. In the general case, the resonance approach could be needed. The local error analysis which is the error estimate on the difference between  $\Pi$  and its approximation  $\Pi^n$  is given in Theorem 4.3.6. It is computed via a recursive definition (see Definition 4.3.5) with Taylor remainders of  $\Pi^n$  given in Lemma 4.3.4. Let us mention that Definition 4.3.5 is more involved as the one given in [BS22], as the pairing among the leaves required to be more precise in the estimation of the local error.

In Section 4.4, we introduce truncated series of decorated trees that solve up to order r equation (4.1). Then, we compute the second moment of this series in Proposition 4.4.2 by using Wick's formula (see Proposition 4.4.1). This is where we rely on paired decorated forests for describing the new series obtained. From this series, we built another paired decorated forests series by replacing the character  $\Pi$  by its approximation  $\Pi^n$ . Then, we can write the general scheme (4.49). In the end, we compute its local error structure (see Theorem 4.4.3) based on the local error between  $\Pi$  and  $\Pi^n$  for each paired decorated forest that appears in the expansion of the scheme.

In Section 4.5, we illustrate our general framework on two fundamental examples: the nonlinear Schrödinger (NLS) and the Korteweg–de Vries (KdV) equation.

Acknowledgements. This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 850941). Y. B. thanks the Max Planck Institute for Mathematics in the Sciences (MiS) in Leipzig for supporting his research via a long stay in Leipzig from January to June 2022 where this work was written. Y. B. is funded by the ANR via the project LoRDeT (Dynamiques de faible régularité via les arbres décorés) from the projects call T-ERC\_STG. Y. A. B. thanks ICERM, and the Simons Fondation for their support, as well as the Max Planck Institute for Mathematics in the Sciences (MiS) in Leipzig for a short stay when this project was started.

## 4.2 Decorated tree based structures for oscillatory iterated integrals

In this section, we introduce the combinatorial structure for describing oscillatory integrals that stem from Duhamel's formula (4.9). We follow the formalism of decorated trees given in [BS22] which has been used for a low regularity approximation of (4.1). In order to describe  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$  we need to introduce a new structure: paired decorated forests which are decorated forests in the sense of [BS22] satisfying extra constraints on the decorations that encode some pairings. These pairings are coming from the Wick formula used for computing  $\mathbb{E}(|u_k(t,v^{\eta})|^2)$ . Such paired structures have been used in wave turbulence theory in more specific set-ups (see [DH21, DH23, ACG21, ST21]).

#### 4.2.1 Decorated trees

We recall briefly the structure of decorated trees introduced in [BS22, Sec. 2]. We assume a finite set  $\mathfrak{L}$  and frequencies  $k_1, ..., k_m \in \mathbb{Z}^d$ . The set  $\mathfrak{L}$  parametrizes a set of differential operators with constant coefficients, whose symbols are given by the polynomials  $(P_{\mathfrak{t}})_{\mathfrak{t}\in\mathfrak{L}}$ . We define the set of decorated trees  $\hat{\mathcal{T}}$  as elements of the form  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}} = (T,\mathfrak{n},\mathfrak{o},\mathfrak{e})$  where

- T is a non-planar rooted tree with root  $\rho_T$ , node set  $N_T$  and edge set  $E_T$ . We denote the leaves of T by  $L_T$ . T must also be a planted tree which means that there is only one edge connecting the root to the rest of the tree.
- the map  $\mathfrak{e}: E_T \to \mathfrak{L} \times \{0, 1\}$  encodes edge decorations. The set  $\{0, 1\}$  encodes the action of taking the conjugate, and determines the sign of the frequencies at the top of this edge. Namely, we have that 1 corresponds to a conjugate and to multiplying by (-1) the frequency on the node above and adjacent to this edge.
- the map  $\mathfrak{n} : N_T \setminus \{\varrho_T\} \to \mathbf{N}$  encodes node decorations. For every inner node v, this map encodes a monomial of the form  $\xi^{\mathfrak{n}(v)}$  where  $\xi$  is a time variable.
- the map  $\boldsymbol{o}: N_T \setminus \{\varrho_T\} \to \mathbf{Z}^d$  also encodes node decorations. These decorations are frequencies that satisfy for every inner node u:

$$(-1)^{\mathfrak{p}(e_u)}\mathfrak{o}(u) = \sum_{e=(u,v)\in E_T} (-1)^{\mathfrak{p}(e)}\mathfrak{o}(v)$$
(4.28)

where  $\mathfrak{e}(e) = (\mathfrak{t}(e), \mathfrak{p}(e))$  and  $e_u$  is the edge outgoing u of the form (v, u). From this definition, one can see that the node decorations at the leaves  $(\mathfrak{o}(u))_{u \in L_T}$  determine the decoration of the inner nodes. One can call this identity Kirchoff's law. We assume that the node decorations at the leaves are linear combinations of the  $k_i$  with coefficients in  $\{-1, 0, 1\}$ . In applications, the leaves decoration are either  $k_i$  or  $-k_i$ , the term  $-k_i$  is coming from a pairing with the Wick formula.

- we assume that the root of T has no decoration.

When the node decoration  $\mathfrak{n}$  is zero, we will denote the decorated trees  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}$  as  $T_{\mathfrak{e}}^{\mathfrak{o}} = (T,\mathfrak{o},\mathfrak{e})$ . The set of decorated trees satisfying such a condition is denoted by  $\hat{\mathcal{T}}_0$ . We set  $\hat{H}$  (resp.  $\hat{H}_0$ ) the (unordered) forests composed of trees in  $\hat{\mathcal{T}}$  (resp.  $\hat{\mathcal{T}}_0$ ) with linear spans  $\hat{\mathcal{H}}$  and  $\hat{\mathcal{H}}_0$ . The forest product is denoted by  $\cdot$ , the empty forest by **1**. Elements in  $\hat{\mathcal{T}}$  are abstract representation of iterated time integrals and elements in  $\hat{H}$  are product of them.

We now introduce how one can represent uniquely decorated trees by using symbolic notations. We denote by  $\mathcal{I}_o$ , an edge decorated by  $o = (\mathfrak{t}, \mathfrak{p}) \in \mathfrak{L} \times \{0, 1\}$ . We introduce the operator  $\mathcal{I}_o(\lambda_k^\ell \cdot) : \hat{\mathcal{H}} \to \hat{\mathcal{H}}$  that merges all the roots of the trees composing the forest into one node decorated by  $(\ell, k) \in \mathbf{N} \times \mathbf{Z}^d$ . The new decorated tree is then grafted via an edge decorated by o onto a new root with no decoration. If the condition (4.28) is not satisfied on the argument then  $\mathcal{I}_o(\lambda_k^\ell \cdot)$  gives zero. If  $\ell = 0$ , then the term  $\lambda_k^\ell$  is denoted by  $\lambda_k$  as a short hand notation for  $\lambda_k^0$ . We have choosen to put no decorations at the root in order to define the operator  $\mathcal{I}_o(\lambda_k^\ell \cdot)$ . that will assign the decoration k. For us, these decorated trees represent oscillatory iterated integrals and there is no need for having a decoration at the root for encoding them.

The forest product between  $\mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1)$  and  $\mathcal{I}_{o_2}(\lambda_{k_2}^{\ell_2}F_2)$  is given by:

$$\mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1)\mathcal{I}_{o_2}(\lambda_{k_2}^{\ell_2}F_2) := \mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1) \cdot \mathcal{I}_{o_2}(\lambda_{k_2}^{\ell_2}F_2)$$

Any decorated tree T is uniquely represented as

$$T = \mathcal{I}_o(\lambda_k^{\ell} F), \quad F \in \hat{H}.$$

Given an iterated integral, its size is given by the number of integrations in time. Therefore, we suppose given a subset  $\mathfrak{L}_+$  of  $\mathfrak{L}$  that encodes edge decorations which correspond to time integrals that we have to approximate.

**Example 14** We illustrate the definitions introduced above with decorated trees coming from the NLS equation. We consider the following decorated tree:

$$T = \mathcal{I}_{(\mathfrak{t}_1,0)} \left( \lambda_k \mathcal{I}_{(\mathfrak{t}_2,0)} \left( \lambda_k \mathcal{I}_{(\mathfrak{t}_1,1)} (\lambda_{k_1}) \mathcal{I}_{(\mathfrak{t}_1,0)} (\lambda_{k_2}) \mathcal{I}_{(\mathfrak{t}_1,0)} (\lambda_{k_3}) \right) \right) =$$

where  $k = -k_1 + k_2 + k_3$ ,  $\mathfrak{L} = \{\mathfrak{t}_1, \mathfrak{t}_2\}$ ,  $\mathfrak{L}_+ = \{\mathfrak{t}_2\}$ ,  $P_{\mathfrak{t}_1}(\lambda) = -\lambda^2$  and  $P_{\mathfrak{t}_2}(\lambda) = \lambda^2$ . For the graphical notation, we put the frequencies decorations only on the leaves that determine those on the inner nodes. An edge | (resp.  $\ddagger$ ) corresponds to a decoration  $(\mathfrak{t}_1, 0)$  (resp.  $(\mathfrak{t}_1, 1)$ ) and the operator associated is  $e^{itP_{\mathfrak{t}_1}(k)} = e^{-itk^2}$  (resp.  $e^{-itP_{\mathfrak{t}_1}(k)} = e^{itk^2}$ ), while an edge | (resp. |) corresponds to a decoration  $(\mathfrak{t}_2, 0)$  (resp.  $(\mathfrak{t}_2, 1)$ ) associated to the integral  $-i\int_0^{\tau} e^{i\xi P_{\mathfrak{t}_2}(k)} \cdots d\xi = -i\int_0^{\tau} e^{i\xi k^2} \cdots d\xi$  (resp.  $-i\int_0^{\tau} e^{-i\xi P_{\mathfrak{t}_2}(k)} \cdots d\xi = -i\int_0^{\tau} e^{-i\xi k^2} \cdots d\xi$ ). Therefore, T is an abstract version of the following integral:

$$-ie^{-i\tau k^2} \int_0^\tau e^{i\xi k^2} e^{i\xi k_1^2} e^{-i\xi k_2^2} e^{-i\xi k_3^2} d\xi$$

The next combinatorial structure, we recall from [BS22] is abstract versions of a discretization of an oscillatory integral. We denote by  $\mathcal{T}$  the set of decorated trees  $T_{\mathfrak{e},r}^{\mathfrak{n},\mathfrak{o}} = (T,\mathfrak{n},\mathfrak{o},\mathfrak{e},r)$  where

$$- T^{\mathfrak{n},\mathfrak{o}}_{\mathfrak{e}} \in \hat{\mathcal{T}}$$

— The decoration of the root is given by  $r \in \mathbf{Z}, r \geq -1$  such that

$$r+1 \ge \deg(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}) \tag{4.29}$$

where deg is defined recursively by

$$\deg(\mathbf{1}) = 0, \quad \deg(F_1 \cdot F_2) = \max(\deg(F_1), \deg(F_2))$$
$$\deg(\mathcal{I}_{(\mathfrak{t},\mathfrak{p})}(\lambda_k^{\ell}F_1)) = \ell + \mathbf{1}_{\{\mathfrak{t}\in\mathcal{L}_+\}} + \deg(F_1)$$

where  $F_1, F_2$  are forests composed of trees in  $\mathcal{T}$ . The quantity  $\deg(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}})$  is the maximum number of edges with type in  $\mathfrak{L}_+$ , corresponding to time integrations, and of node decorations  $\mathfrak{n}$  lying on the same path from one leaf to the root.

We call decorated trees in  $\mathcal{T}$  approximated decorated trees. We allow in the definition of the approximated decorated trees a new decoration at the root r, which corresponds to the order of the approximation. We now define the symbol  $\mathcal{I}_o^r(\lambda_k^\ell): \mathcal{H} \to \mathcal{H}$ , which plays the same role as the previously defined symbol  $\mathcal{I}_o^r(\lambda_k^\ell)$ , with the added adjunction of the decoration r which constrains the time-approximations to be of order r. We now define a projection operator  $\mathcal{D}^r$  which depends on the order r of the approximation and which is used during the construction of the numerical schemes in order to only retain the terms of order at most r. We define the map  $\mathcal{D}^r: \hat{\mathcal{H}} \to \mathcal{H}$  which assigns to the root of a decorated tree

a decoration r and performs the projection along the identity (4.29). It is given by

$$\mathcal{D}^{r}(\mathbf{1}) = \mathbf{1}_{\{0 \le r+1\}}, \quad \mathcal{D}^{r}\left(\mathcal{I}_{o}(\lambda_{k}^{\ell}F)\right) = \mathcal{I}_{o}^{r}(\lambda_{k}^{\ell}F)$$
(4.30)

and we extend it multiplicatively to any forest in  $\hat{\mathcal{H}}$ .

**Example 15** We illustrate the action of the map  $\mathcal{D}^r$  on the decorated tree T introduced in Example 14. First, we notice that:

$$\deg(T) = 1, \quad \mathcal{D}^{r}(T) = 0, r > 1.$$

Graphically, we just add r at the root:



#### 4.2.2 Paired decorated forests

Our aim is to discretise the second moment of  $u_k^r(\tau, v^\eta)$  which can be expanded as a tree series. By applying the Wick product, one gets products of iterated integrals whose initial data  $v^\eta$  are paired. At the level of the trees, this correspond to a pairing on the leaves. Therefore, one has to introduce a more rigid structure for encoding these new terms. We define  $\hat{\mathcal{G}}$  as the set of paired decorated forests F such that:

— F is a forest containing solely two trees denoted by T and  $\overline{T}$ .

— One has the following assumptions on the decoration  $\mathfrak{o}$ :

$$(-1)^{\mathfrak{p}(e_{\varrho_T})}\mathfrak{o}(u_T) = (-1)^{\mathfrak{p}(e_{\varrho_{\bar{T}}})}\mathfrak{o}(u_{\bar{T}})$$

$$(4.31)$$

where  $u_T$  (resp.  $u_{\bar{T}}$ ) are the nodes in T (resp.  $\bar{T}$ ) connected to  $\varrho_T$  (resp.  $\varrho_{\bar{T}}$ ). The decoration  $\mathfrak{p}(e_{\varrho_T})$  corresponds to the second decoration component on the edge  $e_{\varrho_T}$  connecting the root  $\varrho_T$  to  $u_T$ .

— For any leaf  $u \in L_F$ , there exists exactly one leaf  $v \in L_F$  such that

$$(-1)^{\mathfrak{p}(e_u)}\mathfrak{o}(u) = (-1)^{\mathfrak{p}(e_v)+1}\mathfrak{o}(v), \quad \text{if } \{u,v\} \subset L_T \text{ or } \{u,v\} \subset L_{\bar{T}}$$

$$(-1)^{\mathfrak{p}(e_u)}\mathfrak{o}(u) = (-1)^{\mathfrak{p}(e_v)}\mathfrak{o}(v), \quad \text{otherwise}$$

$$(4.32)$$

where  $e_u$  (resp.  $e_v$ ) is the edge connecting u (resp. v) to the rest of the tree, and  $\mathfrak{p}(e) \in \{0, 1\}$  is the second component of the edge decoration associated to e.

Approximated paired decorated forests F are described by

$$F = \mathcal{D}^r(\bar{F}), \quad r \in \mathbf{Z}, \ \bar{F} \in \hat{\mathcal{G}}.$$

We denote this set as  $\mathcal{G}$  and by  $\mathcal{G}_k^r$  decorated paired forests having at most r edges with type in  $\mathfrak{L}_+$  and both sides of the identity (4.31) are equal either to k or -k.

Example 16 We provide an example of paired decorated forests:

$$F = T_1 \cdot T_2, \quad T_1 = \checkmark, \quad T_2 = \checkmark$$

Let u denote the single leaf in the tree  $T_1$ , and let  $\tilde{u}, \tilde{v}, v$  denote the leaves, from left to right, in the tree  $T_2$ . We have that  $\mathfrak{o}(e_u) = \mathfrak{o}(e_v) = k$ , and  $\mathfrak{o}(e_{\tilde{u}}) = k_1 = \mathfrak{o}(e_{\tilde{v}})$ . Moreover,  $\mathfrak{p}(e_u) = \mathfrak{p}(e_{\tilde{v}}) = \mathfrak{p}(e_u) = 0$  and  $\mathfrak{p}(e_{\tilde{u}}) = 1$  since the edge  $e_{\tilde{u}}$  is a dotted brown edge, which encodes that the frequency is preceded by a negative sign. Hence, from (4.32) we read:

$$(-1)^{1}k_{1} = (-1)^{1}k_{1}, \quad (-1)^{0}k = (-1)^{0}k_{1}$$

For the first condition (4.31), one has  $\mathfrak{p}(e_{\varrho_{T_1}}) = 0$  and  $\mathfrak{p}(e_{\varrho_{T_2}}) = 0$ . Therefore, one similarly obtains

$$(-1)^0 k = (-1)^0 k.$$

**Remark 4.2.1** In the above definition, paired decorated forests are formed of only two trees with pairings among the leaves. The pairings among the leaves reflect the computation of the second moment of products of Gaussian random variables, dictated by Wick's theorem. One can easily generalise this definition depending on one needs. Indeed, first instead of computing second order moments, one can look at higher order moments, namely moments of order 2p, and consider forests consisting of 2p trees. The pairing on the leaves between the 2p trees would follow the same condition (4.31). Secondly, one can consider non-Gaussian random initial conditions together with the use of cumulants to compute second moments, see Remark 4.1.2. In this case, the above definition can be replaced by pairings between bigger clusters of leaves and one has to change (4.32) accordingly, by including more than two leaves.

### 4.3 Discretizing oscillatory iterated integrals

In this section, we recall the definition of the map  $\Pi : \hat{\mathcal{H}} \to \mathcal{C}$  that interprets decorated forests as oscillatory integrals and its low regularity discretization  $\Pi^n$  introduced both in [BS22]. The main simplification in comparison to the resonance method exposed is the definition of  $\mathcal{K}_{o_2}^{k,r}$  which is now performing the full Taylor expansion. This is due to the convolution structures observed in the computation of  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  (see remark 4.3.2). This will be enough for the examples treated in this paper but in general a resonance analysis could be needed. The local error analysis relies on the error introduced by the operator  $\mathcal{K}_{o_2}^{k,r}$  via Lemma 4.3.4. This error is propagated through a recursive definition of the error namely  $\mathcal{L}_{low}^r(F, n)$ . This step is more involved in comparison to [BS22]. Indeed, the error introduced by the approximation of a product of iterated integrals needs a more careful treatment, see the new formulation in Definition 4.3.5.

We adopt in this section the following notations: an element of  $\mathfrak{L}_+$  (resp.  $\mathfrak{L}_+ \times \{0, 1\}$ ) is denoted by  $\mathfrak{t}_2$  (resp.  $o_2$ ) and an element of  $\mathfrak{L} \setminus \mathfrak{L}_+$  (resp.  $\mathfrak{L} \setminus \mathfrak{L}_+ \times \{0, 1\}$ ) is denoted by  $\mathfrak{t}_1$  (resp.  $o_1$ ). We define  $\mathcal{C}$  as the space of functions of the form  $z \mapsto \sum_j Q_j(z) e^{izP_j(k_1,...,k_m)}$  where the  $Q_j(z)$  are polynomials in z and the  $P_j$  are polynomials in  $k_1,...,k_n \in \mathbb{Z}^d$ . The  $Q_j$  may also depend on  $k_1,...,k_m$ . Equipped with the pointwise product  $\mathcal{C}$  is an algebra. Iterated integrals and their discretisation will be characters from decorated forests into  $\mathcal{C}$ . For a character  $g: \hat{\mathcal{H}} \to \mathcal{C}$ , one has:

$$g(F \cdot \overline{F}) = g(F)g(\overline{F}), \quad F, \overline{F} \in \hat{\mathcal{H}}.$$

We define the character  $\Pi : \hat{\mathcal{H}} \to \mathcal{C}$  by

$$\Pi \left( F \cdot \bar{F} \right)(\tau) = (\Pi F)(\tau)(\Pi \bar{F})(\tau),$$

$$\Pi \left( \mathcal{I}_{o_1}(\lambda_k^{\ell} F) \right)(\tau) = e^{i\tau P_{o_1}(k)} \tau^{\ell}(\Pi F)(\tau),$$

$$\Pi \left( \mathcal{I}_{o_2}(\lambda_k^{\ell} F) \right)(\tau) = -i |\nabla|^{\alpha}(k) \int_0^{\tau} e^{i\xi P_{o_2}(k)} \xi^{\ell}(\Pi F)(\xi) d\xi,$$
(4.33)

where  $F, \overline{F} \in \hat{\mathcal{H}}$ . We have used the short hand notation  $P_{o_1}$  given for  $o_1 = (\mathfrak{t}_1, \mathfrak{p}_1)$  by:

$$P_{o_1}(k) = (-1)^{\mathfrak{p}_1} P_{\mathfrak{t}_1}((-1)^{\mathfrak{p}_1} k).$$

For the discretisation, we consider a new family of characters defined now on  $\mathcal{H}$  and parametrized by  $n \in \mathbf{N}$ :

$$\Pi^{n}\left(F\cdot\bar{F}\right)(\tau) = (\Pi^{n}F)(\tau)\left(\Pi^{n}\bar{F}\right)(\tau), \quad (\Pi^{n}\lambda^{\ell})(\tau) = \tau^{\ell},$$

$$\left(\Pi^{n}\mathcal{I}_{o_{1}}^{r}(\lambda_{k}^{\ell}F)\right)(\tau) = \tau^{\ell}e^{i\tau P_{o_{1}}(k)}\left(\Pi^{n}\mathcal{D}^{r-\ell}(F)\right)(\tau),$$

$$\left(\Pi^{n}\mathcal{I}_{o_{2}}^{r}(\lambda_{k}^{\ell}F)\right)(\tau) = \mathcal{K}_{o_{2}}^{k,r}\left(\Pi^{n}\left(\lambda^{\ell}\mathcal{D}^{r-\ell-1}(F)\right),n\right)(\tau).$$
(4.34)

In the sequel, we will use the short hand notations:

$$\Pi^{n,r} = \Pi^n \mathcal{D}^r, \quad (\bar{\Pi}T)(\tau) = \overline{(\Pi T)}(\tau), \quad (\bar{\Pi}^{n,r}T)(\tau) = \overline{(\Pi^{n,r}T)}(\tau).$$

We provide the definition of the approximation operator  $\mathcal{K}$  below, when one performs the full Taylor expansion of the oscillatory integrals.

**Definition 4.3.1** Assume that  $G: \xi \mapsto \xi^q e^{i\xi P(k_1,...,k_m)}$  where P is a polynomial in the frequencies  $k_1, ..., k_m$  and let

 $o_2 = (\mathfrak{t}_2, p) \in \mathfrak{L}_+ \times \{0, 1\}$  and  $r \in \mathbb{N}$ . Let k be a linear map in  $k_1, \dots, k_m$  using coefficients in  $\{-1, 0, 1\}$ . We set

$$\mathcal{K}_{o_2}^{k,r}(G,n)(\tau) = -i|\nabla|^{\alpha}(k) \sum_{\ell \le r-q} \frac{\tau^{\ell+q+1}}{\ell!(\ell+q+1)} (iP_{o_2}(k) + iP(k_1,...,k_m))^{\ell}.$$
(4.35)

**Remark 4.3.2** The definition (4.35) for  $\mathcal{K}$  is quite simple and encodes the approximation of the oscillatory integral

$$-i|\nabla|^{\alpha}(k)\int_{0}^{\tau}e^{i\xi P_{o_{2}}(k)}G(\xi)d\xi$$

by a full Taylor expansion of the operator appearing in the integrand. This approximation does not require a careful resonance based analysis as was the case in [BS22, Def. 3.1]. This is due to the fact that we are not approximating  $u_k(\tau)$  but its second moment  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$ . Therefore, our local error terms involve products of two Fourier coefficients. This allows us to repartition half of the regularity on each of the Fourier coefficients, thereby obtaining a similar low-regularity local error structure as in [BS22] without requiring a more delicate analysis. Indeed, to each of the frequencies  $k_1, ..., k_m$  we now consider terms of the form  $v_{k_i}^2$ ,  $|v_{k_i}|$  or  $\bar{v}_{k_i}v_{k_i}$ , instead of  $v_{k_i}$  or  $\bar{v}_{k_i}$ . The new quadratic terms can be interpreted as convolutions or are better summable. In the case of the convolution, the two following terms ask for the same regularity:

$$k^2 v_k^2 \equiv \partial_x v * \partial_x v, \quad k v_k \equiv \partial_x v$$

where \* is the spatial convolution. One can see the difference between the discretization presented here and the one in [BS22] by considering the following oscillatory integral coming from NLS:

$$I = \int_0^\tau e^{-i\xi k^2} e^{-i\xi k_1^2} e^{i\xi k_2^2} e^{i\xi k_3^2} d\xi, \quad k = -k_1 + k_2 + k_3.$$

The idea of the resonance analysis is to split the integrand into the dominant and lower order parts (cf.(4.23))

$$\mathcal{L} = k^2 + k_1^2 - k_2^2 - k_3^2 = \mathcal{L}_{ ext{dom}} + \mathcal{L}_{ ext{low}}$$
  
 $\mathcal{L}_{ ext{dom}} = 2k_1^2, \quad \mathcal{L}_{ ext{low}} = -k_1(k_2 + k_3) + k_2k_3$ 

and to Taylor expand only the term of lower degree  $e^{i\xi \mathcal{L}_{low}}$ . This yields the following discretisation

$$I = \int_0^\tau e^{-2i\xi k_1^2} (1 + \mathcal{O}(\xi P_1)) d\xi, \quad P_1 = k_1(k_2 + k_3) - k_2 k_3$$

with

$$I = \int_0^\tau e^{-2i\xi k_1^2} d\xi + \mathcal{O}(\tau^2 P_1).$$

In contrast, a full classical Taylor expansion gives

$$I = \int_0^\tau e^{-2i\xi k_1^2} d\xi + \mathcal{O}(\tau^2 P_2), \quad P_2 = -2k_1^2 + k_1(k_2 + k_3) - k_2k_3$$

The difference between the two approximations is that the second one asks for two derivatives instead of one. In our setting, the latter is, however, absorbed by the convolution of initial data.

**Remark 4.3.3** For many equations, the main difference between resonance based and classical integrators lies in the gain of one derivative (at order one). Therefore, there is no need to carry out a resonance based discretization for  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  due to the improved resonance structure of the latter. Resonance based schemes are, however, needed if the gap between dominant and lower parts of the resonance is bigger. Then, the full definition [BS22, Def. 3.1] is required. This can happen when one looks at systems of dispersive PDEs as for instance system (4.27).

**Example 17** With the aid of (4.33), one can compute recursively the following oscillatory integrals arising in the cubic NLS equation (4.5)

$$(\Pi \stackrel{(k_2)}{\downarrow})(\tau) = (\Pi \mathcal{I}_{(t_1,0)}(\lambda_{k_2}))(\tau) = e^{-i\tau k_2^2},$$
$$(\Pi \stackrel{(k_1)}{\bullet})(\tau) = (\Pi \mathcal{I}_{(t_1,1)}(\lambda_{k_1}))(\tau) = e^{i\tau k_1^2},$$

$$(\Pi^{(k_1)}, \overset{(k_2)}{\longrightarrow}, \overset{(k_3)}{\longrightarrow})(\tau) = (\Pi^{(k_1)}, (\tau) (\Pi^{(k_2)}, (\tau) (\Pi^{(k_3)}, (\tau)) = e^{i\tau(k_1^2 - k_2^2 - k_3^2)})(\tau) = -i \int_0^\tau e^{is(-k_1 + k_2 + k_3)^2} e^{is(k_1^2 - k_2^2 - k_3^2)} ds.$$

For a second-order order approximation of the above integral, one has the following discretisation which in the end asks for two derivatives on the initial data,

$$(\Pi^{2,1})(\tau) = -i\tau + 2\tau^2 (k_1^2 - k_1(k_2 + k_3) + k_2k_3).$$

We recall [BS22, Lemma 3.3] for a function G as given in Definition 4.3.1.

**Lemma 4.3.4** We keep the notations of Definition 4.3.1. We suppose that  $q \leq r$  then one has

$$-i|\nabla|^{\alpha}(k)\int_{0}^{\tau}\xi^{q}e^{i\xi\left(P_{o_{2}}(k)+P(k_{1},\dots,k_{m})\right)}d\xi-\mathcal{K}_{o_{2}}^{k,r}(G,n)(\tau)=\mathcal{O}(\tau^{r+2}R_{o_{2},n}^{k,r}(G))$$

where  $R_{o_2,n}^{k,r}(G)$  depends on  $n, \alpha$  and the frequencies  $k_1, ..., k_m$ . When  $\mathcal{K}_{o_2}^{k,r}$  performs a full Taylor expansions, it is given by

$$(P_{o_2}(k) + P(k_1, ..., k_m))^{r+1-q} k^{\alpha}.$$
(4.36)

Otherwise, one considers the resonance analysis and decomposes  $P_{o_2} + P$  into

$$P_{o_2} + P = \mathcal{L}_{dom} + \mathcal{L}_{low}$$

and one has

$$R^{k,r}_{o_2,n}(G) = k^{\bar{n}}, \quad \bar{n} = \max(n, \alpha + \deg(\mathcal{L}^{r-q+1}_{low})).$$

We define inductively the term  $(\Pi \cdot)_0$  on decorated forests:

$$\begin{split} &\Pi \left( F \cdot \bar{F} \right)_0 = (\Pi F)_0 (\Pi \bar{F})_0, \quad \Pi \left( \mathcal{I}_{o_1} (\lambda_k^{\ell} F) \right)_0 = (\Pi F)_0 \\ &\Pi \left( \mathcal{I}_{o_2} (\lambda_k^{\ell} F) \right)_0 = |\nabla|^{\alpha} (k) (\Pi F)_0. \end{split}$$

For every decorated forest, there exists  $c_F$  independent of the frequencies  $k_1, ..., k_m$  such that:

$$|(\Pi F)(\tau)| \le c_F \tau^{\hat{n}_+(F)} (\Pi F)_0$$

where  $\hat{n}_+(F) = n_+(F) + \sum_{u \in N_F} \mathfrak{n}(u)$ ,  $n_+(F)$  gives the number of edges with decoration  $o_2 = (\mathfrak{t}_2, \mathfrak{p}_2)$  with  $\mathfrak{t}_2 \in \mathfrak{L}_+$ . In the sequel, we will use the following notation:  $\Pi^{n,r} = \Pi^n \mathcal{D}^r$ . Given a decorated forest, we set  $(\Pi^{n,r}F)_\ell$  to be the term of order  $\tau^\ell$  in  $(\Pi^{n,r}F)(\tau)$  in the sense that one has:

$$\mathcal{Q}_{\leq r+1}(\Pi^{n,r}F)(\tau) = \sum_{\hat{n}_+(F) \leq \ell \leq r+1} (\Pi^{n,r}F)_\ell(\tau)\tau^\ell$$

where each  $(\Pi^{n,r}F)_{\ell}(\tau)$  is bounded, by denoting the bound by  $(\Pi^{n,r}F)_{\ell}$ , we have  $\forall \tau$ ,  $(\Pi^{n,r}F)_{\ell}(\tau) \leq (\Pi^{r,n}F)_{\ell}$ .

**Definition 4.3.5** Let  $n \in \mathbf{N}, r \in \mathbf{Z}$ . We recursively define  $\mathcal{L}_{low}^{r}(\cdot, n)$  as

$$\mathcal{L}^r_{\text{low}}(F,n) = 1, \quad r < 0$$

Else

$$\mathcal{L}_{low}^{r}(\mathbf{1}, n) = 1, \quad \mathcal{L}_{low}^{r}(\mathcal{I}_{o_{1}}(\lambda_{k}^{\ell}F), n) = \mathcal{L}_{low}^{r-\ell}(F, n)$$
$$\mathcal{L}_{low}^{r}(F_{1} \cdot F_{2}, n) = (\Pi F_{2})_{0} \mathcal{L}_{low}^{r-\hat{n}_{+}(F_{2})}(F_{1}, n)$$
$$+ \sum_{\ell \leq r+1-\hat{n}_{+}(F_{2})} (\Pi^{n, r-\hat{n}_{+}(F_{2})}F_{1})_{\ell} \mathcal{L}_{low}^{r-\ell}(F_{2}, n)$$

$$\mathcal{L}_{low}^{r}(\mathcal{I}_{o_{2}}(\lambda_{k}^{\ell}F),n) = k^{\alpha}\mathcal{L}_{low}^{r-\ell-1}(F,n) + R_{o_{2},n}^{k,r}(\xi^{\ell}(\Pi^{n,r-\ell-1}F)(\xi)).$$

**Example 18** We illustrate Definition 4.3.5 on the following paired trees stemming from the NLS equation:

$$F_2 = T_1 \cdot T_2, \quad T_1 = \mathcal{I}_{o_1}(\lambda_k) = \begin{pmatrix} k \\ \bullet \end{pmatrix}, \quad T_2 = \begin{pmatrix} k \\ \bullet \end{pmatrix}, \quad (4.37)$$

(i)

where  $\bar{k} = -k$ . We have,

$$\mathcal{L}^{0}_{\text{low}}(T_1 \cdot T_2, 1) = (\Pi T_2)_0 \mathcal{L}^{-1}_{\text{low}}(T_1, 1) + (\Pi^{1, -1} T_1)_0 \mathcal{L}^{0}_{\text{low}}(T_2, 1),$$
(4.38)

where  $\Pi^{1,-1} = \Pi^1 \mathcal{D}^{-1} = \Pi^1$ ,  $(\Pi^{1,-1}T_1)_0 = e^{-i\tau k^2}$ ,  $\mathcal{L}^{-1}_{low}(T_1,1) = 1$  and  $(\Pi T_2)_0 = 1$ . Next, we have  $\mathcal{L}^0_{low}(T_2,1) = \mathcal{L}^0_{low}(G_2,1) + R^{k,0}_{o_2,1}(\Pi^{1,-1}G_2)$ ,

with  $T_2 = \mathcal{I}_{o_1}(\lambda_k \mathcal{I}_{o_2}(\lambda_k G_2))$ , and  $G_2 = \mathcal{I}_{\bar{o}_1}(\lambda_{-k})\mathcal{I}_{o_1}(\lambda_{-k_2})\mathcal{I}_{o_1}(\lambda_{k_2})$ . A quick computation allow to check that  $\mathcal{L}^0_{low}(G_2, 1)$  is bounded by

$$\mathcal{L}^{0}_{\text{low}}(\mathcal{I}_{\bar{o}_{1}}(\lambda_{-k}),1) + \mathcal{L}^{0}_{\text{low}}(\mathcal{I}_{o_{1}}(\lambda_{-k_{2}}),1) + \mathcal{L}^{0}_{\text{low}}(\mathcal{I}_{o_{1}}(\lambda_{k_{2}}),1)$$

and one has:

$$\mathcal{L}^{0}_{low}(\mathcal{I}_{\bar{o}_{1}}(\lambda_{-k}), 1) = \mathcal{L}^{0}_{low}(\mathcal{I}_{o_{1}}(\lambda_{-k_{2}}), 1) = \mathcal{L}^{0}_{low}(\mathcal{I}_{o_{1}}(\lambda_{k_{2}}), 1) = \mathcal{L}^{0}_{low}(\mathbf{1}, 1) = 1.$$

As in [BS22, Thm. 3.17], we get the following error for our approximation under the assumption:

$$\mathcal{Q}_{\leq r+1}\Pi^{n,r} = \mathcal{Q}_{\leq r+1}\Pi^{n,r'}, \quad r \leq r'.$$
(4.39)

**Theorem 4.3.6** For every forests  $F_1$  and  $F_2$ , one has,

$$(\Pi(F_1 \cdot F_2) - \mathcal{Q}_{\leq r+1}\Pi^{n,r}(F_1 \cdot F_2))(\tau) = \mathcal{O}(\tau^{r+2}\mathcal{L}^r_{low}(F_1 \cdot F_2, n))$$
(4.40)

where  $\Pi$  is defined in (4.33),  $\Pi^n$  is given in (4.34) and  $\Pi^{n,r} = \Pi^n \mathcal{D}^r$  satisfying (4.39).

*Proof.* The proof follows the same steps given in [BS22, Thm. 3.17]. However, one has to be more precise for dealing with products as now frequencies  $k_1, ..., k_m$  can appear several times. We proceed with the following decomposition:

$$\begin{aligned} \Pi(F_{1} \cdot F_{2})(\tau) &= \mathcal{Q}_{\leq r+1} \Pi^{n,r} (F_{1} \cdot F_{2})(\tau) \\ &= \left( \Pi F_{1} - \mathcal{Q}_{\leq r+1-\hat{n}_{+}(F_{2})} \Pi^{n,r-\hat{n}_{+}(F_{2})} F_{1} \right)(\tau) (\Pi F_{2})(\tau) \\ &+ \mathcal{Q}_{\leq r+1-\hat{n}_{+}(F_{2})} \left( \Pi^{n,r-\hat{n}_{+}(F_{2})} F_{1} \right)(\tau) (\Pi F_{2})(\tau) - \mathcal{Q}_{\leq r+1} \Pi^{n,r} (F_{1} \cdot F_{2})(\tau) \\ &= \left( \Pi F_{1} - \mathcal{Q}_{\leq r+1-\hat{n}_{+}(F_{2})} \Pi^{n,r-\hat{n}_{+}(F_{2})} F_{1} \right)(\tau) (\Pi F_{2})(\tau) \\ &+ \sum_{\ell \leq r+1-\hat{n}_{+}(F_{2})} \tau^{\ell} (\Pi^{n,r} F_{1})_{\ell}(\tau) \left( \Pi F_{2} - \mathcal{Q}_{\leq r+1-\ell} \Pi^{n,r-\ell} F_{2} \right)(\tau) \end{aligned}$$

where we have used the following identities

$$\mathcal{Q}_{\leq r+1-\hat{n}_{+}(F_{2})} \Big( \Pi^{n,r-\hat{n}_{+}(F_{2})} F_{1} \Big)(\tau) = \sum_{\ell \leq r+1-\hat{n}_{+}(F_{2})} \tau^{\ell} (\Pi^{n,r-\hat{n}_{+}(F_{2})} F_{1})_{\ell}(\tau)$$
$$\mathcal{Q}_{\leq r+1} (\Pi^{n,r}(F_{1} \cdot F_{2}))(\tau) = \sum_{\ell \leq r+1-\hat{n}_{+}(F_{2})} \tau^{\ell} (\Pi^{n,r} F_{1})_{\ell}(\tau) \mathcal{Q}_{\leq r+1-\ell} (\Pi^{n,r-\ell} F_{2})(\tau)$$

and the fact that by Assumption (4.39), one has:

$$(\Pi^{n,r-\hat{n}_+(F_2)}F_1)_\ell(\tau) = (\Pi^{n,r}F_1)_\ell(\tau).$$

Then, we apply recursively the bounds on  $F_1$  and  $F_2$  to get:

$$\left( \Pi F_1 - \mathcal{Q}_{\leq r+1-\hat{n}_+(F_2)} \Pi^{n,r-\hat{n}_+(F_2)} F_1 \right) (\tau) = \mathcal{O} \left( \tau^{r-\hat{n}_+(F_2)+2} \mathcal{L}_{\text{low}}^{r-\hat{n}_+(F_2)} (F_1, n) \right)$$
$$\left( \Pi F_2 - \mathcal{Q}_{\leq r+1-\ell} \Pi^{n,r-\ell} F_2 \right) (\tau) = \mathcal{O} \left( \tau^{r-\ell+2} \mathcal{L}_{\text{low}}^{r-\ell} (F_2, n) \right)$$

which allows us to conclude. It remains to treat the case when  $F = \mathcal{I}_{o_2}(\lambda_k^{\ell} \bar{F})$ . In that case, one has:

$$\begin{aligned} (\Pi - \Pi^{n,r})(F)(\tau) &= -i |\nabla|^{\alpha}(k) \int_{0}^{\tau} \xi^{\ell} e^{i\xi P_{o_{2}}(k)} (\Pi - \Pi^{n,r-1-\ell})(\bar{F})(\xi) d\xi \\ &- i |\nabla|^{\alpha}(k) \int_{0}^{\tau} \xi^{\ell} e^{i\xi P_{o_{2}}(k)} (\Pi^{n,r-1-\ell}\bar{F})(\xi) d\xi - \mathcal{K}_{o_{2}}^{k,r} (\xi^{\ell}(\Pi^{n,r-1-\ell}\bar{F})(\xi))(\tau) \\ &= \int_{0}^{\tau} \mathcal{O} \Big( \xi^{r+1} k^{\alpha} \mathcal{L}_{low}^{r-1-\ell}(\bar{F},n) \Big) d\xi + \mathcal{O}(\tau^{r+2} R_{o_{2},n}^{k,r} \Big( \xi^{\ell}(\Pi^{n,r-1-\ell}\bar{F})(\xi) \Big) \Big) \\ &= \mathcal{O} \Big( \tau^{r+2} \mathcal{L}_{low}^{r}(F,n) \Big). \end{aligned}$$

where the term  $R_{o_2,n}^{k,r}(\xi^{\ell}(\Pi^{n,r-1-\ell}\bar{F})(\xi))$  is obtained by applying Lemma 4.3.4.

**Remark 4.3.7** An immediate consequence of Theorem 4.3.6, is that for a paired decorated forest  $F = T_1 \cdot T_2$ , one has

$$\left(\bar{\Pi}T_{1}\Pi T_{2} - \mathcal{Q}_{\leq r+1}\bar{\Pi}^{n,r}T_{1}\Pi^{n,r}T_{2}\right)(\tau) = \mathcal{O}\left(\tau^{r+2}\mathcal{L}^{r}_{low}(T_{1}\cdot T_{2},n)\right)$$
(4.41)

**Remark 4.3.8** Assumption 4.39 is satisfied by the examples covered in this work as we use mainly full Taylor expansions for describing the numerical scheme. In full generality, Assumption 4.39 does, however, not hold true (see also Remark 4.3.3) and one has to work with a two-parameter family in order to describe the local error. The identity (4.40) becomes:

$$(\Pi(F_1 \cdot F_2) - \mathcal{Q}_{\leq m} \Pi^{n,r}(F_1 \cdot F_2))(\tau) = \mathcal{O}\big(\tau^{r+2} \mathcal{L}^{r,m}_{\text{low}}(F_1 \cdot F_2, n)\big), \quad m \leq r+1$$

where  $\mathcal{L}_{low}^{r,m}(F_1 \cdot F_2, n)$  can be defined in a recursive way as in Definition 4.3.5. The main change occurs in the approximation of a decorated tree given by Lemma 4.3.4. It is changed into

$$- i |\nabla|^{\alpha}(k) \int_{0}^{\tau} \xi^{\ell} e^{i\xi P_{o_{2}}(k)} (\Pi^{n,r-1-\ell}\bar{F})(\xi) d\xi - \mathcal{Q}_{\leq m} \mathcal{K}_{o_{2}}^{k,r}(\xi^{\ell}(\Pi^{n,r-1-\ell}\bar{F})(\xi))(\tau) = \mathcal{O}(\tau^{m+1} R_{o_{2},n}^{k,r,m} \left(\xi^{\ell}(\Pi^{n,r-1-\ell}\bar{F})(\xi)\right))$$

The new quantity  $R_{o_2,n}^{k,r,m}$  is the new building block for recursively defining  $\mathcal{L}_{low}^{r,m}$ .

### 4.4 Low regularity scheme

The mild solution of (4.1) is given by Duhamel's formula

$$u(\tau, v^{\eta}) = e^{i\tau \mathcal{L}(\nabla)} v^{\eta} - i |\nabla|^{\alpha} e^{i\tau \mathcal{L}(\nabla)} \int_{0}^{\tau} e^{-i\xi \mathcal{L}(\nabla)} p(u(\xi), \bar{u}(\xi)) d\xi.$$

$$(4.42)$$

In the sequel, we will focus on nonlinearities of type

$$p(u,\bar{u}) = u^N \bar{u}^M. \tag{4.43}$$

It has been proven in [BS22, Prop. 4.3] that the following tree series expansion is the k-th Fourier coefficient of a solution of (4.42) up to order r + 1:

$$u_{k}^{r}(\tau, v^{\eta}) = \sum_{T \in \hat{\tau}_{0}^{r,k}(R)} \frac{\Upsilon^{p}(T)(v^{\eta})}{S(T)} (\Pi T)(\tau)$$
(4.44)

where

— For a decorated tree  $T_{\mathfrak{e}} = (T, \mathfrak{e})$  with only edge decorations, we define the symmetry factor  $S(T_{\mathfrak{e}})$  inductively by  $S(\mathbf{1})=1$ , while if T is of the form

$$\prod_{i,j} \mathcal{I}_{(\mathfrak{t}_{t_i},p_i)}(T_{i,j})^{\beta_{i,j}},$$

with  $T_{i,j} \neq T_{i,\ell}$  for  $j \neq \ell$ , then

$$S(T) := \left(\prod_{i,j} S(T_{i,j})^{\beta_{i,j}} \beta_{i,j}!\right).$$
(4.45)

We extend this definition to any tree  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}$  in  $\mathcal{T}$  by setting:

$$S(T^{\mathfrak{n},\mathfrak{o}}_{\mathfrak{e}}) := S(T_{\mathfrak{e}}).$$

— Then, we define the map  $\Upsilon^p(T)(v)$  for

$$T = \mathcal{I}_{(\mathfrak{t}_1, a)} \left( \lambda_k \mathcal{I}_{(\mathfrak{t}_2, a)}(\lambda_k \prod_{i=1}^n \mathcal{I}_{(\mathfrak{t}_1, 0)}(\lambda_{k_i} T_i) \prod_{j=1}^m \mathcal{I}_{(\mathfrak{t}_1, 1)}(\lambda_{\tilde{k}_j} \tilde{T}_j)) \right), \quad a \in \{0, 1\}$$

by

$$\Upsilon^{p}(T)(v) := \partial_{v}^{n} \partial_{\bar{v}}^{m} p_{a}(v, \bar{v}) \prod_{i=1}^{n} \Upsilon^{p}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{i}}T_{i}))(v)$$

$$\prod_{j=1}^{m} \Upsilon^{p}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{\tilde{k}_{j}}\tilde{T}_{j}))(v)$$
(4.46)

and

 $\Upsilon^p(\mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_k))(v):=v_k,\quad \Upsilon^p(\mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_k))(v):=\bar{v}_k.$ 

Above, we have used the notation:

$$p_0(v, \bar{v}) = p(v, \bar{v}), \quad p_1(v, \bar{v}) = \overline{p(v, \bar{v})}$$

In the sequel, we will use the following short hand notation:

$$\overline{\Upsilon^p(T)}(v) = \overline{\Upsilon}^p(T)(v).$$

We set

$$\begin{aligned} \hat{\mathcal{T}}_0(R) &= \{ \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_k \mathcal{I}_{(\mathfrak{t}_2,0)}(\lambda_k \prod_{i=1}^N T_i \prod_{j=1}^M \tilde{T}_j)), \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_k) \\ T_i &\in \hat{\mathcal{T}}_0(R), \ \tilde{T}_j \in \bar{\hat{\mathcal{T}}}_0(R), \ k \in \mathbf{Z}^d \} \\ \bar{\hat{\mathcal{T}}}_0(R) &= \{ \mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_k \mathcal{I}_{(\mathfrak{t}_2,1)}(\lambda_k \prod_{i=1}^N T_i \prod_{j=1}^M \tilde{T}_j)), \mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_k) \\ T_i &\in \bar{\hat{\mathcal{T}}}_0(R), \ \tilde{T}_j \in \bar{\hat{\mathcal{T}}}_0(R), \ k \in \mathbf{Z}^d \}. \end{aligned}$$

For a fixed  $k \in \mathbf{Z}^d$ , we denote the set  $\hat{\mathcal{T}}_0^k(R)$  (resp.  $\overline{\hat{\mathcal{T}}}_0^k(R)$ ) as the subset of  $\hat{\mathcal{T}}_0(R)$  (resp.  $\overline{\hat{\mathcal{T}}}_0(R)$ ) whose decorated trees have decorations on the node connected to the root given by k. For  $r \in \mathbf{Z}$ ,  $r \geq -1$ , we set:

$$\hat{\mathcal{T}}_0^{r,k}(R) = \{T_{\mathfrak{e}}^{\mathfrak{o}} \in \hat{\mathcal{T}}_0^k(R), n_+(T_{\mathfrak{e}}^{\mathfrak{o}}) \le r+1\}.$$

In the previous space, we disregard iterated integrals which have more than r+1 integrals and will be of order  $\mathcal{O}(\tau^{r+2})$ . We denote by  $\mathcal{G}_0^{r,k}(R) \subset \mathcal{G}_k^r$  the paired decorated forests  $F = T_1 \cdot T_2$  such that  $T_1, T_2 \in \hat{\mathcal{T}}_0^{r,k}(R)$  and  $n_+(F) \leq r+1$ .

We want to compute the second moment of the truncated sum (4.44) and then provide a discretisation. Before doing so, let us recall Wick's theorem on the higher order moments of Gaussian random variables that is needed.

**Proposition 4.4.1** Let I be a finite set and  $(X_i)_{i \in I}$  a collection of centred jointly Gaussian random variables. Then

$$\mathbb{E}\left(\prod_{i\in I} X_i\right) = \sum_{P\in\mathcal{P}(I)} \prod_{\{i,j\}\in P} \mathbb{E}(X_i X_j)$$

where  $\mathcal{P}(I)$  are partitions of I with two elements of I in each block of the partition.

Proposition 4.4.2 One has:

$$\mathcal{Q}_{\leq r+1}\mathbb{E}(|u_k^r(\tau, v^{\eta})|^2) = \sum_{F=T_1 \cdot T_2 \in \mathcal{G}_0^{r,k}(R)} m_F \frac{\bar{\Upsilon}^p(T_1)(v)\,\Upsilon^p(T_2)(v)}{S(T_1)S(T_2)} (\bar{\Pi}T_1)(\tau)\,(\Pi T_2)(\tau).$$

where the  $m_F$  belongs to  $\mathbf{N}$ .

*Proof.* We first notice that:

$$\mathbb{E}(|u_{k}^{r}(\tau, v^{\eta})|^{2}) = \sum_{T_{1}, T_{2} \in \hat{\mathcal{T}}_{0}^{r, k}(R)} \mathbb{E}\left(\frac{\bar{\Upsilon}^{p}(T_{1})(v^{\eta})\,\Upsilon^{p}(T_{2})(v^{\eta})}{S(T_{1})S(T_{2})}\right)$$

$$(\bar{\Pi}T_{1})(\tau)\,(\Pi T_{2})(\tau)$$
(4.47)

Now, we use the fact that for each decorated tree T, one has

$$\Upsilon^{p}(T)(v^{\eta}) = \Upsilon^{p}(T)(v) \prod_{u \in L_{T}} \eta^{u}_{k_{\mathfrak{o}(u)}}$$

where  $\eta^u_{k_{\mathfrak{o}(u)}}$  is given by

$$\eta^u_{k_{\mathfrak{o}(u)}} = \overline{\eta_{k_{\mathfrak{o}(u)}}}, \quad \text{if } \mathfrak{p}(e_u) = 1, \quad \eta^u_{k_{\mathfrak{o}(u)}} = \eta_{k_{\mathfrak{o}(u)}}, \quad \text{if } \mathfrak{p}(e_u) = 0.$$

where  $\mathfrak{p}(e_u)$  is the second decoration on the edge  $e_u$  connecting the leaf u to the rest of the tree T. Then, using the Wick theorem, one gets:

$$\mathbb{E}(\tilde{\Upsilon}^{p}(T_{1})(v^{\eta})\,\Upsilon^{p}(T_{2})(v^{\eta}))$$
  
= 
$$\sum_{P\in\mathcal{P}(L_{T_{1}}\sqcup L_{T_{2}})}\prod_{\{u,v\}\in P}\mathbb{E}(\tilde{\eta}^{u}_{k_{\mathfrak{o}(u)}}\tilde{\eta}^{v}_{k_{\mathfrak{o}(v)}})\tilde{\Upsilon}^{p}(T_{1})(v)\,\Upsilon^{p}(T_{2})(v).$$

where

$$\tilde{\eta}_{k_{\mathfrak{o}(v)}}^{v} = \begin{cases} \overline{\eta}_{k_{\mathfrak{o}(v)}}^{v}, \text{ if } v \in L_{T_{1}}, \\\\ \eta_{k_{\mathfrak{o}(v)}}^{v}, \text{ if } v \in L_{T_{2}}, \end{cases}$$

We obtain

$$\mathbb{E}(\tilde{\eta}_{k_{\mathfrak{o}(u)}}^{u}\tilde{\eta}_{k_{\mathfrak{o}(v)}}^{v}) = \begin{cases} \delta_{(-1)^{\mathfrak{p}(e_{u})}\mathfrak{o}(u)+(-1)^{\mathfrak{p}(e_{v})}\mathfrak{o}(v),0}, \text{ if } \{u,v\} \subset L_{T_{1}} \text{ or } \{u,v\} \subset L_{T_{2}}, \\ \delta_{(-1)^{\mathfrak{p}(e_{u})+1}\mathfrak{o}(u)+(-1)^{\mathfrak{p}(e_{v})}\mathfrak{o}(v),0}, \text{ otherwise.} \end{cases}$$

$$(4.48)$$

that fixes the condition (4.32). The map  $\delta_{\ell_1,\ell_2}$  is given by

 $\delta_{\ell_1,\ell_2} = 1$  if  $\ell_1 + \ell_2 = 0$ , and 0 otherwise.

The coefficients  $m_F$  reflects the number of pairings that can give the same F which allows us to conclude. From Proposition 4.4.2, our general scheme is given by:

$$V_{k}^{n,r}(\tau,v) = \sum_{F=T_{1}: T_{2} \in \mathcal{G}_{0}^{r,k}(R)} m_{F} \frac{\tilde{\Upsilon}^{p}(T_{1})(v) \,\Upsilon^{p}(T_{2})(v)}{S(T_{1})S(T_{2})}$$

$$\mathcal{Q}_{\leq r+1} (\bar{\Pi}^{n,r}T_{1}\Pi^{n,r}T_{2})(\tau)$$
(4.49)

The second moment of the k-th Fourier coefficient  $u_k$  of the solution of (4.42) is given by:  $V_k = \mathbb{E}(|u_k|^2)$ .

**Theorem 4.4.3 (Local error)** The numerical scheme (4.49) with initial value v = u(0) approximates the exact second moment  $V_k(\tau, v)$  up to a local error of type

$$V_k^{n,r}(\tau,v) - V_k(\tau,v) = \sum_{T_1 \cdot T_2 \in \mathcal{G}_0^{r,k}(R)} \mathcal{O}\big(\tau^{r+2} \mathcal{L}_{low}^r(T_1 \cdot T_2, n) \bar{\Upsilon}^p(T_1)(v) \Upsilon^p(T_2)(v)\big)$$

$$+ \mathcal{O}\Big(\tau^{r+2} |\nabla|^{\alpha(r+2)}(k) \mathbb{E}(\tilde{p}_k(u(t), v^{\eta}))\Big)$$

for some polynomial  $\tilde{p}_k$  and  $0 \leq t \leq \tau$  and where the operator  $\mathcal{L}_{low}^r(T_1 \cdot T_2, n)$ , given in Definition 4.3.5, embeds the necessary regularity of the solution.

*Proof.* The second moment of the k-th Fourier coefficient of the solution up to order r is given by

$$V_k^r(\tau, v) = \sum_{F=T_1 \cdot T_2 \in \mathcal{G}_0^{r,k}(R)} m_F \frac{\Upsilon^p(T_1)(v) \,\Upsilon^p(T_2)(v)}{S(T_1)S(T_2)} (\bar{\Pi}T_1\Pi T_2)(\tau)$$

which satisfies

$$V_k(\tau, v) - V_k^r(\tau, v) = \mathcal{O}\left(\tau^{r+2} |\nabla|^{\alpha(r+2)}(k) \mathbb{E}(\tilde{p}_k(u(t), v^\eta))\right)$$
(4.50)

for some polynomial  $\tilde{p}_k$  and  $0 \le t \le \tau$ . Thanks to Theorem 4.3.6, we furthermore obtain that

$$V_{k}^{n,r}(\tau,v) - V_{k}^{r}(\tau,v) = \sum_{F=T_{1} \cdot T_{2} \in \mathcal{G}_{0}^{r,k}(R)} m_{F} \frac{\bar{\Upsilon}^{p}(T_{1})\Upsilon^{p}(T_{2})}{S(T_{1})S(T_{2})}(v)$$

$$(\bar{\Pi}T_{1}\Pi T_{2} - \mathcal{Q}_{\leq r+1}\bar{\Pi}^{n,r}T_{1}\Pi^{n,r}T_{2})(\tau)$$

$$= \sum_{T_{1} \cdot T_{2} \in \mathcal{G}_{0}^{r,k}(R)} \mathcal{O}(\tau^{r+2}\mathcal{L}_{low}^{r}(T_{1} \cdot T_{2}, n)\bar{\Upsilon}^{p}(T_{1})(v)\Upsilon^{p}(T_{2})(v)).$$
(4.51)

Next we write

$$V_k^{n,r}(\tau, v) - V_k(\tau, v) = V_k^{n,r}(\tau, v) - V_k^r(\tau, v) + V_k^r(\tau, v) - V_k(\tau, v)$$

where by the definition of  $\mathcal{L}_{low}^r(T_1 \cdot T_2, n)$  we easily see that the approximation error (4.51) is in general dominant compared to (4.50) (see KdV at order one for a counter-example in Section 4.5.2).

We note that generally the terms  $\mathcal{L}_{low}^{r}(T_{1} \cdot T_{2}, n)$  are the leading terms in the local error analysis.

## 4.5 Applications

In this section we detail our general numerical scheme (4.49) together with its error analysis (see Theorem 4.4.3) on two concrete examples: The cubic Schrödinger equation (see Section 4.5.1) and the Korteweg–de Vries equation (see Section 4.5.2).

#### 4.5.1 Nonlinear Schrödinger

As a first example let us consider the nonlinear Schrödinger (NLS) equation

$$i\partial_t u(t,x) + \Delta u(t,x) = |u(t,x)|^2 u(t,x) \quad (t,x) \in \mathbf{R} \times \mathbf{T}^d$$

$$(4.52)$$

together with an initial condition of the form (4.2). The NLS equation (4.52) fits into the general form (4.1) with

$$\mathcal{L}(\nabla) = \Delta, \quad \alpha = 0, \text{ and } p(u, \overline{u}) = |u|^2 u.$$

We set  $\mathfrak{L} = \{\mathfrak{t}_1, \mathfrak{t}_2\}$ ,  $P_{\mathfrak{t}_1} = -X^2$  and  $P_{\mathfrak{t}_2} = X^2$ . An edge decorated by  $(\mathfrak{t}_1, 0)$  is denoted by |, while an edge decorated by  $(\mathfrak{t}_1, 1)$  is denoted by |, and by | an edge decorated by  $(\mathfrak{t}_2, 1)$ .

#### First order scheme

We start by expanding the solution as a power series expansion generated by its Duhamel's iterates (see (4.8)). For a first order approximation we have,

$$u(\tau, v^{\eta}) = u_1(\tau, v^{\eta}) + u_2(\tau, v^{\eta}) + \mathcal{O}(\tau^2),$$
(4.53)

where  $u_m(\tau, v^{\eta})$  denotes the *m*-th Duhamel iterate. Namely we have,

$$u_1(\tau, v^{\eta}) = e^{i\tau\Delta}v^{\eta}, \quad u_2(\tau, v^{\eta}) = e^{i\tau\Delta} \int_0^{\tau} e^{-i\zeta\Delta} (e^{i\zeta\Delta}v^{\eta})^2 (e^{-i\zeta\Delta}\overline{v^{\eta}}) d\zeta.$$

Using (4.10) one can expand the k-th Fourier coefficient  $u_k$  of the above truncated solution (4.53), and of its conjugate, as a sum over all decorated ternary trees of size at most one. As such, using (4.14), we can express a first order truncated approximation of  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  as a sum over paired couples of decorated trees. We now explain how these pairings are made, and construct the set of paired couples  $\mathcal{G}_k^0$ .

For a first order approximation we fix r = 0, and wish to construct an approximation of  $\mathbb{E}(|u_k(\tau, v^{\eta})|^2)$  with a local error of second order. We have

$$\mathbb{E}\left(\left|u_{k}(\tau, v^{\eta})\right|^{2}\right) = \mathbb{E}\left(\left(u_{k,1}(\tau, v^{\eta}) + u_{k,2}(\tau, v^{\eta})\right)\left(\overline{u_{k,1}(\tau, v^{\eta}) + u_{k,2}(\tau, v^{\eta})}\right)\right) + \mathcal{O}(\tau^{2}) \\
= \mathbb{E}\left(\left|u_{k,1}(\tau, v^{\eta})\right|^{2}\right) + 2Re\left(\mathbb{E}\left(\overline{u_{k,1}(\tau, v^{\eta})}u_{k,2}(\tau, v^{\eta})\right)\right) + \mathcal{O}(\tau^{2}).$$
(4.54)

Each of the above terms involves expectations of products of Gaussians. Hence, their computations revolve around applications of Wick's formula, which we have encoded using a decorated tree formalism at equation (4.32). We start by dealing with the first term  $\mathbb{E}\left(u_{k,1}(\tau, v^{\eta})\overline{u_{k,2}(\tau, v^{\eta})}\right)$ . In order for this expectation not to be a trivial one we require the frequencies to be equal. Namely, the couple of paired tree stemming from this pairing is

$$F_1 = T_1 \cdot T_1, \quad T_1 = \mathcal{I}_{(\mathfrak{t}_1, 0)}(\lambda_k) = \overset{k}{\blacktriangleright}.$$

$$(4.55)$$

We now deal with the computation of the second term  $\mathbb{E}\left(\overline{u_{k,1}(\tau,v^{\eta})}u_{k,2}(\tau,v^{\eta})\right)$ . In this case, there are two possible pairings of the frequencies that satisfy equation (4.32), and which yield a nontrivial computation of this second term. See the Example 4.1.1 for more details. The couple of paired trees stemming from these two pairings are:

$$F_2 = T_1 \cdot T_2, \quad T_2 = , \qquad (4.56)$$

where  $\bar{k} = -k$  and in symbolic notation we have

$$T_2 = \mathcal{I}_{(\mathfrak{t}_1,0)}\lambda_k \Big( \mathcal{I}_{(\mathfrak{t}_2,0)} \big( \lambda_k \mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_{-k}) \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{-k_2}) \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{k_2}) \big) \Big).$$

$$(4.57)$$

Finally the third pairing to consider is,

$$F_3 = T_1 \cdot T_3, \quad T_3 = , \qquad (4.58)$$

where in symbolic notation we have,

$$T_3 = \mathcal{I}_{(\mathfrak{t}_1,0)}\lambda_k \Big( \mathcal{I}_{(\mathfrak{t}_2,0)} \big( \lambda_k \mathcal{I}_{(\mathfrak{t}_1,1)} (\lambda_{k_2}) \mathcal{I}_{(\mathfrak{t}_1,0)} (\lambda_{k_2}) \mathcal{I}_{(\mathfrak{t}_1,0)} (\lambda_k) \big) \Big).$$

There is another possible pairing which is equivalent to  $F_3$ , and is given by,

$$\tilde{F}_3 = T_1 \cdot T_3, \quad T_3 = \tag{4.59}$$

The above pairing will yield the same quantity to approximate as the pairing of  $F_3$ , since the resonance structure obtained through both pairings are identical: the resonance structure is zero. Hence, we only consider the paired decorated forest  $F_3$ . We keep track of this by setting  $m_{F_3} = 2$ , while  $m_{F_1} = 1 = m_{F_2}$ . In conclusion, the set of couples of paired decorated forests is given by

$$\mathcal{G}_0^{0,k} = \{T_1 \cdot T_1, \ T_1 \cdot T_2, \ T_1 \cdot T_3\}$$

We are now ready to introduce the first order low-regularity approximation to the second order moments of  $u_k(\tau, v^{\eta})$ , solution of (4.52) with initial data (4.2).

Corollary 4.5.1 At first order our general low regularity scheme (4.15) takes the form:

$$V_k^{1,0}(\tau, v) = v_k \overline{v_k},\tag{4.60}$$

and is locally of order  $\mathcal{O}(\tau^2 |\nabla| v)$ . The above is the Fourier coefficient associated to the following scheme written in physical space:

$$u^{\ell+1} = u^{\ell} * \tilde{u}^{\ell} \tag{4.61}$$

where  $\tilde{u}^{\ell}(x) = \overline{u^{\ell}(-x)}$ .

*Proof.* We read from equation (4.15) that the first-order scheme has the general form,

$$V_{k}^{1,0}(\tau,v) = \sum_{F=T \cdot \tilde{T} \in \mathcal{G}_{0}^{0,k}(R)} m_{F} \frac{\tilde{T}^{p}(T)(v) \Upsilon^{p}(T)(v)}{S(T)S(\tilde{T})}$$

$$\mathcal{Q}_{\leq 1} \Big( (\bar{\Pi}^{1,0}T)(\Pi^{1,0}\tilde{T}) \Big)(\tau), \qquad (4.62)$$

Furthermore, we have that,

$$\Upsilon^{p}(T_{1})(v) = v_{k}, \quad \Upsilon^{p}(T_{2})(v) = 2v_{k}v_{k_{2}}\overline{v_{k_{2}}} = \Upsilon^{p}(T_{3})(v), \quad (4.63)$$
$$S(T_{1}) = 1, \quad S(T_{2}) = 2 = S(T_{3}).$$

Hence, the first-order scheme (4.62) takes the form

$$V_{k}^{1,0}(\tau,v) = v_{k}\overline{v_{k}}\mathcal{Q}_{\leq 1}(\bar{\Pi}^{1,0}(T_{1}))(\Pi^{1,0}(T_{1}))$$

$$+ 2Re\left(v_{k}\overline{v_{k}}\sum_{k_{2}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2}\mathcal{Q}_{\leq 1}\left(\bar{\Pi}^{1,0}(T_{1})(\Pi^{1,0}(T_{2})+2\Pi^{1,0}(T_{3}))\right)(\tau)\right).$$

$$(4.64)$$

We are left to calculate the first-order approximations, encoded in the definition of the character  $\Pi^{1,0}$ , of each of our decorated trees appearing in the couples defined in  $\mathcal{G}_0^{0,k}(R)$ . From (4.55) and by Definition 4.34, given that  $P_{(t_1,0)}(k) = -k^2$  and  $P_{(t_1,1)}(k) = k^2$  we have,

$$(\Pi^{1,0}T_1)(\tau) = e^{i\tau P_{(\mathfrak{t}_1,0)}(k)} = e^{-i\tau k^2}, \quad (\bar{\Pi}^{1,0}T_1)(\tau) = e^{i\tau P_{(\mathfrak{t}_1,1)}(k)} = e^{i\tau k^2}.$$
(4.65)

Henceforth, for notational convenience we will denote  $o_1 = (\mathfrak{t}_1, 0), \overline{o_1} = (\mathfrak{t}_1, 1)$ , and  $o_2 = (\mathfrak{t}_2, 0)$ . Then again by Definition

4.34 and (4.57) it follows that,

$$(\Pi^{1,0}T_{2})(\tau)$$

$$= e^{-i\tau k^{2}} (\Pi^{1,0}\mathcal{I}_{o_{2}}(\lambda_{k}\mathcal{I}_{\bar{o}_{1}}(\lambda_{-k})\mathcal{I}_{o_{1}}(\lambda_{-k_{2}})\mathcal{I}_{o_{1}}(\lambda_{k_{2}})))(\tau)$$

$$= e^{-i\tau k^{2}} \mathcal{K}_{o_{2}}^{k,0} ((\Pi^{n,-1}\mathcal{I}_{\bar{o}_{1}}(\lambda_{-k}))(\xi)(\Pi^{n,-1}\mathcal{I}_{o_{1}}(\lambda_{-k_{2}})(\xi)$$

$$(\Pi^{n,-1}\mathcal{I}_{o_{1}}(\lambda_{k_{2}}))(\xi))(\tau)$$

$$= e^{-i\tau k^{2}} \mathcal{K}_{o_{2}}^{k,0} (e^{i\xi(k^{2}-2k_{2}^{2})}, 1)(\tau)$$

$$= -i\tau e^{-i\tau k^{2}},$$

$$(4.66)$$

where we applied Definition 4.3.1 with  $q = \ell = 0$ ,  $P(k_1, ..., k_n) = k^2 - 2k_2^2$ , and  $P_{o_2}(k) = k^2$ . We note that the resonance structure obtain via this pairing is given by

$$P(k_1, \dots, k_n) + P_{o_2}(k) = 2(k^2 - k_2^2).$$
(4.67)

Similarly we have,

$$(\Pi^{1,0}T_3)(\tau) = e^{-i\tau k^2} \left( \Pi^{1,0}\mathcal{I}_{o_2}(\lambda_k \mathcal{I}_{\bar{o}_1}(\lambda_{k_2})\mathcal{I}_{o_1}(\lambda_{k_2})\mathcal{I}_{o_1}(\lambda_k)) \right)(\tau)$$

$$= e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,0} (e^{i\xi(k_2^2 - k_2^2 - k^2)}, 1)(\tau)$$

$$= -i\tau e^{-i\tau k^2},$$
(4.68)

where we notice that the resonance structure for the integral  $\Pi T_3$  is zero:

$$P(k_1, ..., k_n) + P_{o_2}(k) = k^2 + (k_2^2 - k_2^2 - k^2) = 0.$$
(4.69)

Hence exact integration takes place, and no approximation is necessary when applying the approximation operator  $\mathcal{K}_{o_2}$  on the integral corresponding to  $T_3$ . Collecting the above computations, and plugging it in the expansion (4.64) yields the low-regularity scheme (4.60).

**Local Error:** It remains to show the first order local error bound:  $\mathcal{O}(\tau^2 |\nabla| v)$ . By Theorem 4.4.3 we have

$$V_k^{n,0}(\tau,v) - V_k(\tau,v) = \sum_{T \cdot \tilde{T} \in \mathcal{G}_0^{0,k}(R)} \mathcal{O}(\tau^2 \mathcal{L}^0_{\text{low}}(T \cdot \tilde{T}, 1) \bar{\Upsilon}^p(T)(v) \Upsilon^p(\tilde{T})(v)).$$

We now calculate  $\mathcal{L}^0_{\text{low}}(T \cdot \tilde{T}, 1)$ , for every pair  $T \cdot \tilde{T} \in \mathcal{G}^{0,k}_0(R)$ . First, we consider the pair  $F_1 = T_1 \cdot T_1$ , where we have  $n_+(T_1) = 0$  and hence where no time discretization is necessary. By Definition 4.3.5 of  $\mathcal{L}^r_{\text{low}}$  we have,

$$\mathcal{L}_{low}^{0}(T_{1} \cdot T_{1}, 1) = (\Pi T_{1})_{0} \mathcal{L}_{low}^{0}(T_{1}, 1) + (\Pi^{1,0} T_{1})_{0} \mathcal{L}_{low}^{0}(T_{1}, 1) + (\Pi^{1,0} T_{1})_{1} \mathcal{L}_{low}^{-1}(T_{1}, 1),$$
(4.70)

where  $\mathcal{L}_{low}^0(T_1, 1) = \mathcal{L}_{low}^{-1}(T_1, 1) = 1$ . Furthermore, we have  $(\Pi T_1)_0 = 1$  and  $e^{-i\tau k^2} = (\Pi^{1,0}T_1)_0$ , and finally  $(\Pi^{1,0}T_1)_1 = 0$  since  $T_1$  corresponds to a term of only zero-th order. Hence, from (4.70) it follows that  $\mathcal{L}_{low}^0(T_1 \cdot T_1, 1) = 2e^{-i\tau k^2}$  and as seen previously  $\Upsilon^p(T_1) = v_k$ .

Next we consider the pair  $F_2 = T_1 \cdot T_2$ , where  $n_+(T_2) = 1$ . Again by Definition 4.3.5 of  $\mathcal{L}_{low}^r$  we have,

$$\mathcal{L}_{low}^{0}(T_1 \cdot T_2, 1) = (\Pi T_2)_0 \mathcal{L}_{low}^{-1}(T_1, 1) + (\Pi^{1, -1} T_1)_0 \mathcal{L}_{low}^{0}(T_2, 1),$$
(4.71)

where  $\Pi^{1,-1} = \Pi^1 \mathcal{D}^{-1} = \Pi^1$ ,  $(\Pi^1 T_1)_0 = e^{-i\tau k^2}$ , and  $(\Pi T_2)_0 = 1$ . Next, we have

$$\mathcal{L}^{0}_{\text{low}}(T_2, 1) = \mathcal{L}^{0}_{\text{low}}(G_2, 1) + R^{k,0}_{o_2,1}(\Pi^1 G_2),$$

with  $T_2 = \mathcal{I}_{o_1}(\lambda_k \mathcal{I}_{o_2}(\lambda_k G_2))$ , and  $G_2 = \mathcal{I}_{\bar{o}_1}(\lambda_{-k_2})\mathcal{I}_{o_1}(\lambda_{-k_2})\mathcal{I}_{o_1}(\lambda_{k_2})$ . Given that we are performing a full Taylor approximation (4.3.1) of  $\Pi T_2$ , we have that the local error  $R^{k,0}_{o_2,1}(\Pi^1 G_2)$  is given by (4.36). Given the resonance factor (4.67) it follows that  $R^{k,0}_{o_2,1}(\Pi^1 G_2) = R^{k,0}_{o_2,1}(e^{i\xi(k^2-2k_2^2)}) = 2(k^2 - k_2^2)$ . Lastly,  $\mathcal{L}^0_{\text{low}}(G_2, 1)$  consists of linear combinations involving the propagator  $e^{\pm i\tau k^2}$ , and not of a polynomial of the frequencies  $(k_1, ..., k_n)$ . This is due to the fact that  $G_2$  only consists of terms of zero-th order, and hence these computations are similar to those already made for  $T_1 \cdot T_1$ . Therefore,  $\mathcal{L}^{0}_{low}(T_2, 1) = \mathcal{O}(\sum_{k_2 \in \mathbf{Z}^d} (k^2 - k_2^2))$ , and by (4.71),

$$\mathcal{L}_{low}^{0}(T_1 \cdot T_2, 1)\bar{\Upsilon}^{p}(T_1)\Upsilon^{p}(T_2) = \mathcal{O}(\sum_{k_2 \in \mathbf{Z}^d} (k^2 - k_2^2)\overline{v_k}v_k |v_{k_2}|^2).$$
(4.72)

In physical space this yields an error of  $\mathcal{O}(\tau^2(\nabla v * \nabla \tilde{v}) \|v\|_{L^2}^2 + (v * \tilde{v}) \|\nabla v\|_{L^2}^2)$ .

Finally, the computations for the pair  $F_3 = T_1 \cdot T_3$  is a simpler case of the above analysis for the pair  $F_2$ . Indeed, with the pairing of  $F_3$  the resonance factor (4.69) is zero and hence, no local error is induced by this term: exact integration takes place. Therefore, the local error is given by (4.84) in Fourier space, and in physical space asks for  $\mathcal{O}(\tau |\nabla| v)$ , namely requires only one spacial derivative on the initial data thanks to the use of convolutions.

#### Second order schemes

For a second order approximation we have,

$$\mathbb{E}\left(|u_{k}(\tau,v^{\eta})|^{2}\right) = \mathbb{E}\left(\left(u_{k,1}(\tau,v^{\eta}) + u_{k,2}(\tau,v^{\eta}) + u_{k,3}(\tau,v^{\eta})\right)\right) \\ \left(\overline{u_{k,1}(\tau,v^{\eta}) + u_{k,2}(\tau,v^{\eta}) + u_{k,3}(\tau,v^{\eta})}\right) + \mathcal{O}(\tau^{3}) \\ = \mathbb{E}\left(|u_{k,1}(\tau,v^{\eta})|^{2}\right) + 2Re\mathbb{E}\left(\overline{u_{k,1}(\tau,v^{\eta})}u_{k,2}(\tau,v^{\eta})\right) \\ + 2Re\mathbb{E}\left(\overline{u_{k,1}(\tau,v^{\eta})}u_{k,3}(\tau,v^{\eta})\right) + \mathbb{E}\left(|u_{k,2}(\tau,v^{\eta})|^{2}\right) + \mathcal{O}(\tau^{3}),$$

where  $u_{k,3}$  is the k-th coefficient of the term :

$$u_{3}(\tau, v^{\eta}) = -\int_{0}^{\tau} e^{i(\tau-\xi)\Delta} \Big( \int_{0}^{\xi} e^{i(\xi-\xi_{1})\Delta} \Big( (e^{i\xi_{1}\Delta}v^{\eta})^{2} (e^{-i\xi_{1}\Delta}\overline{v^{\eta}}) \Big) d\xi_{1} \\ (2(e^{i\xi\Delta}v^{\eta})(e^{-i\xi\Delta}\overline{v^{\eta}})) \Big) d\xi \\ + \int_{0}^{\tau} e^{i(\tau-\xi)\Delta} \Big( \Big( \int_{0}^{\xi} e^{-i(\xi-\xi_{1})\Delta} (e^{-i\xi_{1}\Delta}\overline{v^{\eta}})^{2} (e^{i\xi_{1}\Delta}v^{\eta}) d\xi_{1} \Big) (e^{i\xi\Delta}v^{\eta})^{2} \Big) d\xi \\ := u_{3,1}(\tau, v^{\eta}) + u_{3,2}(\tau, v^{\eta}),$$

$$(4.73)$$

and where  $u_{3,1}(\tau, v^{\eta})$  and  $u_{3,2}(\tau, v^{\eta})$  are encoded in Fourier by the following two decorated trees respectively:



Using Wick's formula we can once again obtain the possible pairings which come at play when wanting to compute each of the above terms. This determines the set of decorated pair of trees  $\mathcal{G}_0^{1,k}(R)$ , of size at most two, which we detail below.

From the previous Section 4.5.1, we have that the pair of decorated trees which encodes the first term  $\mathbb{E}(|u_{k,1}(\tau, v^{\eta})|^2)$ is given by  $F_1 = T_1 \cdot T_1$ , defined in (4.55). Similarly, we have that the second term  $\mathbb{E}[\overline{u_{k,1}(\tau, v^{\eta})}u_{k,2}(\tau, v^{\eta})]$  is encoded by the pairings  $F_2 = T_1 \cdot T_2$  and  $F_3 = T_1 \cdot T_3$ , defined in (4.56) and (4.58) respectively. Next, we consider the possible pairings we need to take into account for the computation of  $\mathbb{E}[\overline{u_{k,1}(\tau, v^{\eta})}u_{k,3}(\tau, v^{\eta})]$ , where  $u_3$  is the sum of two second order terms given in (4.73). We start by considering the different pairings between the frequencies of  $\overline{u_{k,1}}$  and  $u_{k,3,1}$ , those for  $u_{k,3,2}$  will then follow. We can classify the different pairings into five groups, which we call the (five) principal pairings. This classification is based upon two factors:

(a) first, whether the frequency pairing is made internally within the same layer/integral in the tree,

(b) secondly, whether the pairing is made between frequencies of the same or opposite sign.

We start by looking at the pairings which consist of at least one pairing of two frequencies made internally within the same integral. There are then two possibilities to consider, which depend on the sign of the frequency pairing. Namely, we have that a pairing between two frequencies of opposite sign will result in a null resonance factor, as was the case in (4.69) for  $F_3$ , and a pairing between two frequencies of the same sign will results in a resonance structure of the form

(4.67), namely as a difference of the squares of the frequencies. Hence, there are solely four cases of pairings which we need to consider:

- First, the case when the pairings are all of opposite signs:  $\{k_1 = k_2, k_4 = k_5, k = k_3\} \cup \{k_1 = k_3, k_2 = k, k_4 = k_5\} \cup \{k_1 = k_2, k_3 = k_4, k = k_5\} \cup \{k_1 = k_3, k_2 = k_4, k = k_5\}$ . Without loss of generality we take the first pairing in the union above as the representative of this class, which we include in the set  $\mathcal{G}_0^{1,k}(R)$ :

$$F_{4} = T_{1} \cdot T_{4}, \quad T_{1} = \overset{(k)}{\downarrow}, \quad T_{4} = \overset{(k)}{\downarrow}. \quad (4.74)$$

— Secondly, the case where frequencies of opposite signs are paired in the inner-integral (or tree), whereas frequencies of the same signs are paired in the outer-integral (or tree):  $\{k_1 = k_2, k_3 = -k_5, k = -k_4\}, \cup\{k_1 = k_3, k_2 = -k_5, k = -k_4\}$ . We once again take the first pairing in the union above as the representative of this class, which we include in the set  $\mathcal{G}_0^{1,k}(R)$ :

$$F_5 = T_1 \cdot T_5, \quad T_1 = \overbrace{}^{k_2}, \quad T_5 =$$

— Thirdly, the case where frequencies of the same signs are paired within the inner-integral (or tree), whereas frequencies of opposite signs are paired in the outer-integral (or tree):  $\{k_2 = -k_3, k = -k_1, k_4 = k_5\} \cup \{k_2 = -k_3, k_4 = -k_1, k_4 = k_5\}$ . We take the first pairing in the union above as the representative of this class:

$$F_6 = T_1 \cdot T_6, \quad T_1 = \overset{(k)}{\downarrow}, \quad T_6 = \overset{(k)}{\downarrow}.$$

- Fourthly, we are left with the case where the internal pairings are of the same sign. The only pairing which corresponds to this case is given by:  $\{k_2 = -k_3, k_1 = k_5, k = -k_4\}$ .



Lastly, going back to point (a.) in the above we are left to consider pairings which are all external, meaning that there are no pairings of two frequencies made internally within the same integral. Hence, no simplification will be made at the level of the resonance structure of each of the integrals. This defines the last class to be considered. There are six possibilities of such pairings:  $\{k = -k_1, k_2 = k_4, k_3 = -k_5\} \cup \{k = -k_1, k_2 = -k_5, k_3 = k_4\} \cup \{k = k_2, k_1 = -k_4, k_3 = -k_5\} \cup \{k = k_2, k_1 = k_5, k_3 = k_4\} \cup \{k = k_3, k_2 = -k_5, k_1 = -k_4\} \cup \{k = k_3, k_2 = k_4, k_3 = -k_5\}$ . We take the first pairing

in the union above as the representative of this class:



Analogously, one can make the analysis for the pairings of  $\overline{u_{k,1}}$  and  $u_{k,3,2}$  to obtain that five representatives are given by:  $F_i = T_1 \cdot T_i$ , for  $i \in \{9, ..., 13\}$ , where

$$T_{9} = \begin{pmatrix} k_{2} & k_{3} & k_{2} & k_{4} & k_{3} \\ k_{2} & k_{4} & k_{4} & k_{4} & k_{4} & k_{4} \\ k_{4} & k_{4} & k_{4} & k_{4} & k_{4} \\ k_{5} & k_{4} & k_{5} & k_{5} & k_{5} \\ k_{6} & k_{5} & k_{6} & k_{6} & k_{6} \\ k_{6} & k_{6} & k_{6} & k_{6} & k_{6} \\ k_{6} & k_{6} & k_{6} & k_{6} & k_{6} \\ k_{6} & k_{6} & k_{6} & k_{6} & k_{6} \\ k_{6} & k_{6} & k_{6} & k_{6} & k_{6} \\ k_{7} & k_{6} & k_{6} & k_{6} & k_{6} \\ k_{7} & k_{7} & k_{7} & k_{6} & k_{6} \\ k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} \\ k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} \\ k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} \\ k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} & k_{7} \\ k_{7} & k_{7} \\ k_{7} & k_{7} \\ k_{7} & k_$$

Finally, also using the same case analysis as made previously, we can construct the possible pairings between  $\overline{u_{k,1}}$  and  $u_{k,1}$  in order to compute the last term  $\mathbb{E}(|u_{k,1}(\tau, v^{\eta})|^2)$ . We have the same case figures as seen previously: six possible external pairings, four different possible pairings which are internal and of opposite signs, by symmetry four possibilities of pairings which are internal and one of opposite and the other of the same signs, and finally one possible internal pairing of the same signs. A representative of each of the above four class is given by:  $F_i = \tilde{T}_i \cdot T_i$ , for  $i \in \{14, ..., 17\}$ , where

$$\tilde{T}_{14} = \begin{pmatrix} k_1 \\ \vdots \\ \vdots \\ \vdots \\ \tilde{T}_{14} = \end{pmatrix}, T_{14} = \begin{pmatrix} k_2 \\ \vdots \\ \vdots \\ \vdots \\ \tilde{T}_{16} = \end{pmatrix} = T_3, \quad \tilde{T}_{15} = \begin{pmatrix} k_1 \\ \vdots \\ \vdots \\ \vdots \\ \tilde{T}_{15} = \end{pmatrix}, T_{15} = T_2$$

We denote by  $m_{F_i}$ ,  $i \in [4, 17]$  the multiplicative constant representing the number of elements in the class who's representative is  $F_i$ . Namely, it follows from the above analysis that

$$m_{F_i} = 6, \text{ for } i \in \{8, 13, 17\},$$
  

$$m_{F_i} = 2, \text{ for } i \in \{5, 10, 6, 11\},$$
  

$$m_{F_i} = 1, \text{ for } i \in \{7, 12, 16\},$$
  

$$m_{F_i} = 4, \text{ for } i \in \{4, 9, 14, 15\}.$$
  
(4.75)

In conclusion, we have that the set of representatives of approximated paired forests is given by:  $\mathcal{G}_0^{1,k}(R) = \{F_i\}_{i=\{1,\ldots,17\}}$ . We now present a second order low-regularity approximation to the second order moments of  $u_k(\tau, v^{\eta})$ , solution of (4.52) with initial data (4.2).

**Corollary 4.5.2** At second order our general low regularity scheme (4.15) takes the form:

$$V_{k}^{2,1}(\tau,v) = v_{k}\overline{v_{k}} + 2\tau^{2}v_{k}\overline{v_{k}}\left(|k|^{2}\sum_{k_{2}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2} - \sum_{k_{2}\in\mathbf{Z}^{d}}|k_{2}|^{2}|v_{k_{2}}|^{2}\right) - 6\tau^{2}\left(6v_{k}\overline{v_{k}}\left(\sum_{k_{2}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2}\right)^{2} - \sum_{\substack{k_{1},k_{2},k_{3}\in\mathbf{Z}^{d}\\-k_{1}+k_{2}+k_{3}=k}}|v_{k_{1}}|^{2}|v_{k_{2}}|^{2}|v_{k_{3}}|^{2}\right).$$
(4.76)

The scheme (4.76) is locally of order  $\mathcal{O}(\tau^3 |\nabla|^2 v)$ . The above is the Fourier coefficient associated to the following scheme written in physical space:

$$u^{\ell+1} = u^{\ell} * \tilde{u}^{\ell} + 2\tau^{2} \Big( (\nabla u^{\ell} * \nabla \tilde{u}^{\ell}) \| u^{\ell} \|_{L^{2}}^{2} - (u^{\ell} * \tilde{u}^{\ell}) \| \nabla u^{\ell} \|_{L^{2}}^{2} \Big) - 6\tau^{2} \Big( 6(u^{\ell} * \tilde{u}^{\ell}) \| u^{\ell} \|_{L^{2}}^{4} - (u^{\ell} * \tilde{u}^{\ell})^{3} \Big),$$

$$(4.77)$$

where  $\tilde{u}^{\ell}(x) = \overline{u^{\ell}(-x)}$ .

**Remark 4.5.3 (Stabilisation technique)** We note that in contrast to the first-order scheme (4.61) the second-order scheme (4.77) involves spatial derivatives on the numerical solution such as for instance in the term  $\nabla u^{\ell} * \nabla \tilde{u}^{\ell}$ . These derivatives cause instability in the discretisation. For practical implementations, one needs to stabilize the scheme in order for it to converge. Different approaches can be taken to stabilize the above scheme, which do not require a Courant-Friedrichs-Lewy (CFL) type condition on the step sizes. Here we propose an approach based on the a posteriori inclusion of well chosen filter functions. We refer to the works of [ABBS22b, AB23a] for more details regarding these questions of stability and of the suitable choice of filter functions used to yield stable schemes with optimal local error. We also refer to [HLW10] for a general introduction to filter function in the case of oscillatory ordinary differential equations.

In this work we introduce the following filter function

$$\operatorname{sinc}^2\left(\tau^{\frac{1}{2}}|\nabla|\right)$$

which in in Fourier space takes the form,

$$\Psi(i\tau|k|^2) = \operatorname{sinc}^2\left(\tau^{\frac{1}{2}}|k|\right) = \frac{1}{(i\tau^{\frac{1}{2}}|k|)^2} \left(e^{i\tau^{\frac{1}{2}}|k|/2} - e^{-i\tau^{\frac{1}{2}}|k|/2}\right)^2.$$
(4.78)

In order to stabilise our scheme (4.77) we pre-multiply both critical terms in the second line of (4.60) with the filter function (4.78) in the corresponding frequencies k and  $k_2$  respectively. This yields the following stabilised version of (4.76)

$$V_{k}^{2,1}(\tau, v) = v_{k}\overline{v_{k}} + 2\tau^{2}v_{k}\overline{v_{k}} \left(\Psi(i\tau|k|^{2})|k|^{2}\sum_{k_{2}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2} - \sum_{k_{2}\in\mathbf{Z}^{d}}\Psi(i\tau|k_{2}|^{2})|k_{2}|^{2}|v_{k_{2}}|^{2}\right) + \dots$$
(4.79)

We have that

$$\Psi(i\tau|k|^2) = 1 + \mathcal{O}(\tau|k|^2)$$

and hence the stabilised scheme (4.79) preserves the low regularity error structure of  $\mathcal{O}(\tau |\nabla|^2 u)$ . This is essential for the local error analysis of the scheme. Furthermore, thanks to the observation that

$$|\Psi(i\tau|k|^2)\tau|k|^2| \le 1$$
 for all  $k \in \mathbf{Z}^d$ 

 $\Psi$  renders a stabilised version of the scheme (4.77) in physical space given by:

$$\begin{split} u^{\ell+1} &= u^{\ell} * \tilde{u}^{\ell} \\ &- 2\tau \bigg( \big( (e^{i\tau^{\frac{1}{2}} |\nabla|/2} - e^{-i\tau^{\frac{1}{2}} |\nabla|/2} \big) u^{\ell} * (e^{i\tau^{\frac{1}{2}} |\nabla|/2} - e^{-i\tau^{\frac{1}{2}} |\nabla|/2} \big) \tilde{u}^{\ell} \big) \| u^{\ell} \|_{L^{2}}^{2} \\ &- (u^{\ell} * \tilde{u}^{\ell}) \| (e^{i\tau^{\frac{1}{2}} |\nabla|/2} - e^{-i\tau^{\frac{1}{2}} |\nabla|/2} ) u^{\ell} \|_{L^{2}}^{2} \bigg) \\ &- 6\tau^{2} \Big( 6(u^{\ell} * \tilde{u}^{\ell}) \| u^{\ell} \|_{L^{2}}^{4} - (u^{\ell} * \tilde{u}^{\ell})^{3} \Big). \end{split}$$

**Remark 4.5.4 (Practical implementation)** In order to allow for a practical implementation of our schemes we want to be able to express the discretization both in physical and Fourier space. This allows us to use the Fast Fourier Transform (FFT) whose computational cost is of order  $O(|K|^d log(|K|^d))$ , where K denotes the highest frequency in the discretization and d is the dimension. Namely, we compute the action of the filter functions on the solution in frequency space, while computing the product of functions in physical space.

**Remark 4.5.5** In (4.77), the scheme  $V_k^{n,r}$  is written in physical space. In fact, one can wonder if it is possible to get a general statement such as [BS22, Prop. 3.18] which shows that low regualrity schemes can always be rewritten in physical space. It is not clear how to prove such a statement in full generality when exact integrations are performed instead of a full Taylor expansion. Indeed, the core of the proof [BS22, Prop. 3.18] relies on [BS22, Assumption 1] which is no longer true in our case: Frenquencies on the leaves are no longer disjoint. We will have to be more cautious in the scheme and probably Taylor expand a bit more as exact integrations will not allow to move back to physical space. It is a challenging open question. In the case of the present work, we are mostly doing Taylor expansions. They guarantee that we are able to go back to physical space.

*Proof.* We are considering a second order scheme and hence, we enter the case where r = 1. Furthermore, we ask for two derivatives on the initial data, and hence we take the regularity parameter n to take on the value of 2. We read from equation (4.15) that the second-order scheme has the general form,

$$V_{k}^{2,1}(\tau,v) = \sum_{F=T_{1} \cdot T_{2} \in \mathcal{G}_{0}^{1,k}(R)} m_{F} \frac{\Upsilon^{p}(T_{1})(v) \Upsilon^{p}(T_{2})(v)}{S(T)S(T_{2})}$$

$$\mathcal{Q}_{\leq 2}((\bar{\Pi}^{2,1}(T_{1}))(\Pi^{2,1}(T_{2})))(\tau),$$
(4.80)

where  $\{m_F\}_{F \in \mathcal{G}_0^{1,k}(R)}$  are given in (4.75), and where in the same spirit of (4.63) we have that for all  $i \in \{4, ..., 13\}$ ,

$$\begin{split} \Upsilon^{p}(T_{4})(v) &= 2\Upsilon^{p}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k_{4}}))(v)\Upsilon^{p}(T_{3})(v)\Upsilon^{p}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{4}}))(v) \\ &= 2\overline{v_{k_{4}}}(2\overline{v_{k_{2}}}v_{k_{2}}v_{k})v_{k_{4}} \\ &= 4|v_{k_{4}}|^{2}|v_{k_{2}}|^{2}v_{k} \\ &= \Upsilon^{p}(T_{i})(v). \end{split}$$

and for  $i \in \{4, ..., 8\}$ ,  $S(T_i) = 2$  while for  $i \in \{9, ..., 13\}$ ,  $S(T_i) = 4$ . Furthermore, we have that for  $i \in \{14, ..., 17\}$ ,

$$\overline{\Upsilon}^p(\widetilde{T}_i)(v)\Upsilon^p(T_i)(v) = 4|v_{k_1}|^2|v_{k_2}|^2|v_{k_3}|^2, \quad S(T_i)^2 = 4.$$

Hence, the second order scheme (4.80) takes the form

$$\begin{aligned} V_{k}^{2,1}(\tau, v) &= v_{k}\overline{v_{k}} \\ &+ 2Re\left(v_{k}\overline{v_{k}}\sum_{k_{2}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2}\mathcal{Q}_{\leq 2}\left(\bar{\Pi}^{2,1}(T_{1})\left(\Pi^{2,1}(T_{2})+2\Pi^{2,1}(T_{3})\right)\right)(\tau)\right) \\ &+ 2Re\left(v_{k}\overline{v_{k}}\sum_{k_{2},k_{4}\in\mathbf{Z}^{d}}|v_{k_{2}}|^{2}|v_{k_{4}}|^{2} \\ \left(\sum_{4\leq i\leq 13}m_{F_{i}}\mathcal{Q}_{\leq 2}\left(\bar{\Pi}^{2,1}(T_{1})\Pi^{2,1}(T_{i})\right)\right)(\tau)\right) \\ &+ v_{k}\overline{v_{k}}\sum_{k_{1},k_{2}\in\mathbf{Z}^{d}}|v_{k_{1}}|^{2}|v_{k_{2}}|^{2}\left(\sum_{14\leq i\leq 16}m_{F_{i}}\mathcal{Q}_{\leq 2}\left(\bar{\Pi}^{2,1}(\bar{T}_{i})\Pi^{2,1}(T_{i})\right)\right)(\tau) \\ &+ \sum_{\substack{k_{1},k_{2},k_{3}\in\mathbf{Z}^{d}\\ -k_{1}+k_{2}+k_{3}=k_{3}}}|v_{k_{1}}|^{2}|v_{k_{2}}|^{2}|v_{k_{3}}|^{2}\left(m_{F_{17}}\mathcal{Q}_{\leq 2}\left(\bar{\Pi}^{2,1}(T_{17})\Pi^{2,1}(T_{17})\right)\right)(\tau). \end{aligned}$$

1. Computation of  $\mathcal{Q}_{\leq 2}(\overline{\Pi}^{2,1}(T_1)(\Pi^{2,1}(T_2) + 2\Pi^{2,1}(T_3)))$ : We apply Definition 4.34 to each of the three decorated trees.

— Computation of  $\overline{\Pi}^{2,1}(T_1)$ : Given that (4.55) is a zero-th order term, as in (4.65) we have,

$$(\bar{\Pi}^{2,1}T_1)(\tau) = e^{i\tau k^2}$$

— Computation of  $\Pi^{2,1}(T_2)$ : It follows from the definition (4.56) and (4.57) of  $T_2$  that,

$$(\Pi^{2,1}T_2)(\tau) = e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1} ((\Pi^{2,0}\mathcal{I}_{\bar{o}_1}(\lambda_{-k}))(\xi) (\Pi^{2,0}\mathcal{I}_{o_1}(\lambda_{-k_2})(\xi)(\Pi^{2,0}\mathcal{I}_{o_1}(\lambda_{k_2}))(\xi))(\tau) = e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1} (e^{i\xi(k^2 - 2k_2^2)}, 2)(\tau) = -i\tau e^{-i\tau k^2} + \tau^2 e^{-i\tau k^2} (k^2 - k_2^2),$$
(4.82)

where we applied Definition 4.3.1 with  $\ell = 1$ . Namely, up to one additional order then made in (4.66) for the first order analysis.

- Computation of  $\Pi^{2,1}(T_3)$ : As for the calculations made in (4.68), given that the resonance structure (4.69) of the integrand in  $\Pi(T_3)$  is zero it follows that,

$$(\Pi^{2,1}T_3)(\tau) = -i\tau e^{-i\tau k^2}.$$
(4.83)

Hence, having only at most second order terms in the above calculations,  $Q_{\leq 2}$  does not play a role and we have:

$$\begin{aligned} \mathcal{Q}_{\leq 2} \big( \bar{\Pi}^{2,1}(T_1) \big( \Pi^{2,1}(T_2) + 2\Pi^{2,1}(T_3) \big)(\tau) &= \bar{\Pi}^{2,1}(T_1) \big( \Pi^{2,1}(T_2) + 2\Pi^{2,1}(T_3) \big)(\tau) \\ &= -3i\tau + \tau^2 (k^2 - k_2^2). \end{aligned}$$

Given that we take the real part of the above approximation, only the second term in the above will contribute to the scheme.

- 2. Computation of  $\mathcal{Q}_{\leq 2}(\overline{\Pi}^{2,1}(T_1)\Pi^{2,1}(T_i))$ , for  $i \in \{4, ..., 13\}$ :
- Computation of  $(\bar{\Pi}^{2,1}(T_1)\Pi^{2,1}(T_4))$ : We recall that  $T_4$  is given by (4.74) and in symbolic notation we have,

$$T_4 = \mathcal{I}_{o_1} \Big( \lambda_k \Big( \mathcal{I}_{o_2}(\lambda_k (\mathcal{I}_{\bar{o_1}}(\lambda_{k_4}) T_3 \mathcal{I}_{o_1}(\lambda_{k_4}))) \Big) \Big),$$

where  $T_3$  is given in (4.58). Hence, it follows that

$$\begin{aligned} (\Pi^{2,1}T_4)(\tau) &= e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1} \left( (\Pi^{2,0}\mathcal{I}_{\bar{o}_1}(\lambda_{k_4}))(\xi) (\Pi^{2,0}T_3)(\xi) (\Pi^{2,0}\mathcal{I}_{o_1}(\lambda_{k_4}))(\xi) \right)(\tau) \\ &= e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1} \left( -i\xi e^{i\xi (k_4^2 - k^2 - k_4^2)} \right)(\tau) \\ &= -ie^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1} (\xi e^{-i\xi k^2}, 2)(\tau) \\ &= -\frac{\tau^2}{2} e^{-i\tau k^2}, \end{aligned}$$

where in the second line we used that  $(\Pi^{2,0}T_3)(\xi) = -i\xi e^{-i\xi k^2}$ , which comes from (4.68), obtained during the first order analysis. We note that when calculating  $\mathcal{K}$  in the last line, the resonance structure is as expected equal to zero. Computation of  $\Pi^{2,1}(T_i)$ , for  $i \in \{5, ..., 13\}$ : These computations follow exactly the same line as in the above case. We have,  $(\Pi^{2,1}T_i)(\tau) = -\frac{\tau^2}{2}e^{-i\tau k^2}$ .

The only difference lies in the expression of the resonance factor which will either be of the form (4.67), be equal to zero (as was the case in the above), or finally in the case of an external pairing ( $\Pi(T_i)$ ,  $i \in \{8, 13\}$ ) of the form (4.23). Nevertheless, given that we can make the full Taylor approximation (4.3.1) when approximating the each integral, the approximation of  $\Pi^{2,1}(T_i)$ ,  $i \in \{4, ..., 13\}$  are the same, the difference lies in the local error produced.

Therefore,  $\mathcal{Q}_{\leq 2}(\bar{\Pi}^{2,1}(T_1)\Pi^{2,1}(T_i))(\tau) = -\frac{\tau^2}{2}$ . 3. Computation of  $\mathcal{Q}_{\leq 2}(\bar{\Pi}^{2,1}(T_i)\Pi^{2,1}(T_i))(\tau)$ , for  $i \in \{14, ..., 17\}$ :

- Computation of  $\overline{\Pi}^{2,1}(T_{14})\Pi^{2,1}(T_{14})$ . Given that  $T_{14}$  has the same null resonance structure as  $T_3$ , it follows as in (4.83) that  $(\Pi^{2,1}T_{14})(\tau) = -i\tau e^{-i\tau k^2}$  and hence,

$$(\bar{\Pi}^{2,1}T_{14}\Pi^{2,1}T_{14})(\tau) = \tau^2$$

For the remaining computations the truncation operator  $Q_{\leq 2}$  will be essential to incorporate the terms of correct order into the scheme.

- Computation of  $\mathcal{Q}_{r\leq 2}\bar{\Pi}^{2,1}(\tilde{T}_{15})\Pi^{2,1}(T_{15})$ ,  $\mathcal{Q}_{r\leq 2}\bar{\Pi}^{2,1}(\tilde{T}_{16})\Pi^{2,1}(T_{16})$ , and  $\mathcal{Q}_{r\leq 2}\bar{\Pi}^{2,1}(\tilde{T}_{17})\Pi^{2,1}(T_{17})$ . First, we have that  $\Pi(T_{15})$  and  $\Pi(T_{16})$  have the same resonance structure as  $\Pi T_2$ , and hence following the calculations in (4.82) it follows that,

$$(\Pi T_{15})(\tau) = (\Pi T_{16})(\tau) = -i\tau e^{-i\tau k^2} + \tau^2 e^{-i\tau k^2} (k^2 - k_2^2).$$

Furthermore, we have that  $\bar{\Pi}(\tilde{T}_{15})$  has a null resonance structure and hence it follows that  $(\bar{\Pi}\tilde{T}_{15})(\tau) = i\tau e^{i\tau k^2}$  and,

$$\mathcal{Q}_{r\leq 2}(\bar{\Pi}^{2,1}\tilde{T}_{15}\Pi^{2,1}T_{15})(\tau) = \mathcal{Q}_{r\leq 2}(\tau^2 + i\tau^3(k^2 - k_2^2))$$
  
=  $\tau^2$ .

Similarly, we have

$$\mathcal{Q}_{r\leq 2}(\bar{\Pi}^{2,1}\tilde{T}_{16}\Pi^{2,1}T_{16})(\tau) = \tau^2 = \mathcal{Q}_{r\leq 2}(\bar{\Pi}^{2,1}\tilde{T}_{17}\Pi^{2,1}T_{17})(\tau),$$

where the third and fourth order terms are truncated by the operator  $\mathcal{Q}_{\leq 2}$ .

Plugging the results obtained in the above computations into (4.81) yields the second order low regularity scheme (4.76), which is given in physical space by (4.77).

**Local Error:** It remains to show the claimed third order local error bound:  $\mathcal{O}(\tau^3 |\nabla|^2 v)$ . We follow along the same lines made for the first order local error analysis in Section 4.5.1. By Theorem 4.4.3 we have

$$V_k^{2,1}(\tau, v) - V_k(\tau, v) = \sum_{T_1 \cdot T_2 \in \mathcal{G}_0^{1,k}(R)} \mathcal{O}(\tau^3 \mathcal{L}^1_{\text{low}}(T_1 \cdot T_2, 2) \bar{\Upsilon}^p(T_1)(v) \Upsilon^p(T_2)(v))$$

We are left to calculate  $\mathcal{L}_{low}^1(T_1 \cdot T_2, 2)$ , for  $T_1 \cdot T_2 \in \mathcal{G}_0^{1,k}(R)$ . We are interested in the integrals  $\Pi T$ ,  $\Pi \tilde{T}$  which have non-zero resonance structure, since oscillatory integrals who's resonance structure is zero are integrated exactly, and hence do not contribute to the local error analysis of the scheme.

— Computation of  $\mathcal{L}^1_{low}(T_1 \cdot T_2, 2)$ , where  $\Pi T_2$  has the non-zero resonance structure (4.67). Following the analysis made in (4.71) for this pairing we have,

$$R_{o_2,1}^{k,1}(\Pi^{n=2}G_2) = R_{o_2,2}^{k,1}(e^{i\xi(k^2 - 2k_1^2)}) = \left(2(k^2 - k_1^2)\right)^2,$$

where we applied definition (4.36) up to an additional order. Therefore, it follows from the previously made steps (4.71) that

$$\mathcal{L}_{low}^{1}(T_{1} \cdot T_{2}, 2)\bar{\Upsilon}^{p}(T_{1})\Upsilon^{p}(T_{2}) = \mathcal{O}(\sum_{k_{1} \in \mathbf{Z}^{d}} (k^{2} - k_{1}^{2})^{2} \overline{v_{k}} v_{k} |v_{k_{1}}|^{2}).$$
(4.84)

In physical space this yields an error of the form

$$\mathcal{O}\Big(\tau^{3}\Big((|\nabla|^{2}v*|\nabla|^{2}\tilde{v})\|v\|_{L^{2}}^{2} + (|\nabla|v*|\nabla|\tilde{v})\||\nabla|v\|_{L^{2}}^{2}\Big)\Big).$$

$$(v*\tilde{v})\||\nabla|^{2}v\|_{L^{2}}^{2} + (|\nabla|v*|\nabla|\tilde{v})\||\nabla|v\|_{L^{2}}^{2}\Big)\Big).$$
(4.85)

The computations for the remaining pairings  $F \in \mathcal{G}_0^{1,k}(R)$  can be made in an analogous fashion and can be shown to produce an error term of the form  $O(\tau^3 |\nabla| v)$ . In consequence, their approximation requires less regularity on the initial data than that for the above pairing  $T_1 \cdot T_2$ . We conclude from the above expression (4.85) that the local error in physical space is of the form  $\mathcal{O}(\tau^3 |\nabla|^2 v)$ , which requires two derivatives on the initial data.

**Remark 4.5.6** Recently new low-regularity integrators ([AB23b]) and resonance-based discretisations ([FMS23, MS23]) have been introduced, which preserve better the underlying structure of the solution over long-times, and which exhibits improved error constants at low-regularity. Encapsulating the idea behind the construction of these low-regularity structure-preserving schemes into the general framework presented in this article would be of interest in the future. This could yield better insight on the long-time behaviour of the solution on a numerical level.

#### 4.5.2 KdV

Let us next consider the Korteweg-de Vries (KdV) equation

$$\partial_t u + \partial_x^3 u = \frac{1}{2} \partial_x u^2, \quad (t, x) \in \mathbf{R} \times \mathbf{T}$$
(4.86)

with a random initial value u(0) = v of the form (4.2). The KdV equation (4.86) fits into the general framework (4.1) with

$$\mathcal{L}(\nabla) = i\partial_x^3, \quad \alpha = 1, \text{ and } p(u,\overline{u}) = p(u) = i\frac{1}{2}u^2.$$

One has  $\mathcal{L} = \{t_1, t_2\}, P_{t_1} = -\lambda^3$  and  $P_{t_2} = \lambda^3$ . Then, we denoted by | an edge decorated by  $(t_1, 0)$  and by | an edge decorated by  $(t_2, 0)$ . The general framework (4.49) derived in Section 4.4 builds the foundation of the first- and second-order schemes presented below for the KdV equation (4.86).

**Corollary 4.5.7** For the KdV equation (4.86) the general scheme (4.44) takes at first order the form

$$u^{\ell+1} = u^{\ell} * \tilde{u}^{\ell} \tag{4.87}$$

with a local error of order  $\mathcal{O}(\tau^2 \partial_x u)$  and at second-order

$$u^{\ell+1} = u^{\ell} * \tilde{u}^{\ell} + \tau^2 \partial_x^2 (u^{\ell} * \tilde{u}^{\ell})^2 + \frac{\tau^2}{2} \partial_x \left( u^{\ell} * \tilde{u}^{\ell} \right) \left( \sum_{k_1 \in \mathbf{Z}^d} k_1 |u_{k_1}^{\ell}|^2 \right) - \frac{\tau^2}{2} \left( \partial_x u^{\ell} * \partial_x \tilde{u}^{\ell} \right) \left( (u_0^{\ell})^2 + (\overline{u_0^{\ell}})^2 \right)$$
(4.88)

with a local error of order  $\mathcal{O}(\tau^3 \partial_x^2 u)$ , and where  $\tilde{u}^{\ell}(x) = \overline{u^{\ell}(-x)}$ .

**Remark 4.5.8 (Stability)** In view of a practical implementation of the second order scheme (4.88) for KdV, we address the need to stabilize the three terms in the above scheme involving spatial derivatives on the numerical solution. As in the Remark 4.5.3, for the stabilisation of the second order scheme for NLS, we introduce three filter functions which we premultiply in front of each of the last three critical term appearing in the scheme (4.88). We introduce the following three filter functions in Fourier space,

$$\begin{split} \Psi_1(k) &= \varphi_1(-i\tau|k|^2) = \frac{1 - e^{-i\tau|k|^2}}{i\tau k^2}, \\ \Psi_2(k,k_1) &= \frac{1}{(i\tau^{\frac{1}{2}}|k|)(i\tau^{\frac{1}{2}}|k_1|)} \left( e^{i\tau^{\frac{1}{2}}|k|/2} - e^{-i\tau^{\frac{1}{2}}|k|/2} \right) \left( e^{i\tau^{\frac{1}{2}}|k_1|/2} - e^{-i\tau^{\frac{1}{2}}|k_1|/2} \right), \\ \Psi_3(k) &= \operatorname{sinc}^2(\tau^{\frac{1}{2}}|k|), \end{split}$$

where  $\Psi_3(k)$  is the same filter function used for the stabilization of the second order scheme for NLS (see equation (4.78)). We have that

$$\begin{split} \Psi_j(k) &= 1 + \mathcal{O}(\tau |k|^2), \quad |\Psi_j(k)\tau |k|^2 | \le 1, \ j \in \{1,3\}, \ k \in \mathbf{Z}, \\ \Psi_2(k,k_1) &= 1 + \mathcal{O}(\tau (|k|^2 + |k_1|^2)), \quad |\Psi_2(k,k_1)\tau |k| |k_1|| \le 1, \ k, k_1 \in \mathbf{Z}. \end{split}$$

Namely, these three filter functions preserve the low regularity error structure of  $\mathcal{O}(\tau \partial_x^2 u)$ , and provides the following stabilized version of the scheme (4.88) which is given in physical space by,

$$\begin{aligned} u^{\ell+1} &= u^{\ell} * \tilde{u}^{\ell} - i\tau (e^{i\tau\partial_x^2} - 1)(u^{\ell} * \tilde{u}^{\ell})^2 \\ &- \frac{\tau}{2} \Big( (e^{\tau^{\frac{1}{2}}\partial_x/2} - e^{-\tau^{\frac{1}{2}}\partial_x/2})(u^{\ell} * \tilde{u}^{\ell}) \Big) \Big( \sum_{k_1 \in \mathbf{Z}^d} (e^{\tau^{\frac{1}{2}}|k_1|/2} - e^{-\tau^{\frac{1}{2}}|k_1|/2})|u_{k_1}|^2 \Big) \\ &+ \frac{\tau}{2} \Big( (e^{\tau^{\frac{1}{2}}\partial_x/2} - e^{-\tau^{\frac{1}{2}}\partial_x/2})u^{\ell} * (e^{\tau^{\frac{1}{2}}\partial_x/2} - e^{-\tau^{\frac{1}{2}}\partial_x/2})\tilde{u}^{\ell} \Big) ((u_0^{\ell})^2 + (\overline{u_0^{\ell}})^2). \end{aligned}$$

Proof. We proceed as for the Nonlinear Schrödinger equation. The construction of the schemes is again based on the

general framework (4.44). We consider for r = 0, 1 the expansion,

$$V_{k}^{n,r}(\tau,v) = \sum_{F=T_{1}\cdot T_{2}\in\mathcal{G}_{0}^{r,k}(R)} m_{F} \frac{\Upsilon^{p}(T_{1})(v)\,\Upsilon^{p}(T_{2})(v)}{S(T_{1})S(T_{2})}$$

$$\mathcal{Q}_{\leq r+1}(\bar{\Pi}^{n,r}T_{1}\Pi^{n,r}T_{2})(\tau).$$
(4.89)

For the first-order scheme (r = 0) we need only to consider the single paired decorated forest:

$$\mathcal{G}_0^{0,k}(R) = \{T_0 \cdot T_0\}, \quad T_0 = \overset{(k)}{\blacktriangleright}.$$
 (4.90)

Indeed, we recall the first order expansion (4.54), where for the KdV equation we have,

$$u_1(\tau, v^{\eta}) = e^{-\tau \partial_x^3} v^{\eta}, \quad u_2(\tau, v^{\eta}) = e^{-\tau \partial_x^3} \int_0^{\tau} e^{\zeta \partial_x^3} \partial_x (e^{-\zeta \partial_x^3} v^{\eta})^2 d\zeta.$$

Given that for the KdV equation the term  $\mathbb{E}[\overline{u_{k,1}(\tau, v^{\eta})}u_{k,2}(\tau, v^{\eta})]$  consists of a product of three Gaussians, we have that this term is equal to zero. Hence, the only term to compute in the expansion (4.54) is

$$\mathbb{E}[\overline{u_{k,1}(\tau,v^{\eta})}u_{k,1}(\tau,v^{\eta})]$$

and the pairing to consider is (4.90).

A straightforward calculation yields,

$$S(T_0) = 1, \quad \Upsilon^p(T_0)(v) = v_k, \quad m_{T_0 \cdot T_0} = 1,$$

and

$$(\Pi^{n,0}T_0)(\tau) = e^{-i\tau k^3}.$$

In the end, the first order scheme takes on the simple form:

$$V_k^{n,0}(\tau, v) = |v_k|^2.$$

One can notice that the only approximation error made comes from the first-order truncation of the tree series (4.49). Namely, by noting that  $\alpha = 1$ , it follows from (4.50) that  $(V_k^{n,0} - V_k)(\tau, v) = \mathcal{O}(\tau^2 k^2 v_k \bar{v}_k)$ . Therefore, in physical space the local error is of order  $\mathcal{O}(\tau^2(\partial_x v * \partial_x \tilde{v}))$ , which solely requires one additional derivative on the initial data.

For a second order scheme, one has to compute more terms. The set of paired decorated forests is given by:

$$\mathcal{G}_{0}^{1,k}(R) = \{T_{0} \cdot T_{0}, T_{1} \cdot T_{1}, T_{0} \cdot T_{2}, T_{0} \cdot T_{3}, k_{i} \in \mathbf{Z}^{d}\},\$$

$$\overset{(k_{1})}{\overset{(k_{2})}{\overset{(k_{2})}{\overset{(k_{3})}{\overset{(k_$$

A straightforward computation gives:

$$S(T_1) = 2, \quad S(T_2) = S(T_3) = 2, \quad \Upsilon^p(T_1)(v) = v_{k_1}v_{k_2}$$
  
$$\Upsilon^p(T_2)(v) = |v_{k_1}|^2 v_k, \quad \Upsilon^p(T_3)(v) = v_0^2 v_k$$
  
$$m_{T_1 \cdot T_1} = 2, \quad m_{T_0 \cdot T_2} = 1, \quad m_{T_0 \cdot T_3} = 2.$$

Further, due to cancellations in the resonance, the integrals encoded by the trees  $T_2$  and  $T_3$  have a null resonance structure, and hence it follows that

$$(\Pi^{n,1}T_0)(\tau) = e^{-i\tau k^3}, \quad (\Pi^{n,1}T_2)(\tau) = -\frac{\tau^2}{2}k_1ke^{-i\tau k^3},$$
$$(\Pi^{n,1}T_3)(\tau) = -\frac{\tau^2}{2}k^2e^{-i\tau k^3}.$$

For the computation of  $(\Pi^{n,1}T_1)(\tau)$ , we proceed by a full Taylor expansion which yields

$$(\Pi^{n,1}T_1)(\tau) = e^{-i\tau k^3} \mathcal{K}^{k,1}_{(t_2,1)} \left( e^{i\xi(-k_1^3 - k_2^3)}, n \right)(\tau)$$
  
=  $e^{-i\tau k^3} \left( i\tau k + \frac{3}{2}\tau^2 k_1 k_2 k^2 \right),$ 

where we used the fact that the resonance structure is given by,

$$P_{(t_2,0)}(k) - k_1^3 - k_2^3 = k^3 - k_1^3 - k_2^3 = 3k_1k_2(k_1 + k_2) = 3k_1k_2k.$$

Then, by taking the product and making the truncation, we obtain:

$$\mathcal{Q}_{\leq 2}(\bar{\Pi}^{n,1}T_1)(\tau)(\Pi^{n,1}T_1)(\tau) = \mathcal{Q}_{\leq 2}\left((-i\tau k + \frac{3}{2}\tau^2 k_1 k_2 k^2)(i\tau k + \frac{3}{2}\tau^2 k_1 k_2 k^2)\right)$$
$$= \tau^2 k^2.$$

The local error introduced is of order  $\mathcal{O}(\tau^3 k^4)$  which corresponds to two derivatives due to the convolution structure. In the end, the scheme in Fourier space is given by:

$$V_k^{n,1}(\tau, v) = |v_k|^2 + \tau^2 k^2 \sum_{k=k_1+k_2} |v_{k_1}|^2 |v_{k_2}|^2 - \sum_{k_1 \in \mathbf{Z}^d} \frac{\tau^2}{2} k_1 k |v_{k_1}|^2 |v_k|^2 - \frac{\tau^2}{2} k^2 |v_k|^2 (v_0^2 + \overline{v_0}^2).$$

By writing the above scheme in physical space we recover the scheme (4.88), which induces a local error of  $\mathcal{O}(\tau^3 \partial_x^2 v)$ .

# Part III

Structure preserving schemes suited for low-regularity approximations

## Chapter 5

## A symmetric low-regularity integrator for the nonlinear Schrödinger equation

This chapter is based on the article [AB23b] published in the IMA Journal of Numerical Analysis.

Abstract. We introduce and analyze a symmetric low-regularity scheme for the nonlinear Schrödinger (NLS) equation beyond classical Fourier-based techniques. We show fractional convergence of the scheme in  $L^2$ -norm, from first up to second order, both on the torus  $\mathbb{T}^d$  and on a smooth bounded domain  $\Omega \subset \mathbb{R}^d$ ,  $d \leq 3$ , equipped with homogeneous Dirichlet boundary condition. The new scheme allows for a symmetric approximation to the NLS equation in a more general setting than classical splitting, exponential integrators, and low-regularity schemes (i.e. under lower regularity assumptions, on more general domains, and with fractional rates). We motivate and illustrate our findings through numerical experiments, where we witness better structure preserving properties and an improved error-constant in low-regularity regimes.

## 5.1 Introduction

We consider the nonlinear Schrödinger (NLS) equation,

$$i\partial_t u(t,x) = -\Delta u(t,x) + |u(t,x)|^2 u(t,x), \quad (t,x) \in \mathbb{R} \times \Omega$$
(5.1)

with  $\Omega \subset \mathbb{R}^d$  or  $\Omega = \mathbb{T}^d$ ,  $d \leq 3$ , and an initial condition

$$u_{|t=0} = u_0. (5.2)$$

When  $\partial \Omega \neq \emptyset$ , we assume that  $\Omega$  is a smooth bounded domain and we assign homogeneous boundary conditions which will be encoded in the choice of the domain of the operator  $i\Delta$ . In the convergence analysis we will consider either periodic or homogeneous Dirichlet boundary conditions. Nevertheless, one could also consider different types of boundary conditions such as homogeneous Neumann boundary conditions by defining the functional spaces accordingly (see Section 5.2).

Throughout this article we will be interested in providing a reliable approximation of (5.1) when the initial data  $u_0$  are non-smooth, in the sense that they belong to Sobolev spaces of low order. Namely, we will be interested in studying numerical schemes which approximate the time dynamics of (5.1) at low regularity. The numerical study of low-regularity approximations to nonlinear evolution equations has gained lots of attention in the past years, and numerous contributions have been made in this direction. The first results were established on the torus for the Korteweg-De Vries (KdV) equation and then the NLS equation with the pioneering works of [HS17b] and [OS18]. These results could be further greatly extended, see for instance [WZ22b, NWZ22, RS22, LW22] and [ORS22b, WY22, LW21, OWY22] for the KdV and NLS equations respectively. More types of dispersive equations could be dealt with, including for example the Dirac equation [SWZ21] or the "good" Boussinesq equation ([LS23]), and a general framework for constructing low-regularity approximations up to arbitrary order and for a class of dispersive equations on the torus was obtained in [BS22].

The construction of these time integrators (called *resonance-based schemes, exponential-type low-regularity integrators*, or *Fourier integrators*) strongly depended on Fourier-based expansions, and hence were restricted to periodic boundary conditions. Recently, this restriction was withdrawn to treat more general domains  $\Omega \subset \mathbb{R}^d$  and boundary conditions, as well as more general nonlinearities (see [RS21, AB23a, ABBS22b, BLW23, LMS21]).

The general aim of low-regularity integrators is that they converge under lower regularity assumptions, contrarily to classical methods (see [ABBS22b] for a comparative analysis on general smooth domains). Their major drawback is

that they do not preserve the geometric structure of the underlying system. The NLS equation (5.1) is time reversible, meaning that u(-t, x) is again solution of (5.1), and both the density and the energy are conserved quantities:

$$\begin{cases} \|u(t)\|_{L^2} &= \|u_0\|_{L^2}, \\ E(t) &= E_0, \end{cases} \quad t \in I,$$
(5.3)

where I is the interval of existence of the solution, and where for  $H^1$ -solutions we have

$$E(t) = \frac{1}{2} \int |\nabla u|^2(t, x) dx + \frac{1}{4} \int |u|^4(t, x) dx.$$

Hence, when designing a numerical scheme it is natural to take into account both of these conserved quantities, and to retain (as much as possible) these properties also on the discrete level by introducing so-called structure preserving schemes, see [HLW10] for an extensive introduction on the subject. The latter has received great interest thanks to their good long-time near-preservation of the actions of the integrable properties of the equation, and have been successfully studied in the past for the approximation of the NLS equation (5.1). Examples of such schemes are splitting schemes ([Lub08], [Fao12]), relaxation finite difference type schemes ([Bes04]), symmetric exponential integrators ([CC008]) or Crank-Nicolson Galerkin methods ([HP17]), just to name a couple of them. For an overview of symmetric methods for NLS see [CC008, Fao12, ABB13]. While these classical structure preserving schemes provide excellent approximations to smooth solutions in general even up to long times, they often break down and lead to severe loss of convergence for non-smooth solutions. Low-regularity integrators which are suited for non-smooth solutions on the other hand do not preserve the structure of the underlying equation. The natural question which thus arises is: *What about low-regularity structure preserving schemes for solving the NLS equation* (5.1)? Only very little is known in this direction, see the work of [MS22] on the KdV equation, [WZ22a] on the cubic Klein-Gordon equation, and [BMS22] on the isotropic Landau–Lifschitz equation. Also worth to be mentioned is the work of [WY22] which introduces for the first time a first-order Fourier integrator for the NLS equation (5.1) set on  $\mathbb{T}$  which almost conserves the mass.

In this article, we introduce a symmetric low-regularity integrator for solving the NLS equation (5.1) which allows for low-regularity approximation while maintaining good long-time preservation of the two conserved properties (5.3) on the discrete level. We carry-out a rigorous convergence analysis in  $L^2(\Omega)$  on smooth domains  $\Omega \subset \mathbb{R}^d$  and obtain improved error estimates at low-regularity compared to classical symmetric methods. Our numerical findings not only show better structure preservation properties but also show a much better error constant at low-regularity than previously proposed methods (see Figure 5.1).

In the finite dimensional ODE setting it is well-known that symmetric methods are of even order. In the context of PDEs this is a much more delicate question as convergence is met only when sufficient regularity assumptions are imposed on the solution. Thanks to the gain of symmetry, we show second order  $L^2$ -convergence of the symmetric scheme under less regularity assumptions than what is required by classical symmetric schemes ([Lub08, Bes04, BDDLV21, HP17]), while asking for slightly more regularity than asymmetric second-order low-regularity schemes ([AB23a]) which however do not preserve the structure of the system (see Figures 5.2 and 5.3). Optimal first order low-regularity convergence rates could be obtained. See Section 5.1.2 for a detailed discussion on the subject.

The scheme we present here is based on the first-order low-regularity scheme first introduced in [OS18] which is given by,

$$\Phi_{\tau}(u^n) := e^{i\tau\Delta} \left( u^n - i\tau(u^n)^2 \varphi_1(-2i\tau\Delta)\overline{u^n} \right), \quad u^0 = u_0,$$
(5.4)

where  $\varphi_1(z) = \frac{e^z - 1}{z}$ , and  $\tau$  is the time step. In order to symmetrize the above scheme we introduce the adjoint method as the map

$$\hat{\Phi}_{\tau} = \Phi_{-\tau}^{-1}$$

and compute (see [HLW10])

$$\Phi_{\tau/2} \circ \Phi_{\tau/2}$$

This yields the following implicit symmetric low-regularity scheme,

$$u^{n+1} = \varphi^{\tau}(u^n) = e^{i\tau\Delta}u^n - i\frac{\tau}{2}e^{i\tau\Delta}\left((u^n)^2\varphi_1(-i\tau\Delta)\overline{u^n}\right) - i\frac{\tau}{2}\left((u^{n+1})^2\varphi_1(i\tau\Delta)\overline{u^{n+1}}\right)$$
(5.5)  
$$= e^{i\tau\Delta}u^n + \psi_E^{\tau/2}(u^n) + \psi_I^{\tau/2}(u^{n+1})$$
$$= e^{i\tau\Delta}u^n + \Psi^{\tau}(u^n, u^{n+1}),$$

which satisfies the discrete analogue of the time-reversible property of (5.1).

We highlight the properties which the scheme (5.5) inherits through numerical experiments, where we couple the time-integrators with the standard Fourier pseudo-spectral method which encodes periodic boundary conditions. The case of homogeneous Dirichlet boundary conditions remains very similar, yet for completeness we also include a convergence plot in this case where we expand the solution as a sine series expansion. First, in the case of periodic boundary

#### 5.1. Introduction

conditions, we observe in Figures 5.1a and 5.1b the favorable convergence properties of the scheme (5.5) for  $H^1$  and  $H^2$  data respectively. We notice that the error constant of the symmetric scheme is much better than the asymmetric first-order low-regularity integrator (Low-reg 1), and is also better than the asymmetric second-order low-regularity integrator (Low-reg 2). Figure 5.1c similarly shows the favorable convergence behavior when considering homogeneous Dirichlet boundary conditions. Secondly, we study in Figures 5.2 and 5.3 the structure preserving properties of the new symmetric low-regularity integrator (5.5) against previous asymmetric low-regularity integrators. We witness that the asymmetric first and second-order low-regularity integrators (Low-reg 1, Low-reg 2) are unable to preserve the density and energy (see (5.3)), whereas the symmetric integrator (5.5) appears to nearly-preserve both conserved properties over long-times. We note that in the finite dimensional ODE setting a general theory for symmetric methods applied to integrable reversible systems has been established in [HLW10] allowing for long-time near-conservation of first-integrals. In the infinite dimensional case the understanding of the long-time behavior of numerical solutions is an ongoing challenge in the field of geometric integration and few results are known, see for example [Fao12, FGP10a, GL10b, CHL08]. We expect that it would be possible to prove long-time near-preservation of the density and energy of the scheme (5.5) by using the results of [HLW10], and by benefitting of an analysis using modulated Fourier expansions (see [CHL08, GL10b]) or using normal form techniques ([FGP10a, BG22]) to show near-conservation of the energy. This delicate analysis is out of scope for this paper, where here we focus on the low-regularity error estimates on the solution itself. Finally, we refer



Figure 5.1 – Convergence plot for data in  $H^1$  (Figure (a)) and data in  $H^2$  (Figure (b) and (c)) of the asymmetric first and second-order low-regularity integrators (pink, dark blue), the symmetric method (5.5) (red), the classical Lie splitting, Strang splitting, and Euler Exponential method (light blue, yellow, and green). We observe order reduction of the classical Euler Exponential and Lie and Strang splitting methods (Figure (a),  $H^1(\mathbb{T})$ -data), and of the Strang splitting method (Figure (b),  $H^2(\mathbb{T})$ -data, and Figure (c),  $H^2([0,1])$ -data). Figures (a) and (b) are with periodic boundary conditions, while Figure (c) is with homogeneous Dirichlet boundary conditions. The slopes of the continuous black lines are one and two, respectively. We took the final time T = 1, and the number of Fourier modes  $K = 2^{11}$ . The figure is taken from [AB23b].



Figure 5.2 – Plot (a) : We graph the  $L^2$ -norm  $||u(t)||_{L^2(\mathbb{T})}$  up until time T = 50 of three low-regularity integrators. The asymmetric first and second-order low-regularity integrators (pink, blue), and the new symmetric low-regularity integrator (red). We also graph the exact value  $||u_0||_{L^2(\mathbb{T})}$  (black). Plot (b): We only graph the asymmetric second-order low-regularity integrators (blue), and the symmetric low-regularity integrators (blue), and the fixed the number of Fourier modes  $K = 2^9$ , the time step  $\tau = 0.05$ , and took an initial data  $u_0 \in H^2$ . The figures is taken from [AB23b].



Figure 5.3 – Plot (a) : We graph the relative energy  $E(t)/E(t_0)$  (see (5.3)) up to T = 50 of the same three low-regularity integrators as in Figures 5.2a and 5.2b. Plot (b): We only graph the relative energy of the asymmetric second order low-regularity integrators (blue), and the symmetric low-regularity integrator (red). We again fixed the number of Fourier modes  $K = 2^9$ , the time step  $\tau = 0.05$ , and took an initial data  $u_0 \in H^2$ .

to Figure 5.4 for a broad indication of the relative computational cost of each of the three low-regularity integrators, and discuss the added cost of implementing the symmetric implicit scheme (5.5). We observe that the asymmetric second-order low-regularity integrator (Low-reg 2) costs in CPU-time approximately the same as the symmetric integrator (5.5). Whereas when comparing with the asymmetric first-order scheme (Low-reg 1) we have that the improved convergence properties of the scheme (5.5) make up for the extra cost of solving the implicit system (5.5) at every time step.

This numerical study motivates the use of the scheme (5.5), which conserves better the underlying geometric structure of the equation, exhibits a better error constant, and can be implemented at relatively low additional cost despite its implicit nature.

**Remark 5.1.1 (Implicit versus explicit low-regularity schemes)** We make the important remark that unlike the previous (asymmetric) low-regularity integrators [OS18, BS22, ORS21, ORS22b, RS21, AB23a, ABBS22b] the above symmetrized scheme (5.5) is an *implicit* one. We have witnessed that the implicit nature of the scheme does not adversely affect the computational cost of the method (see Figure 5.4). Nevertheless, one could query on the necessity of the



Figure 5.4 – We plot the CPU time versus the  $L^2$ -error. Namely, we compare the computational cost for running the first and second-order low-regularity scheme (pink, dark blue), with the low-regularity symmetric integrator (5.5) (red). We took the same parameter values as in Figure 5.1. The figure is taken from [AB23b].

implicit nature of the symmetric low-regularity scheme. In the case of second-order wave-type equations, instead of considering implicit symmetrized schemes one could study explicit three-time step symmetric schemes using Gautschi-type methods. Indeed, for the cubic Klein-Gordon equation set on  $\mathbb{T}$ , an *explicit* symmetric three time-step low-regularity integrator could be obtained by [WZ22a]. While this approach is suited to second order equations, by combining the work of [WZ22a] together with the uniformly accurate low-regularity integrator [CS22a] adapted to the non-relativistic regime, an interesting open problem would be to obtain in the non-relativistic limit an *explicit* symmetric three time-step low-regularity approximation to the NLS equation (5.1).

Having motivated the scheme (5.5) we now provide the underlying idea behind its construction. To provide a better intuition to the reader we will work in the twisted variable and place ourselves on the torus  $\Omega = \mathbb{T}$  in order to make use of Fourier-based expansions (see also [OS18]).

#### 5.1.1 Derivation of the scheme

We switch to the twisted variable  $v = e^{-it\Delta}u$ . We observe that v satisfies

$$i\partial_t v = e^{-it\Delta} (|e^{it\Delta}v|^2 e^{it\Delta}v), \quad v_0 = u_0.$$

Equivalently, by integrating the above and mapping Duhamel's formula in Fourier space we have

$$v(t_{n+1}) = v(t_n) - i \sum_{k=-k_1+k_2+k_3} e^{ikx} e^{it_n(k^2+k_1^2-k_2^2-k_3^2)} I^{\tau},$$
(5.6)

where the oscillatory integral is given by

and  $h(s) = e^{i\omega_2 s} g(s), \, \omega_1 = 2k_1^2, \, \omega_2 = k^2 - k_1^2 - k_2^2 - k_1^2 - k_1^2 - k_2^2 - k_1^2 - k_1^2 - k_2^2 - k_1^2 - k_$ 

$$I^{\tau} = \int_{0}^{\tau} e^{i\omega_{1}s} h(s) ds,$$

$$k_{2}^{2} - k_{3}^{2} = -2k_{1}(k_{2} + k_{3}) + k_{2}k_{3} \text{ and}$$
(5.7)

$$g(s) = \overline{v}_{k_1}(t_n + s)v_{k_2}(t_n + s)v_{k_3}(t_n + s).$$

The central question revolves around making a suitable choice of discretization of the oscillatory integral  $I^{\tau}$ , with the aim of minimizing the regularity assumption required by this approximation. The underlying idea behind the construction of the previous (asymmetric) low-regularity integrators (or *resonance-based schemes*) is to choose an approximation of the integral  $I^{\tau}$  which allows for a practical implementation (by not performing exact integration), while optimizing the local error in the sense of regularity. Namely, by recalling that  $2k_1^2$  corresponds to second order derivatives in Fourier space while the terms  $k_m k_i$  (for  $m \neq j$ ) correspond to product of first order derivatives, the idea is to separate the dominant
$(\omega_1)$  and lower-order  $(\omega_2)$  frequencies. The lower-order and non-oscillatory part h(s) is then approximated by a Taylor series expansion centered at s = 0,

$$h(s) = h(0) + O(sw_2g),$$

and the dominant part  $e^{i\omega_1 s}$  is integrated exactly. This yields the first-order low-regularity scheme (5.4) with a local error of  $O(\tau^2 \partial_x v)$ . At low regularity this is more advantageous than classical techniques (such as exponential integrators [HO10] or splitting methods [Lub08]) which do not embed the dominant frequency interactions into the scheme and obtain a local error of  $O(\tau^2 \partial_x^2 v)$ .

The key idea behind obtaining the symmetric scheme (5.1) is to make a different Taylored discretization of the lower-order and non-oscillatory part h(s). Namely, we again integrate exactly and embed the dominant part  $e^{i\omega_1 s}$  into the numerical scheme, while this time approximating the non-dominant part in the following symmetric fashion,

$$h(s) \approx h(0)\mathbb{1}_{[0,\tau/2]} + h(\tau)\mathbb{1}_{[\tau/2,\tau]}, \quad s \in [0,\tau],$$
(5.8)

where  $\mathbb{1}_A$  is the indicator function on the set A. By plugging this approximation for h into the oscillatory integral (5.7) yields two terms: an explicit and an implicit one. The explicit term is given by,

$$\int_{0}^{\tau/2} e^{i\omega_{1}s} h(0) ds = \tau \frac{e^{i\omega_{1}\tau/2} - 1}{i\omega_{1}\tau} \overline{v_{k_{1}}^{n}}(t_{n}) v_{k_{2}}^{n}(t_{n}) v_{k_{3}}^{n}(t_{n}) = \frac{\tau}{2} \varphi_{1}(ik_{1}^{2}\tau) \overline{v_{k_{1}}^{n}}(t_{n}) v_{k_{2}}^{n}(t_{n}) v_{k_{3}}^{n}(t_{n}).$$

Using the definition of the twisted variable, equation (5.6), and by mapping the above back to physical space yields the explicit nonlinear term  $\psi_E^{\tau/2}(u^n)$  in the scheme (5.5). Similarly, one obtains the nonlinear implicit term in (5.5) by using the definitions  $\omega_1 = 2k_1^2$ ,  $\omega_2 = k^2 - k_1^2 - k_2^2 - k_3^2$  and noticing that

$$h(\tau) = e^{ik^2\tau} \left( \left( e^{-ik_2^2\tau} v_{k_2}^n(t_{n+1}) \right) \left( e^{-ik_3^2\tau} v_{k_3}^n(t_{n+1}) \right) \left( e^{-ik_1^2\tau} \overline{v_{k_1}^n}(t_{n+1}) \right) \right)$$

and

$$\begin{split} \int_{\tau/2}^{\tau} e^{i\omega_1 s} h(\tau) ds &= e^{ik^2 \tau} \left( \left( e^{-ik_2^2 \tau} v_{k_2}^n(t_{n+1}) \right) \left( e^{-ik_3^2 \tau} v_{k_3}^n(t_{n+1}) \right) \int_0^{\tau/2} e^{2ik_1^2(\tau-s)} ds \left( e^{-ik_1^2 \tau} \overline{v_{k_1}^n}(t_{n+1}) \right) \right) \\ &= e^{ik^2 \tau} \left( \left( e^{-ik_2^2 \tau} v_{k_2}^n(t_{n+1}) \right) \left( e^{-ik_3^2 \tau} v_{k_3}^n(t_{n+1}) \right) \frac{\tau}{2} \varphi_1(-i\tau k_1^2) \left( e^{+ik_1^2 \tau} \overline{v_{k_1}^n}(t_{n+1}) \right) \right). \end{split}$$

We note that a general approach to obtain the approximation (5.8) is to first give a symmetric approximation to the non-oscillatory part g(s) by iterating Duhamel's formula inside  $v_k(t_n + s)$  in a symmetric fashion. Namely,  $v_k(t_n + s)$  is approximated on  $[0, \frac{\tau}{2}]$  by the linear term in the Duhamel formula centered about s = 0 (yielding the approximation g(0), see (5.6)). While on  $(\frac{\tau}{2}, \tau]$ ,  $v_k(t_n + s)$  is approximated by the linear term in the Duhamel formula centered about  $s = \tau$  (yielding the approximation  $g(\tau)$ ). We then proceed by approximating the lower-order oscillatory part  $e^{i\omega_2 s}$  in a symmetric fashion. In order to obtain higher-order symmetric low-regularity approximations, we would iterate inside g(s) both of these Duhamel expansions (centered about s = 0 on  $[0, \frac{\tau}{2}]$  and about  $s = \tau$  on  $(\frac{\tau}{2}, \tau]$ ) up to higher order. The construction of higher-order low-regularity symmetric integrators will be dealt with in future work.

Remark 5.1.2 (Third order local error bound) We make an important point related to the third-order local error structure of the scheme (5.5). Thanks to the symmetry of the scheme (5.5) we can expect to have second-order convergence under suitable regularity assumptions on the solution. From the above calculations in Fourier one easily observes that we naturally need three additional derivatives in order to obtain a third-order local error bound of the scheme (5.5). Indeed, by Taylor expanding around the midpoint one observes that the error induced by the discretization (5.8) of  $I^{\tau}$  requires the boundedness of a term of the form,

$$\int_{-\tau}^{0} e^{i\omega_{1}(\frac{s+\tau}{2})} \left( e^{i\omega_{2}(\frac{s+\tau}{2})} - 1 \right) dsg(0)$$

$$+ \int_{0}^{\tau} e^{i\omega_{1}(\frac{s+\tau}{2})} \left( e^{i\omega_{2}(\frac{s+\tau}{2})} - e^{i\omega_{2}\tau} \right) dsg(0).$$
(5.9)

For the above to yield a third order term one needs to bound a term of order  $O(\tau^3 \omega_1 \omega_2 \bar{v}_{k_1} v_{k_2} v_{k_3})$  which corresponds in physical space to asking for three additional derivatives on  $\bar{v}$ . This is to be compared with classical (symmetric) schemes which usually have a local error of  $O(\tau^3 \partial_x^4 v)$  (see for example [Lub08] for splitting schemes), and to asymmetric resonance-based schemes which merely asks for  $O(\tau^3 \partial_x^2 v)$ , ([BS22, OWY22]).

While we motivated this symmetric low-regularity integrator on a periodic domain, we show that it also allows for a low-regularity approximation on general smooth domains by establishing its convergence at low-regularity (see Section 5.1.2). Throughout the remainder of this article we will work on general smooth domains  $\Omega \subset \mathbb{R}^d$  and make use of semi-group theory to derive our scheme and establish our convergence result on general domains (see Section 5.4). This differs from the first structure preserving low-regularity integrators [MS22, BMS22] which are restricted to periodic boundary conditions.

We now enter the main bulk of this paper, which answers the question of what can be rigorously proven on the  $L^2$ -convergence of the scheme (5.5) when set on a general smooth domain.

We state and prove  $L^2$ -fractional convergence results, from first to second order, both on the torus  $\mathbb{T}^d$  and on a smooth bounded domain  $\Omega \subset \mathbb{R}^d$ , under moderate regularity assumptions on the solution u. These are stronger convergence results than the more typical  $H^{\sigma}(\mathbb{T}^d)$  ( $\sigma > \frac{d}{2}$ ) -convergence analysis, which is restricted to an analysis in smooth Sobolev spaces and to periodic boundary conditions. We state our results in the next subsection.

### 5.1.2 Result

**Theorem 5.1.1** ( $\Omega = \mathbb{T}^d$ ) Let T > 0,  $d \leq 3$ , and  $u_0 \in H^{\alpha}(\mathbb{T}^d)$  with  $\alpha \in [1 + \frac{d}{4}, 3]$ . Let  $u \in \mathcal{C}([0, T], H^{\alpha}(\mathbb{T}^d))$  be the unique solution of (5.1). Then there exist  $\tau_{\min} > 0$  depending on T and on  $||u_0||_{H^{\alpha}}$ , and  $C_T$  a positive function depending on T and  $\sup_{[0,T]} ||u(t)||_{H^{\alpha}}$ , such that for every time step size  $\tau \leq \tau_{\min}$  the numerical solution  $u^n$  given in equation (5.5) has the following error bound:

$$\|u(n\tau) - u^n\|_{L^2} \le C_T(\sup_{[0,T]} \|u(t)\|_{H^{\alpha}})\tau^{1+\gamma}, \quad 0 \le n\tau \le T,$$
(5.10)

for  $\alpha$  and  $\gamma \in [0, 1]$  which satisfy

$$\begin{cases} \alpha > 1 + \frac{d}{2} \text{ and } 0 \le \gamma \le \frac{\alpha - 1}{2}, \\ \alpha = 1 + \frac{d}{2} \text{ and } 0 \le \gamma < \frac{d}{4}, \\ \alpha < 1 + \frac{d}{2} \text{ and } 0 \le \gamma \le \alpha - 1 - \frac{d}{4}. \end{cases}$$
(5.11)

We now consider the case where  $\Omega$  is a smooth bounded domain. Given that in this case the space  $X^s$  (see Section 5.2) in which the solution belongs depends not only on Sobolev regularity but also on compatibility conditions which the solution must satisfy on the boundary, we divide the statement of our results depending on the compatibility conditions imposed on  $u|_{\partial\Omega}$ , (and on the order of convergence).

**Theorem 5.1.2** ( $\Omega \subset \mathbb{R}^d$  smooth bounded domain) Let  $\Omega$  be a smooth bounded domain of  $\mathbb{R}^d$ . We consider the NLS equation (5.1), equipped with homogeneous Dirichlet boundary conditions. Given any T > 0, and  $d \leq 3$ , there exists  $\tau_{min} > 0$  depending on T and the norm of the initial data such that we have the following:

1. Given any  $u_0 \in (H^{1+d/4} \cap H_0^1)(\Omega)$  we have first-order convergence of the symmetric scheme (5.5),

$$\|u^n - u(n\tau)\|_{L^2} \le C_T \tau,$$

for all  $\tau \leq \tau_{\min}$ , and  $0 \leq n\tau \leq T$ .

2. More generally, given any  $u_0 \in (H^{\alpha} \cap H_0^1)(\Omega)$  with  $\alpha \in [1 + \frac{d}{4}, 2]$  we have the fractional convergence estimates (5.10) for  $\alpha$  and  $\gamma$  which satisfies (5.26). In particular, we have

$$\|u(n\tau) - u^n\|_{L^2} \le C_T \begin{cases} \tau^{1+\frac{\alpha-1}{2}} & \text{if } 1 + \frac{d}{2} < \alpha \le 2, \\ \tau^{1+\frac{d}{4}-\epsilon} & \text{for } \alpha = 1 + \frac{d}{2}, \\ \tau^{\alpha-d/4}, & \text{for } 1 + \frac{d}{4} \le \alpha < 1 + \frac{d}{2}, \end{cases}$$

for  $0 \le n\tau \le T$ ,  $\tau \le \tau_{min}$ , and for any  $\epsilon > 0$ .

3. By allowing for more compatibility condition on the boundary we have the following second-order convergence result for an initial data  $u_0 \in X^3 = \{u \in H^3(\Omega) : u|_{\partial\Omega} = 0, \Delta u|_{\partial\Omega} = 0 \text{ in } L^2(\partial\Omega)\},\$ 

$$||u^n - u(n\tau)||_{L^2} \le C_T \tau^2$$

for all  $\tau \leq \tau_{\min}$ , and  $0 \leq n\tau \leq T$ .

We start by making a few remarks on Theorem 5.1.1, set on the torus  $\mathbb{T}^d$ . Let us first mention that the symmetric low-regularity integrator (5.5) requires less regularity assumptions than classical symmetric schemes (see [ESS16, Lub08, BDDLV21, HP17]). Indeed, for example the authors [ESS16] require  $H^2$  solutions to obtain first order convergence of a Lie splitting scheme for NLS, and the author [Lub08] requires  $H^4$ -solutions for second-order convergence of a Strang splitting method, whereas we require  $H^{1+\frac{d}{4}}$  and  $H^3$  to obtain first and resp. second order convergence. We compare this result to previous convergence results of explicit low-regularity integrators for the NLS equation (5.1), which are not symmetric and hence do not have good structure preservation properties (see Figures 5.2, 5.3). To the best of our knowledge, this is the first fractional convergence results of a low-regularity schemes to be obtained from first to second order. We compare our full first and second-order convergence result with the work of [AB23a], which also obtains first order convergence in  $L^2(\mathbb{T}^d)$  for solutions  $u(t) \in H^{1+d/4}(\mathbb{T}^d)$ . For the second order convergence in  $L^2$  of their asymmetric second-order low-regularity integrator the author [AB23a] asks for solutions  $u(t) \in H^{2+d/4}$ , whereas the symmetric low-regularity integrator (5.5) requires a bit more regularity, namely  $H^3$  solutions. Moreover, convergence of order  $\tau^{1+\gamma}$  in  $H^r$ -norm, r > d/2, for  $u_0 \in H^{2\gamma+r+1}(\mathbb{T}^d)$  easily follows from the proof of Theorem 5.1.1. This is to be compared with asymmetric resonance-based schemes which would typically ask for  $u_0 \in H^{\gamma+r+1}(\mathbb{T}^d)$ ,  $r > \frac{d}{2}$ . See [OS18, OWY22] for a first and resp. second order analysis. We refer to Remark 5.1.2 which discusses the necessity of requiring three additional derivatives on the solution to obtain second-order convergence of the scheme (5.5).

We finish by comparing our result with the work of [BMS22], which introduced a symmetrized low-regularity integrator for the Schrödinger map (SM), where they relate the SM flow to the NLS equation set on 1-d torus  $\mathbb{T}$  via the Hasimoto transform. The analysis of their scheme is however restricted to the 1-d torus, and to first order convergence in smooth Sobolev spaces  $H^r(\mathbb{T})$ , r > 1/2. The results we present here go beyond the more typical  $H^r(\mathbb{T})$  error analysis  $(r > \frac{1}{2})$ , by pushing down the error analysis to  $L^2$  for first and up to second order convergence. Furthermore, we do not restrict ourselves to Fourier-based techniques, and hence to periodic boundary conditions, as is testified by Theorem 5.1.2. Using the techniques presented in this article, one can also obtain a symmetric low-regularity approximation to the Schrödinger map in a more general setting than [BMS22].

We now comment upon Theorem 5.1.2. To our knowledge, this is the first convergence result which goes beyond the first-order convergence analysis of a low-regularity integrator when set on a smooth bounded domain  $\Omega \subset \mathbb{R}^d$ . We refer to [RS21, Corollary 20] where the authors show first order convergence in  $L^2(\Omega)$  of the asymmetric low-regularity scheme (5.4) while analogously asking for  $(H^{1+\frac{d}{4}} \cap H^1_0)(\Omega)$  solutions.

We also compare our result to the work of [HP17] which introduces a mass and energy conserving variant of the Crank-Nicolson method as its time-discretization. They show first order convergence on a smooth bounded domain  $\Omega \subset \mathbb{R}^d$  under -among other assumptions-  $u_t \in L^2(0,T; H^2(\Omega))$ , and obtain second order convergence under -among other assumptions-  $u_{tt} \in L^2(0,T; H^2(\Omega))$ , while assuming  $u \in C([0,T], H^2(\Omega))$  throughout their analysis. In contrast to the above classical results Theorem 5.1.2 permits less regularity assumptions on u(t), namely less than  $H^2$ -solutions for first order, and less than  $H^4$ -solutions for second order. We note that the analysis presented here works analogously when adding a potential term uV to equation (5.1), as is considered in [HP17]. One would need to ask for the same regularity assumption (and boundary conditions) on V as is required on u in the above theorem. This follows exactly as done in [AB23a]. The case of a rougher potential ( $V \in L^{\infty}(\Omega)$ ) is dealt in the works of [HP17, BW23b, BMW23, MWZ24].

We also mention that for the 1-d NLS equation with Neumann boundary conditions a low-regularity integrator has been introduced by [BLW23], where using harmonic analysis techniques they could prove up to almost first order convergence with  $H^1$ -data.

## 5.1.3 Outline of the paper

In Section 5.2 we set the scene and introduce the spaces and norms, together with crucial nonlinear estimates, which we will work with throughout the error analysis section. In Section 5.3 we analyze the implicit nature of the scheme; we show that it is well-defined and establish a crucial a priori estimate on the numerical solution. Finally, in Section 5.4 we prove the fractional global error estimates presented in Theorems 5.1.1 and 5.1.2. First, in Section 5.4.1 the fractional local error bounds are obtained, followed by Section 5.4.2 where the stability estimate is shown, and from which the convergence results then naturally follow.

# 5.2 Norms, spaces, and nonlinear estimates

The norm and space used during the error analysis will depend on the domain  $\Omega$  and boundary conditions imposed. We will treat the case where  $\Omega = \mathbb{T}^d$  with periodic boundary conditions, and the case of homogeneous Dirichlet boundary conditions when placed on a smooth bounded domain of  $\mathbb{R}^d$ . In the case where  $\Omega = \mathbb{T}^d$  the domain of the operator  $\mathcal{L} = -\Delta$  is  $D(\mathcal{L}) = H^2(\mathbb{T}^d)$ , whereas for Dirichlet boundary conditions we have that  $D(\mathcal{L}) = (H^2 \cap H_0^1)(\Omega)$ . We can define powers of  $\mathcal{L}$ ,  $\mathcal{L}^s$ , for  $s \ge 0$  using the spectral resolution, and define the space  $X^s(\Omega) = \mathcal{D}(\mathcal{L}^{s/2})$  as the domain of the operator  $\mathcal{L}^{s/2}$ , where  $X^0(\Omega) = L^2(\Omega)$ . We define the norm on  $X^s(\Omega)$  by the usual graph norm

$$||u||_{s}^{2} = ||u||^{2} + ||\mathcal{L}^{s/2}u||^{2}, \quad s \ge 0$$

where  $||u|| = ||u||_{L^2}$  is the  $L^2(\Omega)$ -norm. We will be interested in characterizing the space  $X^s(\Omega)$  depending on the domain  $\Omega$  at study.

# 5.2.1 The case of periodic boundary conditions

In the case of periodic boundary conditions we have that

$$X^{s}(\mathbb{T}^{d}) = H^{s}(\mathbb{T}^{d}) := \left\{ u = \sum_{k \in \mathbb{Z}^{d}} u_{k} \frac{e^{ikx}}{\sqrt{(2\pi)^{d}}} \in L^{2}(\mathbb{T}^{d}) : |u|_{s}^{2} \triangleq \sum_{k \in \mathbb{Z}^{d}} |k|^{2s} |u_{k}|^{2} < \infty \right\}$$

with equivalence of norms

$$||u||_{s}^{2} = ||u||_{L^{2}(\mathbb{T}^{d})}^{2} + ||(-\Delta)^{s/2}u||_{L^{2}(\mathbb{T}^{d})}^{2} = \sum_{k \in \mathbb{Z}^{d}} (1 + |k|^{2s})|u_{k}|^{2} = ||u||_{H^{s}}^{2},$$

where  $u_k = \frac{1}{\sqrt{(2\pi)^d}} \int_{\mathbb{T}^d} u e^{-ikx} dx.$ 

# 5.2.2 The case of Dirichlet boundary conditions

We will be interested in characterizing the domain  $X^{s}(\Omega)$  for  $s \in [0,2] \cup \mathbb{N}$  (see Theorem 5.1.2).

In the case where  $s = m \in \mathbb{N}$  we have the following characterization (see [Tho07, Lemma 3.1])

$$X^m = \{ u \in H^m(\Omega) : \Delta^j u = 0 \text{ in } L^2(\partial\Omega) \text{ for } j < m/2 \},\$$

with equivalence of the norms on  $H^m(\Omega)$  and  $X^m$  for functions in  $X^m$ .

To treat the case where s is not an integer we first introduce the following fractional Sobolev-type spaces known as the Sobolev-Slobodetskij, Gagliardo or Aronszajn space. Given any s > 0 of the form  $s = m + \sigma$ , with  $m \in \mathbb{N}$  and  $\sigma \in (0, 1)$ , we define

$$H^{s}(\Omega) = \{ u \in H^{m}(\Omega) : D^{\alpha}u \in H^{\sigma}(\Omega) \text{ for any } \alpha \text{ s.t. } |\alpha| = m \},\$$

endowed with the norm

$$||u||_{H^s}^2 = \sum_{|\alpha|=0}^m ||D^{\alpha}u||^2 + \sum_{|\alpha|=m} |D^{\alpha}u|_{H^{\sigma}}^2.$$

For s = m an integer the space  $H^{s}(\Omega)$  coincides with the usual Sobolev space  $H^{m}(\Omega)$ , and for  $\sigma \in (0,1)$  we have

$$H^{\sigma}(\Omega) = \left\{ u \in L^{2}(\Omega) : |u|_{H^{\sigma}}^{2} := \int_{\Omega} \int_{\Omega} \frac{|u(x) - u(y)|^{2}}{|x - y|^{d + 2\sigma}} dx dy < \infty \right\}.$$

We note that all of the fractional Sobolev spaces which we introduce here can also be defined by using interpolation theory. Indeed, the above space is an intermediary Banach space between  $L^2(\Omega)$  and  $H^1(\Omega)$ , and can be defined by interpolation as

$$H^{\sigma}(\Omega) = [L^{2}(\Omega), H^{1}(\Omega)]_{\sigma},$$

see [BSV15, Appendix 1] and [Yag09]. Finally, for  $s \in (1/2, 2]$  we define

$$H_D^s(\Omega) = \{ u \in H^s(\Omega) : u|_{\partial\Omega} = 0 \text{ in } L^2(\partial\Omega) \},\$$

it follows from the above that  $D(\mathcal{L}) = H_D^2(\Omega)$ . We can now express  $X^s$  in terms of Sobolev spaces for  $s \in [0,2] \setminus \frac{1}{2}$  (see [Yag09, Theorem 16.12]),

$$X^{s}(\Omega) = \begin{cases} H^{s}(\Omega) & \text{if } 0 \le s < \frac{1}{2} \\ H^{s}_{D}(\Omega) & \text{if } \frac{1}{2} < s \le 2 \end{cases}$$

with norm equivalence

$$C^{-1}||u||_{H^s} \le ||u||_s \le C||u||_{H^s}, \quad u \in X^s,$$
(5.12)

for some constant C > 0. In the special case where s = 1/2 we have that  $X^{1/2} = H_{00}^{1/2}$  is the intermediate space defined by

$$H_{00}^{1/2}(\Omega) := \{ u \in H^{1/2}(\Omega) : |u|_{H_{00}^{1/2}}^2 := \int_{\Omega} \frac{u^2(x)}{\operatorname{dist}(x, \partial\Omega)} dx < \infty \}.$$

with equivalence of norms on  $X^{1/2}$  as in (5.12), see [APR18, Prop 2.2].

#### Bilinear and nonlinear estimates

In this section we introduce bilinear estimates that are fundamental for the global error analysis, which we now motivate. The results we present in this article go beyond the more typical  $H^s$  error analysis  $(s > \frac{d}{2})$ , by pushing down the analysis to  $L^2$  and obtaining fractional rates of convergence, from first up to second order. In particular, to obtain these fractional rates when  $\gamma < d/4$ , we need to work in the low-order Sobolev spaces  $H^{2\gamma}$  (see Section 5.4). In order to obtain sharp low-regularity error estimates in these spaces we call upon three bilinear estimates (see equations (5.14), (5.15), and (5.16) below) which are taylored to require the least regularity assumptions on u when bounding the local error terms (see also Remark 5.2.1).

Let  $\gamma \ge 0$  and  $\epsilon > 0$ . Throughout the error analysis we will use the following bilinear estimates, depending on the values of  $\gamma$ . In the regime  $\gamma > d/4$  we call upon the classical bilinear estimate

$$||uv||_{H^{2\gamma}} \lesssim ||u||_{H^{2\gamma}} ||v||_{H^{2\gamma}}, \text{ for } \gamma > \frac{d}{4},$$
(5.13)

whereas in the regime  $\gamma \in [0, d/4)$  we exploit the following three bilinear estimates,

$$\|uv\| \lesssim \|u\|_{H^{\frac{d}{4}+\gamma}} \|v\|_{H^{\frac{d}{4}-\gamma}} \text{ for } 0 \le \gamma < \frac{d}{4},$$
(5.14)

$$\|uv\|_{H^{2\gamma}} \lesssim \|u\|_{H^{\frac{d}{4}+\gamma}} \|v\|_{H^{\frac{d}{4}+\gamma}} \text{ for } 0 \le \gamma < \frac{d}{4},$$
(5.15)

$$\|uv\|_{H^{2\gamma}} \lesssim \|u\|_{H^{\frac{d}{2}+\epsilon}} \|v\|_{H^{2\gamma}} \text{ for } 0 \le \gamma \le \frac{d}{4},$$
(5.16)

for any  $\epsilon > 0$ . The above estimates are particular cases of [Hör97, Theorem 8.3.1], valid either on  $\mathbb{R}^d$  or  $\mathbb{T}^d$ . Furthermore, for a smooth bounded domain  $\Omega \subset \mathbb{R}^d$ , Stein's extension theorem ([AF03, p.154]) guarantees the existence of a total extension operator, bounded both from  $L^2(\Omega)$  to  $L^2(\mathbb{R}^d)$  and from  $H^m(\Omega)$  to  $H^m(\mathbb{R}^d)$ , for any  $m \in \mathbb{N}$ . By interpolation, this operator is bounded from  $H^s(\Omega)$  to  $H^s(\mathbb{R}^d)$  for any  $s \leq m$  (see [AF03, p. 208]). The estimates (5.13), (5.14), (5.15) and (5.16) consequently hold on  $\Omega$  by extending u and v to  $\mathbb{R}^d$ , applying the estimates on their extensions, and restricting their product to  $\Omega$ .

Remark 5.2.1 (Bilinear estimates in low-order Sobolev spaces  $H^{2\gamma}$ ,  $\gamma < d/4$ ) A natural bilinear estimate which is essential for an analysis in the spaces  $H^{2\gamma}$ ,  $\gamma < d/4$ , is the estimate (5.16). This estimate is an analogue of the estimate (5.13) in the smooth case where  $\gamma > d/4$ . These two estimates allow to start and fall back on the same space  $H^{2\gamma}$ . However, the estimate (5.16) requires more regularity on u than on v ( $\frac{d}{2} + \epsilon > 2\gamma$ , for  $\gamma < d/4$ ), and asks for  $H^{\frac{d}{2}+\epsilon}$ -regularity on the solution. One can obtain more optimal bounds which require less regularity assumptions by equally distributing the regularity on u and on v. Indeed, by assuming that u and v have the same regularity, applying the estimate (5.15) requires  $\frac{d}{4} + \gamma$  additional derivatives, which is better than  $\frac{d}{2} + \epsilon$  for  $\gamma < d/4$ . While the estimate (5.14) is used when v requires  $2\gamma$  derivatives more than u, and balances the regularity requirement to again ask for  $\frac{d}{4} + \gamma$ on both u and v (see Proposition 5.4.2).

We now consider the nonlinearity, which we denote by

$$f(u,\bar{u})(t,x) = -iu^2(t,x)\bar{u}(t,x).$$
(5.17)

One can easily deduce from the inequalities (5.13) and (5.16) together with the equivalence of norms on  $X^s$  the following estimates on the nonlinearity (5.17)

$$\|f(w,\bar{w})\|_{s} \leq c_{s,\sigma} \|w\|_{\sigma}^{2} \|w\|_{s} \leq C_{s,\sigma}(\|w\|_{\sigma})\|w\|_{s}$$

$$\|f(v,\bar{v}) - f(w,\bar{w})\|_{s} \leq c_{s,\sigma} \|v - w\|_{s} \sum_{k=0}^{2} \|v\|_{\sigma}^{k} \|w\|_{\sigma}^{2-k} \leq C_{s,\sigma}(\|v\|_{\sigma}, \|w\|_{\sigma})\|v - w\|_{s}$$
(5.18)

where  $\sigma = \frac{d}{2} + \epsilon$ ,  $c_{s,\sigma} > 0$ , and  $C_{s,\sigma}(||u||, ||v||)$  denotes a generic constant which depends continuously on the bounded arguments ||u|| and ||v||. In the regime  $s > \frac{d}{2}$  the above holds with  $\sigma = s$ .

Remark 5.2.2 (An analysis for very rough solutions) The main ingredient throughout the error analysis section of this article rests upon the crucial bilinear estimates given above, and restricts the solution to belong to the Sobolev space  $H^s$ , s > d/2. In order to consider very rough solutions  $u \in H^s$ ,  $s \le d/2$  one needs to call upon more refined tools such as discrete Bourgain spaces when working on the torus ([ORS22b]), and discrete Strichartz estimates when working on the full space ([ORS21]). This delicate error-analysis is out of scope for this paper.

130

Lastly, as we are interested in obtaining fractional error estimates we will call upon the following estimate several times throughout the error analysis section. For  $\gamma \in [0, 1]$ , we have,

$$\left\|\frac{(e^{it\Delta}-1)}{(-t\Delta)^{\gamma}}u\right\| \le 2^{1-\gamma}\|u\|.$$
(5.19)

The above estimate easily follows from the usual bound

$$|e^{ix} - 1| \le 2^{1-\gamma} |x|^{\gamma}, \quad \gamma \in [0, 1],$$

and using the discrete spectral decomposition of the operator  $\mathcal{L} = -\Delta$ .

We finish this section by stating the definition of a commutator-type term, which is used in order to obtain low-regularity error estimates (see Section 5.4.1). For  $H(v_1, \dots, v_n)$ ,  $n \ge 1$ , a function and L a linear operator, we define the commutator-type term C[H, L] as

$$\mathcal{C}[H,L](v_1,\cdots,v_n) = -L(H(v_1,\cdots,v_n)) + \sum_{j=1}^n \partial_{v_j} H(v_1,\cdots,v_n) \cdot Lv_j.$$

In our setting,  $H(v_1, v_2) = f(v_1, v_2) = -iv_1^2 v_2$  is the nonlinearity given in (5.17),  $L = i\Delta$ , and hence

$$\partial_{v_1} f(v_1, v_2) = -2iv_1 v_2 \quad \text{and} \quad \partial_{v_2} f(v_1, v_2) = -iv_1^2$$
(5.20)

and

$$\mathcal{C}[f, i\Delta](v_1, v_2) = -\Delta(v_1^2 v_2) + 2v_1 v_2 \Delta v_1 + v_1^2 \Delta v_2 = -2(|\nabla v_1|^2 v_2 + 2v_1 \nabla v_1 \cdot \nabla v_2).$$
(5.21)

# 5.3 The implicit nature of the scheme

In this section we deal with the question of solving the nonlinear equation (5.5) at a given time step. We also provide an a priori bound on the numerical solution  $\varphi^{\tau}(v)$  in terms of v, which is crucial for the convergence analysis. We recall from Figure 5.4 that the implicit nature of the scheme does not adversely affect the computational cost of the method.

In the following, we fix  $v \in X^{\sigma}$  for some  $\sigma > d/2$ . We note that v will play the role of the element  $u^n$  in the scheme (5.5). We then introduce the map

$$z \mapsto S(z) = e^{i\tau\Delta}v + \psi_E^{\tau/2}(v) + \psi_I^{\tau/2}(z),$$

and wish to prove that it admits a unique fixed point given by  $\varphi^{\tau}(v)$ .

We start by introducing some useful estimates on the map S.

**Proposition 5.3.1** Given any  $\sigma > d/2$  we have

$$\|S(z_1) - S(z_2)\|_{\sigma} \le \tau M(\|z_1\|_{\sigma}, \|z_2\|_{\sigma})\|z_1 - z_2\|_{\sigma}, \quad and \quad \|S(e^{i\tau\Delta}z_1) - e^{i\tau\Delta}z_1\|_{\sigma} \le \tau \widetilde{M}(\|z_1\|_{\sigma}),$$

where  $M(||z_1||, ||z_2||)$  and  $\widetilde{M}(||z_1||)$  denote generic constants which depend continuously on their arguments  $||z_1||$  and  $||z_2||$ .

Proof of Proposition 5.3.1. The proof follows directly from the definition of the map S, the scheme (5.5) and of the estimate (5.18).

The following theorem shows that the implicit scheme (5.5) is well-defined, and admits an a priori bound.

**Theorem 5.3.1** Let R > 0 and  $\sigma > d/2$ . There exists  $\tau_R > 0$  such that, for all  $\tau \leq \tau_R$  and  $v \in X^{\sigma}$  with  $||v||_{\sigma} \leq R$ , we have that  $\varphi^{\tau}(v)$  defined in (5.5) is given by

$$\varphi^{\tau}(v) \stackrel{H^{\sigma}}{=} \lim_{j \to +\infty} S^{j}(e^{i\tau\Delta}v).$$
(5.22)

Moreover, under the same conditions, we have

$$\|\varphi^{\tau}(v)\|_{\sigma} \le 2R. \tag{5.23}$$

Proof of Theorem 5.3.1. For notational convenience we let  $x_j = S^j(x_0)$ , with  $x_0 = e^{i\tau\Delta}v$ . We first show by induction that for sufficiently small  $\tau$  we have the bound

$$||x_j||_{\sigma} \le 2R, \quad j \ge 0.$$
 (5.24)

We choose  $\tau_R > 0$  such that  $\tau_R M(2R, 2R) \leq 1/2$  and  $\tau_R \widetilde{M}(R) \leq R/2$ , with M and  $\widetilde{M}$  from Proposition 5.3.1. We assume that  $||x_j||_{\sigma} \leq 2R$ ,  $\forall j \leq J$ . It follows that for  $\tau \leq \tau_R$  we have,

$$||x_{J+1} - x_0||_{\sigma} \leq \sum_{j=1}^{J} ||S(x_j) - S(x_{j-1})||_{\sigma} + ||S(x_0) - x_0||_{\sigma}$$
$$\leq \sum_{j=0}^{J} \left( \prod_{k=1}^{j} \tau M(||x_k||_{\sigma}, ||x_{k-1}||_{\sigma}) \right) ||S(x_0) - x_0||_{\sigma}$$
$$\leq \frac{R}{2} \sum_{j=0}^{J} \frac{1}{2^j}$$
$$\leq R.$$

By recalling that, by assumption,  $||x_0|| = ||v|| \le R$ , we conclude from the above that

$$||x_{J+1}||_{\sigma} \le 2R,$$

and hence by induction bound (5.24) holds.

It then follows that for all  $\tau \leq \tau_R$ ,  $(x_j)_{j \in \mathbb{N}}$  is a Cauchy sequence. Indeed, for m > p we have

$$\|x_m - x_p\|_{\sigma} \le \sum_{j=p}^{m-1} \|S(x_j) - S(x_{j-1})\|_{\sigma} \le \frac{R}{2} \sum_{j=p}^{\infty} \frac{1}{2^j} \underset{p \to \infty}{\longrightarrow} 0$$

This implies that the sequence  $(x_j)_{j \in \mathbb{N}}$  converges in  $X^{\sigma}$  to the unique fixed-point  $\varphi^{\tau}(v)$  of S, and the characterization (5.22) follows. Finally, by passing to the limit in (5.24) we obtain the desired a priori bound (5.23) on  $\varphi^{\tau}(v)$ , which concludes the proof.

# 5.4 Error Analysis

In this section we will prove the following proposition.

**Proposition 5.4.1** Let T > 0, and  $\gamma \in [0,1] \setminus \{\frac{d}{4}\}$ . Then there exists  $\tau_{\min} > 0$  such that for every time step  $\tau \leq \tau_{\min}$  the numerical solution  $u^n$  given in equation (5.5) has the following error bound:

$$\|u(n\tau) - u^n\|_{L^2} \le C_T(\sup_{[0,T]} \|u(t)\|_{H^{\alpha}})\tau^{1+\gamma}, \quad 0 \le n\tau \le T,$$
(5.25)

where  $\alpha$  is given by

$$\begin{cases} \alpha = 2\gamma + 1 & \text{if } \frac{d}{4} < \gamma \le 1\\ \alpha = \gamma + 1 + \frac{d}{4} & \text{if } 0 \le \gamma < \frac{d}{4} \end{cases}$$
(5.26)

and where  $\tau_{\min}$  depends on T and on  $\|u_0\|_{H^{\alpha}}$ , and  $C_T$  is a positive function of its argument, depending on T.

Given a fixed convergence rate this proposition expresses the regularity assumptions needed in order to obtain this rate, while on the other hand given a fixed regularity assumption on the initial data Theorems 5.1.1 and 5.1.2 express the convergence rates one can attain with the method (5.5). We now link these results.

Proof of Theorem 5.1.1 and Theorem 5.1.2. By writing the convergence rate  $\gamma$  in terms of the regularity assumptions needed on the solution, it directly follows from the above proposition that the convergence rate (5.10) holds for  $\alpha$  and  $\gamma$  which satisfy:

$$\begin{cases} 1 + \frac{d}{2} < \alpha \le 3, \quad \gamma = \frac{\alpha - 1}{2} \\ 1 + \frac{d}{4} \le \alpha < 1 + \frac{d}{2}, \quad \gamma = \alpha - 1 - \frac{d}{4}. \end{cases}$$
(5.27)

The proof of Theorem 5.1.1 then follows by using the fact that if (5.10) holds for some  $\tilde{\alpha} \in [1 + \frac{d}{4}, 3], \tilde{\gamma} \in [0, 1]$ , then the error bound also holds for any  $\alpha \geq \tilde{\alpha}$ , and  $\gamma \leq \tilde{\gamma}$ . In particular, we recover the case  $\alpha = 1 + \frac{d}{2}$  in Theorem 5.1.1 by applying the second line in (5.27) with  $(\tilde{\alpha}, \tilde{\gamma}) = (1 + \frac{d}{2} - \epsilon, \frac{d}{4} - \epsilon)$  to obtain convergence for  $0 \leq \gamma < \frac{d}{4}$ . See Figure 5.5 which illustrates graphically the convergence result.



Figure 5.5 – Illustration of the interplay between the regularity parameter  $\alpha$  and the convergence rate parameter  $\gamma$  in the convergence result stated in Proposition (5.4.1) and Theorem 5.1.1. We plot the regularity assumption  $(u(t) \in H^{\alpha})$  needed in order to obtain convergence of order  $\tau^{1+\gamma}$ .

The proof of Theorem 5.1.2 follows in the same manner, with the added constraint of the boundary conditions. Namely, we require  $u \in C([0,T], X^s)$  and  $s = 2\gamma + 1$  or  $\gamma + 1 + d/4$ , where the boundary conditions are imposed in the definition of the space  $X^s$ . See Section 5.2 for the definition of the spaces  $X^s$ .

In order to prove Proposition 5.4.1 we combine local error bounds together with a stability argument to conclude via a Lady Windermere's fan argument. We start by showing the local error bound of order  $\tau^{2+\gamma}$  with the regularity assumptions stated in Proposition 5.4.1.

### 5.4.1 Local error analysis

We decompose the local error term as follows,

$$u(t_{n}+\tau) - \varphi^{\tau}(u(t_{n})) = \int_{0}^{\tau} e^{i(\tau-s)\Delta} f(u(t_{n}+s), \bar{u}(t_{n}+s)) ds - \psi_{E}^{\tau/2}(u(t_{n})) - \psi_{I}^{\tau/2}(\varphi^{\tau}(u(t_{n}))))$$
  
=  $\mathcal{R}_{1}(\tau, t_{n}) + \mathcal{R}_{2}(\tau, t_{n}) + \mathcal{R}_{3}(\tau, t_{n}),$ 

with

$$\mathcal{R}_{1}(\tau, t_{n}) = \left( \int_{0}^{\tau/2} e^{i(\tau-s)\Delta} f\left(e^{is\Delta}u(t_{n}), e^{-is\Delta}\bar{u}(t_{n})\right) ds - \psi_{E}^{\tau/2}\left(u(t_{n})\right) \right) + \left( \int_{\tau/2}^{\tau} e^{i(\tau-s)\Delta} f\left(e^{is\Delta}u(t_{n}), e^{-is\Delta}\bar{u}(t_{n})\right) ds - \psi_{I}^{\tau/2}\left(e^{i\tau\Delta}u(t_{n})\right) \right),$$
(5.28)

$$\mathcal{R}_2(\tau, t_n) = \psi_I^{\tau/2} \left( e^{i\tau\Delta} u(t_n) \right) - \psi_I^{\tau/2} \left( \varphi_\tau \left( u(t_n) \right) \right), \tag{5.29}$$

and

$$\mathcal{R}_{3}(\tau,t_{n}) = \int_{0}^{\tau} e^{i(\tau-s)\Delta} \Big( f\big(u(t_{n}+s),\bar{u}(t_{n}+s)\big) - f\big(e^{is\Delta}u(t_{n}),e^{-is\Delta}\bar{u}(t_{n})\big) \Big) ds.$$

We start by estimating the first error term  $\mathcal{R}_1(\tau, t_n)$ . This term is the one which asks for the most regularity, and hence dictates the regularity assumptions required on the solution, and thereby the initial data. We then proceed by estimating each of the terms  $\mathcal{R}_2(\tau, t_n)$  and  $\mathcal{R}_3(\tau, t_n)$ , to obtain a cancelation in their sum thanks to the symmetry of the scheme, yielding the desired local error estimate. **Proposition 5.4.2** The error term  $\mathcal{R}_1(\tau, t_n)$  satisfies the following bound,

$$\|\mathcal{R}_{1}(\tau, t_{n})\| \leq \begin{cases} C_{T}\left(\sup_{t \in [0,T]} \|u(t)\|_{X^{2\gamma+1}}\right) \tau^{2+\gamma} & \text{if } \gamma > \frac{d}{4} \\ C_{T}\left(\sup_{t \in [0,T]} \|u(t)\|_{X^{\gamma+1+\frac{d}{4}}}\right) \tau^{2+\gamma} & \text{if } 0 \leq \gamma < \frac{d}{4} \end{cases},$$
(5.30)

for  $0 \leq t_n \leq T$ .

Proof of Proposition 5.4.2. We define the filtered function as

$$\mathcal{N}(\tau, s, \zeta, \Delta, v) = -ie^{i(\tau-s)\Delta} [(e^{is\Delta}v)^2 (e^{is\Delta}e^{-2i\zeta\Delta}\bar{v})] = e^{i(\tau-s)\Delta} f(e^{is\Delta}v, e^{i(s-2\zeta)\Delta}\bar{v})$$
(5.31)

which plays a fundamental role in the derivation and analysis of our scheme on general domains. In the above expression we duplicate the time variable into s and  $\zeta$ , pulling out a factor  $e^{is\Delta}$  in front of the conjugate term  $e^{i(s-2\zeta)\Delta}\bar{v}$ . Taylor expanding in the variable s yields the right cancellation with the factor  $e^{i(\tau-s)\Delta}$  to recover, after integrating in the variable  $\zeta$ , the explicit term  $\psi_E^{\tau/2}(v)$  in the scheme (5.5), as is detailed below. Similar filtering techniques are used in [RS21, AB23a, ABBS22b].

Let  $v = u(t_n)$ . By Taylor expanding the filtering function (5.31) around s = 0, we obtain that the first term in (5.28) satisfies

$$\int_{0}^{\tau/2} e^{i(\tau-\zeta)\Delta} f(e^{i\zeta\Delta}v, e^{-i\zeta\Delta}\bar{v}) d\zeta = \int_{0}^{\tau/2} \mathcal{N}(\tau, \zeta, \zeta, \Delta, v) d\zeta$$

$$= \int_{0}^{\tau/2} \mathcal{N}(\tau, 0, \zeta, \Delta, v) d\zeta + \int_{0}^{\tau/2} \int_{0}^{\zeta} \partial_{s} \mathcal{N}(\tau, s, \zeta, \Delta, v) ds d\zeta$$

$$= \psi_{E}^{\tau/2}(v) + \int_{0}^{\tau/2} \int_{0}^{\zeta} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta](e^{is\Delta}v, e^{i(s-2\zeta)\Delta}\bar{v}) ds d\zeta,$$
(5.32)

where to obtain the last line we used the definition of the  $\varphi_1$  function (see (5.4)) to obtain that

$$\int_0^{\tau/2} e^{-2i\zeta\Delta} d\zeta = \frac{\tau}{2} \frac{e^{-i\tau\Delta} - 1}{-i\tau\Delta} = \frac{\tau}{2} \varphi_1(-i\tau\Delta)$$

and the definition of commutator term  $C[f, i\Delta](u, v)$  given in (5.21).

Similarly, using the filtering function (5.31), we can treat the second line in (5.28) by Taylor expanding around  $s = \tau$ . This yields

$$\int_{\tau/2}^{\tau} e^{i(\tau-\zeta)\Delta} f(e^{i\zeta\Delta}v, e^{-i\zeta\Delta}\bar{v}) d\zeta = \int_{\tau/2}^{\tau} \mathcal{N}(\tau, \zeta, \zeta, \Delta, v) d\zeta$$
$$= \int_{\tau/2}^{\tau} \mathcal{N}(\tau, \tau, \zeta, \Delta, v) d\zeta - \int_{\tau/2}^{\tau} \int_{\zeta}^{\tau} \partial_s \mathcal{N}(\tau, s, \zeta, \Delta, v) ds d\zeta$$
$$= \psi_I^{\tau/2} (e^{i\tau\Delta}v) - \int_{\tau/2}^{\tau} \int_{\zeta}^{\tau} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta] (e^{is\Delta}v, e^{i(s-2\zeta)\Delta}\bar{v}) ds d\zeta,$$
(5.33)

where to go from the second line in the above to the third we used that

$$\begin{split} \int_{\tau/2}^{\tau} \mathcal{N}(\tau,\tau,\zeta,\Delta,v) d\zeta &= -i(e^{i\tau\Delta}v)^2 \left( \left( \int_{\tau/2}^{\tau} e^{-2i\zeta\Delta} d\zeta \right) e^{i\tau\Delta} \bar{v} \right) \\ &= -i(e^{i\tau\Delta}v)^2 \left( \left( \int_{0}^{\tau/2} e^{-2i(\tau-\zeta)\Delta} d\zeta \right) e^{i\tau\Delta} \bar{v} \right) \\ &= -i\frac{\tau}{2} (e^{i\tau\Delta}v)^2 \varphi_1(i\tau\Delta) (e^{-i\tau\Delta} \bar{v}), \\ &= \psi_I^{\tau/2} (e^{i\tau\Delta}v). \end{split}$$

By definition of  $\mathcal{R}_1(\tau, t_n)$  and by using (5.32) and (5.33) we have that

$$\mathcal{R}_{1}(\tau, t_{n}) = \int_{0}^{\tau/2} \int_{0}^{\zeta} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta](e^{is\Delta}v, e^{is\Delta}e^{-2i\zeta\Delta}\bar{v}) dsd\zeta$$
  
$$-\int_{\tau/2}^{\tau} \int_{\zeta}^{\tau} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta](e^{is\Delta}v, e^{is\Delta}e^{-2i\zeta\Delta}\bar{v}) dsd\zeta$$
  
$$= \int_{0}^{\tau/2} \int_{0}^{\zeta} e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta](e^{is\Delta}v, e^{is\Delta}e^{-2i\zeta\Delta}\bar{v})$$
  
$$- e^{is\Delta} \mathcal{C}[f, i\Delta](e^{i(\tau-s)\Delta}v, e^{i(\tau-s)\Delta}e^{-2i(\tau-\zeta)\Delta}\bar{v}) dsd\zeta.$$
(5.34)

Note that the local error structure lead by the commutator-type terms in the above expression requires less regularity assumptions than what is required by classical methods, such as exponential integrators of splitting methods (see [HO10, Lub08]). Indeed, from the explicit form (5.21) of the commutator term we see that this error term requires only one additional derivative on the initial datum rather than two (see also [AB23a, ABBS22b]).

We now show that thanks to the symmetry of the scheme, we obtain a cancelation in the second-order error term (5.34) yielding (up to) a third-order remainder. For notational convenience we let  $w_1(s) = e^{is\Delta}v$ ,  $w_2(s,\zeta) = e^{i(s-2\zeta)\Delta}\bar{v}$ ,  $z_1(s) = e^{i(\tau-s)\Delta}v$ , and  $z_2(s,\zeta) = e^{i(2\zeta-\tau-s)\Delta}\bar{v}$  and we denote the integrand by

$$R_r(\tau, s, \zeta, v) = e^{i(\tau-s)\Delta} \mathcal{C}[f, i\Delta](w_1(s), w_2(s, \zeta)) - e^{is\Delta} \mathcal{C}[f, i\Delta](z_1(s), z_2(s, \zeta)).$$
(5.35)

It follows from the above equations (5.34) and (5.35) that

$$\|\mathcal{R}_1(\tau, t_n)\| \le \frac{\tau^2}{4} \sup_{s,\zeta \in [0, \tau/2]} \|R_r(\tau, s, \zeta, v)\|.$$

It remains to show that

$$\sup_{s,\zeta \in [0,\tau/2]} \|R_r(\tau, s, \zeta, v)\| \le \begin{cases} C(\|v\|_{2\gamma+1})\tau^{\gamma} \text{ if } \gamma > d/4\\ C(\|v\|_{\gamma+1+\frac{d}{4}})\tau^{\gamma} \text{ if } \gamma < d/4 \end{cases}$$
(5.36)

We first approximate the exponentials appearing in front of both commutator terms in (5.35) to obtain the following first approximation result on  $R_r$ .

Lemma 5.4.3 We have

$$R_r(\tau, s, \zeta, v) = R_r^1(\tau, s, \zeta, v) + R_r^2(\tau, s, \zeta, v)$$

and

with

$$R_{r}^{1}(\tau, s, \zeta, v) = (e^{i(\tau-s)\Delta} - 1)\mathcal{C}[f, i\Delta](w_{1}(s), w_{2}(s, \zeta)) - (e^{is\Delta} - 1)\mathcal{C}[f, i\Delta](z_{1}(s), z_{2}(s, \zeta))$$

$$R_r^2(\tau, s, \zeta, v) = \mathcal{C}[f, i\Delta](w_1(s), w_2(s, \zeta)) - \mathcal{C}[f, i\Delta](z_1(s), z_2(s, \zeta)),$$

which satisfy

$$\sup_{s,\zeta \in [0,\tau/2]} \|R_r^i(\tau, s, \zeta, \tau, v)\| \le \begin{cases} C(\|v\|_{2\gamma+1})\tau^{\gamma} & \text{if } \gamma > d/4, \\ C(\|v\|_{\gamma+1+\frac{d}{4}})\tau^{\gamma} & \text{if } \gamma < d/4, \end{cases}$$
(5.37)

for i = 1, 2.

Proof of Lemma 5.4.3. We write the error term  $R_r^1$  as

$$R_{r}^{1}(s,\zeta,\tau,v,\bar{v}) = (\tau-s)^{\gamma} \left(\frac{e^{i(\tau-s)\Delta}-1}{(\tau-s)^{\gamma}(-\Delta)^{\gamma}}\right) (-\Delta)^{\gamma} \mathcal{C}[f,i\Delta](w_{1}(s),w_{2}(s,\zeta)) -s^{\gamma} \left(\frac{e^{is\Delta}-1}{s^{\gamma}(-\Delta)^{\gamma}}\right) (-\Delta)^{\gamma} \mathcal{C}[f,i\Delta](z_{1}(s),z_{2}(s,\zeta)).$$
(5.38)

Using the bound given in equation (5.19), and the boundedness of  $e^{it\Delta}$  on Sobolev spaces, it follows from equation (5.38) that we are left to provide a bound on  $C[f, i\Delta](v, \bar{v})$  of the form

$$\|\mathcal{C}[f, i\Delta](v, \bar{v})\|_{2\gamma} \le \begin{cases} C(\|v\|_{2\gamma+1}) \text{ if } \gamma > d/4\\ C(\|v\|_{\gamma+1+\frac{d}{4}}) \text{ if } \gamma < d/4 \end{cases}$$
(5.39)

From the definition of the commutator (5.21), and by using the equivalence of norms  $\|\cdot\|_s$  and  $\|\cdot\|_{H^s}$  on  $X^s$  (see

Section 5.2), it follows that for  $d/4 < \gamma \leq 1$ 

$$\begin{aligned} \||\nabla v|^2 \bar{v}\|_{2\gamma} &\lesssim \||\nabla v|^2 \bar{v}\|_{H^{2\gamma}} \\ &\lesssim \|v\|_{H^{2\gamma}} \|\nabla v \cdot \nabla v\|_{H^{2\gamma}} \\ &\lesssim \|v\|_{H^{2\gamma}} \|v\|_{H^{2\gamma+1}}^2 \\ &\lesssim C(\|v\|_{2\gamma+1}), \end{aligned}$$

where we used the estimate (5.13). Similarly, in the case  $0 \le \gamma < d/4$  we have

$$\begin{aligned} \||\nabla v|^2 \bar{v}\|_{2\gamma} &\lesssim \|v\|_{H^{d/2+\epsilon}} \|\nabla v \cdot \nabla v\|_{H^{2\gamma}} \\ &\lesssim \|v\|_{H^{d/2+\epsilon}} \|v\|_{H^{\gamma+1+\frac{d}{4}}}^2 \\ &\lesssim C(\|v\|_{\gamma+1+\frac{d}{4}}), \end{aligned}$$

where to obtain the first line we used estimate (5.16), to go from the first to the second line we used the estimate (5.15) and concluded using the fact that  $d/2 + \epsilon < \gamma + 1 + \frac{d}{4}$  together with the equivalence of norms on  $X^{\gamma+1+\frac{d}{4}}$ .

We can bound the second term in the commutator-type term (5.21) in the same manner to obtain the desired bound (5.39). The estimate (5.37) on  $R_r^1$  follows immediately.

We now deal with the approximation of the second error term  $R_r^2$  by approximating each of the exponentials appearing in the arguments of the commutator terms, namely on  $w_1, w_2, z_1$  and  $z_2$ . Given the form of the commutator (5.21) and of  $w_1, w_2, z_1, z_2$ , each term to approximate will either be of the form

$$\left\| w\nabla((e^{i\xi\Delta} - 1)u) \cdot \nabla z \right\| \tag{5.40}$$

or

$$\left( \left( e^{i\xi\Delta} - 1 \right) w \right) \nabla u \cdot \nabla z \right\|$$
(5.41)

for  $\xi \in [0, \tau]$ , and where we use the boundedness of  $e^{it\Delta}$  on Sobolev spaces  $(t \in \mathbb{R})$ . We can approximate (5.40) as follows, given any  $\epsilon > 0$ ,

$$\begin{split} \left\| w \nabla ((e^{i\xi\Delta} - 1)u) \cdot \nabla z \right\| &\lesssim \xi^{\gamma} \begin{cases} \|w\|_{H^{2\gamma}} \left\| \nabla (-\Delta)^{\gamma} \left( \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} u \right) \right\| \|\nabla z\|_{H^{2\gamma}} \text{ if } \gamma > d/4 \\ \|w\|_{H^{\frac{d}{2} + \epsilon}} \left\| \nabla (-\Delta)^{\gamma} \left( \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} u \right) \right\|_{H^{\frac{d}{4} - \gamma}} \|\nabla z\|_{H^{\frac{d}{4} + \gamma}} \text{ if } \gamma < d/4 \\ &\leq \xi^{\gamma} \begin{cases} C(\|w\|_{2\gamma}, \|u\|_{2\gamma + 1}, \|z\|_{2\gamma + 1}) \text{ if } \gamma > d/4 \\ C(\|w\|_{\frac{d}{2} + \epsilon}, \|u\|_{\gamma + 1 + \frac{d}{4}}, \|z\|_{\gamma + 1 + \frac{d}{4}}) \text{ if } \gamma < d/4 \end{cases}, \end{split}$$

where we used the Sobolev embedding  $H^{\sigma} \hookrightarrow L^{\infty}$ , for  $\sigma > \frac{d}{2}$ , and the estimate (5.14) to obtain the first inequality. To obtain the second line in the above we used the equivalence of norms, thanks to the fact that u, v, w belong to  $X^{2\gamma+1}$  or  $X^{\gamma+1+d/4}$  respectively, as well as the estimate (5.19). Hence, given that  $\frac{d}{2} + \epsilon < \gamma + 1 + \frac{d}{4}$ , the term above satisfies the desired bound of the form (5.37).

Furthermore, for the expression (5.41) we have,

$$\begin{split} \left\| \left( (e^{i\xi\Delta} - 1)w \right) \nabla u \cdot \nabla z \right\| &\lesssim \xi^{\gamma} \begin{cases} \left\| (-\Delta)^{\gamma} \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} w \right\| \| \nabla u \cdot \nabla z \|_{H^{2\gamma}} \text{ if } \gamma > d/4 \\ & \left\| (-\Delta)^{\gamma} \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} w \right\|_{H^{1+\frac{d}{4}-\gamma}} \| \nabla u \cdot \nabla z \|_{H^{2\gamma}} \text{ if } \gamma < d/4 \\ & \lesssim \xi^{\gamma} \begin{cases} \left\| (-\Delta)^{\gamma} \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} w \right\| \| \nabla u \|_{H^{2\gamma}} \| \nabla z \|_{H^{2\gamma}} \text{ if } \gamma > d/4 \\ & \left\| (-\Delta)^{\gamma} \frac{(e^{i\xi\Delta} - 1)}{(-\xi\Delta)^{\gamma}} w \right\|_{H^{1+\frac{d}{4}-\gamma}} \| \nabla u \|_{H^{\frac{d}{4}+\gamma}} \| \nabla z \|_{H^{\frac{d}{4}+\gamma}} \text{ if } \gamma < d/4 \\ & \leq \xi^{\gamma} \begin{cases} C(\|w\|_{2\gamma}, \|u\|_{2\gamma+1}, \|z\|_{2\gamma+1}) \text{ if } \gamma > d/4 \\ C(\|w\|_{\gamma+1+\frac{d}{4}}, \|u\|_{\gamma+1+\frac{d}{4}}, \|z\|_{\gamma+1+\frac{d}{4}}) \text{ if } \gamma < d/4 \end{cases} \end{split}$$

where to obtain the first line we used the Sobolev embedding  $H^{\sigma} \hookrightarrow L^{\infty}$ ,  $\sigma > \frac{d}{2}$ , and the estimate  $||uz|| \leq ||u||_{H^{1+\frac{d}{4}-\gamma}} ||z||_{H^{2\gamma}}$ ,  $\gamma \in [0, 1]$ , (see [Hör97, Theorem 8.3.1]). In order to obtain the second line in the above we again used the estimates (5.13) and (5.15), and to obtain the third line we used the equivalence of norms on the spaces  $X^s$  together with the

estimate (5.19).

By approximating each of the exponentials in the commutator terms defining  $R_r^2$  and by collecting the error terms which are either of the form (5.40) or (5.41) we recuperate the desired  $\tau^{\gamma}$  bound in (5.37).

We can conclude from Lemma 5.4.3 that we have the desired bound (5.36), and hence the bound (5.30) on  $\mathcal{R}_1(\tau, t_n)$ .

We now continue with the bound on the two remaining terms  $(\mathcal{R}_2 + \mathcal{R}_3)(\tau, t_n)$ , which asks for less regularity assumptions than the boundedness of  $\mathcal{R}_1(\tau, t_n)$ , as shown below.

**Proposition 5.4.4** For  $\gamma \in [0, 1]$ , we have the following fractional bound,

$$\|(\mathcal{R}_2 + \mathcal{R}_3)(\tau, t_n)\| \le C \left( \sup_{[0,T]} \|u(t)\|_{2\gamma}, \sup_{[0,T]} \|u(t)\|_{\sigma} \right) \tau^{2+\gamma},$$
(5.42)

given any  $\sigma > \frac{d}{2}$ . In particular we have the bound,

$$\|(\mathcal{R}_{2} + \mathcal{R}_{3})(\tau, t_{n})\| \leq \begin{cases} C_{T,\gamma} \Big( \sup_{t \in [0,T]} \|u(t)\|_{2\gamma+1} \Big) \tau^{2+\gamma}, & \text{if } \gamma > \frac{d}{4} \\ C_{T,\gamma} \Big( \sup_{t \in [0,T]} \|u(t)\|_{\gamma+1+\frac{d}{4}} \Big) \tau^{2+\gamma}, & \text{if } 0 \leq \gamma < \frac{d}{4} \end{cases},$$
(5.43)

for  $0 \leq t_n \leq T$ .

Proof of Proposition 5.4.4. First, we rewrite the error term  $\mathcal{R}_3(\tau, t_n)$  by making suitable Taylor expansions on f. We start by expanding  $u(t_n + \zeta)$  locally up to second order :

$$u(t_n + \zeta) = e^{i\zeta\Delta}u(t_n) + \zeta f^n + \tilde{R}(\zeta, t_n)$$
(5.44)

where  $f^n = f(u(t_n), \bar{u}(t_n))$  and

$$\tilde{R}(\zeta, t_n) = \int_0^{\zeta} e^{i(\zeta - s)\Delta} f(u(t_n + s), \bar{u}(t_n + s)) ds - \zeta f^n.$$
(5.45)

Using the above expansion for u we rewrite the error term as

$$\mathcal{R}_{3}(\tau,t_{n}) = \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} \left( f\left(e^{i\zeta\Delta}u(t_{n}) + \zeta f^{n} + \tilde{R}(\zeta,t_{n}), \ e^{-i\zeta\Delta}\bar{u}(t_{n}) + \zeta \overline{f^{n}} + \overline{\tilde{R}}(\zeta,t_{n})\right) - f\left(e^{i\zeta\Delta}u(t_{n}), e^{-i\zeta\Delta}\bar{u}(t_{n})\right) \right) d\zeta.$$
(5.46)

For notational convenience we let  $a_1 := e^{i\zeta\Delta}u(t_n) + \zeta f^n$ . By Taylor expanding f around  $(a_1, \bar{a}_1)$  and  $(e^{i\zeta\Delta}u(t_n), e^{-i\zeta\Delta}\bar{u}(t_n))$  respectively we obtain,

$$f\left(a_{1}+\tilde{R}(\zeta,t_{n}),\bar{a}_{1}+\overline{\tilde{R}(\zeta,t_{n})}\right) = f(a_{1},\bar{a}_{1}) + E_{1}(\zeta)$$

$$f(a_{1},\bar{a}_{1}) = f\left(e^{i\zeta\Delta}u(t_{n}),e^{-i\zeta\Delta}\bar{u}(t_{n})\right) + E_{2}(\zeta)$$
(5.47)

where

$$E_{1}(\zeta) = \int_{0}^{1} \partial_{v_{1}} f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}(\zeta, t_{n})}\right) \cdot \tilde{R}(\zeta, t_{n}) + \partial_{v_{2}} f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}(\zeta, t_{n})}\right) \cdot \overline{\tilde{R}(\zeta, t_{n})} d\theta$$
(5.48)

$$E_{2}(\zeta) = \zeta \int_{0}^{1} \left[ \partial_{v_{1}} f\left(e^{i\zeta\Delta} u(t_{n}) + \theta\zeta f^{n}, e^{-i\zeta\Delta} \bar{u}(t_{n}) + \theta\zeta \overline{f^{n}}\right) \cdot f^{n} + \partial_{v_{2}} f\left(e^{i\zeta\Delta} u(t_{n}) + \theta\zeta f^{n}, e^{-i\zeta\Delta} \bar{u}(t_{n}) + \theta\zeta \overline{f^{n}}\right) \cdot \overline{f^{n}} \right] d\theta,$$
(5.49)

and  $\partial_{v_1} f$  together with  $\partial_{v_2} f$  are given in (5.20). Hence, plugging the above into equation (5.46) yields,

$$\mathcal{R}_{3}(\tau, t_{n}) = \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} E_{1}(\zeta) d\zeta + \int_{0}^{\tau} e^{i(\tau-\zeta)\Delta} E_{2}(\zeta) d\zeta$$

$$= \mathcal{E}_{1}(\tau, t_{n}) + \mathcal{E}_{2}(\tau, t_{n}).$$
(5.50)

We first deal with the term in the decomposition above which is of highest order and hence is the simplest to bound, namely the third order term  $\mathcal{E}_1(\tau, t_n)$ . In view of obtaining the bound (5.42) on  $(\mathcal{R}_2 + \mathcal{R}_3)(\tau, t_n)$ , we first show that  $\mathcal{E}_1(\tau, t_n)$  satisfies this bound.

**Lemma 5.4.5** We have the following fractional bound on  $\mathcal{E}_1(\tau, t_n)$ ,

$$\|\mathcal{E}_1(\tau, t_n)\| \le C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\tau^{2+\gamma},$$

for any  $\gamma \in [0,1]$ , and  $\sigma > d/2$ .

*Proof.* It follows from (5.48) that in order to obtain the above bound we need to show that

$$||E_1(\zeta)|| \le C(\sup_{[0,T]} ||u(t)||_{\sigma}, \sup_{[0,T]} ||u(t)||_{2\gamma}) \zeta^{1+\gamma}.$$

By equation (5.48) and by using the Sobolev embedding  $H^{\sigma} \hookrightarrow L^{\infty}$  we have that for all  $\sigma > \frac{d}{2}$ ,

$$\begin{aligned} \|E_{1}(\zeta)\| &\leq \sup_{\theta \in ]0,1[} \left( \|\partial_{v_{1}} f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}}(\zeta, t_{n})\right) \|_{\sigma} \right. \\ &+ \|\partial_{v_{2}} f\left(a_{1} + \theta \tilde{R}(\zeta, t_{n}), \bar{a}_{1} + \theta \overline{\tilde{R}}(\zeta, t_{n})\right) \|_{\sigma} \right) \|\tilde{R}(\zeta, t_{n})\| \\ &\leq C \left( \sup_{t \in [0,T]} \|u(t)\|_{\sigma}, \sup_{(\zeta,t) \in [0,\tau] \times [0,T]} \|\tilde{R}(\zeta,t)\|_{\sigma} \right) \|\tilde{R}(\zeta, t_{n})\|, \end{aligned}$$
(5.51)

where the last inequality follows by using the explicit form of the derivatives (5.20), the bilinear inequality (5.13), the first estimate of equation (5.18), and the fact that  $e^{i\zeta\Delta}$  is an isometry on Sobolev spaces. Next, we show that

$$\sup_{(\zeta,t)\in[0,\tau]\times[0,T]} \|\tilde{R}(\zeta,t)\|_{\sigma} < \infty, \quad \text{and} \quad \|\tilde{R}(\zeta,t_n)\| \le C_T(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\zeta^{1+\gamma}.$$
(5.52)

We obtain the first bound by using the first estimate of equation (5.18) on f with  $r = \sigma$ ,

$$\sup_{(\zeta,t)\in[0,\tau]\times[0,T]} \|\tilde{R}(\zeta,t)\|_{\sigma} \le \tau C(\sup_{t\in[0,T]} \|u(t)\|_{\sigma}) < +\infty.$$

Next, we obtain the second fractional estimate of equation (5.52) by making the following decomposition,

$$\tilde{R}(\zeta, t_n) = \tilde{R}_1(\zeta, t_n) + \tilde{R}_2(\zeta, t_n), \tag{5.53}$$

with

$$\tilde{R}_1(\zeta, t_n) = \int_0^{\zeta} (\zeta - s)^{\gamma} \frac{(e^{i(\zeta - s)\Delta} - 1)}{(-(\zeta - s)\Delta)^{\gamma}} (-\Delta)^{\gamma} f(u(t_n + s), \bar{u}(t_n + s)) ds$$

$$\tilde{R}_2(\zeta, t_n) = \int_0^{\zeta} f(u(t_n + s), \overline{u}(t_n + s)) ds - \zeta f^n.$$

Using the fractional bound (5.19) and the nonlinear estimate (5.18) we have that  $\hat{R}_1(\zeta, t_n)$  is bounded by

$$\|\tilde{R}_1(\zeta, t_n)\| \le \zeta^{1+\gamma} C(\sup_{t \in [0,T]} \|u(t)\|_{\sigma}, \sup_{t \in [0,T]} \|u(t)\|_{2\gamma}).$$

Next, by iterating Duhamel's formula in the first term of  $\tilde{R}_2(\zeta, t_n)$  we obtain the following expansion for  $\tilde{R}_2(\zeta, t_n)$ ,

$$\tilde{R}_{2}(\zeta, t_{n}) = \tilde{R}_{2,1}(\zeta, t_{n}) + \tilde{R}_{2,2}(\zeta, t_{n}),$$
(5.54)

where

$$\tilde{R}_{2,1}(\zeta,t_n) = \int_0^{\zeta} f(e^{is\Delta}u(t_n), e^{-is\Delta}\bar{u}(t_n))ds - \zeta f^n,$$

$$\tilde{R}_{2,2}(\zeta,t_n) = \int_0^{\zeta} \int_0^1 \partial_{v_1} f\left(e^{is\Delta}u(t_n) + \theta \tilde{R}_{2,2}^r(s), e^{-is\Delta}\bar{u}(t_n) + \theta \overline{\tilde{R}_{2,2}^r(\zeta)}\right) \cdot \tilde{R}_{2,2}^r(s) \\ + \partial_{v_2} f\left(e^{is\Delta}u(t_n) + \theta \tilde{R}_{2,2}^r(s), e^{-is\Delta}\bar{u}(t_n) + \theta \overline{\tilde{R}_{2,2}^r(s)}\right) \cdot \overline{\tilde{R}_{2,2}^r(s)} d\theta ds,$$

and  $\tilde{R}_{2,2}^r(s) = \int_0^s e^{i(s-s_1)\Delta} f(u(t_n+s_1), \bar{u}(t_n+s_1)) ds_1.$ 

Using the nonlinear estimate (5.18), one easily obtains the bound

$$\|\tilde{R}_{2,2}(\zeta, t_n)\| \le C(\sup_{[0,T]} \|u(t)\|_{\sigma})\zeta^2,$$

and hence in particular the  $\zeta^{1+\gamma}$  bound for  $\gamma \in [0, 1]$ .

In order to deal with the first term in the decomposition (5.54), we Taylor expand the exponentials appearing in  $\tilde{R}_{2,1}(\zeta, t_n)$  which yields,

$$\begin{split} \tilde{R}_{2,1}(\zeta,t_n) &= \int_0^{\zeta} \int_0^1 \partial_{v_1} f\Big( u(t_n) + \theta(e^{is\Delta} - 1)u(t_n), \bar{u}(t_n) + \theta(e^{-is\Delta} - 1)\bar{u}(t_n) \Big) \cdot \tilde{R}_{2,1}^r(s) \\ &+ \partial_{v_2} f\Big( u(t_n) + \theta(e^{is\Delta} - 1)u(t_n), \bar{u}(t_n) + \theta(e^{-is\Delta} - 1)\bar{u}(t_n) \Big) \cdot \overline{\tilde{R}_{2,1}^r(s)} d\theta ds, \end{split}$$

with  $\tilde{R}_{2,1}^r(s) = s^{\gamma} \frac{(e^{is\Delta}-1)}{(-s\Delta)^{\gamma}} (-\Delta)^{\gamma} u(t_n)$ . Using the fractional estimate (5.19) we obtain the bound

$$\|\tilde{R}_{2,1}^r(s)\| \le C_T(\sup_{[0,T]} \|u(t)\|_{2\gamma})s^{\gamma}.$$

Therefore, by using the usual bilinear inequality (5.13) we achieve the desired bound on  $\tilde{R}_{2,1}$ ;

$$\|\tilde{R}_{2,1}(\zeta)\| \le C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\zeta^{1+\gamma},$$

which concludes the proof of Lemma 5.4.5.

-	_

Now that we have dealt with the third order term  $\mathcal{E}_1(t_n, \tau)$  in the decomposition (5.50) of  $\mathcal{R}_3(t_n, \tau)$ , we are left to consider the term  $\mathcal{E}_2(t_n, \tau)$ , together with the term  $\mathcal{R}_2(t_n, \tau)$  defined at equation (5.29). First, we rewrite  $\mathcal{E}_2(t_n, \tau)$  as a second order term with a third order remainder. The goal being that this second order term cancels with the second order part of the term  $\mathcal{R}_2(t_n, \tau)$ , thereby only leaving third order remainders.

By using the definition of  $\partial_{v_1} f$  and  $\partial_{v_2} f$  given in (5.20) we have that

$$E_2(\zeta) = -i\zeta \int_0^1 [2(e^{i\zeta\Delta}u(t_n) + \theta\zeta f^n)(e^{-i\zeta\Delta}\bar{u}(t_n) + \theta\zeta \overline{f^n})f^n + (e^{i\zeta\Delta}u(t_n) + \theta\zeta f^n)^2\overline{f^n}]d\theta.$$

We can separate the first order terms in the above with the higher order ones to obtain the following decomposition for  $E_2(\zeta)$ ,

$$E_2(\zeta) = \tilde{E}_2(\zeta) + E_2^r(\zeta),$$

with

$$\tilde{E}_2(\zeta) = -i\zeta \left( 2(e^{i\zeta\Delta}u(t_n))(e^{-i\zeta\Delta}\bar{u}(t_n))f^n + (e^{i\zeta\Delta}u(t_n))^2\overline{f^n} \right)$$

and where one can easily show using the nonlinear estimate (5.18) with  $r = \sigma$  that  $E_2^r(\zeta)$  has the following bound:  $||E_2^r(\zeta)|| \leq C_T(\sup_{[0,T]} ||u(t)||_{\sigma})\zeta^2$ .

We let

$$\tilde{\mathcal{E}}_2(\tau, t_n) = \int_0^\tau e^{i(\tau-\zeta)\Delta} \tilde{E}_2(\zeta) d\zeta \quad \text{and} \quad \tilde{\mathcal{E}}_2^r(\tau, t_n) = \int_0^\tau e^{i(\tau-\zeta)\Delta} \tilde{E}_2^r(\zeta) d\zeta,$$

where the error term produced has the bound  $\|\tilde{\mathcal{E}}_{2}^{r}(\tau, t_{n})\| \leq C(\sup_{[0,T]} \|u(t)\|_{\sigma})\tau^{3}$ , which in particular satisfies the  $\tau^{2+\gamma}$  bound for  $\gamma \in [0, 1]$ .

It remains to show that the sum of the remaining terms to bound  $(\mathcal{R}_2 + \tilde{\mathcal{E}}_2)(\tau, t_n)$  also satisfy the  $\tau^{2+\gamma}$  bound given in equation (5.42). In view of this, one last approximation step is made on the term  $\tilde{\mathcal{E}}_2(\tau, t_n)$  before estimating its sum with the term  $\mathcal{R}_2(\tau, t_n)$ . By Taylor expanding around  $\zeta = \tau$  the function  $\zeta \mapsto e^{i(\tau-\zeta)\Delta}(\tilde{\mathcal{E}}_2(\zeta)/\zeta)$  we obtain the following approximation of  $\tilde{\mathcal{E}}_2$ ,

$$\tilde{\mathcal{E}}_2(\tau, t_n) = \tilde{\mathcal{E}}_{2,1}(\tau, t_n) + \tilde{\mathcal{E}}_{2,1}^r(\tau, t_n)$$

with

$$\tilde{\mathcal{E}}_{2,1}(\tau,t_n) = -i\frac{\tau^2}{2} \Big( 2(e^{i\tau\Delta}u(t_n))(e^{-i\tau\Delta}\bar{u}(t_n))f^n + (e^{i\tau\Delta}u(t_n))^2\overline{f^n} \Big),$$
(5.55)

and where  $\tilde{\mathcal{E}}_{2,1}^r(\tau)$  satisfies

$$\|\tilde{\mathcal{E}}_{2,1}^{r}(\tau)\| \leq C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\tau^{2+\gamma}$$

The above estimate follows from the definition of  $\tilde{E}_2$ , the estimate (5.19), and the expansion

$$e^{i\tau\Delta}v + (e^{i\zeta\Delta} - e^{i\tau\Delta})v = e^{i\tau\Delta}v + \left(\zeta^{\gamma}\frac{(e^{i\zeta\Delta} - 1)}{(-\zeta\Delta)^{\gamma}} + \tau^{\gamma}\frac{(1 - e^{i\tau\Delta})}{(-\tau\Delta)^{\gamma}}\right)(-\Delta)^{\gamma}v.$$

The first estimate (5.42) of Proposition 5.4.4 then follows directly once the following lemma is established.

Lemma 5.4.6 The remaining error terms have the following bound,

$$\|\mathcal{R}_{2}(\tau, t_{n}) + \tilde{\mathcal{E}}_{2,1}(\tau, t_{n})\| \leq C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\tau^{2+\gamma},$$
(5.56)

for any  $\sigma > d/2$ .

*Proof.* We perform a very similar analysis as was done on the term  $\mathcal{R}_3(\tau, t_n)$  to the term  $\mathcal{R}_2(\tau, t_n)$  to show that it can be decomposed as a second and third order term. We then conclude by showing that this second order part coincides with the second order term  $-\tilde{\mathcal{E}}_{2,1}(\tau, t_n)$ .

First, we expand  $\varphi^{\tau}(u(t_n))$  as follows,

$$\varphi^{\tau}(u(t_n)) = e^{i\tau\Delta}u(t_n) + \tau f^n + R^1(\tau, t_n),$$
(5.57)

where

$$R^{1}(\tau, t_{n}) = \Psi^{\tau}(u(t_{n}), \varphi^{\tau}(u(t_{n}))) - \tau f^{n},$$

and  $\Psi^{\tau}$  is the nonlinear part of the numerical scheme (5.5). We show the same bounds in  $X^{\sigma}$  and  $L^2$  given in equation (5.52) on the error term  $R^1(\tau, t_n)$ . First, using the estimate (5.18) with  $r = \sigma$  we have that

$$\sup_{t \in [0,T]} \|R^{1}(\tau,t)\|_{\sigma} \leq \tau C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|\varphi^{\tau}(u(t))\|_{\sigma}) \leq C(\sup_{[0,T]} \|u(t)\|_{\sigma}) < \infty,$$

where we use the bound

$$\|\varphi^{\tau}(u(t))\|_{\sigma} \le 2\|u(t)\|_{\sigma},$$
(5.58)

which follows from equation (5.23). We note that in what follows we will always use the above a priori bound (5.58) when bounding the term  $\varphi^{\tau}(u(t_n))$ , and hence will not show its explicit dependance. In order to obtain the  $L^2$ -bound on  $R^1(\tau, t_n)$  we use the following expansion

$$R^{1}(\tau, t_{n}) = -i\frac{\tau}{2}e^{i\tau\Delta} \Big( (u(t_{n}))^{2}\varphi_{1}(-i\tau\Delta)\overline{u(t_{n})} \Big) - i\frac{\tau}{2} \left( \Big( e^{i\tau\Delta}u(t_{n}) + \Psi^{\tau}(u(t_{n}), \varphi^{\tau}(u(t_{n}))) \Big)^{2} \varphi_{1}(i\tau\Delta) \Big( e^{-i\tau\Delta}\overline{u(t_{n})} + \overline{\Psi^{\tau}(u(t_{n}), \varphi^{\tau}(u(t_{n})))} \Big) \Big) - \tau f^{n}$$

$$(5.59)$$

where we used equation (5.5) and simply inserted into the term  $\Psi^{\tau}(u(t_n), \varphi^{\tau}(u(t_n)))$  the definition of the scheme

$$\varphi^{\tau}(u(t_n)) = e^{i\tau\Delta}u(t_n) + \Psi^{\tau}(u(t_n), \varphi^{\tau}(u(t_n))).$$

By expanding about  $e^{i\tau\Delta}u(t_n)$  the second term in (5.59), and then by approximating the remaining exponentials using the usual fractional estimate (5.19), we obtain that the first-order terms cancel leaving the following bound on  $R^1(\tau, t_n)$ ,

$$\begin{aligned} \|R^{1}(\tau,t_{n})\| &\leq \tau^{1+\gamma} C(\sup_{[0,T]} \|u(t)\|_{2\gamma}, \sup_{[0,T]} \|u(t)\|_{\sigma}) + \tau C(\sup_{n\tau \leq T} \|\Psi^{\tau}(u(t_{n}),\varphi^{\tau}(u(t_{n})))\|_{\sigma}) \\ &\leq \tau^{1+\gamma} C(\sup_{[0,T]} \|u(t)\|_{2\gamma}, \sup_{[0,T]} \|u(t)\|_{\sigma}), \end{aligned}$$

where in order to obtain the second line we used the estimate

$$\|\Psi^{\tau}(u(t_n),\varphi^{\tau}(u(t_n)))\| \le C(\sup_{[0,T]} \|u(t)\|_{\sigma})\tau.$$

We make note that in order to approximate the  $\varphi_1$  functions appearing in (5.59) we use the following expansion

$$\tau\varphi_1(i\tau\Delta)v = \int_0^\tau e^{is\Delta}dsv = \tau v + \int_0^\tau \frac{(e^{is\Delta} - 1)}{(-s\Delta)^\gamma} s^\gamma ds (-\Delta)^\gamma v.$$
(5.60)

We conclude from the above calculations that the error term  $R^1(\tau, t_n)$  satisfies the  $X^{\sigma}$  and  $L^2$  bounds,

$$\sup_{t \in [0,T]} \|R^{1}(\tau,t)\|_{\sigma} < \infty \quad \text{and} \quad \|R^{1}(\tau,t_{n})\| \le C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma})\tau^{1+\gamma}.$$
(5.61)

We now return to the definition (5.29) of  $\mathcal{R}_2(\tau, t_n)$  and to the expansion (5.57). By letting  $b_1 = e^{i\tau\Delta}u(t_n) + \tau f^n$  we have

$$\psi_{I}^{\tau/2}(b_{1} + R^{\tau}(\tau, t_{n})) = \psi_{I}^{\tau/2}(b_{1}) + \mathcal{E}_{3}(\tau, t_{n})$$
  
$$\psi_{I}^{\tau/2}(b_{1}) = \psi_{I}^{\tau/2}(e^{i\tau\Delta}u(t_{n})) + \mathcal{E}_{4}(\tau, t_{n}),$$
  
(5.62)

where using the estimates in equation (5.61) and the definition of  $\psi_I^{\tau/2}$  it follows that

$$\|\mathcal{E}_{3}(\tau, t_{n})\| \leq \tau^{2+\gamma} C(\sup_{[0,T]} \|u(t)\|_{\sigma}, \sup_{[0,T]} \|u(t)\|_{2\gamma}).$$

Furthermore, it follows from equation (5.62) that by isolating the second order terms with the higher order ones we have the following expansion for  $\mathcal{E}_4(\tau, t_n)$ ,

$$\mathcal{E}_4(\tau, t_n) = -i\frac{\tau^2}{2} \left( 2f^n (e^{i\tau\Delta} u(t_n))(\varphi_1(i\tau\Delta)(e^{-i\tau\Delta}\overline{u(t_n)})) + (e^{i\tau\Delta} u(t_n))^2 \varphi_1(i\tau\Delta)\overline{f^n} \right) + \mathcal{E}_4^r(\tau, t_n),$$
(5.63)

where from a simple calculation one obtains that  $\mathcal{E}_4^r(\tau, t_n)$  satisfies

$$\|\mathcal{E}_{4}^{r}(\tau, t_{n})\| \leq C(\sup_{[0,T]} \|u(t)\|_{\sigma})\tau^{3} \leq C_{T}\tau^{2+\gamma}$$

By approximating the  $\varphi_1$  functions in (5.63) following the expansion given in equation (5.60), and by using once again the fractional estimate (5.19) we conclude from the above equations together with definition (5.55) of  $\tilde{\mathcal{E}}_{2,1}$  that the bound (5.56) is met. This concludes the proof of Lemma 5.4.6.

The proof of the above lemma concludes the proof of the first estimate (5.42) on  $(\mathcal{R}_2 + \mathcal{R}_3)(\tau, t_n)$  of Proposition 5.4.4. The second estimate (5.43) of Proposition 5.4.4 follows directly from the first estimate (5.42) by noticing that for some small  $\epsilon > 0$  and with  $\sigma = d/2 + \epsilon$ , we have that  $2\gamma$  and  $\sigma$  are smaller than  $2\gamma + 1$  for  $\gamma > d/4$ , and are also smaller than  $\gamma + 1 + d/4$  for  $d \leq 3$  (and  $\gamma \in [0, 1]$ ).

Remark 5.4.7 Another way of writing the local error terms is as follows,

$$u(t_n + \tau) - \varphi^{\tau} \left( u(t_n) \right) = \mathcal{R}(\tau, t_n) + \mathcal{R}(\tau, t_n),$$
(5.64)

with

$$\mathcal{R}(\tau, t_n) = \int_0^{\tau/2} e^{i(\tau-s)\Delta} \Big( f\big(u(t_n+s), \bar{u}(t_n+s)\big) - f\big(e^{is\Delta}u(t_n), e^{-is\Delta}\bar{u}(t_n)\big) \Big) ds \\ + \int_{\tau/2}^{\tau} e^{i(\tau-s)\Delta} \Big( f\big(u(t_n+s), \bar{u}(t_n+s)\big) - f\big(e^{i(s-\tau)\Delta}u(t_{n+1}), e^{-i(s-\tau)\Delta}\bar{u}(t_{n+1})\big) \Big) ds$$

and

$$\begin{aligned} \widetilde{\mathcal{R}}(\tau,t_n) &= \left( \int_0^{\tau/2} e^{i(\tau-s)\Delta} f\left(e^{is\Delta} u(t_n), e^{-is\Delta} \overline{u}(t_n)\right) ds - \psi_E^{\tau/2} \left(u(t_n)\right) \right) \\ &+ \left( \int_{\tau/2}^{\tau} e^{i(\tau-s)\Delta} f\left(e^{i(s-\tau)\Delta} u(t_{n+1}), e^{-i(s-\tau)\Delta} \overline{u}(t_{n+1})\right) ds - \psi_I^{\tau/2} \left(\varphi^{\tau} \left(u(t_n)\right)\right) \right) \end{aligned}$$

The above error decomposition uses the fact that on  $[0, \frac{\tau}{2}]$  we center the approximation at the left-end point and on  $(\frac{\tau}{2}, \tau]$ 

at the right-end point. Hence, on each interval respectively we iterate the Duhamel expansions

$$u(t_n + s) = e^{is\Delta}u(t_n) + \int_0^s e^{i(s-s_1)\Delta}f(u(t_n + s_1), \bar{u}(t_n + s_1))ds_1, \quad s \in [0, \frac{\tau}{2}],$$
$$u(t_n + s) = e^{i(s-\tau)\Delta}u(t_{n+1}) - \int_s^\tau e^{i(s-s_1)\Delta}f(u(t_n + s_1), \bar{u}(t_n + s_1))ds_1, \quad s \in (\frac{\tau}{2}, \tau].$$

Using the tools in Section 5.4.1 one can bound the local error terms (5.64) in an analogous manner.

# 5.4.2 Stability

**Theorem 5.4.1** Let R > 0,  $s \ge 0$ . There exists  $\tau_R > 0$  and  $\sigma > d/2$  such that for any  $\tau \le \tau_R$  and  $w, v \in X^{\sigma}$ , such that  $\|w\|_{\sigma} \le R$  and  $\|v\|_{\sigma} \le R$  we have,

$$\varphi^{\tau}(v) - \varphi^{\tau}(w) \|_{s} \le e^{\tau C_{R}} \|v - w\|_{s},$$

where  $C_R$  denotes a generic constant depending on R (and on s).

Proof of Theorem 5.4.1. Using the second estimate in equation (5.18) we have

$$\|\varphi^{\tau}(v) - \varphi^{\tau}(w)\|_{s} \le (1 + \tau C_{s}(\|v\|_{\sigma}, \|w\|_{\sigma}))\|v - w\|_{s} + \tau C_{s}(\|\varphi^{\tau}(v)\|_{\sigma}, \|\varphi^{\tau}(w)\|_{\sigma})\|\varphi^{\tau}(v) - \varphi^{\tau}(w)\|_{s},$$

with  $\sigma = d/2 + \epsilon$  if  $s \le d/2$  and  $\sigma = s$  if s > d/2. By Theorem 5.3.1 we have that there exists  $\tau_R > 0$  such that for all  $\tau \le \tau_R$  we have the bounds:  $\|\varphi^{\tau}(v)\|_{\sigma} \le 2R$  and  $\|\varphi^{\tau}(w)\|_{\sigma} \le 2R$ . Hence, it follows from the above that,

$$\|\varphi^{\tau}(v) - \varphi^{\tau}(w)\|_{s} \leq \frac{(1 + \tau C_{s}(R, R))}{(1 - \tau C_{s}(2R, 2R))} \|v - w\|_{s} \leq e^{\tau C_{R}} \|v - w\|_{s},$$

for some  $C_R > 0$ .

It remains to combine the stability argument presented in Section 5.4.2 together with the local error bounds of Section 5.4.1 to prove the global convergence result stated in Proposition 5.4.1.

Proof of Proposition 5.4.1. We let  $e^n = u^n - u(t_n)$ , with  $e^0 = 0$ . First, thanks to Proposition 5.4.2 and 5.4.4 we have the local error bound in  $L^2$  required for the global convergence analysis, where the regularity requirements on u have been optimized depending on the fractional order of convergence desired. In order to apply the stability bound stated in Theorem 5.4.1 and to conclude with a Lady Windermere's fan type argument, one needs to show the following uniform bound on the numerical solution,

$$\|u^n\|_{\sigma} \le M_T, \quad \forall n\tau \le T, \tag{5.65}$$

for some  $\sigma > d/2$  and  $M_T > 0$ .

We let  $\sigma = d/2 + \epsilon$ , for some small  $\epsilon > 0$ . To obtain the bound (5.65) we show that there exists  $\delta > 0$ , a constant  $C_{R_n} = C_T(||u^n||_{\sigma})$  depending on  $||u^n||_{\sigma}$  and on  $\sup_{[0,T]} ||u(t)||_{\sigma}$ , and some  $\tau_{R_n} > 0$  also depending on  $||u^n||_{\sigma}$  such that the following global error bound is met,

$$\|e^{n+1}\|_{\sigma} \le \|\varphi^{\tau}(u(t_n)) - u(t_{n+1})\|_{\sigma} + \|\varphi^{\tau}(u^n) - \varphi^{\tau}(u(t_n))\|_{\sigma} \le C_{T,\gamma}\tau^{1+\delta} + e^{\tau C_{R_n}}\|e^n\|_{\sigma}$$
(5.66)

for all  $\tau \leq \tau_{R_n}$ . One can obtain the second term in the above estimate for  $\tau \leq \tau_{R_n}$ , where  $\tau_{R_n}$  depends on  $||u^n||_{\sigma}$  and on  $\sup_{[0,T]} ||u(t)||_{\sigma}$ , by applying Theorem 5.4.1 with  $s = \sigma$ . Hence, it remains to obtain the first term in the above estimate, which corresponds to the local error bound in  $X^{\sigma}$ . Namely, by letting  $\mathcal{R}(\tau, t_n) = u(t_{n+1}) - \varphi^{\tau}(u(t_n))$  we show that there exists  $\delta > 0$  such that

$$\|\mathcal{R}(\tau, t_n)\|_{\sigma} \le C_{T,\gamma} \tau^{1+\delta},\tag{5.67}$$

where  $C_{T,\gamma}$  is a constant depending on T and on the regularity assumptions on u (which in turn depend on  $\gamma$ , the fractional order of convergence required). We establish the above local error estimate (5.67) by using the following interpolation bound,

$$||\mathcal{R}(\tau, t_n)||_{\sigma} \leq \begin{cases} ||\mathcal{R}(\tau, t_n)||^{\theta} ||\mathcal{R}(\tau, t_n)||_{2\gamma+1}^{1-\theta} & \text{if } \gamma > \frac{d}{4} \\ ||\mathcal{R}(\tau, t_n)||^{\widetilde{\theta}} ||\mathcal{R}(\tau, t_n)||_{\gamma+1+\frac{d}{4}}^{1-\widetilde{\theta}} & \text{if } 0 \le \gamma < \frac{d}{4} \end{cases},$$
(5.68)

where  $(\theta, \tilde{\theta}) \in (0, 1)^2$  satisfies  $\sigma = (1 - \theta)(2\gamma + 1) = (1 - \tilde{\theta})(\gamma + 1 + d/4)$ . We have already established the  $L^2$ -bound on

 $\mathcal{R}(\tau, t_n)$  in Section 5.4.1, which is given by,

$$\|\mathcal{R}(\tau, t_n)\| \le \tau^{2+\gamma} \begin{cases} C_{T,\gamma} \left( \sup_{t \in [0,T]} \|u(t)\|_{2\gamma+1} \right) & \text{if } \gamma > \frac{d}{4} \\ C_{T,\gamma} \left( \sup_{t \in [0,T]} \|u(t)\|_{\gamma+1+\frac{d}{4}} \right) & \text{if } 0 \le \gamma < \frac{d}{4} \end{cases}.$$
(5.69)

To obtain the  $X^{2\gamma+1}$  and  $X^{\gamma+1+\frac{d}{4}}$  bound on  $\mathcal{R}(\tau, t_n)$  we simply express the local error using Duhamel's formula and the scheme (5.5),

$$\mathcal{R}(\tau,t_n) = \int_0^\tau e^{i(\tau-s)\Delta} f(u(t_n+s),\overline{u}(t_n+s))ds - i\frac{\tau}{2}e^{i\tau\Delta} ((u(t_n))^2\varphi_1(-i\tau\Delta)\overline{u}(t_n)) - i\frac{\tau}{2} ((\varphi^\tau(u(t_n)))^2\varphi_1(i\tau\Delta)\overline{\varphi^\tau(u(t_n)})).$$

One can bound each of the above terms separately using the first estimate in equation (5.18) with  $s = 2\gamma + 1$  and  $s = \gamma + 1 + \frac{d}{4}$  together with equation (5.23) (with  $R = \sup_{[0,T]} ||u(t)||_s$ ) to obtain that there exists some  $\tilde{\tau}_0 > 0$  depending on  $u_0$  and T such that for all  $\tau \leq \tilde{\tau}_0$ 

$$||\mathcal{R}(\tau, t_n)||_s \le C(\sup_{[0,T]} ||u(t)||_s)\tau \le C_{T,s}\tau.$$
(5.70)

We conclude that the bound (5.67) follows from equation (5.68) (with  $\delta = (1 + \gamma)\theta$  for  $\gamma > d/4$  and  $\delta = (1 + \gamma)\tilde{\theta}$  for  $\gamma < d/4$ ), and where the constant  $C_{T,\gamma}$  is given in equation (5.69) by the  $L^2$  local error bound. We then proceed by induction on (5.66) to obtain that there exists a  $\tau_0 > 0$  which depends on T and  $u_0$  for which the uniform bound (5.65) is true for all  $\tau \leq \tau_0$ .

Finally, by taking s = 0,  $\sigma = d/2 + \epsilon$ , and  $R = \max\{M_T, \sup_{[0,T]} ||u(t)||_{\sigma}\}$  in Theorem 5.4.1 yields the existence of a  $\tau_R$  which depends on T and  $u_0$  such that for all  $\tau \leq \tau_R$ ,

$$||e^{n+1}|| \leq C_{T,\gamma} \tau^{2+\gamma} + e^{\tau C_R} ||e^n||, \quad n\tau \leq T_{\tau}$$

where  $C_{T,\gamma}$  is given in equation (5.69). The global error bound of Proposition 5.4.1 follows by iterating the above estimate and taking  $\tau_{\min} = \min\{\tau_0, \tau_R\}$ , which concludes the proof.

Acknowledgements. This project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 850941). The author would like to express her thanks to David Lee, Georg Maierhofer, Frédéric Rousset, and Katharina Schratz for helpful discussions.

# Chapter 6

# Symmetric resonance based integrators and forest formulae

This chapter is based on the article [ABBMS23], and the numerical simulations presented here are taken from that article.

Abstract. In the present work we introduce a unified framework that allows for the very first systematic construction of symmetric resonance-based integrators to approximate a wide class of nonlinear dispersive equations at low-regularity. The inclusion of symmetries in the construction of resonance-based schemes presents serious challenges and induces a need for a significant extension of prior approaches to allow for sufficient number of degrees of freedom in the resulting schemes while preserving the favorable low-regularity convergence properties of prior constructions. Motivated by recent work [BS22], we achieve this by introducing a novel formalism based on forest formulae that allows us to encode a wider range of possibilities of iterating Duhamel's formula and interpolatory approximations of lower order parts in the construction of these time-stepping methods. The forest formulae allow for a simple characterisation of symmetric schemes and provides a fascinating algebraic structure in its own right which echo those used in Quantum Field Theory for renormalising Feynman diagrams and those used for the renormalisation of singular SPDEs via the theory of Regularity Structures. Our constructions lead to the development of several new symmetric resonance-based integrators that exhibit remarkable structure preservation and convergence properties which are witnessed in numerical experiments.

# 6.1 Introduction

We consider a general class of dispersive differential equations of the form

$$i\partial_t u(t,x) + \mathcal{L}\left(\nabla, \frac{1}{\varepsilon}\right) u(t,x) = |\nabla|^{\alpha} p(u(t,x), \overline{u}(t,x)),$$
  
$$u(0,x) = v(x),$$
  
(6.1)

equipped with periodic boundary conditions  $x \in \mathbb{T}^d$ . Throughout, we assume that p is a polynomial nonlinearity, and that the structure of (6.1) implies at least local well-posedness of the problem on a finite time interval [0, T],  $T < \infty$ , in an appropriate functional space. This class of equations captures a number of physically important models, including the Korteweg de Vries (KdV) equation

$$\partial_t u - i\mathcal{L}(\nabla)u = \frac{1}{2}\partial_x u^2, \quad \mathcal{L}(\nabla) = i\partial_x^3, \quad |\nabla|^\alpha = \partial_x, \quad x \in \mathbb{T},$$
(6.2)

and the nonlinear Schrödinger (NLS) equation,

$$i\partial_t u + \mathcal{L}(\nabla)u = |u|^2 u, \quad \mathcal{L}(\nabla) = \Delta, \quad x \in \mathbb{T}^d.$$
 (6.3)

Like those two examples many physical equations in this class possess conservation laws, or are indeed integrable systems (for example the KdV equation is a completely integrable parity-time invariant system). It is known that symmetric numerical schemes have favourable long-time behaviour when applied to such reversible integrable systems, such as linear (slow) growth in error as a function of the integration time, and near conservation of first integrals over long times [HL04, HLW10, BG94]. At the same time the numerical approximation of the Cauchy problem in low-regularity regimes requires the design of designated methods, amongst which resonance-based schemes have seen significant success over recent years. Firstly developed for specific equations, including the KdV equation [HS17b, WZ22b, LW22], the NLS

equation [OS18, OWY22, AB23b, ORS21, ORS22b, WY22, BLW23], the Gross-Pitaevskii equation [AB23a] and the Navier-Stokes equations [NS15], more recent work has started to establish a more general framework for resonance based low-regularity integrators [ABBS22b, ABBS22a, RS22]. The key idea of these schemes lies in embedding the underlying structure of resonances - triggered by the nonlinear frequency interactions between the leading differential operator  $\mathcal{L}(\nabla, \frac{1}{\varepsilon})$  and the nonlinearity  $p(u(t, x), \overline{u}(t, x))$  - into the numerical discretisation. These nonlinear interactions are in general neglected by classical approximation techniques such as Runge-Kutta methods, splitting methods or exponential integrators. While for smooth solutions these nonlinear interactions are indeed negligible, they do play a central role at low regularity and high oscillations. The accurate resolution of these interactions has been achieved in broad generality only in the recent few years in [BS22, RS21, ABBS22b]. Yet, while the design of such schemes has seen a wide range of developments, prior work has focussed mostly on explicit schemes with desired convergence properties. A few recent results [BMS22, AB23b, FMS23, MS22] have introduced first or second order implicit symmetric integrators at low-regularity fitting the particular structure of the equation with better conservation properties. Nevertheless, a central question remained unanswered: *Can we systematically construct structure preserving resonance based schemes up to arbitrary order which preserve central symmetries of the underlying continuous equation?* 

For ordinary differential equations (ODEs) the theory of *structure preservation* in numerical schemes is thoroughly established [HLW10], specifically there is a extensive amount of literature on the characterisation of symmetric and symplectic Runge–Kutta methods [Kul03, SS88, IZ00] and, more broadly, B-series methods [BS94, CM07]; on the favourable long-time behaviour of such methods when applied for finite-dimensional integrable reversible systems and Hamiltonian systems [HL04, HLW10, BG94] respectively; and even on the limitations on types of structure that can be preserved with B-series methods [IQT07]. Even though the long-time analysis of such methods in the case of PDEs is much less straightforward [Fao12, GL10b], these favourable structure preservation properties have motived the study of symmetric methods for PDEs, for example in the classification of symmetric splitting methods [MQ02] and symmetric exponential integrators [CCO08].

In general, resonance based schemes are *not* structure preserving and do not preserve the symmetries in the system. We can consider for example the second order resonance based scheme introduced by [BS22, Section 5.1.2], referred to henceforth as 'Bruned & Schratz 2022', and given by

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau e^{i\tau\Delta} \Big( (u^n)^2 (\varphi_1(-2i\tau\Delta) - \varphi_2(-2i\tau\Delta))\overline{u^n} \Big) - i\tau \Big( e^{i\tau\Delta}u^n \Big)^2 \varphi_2(-2i\tau\Delta) (e^{i\tau\Delta}\overline{u^n}) - \frac{\tau^2}{2} e^{i\tau\Delta} \Big( |u^n|^4 u^n \Big),$$
(6.4)

where  $\varphi_1(\sigma) = \frac{e^{\sigma} - 1}{\sigma}$  and  $\varphi_2(\sigma) = \frac{e^{\sigma} - \varphi_1(\sigma)}{\sigma}$ . Symmetry of a numerical scheme is defined by considering its adjoint method: For a given method  $u^n \mapsto u^{n+1} = \Phi_{\tau}(u^n)$  its adjoint method is defined as  $\widehat{\Phi}_{\tau} := \Phi_{-\tau}^{-1}$ .

### Definition 1 (See for example Definition V.1.4 in [HLW10]) The method $\Phi_{\tau}$ is called symmetric if $\Phi_{\tau} = \widehat{\Phi}_{\tau}$ .

The scheme (6.4) is not symmetric in the sense of definition (1) because the adjoint method is given by

$$u^{n+1} = e^{i\tau\Delta}u^n - i\tau \left( \left(u^{n+1}\right)^2 (\varphi_1(2i\tau\Delta) - \varphi_2(2i\tau\Delta))\overline{u^{n+1}} \right) - i\tau e^{i\tau\Delta} \left( \left( e^{-i\tau\Delta}u^{n+1} \right)^2 \varphi_2(2i\tau\Delta) (e^{-i\tau\Delta}\overline{u^{n+1}}) \right) + \frac{\tau^2}{2} \left( \left| u^{n+1} \right|^4 u^{n+1} \right),$$

which is implicit as opposed to the original scheme (6.4), which is explicit.

The derivation of new schemes which are structure preserving and at the same time allow for low-regularity approximations was first addressed in the specific case of the KdV, the Klein-Gordon (KG), the NLS equation and the isotropic Landau–Lifschitz equation in the recent work of [MS23], [WZ22a], [AB23b] and [BMS22] respectively. A further symmetric low-regularity integrator with good long time behaviour was introduced in [FMS23]; see also [MS23] for the construction of symplectic resonance-based schemes. Let us also highlight the work [WY22] which was the first low regularity method which allowed for high order mass conservation (for fixed time). However, all these results are yet again tailored to the particular structure of the equation, and bespoke calculations made on individual resonance structure of the equation at hand. Furthermore, they are restricted to second order and not always optimal in the sense of regularity.

This motivates the study of systematic constructions of symmetric resonance based schemes that we address in the present work. In particular, we develop a unified framework of symmetric resonance based schemes which preserve central symmetries of the system (6.1) while allowing for good approximation in the regimes treated by [BS22]. We extend the resonance decorated trees approach introduced in [BS22] to a richer framework by exploring different ways of iterating Duhamel's formula, capturing the dominant parts while interpolating the lower parts of the resonances in a symmetric

### 6.1. Introduction

manner. This gives a range of new numerical schemes with more degrees of freedom than the original framework from [BS22]. Our new framework allows us to recover previously constructed low-regularity symmetric schemes such as [AB23b], but also introduce new symmetric low regularity schemes which are optimal in the sense of regularity - in the spirit of [BS22]. An example of such a method introduced in the present work is (6.63) matching the regularity obtained for the non-symmetric scheme given in [BS22]. In addition, as opposed to the previous works [BS22, ABBS22a, ABBS22b] the schemes we introduce here do not need to be accompanied by well-chosen filter functions in order to obtain stability of the scheme. Indeed, our construction based on interpolation rather than Taylor series expansion of the non-dominant parts of the nonlinear frequency interactions directly leads to stable schemes, see also [RS21, AB23a].

Our main result is the new general resonance based scheme presented in Definition (6.3.9), with its error structure given in Theorem (6.3.15), the latter of which is a consequence of [BS22]. We show that this scheme is symmetric in Theorem (6.4.3) and that it is contained within a forest formula in Theorem (6.3.11). Our general framework is illustrated on concrete examples in Section (6.4.3) and simulations show the better structure preserving properties as well as the convergence properties of the scheme. This was only possible through a significant extension of the algebraic structures proposed in [BS22] by introducing a new forest formula in Theorem (6.3.6). This formula is used for finding new symmetric schemes and have their own interest by providing a new parametrisation of low regularity schemes allowing for implicitness in the schemes and thus resembling more closely the formulation of classical schemes such as Runge–Kutta methods or exponential integrators. We derive conditions on the coefficients of these formulae for having a symmetric scheme, see Proposition (6.4.7).

**Remark 6.1.1** Up to now we were faced with a choice between structure preservation and low-regularity approximation properties. This is exhibited in Figure (6.1) where we study the cubic NLS equation and compare the preservation of energy of the Strang splitting (a symmetric splitting method) against previous resonance based integrators (Bruned & Schratz 2022 [BS22]) for smooth  $C^{\infty}$  and  $H^2$  data.



(b) Low regularity data,  $u_0 \in H^2$  and M = 64.

Figure 6.1 – Long-time relative error in the energy of the NLS equation with time-step  $\tau = 0.02$ .

The Strang splitting almost preserves the energy over long times for smooth solutions, but suffers from numerical energy blow up for rougher data. The resonance based integrator Bruned & Schratz 2022 [BS22] on the other hand only achieves approximate energy preservation up to short times (for both smooth and rougher data). Our novel resonance based midpoint method (6.63) bridges this gap allowing for numerical long-time approximate energy conservation even at low regularity, see Figure (6.1b).

Let us now take a closer look at smooth solutions, where we find a surprising additional characteristic of our new scheme (6.63). Note that long-time structure preservation properties apply only subject to a CFL condition for Strang splitting methods applied to the NLS equation. More precisely the time step size  $\tau$  has to be chosen such that  $\tau \leq M^{-2}$  where M is the number of degrees of freedom in the spatial discretisation, see for instance [Fao12] and references therein for a detailed discussion. This step size restriction is not only a theoretical technicality, but also observed in numerical experiments. The long-time energy preservation in the Strang splitting drastically breaks down if we start to increase the number of Fourier modes M, i.e., move from "ODE to PDE", see Figure (6.1a) versus Figure (6.2), where we double the Fourier modes in our discretisation. A very interesting feature of our new resonance-based constructions appears to be that in numerical experiments the long-time behaviour of the method seemingly does not depend on the number M of spatial modes used.



Figure 6.2 – Long-time relative error in the energy of the NLS equation with time-step  $\tau = 0.02$ .

In summary, the long-time dynamics shown in Figures (6.1)-(6.3) is representative of the behaviour of these methods and is observed in a large number of numerical experiments. Namely, we have that the Strang splitting is able to approximately preserve the energy over long times only for a small number of spatial discretisation points and for smooth initial data ( $M \ll \tau^{-1/2}$ ). In contrast, our proposed symmetric low-regularity integrators can achieve this feat even for low regularity solutions and with a large number of spatial discretisation points.

Outline of the article



(b) Zoom on time interval  $t \in [0, 40]$ , low-regularity data  $u_0 \in H^2$  and M = 256.

Figure 6.3 – Long-time relative error in the energy of the NLS equation with time-step  $\tau = 0.02$ .

The remainder of this manuscript is structured as follows. To begin with, in section (6.2), we outline the main ideas in the construction of symmetric resonance based schemes, before formalising those ideas more rigorously in the subsequent sections. In particular, in section (6.3) we firstly recall the decorated tree framework introduced in [BS22] for nonsymmetric resonance based schemes. We then generalise this framework in order to capture a broader class of resonance based integrators allowing for polynomial interpolation of lower order parts in the approximation (cf. section (6.3.1)). This leads to a general framework taking the form of a forest formula that can capture a wide class of implicit and explicit resonance based schemes and which is introduced in section (6.3.2). This forest formula motivates our consideration of a particular way of iterating Duhamel's formula to generate a subclass of resonance based schemes described by this general framework in section (6.3.3) which turns out to be sufficiently general to allow us to find symmetric resonance based schemes of arbitrary order in this class. In section (6.4) we then describe how symmetric interpolation in the construction from section (6.4.2). We conclude the section with examples of the new symmetric resonance based integrators that can be found using our novel framework in section (6.4.3). In section (6.5), we provide numerical experiments demonstrating the favorable practical performance of the new symmetric resonance based schemes that we were able to develop using our formalism.

Acknowledgements. K. S. and Y. A. B. received financial support from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (grant agreement No. 850941). Y. B. received financial support from the European Research Council (ERC) (ERC Starting Grant grant agreement No. 101075208) and from the ANR via the project LoRDeT (Dynamiques de faible régularité via les arbres décorés) from the projects call T-ERC\_STG. G. M. gratefully acknowledges funding from the European Union's Horizon Europe research and innovation programme under the Marie Sklodowska–Curie grant agreement No. 101064261.

# 6.2 Main ideas of the derivation of symmetric resonance based schemes

Before diving into a more abstract construction of the algebraic structures describing our novel resonance based schemes let us begin by outlining the main assumptions on the type of equation we consider as well as the blueprint for the construction of implicit (and specifically symmetric) resonance based integrators for equations of the form (6.1).

### Assumptions

We impose periodic boundary conditions, i.e.  $x \in \mathbf{T}^d$ . We assume that the differential operator  $\mathcal{L}$  is real and that the differential operators  $\mathcal{L}(\nabla)$  and  $|\nabla|^{\alpha}$  shall cast in Fourier space into the form

$$\mathcal{L}(\nabla)(k) = k^{\sigma} + \sum_{\gamma:|\gamma| < \sigma} a_{\gamma} \prod_{j} k_{j}^{\gamma_{j}}, \qquad |\nabla|^{\alpha}(k) = \sum_{\gamma:|\gamma| \le \alpha} \prod_{j=1}^{a} k_{j}^{\gamma_{j}}$$
(6.5)

for some  $\alpha \in \mathbf{R}$ ,  $\sigma \in \mathbf{N}$ ,  $\gamma \in \mathbf{Z}^d$  and  $|\gamma| = \sum_i \gamma_i$ , where for  $k = (k_1, \ldots, k_d) \in \mathbf{Z}^d$  and  $m = (m_1, \ldots, m_d) \in \mathbf{Z}^d$  we set

$$k^{\sigma} = k_1^{\sigma} + \ldots + k_d^{\sigma}, \qquad k \cdot m = k_1 m_1 + \ldots + k_d m_d.$$

### Construction of implicit resonance based schemes

We first rewrite (6.1) in Duhamel's form

$$u(t,x) = e^{it\mathcal{L}}v(0,x) - ie^{it\mathcal{L}} \int_0^t e^{-is\mathcal{L}} |\nabla|^{\alpha} p(u(s,x),\bar{u}(s,x)) ds$$

where we have used  $\mathcal{L} = \mathcal{L}(\nabla, \frac{1}{\varepsilon})$  as a short hand notation. Then, if we move to Fourier space by denoting  $u_k$  and  $v_k$  the k-th Fourier coefficients of u and v, one obtains:

$$u_{k}(t) = e^{it\mathcal{L}(k)}v_{k}(0) - ie^{it\mathcal{L}(k)} \int_{0}^{t} e^{-is\mathcal{L}(k)} |\nabla|^{\alpha}(k)p_{k}(u(s,x),\bar{u}(s,x))ds$$
(6.6)

where  $\mathcal{L}(k)$  and  $|\nabla|^{\alpha}(k)$  are the differential operators  $\mathcal{L}$  and  $|\nabla|^{\alpha}$  mapped in Fourier space. The term  $p_k(u(s, x), \bar{u}(s, x))$ stands for the Fourier transform of the product. For example, in the case of NLS we have  $\alpha = 0$ ,  $\mathcal{L} = \Delta$ , and  $p(u, \bar{u}) = u^2 \bar{u}$ . The equation (6.6) becomes

$$u_k(t) = e^{-itk^2} v_k(0) - ie^{-itk^2} \int_0^t e^{isk^2} \left( \sum_{k=-k_1+k_2+k_3} \bar{u}_{k_1}(s) u_{k_2}(s) u_{k_3}(s) \right) ds.$$

The product  $|u|^2 u$  becomes a convolution on the coefficients in Fourier space, where we note that the minus pre-multiplying  $k_1$  is due to the conjugate. We have also used the fact that the Fourier transform of  $e^{it\Delta}$  is  $e^{-itk^2}$ . The first step in the construction of resonance based schemes consists in iterating (6.6) inside the nonlinearity which produces a sum of oscillatory integrals that can be described by decorated trees as introduced in [BS22]. Namely, in [BS22] we iterate only with (6.6) which corresponds to a left end point iteration, meaning that given a time step  $\tau$ , we always take the left end point approximation in the linear part  $\exp(-itk^2)v_k(0)$  on the interval  $[0, \tau]$ . Hence, in general we do not obtain a symmetric scheme. Indeed, one has the possibility to write Duhamel's formula around any point in the interval  $[0, \tau]$ . In particular, if we set for  $s \in [0, \tau]$ 

$$I(k, u, s, t) = e^{i(t-s)\mathcal{L}(k)}u_k(s)$$

$$-ie^{it\mathcal{L}(k)}\int_s^t e^{-i\tilde{s}\mathcal{L}(k)}|\nabla|^{\alpha}(k)p_k(u(\tilde{s}, x), \bar{u}(\tilde{s}, x))d\tilde{s}$$

$$(6.7)$$

then we have the identity:

$$u_k(t) = I(k, u, v, s, t).$$

From the identity (6.7), we will obtain implicit schemes. We can take a weighted sum of the various Duhamel's iterations (6.7) ultimately to arrive at schemes with a large number of additional degrees of freedom. The sum that we will use for a large part of this paper is the midpoint rule, that is:

$$u_k(t) = \frac{1}{2}(I(k, u, 0, t) + I(k, u, \tau, t)).$$
(6.8)

For example, to construct a symmetric resonance based scheme of a desired order we can start the first iteration for  $u_k(\tau)$  with the left end point Duhamel's formula and then we iterate the midpoint rule (6.8). We can express this iteration in terms of the following tree series

$$U_{\text{mid},k}^{r}(\tau) = e^{i\tau\mathcal{L}}u_{k}(0) + \sum_{T \in \mathcal{V}_{k}^{r}} \frac{\Upsilon_{\text{mid}}^{p}(T)(u,\tau)}{S(T)} (\Pi_{\text{mid}}T)(\tau)$$
(6.9)

where  $\mathcal{V}_k^r$  is a set of decorated trees of size at most r+1 which incorporate the frequency k. These trees encode via  $(\Pi_{\text{mid}}T)(\tau)$ , iterated integrals of depth at most r+1. The coefficient S(T) is the symmetry factor associated to the tree T and  $\Upsilon_{\text{mid}}^p(T)(u,\tau)$  is the coefficient appearing in the iteration of Duhamel's formulation depending on the nonlinearity p. The coefficients  $\Upsilon_{\text{mid}}^p(T)(u,\tau)$  depend on  $\tau$  as with our midpoint iteration, we finish always on terms of the form:

$$\frac{1}{2} \Big( e^{is\mathcal{L}} u_{k_j}(0) + e^{i(s-\tau)\mathcal{L}} u_{k_j}(\tau) \Big) = e^{is\mathcal{L}} \frac{1}{2} \Big( u_{k_j}(0) + e^{-i\tau\mathcal{L}} u_{k_j}(\tau) \Big).$$

It is natural to absorb the term  $e^{is\mathcal{L}}$  in the definition of  $(\Pi_{\text{mid}}T)(\tau)$  and

$$\frac{1}{2} \Big( u_{k_j}(0) + e^{-i\tau \mathcal{L}} u_{k_j}(\tau) \Big)$$

into the definition of  $\Upsilon_{\text{mid}}^p(T)(u,\tau)$ . These aforementioned quantities are described in detail in Section (6.3.3). The sum (6.9) can be viewed as a first numerical approximation by keeping only the iterated integrals of order below r of the infinite series describing formally the solution of (6.1). In Proposition (6.4.1), we show that (6.9) is a symmetric scheme.

In order for the scheme to have a suitable local error when applied to low-regularity solutions, it is necessary to replace each oscillatory integral  $(\Pi_{mid}T)(\tau)$  appearing in the finite sum (6.9) by a low regularity approximation that embeds the resonance structure into the numerical discretization. Our novel approach is to try and perform this approximation in a symmetric manner. Let us explain briefly how it works. Suppose we aim to discretise an oscillatory integral of the form

$$\int_0^t e^{is\mathcal{L}} ds, \quad \mathcal{L} = \mathcal{L}_{\rm dom} + \mathcal{L}_{\rm low},$$

where we have split the operator into a dominant part  $\mathcal{L}_{dom}$  that we will integrate exactly and a lower part  $\mathcal{L}_{dom}$  that we will approximate (cf. section (6.3.1) for a definition of these quantities). A typical example arising in the case of NLS is

$$\mathcal{L} = k^2 + k_1^2 - k_2^2 - k_3^2, \quad \mathcal{L}_{dom} = 2k_1^2 \quad \mathcal{L}_{low} = -2k_1(k_2 + k_3) + 2k_2k_3,$$

where  $k = -k_1 + k_2 + k_3$ . We see that the exact integration of

$$\int_0^t e^{is\mathcal{L}_{\text{dom}}} ds = \frac{e^{2isk_1^2} - 1}{2ik_1^2}$$

can be mapped back to physical space as  $1/k^2$  corresponds to  $\Delta^{-1}$ . This property of being able to write the scheme in physical space is crucial for efficient numerical implementation: If we have an expression in physical space the differential operators can be computed quickly in *frequency* space using the Fast Fourier Transform (FFT) while any polynomial-type nonlinear term can be computed quickly in *physical* space since it corresponds to a local operation on the function values on a grid. Now, it remains to approximate the lower part. For this task, we use a polynomial interpolation with m + 1 points on  $[0, \tau]$  denoted by  $a_j \tau$ . We note that the error incurred by this polynomial interpolation will be one of the determining factors of the convergence order of our overall numerical scheme, and thus we highlight that this interpolation can be done to any given order. We take r + 1 distinct interpolation points  $0 \le a_0 < a_1 < \cdots < a_r \le 1$  which are symmetrically distributed such that  $a_j = 1 - a_{r-j}, j = 0, \ldots, r$ . Let us denote the corresponding nodal polynomials by  $p_{j,r}$  such that

$$p_{j,r}(a_m\tau) = \delta_{j,m}.$$

Then, we define the following approximation

$$\tilde{p}_r(f,\xi) = \sum_{j=0}^r f(a_j) p_{j,r}(\xi), \quad f(a_j) = e^{ia_j \tau \mathcal{L}_{\text{low}}}.$$

We have the following local error

$$f(\xi) - \tilde{p}_r(f,\xi) = \mathcal{O}(\prod_{j=0}^r (\xi - a_j \tau) (i\mathcal{L}_{\text{low}})^{r+1})$$
(6.10)

which requires less regularity than if we had chosen to base our approximation on  $\mathcal{L}$  instead of  $\mathcal{L}_{low}$  as classical schemes do.

In order to arrive at a numerical scheme, we will in the following introduce the low-regularity symmetric approximation operator of  $\Pi_{\text{mid}}$  denoted by by  $\Pi_{\text{mid}}^{n,r}$ . Here, r corresponds to the order of the discretization and n is the a priori regularity assumed on the initial data v. Namely, we assume  $v \in H^n$ , where  $H^n$  is a Sobolev space. The general scheme then takes the form:

$$U_{\text{mid},k}^{n,r}(\tau,u) = \sum_{T \in \mathcal{V}_k^r} \frac{\Upsilon_{\text{mid}}^p(T)(u,\tau)}{S(T)} (\Pi_{\text{mid}}^{n,r}T)(\tau).$$
(6.11)

We will show in Theorem (6.4.3) that this scheme is symmetric. The local error structure for each approximated iterated integral is given by

$$(\Pi_{\mathrm{mid}}T - \Pi_{\mathrm{mid}}^{n,r}T)(\tau) = \mathcal{O}\big(\tau^{r+2}\mathcal{L}^r_{\mathrm{low}}(T,n)\big),\tag{6.12}$$

where  $\mathcal{L}_{low}^r$  involves only lower order derivatives. Its proof is exactly the same as in [BS22]. The local error does not depend on the choice of Duhamel's iteration and polynomial interpolations. The form of the scheme draws its inspiration from the treatment of singular stochastic partial differential equations (SPDEs) via Regularity Structures in [Hai14, BHZ19, BCCH20, BHZ20]. These decorated tree expansions are generalization of the B-series widely used for ordinary differential equations, we refer to [But72, CCO08, HLW10, MKL13] and tree series used for dispersive equations [Chr07, GKO13, Gub12, LO13, HLO20]. In the end, one obtains an approximation of u under much lower regularity assumptions than classical methods (e.g., splitting methods, exponential integrators [CG12, Fao12, HLW10, HO10, Ho110, Law67, Lub08, LR04, MQ02, SSC18]) require, which in general introduce the local error

$$\mathcal{O}\big(\tau^{r+2}\mathcal{L}^r(T,n)\big) \tag{6.13}$$

involving the full high order differential operator  $\mathcal{L}^r$ . Indeed, denoting by  $\mathcal{D}(\cdot)$  the domain of a given operator, we have that  $\mathcal{D}(\mathcal{L}_{low}) \supset \mathcal{D}(\mathcal{L})$ , meaning that the local error structure (6.12) allows us to deal with a rougher class of solutions than the classical error bound (6.13). Let us mention that the local error analysis can be nicely understood via a Birkhoff factorisation of the character  $\Pi_{\text{mid}}^{n,r}$  (see [BS22, BEF20]) that involves a deformed Butcher-Connes-Kreimer coproduct (see [But72, CK99, CK00, BS22, BM23]). In the present work, we push further the algebraic perspective by writing several forest formulae that can be used to represent a larger class of low regularity schemes. These forest formulae take the following form:

$$u_{k}^{\ell+1} = e^{i\tau\mathcal{L}}u_{k}^{\ell} + e^{i\tau\mathcal{L}}\sum_{T\in\tilde{\mathcal{V}}_{k}^{T}}\sum_{\mathbf{a}\in[0,1]^{\tilde{E}_{T}}}\sum_{\chi\in\{0,1\}^{L_{T}}}\sum_{T_{0}\cdot T_{1}...\cdot T_{m}\subset T}C_{T}$$

$$b_{\mathbf{a},\chi,T,T_{0}\cdot...\cdot T_{m}}(\tau,i\tau\mathscr{F}_{\mathrm{dom}}(T_{j}),j\in\{0,...,m\})$$

$$\prod_{e\in\tilde{E}_{T_{j}}}e^{i\tau a_{e}\mathscr{F}_{\mathrm{low}}(T_{j}^{e})}\frac{\Upsilon_{\chi}^{p}(T)(u_{k_{v}}^{\ell+\chi_{v}},v\in L_{T},\tau)}{S(T)}.$$
(6.14)

Below, we give a brief description of the notation of this forest formula before introducing each term in full detail in section (6.3). Here,  $\mathcal{L}$  si the full operator of (6.1),  $\tilde{\mathcal{V}}_k^r$  is a finite set of decorated trees,  $\tilde{E}_T$  denotes the edges of T that correspond to a time integration. These time integrals are discretised with a low regularity approximation. Therefore, we have to use a map **a** on these edges that specifies which interpolation points have been used. This corresponds to the following term

$$\prod_{e \in \tilde{E}_{T_i}} e^{i\tau a_e \mathscr{F}_{\text{low}}(T_j^e)}$$

Here  $\mathscr{F}_{low}(T_j^e)$  denote the lower part of the various discretisations where  $T_j^e$  is included into  $T_j$ . The set  $L_T$  are the leaves of T associated to some  $u_{k_u}$  and the map  $\chi$  specifies if they are evaluated at the right  $(u_{k_u}^\ell)$  or left end point  $(u_{k_u}^{\ell+1})$ . The coefficients  $\Upsilon^p_{\chi}(T)(u_{k_v}^{\ell+\chi_v}, v \in L_T, \tau)$  depend on the structure of the equation and the way one iterates Duhamel's formula. One essential choice of this forest formula is the splitting of T into a forest  $T_0 \cdot \ldots \cdot T_m$  where the  $T_i$  are decorated trees. This allows us to encode all the lower parts of the resonances  $\mathscr{F}_{low}(T_j^e)$  and all their dominant

parts  $\mathscr{F}_{dom}(T_j)$  that appear in the low regularity discretisation. As we shall see below, this forest splitting is a crucial novelty necessary for the construction and classification of symmetric low-regularity integrators. For the splitting, one can use a Butcher-Connes-Kreimer coproduct (see Section (6.3.2)) or a deformed Butcher-Connes-Kreimer coproduct used for the local error analysis. In our applications so far, the forest formula without deformation is enough for finding symmetric schemes hence this is the one presented in this article. We derive a condition on the coefficients  $b_{\mathbf{a},\chi,T,T_0,\ldots,T_m}$  in Proposition(6.4.7) that allows to find symmetric schemes. The coefficients  $b_{\mathbf{a},\chi,T,T_0,\ldots,T_m}$  do not depend on the frequencies that are node decorations of the trees  $T, T_j$ . One can see them only as functions of the dominant parts of the various operators encountered during the discretisation. The term  $C_T$  is a structure term depending on the frequencies that encode the various operators that appear in the iterated integral given by T. We conclude this section with a few remarks concerning the structures introduced and the properties of the resulting low-regularity schemes.

**Remark 6.2.1** The forest formula appear in the BPHZ algorithm [BP57, Hep69, Zim69] for renormalising Feynman diagrams and was later used for renormalising singular SPDEs in [BHZ19, CH16] with an extension of the algebraic structure.

**Remark 6.2.2** The scheme (6.11) has been generalized to non-polynomial nonlinearities and to parabolic equations in [ABBS22b] with the use of nested commutators first introduced in [RS21]. The Birkhoff factorisation discovered in [BS22] is not available in this case. It is also not obvious to translate forest formulae into this context. Indeed, due to the fact that the formula is written in Fourier space, there is no order on the operators written in Fourier space. This is not the case in physical space. But one can repeat the construction of the scheme  $U_k^{n,r}$  in this context and this scheme should be symmetric as the recursive proofs in Section (6.4.1) seem robust to this case.

**Remark 6.2.3** The schemes presented in [BS22] have been adapted to a probabilistic setting by proposing a low regularity approximation [ABBS22a] of the second moment of the Fourier coefficient of the solution, i.e.  $\mathbb{E}(|u(v^{\eta}, \tau)|^2)$  where  $v^{\eta}$  is a random initial data. In this context, one has to work with paired decorated trees. It possible to write symmetric schemes for approximating this second moment using our approach. Also, one can set up an equivalent forest formula on these paired decorated trees. One open direction is to understand the connection between the algebraic tools developed for these numerical schemes and the tools used for the rigorous derivation of the wave kinetic equation (WKE) for NLS is performed in [DH21, DH23, ACG21]

**Remark 6.2.4** The central novelty of the present work is the structured understanding of implicit and, in particular, symmetric low-regularity integrators. The local error bounds we use in this paper often rely on the previous local error derivations first introduced in [BS22]. Indeed, the scheme  $U_k^{n,r}(\tau, u)$  is of the form (6.14) but the local error analysis comes from the fact that it is defined recursively via the character  $\Pi_{\text{mid}}^{n,r}$  and therefore the tools from [BS22] are available. If one found a new scheme by choosing the coefficients  $b_{\mathbf{a},\chi,T,T_0\cdots T_m}$ , it is not clear how to get directly the local error analysis and check that the scheme is optimal in terms of regularity.

We also make the important remark that given that we derive schemes which are of implicit nature, an additional fixed-point argument needs to be performed on the numerical flow in order to rigorously buckle the local error bounds, we refer to the works of [MS22, AB23b] where this analysis is made in detail.

**Remark 6.2.5** On this forest formula, we have identified symmetric schemes and, in addition, we have provided a general recursive mechanism to derive symmetric schemes for a large class of PDEs. One can wonder if such an approach could be repeated for other symmetries. Indeed, we believe that our techniques are fairly general. The degrees of freedom offered by different Duhamel's iterations and interpolations should allow us to capture other symmetries at low regualrity using variants of the recursive scheme  $U_k^{n,r}(\tau, u)$ . One degree of freedom which has not been used in full generality is the splitting of the operator into dominant and lower part :

$$\mathcal{L} = \mathcal{L}_{\mathrm{dom}} + \mathcal{L}_{\mathrm{low}}.$$

Right now, it is governed by Definition 6.3.1 that guarantees to get a resonance-based scheme and a scheme which can be written in physical space. For symplectic schemes, one expects to have symmetries between the frequencies of  $\mathcal{L}_{dom}$ and those  $\mathcal{L}_{low}$ . One should have the possibility of refining this splitting for encapsulating some symmetries as has been done for the 1D NLSE and the KdV equation in recent work [MS23]. The rest of the construction of the scheme remains unchanged. Consequently, a natural line of future research is the study of such symmetries ( $\rho$ -reversibility, preservation of quadratic invariants, etc.) directly on a structured tree or forest expansion of the numerical schemes comparable to the use of B-series in the study of structure preservation properties of methods for ODEs. We believe that the forest formulae presented in the current work take a first step in this direction. **Remark 6.2.6** Let us close this section with an interesting, but crucial observation: In the context of ODEs it is well known that symmetric methods are of even order (cf. [HLW10, Theorem IX.2.2]). In general this is, however, *not* the case for PDEs as the rate of convergence depends intrinsically on the regularity of the solution, and hence convergence at even order only holds if sufficient regularity requirements are met by the solution. For instance, the resonance based midpoint method for the NLS equation takes the form (see Section (6.4.3) below for its derivation)

$$u^{n+1} = e^{i\tau\Delta}u^{n} -i\frac{\tau}{16}e^{i\tau\Delta}\Big((u^{n} + e^{-i\tau\Delta}u^{n+1})^{2}\varphi_{1}(-2i\tau\Delta)\Big(\overline{u^{n}} + e^{i\tau\Delta}\overline{u^{n+1}}\Big)\Big) -i\frac{\tau}{16}\Big((e^{i\tau\Delta}u^{n} + u^{n+1})^{2}\varphi_{1}(2i\tau\Delta)\Big(e^{-i\tau\Delta}\overline{u^{n}} + \overline{u^{n+1}}\Big)\Big).$$
(6.15)

This scheme is symmetric and first order with optimal local error structure in the sense of [BS22], as its first order local error structure  $\mathcal{O}(\tau^2 \nabla u)$  does not require more regularity on the solution than the asymmetric first order resonance based schemes of [BS22, OS18]. As the scheme (6.15) is symmetric it is, for  $C^{\infty}$  solutions, naturally of even order, hence, not only of order one, but also of order two. However, a closer look shows that its second order convergence is only attained for sufficiently regular solutions: With a similar error analysis as introduced in [AB23b] one can show that at second order the symmetric scheme (6.15) introduces a local error of type  $\mathcal{O}(\tau^3 \nabla \Delta u)$  which requires the boundedness of three additional derivatives in order to attain second order convergence. For initial data in lower order spaces than  $H^3$ , one can obtain fractional convergence of order *less than two*, see [AB23b]. We make the additional remark that in view of [BS22], requiring a local error of  $\mathcal{O}(\tau^3 \nabla \Delta u)$  is not optimal in the sense of regularity. Indeed, we recall that the second-order non-symmetric resonance based integrators [BS22] obeys the favourable error structure  $\mathcal{O}(\tau^3 \Delta u)$ , hence asking for one less derivative on the solution.

# 6.3 Decorated trees and generalised resonance based schemes

The main object of this manuscript is to formalise the construction of symmetric resonance based schemes as outlined in section (6.2). To achieve this we resort to a new, generalised tree formalism which has already seen (in much simpler version) significant success in the construction of explicit (asymmetric) resonance based schemes (cf. [BS22]). In the present section we will begin by recalling some of the main definitions in this framework before generalising the construction to incorporate the possibility of implicit low-regularity integrators before ultimately culminating in a forest formula (6.31) which captures a broad class of resonance based numerical schemes in such way that we can later characterise those schemes in this class which are symmetric in the sense of definition (1).

We recall briefly the structure of decorated trees introduced in [BS22, Sec. 2]. Let  $\mathfrak{L}$  a finite set and frequencies  $k_1, ..., k_m \in \mathbf{Z}^d$ . We suppose we are given a fixed time step  $\tau > 0$ . The set  $\mathfrak{L}$  parametrizes a set of differential operators with constant coefficients, whose symbols are given by the polynomials  $(P_t)_{t \in \mathfrak{L}}$ . These operators are given in Fourier space and therefore the polynomials will be evaluated in the frequencies  $k_i$ . We define the set of decorated trees  $\hat{\mathcal{T}}$  as elements of the form  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}} = (\mathcal{T},\mathfrak{n},\mathfrak{o},\mathfrak{e})$  where

- T is a non-planar rooted tree with root  $\rho_T$ , node set  $N_T$  and edge set  $E_T$ . We denote the leaves of T by  $L_T$ . T must also be a planted tree which means that there is only one edge connecting the root to the rest of the tree.
- the map  $\mathfrak{e}: E_T \to \mathfrak{L} \times \{0, 1\}$  are edge decorations. The set  $\{0, 1\}$  encodes the action of taking the conjugate, and determines the sign of the frequencies at the top of this edge. Namely, we have that 1 corresponds to a conjugate and to multiplying by (-1) the frequency on the node above and adjacent to this edge.
- the map  $\mathbf{n}: N_T \setminus \{\varrho_T\} \to \mathbf{N}^2$  are node decorations. For every inner node v, this map encodes a monomial of the form  $\xi^{\mathfrak{n}_1(v)}\tau^{\mathfrak{n}_2(v)}$  where  $\xi$  is a free time variable belonging to  $[0, \tau]$ . This is a novelty from [BS22] where we do not have factors in  $\tau$ . We need it as in the sequel, we will consider integrals of the form  $\int_{\tau}^{\xi} ... ds$ .
- the map  $\mathbf{o}: N_T \setminus \{\varrho_T\} \to \mathbf{Z}^d$  are node decorations. These decorations are frequencies that satisfy for every inner node u:

$$(-1)^{\mathfrak{p}(e_u)}\mathfrak{o}(u) = \sum_{e=(u,v)\in E_T} (-1)^{\mathfrak{p}(e)}\mathfrak{o}(v)$$
(6.16)

where  $\mathfrak{e}(e) = (\mathfrak{t}(e), \mathfrak{p}(e))$  is the edge decoration of e with  $\mathfrak{t}(e) \in \mathfrak{L}$  and  $\mathfrak{p}(e) \in \{0, 1\}$  and  $e_u$  is the unique edge outgoing from u which is part of the path connecting u to the root. We denote this edge by (v, u). From this definition, one can see that the node decorations at the leaves  $(\mathfrak{o}(u))_{u \in L_T}$  determine the decoration of the inner nodes. One can call this identity Kirchhoff's law. We assume that the node decorations at the leaves are linear combinations of the  $k_i$ with coefficients in  $\{-1, 0, 1\}$ .

– we assume that the root of T has no decoration.

When the node decoration  $\mathfrak{n}$  is zero, we will denote the decorated trees  $T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}$  as  $T_{\mathfrak{e}}^{\mathfrak{o}} = (T,\mathfrak{o},\mathfrak{e})$ . The set of decorated

trees satisfying such a condition is denoted by  $\hat{\mathcal{T}}_0$ . We set  $\hat{H}$  (resp.  $\hat{H}_0$ ) the (unordered) forests composed of trees in  $\hat{\mathcal{T}}$  (resp.  $\hat{\mathcal{T}}_0$ ) with linear spans  $\hat{\mathcal{H}}$  and  $\hat{\mathcal{H}}_0$ . The forest product is denoted by  $\cdot$ , the empty forest by **1**. Elements in  $\hat{\mathcal{T}}$  are abstract representation of iterated time integrals and elements in  $\hat{H}$  are a product of them.

We now introduce how one can represent uniquely decorated trees by using symbolic notations. We denote by  $\mathcal{I}_o$ , an edge decorated by  $o = (\mathfrak{t}, \mathfrak{p}) \in \mathfrak{L} \times \{0, 1\}$ . We introduce the operator  $\mathcal{I}_o(\lambda_k^{\ell} \cdot) : \hat{\mathcal{H}} \to \hat{\mathcal{H}}$  that merges all the roots of the trees composing the forest into one node decorated by  $(\ell, k) \in \mathbf{N}^2 \times \mathbf{Z}^d$ . The new decorated tree is then grafted onto a new root with no decoration. If the condition (6.16) is not satisfied on the argument then  $\mathcal{I}_o(\lambda_k^{\ell} \cdot)$  gives zero. If  $\ell = 0$ , then the term  $\lambda_k^{\ell}$  is denoted by  $\lambda_k$  as a short hand notation for  $\lambda_k^0$ . The forest product between  $\mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1)$  and  $\mathcal{I}_{o_2}(\lambda_k^{\ell_2}F_2)$  is given by:

$$\mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1)\mathcal{I}_{o_2}(\lambda_{k_2}^{\ell_2}F_2) := \mathcal{I}_{o_1}(\lambda_{k_1}^{\ell_1}F_1) \cdot \mathcal{I}_{o_2}(\lambda_{k_2}^{\ell_2}F_2).$$

The right hand side of the previous equality could be understood as a set where we can repeat elements and the forest product is the disjoint union of these sets. Any decorated tree T is uniquely represented as

$$T = \mathcal{I}_o(\lambda_k^{\ell} F), \quad F \in \hat{H}.$$

Given an iterated integral, its size is given by the number of integrations in time. Therefore, we suppose we are given a subset  $\mathfrak{L}_+$  of  $\mathfrak{L}$  that encodes edge decorations which correspond to time integrals that we have to approximate.

**Example 19** We illustrate the definitions introduced above with decorated trees coming from the NLS equation. We consider the following decorated tree:

$$T = \mathcal{I}_{(\mathfrak{t}_{1},0)} \left( \lambda_{k} \mathcal{I}_{(\mathfrak{t}_{2},0)} \left( \lambda_{k} \mathcal{I}_{(\mathfrak{t}_{1},1)} (\lambda_{k_{1}}) \mathcal{I}_{(\mathfrak{t}_{1},0)} (\lambda_{k_{2}}) \mathcal{I}_{(\mathfrak{t}_{1},0)} (\lambda_{k_{3}}) \right) \right) = \begin{pmatrix} k_{2} \\ k_{3} \\ k_{3} \\ k_{3} \end{pmatrix}$$

where  $k = -k_1 + k_2 + k_3$ ,  $\mathfrak{L} = \{\mathfrak{t}_1, \mathfrak{t}_2\}$ ,  $\mathfrak{L}_+ = \{\mathfrak{t}_2\}$ ,  $P_{\mathfrak{t}_1}(\lambda) = -\lambda^2$  and  $P_{\mathfrak{t}_2}(\lambda) = \lambda^2$ . We put the frequency decorations only on the leaves as those on the inner nodes are uniquely determined by them. In the table below, we explain the coding of the edges

Edge	Decoration	Operator
-	$(\mathfrak{t}_1, 0)$	$e^{itP_{\mathfrak{t}_1}(k)} = e^{-itk^2}$
	$(\mathfrak{t}_1,1)$	$e^{-itP_{\mathbf{t}_1}(k)} = e^{itk^2}$
	$(\mathfrak{t}_2,0)$	$-i \int_{0}^{t} e^{i\xi P_{t_{2}}(k)} \cdots d\xi = -i \int_{0}^{t} e^{i\xi k^{2}} \cdots d\xi$
	$(\mathfrak{t}_2,1)$	$-i\int_{0}^{t} e^{-i\xi P_{t_{2}}(k)} \cdots d\xi = -i\int_{0}^{t} e^{-i\xi k^{2}} \cdots d\xi$

In the end, T is an abstract version of the following integral:

$$-ie^{-itk^2} \int_0^t e^{i\xi k^2} e^{i\xi k_1^2} e^{-i\xi k_2^2} e^{-i\xi k_3^2} d\xi$$

The next combinatorial structure which we recall from [BS22] encodes abstract versions of a discretization of an oscillatory integral. We denote by  $\mathcal{T}$  the set of decorated trees  $T_{\mathfrak{e},r}^{\mathfrak{n},\mathfrak{o}} = (T,\mathfrak{n},\mathfrak{o},\mathfrak{e},r)$  where

$$- T^{\mathfrak{n},\mathfrak{o}}_{\mathfrak{e}} \in \hat{\mathcal{T}}$$

— The decoration of the root is given by  $r \in \mathbf{Z}, r \geq -1$  such that

$$r+1 \ge \deg(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}) \tag{6.17}$$

where deg is defined recursively by

$$\deg(\mathbf{1}) = 0, \quad \deg(F_1 \cdot F_2) = \max(\deg(F_1), \deg(F_2)),$$
$$\deg(\mathcal{I}_{(\mathfrak{t}, \mathfrak{p})}(\lambda_k^{\ell} F_1)) = |\ell| + \mathbf{1}_{\{\mathfrak{t} \in \mathfrak{L}_+\}} + \deg(F_1)$$

where  $\ell = (\ell_1, \ell_2), |\ell| = \ell_1 + \ell_2, F_1, F_2$  are forests composed of trees in  $\mathcal{T}$ . The quantity deg $(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}})$  is the maximum number of edges with type in  $\mathfrak{L}_+$ , corresponding to time integrations, and of node decorations  $\mathfrak{n}$  lying on the same path from one leaf to the root.

We call decorated trees in  $\mathcal{T}$  approximated decorated trees. The order of the approximation is encoded by a new decoration

at the root r. We denote by  $\mathcal{H}$  the vector space spanned by forests composed of trees in  $\mathcal{T}$  and  $\lambda^{\ell}$ ,  $\ell \in \mathbf{N}^2$  where  $\lambda^{\ell}$  is the tree with one node decorated by  $\ell$ . When the decoration  $\ell$  is equal to zero we identify this tree with the empty forest:  $\lambda^0 = \mathbf{1}$ . We now define the symbol  $\mathcal{I}_o^r(\lambda_k^{\ell} \cdot) : \mathcal{H} \to \mathcal{H}$ , as the same as  $\mathcal{I}_o(\lambda_k^{\ell} \cdot)$ , with the added adjunction of the decoration r which constrains the time-approximations to be of order r. It is given by:

$$\mathcal{I}_{o}^{r}(\lambda_{k}^{\ell}(\prod_{j}\lambda^{m_{j}}\prod_{i}\mathcal{I}_{o_{i}}^{r_{i}}(\lambda_{k_{i}}^{\ell_{i}}F_{i}))):=\mathcal{I}_{o}^{r}(\lambda_{k}^{\ell+\sum_{j}m_{j}}(\prod_{i}\mathcal{I}_{o_{i}}(\lambda_{k_{i}}^{\ell_{i}}F_{i}))).$$

We define a projection operator  $\mathcal{D}^r$  which depends on r and which is used during the construction of the numerical schemes in order to only retain the terms of order at most r. We define the map  $\mathcal{D}^r : \hat{\mathcal{H}} \to \mathcal{H}$  which assigns r to the root of a decorated tree. This implies a projection along the identity (6.17). It is given by

$$\mathcal{D}^{r}(\mathbf{1}) = \mathbf{1}_{\{0 \le r+1\}}, \quad \mathcal{D}^{r}\left(\mathcal{I}_{o}(\lambda_{k}^{\ell}F)\right) = \mathcal{I}_{o}^{r}(\lambda_{k}^{\ell}F)$$
(6.18)

and we extend it multiplicatively to any forest in  $\hat{\mathcal{H}}$ .

**Example 20** We illustrate the action of the map  $\mathcal{D}_r$  on the decorated tree T introduced in Example (19). One has:

$$\deg(T) = 1, \quad \mathcal{D}_r(T) = 0, r > 1, \quad \mathcal{D}_r(T) = \mathbf{r}$$

# 6.3.1 Dominant part and polynomial interpolation

Let us now introduce the operations used when approximating integrals represented by tree formalism as described above. We first recall [BS22, Def.2.2] that select higher degree terms in a polynomial of the frequencies.

**Definition 6.3.1** Let  $P(k_1, ..., k_n)$  a polynomial in the  $k_i$ . If the highest-degree monomials of P are of the form

$$a\sum_{i=1}^{n} (a_i k_i)^m, \quad a_i \in \{0, 1\}, \ a \in \mathbf{Z},$$

then we define  $\mathcal{P}_{dom}(P)$  as

 $\mathcal{P}_{\rm dom}(P) = a \left(\sum_{i=1}^{n} a_i k_i\right)^m.$ (6.19)

Otherwise, it is zero.

This definition is used for splitting an operator between a lower part and a dominant part. Indeed, if we consider the polynomial

$$P(k_1, k_2, k_3) = k^2 + k_1^2 - k_2^2 - k_3^3, \quad k = -k_1 + k_2 + k_3.$$

coming from the NLS equation, we observe that P can be rewritten into the form:

$$P(k_1, k_2, k_3) = 2k_1^2 - 2k_1(k_2 + k_3) + 2k_2k_3.$$

Then, we set

$$\mathcal{L}_{\text{dom}} = \mathcal{P}_{\text{dom}}(P) = 2k_1^2, \quad \mathcal{L}_{\text{low}} = (\text{id} - \mathcal{P}_{\text{dom}})(P).$$

We note that  $\mathcal{L}_{\text{dom}}$  asks for boundedness of two derivatives due to the factor  $k_1^2$  and  $\mathcal{L}_{\text{low}}$  only one because the latter consists only of cross products  $k_i k_j$ ,  $i \neq j$ . Another main reason for this splitting is to be able to map back to physical space the following integral:

$$\int_0^t e^{is\mathcal{L}_{\text{dom}}} ds = \frac{e^{it\mathcal{L}_{\text{dom}}} - 1}{i\mathcal{L}_{\text{dom}}}.$$

We observe that it is essential to map back to physical space the term  $\frac{1}{\mathcal{L}_{dom}}$  equal to  $\frac{1}{2k_1^2}$ . Such a term is given by  $\Delta^{-1}$  in physical space.

The next definition extracted from [BS22, Def. 2.6] allows us to compute recursively the various frequency interactions by extracting dominant and lower parts. Such a definition is required for the local error analysis and the forest formula given in the sequel.

**Definition 6.3.2** We recursively define  $\mathscr{F}_{dom}, \mathscr{F}_{low} : \hat{H}_0 \to \mathbb{R}[\mathbf{Z}^d]$  as:

$$\begin{aligned} \mathscr{F}_{\rm dom}(\mathbf{1}) &= 0 \quad \mathscr{F}_{\rm dom}(F \cdot \bar{F}) = \mathscr{F}_{\rm dom}(F) + \mathscr{F}_{\rm dom}(\bar{F}) \\ \mathscr{F}_{\rm dom}\left(\mathcal{I}_{(\mathfrak{t},\mathfrak{p})}(\lambda_k F)\right) &= \begin{cases} \mathcal{P}_{\rm dom}\left(P_{(\mathfrak{t},\mathfrak{p})}(k) + \mathscr{F}_{\rm dom}(F)\right), \text{ if } \mathfrak{t} \in \mathfrak{L}_+, \\ P_{(\mathfrak{t},\mathfrak{p})}(k) + \mathscr{F}_{\rm dom}(F), & \text{ otherwise} \end{cases} \\ \mathscr{F}_{\rm low}\left(\mathcal{I}_{(\mathfrak{t},\mathfrak{p})}(\lambda_k F)\right) &= (\mathrm{id} - \mathcal{P}_{\rm dom})\left(P_{(\mathfrak{t},\mathfrak{p})}(k) + \mathscr{F}_{\rm dom}(F)\right), \end{aligned}$$

where we recall that  $\mathfrak{L}_+$  is a subset of  $\mathfrak{L}$  that encodes edge decorations which correspond to time integrals. We extend these two maps to  $\hat{H}$  by ignoring the node decorations  $\mathfrak{n}$ .

In a nutshell the above recursive definition means that in the set  $\mathfrak{L} \setminus \mathfrak{L}_+$ , i.e. operators that do not correspond to integration, we collect all frequency contributions, and in the set  $\mathfrak{L}_+$ , i.e. operators that correspond to integration, we extract the dominant frequencies of the full integrand.

Example 21 We illustrate the previous definition on a simple decorated tree coming from the NLS equation.

$$T = \underbrace{\mathcal{I}_{(t_2,0)}(\lambda_k F)}_{(t_1,0)} = \mathcal{I}_{(t_1,1)}(\lambda_{k_1})\mathcal{I}_{(t_1,0)}(\lambda_{k_2})\mathcal{I}_{(t_1,0)}(\lambda_{k_3})$$

with  $k = -k_1 + k_2 + k_3$ . One has

$$\mathscr{F}_{\mathrm{dom}}(T) = \mathcal{P}_{\mathrm{dom}}(P_{(\mathfrak{t}_2,0)}(k) + \mathscr{F}_{\mathrm{dom}}(F))$$

because  $\mathfrak{t}_2 \in \mathfrak{L}_+$ . Then, we use the fact that

$$P_{(\mathfrak{t}_2,0)}(k) = k^2, \quad P_{(\mathfrak{t}_1,0)}(k) = -k^2, \quad P_{(\mathfrak{t}_1,1)}(k) = k^2$$

and

$$\mathscr{F}_{\text{dom}}(F) = \mathscr{F}_{\text{dom}}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k_{1}})) + \mathscr{F}_{\text{dom}}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{2}})) + \mathscr{F}_{\text{dom}}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{3}}))$$
  
$$= P_{(\mathfrak{t}_{2},1)}(k_{1}) + P_{(\mathfrak{t}_{2},0)}(k_{2}) + P_{(\mathfrak{t}_{2},0)}(k_{3})$$
  
$$= k_{1}^{2} - k_{2}^{2} - k_{3}^{2}.$$

Therefore,

$$\mathscr{F}_{\text{dom}}(T) = \mathcal{P}_{\text{dom}} \left( k^2 + k_1^2 - k_2^2 - k_3^2 \right)$$
$$= \mathcal{P}_{\text{dom}} \left( 2k_1^2 - 2k_1(k_2 + k_3) + 2k_2k_3 \right)$$
$$= 2k_1^2$$

One observes that the projection  $\mathcal{P}_{dom}$  projects to zero the cross terms  $k_i k_j$  with  $i \neq j$ .

A central novel idea which we introduce in our present work is that we proceed to interpolate the exponential of the lower part of the operator in place of a direct Taylor series expansion. The advantage of this procedure is firstly that it allows us to immediately arrive at stable schemes without the need for filter functions (the spectrum of  $i\mathcal{P}_{\text{low}} = i\mathcal{P} - i\mathcal{P}_{\text{dom}}$  typically lies on the imaginary axis so terms involving the exponential of the operator are all bounded). Secondly, through this interpolation process we are able to arrive at numerical schemes whose adjoint has the same functional form which is essential in the construction of symmetric methods. Classical Taylor expansion for the lower part gives:

$$e^{i\mathcal{L}_{\text{low}}} = \sum_{\ell \le r} \frac{\xi^{\ell}}{\ell!} (i\mathcal{L}_{\text{low}})^{\ell} + \mathcal{O}(\xi^{r+1} (i\mathcal{L}_{\text{low}})^{r+1})$$

Now, for reasons of stability, we would like to use a polynomial interpolation that will give the same local error analysis. We suppose given r + 1 distinct interpolation points  $0 \le a_0 < a_1 < \cdots < a_r \le 1$  associated to the polynomials  $p_{j,r}(\cdot, \tau)$  such that

$$p_{j,r}(a_m\tau,\tau)=\delta_{j,m}.$$

Then, we define the following approximation

$$\tilde{p}_{r}(f,\xi) = \sum_{j=0}^{r} f(a_{j}\tau) p_{j,r}(\xi,\tau), \quad f(a_{j}\tau) = e^{ia_{j}\tau\mathcal{L}_{low}}, j = 0, \dots, r,$$

where we have suppressed the implicit  $\tau$ -dependency of  $\tilde{p}_r(f,\xi)$  for notational simplicity. One has the following local error

$$f(\xi) - \tilde{p}_r(f,\xi) = \mathcal{O}\left(\prod_{j=0}^r (\xi - a_j\tau)(i\mathcal{L}_{\rm low})^{r+1}\right).$$
 (6.20)

In the sequel, we will write the polynomial interpolation as:

$$\tilde{p}_r(f,\xi) = \sum_{j=0}^r \hat{p}_{j,r}(f,\tau)\xi^j$$
(6.21)

where the  $\hat{p}_{j,\tau}(f,\tau)$  are bounded in  $\tau$  because they correspond to linear combinations of terms of the form  $\exp(ia_j\tau \mathcal{L}_{low})$ . We provide below one example with two points 0 and  $\tau$ 

$$\tilde{p}_1(f,\xi) = 1 + \frac{s}{\tau} \left( e^{is\mathcal{L}_{\text{low}}} - 1 \right), \tag{6.22}$$

and

$$\hat{p}_{0,1}(f,\tau) = 1, \quad \hat{p}_{1,1}(f,\tau) = \frac{e^{is\mathcal{L}_{\text{low}}} - 1}{\tau},$$
$$p_{0,1}(f,\xi) = \frac{\tau - s}{\tau}, \quad p_{1,1}(f,\xi) = \frac{s}{\tau}e^{is\mathcal{L}_{\text{low}}}$$

When r = 0 we can, for example, pick

$$p_0(f,\xi) = \hat{p}_{0,0}(f,\tau) = p_{0,0}(f,\xi) = f\left(\frac{\tau}{2}\right)$$

In practice, we will also consider

$$\hat{p}_{0,0}(f,\xi) = \frac{f(0) + f(\xi)}{2}.$$

The next definition is a slight modification of [BS22, Def. 3.1] where Taylor expansions around zero are replaced by an interpolation on the interval  $[0, \tau]$  and we take into account monomials in  $\tau$  for the discretisation.

**Definition 6.3.3** Assume that  $G: \xi \mapsto \tau^m \xi^q e^{i\xi P(k_1,...,k_n)}$  where P is a polynomial in the frequencies  $k_1, ..., k_n$  and let  $o_2 = (\mathfrak{t}_2, p) \in \mathfrak{L}_+ \times \{0, 1\}$  and  $r \in \mathbb{N}$ . Let k be a linear combination of  $k_1, ..., k_n$  using coefficients in  $\{-1, 0, 1\}$  and

$$\mathcal{L}_{\text{dom}} = \mathcal{P}_{\text{dom}}(P_{o_2}(k) + P), \quad \mathcal{L}_{\text{low}} = \mathcal{P}_{\text{low}}(P_{o_2}(k) + P)$$
$$f(\xi) = e^{i\xi\mathcal{L}_{\text{dom}}}, \quad g(\xi) = e^{i\xi\mathcal{L}_{\text{low}}}, \quad \tilde{g}(\xi) = e^{i\xi\left(P_{o_2}(k) + P\right)}.$$

Then, we define for  $n \in \mathbf{N}$ ,  $r \ge q$ ,  $\tilde{r} = r - q - m$  and  $\bar{n} = \deg(\mathcal{L}_{dom}^{r+1}) + \alpha$ 

$$\mathcal{K}_{o_2}^{k,r}(G,n)(s) = \begin{cases} -i|\nabla|^{\alpha}(k)\sum_{\ell\leq\tilde{r}}\hat{p}_{\ell,\tilde{r}}(\tilde{g},\tau)\int_0^s\tau^m\xi^{q+\ell}d\xi, \text{ if } n\geq\bar{n},\\ -i|\nabla|^{\alpha}(k)\sum_{\ell\leq\tilde{r}}\tau^m\hat{p}_{\ell,\tilde{r}}(g,\tau)\Psi_{n,q}^r(\mathcal{L}_{\mathrm{dom}},\ell)(s), & \text{otherwise.} \end{cases}$$
(6.23)

Thereby we set for  $(r - q - m - \ell + 1) \deg(\mathcal{L}_{dom}) + \ell \deg(\mathcal{L}_{low}) + \alpha > n$ 

$$\Psi_{n,q}^{r}(\mathcal{L}_{\text{dom}},\ell)(s) = \int_{0}^{s} \xi^{q+\ell} f(\xi) d\xi.$$
(6.24)

Otherwise,

$$\Psi_{n,q}^{r}(\mathcal{L}_{\text{dom}},\ell)(s) = \sum_{j \le \hat{r}} \hat{p}_{j,\hat{r}}(f,\tau) \int_{0}^{s} \xi^{q+\ell+j} d\xi.$$
(6.25)

Here  $\hat{r} = r - q - m - \ell$ , deg( $\mathcal{L}_{\text{dom}}$ ) and deg( $\mathcal{L}_{\text{low}}$ ) denote the degree of the polynomial  $\mathcal{L}_{\text{dom}}$  and  $\mathcal{L}_{\text{low}}$ , respectively and  $|\nabla|^{\alpha}(k) = \prod_{\alpha = \sum \gamma_j < \text{deg}(\mathcal{L})} k_j^{\gamma_j}$ . If r < q + m, the map  $\mathcal{K}_{o_2}^{k,r}(G, n)(s)$  is equal to zero.

We perform an example to illustrate the polynomial interpolation.

**Example 22** We consider  $P_{t_2}(\lambda) = -\lambda^2$ , p = 0,  $\alpha = 0$ ,  $k = -k_1 + k_2 + k_3$  and

$$G(\xi) = \xi e^{i\xi(k_1^2 - k_2^2 - k_3^2)}$$

With the notation of Definition (6.3.3) we observe that

$$\mathcal{L}_{
m dom} = 2k_1^2, \quad \mathcal{L}_{
m low} = -2k_1(k_2+k_3)+2k_2k_3,$$

Furthermore, we observe as  $\deg(\mathcal{L}_{dom}) = 2$ ,  $\deg(\mathcal{L}_{low}) = 1$  and q = 1 that

$$(r-q-\ell+1)\deg(\mathcal{L}_{dom})+\ell\deg(\mathcal{L}_{low})>n \quad \text{if} \quad 2r-n>\ell.$$
(6.26)

We consider the polynomial interpolation given in (6.22) and focus on some cases

— Case r = 1 and n = 1: We obtain

$$\begin{split} \mathcal{K}_{o_2}^{k,1}(G,n)(s) &= -i\hat{p}_{0,0}(f,\tau)\Psi_{n,1}^1(\mathcal{L}_{\text{dom}},0)(s) \\ &= -i\hat{p}_{0,0}(f,\tau)\int_0^s\xi f(\xi)d\xi \\ &= \frac{1}{2ik_1^2}\bigg(se^{2isk_1^2} - \frac{e^{2isk_1^2} - 1}{2ik_1^2}\bigg)\bigg(\frac{1+e^{is\mathcal{L}_{\text{low}}}}{2}\bigg) \end{split}$$

as condition (6.26) takes for  $\ell = 0$  the form 2 - n > 0. **Case** r = 2 and n = 2: We have that

$$\mathcal{K}^{k,2}_{\alpha}(G,n)(s) = -i\big(\hat{p}_{0,1}(g,\tau)\Psi^2_{n,1}(\mathcal{L}_{\rm dom},0)(s) + \hat{p}_{1,1}(g,\tau)\Psi^2_{n,1}(\mathcal{L}_{\rm dom},1)(s)\big)$$

and condition (6.26) takes the form  $4 - n > \ell$ . If  $\ell = 1$  we thus obtain

$$\Psi_{n,1}^2(\mathcal{L}_{\text{dom}},1)(s) = \int_0^s \xi^2 f(\xi) d\xi = \frac{s^2}{2ik_2^2} \Big( e^{2isk_1^2} - 2\Psi_{1,1}^1(\mathcal{L}_{\text{dom}},0) \Big).$$

If  $\ell = 0$ , on the other hand, condition (6.26) holds. Henceforth, we have that

$$\Psi_{n,1}^2(\mathcal{L}_{\text{dom}},0)(s) = \int_0^s \xi f(\xi) d\xi$$

Following a similar proof as for [BS22, Lem. 3.3] by using (6.20), one gets

**Lemma 6.3.4** We keep the notations of Definition (6.3.3). We suppose that  $q + m \leq r$  then one has for  $s \in [0, \tau]$ 

$$-i|\nabla|^{\alpha}(k)\int_{0}^{s}\tau^{m}\xi^{q}e^{i\xi(\mathcal{L}_{dom}+\mathcal{L}_{low})}d\xi - \mathcal{K}_{o_{2}}^{k,r}(G,n)(s) = \mathcal{O}(\tau^{r+2}k^{\bar{n}})$$
(6.27)

where  $\bar{n} = \max(n, \deg(\mathcal{L}_{low}^{r-q-m+1}) + \alpha).$ 

# 6.3.2 A forest formula for resonance based schemes

We recall the characters defined now on  $\mathcal{H}$  and parametrised by  $n \in \mathbf{N}$  where *n* here the a priori regularity assumed on the initial value, that is  $v \in H^n$  where  $H^n$  is the periodic Sobolev space of order *n*. These characters give a low regularity discretisation of some iterated integrals:

$$\Pi^{n} \left( F \cdot \bar{F} \right)(s,\tau) = (\Pi^{n} F)(s,\tau) (\Pi^{n} \bar{F})(s,\tau), \quad (\Pi^{n} \lambda^{\ell})(s,\tau) = s^{\ell_{1}} \tau^{\ell_{2}}, (\Pi^{n} \mathcal{I}_{o_{1}}^{r} (\lambda_{k}^{\ell} F))(s,\tau) = s^{\ell_{1}} \tau^{\ell_{2}} e^{isP_{o_{1}}(k)} (\Pi^{n} \mathcal{D}^{r-|\ell|}(F))(s,\tau), \left( \Pi^{n} \mathcal{I}_{o_{2}}^{r} (\lambda_{k}^{\ell} F) \right)(s,\tau) = \mathcal{K}_{o_{2}}^{k,r} \left( \Pi^{n} \left( \lambda^{\ell} \mathcal{D}^{r-|\ell|-1}(F) \right)(\cdot,\tau), n \right)(s).$$
(6.28)

where  $o_2 = (\mathfrak{t}_2, p_2)$  with  $\mathfrak{t}_2 \in \mathfrak{L}_+$  and  $o_1 = (\mathfrak{t}_1, p_1)$  with  $\mathfrak{t}_1 \in \mathfrak{L} \setminus \mathfrak{L}_+$  and  $\ell = (\ell_1, \ell_2) \in \mathbb{N}^2$ . We will use frequently the notations  $o_i$  in the sequel. The main difference with [BS22] is the use of the polynomial interpolation in Definition (6.3.3). In the next theorem, we state a forest formula for the resonance scheme in the sense that we exhibit a general formula for the terms  $(\Pi^{n,r}F)(t)$  where  $\Pi^{n,r}$  is short hand notation for  $\Pi^n \mathcal{D}_r$ . This new forest formula is a significant extension of contributions in [BS22] since it incorporates not just the aforementioned polynomial interpolants but also allows for implicit discretisations in the unknown v. We first need to introduce some notations that are needed for its formulation. We denote by  $\tilde{E}_T$  the edges of T associated to an integration in time. They carry a decoration of the type  $o_2$ 

$$\dot{E}_T = \{ e \in E_T \mid \mathfrak{e}(e) \in \mathfrak{L}_+ \times \{0, 1\} \}.$$

The notation  $T^e$  means that we consider the planted tree above the edge e in T. This tree has its root connected to the rest of its nodes by the edge e. By  $F_0 \cdot T_1 \dots \cdot T_m \subset F$ , we mean that the forest

$$F_0 \cdot T_1 \dots \cdot T_m \subset F \tag{6.29}$$

is a splitting of F where :

- The  $T_i$  are planted trees with the edge connecting the root decorated by an edge decoration of type  $o_2$ .
- $F_0$  is a forest either empty or taking the form:

$$F_0 = \prod_{j=1}^{m_0} T_{0,j}$$

where the  $T_{0,j}$  are subtrees at the root of some trees appearing in the decomposition of the forest F into product of planted trees.

**Example 23** We provide an example of the forests such that  $F_0 \cdot T_1 \dots \cdot T_m \subset F$ . Let us consider F to be the following decorated trees coming from the NLS equation



Because T starts with a brown edge that is an edge decorated by  $(t_1, 0)$ ,  $F_0$  is not empty. Below, we list all the possible splitting respecting this rule and also that the  $T_i$  with  $i \ge 1$  must be planted trees with a blue edge (decorated by  $(t_2, 0)$ ) at their root.



where  $\ell = -k_1 + k_2 + k_3$  and  $k = -k_4 + \ell + k_5$ . We also give below the decomposition for the decorated tree  $\overline{T}$ . Notice

that now  $\overline{T}_0$  could be empty.



We can obtain these terms by iterating a Butcher-Connes-Kreimer type copoduct  $\Delta_{BCK} : \hat{\mathcal{H}}_0 \to \hat{\mathcal{H}}_0 \otimes \hat{\mathcal{H}}_0$ , a simple version of the one introduced in [BS22]. It is defined recursively by

$$\Delta_{\mathrm{BCK}} \mathcal{I}_{o_1}(\lambda_k^{\ell} F) = \left( \mathcal{I}_{o_1}(\lambda_k^{\ell} \cdot) \otimes \mathrm{id} \right) \Delta_{\mathrm{BCK}} F,$$
  
$$\Delta_{\mathrm{BCK}} \mathcal{I}_{o_2}(\lambda_k^{\ell} F) = \left( \mathcal{I}_{o_2}(\lambda_k^{\ell} \cdot) \otimes \mathrm{id} \right) \Delta_{\mathrm{BCK}} F + \mathbf{1} \otimes \mathcal{I}_{o_2}(\lambda_k^{\ell} F)$$

and then extended multiplicatively for the forest product. Below, we provide some examples of computations:

$$\Delta_{\rm BCK} \stackrel{(k_2)}{\longrightarrow} = \stackrel{(k_2)}{\longrightarrow} \stackrel{(k_3)}{\longrightarrow} \stackrel{(k$$

Below, we introduce recursive maps  $\psi_{BCK}$  and  $\tilde{\psi}_{BCK}$  that can compute the splitting describe above with the coproduct  $\Delta_{BCK}$ :

$$\psi_{\rm BCK} = \left( {\rm id} \otimes \tilde{\psi}_{\rm BCK} \right) \Delta_{\rm BCK},$$
  
$$\tilde{\psi}_{\rm BCK} = \mathcal{M} \left( \tilde{\psi}_{\rm BCK} \otimes P_{\mathbf{1}} \right) \Delta_{\rm BCK}, \quad \tilde{\psi}_{\rm BCK}(\mathbf{1}) = \mathbf{1}$$
(6.30)

where  $\mathcal{M}$  is the forest product and  $P_1 = \mathrm{id} - \mathbf{1}^*$  is is the augmentation projector. Here  $\mathbf{1}^*$  is the co-unit which is non-zero and equal to one only on the empty forest. The projector  $P_1$  forces at least one cut at each iteration and therefore the recursion is well-defined. If we apply  $\psi$  to T, we obtain a linear combination of the terms of the form  $T_0 \otimes T_1 \cdot \ldots \cdot T_m$  that corresponds exactly to the splitting described above. We do not get a forest in the end but a term with a tensor product. This is for distinguishing the root as it is needed in our splitting. As an example of computation of those maps, one has

$$\tilde{\psi}(\overset{k_{2}}{\overset{k_{3}}{\overset{k}}{\overset{k_{3}}{\overset{k_{3}}}{\overset{k_{3}}{\overset{k_{3}}}{\overset{k}}{\overset{k_{3}}}{\overset{k_{3}}$$

**Theorem 6.3.5** For every forest  $F \in \hat{H}_0$ ,  $(\Pi^{n,r}F)(t,\tau)$  takes the form:

$$\sum_{\mathbf{a}\in[0,1]}\sum_{\bar{E}_{F}}\sum_{F_{0}\cdot T_{1}\ldots\cdot T_{m}\subset F}C_{F}e^{it\mathscr{F}_{dom}(F_{0})}\prod_{j=0}^{m}\prod_{e\in\bar{E}_{T_{j}}}e^{i\tau a_{e}}\mathscr{F}_{low}(T_{j}^{e})$$

$$\mathbf{b}_{\mathbf{a},F,F_{0}\cdot\ldots\cdot T_{m}}(t,\tau,i\tau\mathscr{F}_{dom}(T_{j}),i\tau\mathscr{F}_{dom}(T_{0,\bar{j}}))$$

$$(6.31)$$
with the convention  $T_0 = F_0$ . The coefficients  $C_F$  depend only on the node decorations of F. The coefficients b are polynomial in t and are non zero only for a finite number of values of  $\mathbf{a}$ . They are uniformly bounded in  $\tau$ . Moreover, they do not depend on the node decorations of the  $T_i$ ,  $F_0$  and F that correspond to the frequencies.

We recall that in the above notation the parameters n and r in  $(\Pi^{n,r}F)(t,\tau)$  denote the regularity requirements and maximum length of trees in the approximation respectively.

*Proof.* We proceed by induction on the size of the forest F. For the empty forest, the sum is equal to one by convention and

$$(\Pi^{n,r}\mathbf{1})(t,\tau) = 1.$$

Let  $F_1, F_2$  two decorated forests with  $F = F_1 \cdot F_2$  for which we have (6.31). We apply the induction hypothesis and get

$$(\Pi^{n,\gamma} F)(s,\tau) = (\Pi^{n,\gamma} F_1)(s,\tau)(\Pi^{n,\gamma} F_2)(s,\tau)$$

$$= \sum_{\mathbf{a}_1 \in [0,1]} \sum_{\tilde{E}_{F_1}} \sum_{\mathbf{a}_2 \in [0,1]} \sum_{\tilde{E}_{F_2}} \sum_{F_{1,0} \cdot T_{1,1} \dots \cdot T_{1,m_1} \subset F_1} \sum_{F_{2,0} \cdot T_{2,1} \dots \cdot T_{2,m_2} \subset F_2} C_{F_1} C_{F_2}$$

$$e^{it \left(\mathscr{F}_{\text{dom}}(F_{1,0}) + \mathscr{F}_{\text{dom}}(F_{2,0})\right)} \prod_{j=0}^{m_1} \prod_{e \in \tilde{E}_{T_{1,j}}} e^{i\tau a_{1,e} \cdot \mathscr{F}_{\text{low}}(T_{1,j}^e)} \times \prod_{j=0}^{m_2} \prod_{e \in \tilde{E}_{T_{2,j}}} e^{i\tau a_{2,e} \cdot \mathscr{F}_{\text{low}}(T_{2,j}^e)}$$

$$b_{\mathbf{a}_1,F_1,F_{1,0} \dots \cdot T_{1,m_1}} \times b_{\mathbf{a}_2,F_2,F_2,0 \dots \cdot T_{2,m_2}}.$$

By using Definition (6.3.2), we have

$$\mathscr{F}_{dom}(F_{1,0}) + \mathscr{F}_{dom}(F_{2,0}) = \mathscr{F}_{dom}(F_0), \quad F_0 = F_{1,0} \cdot F_{2,0}.$$

Then, we can perform the disjoint sum of  $\mathbf{a}_1$  and  $\mathbf{a}_1$ :

$$\mathbf{a} = \mathbf{a}_1 + \mathbf{a}_2$$

by extending  $\mathbf{a}_1$  (resp.  $\mathbf{a}_2$ ) on the edges of  $F_2$  (resp.  $F_1$ ) by zero. Then,  $\mathbf{a}$  is defined on the edges of F. We can gather the sum on the forests by:

$$\sum_{F_{1,0} \cdot T_{1,1} \dots \cdot T_{1,m_1} \subset F_1} \sum_{F_{2,0} \cdot T_{2,1} \dots \cdot T_{2,m_2} \subset F_2} = \sum_{F_0 \cdot T_{1,1} \dots \cdot T_{1,m_1} \cdot T_{2,1} \dots \cdot T_{2,m_2} \subset F}$$
$$= \sum_{F_0 \cdot T_1 \dots \cdot T_m \subset F}$$

and we can also set

$$\begin{aligned} b_{\mathbf{a},F,F_0\cdot\ldots\cdot T_1\cdot\ldots\cdot T_m} &= b_{\mathbf{a}_1,F_1,F_{1,0}\cdot\ldots\cdot T_{1,m_1}} \times b_{\mathbf{a}_2,F_2,F_{2,0}\cdot\ldots\cdot T_{2,m_2}} \\ C_F &= C_{F_1}\times C_{F_2} \end{aligned}$$

and see that the properties of the coefficient b are preserved by multiplication. Indeed, we have a bijection between partitions of  $F = F_1 \cdot F_2$  into a product of trees and the forest product of partitions of  $F_i$ . For a tree of the form  $\mathcal{I}_{o_1}(\lambda_k^{\ell}F)$ , one has

$$(\Pi^{n} \mathcal{I}_{o_{1}}^{r}(\lambda_{k}^{\ell} F))(t,\tau) = t^{\ell_{1}} \tau^{\ell_{2}} e^{it P_{o_{1}}(k)} (\Pi^{n,r-\ell} F)(t,\tau).$$

We multiply the formula for F obtained by the induction hypothesis by  $e^{itP_{o_1}(k)}$ :

$$\sum_{\mathbf{a}\in[0,1]\tilde{E}_{F}}\sum_{F_{0}\cdot T_{1}\ldots\cdot T_{m}\subset F}C_{F}e^{it(\mathscr{F}_{\mathrm{dom}}(F_{0})+P_{o_{1}}(k))}\prod_{j=0}^{m}\prod_{e\in\tilde{E}_{T_{j}}}e^{i\tau a_{e}\mathscr{F}_{\mathrm{low}}(T_{j}^{e})}$$
$$t^{\ell}b_{\mathbf{a},F,F_{0}}\ldots\cdot T_{m}(t,\tau,i\tau\mathscr{F}_{\mathrm{dom}}(T_{j}),i\tau\mathscr{F}_{\mathrm{dom}}(T_{0,\bar{j}})).$$

From Definition (6.3.2), we have

$$\mathscr{F}_{\operatorname{dom}}(F_0) + P_{o_1}(k) = \mathscr{F}_{\operatorname{dom}}(\mathcal{I}_{o_1}(\lambda_k^{\ell} F_0)).$$

Moreover, for  $\mathcal{I}_{o_1}(\lambda_k^{\ell}F)$ , the forest at the root must be of the form  $\mathcal{I}_{o_1}(\lambda_k^{\ell}F_0)$ . For the coefficients b, we have

$$t^{\ell_1}\tau^{\ell_2}b_{\mathbf{a},F,F_0\cdot\ldots\cdot T_m} = b_{\mathbf{a},\mathcal{I}_{o_1}(\lambda_k^{\ell}F),\mathcal{I}_{o_1}(\lambda_k^{\ell}F_0)\cdot\ldots\cdot T_m}, \quad C_{\mathcal{I}_{o_1}(\lambda_k^{\ell}F)} = C_F$$

It remains to prove the forest formula for a tree of the form  $\mathcal{I}_{o_2}(\lambda_k^{\ell} F)$ . We have

$$\left(\Pi^{n}\mathcal{I}_{o_{2}}^{r}(\lambda_{k}^{\ell}F)\right)(t,\tau) = \mathcal{K}_{o_{2}}^{k,r}\left(\Pi^{n}\left(\lambda^{\ell}\mathcal{D}^{r-|\ell|-1}(F)\right)(\cdot,\tau),n\right)(t).$$

We apply the induction hypothesis on  $\Pi^{n,r-|\ell|-1}$ . Then, the proof boils down to understand how the operator  $\mathcal{K}_{o_2}^{k,r}$  acts on the forest formula. This operator computes first the dominant part of the oscillation. It is given for a fixed forest  $F_0 \cdot T_1 \dots \cdot T_m \subset F$  by

$$\mathcal{P}_{\mathrm{dom}}(P_{o_2}(k) + \mathscr{F}_{\mathrm{dom}}(F_0)) = \mathscr{F}_{\mathrm{dom}}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0)).$$

If this dominant part is integrated exactly, we obtain a factor of the form

$$e^{it\mathscr{F}_{\operatorname{dom}}(\mathcal{I}_{o_2}(\lambda_k^{\ell}F_0))}$$

This will correspond to forests  $\mathcal{I}_{o_2}(\lambda_k^{\ell}F_0) \cdot \ldots \cdot T_m$ . In this exact integration, we have terms without this factor which corresponds to the forest  $\mathbf{1} \cdot \mathcal{I}_{o_2}(\lambda_k^{\ell}F_0) \cdot \ldots \cdot T_m$  as the forest connected to the root could be the empty forest. For the lower part given by

$$(\mathrm{id} - \mathcal{P}_{\mathrm{dom}})(P_{o_2}(k) + \mathscr{F}_{\mathrm{dom}}(F)) = \mathscr{F}_{\mathrm{low}}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0))$$

we perform an interpolation that produces terms of the form

$$e^{i\tau a \mathscr{F}_{\text{low}}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0)))}, \quad a \in [0, 1].$$

The operator  $\mathcal{K}_{o_2}^{k,r}$  depends on *n* which is the a priori regularity assumed on the initial data. With this information, if *n* is sufficiently big, we can perform a full Taylor expansion via an interpolation that will produce terms of the form:

$$e^{i\tau a \mathscr{F}_{\text{low}}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0)))} e^{i\tau a' \mathscr{F}_{\text{dom}}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0)))}, \quad a, a' \in [0, 1]$$

These terms will be associated to a forest of the form  $\mathbf{1} \cdot \mathcal{I}_{o_2}(\lambda_k^{\ell} F_0) \cdot \ldots \cdot T_m$ . The factor  $e^{i\tau a' \mathscr{F}_{dom}(\mathcal{I}_{o_2}(\lambda_k^{\ell} F_0)))}$  will be inside the coefficients b. The interpolation also produces monomials in t and  $\tau$  which implies the polynomial structure of the coefficients b in t. Also, it produces coefficients bounded in  $\tau$  such as the  $\hat{p}_{j,r}(f,\tau)$  given in (6.21). The choice of the  $a \in [0,1]$  are fixed by the interpolation method and one uses only a finite number of them which implies that the coefficients b are non-zero on a finite set of the  $\mathbf{a}$ . Finally, we have

$$C_{\mathcal{I}_{o_2}(\lambda_k^{\ell}F)} = -i|\nabla|^{\alpha}(k)C_F.$$

**Theorem 6.3.6** For every decorated tree  $T = \mathcal{I}_{o_2}(\lambda_k F)$ ,  $(\Pi^{n,r}T)(\tau,\tau)$  takes the form:

$$\sum_{\in [0,1]} \sum_{\tilde{E}_F} \sum_{T_0 \cdot T_1 \dots \cdot T_m \subset T} C_T \prod_{j=0}^m \prod_{e \in \tilde{E}_{T_j}} e^{i\tau a_e \mathscr{F}_{low}(T_j^e)} b_{\mathbf{a},T,T_0 \cdot \dots \cdot T_m}(\tau, i\tau \mathscr{F}_{dom}(T_j))$$
(6.32)

where the coefficients b are polynomial in  $\tau$  with bounded coefficient in  $\tau$  and are non zero for finite values of **a**. Moreover, they do not depend on the nodes decorations of the  $T_i$ ,  $F_0$  and F that correspond to the frequencies.

*Proof.* The proof works mostly in the same way as for Theorem (6.3.5) but with  $t = \tau$ . The main difference is that when we apply the operator  $\mathcal{K}_{o_2}^{k,r}$ , we put factors of the form  $e^{it\mathscr{F}_{dom}(\mathcal{I}_{o_2}(\lambda_k F_0))}$  in the coefficients b.

## 6.3.3 Midpoint general resonance based schemes

a

In this section, we introduce new resonance based schemes where we iterate Duhamel's formula in slightly different manner. The iteration chosen follows the mid point rule. These schemes turn out to be a subclass of the general forest formula (6.45). In this subclass it is also possible to work with a fairly general framework more closely aligned with [BS22] which allows for an automatic handle on the local error of these schemes. In particular, in order to incorporate the midpoint iterations and subsequent lower part interpolations we need to add more edge decorations on the trees

representing the iterated integrals. For example, we can consider decorated trees of the form  $T_{\mathfrak{c},\chi}^{\mathfrak{n},\mathfrak{o}} = (T,\mathfrak{n},\mathfrak{o},\mathfrak{c},\chi)$  where  $\chi: E_T \to \mathfrak{D}$ . Here  $\mathfrak{D}$  is a finite set and encodes the following information for an edge  $e \in N_T$  decorated by  $(\mathfrak{t}, p)$ :

- For  $\mathfrak{t} \notin \mathfrak{L}_+$ , the edge *e* is associated with a term of the form  $e^{iP(\mathfrak{t},p)(k_v)}$ . Then,  $\chi(e)$  corresponds of a way of iterating Duhamel's formula not only using the leftmost point of the interval but a weighted sum of iterations on various points in  $[0, \tau]$ .
- For  $\mathfrak{t} \in \mathfrak{L}_+$ , the edge *e* is associated with a term of the form  $\int_{a\tau}^t e^{iP_{(\mathfrak{t},p)}(k_v)} \cdots ds$  where  $a \in [0,1]$  and it corresponds to a different Duhamel's iteration. Now,  $\chi(e)$  gives a choice of a polynomial interpolation for the lower part of the resonance in the discretisation.

For reasons of presentation we focus on the midpoint rule, however, we could also follow other types of Duhamel iterations. Moreover, in this subclass we take the polynomial interpolation to be fixed.

To illustrate the central idea of a different way of iterating Duhamel's formula, let us first consider the Nonlinear Schrödinger equation. The usual iteration is given by

$$u(t_{n}+s) = e^{i\tau\Delta}u(t_{n}) - ie^{i\tau\Delta} \int_{0}^{\tau} e^{-is\Delta} (|u(t_{n}+s)|^{2}u(t_{n}+s)) ds.$$

In Fourier space, we obtain

$$u_{k}(t_{n}+\tau) = e^{-i\tau k^{2}} u_{k}(t_{n})$$

$$-ie^{-i\tau k^{2}} \sum_{k=-k_{1}+k_{2}+k_{3}} \int_{0}^{\tau} e^{isk^{2}} \overline{\hat{u}_{k_{1}}(t_{n}+s)} u_{k_{2}}(t_{n}+s) u_{k_{3}}(t_{n}+s) ds.$$
(6.33)

We can now choose to iterate this expression using two possible ways:

$$u_{k}(t_{n}+s) = e^{-isk^{2}}u_{k}(t_{n})$$

$$-ie^{-isk^{2}}\sum_{k=-k_{1}+k_{2}+k_{3}}\int_{0}^{s}e^{i\tilde{s}k^{2}}\overline{u_{k_{1}}(t_{n}+\tilde{s})}u_{k_{2}}(t_{n}+\tilde{s})u_{k_{3}}(t_{n}+\tilde{s})d\tilde{s}.$$
(6.34)

and

$$u_{k}(t_{n}+s) = e^{-i(s-\tau)k^{2}}u_{k}(t_{n}+\tau)$$

$$-ie^{-isk^{2}}\sum_{k=-k_{1}+k_{2}+k_{3}}\int_{\tau}^{s}e^{i\tilde{s}k^{2}}\overline{u_{k_{1}}(t_{n}+\tilde{s})}u_{k_{2}}(t_{n}+\tilde{s})u_{k_{3}}(t_{n}+\tilde{s})d\tilde{s}.$$
(6.35)

The iteration (6.34) corresponds to the left end point of the interval  $[0, \tau]$  while (6.35) is the right end point. We now have a choice over each term in (6.33) if we want the iteration of Duhamel's formula to begin with (6.34) or (6.35). The average of the two iterations gives the midpoint rule.

There are quite a lot of degrees of freedom as one can choose various linear combinations of Duhamel's formulae in different points. Let us mention that the tree structure is not modified if one changes the iteration but the definition of  $\Pi$  has to reflect this new formulation. The midpoint rule oscillatory integrals are given by

$$\Pi_{\rm mid}(F \cdot \bar{F})(s,\tau) = (\Pi_{\rm mid}F)(s,\tau)(\Pi_{\rm mid}\bar{F})(s,\tau),$$

$$(\Pi_{\rm mid}\mathcal{I}_{o_1}(\lambda_k F))(s,\tau) = \frac{1}{2}e^{isP_{o_1}(k)}((\Pi_{\rm mid,1}F)(s,\tau) + (\Pi_{\rm mid,2}F)(s,\tau)),$$

$$(\Pi_{\rm mid,1}\mathcal{I}_{o_2}(\lambda_k F))(s,\tau) = -i|\nabla|^{\alpha}(k)\int_{\tau}^{s}e^{i\xi P_{o_2}(k)}(\Pi_{\rm mid}F)(\xi,\tau)d\xi,$$

$$(\Pi_{\rm mid,2}\mathcal{I}_{o_2}(\lambda_k F))(s,\tau) = -i|\nabla|^{\alpha}(k)\int_{0}^{s}e^{i\xi P_{o_2}(k)}(\Pi_{\rm mid}F)(\xi,\tau)d\xi.$$
(6.36)

We notice that, in this definition, we have to keep track of the time step  $\tau$  in order to remember the interval  $[0, \tau]$ . In this definition, we have assumed that the Duhamel iteration corresponds to edges decorated by  $\mathfrak{t}_1$ . Moreover, we have supposed that F is not empty for  $(\prod_{\mathrm{mid}} \mathcal{I}_{o_1}(\lambda_k F))(s, \tau)$  and  $s \neq \tau$ . If F is empty, we set

$$(\Pi_{\mathrm{mid}}\mathcal{I}_{o_1}(\lambda_k \mathbf{1}))(s,\tau) = e^{isP_{o_1}(k)}.$$
(6.37)

If  $\tau = s$ , we set

$$(\Pi_{\mathrm{mid}}\mathcal{I}_{o_1}(\lambda_k F))(\tau,\tau) = (\Pi_{\mathrm{mid}}\mathcal{I}_{o_1}(\lambda_k F))(\tau) = e^{i\tau P_{o_1}(k)}(\Pi_{\mathrm{mid},2}F)(\tau,\tau).$$
(6.38)

The last two specific cases are necessary for building up the scheme. Indeed, (6.37) corresponds to the leaves of our trees or when we terminate on an initial data  $u_{k_i}$ . Here, we will apply the midpoint rule in the sequel (see (6.41) in the definition of  $\Upsilon^p_{mid}(T)(v,\tau)$ )

$$\frac{1}{2}u_{k_j}(0) + e^{-i\tau P_{o_1}(k_j)} \frac{1}{2}u_{k_j}(\tau).$$

The second condition (6.38) corresponds to the fact that the first is not the midpoint rule approximation as we do not need to perform it as  $\tau = s$ .

**Remark 6.3.7** This approach also works for the more general scheme given in [ABBS22b]. The main difference is that now  $\Upsilon^p_{\text{mid}}(T)(v,\tau)$  defined in the sequel is part of the definition of  $\Pi_{\text{mid}}$ .

The scheme  $\Pi_{\text{mid}}^{n,r}$  is defined as the same as for  $\Pi_{\text{mid}}$  but now we discretise the time integrals:

$$\left(\Pi_{\mathrm{mid},j}^{n,r}\mathcal{I}_{o_2}(\lambda_k F)\right)(s,\tau) = \mathcal{K}_{o_2,j}^{k,r}(\Pi_{\mathrm{mid}}^{n,r}(F)(\cdot,\tau),n)(s,\tau), \quad j \in \{1,2\}$$

where the map  $\mathcal{K}_{o_2,1}^{k,r}(\cdot)(s,\tau)$  uses the exact integration  $\int_s^{\tau} ...d\xi$  and  $\mathcal{K}_{o_2,2}^{k,r}$  the one given by  $\int_0^s ...d\xi$ . We first introduce some notations:

**Definition 2** — For a decorated tree  $T_{\mathfrak{e}} = (T, \mathfrak{e})$  with only edge decorations, we define the symmetry factor  $S(T_{\mathfrak{e}})$  inductively by  $S(\mathbf{1}) = 1$ , while if T is of the form

$$\prod_{i,j} \mathcal{I}_{(\mathfrak{t}_{t_i},p_i)}(T_{i,j})^{\beta_{i,j}}$$

with  $T_{i,j} \neq T_{i,\ell}$  for  $j \neq \ell$ , then

$$S(T) := \left(\prod_{i,j} S(T_{i,j})^{\beta_{i,j}} \beta_{i,j}!\right).$$
(6.39)

We extend this definition to any tree  $T_{\mathfrak{c}}^{\mathfrak{n},\mathfrak{o}}$  in  $\mathcal{T}$  by setting:

$$S(T_{\mathfrak{e}}^{\mathfrak{n},\mathfrak{o}}) := S(T_{\mathfrak{e}}).$$

Let us stress that the symmetric factor depends only on the edges decorations but not on the nodes decorations given by the frequencies.

— Then, we define the map  $\Upsilon^p_{mid}(T)(v,\tau)$  for

$$T = \mathcal{I}_{(\mathfrak{t}_1,a)} \left( \lambda_k \mathcal{I}_{(\mathfrak{t}_2,a)}(\lambda_k \prod_{i=1}^n \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{k_i} T_i) \prod_{j=1}^m \mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_{\tilde{k}_j} \tilde{T}_j)) \right), \quad a \in \{0,1\}$$

by

$$\Upsilon^{p}_{mid}(T)(v,\tau) := \partial^{n}_{v} \partial^{m}_{\bar{v}} p_{a}(v,\bar{v}) \prod_{i=1}^{n} \Upsilon^{p}_{mid}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{i}}T_{i}))(v,\tau)$$

$$\prod_{j=1}^{m} \Upsilon^{p}_{mid}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{\bar{k}_{j}}\tilde{T}_{j}))(v,\tau)$$
(6.40)

and

$$\Upsilon^{p}_{mid}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}))(v,\tau) := \frac{1}{2}v_{k}(0) + \frac{1}{2}e^{-iP_{o_{1}}(k)\tau}v_{k}(\tau) 
\Upsilon^{p}_{mid}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k}))(v,\tau) := \frac{1}{2}\bar{v}_{k}(0) + \frac{1}{2}e^{iP_{o_{1}}(k)\tau}\bar{v}_{k}(\tau).$$
(6.41)

Above, we have used the notation:

 $p_0(v, \overline{v}) = p(v, \overline{v}), \quad p_1(v, \overline{v}) = \overline{p(v, \overline{v})}$ 

In the sequel, we will use the following short hand notation:

$$\overline{\Upsilon^p_{mid}(T)}(v,\tau) = \overline{\Upsilon^p_{mid}}(T)(v,\tau).$$

— We set

$$\begin{aligned} \hat{\mathcal{T}}_{0}(R) &= \{ \mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}\mathcal{I}_{(\mathfrak{t}_{2},0)}(\lambda_{k}\prod_{i=1}^{N}T_{i}\prod_{j=1}^{M}\tilde{T}_{j})), \mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}) \\ T_{i} &\in \hat{\mathcal{T}}_{0}(R), \ \tilde{T}_{j} \in \bar{\hat{\mathcal{T}}}_{0}(R), \ k \in \mathbf{Z}^{d} \} \\ \bar{\hat{\mathcal{T}}}_{0}(R) &= \{ \mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k}\mathcal{I}_{(\mathfrak{t}_{2},1)}(\lambda_{k}\prod_{i=1}^{N}T_{i}\prod_{j=1}^{M}\tilde{T}_{j})), \mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k}) \\ T_{i} &\in \bar{\hat{\mathcal{T}}}_{0}(R), \ \tilde{T}_{j} \in \hat{\mathcal{T}}_{0}(R), \ k \in \mathbf{Z}^{d} \} \\ \hat{\mathcal{T}}_{2}(R) &= \{ \mathcal{I}_{(\mathfrak{t}_{2},0)}(\lambda_{k}\prod_{i=1}^{N}T_{i}\prod_{j=1}^{M}\tilde{T}_{j}), \ T_{i} \in \hat{\mathcal{T}}_{0}(R), \ \tilde{T}_{j} \in \bar{\hat{\mathcal{T}}}_{0}(R), \ k \in \mathbf{Z}^{d} \} \end{aligned}$$

For a fixed  $k \in \mathbf{Z}^d$ , we denote the set  $\hat{\mathcal{T}}_0^k(R)$  (resp.  $\overline{\hat{\mathcal{T}}}_0^k(R)$  and  $\hat{\mathcal{T}}_2^k(R)$ ) as the subset of  $\hat{\mathcal{T}}_0(R)$  (resp.  $\overline{\hat{\mathcal{T}}}_0(R)$  and  $\overline{\hat{\mathcal{T}}}_2^k(R)$ ) whose decorated trees have decorations on the node connected to the root given by k. For  $r \in \mathbf{Z}$ ,  $r \geq -1$ , we set:

$$\hat{\mathcal{T}}_0^{r,k}(R) = \{T_{\mathfrak{e}}^{\mathfrak{o}} \in \hat{\mathcal{T}}_0^k(R), n_+(T_{\mathfrak{e}}^{\mathfrak{o}}) \le r+1\}.$$

In the previous space, we disregard iterated integrals which have more than r + 1 integrals and will be of order  $\mathcal{O}(\tau^{r+2})$ . The set  $\hat{\mathcal{T}}_2^{r,k}(R)$  is defined as the same from  $\hat{\mathcal{T}}_2^k(R)$ . In the sequel, we will use the short hand notation for  $T \in \hat{\mathcal{T}}_2^k(R)$ :

$$\Upsilon^p_{mid}(T)(v,\tau) = \Upsilon^p_{mid}(\mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_k T))(v,\tau).$$

This truncation leads exactly to the current local error behaviour as shown in the following proposition which forms the basis of our local error analysis in Theorem (6.3.15).

Proposition 6.3.8 The tree series given by

$$U_{mid,k}^{r}(\tau,v) = \sum_{T \in \hat{\tau}_{0}^{r,k}(R)} \frac{\Upsilon_{mid}^{p}(T)(v,\tau)}{S(T)} (\Pi_{mid}T)(\tau)$$
(6.42)

where  $o_1 = (\mathfrak{t}_1, 0)$ , is the k-th Fourier coefficient of a solution of (6.33) with the midpoint rule expansion up to order r + 1. Proof. The proof follows the same lines as the one given in [BS22, Prop. 4.3].

We are now able to define the main resonance based scheme:

Definition 6.3.9 The midpoint resonance based scheme is given by:

$$U_{\text{mid},k}^{n,r}(\tau,v) = \sum_{T \in \hat{\tau}_0^{r,k}(R)} \frac{\Upsilon_{\text{mid}}^p(T)(v,\tau)}{S(T)} (\Pi_{\text{mid}}^{n,r}T)(\tau)$$
(6.43)

It is obtained by replacing the character  $\Pi_{\text{mid}}^r$  by  $\Pi_{\text{mid}}^{n,r}$  in (6.42).

The new scheme (6.43) can be described by the same type of forest formula introduced before.

**Proposition 6.3.10** For every forest F,  $(\prod_{mid}^{n,r}F)(t,\tau)$  is of the form of (6.31). For every decorated tree  $T = \mathcal{I}_{o_1}(\lambda_k \mathcal{I}_{o_2}(\lambda_k F))$ ,  $(\prod_{mid}^{n,r}T)(\tau)$  is of the form (6.32).

*Proof.* The proof follow by induction as in Theorem-(6.3.5) and Theorem (6.3.6).

Before stating our main result connecting the midpoint resonance based schemes to our earlier forest formula, we need to introduce a new map

$$\Upsilon^p_{\chi}(T)(u^{n+\chi_v}_{k_v}, v \in L_T, \tau)$$

defined as the same as  $\Upsilon^p_{\mathrm{mid}}(T)(v,\tau)$  except that for the leaves we use

$$e^{-i\chi_u \tau P_{o_{e_u}}(k_u)} u_{k_u}^{n+\chi_u}, (6.44)$$

where  $e_u$  is the outgoing edge of u in T and  $o_{e_u}$  corresponds to the edge decoration of  $e_u$ . The map  $\Upsilon^p_{\chi}(T)$  allows to parametrise implicit schemes as the scheme given by the midpoint rule.

**Theorem 6.3.11** The low regularity scheme  $U_{mid,k}^{n,r}$  is of the form:

$$u_{k}^{n+1} = e^{i\tau P_{o_{1}}(k)}u_{k}^{n} + e^{i\tau P_{o_{1}}(k)} \sum_{T \in \hat{\tau}_{2}^{\tau,k}(R)} \sum_{\mathbf{a} \in [0,1]^{\tilde{E}_{F}}} \sum_{\chi \in \{0,1\}^{L_{T}}} \sum_{T_{0} \cdot T_{1} \dots \cdot T_{m} \subset T} \sum_{C_{T} b_{\mathbf{a},\chi,T,T_{0}} \dots \cdot T_{m}} (\tau, i\tau \mathscr{F}_{dom}(T_{j}), j \in \{0, \dots, m\})$$

$$\prod_{e \in \tilde{E}_{T_{j}}} e^{i\tau a_{e}} \mathscr{F}_{low}(T_{j}^{e})} \frac{\Upsilon_{\chi}^{p}(T)(u_{k_{v}}^{n+\chi_{v}}, v \in L_{T}, \tau)}{S(T)},$$
(6.45)

where  $\hat{\mathcal{T}}_2^{r,k}(R)$  was introduced in Definition (2).

*Proof.* First, we notice that

$$U_{\text{mid},k}^{n,r}(\tau,v) = e^{i\tau P_{o_1}(k)} u_k^n + e^{i\tau P_{o_1}(k)} \sum_{T \in \hat{\tau}_2^{r,k}(R)} \frac{\Upsilon_{\text{mid}}^p(T)(v,\tau)}{S(T)} (\Pi_{\text{mid}}^{n,r}T)(\tau).$$

Then, the result is just a consequence of Theorem (6.3.6) applied to each of the  $(\Pi_{\text{mid}}^{n,r}T)(\tau)$ . Indeed, one multiplies the coefficients for a decorated trees (6.32) with  $\Upsilon_{\text{mid}}^{p}(T)(v,\tau)$ .

For the local error, we can adapt [BS22, Def. 3.11].

**Definition 6.3.12** Let  $n \in \mathbf{N}$ ,  $r \in \mathbf{Z}$ . We recursively define  $\mathcal{L}_{low}^{r}(\cdot, n)$  as

$$\mathcal{L}^r_{\text{low}}(F,n) = 1, \quad r < 0.$$

Else, when  $r \ge 0$ , we let:

$$\mathcal{L}_{\text{low}}^{r}(\mathbf{1},n) = 1, \quad \mathcal{L}_{\text{low}}^{r}(F \cdot \bar{F},n) = \mathcal{L}_{\text{low}}^{r}(F,n) + \mathcal{L}_{\text{low}}^{r}(\bar{F},n)$$
$$\mathcal{L}_{\text{low}}^{r}(\mathcal{I}_{o_{1}}(\lambda_{k}^{\ell}F),n) = \mathcal{L}_{\text{low}}^{r-|\ell|}(F,n)$$
$$\mathcal{L}_{\text{low}}^{r}(\mathcal{I}_{o_{2}}(\lambda_{k}^{\ell}F),n) = k^{\alpha}\mathcal{L}_{\text{low}}^{r-|\ell|-1}(F,n) + \mathbf{1}_{\{r-|\ell| \ge 0\}} \sum_{i} k^{\bar{n}_{i}}$$

where

$$\bar{n}_{j} = \max_{m} \left( n, \deg \left( P_{(F_{j}^{(1)}, F_{j}^{(2)}, m)} \mathscr{F}_{\text{low}}(\mathcal{I}_{(\mathfrak{t}_{2}, p)}(\lambda_{k}^{\ell} F_{j}^{(1)}))^{r-|\ell|+1-m} + \alpha \right) \right)$$

with

$$\Delta \mathcal{D}^{r-|\ell|-1}(F) = \sum_{j} F_{j}^{(1)} \otimes F_{j}^{(2)},$$
$$A^{n}(F_{j}^{(2)})B^{n}\left(F_{j}^{(1)}\right)(\xi,\tau) = \sum_{|m| \le r-|\ell|-1} \frac{P_{(F_{j}^{(1)},F_{j}^{(2)},m)}}{Q_{(F_{j}^{(1)},F_{j}^{(2)},m)}} \xi^{m_{1}}\tau^{m_{2}}$$

and  $\mathscr{F}_{low}$  is defined in Definition (6.3.2).

**Remark 6.3.13** The main difference between the definition above and [BS22, Def. 3.11] is the fact that we deal with monomials of the form  $s^{m_1}\tau^{m_2}$  due to the fact that  $\tau$  appears in the exact integrations. These modifications are minor from the original structure as the formalism is robust from moving from decorations on the edges in N to N<sup>2</sup>.

With the previous definition, one is able to give the local error of the approximations of the oscillatory integrals and for the schemes. The proofs are exactly the same as in [BS22, Section 3.3].

**Theorem 6.3.14** For every  $T \in \mathcal{T}$  one has,

$$(\Pi_{mid}T - \Pi_{mid}^{n,r}T)(\tau) = \mathcal{O}\big(\tau^{r+2}\mathcal{L}_{low}^r(T,n)\big)$$

The numerical scheme (6.42) approximates the exact solution locally up to order r + 2. More precisely, the following Theorem holds:

**Theorem 6.3.15 (Local error)** The numerical scheme (6.42) with initial value v = u(0) approximates the exact solution  $U_k(\tau, v)$  up to a local error of type

$$U_{mid,k}^{n,r}(\tau,v) - U_k(\tau,v) = \sum_{T \in \tilde{\tau}_0^{r,k}(R)} \mathcal{O}\left(\tau^{r+2} \mathcal{L}_{low}^r(T,n) \Upsilon_{mid}^p(\lambda_k T)(v,\tau)\right)$$

where the operator  $\mathcal{L}_{low}^{r}(T,n)$ , given in Definition (6.3.12), embeds the necessary regularity of the solution.

**Remark 6.3.16** The local error of the resonance based low regularity schemes does not depend on the choice of the polynomial interpolation and the iteration of Duhamel's formula but only on the structure of the resonances.

**Remark 6.3.17** As in [BS22, Prop. 3.18], one can always map back to physical space the scheme  $U_{\text{mid},k}^{n,r}$ . This due to the fact that the structure of the resonances and their exact integration is the same in this context.

# 6.4 Symmetric schemes

Having introduced the general forest formula (6.31) and the general subclass of midpoint general resonance based schemes, we now seek to answer the central question of this manuscript: "Which schemes in these classes are symmetric in the sense of definition (1)?" For this we take two routes: Firstly, for the subclass of midpoint general resonance based schemes it turns out that symmetry of the interpolation nodes is sufficient for the symmetry of the schemes. Secondly, for schemes captured by the forest formula (6.31) we can study the form of their adjoint method and find conditions on the coefficients of these schemes under which the methods are symmetric. We recall the adjoint method of a numerical scheme  $v^{n+1} = \Phi_{\tau}v^n$  is defined by  $\hat{\Phi}_{\tau} := \Phi_{-\tau}^{-1}$  and the method is said to be symmetric if  $\hat{\Phi}_{\tau} = \Phi_{\tau}$ . We can find the adjoint method of a scheme simply by the operations  $n \leftrightarrow n+1$  and  $\tau \leftrightarrow -\tau$ . The swapping of n and n+1 corresponds in our case to changing  $v_k(0)$  into  $v_k(\tau)$ . We define  $\tilde{\Upsilon}^p_{mid}(T)(v,\tau)$  as the same as  $\Upsilon^p_{mid}(T)(v,\tau)$  exept that we exchange  $v_k(0)$  and  $v_k(\tau)$  in the definition:

$$\begin{split} \tilde{\Upsilon}^{p}_{\mathrm{mid}}(\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}))(v,\tau) &:= \frac{1}{2}v_{k}(\tau) + \frac{1}{2}e^{iP_{o_{1}}(k)\tau}v_{k}(0)\\ \tilde{\Upsilon}^{p}_{\mathrm{mid}}(\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k}))(v,\tau) &:= \frac{1}{2}\bar{v}_{k}(\tau) + \frac{1}{2}e^{-iP_{o_{1}}(k)\tau}\bar{v}_{k}(0). \end{split}$$

## 6.4.1 Symmetric interpolation

We prove in the next proposition that the Duhamel's midpoint iteration truncated up to order r + 1 gives a symmetric scheme. The proof uses the recursive construction of the iterated integrals.

**Proposition 6.4.1** The scheme defined by (6.42) is symmetric.

*Proof.* We first observe that the scheme is given by

$$u_{k}(\tau) = e^{i\tau P_{o_{1}}(k)} u_{k}(0) + \sum_{T \in \hat{\tau}_{0}^{r,k}(R) \setminus \{\mathcal{I}_{(t_{1},0)}(\lambda_{k}\mathbf{1})\}} \frac{\Upsilon_{\mathrm{mid}}^{p}(T)(u,\tau)}{S(T)} (\Pi_{\mathrm{mid}}T)(\tau)$$

Now we swap n and n+1, and we also send  $\tau$  onto  $-\tau$ , we obtain

$$u_{k}(\tau) = e^{i\tau P_{o_{1}}(k)} u_{k}(0) - e^{i\tau P_{o_{1}}(k)} \sum_{T \in \hat{\tau}_{0}^{r,k}(R) \setminus \{\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}\mathbf{1})\}} \frac{\Upsilon_{\mathrm{mid}}^{p}(T)(u,-\tau)}{S(T)} (\Pi_{\mathrm{mid}}T)(-\tau).$$

Then, one has to show that two sums coincide for proving that the scheme is symmetric. We prove that this is the case for each term of the sum namely, one has:

$$-e^{-i\tau P_{o_1}(k)}\frac{\tilde{\Upsilon}_{\rm mid}^p(T)(u,-\tau)}{S(T)}(\Pi_{\rm mid}T)(-\tau) = \frac{\Upsilon_{\rm mid}^p(T)(u,\tau)}{S(T)}(\Pi_{\rm mid}T)(\tau).$$
(6.46)

We proceed by induction on the construction of the trees for showing (6.46). Decorated trees in  $\hat{\mathcal{T}}_{0}^{r,k}(R) \setminus \{\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k}\mathbf{1})\}$  are necessarily of the form

$$T = \mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_k \mathcal{I}_{(\mathfrak{t}_2,0)}(\lambda_k F)).$$

We notice that

$$(\Pi_{\rm mid}T)(-\tau) = -i|\nabla|^{\alpha}(k)e^{i\tau P_{(t_1,0)}(k)} \int_0^{-\tau} e^{-isP_{(t_1,0)}(k)}(\Pi F)_{\rm mid}(s,-\tau)ds$$

By performing the change of variable  $s = s + \tau$ , one gets

$$(\Pi_{\rm mid}T)(-\tau) = i|\nabla|^{\alpha}(k) \int_0^{\tau} e^{isP_{(t_1,0)}(k)} (\Pi_{\rm mid}F)(s-\tau,-\tau)ds.$$

It remains to show that

$$\frac{\widetilde{\Upsilon}^p_{\mathrm{mid}}(T_j)(u,-\tau)}{S(T_j)}(\Pi_{\mathrm{mid}}T_j)(s-\tau,-\tau) = \frac{\Upsilon^p_{\mathrm{mid}}(T_j)(u,\tau)}{S(T_j)}(\Pi_{\mathrm{mid}}T_j)(s,\tau),$$

where  $T_j = \mathcal{I}_{o_1}(\lambda_{k_j}F_j)$  is a decorated tree appearing in the decomposition of F into a product of planted trees. If  $F_j = \mathbf{1}$ , with loss of generality, we suppose that  $o_1 = (\mathfrak{t}_1, 0)$ , then

$$\frac{\Upsilon^{p}_{\mathrm{mid}}(T_{j})(u,\tau)}{S(T_{j})}(\Pi_{\mathrm{mid}}T_{j})(s,\tau) = \left(\frac{1}{2}e^{-i\tau P_{o_{1}}(k_{j})}u_{k_{j}}(\tau) + \frac{1}{2}u_{k_{j}}(0)\right)e^{isP_{o_{1}}(k_{j})}$$
$$= \frac{1}{2}e^{i(s-\tau)P_{o_{1}}(k_{j})}u_{k_{j}}(\tau) + \frac{1}{2}e^{isP_{o_{1}}(k_{j})}u_{k_{j}}(0)$$

and

$$\frac{\tilde{\Upsilon}_{\mathrm{mid}}^{p}(T_{j})(u,-\tau)}{S(T_{j})}(\Pi_{\mathrm{mid}}T_{j})(s-\tau,-\tau) = \left(\frac{1}{2}e^{i\tau P_{o_{1}}(k_{j})}u_{k_{j}}(0) + \frac{1}{2}u_{k_{j}}(\tau)\right)e^{i(s-\tau)P_{o_{1}}(k_{j})}$$
$$= \frac{1}{2}e^{i(s-\tau)P_{o_{1}}(k_{j})}u_{k_{j}}(\tau) + \frac{1}{2}e^{isP_{o_{1}}(k_{j})}u_{k_{j}}(0).$$

For  $(\mathfrak{t}_1, 1)$ , we proceed analogously with the conjugate. For a more general  $F_j$ , we have:

$$(\Pi_{\mathrm{mid}}T_j)(s-\tau,-\tau) = \frac{1}{2}e^{i(s-\tau)P_{o_1}(k_j)}((\Pi_{\mathrm{mid},1}F_j)(s-\tau,-\tau) + (\Pi_{\mathrm{mid},2}F_j)(s-\tau,0)).$$

Then  $F_j$  is of the form  $\mathcal{I}_{o_2}(\lambda_{k_j}\hat{F}_j)$ . Thus, we have

$$\left( \Pi_{\mathrm{mid},1} \mathcal{I}_{o_2}(\lambda_{k_j} \hat{F}_j) \right) (s-\tau,-\tau) = -i |\nabla|^{\alpha} (k_j) \int_{-\tau}^{s-\tau} e^{i\xi P_{o_2}(k_j)} (\Pi_{\mathrm{mid}} \hat{F}_j)(\xi,-\tau) d\xi,$$
  
=  $-i |\nabla|^{\alpha} (k_j) \int_0^s e^{i(\xi-\tau) P_{o_2}(k_j)} (\Pi_{\mathrm{mid}} \hat{F}_j)(\xi-\tau,-\tau) d\xi.$ 

and

$$\left( \Pi_{\text{mid},2} \mathcal{I}_{o_2}(\lambda_{k_j} \hat{F}_j) \right) (s-\tau,0) = -i |\nabla|^{\alpha} (k_j) \int_0^{s-\tau} e^{i\xi P_{o_2}(k)} (\Pi_{\text{mid}} \hat{F}_j)(\xi,-\tau) d\xi$$
  
=  $-i |\nabla|^{\alpha} (k_j) \int_{\tau}^s e^{i(\xi-\tau) P_{o_2}(k)} (\Pi_{\text{mid}} \hat{F}_j)(\xi-\tau,-\tau) d\xi$ 

We conclude by applying the induction hypothesis on  $\hat{F}_j$  that is

$$\frac{\tilde{\Upsilon}^p_{\mathrm{mid}}(F_j)(u,-\tau)}{S(F_j)}(\Pi_{\mathrm{mid}}\hat{F}_j)(s-\tau,-\tau) = \frac{\Upsilon^p_{\mathrm{mid}}(F_j)(u,\tau)}{S(F_j)}(\Pi_{\mathrm{mid}}\hat{F}_j)(s,\tau).$$

We recall the scheme given by the midpoint rule (6.43)

$$U_{\text{mid},k}^{n,r}(\tau,v) = \sum_{T \in \hat{\tau}_0^{r,k}(R)} \frac{\Upsilon_{\text{mid}}^p(T)(v,\tau)}{S(T)} (\Pi_{\text{mid}}^{n,r}T)(\tau).$$
(6.47)

The terms  $(\Pi_{\text{mid}}^{n,r}T)(\tau)$  are constructed in a similar way as  $(\Pi_{\text{mid}}T)(\tau)$ . The main difference happens for the computation of the time integrals. Indeed,  $(\Pi_{\text{mid}}^{n,r}T)(\tau)$  performs an approximation with a polynomial interpolation and we need to do it in a symmetric way. We need the following lemma on the polynomial interpolation in order to guarantee this property:

**Lemma 6.4.2** If the interpolation nodes  $a_j \in [0, 1], j = 0, ..., r$  are symmetrically distributed, i.e.  $a_j = 1 - a_{r-j}, j = 0, ..., r$ , then

$$\tilde{p}_r(s-\tau,-\tau) = \sum_{j=0}^r e^{-ia_j\tau\mathcal{L}_{low}} p_j(s-\tau,-\tau) = e^{-i\tau\mathcal{L}_{low}} \tilde{p}_r(s,\tau),$$
(6.48)

where we have used the short hand notation

$$\tilde{p}_r(s,\tau) := \tilde{p}_r(\exp(is\mathcal{L}_{low}),\tau).$$

*Proof.* To begin with, for any  $j = 0, \ldots, r$  we have by definition of the interpolating polynomials  $\tilde{p}_r(s, \tau)$  and  $\tilde{p}_r(s, -\tau)$ 

$$\tilde{p}_r(a_j\tau,\tau) = e^{ia_j\tau\mathcal{L}_{\text{low}}}, \quad \tilde{p}_r(-a_j\tau,-\tau) = e^{ia_j\tau\mathcal{L}_{\text{low}}}.$$

Thus in particular we have

$$\tilde{p}_r(a_j\tau - \tau, -\tau) = \tilde{p}_r(-a_{r-j}\tau, -\tau) = e^{ia_{r-j}\tau\mathcal{L}_{low}}$$
$$= e^{i\tau\mathcal{L}_{low}}e^{-ia_j\tau\mathcal{L}_{low}} = e^{i\tau\mathcal{L}_{low}}\tilde{p}_r(a_j\tau, \tau)$$

for each j = 0, ..., r. Thus, for any given  $\tau$ ,  $\tilde{p}_r(s - \tau, -\tau)$  and  $e^{-i\tau \mathcal{L}_{low}} \tilde{p}_r(s, \tau)$  are two polynomials in s of degree  $\leq r$  which match at r + 1 distinct points, so they are identical.

**Theorem 6.4.3** The scheme given by (6.43) is symmetric.

*Proof.* The proof works in the same manner as for Proposition (6.4.1). The main difference is the use of the operator  $\mathcal{K}_{o_2,j}^{k,r}$ . We suppose that

$$\frac{\tilde{\Upsilon}_{\rm mid}^p(F_j)(u,-\tau)}{S(F_j)}(\Pi_{\rm mid}^{n,r}\hat{F}_j)(s-\tau,-\tau) = \frac{\Upsilon_{\rm mid}^p(F_j)(u,\tau)}{S(F_j)}(\Pi_{\rm mid}^{n,r}\hat{F}_j)(s,\tau)$$
(6.49)

and we consider

$$\left(\Pi_{\mathrm{mid},1}^{n,r}\mathcal{I}_{o_2}(\lambda_{k_j}\hat{F}_j)\right)(s-\tau,-\tau) = \mathcal{K}_{o_2,1}^{k_j,r}\left((\Pi_{\mathrm{mid}}^{n,r}\hat{F}_j)(\cdot,-\tau),n\right)(s-\tau,-\tau)$$

From (6.49), we know that  $\prod_{mid}^{n,r} \hat{F}_j$  is of the form

$$(\Pi_{\mathrm{mid}}^{n,r}\hat{F}_j)(s,\tau) = e^{i(s-\tau)\mathcal{F}_{\mathrm{dom}}(\hat{F}_j)}A(s-\tau) + e^{is\mathcal{F}_{\mathrm{dom}}(\hat{F}_j)}A(s)$$

Now, when we apply the operator  $\mathcal{K}^{k,r}_{o_2,j}$ , we get among various cases the exact integration

$$\begin{split} &\sum_{\ell=0}^{r} \int_{-\tau}^{s-\tau} e^{i\xi\mathcal{L}_{\mathrm{dom}}} e^{-ia_{j}\tau\mathcal{L}_{\mathrm{low}}} p_{\ell,r}(\xi,-\tau) \Big( e^{i\tau\mathcal{F}_{\mathrm{dom}}(\hat{F}_{j})} A(\xi+\tau) + A(\xi) \Big) d\xi \\ &= \sum_{\ell=0}^{r} \int_{0}^{s} e^{i(\xi-\tau)\mathcal{L}_{\mathrm{dom}}} e^{-ia_{j}\tau\mathcal{L}_{\mathrm{low}}} p_{\ell,r}(\xi-\tau,-\tau) \Big( e^{i\tau\mathcal{F}_{\mathrm{dom}}(\hat{F}_{j})} A(\xi) + A(\xi-\tau) \Big) d\xi \\ &= \int_{0}^{s} e^{i(\xi-\tau)\mathcal{L}_{\mathrm{dom}}} e^{-i\tau\mathcal{L}_{\mathrm{low}}} \tilde{p}_{r}(s,\tau) \Big( e^{i\tau\mathcal{F}_{\mathrm{dom}}(\hat{F}_{j})} A(\xi) + A(\xi-\tau) \Big) d\xi \end{split}$$

where from the second to the third line, we have used the assumption (6.48) and we have

 $\mathcal{L}_{\text{dom}} + \mathcal{L}_{\text{low}} = P_{o_2}(k_j) + \mathcal{F}_{\text{dom}}(\hat{F}_j).$ 

Therefore, we obtain in the end

$$e^{-i\tau P_{o_2}(k_j)} \int_0^s e^{i\xi\mathcal{L}_{\text{dom}}} \tilde{p}_r(s,\tau) \Big( A(\xi) + e^{-i\tau\mathcal{F}_{\text{dom}}(\hat{F}_j)} A(\xi-\tau) \Big) d\xi$$

which allows us to conclude the symmetry of the method.

## 6.4.2 Conditions for symmetry

Based on the general expression of the scheme (6.45) we can arrive at sufficient conditions for the methods to be symmetric. The following observation is crucial:

**Lemma 6.4.4** Let T a decorated tree in  $\hat{\mathcal{T}}_0^k(R)$  as introduced in (6.29). We have

$$\mathscr{F}_{dom}(T) + \sum_{e \in \tilde{E}_T} \mathscr{F}_{low}(T^e) = \sum_{v \in L_T} P_{o_{e_v}}(k_v)$$

where  $e_v$  is the outgoing edge of v in T and  $o_{e_v}$  corresponds to the edge decoration of  $e_v$ . The  $k_v$  are the leaves decorations corresponding to the frequencices. The dominant part  $\mathscr{F}_{dom}(T)$  and the lower parts  $\mathscr{F}_{low}(T^e)$  depend on them.

Before proving this statement let us briefly exhibit the meaning based on a simple example already introduced in Example (21).

Example 24 We consider the simple decorated tree from the NLSE

$$T = \mathcal{I}_{(\mathfrak{t}_1, 0)}(\lambda_k \tilde{T}) = ,$$

with  $k = -k_1 + k_2 + k_3$  we use the fact that

$$P_{(\mathfrak{t}_1,0)}(k) = -k^2, \quad P_{(\mathfrak{t}_1,1)}(k) = k^2$$

to find

$$\sum_{v \in L_T} P_{o_{e_v}}(k_v) = P_{(\mathfrak{t}_1,1)}(k_1) + P_{(\mathfrak{t}_1,0)}(k_2) + P_{(\mathfrak{t}_1,0)}(k_3) = k_1^2 - k_2^2 - k_3^2$$

Moreover we have already established in Example (21) that

$$\mathscr{F}_{\operatorname{dom}}(T) = \mathscr{F}_{\operatorname{dom}}(\tilde{T}) + P_{(\mathfrak{t}_1,0)}(k) = 2k_1^2 - k^2$$

On the other hand, we have

$$\mathscr{F}_{\text{low}}(\tilde{T}) = k^2 - k_1^2 - k_2^2 - k_3^2$$

and the set  $\tilde{E}_T$  is composed of only one edge which is the only blue edge in T. We have  $T^e = \tilde{T}$ . In the end

$$\begin{aligned} \mathscr{F}_{\rm dom}(T) + \sum_{e \in \tilde{E}_T} \mathscr{F}_{\rm low}(T^e) &= \mathscr{F}_{\rm dom}(T) + \mathscr{F}_{\rm low}(T) \\ &= 2k_1^2 - k^2 + k^2 - k_1^2 - k_2^2 - k_3^2 \\ &= k_1^2 - k_2^2 - k_3^2 \\ &= \sum_{v \in L_T} P_{o_{e_v}}(k_v). \end{aligned}$$

*Proof of Lemma (6.4.5).* We proceed by induction on the size of the trees. If T is of the form  $\mathcal{I}_{(t_1,a)}(\lambda_k \mathbf{1})$  then

$$\mathscr{F}_{\operatorname{dom}}(\mathcal{I}_{(\mathfrak{t}_1,a)}(\lambda_k \mathbf{1})) = P_{(\mathfrak{t}_1,a)}(k)$$

which allows us to conclude the desired result since  $\tilde{E}_T$  is empty and this tree has only one leaf giving a contribution  $P_{(\mathfrak{t}_1,a)}(k)$ . If T is of the form  $\mathcal{I}_{(\mathfrak{t}_1,a)}(\lambda_k \mathcal{I}_{(\mathfrak{t}_2,a)}(\lambda_k F))$  then

$$\mathscr{F}_{\text{dom}}(T) + \sum_{e \in \tilde{E}_T} \mathscr{F}_{\text{low}}(T^e) = P_{(\mathfrak{t}_1, a)}(k) + \mathscr{F}_{\text{dom}}(\mathcal{I}_{(\mathfrak{t}_2, a)}(\lambda_k F)) + \mathscr{F}_{\text{low}}(\mathcal{I}_{(\mathfrak{t}_2, a)}(\lambda_k F)) + \sum_{e \in \tilde{E}_T \setminus \{\bar{e}\}} \mathscr{F}_{\text{low}}(T^e)$$

where  $\bar{e}$  denotes the edge such that  $T^{\bar{e}} = \mathcal{I}_{(t_2,a)}(\lambda_k F)$ . Now, we use Definition (6.3.2) to notice that

$$\mathscr{F}_{\mathrm{dom}}(\mathcal{I}_{(\mathfrak{t}_{2},a)}(\lambda_{k}F)) + \mathscr{F}_{\mathrm{low}}(\mathcal{I}_{(\mathfrak{t}_{2},a)}(\lambda_{k}F)) = P_{(\mathfrak{t}_{2},a)}(k) + \mathscr{F}_{\mathrm{dom}}(F).$$

By definition, we have also

$$P_{(\mathfrak{t}_2,a)}(k) + P_{(\mathfrak{t}_1,a)}(k) = 0.$$
(6.50)

We got in the end

$$\begin{aligned} \mathscr{F}_{\mathrm{dom}}(T) + \sum_{e \in \tilde{E}_T} \mathscr{F}_{\mathrm{low}}(T^e) &= \mathscr{F}_{\mathrm{dom}}(F) + \sum_{e \in \tilde{E}_T \setminus \{\bar{e}\}} \mathscr{F}_{\mathrm{low}}(T^e) \\ &= \mathscr{F}_{\mathrm{dom}}(F) + \sum_{e \in \tilde{E}_F} \mathscr{F}_{\mathrm{low}}(F^e). \end{aligned}$$

We continue the induction by observing that F is a product of trees in  $\tilde{\mathcal{T}}_0^{r,k}(R)$ . We apply the induction hypothesis on each of these trees and use the fact that  $\mathscr{F}_{dom}$  is additive for the forest product in order to conclude.

**Lemma 6.4.5** Let  $T_0 \cdot T_1 \ldots \cdot T_m \subset \mathcal{I}_{(t_2,0)}(\lambda_k F) \in \hat{\mathcal{T}}_2^k(R)$  be a splitting of F as introduced in (6.29). Then we have

$$\sum_{j=0}^{m} \left( \mathscr{F}_{dom}(T_j) + \sum_{e \in \tilde{E}_{T_j}} \mathscr{F}_{low}(T_j^e) \right) = \sum_{v \in L_F} P_{o_{e_v}}(k_v) + P_{(\mathfrak{t}_2,0)}(k).$$
(6.51)

*Proof.* This is a consequence of Lemma (6.4.4) applied to each of the  $T_j$ . In the end, we do not get all the leaves of the  $T_j$  but only the ones in F because the root of the  $T_j$   $(j \ge 0)$  is associated with a leaf in a  $T_i$ . Indeed, this introduced a cancellation of the type (6.50).

Example 25 Let us consider the following forest

such that  $T = \mathcal{I}_{(\mathfrak{t}_2,0)}(\lambda_k F)$  is identified with

with  $k = -k_1 - k_4 + k_2 + k_3 + k_5$ . We consider the following forest splitting of T, with  $T_0 \cdot T_1 \subset T$ :

$$T_0 = \underbrace{k_4}^{\ell} \underbrace{k_5}^{\ell}, T_1 = \underbrace{k_1}^{k_2} \underbrace{k_3}^{k_3}$$

with  $k = k_4 - \ell - k_5$  and  $\ell = -k_1 + k_2 + k_3$ . Let us compute both sides of the identity (6.51) for this forest splitting: Beginning with the left hand side where we have

$$\mathscr{F}_{\rm dom}(T_0) = 2k_4^2$$

Moreover, since the tree has just one single blue edge the sum simplifies and we find

$$\sum_{e \in \tilde{E}_{T_0}} \mathscr{F}_{\text{low}}(T_0) = \mathscr{F}_{\text{low}}(T_0^e) = -k_4^2 - k_5^2 - \ell^2 + k_5^2$$

and, similarly for  $T_1$  we have

$$\begin{split} \mathscr{F}_{\text{dom}}(T_1) &= 2k_1^2, \\ \sum_{e \in \tilde{E}_{T_1}} \mathscr{F}_{\text{low}}(T_1^e) &= -k_1^2 - k_2^2 - k_3^2 + \ell^2. \end{split}$$



For the right hand side on the other hand we obtain

$$\sum_{v \in L_F} P_{o_{e_v}}(k_v) = k_4^2 + k_1^2 - k_2^2 - k_3^2 - k_5^2,$$
$$P_{(i_2,0)}(k) = -k^2.$$

Combining all of the above expressions clearly shows that the identity (6.51) is indeed satisfied in the present example.

**Remark 6.4.6** The appearance of the term  $P_{(t_2,0)}(k)$  in (6.51) is due to the fact that here we consider forest splittings of trees inside  $\hat{\mathcal{T}}_2^k(R)$ . Had we instead chosen to work with trees inside  $\hat{\mathcal{T}}_0^k(R)$  this term would disappear from the above identity. Essentially, like in the previous proof, if we want to express the sum of the dominant and lower order parts in a forest splitting we can take advantage of cancellations of the form (6.50), meaning as soon as a blue dotted and brown solid edge are adjacent this leads to cancellation of the contribution from the nodal decoration in the overall identity. This means the only terms left are those which cannot be paired with an edge of conjugate colour, in particular the root and all the leaves of the resulting tree.

Proposition 6.4.7 If the coefficients b, satisfy the following simple relation

$$-\left(\prod_{j=0}^{m} e^{z_j}\right) b_{\mathbf{a},\chi,T,T_0\cdot\ldots\cdot T_m}(-\tau,-z_j) = b_{1-\mathbf{a},1-\chi,T,T_0\cdot\ldots\cdot T_m}(\tau,z_j)$$
(6.52)

for every  $\mathbf{a} \in [0,1]^{\tilde{E}_F}$ ,  $\chi \in \{0,1\}^{L_T}$  and any splitting  $T_0 \cdot T_1 \dots \cdot T_m \subset T$ , then the method (6.45) is symmetric.

*Proof.* Let us consider the adjoint method  $\widehat{\Phi}_{n \to n+1} = \Phi_{n+1 \to n}^{-1}$ , which can be expressed as follows

$$\begin{split} u_k^{n+1} &= e^{i\tau P_{o_1}(k)} u_k^n - \sum_{T \in \hat{\tau}_2^{r,k}(R)} \sum_{\mathbf{a} \in [0,1]^{\tilde{E}_F}} \sum_{\chi \in \{0,1\}^{L_T}} \sum_{T_0 \cdot T_1 \dots \cdot T_m \subset T} C_T \\ b_{\mathbf{a},\chi,T,T_0 \cdot \dots \cdot T_m}(\tau, i\tau \mathscr{F}_{\mathrm{dom}}(T_j), j \in \{0, \dots, m\}) \\ &\prod_{e \in \hat{E}_{T_j}} e^{i\tau a_e \mathscr{F}_{\mathrm{low}}(T_j^e)} \frac{\Upsilon_{\chi}^p(T)(u_{k_v}^{n+\chi_v}, v \in L_T), \tau}{S(T)}. \end{split}$$

By using Lemma (6.4.5), we have

$$u_{k}^{n+1} = e^{i\tau P_{o_{1}}(k)} u_{k}^{n} - e^{i\tau P_{o_{1}}(k)} \sum_{T \in \tilde{\mathcal{T}}_{0}^{r,k}(R)} \sum_{\mathbf{a} \in [0,1]^{\tilde{E}_{F}}} \sum_{\chi \in \{0,1\}^{L_{T}}} \sum_{T_{0} \cdot T_{1} \dots \cdot T_{m} \subset T} C_{T} \\ \left( \prod_{j=0}^{m} e^{i\tau \mathscr{F}_{dom}(T_{j})} \right) b_{\mathbf{a},\chi,T,T_{0} \dots \cdot T_{m}} (-\tau, -i\tau \mathscr{F}_{dom}(T_{j}), j \in \{0, \dots, m\}) \\ \prod_{e \in \tilde{E}_{T_{j}}} e^{i\tau (1-a_{e}) \mathscr{F}_{low}(T_{j}^{e})} \frac{\Upsilon_{\chi}^{p}(T) (u_{k_{v}}^{n+1-\chi_{v}}, v \in L_{T}, \tau)}{S(T)} \prod_{v \in L_{T}} e^{-i\tau P_{o_{e_{v}}}(k_{v})}.$$

$$(6.53)$$

Here we have used the fact that

$$\Upsilon^{p}_{\chi}(T)(u_{k_{v}}^{n+1-\chi_{v}}, v \in L_{T}, \tau) \prod_{v \in L_{T}} e^{-i\tau P_{o_{e_{v}}}(k_{v})} = \Upsilon^{p}_{\chi}(T)(u_{k_{v}}^{n+\chi_{v}}, v \in L_{T}, \tau)$$

which can be proved easily by induction on T using the recursive definition of  $\Upsilon^p_{\chi}$ . We have also used the identity

$$P_{o_1}(k) = -P_{(\mathfrak{t}_2,0)}(k).$$

By comparing (6.45) and (6.53) we immediately obtain the conditions (6.52) for symmetry.

## 6.4.3 Examples

In this section we illustrate the general framework introduced previously on two examples: the nonlinear Schrödinger equation (see Section (6.4.3)) and the Korteweg-de Vries equation (see Section (6.4.3)).

## The Nonlinear Schrödinger equation

As a first example let us consider the cubic nonlinear Schrödinger (NLS) equation

$$i\partial_t u(t,x) + \Delta u(t,x) = |u(t,x)|^2 u(t,x) \quad (t,x) \in \mathbf{R} \times \mathbb{T}^d$$
(6.54)

with an initial condition

$$u_{|t=0} = u_0. (6.55)$$

We start with the construction of a first-order symmetric low-regularity scheme for (6.54) and illustrate how our general framework covers both the previous explicit low-regularity schemes [BS22], [ABBS22b] and the case of symmetric low-regularity schemes for NLS which was recently introduced and then studied in [AB23b] and [FMS23]. We then exhibit our new symmetric midpoint rule framework (6.43), which in particular allows for a symmetric second order scheme which is optimal in the sense of regularity.

Note that the Schrödinger equation (6.54) fits into the general framework (6.1) with

$$\mathcal{L}\left(\nabla, \frac{1}{\varepsilon}\right) = \Delta, \quad \alpha = 0 \quad \text{and} \quad p(u, \overline{u}) = u^2 \overline{u}.$$

Here  $\mathcal{L} = \{\mathbf{t}_1, \mathbf{t}_2\}, P_{\mathbf{t}_1} = -\lambda^2$  and  $P_{\mathbf{t}_2} = \lambda^2$ , and the structure constant  $C_T = 1$  for all  $T \in \hat{\mathcal{T}}_0^{r,k}(R)$ , for any  $r \in \mathbf{N}$ . Then, we denote by | an edge decorated by  $(\mathbf{t}_1, 0)$ ,  $\vdots$  an edge denoted by  $(\mathbf{t}_1, 1)$  by | an edge decorated by  $(\mathbf{t}_2, 0)$  and by | an edge decorated by  $(\mathbf{t}_2, 1)$ . The set  $\hat{\mathcal{T}}_0^{0,k}(R)$  is given by:

$$\hat{T}_{0}^{0,k}(R) = \left\{ T_{0}, T_{1}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{0} = \overset{(k)}{\longrightarrow}, \quad T_{1} = \overset{(k)}{\longrightarrow}, \quad (6.56)$$

and  $\hat{\mathcal{T}}_0^{1,k}(R)$  is given by:

$$\hat{T}_{0}^{1,k}(R) = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{2} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{2} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{2} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{2} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf{Z}^{d} \right\}, \quad T_{3} = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, k_{i} \in \mathbf$$

If we take all coefficients equal to zero whenever T is not given by



and we consider only the forest F = T, then the general formula (6.45) reduces to a single term of the form:

$$u_{k}^{n+1} = e^{-i\tau k^{2}} u_{k}^{n} + e^{-i\tau k^{2}} \sum_{a \in [0,1]} \sum_{\chi \in \{0,1\}^{L_{T}}} b_{a,\chi}(\tau, i\tau \mathscr{F}_{\text{dom}}(T))$$
$$e^{i\tau a \mathscr{F}_{\text{low}}(T)} \frac{\Upsilon_{\chi}^{p}(T)(u_{k_{v}}^{n+\chi_{v}}, v \in L_{T})}{S(T)}.$$

Note here  $|L_T| = 3$  so we can equivalently write the above expression in the following form in Fourier coordinates. Indeed, by noting that  $\mathscr{F}_{\text{dom}}(T) = 2k_1^2$ ,  $\mathscr{F}_{\text{low}}(T) = 2k_2k_3 - 2k_1k_2 - 2k_1k_3$  we have

$$u_{k}^{n+1} = e^{-i\tau k^{2}} u_{k}^{n} + e^{-i\tau k^{2}} \sum_{\substack{k=-k_{1}+k_{2}+k_{3}}} \sum_{\substack{a\in[0,1]}} \sum_{\chi\in\{0,1\}^{3}} b_{a,\chi}(\tau,2i\tau k_{1}^{2})} e^{i\tau a2(k_{2}k_{3}-k_{1}k_{2}-k_{1}k_{3})} \overline{\hat{v}_{k_{1}}^{n+\chi_{1}}} \hat{v}_{k_{2}}^{n+\chi_{2}} \hat{v}_{k_{3}}^{n+\chi_{3}}.$$
(6.58)

First of all we note that the first order integrator developed in [OS18] falls in this category: Take  $b_{0,(0,0,0)}(\tau,z) = -i\tau\varphi_1(z)$ 

and all other coefficients to zero then we find

$$u_k^{n+1} = e^{-i\tau k^2} u_k^n - i e^{-i\tau k^2} \sum_{k=-k_1+k_2+k_3} \tau \varphi_1(2i\tau k_1^2) \overline{u_{k_1}^n} u_{k_2}^n u_{k_3}^n,$$

which is exactly equal to the integrator introduced in [OS18, (4)]. In physical space the above scheme is given by

$$u^{n+1} = \Phi^{\tau}_{\mathrm{NLS},1}(u^n) = e^{i\tau\Delta} u^n - ie^{i\tau\Delta} \left( (u^n)^2 \varphi_1(-2i\tau\Delta)\overline{u^n} \right), \tag{6.59}$$

where the filter function  $\varphi_1$  is defined as  $\varphi_1(\sigma) = \frac{e^{\sigma} - 1}{\sigma}$ .

Let us now consider symmetric schemes. Following Proposition (6.4.7) the scheme (6.45) is symmetric if the following equality is satisfied for all  $a \in [0, 1], \chi \in \{0, 1\}^3$ :

$$-e^{z}b_{a,\chi}(-\tau,-z) = b_{1-a,1-\chi}(\tau,z).$$
(6.60)

Intuitively speaking the above equations provide a sufficient condition relating the coefficients  $b_{a,\chi}$  and  $b_{1-a,1-\chi}$  therefore allowing us to find symmetric schemes if we specify one of the two for each value of  $a, \chi$ . There are a large class of first order schemes in this form, but perhaps one of the simplest ones is the following symmetrised version of the integrator from [OS18] which was recently introduced in [AB23b]: Take  $b_{0,(0,0,0)}(\tau, z) = i/2\tau\varphi_1(z/2)$ , then by (6.60) we should choose  $b_{1,(1,1,1)}(\tau, z) = i/2\tau\varphi_1(-z/2)$ . We take all other coefficients equal to zero, which results precisely in the following integrator:

$$u^{n+1} = \Phi^{\tau}_{\text{NLS},2}(u^n) = e^{i\tau\Delta}u^n - i\frac{\tau}{2}e^{i\tau\Delta}\left((u^n)^2\varphi_1(-i\tau\Delta)\overline{u^n}\right) - i\frac{\tau}{2}\left((u^{n+1})^2\varphi_1(i\tau\Delta)\overline{u^{n+1}}\right).$$
(6.61)

Note this is not the only symmetric first order integrator that can be found in this way. For example we could have taken

$$b_{0,(1,0,0)}(\tau,z) = i/2\tau\varphi_1(z/2), \quad b_{1,(0,1,1)}(\tau,z) = i/2\tau\varphi_1(-z/2),$$

and all other coefficients zero.

Next we choose the coefficients

$$b_{a,(\chi_1,\chi_2,\chi_3)}(\tau,z) = -i\frac{\tau}{16}\varphi_1(z)$$

for every  $a, \chi_j \in \{0, 1\}$ . The other coefficients are set to be zero which leads to the following symmetric scheme (6.54)

$$u^{n+1} = \Phi^{\tau}_{\text{NLS},3}(u^{n})$$

$$= e^{i\tau\Delta}u^{n} - i\frac{\tau}{16}e^{i\tau\Delta} \Big( (u^{n} + e^{-i\tau\Delta}u^{n+1})^{2}\varphi_{1}(-2i\tau\Delta) \Big(\overline{u^{n}} + e^{i\tau\Delta}\overline{u^{n+1}}\Big) \Big)$$

$$- i\frac{\tau}{16} \Big( (e^{i\tau\Delta}u^{n} + u^{n+1})^{2}\varphi_{1}(2i\tau\Delta) \Big( e^{-i\tau\Delta}\overline{u^{n}} + \overline{u^{n+1}} \Big) \Big).$$
(6.62)

The above scheme can also be recursively derived by the general framework of the midpoint rule (6.43) and therefore allows for higher order symmetric counterparts which are optimal in the sense of regularity. Our characterisation of symmetric schemes in Proposition (6.4.7) immediately confirm this method to be symmetric, since for all  $a, \chi$  we have

$$-e^{z}b_{a,\chi}(-\tau,-z) = -i\frac{\tau}{16}e^{z}\varphi_{1}(-z) = -i\frac{\tau}{16}\frac{1-e^{z}}{-z} = \varphi_{1}(z)b_{1-a,1-\chi}(\tau,z)$$

**Proposition 6.4.8** The scheme (6.62) can be derived from the general tree series expansion (6.43).

*Proof.* At first order it follows from (6.43) that we have

$$U_{k}^{r}(\tau, u) = \sum_{T \in \hat{\tau}_{0}^{0,k}(R)} \frac{\Upsilon_{\text{mid}}^{p}(T)(u, \tau)}{S(T)} (\Pi_{\text{mid}}^{n,r}T_{0})(\tau)$$
  
=  $\frac{\Upsilon_{\text{mid}}^{p}(T_{0})(u, \tau)}{S(T_{0})} (\Pi_{\text{mid}}^{n,r}T_{0})(\tau) + \sum_{k=-k_{1}+k_{2}+k_{3}} \frac{\Upsilon_{\text{mid}}^{p}(T_{1})(u, \tau)}{S(T_{1})} (\Pi_{\text{mid}}^{n,r}T_{1})(\tau)$ 

From the definition of the symmetry factor, one has

$$S(T_0) = 1, \quad S(T_1) = 2$$

for  $S(T_1)$  the factor two is due to the fact that we have two solid brown edges | attached to the same node in  $T_1$ . Moreover, we have:

$$\begin{split} \Upsilon^p_{\rm mid}(T_0)(u,\tau) &= u_k^\ell \\ \hat{\Upsilon}^p_{\rm mid}(T_0)(u,\tau) &= \frac{1}{2} (e^{i\tau k^2} u_k^{\ell+1} + u_k^\ell) \end{split}$$

where we have used  $\ell$  instead of n such that to not create confusion with  $\Pi^{n,r}$ . By multiplicativity, for the following tree:

$$\tilde{T}_1 = \underbrace{k_1}^{k_2} \underbrace{k_3}^{k_3},$$

we have

$$\begin{split} \Upsilon^{p}_{\mathrm{mid}}(T_{1})(u,\tau) &= \hat{\Upsilon}^{p}_{\mathrm{mid}}(\tilde{T}_{1})(u,\tau) \\ &= 2\frac{1}{2} (e^{-i\tau k_{1}^{2}} \bar{u}_{k_{1}}^{\ell+1} + \bar{u}_{k_{1}}^{\ell}) \frac{1}{2} (e^{i\tau k_{2}^{2}} u_{k_{2}}^{\ell+1} + u_{k_{2}}^{\ell}) \frac{1}{2} (e^{i\tau k_{3}^{2}} u_{k_{3}}^{\ell+1} + u_{k_{3}}^{\ell}) \end{split}$$

On the other hand, we have

$$(\Pi_{\mathrm{mid}}^{n,r}T_0)(\tau) = e^{-i\tau k^2}, \quad (\Pi_{\mathrm{mid}}^{n,r}\tilde{T}_1)(s,\tau) = e^{i(k_1^2 - k_2^2 - k_3^2)},$$

Then,

$$(\Pi_{\text{mid}}^{n,r} T_1)(\tau) = e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,r}((\Pi_{\text{mid}}^{n,r-1} \tilde{T}_1)(\cdot,\tau), n)(\tau)$$

We compute the scheme for n = 1 and r = 0. We obtain the following term:

$$\mathcal{K}_{o_2}^{k,r}((\Pi_{\mathrm{mid}}^{n,r-1}\tilde{T}_1)(\cdot,\tau),n)(\tau) = -i\int_0^\tau e^{is\mathcal{L}_{\mathrm{dom}}}ds\bigg(\frac{1+e^{i\tau\mathcal{L}_{\mathrm{low}}}}{2}\bigg)$$

where

$$\mathcal{L}_{\text{dom}} = 2k_1^2, \quad \mathcal{L}_{\text{low}} = k^2 - k_1^2 - k_2^2 - k_3^2$$

In the end, we have

$$(\Pi_{\text{mid}}^{n,r}T_1)(\tau) = -i\tau\varphi_1(2i\tau k_1^2) \left(\frac{e^{-i\tau k^2} + e^{-i\tau(k_1^2 + k_2^2 + k_3^2)}}{2}\right).$$

We note that

$$\varphi_1(2i\tau k_1^2)e^{-2i\tau k_1^2} = \varphi_1(-2i\tau k_1^2).$$

Therefore, we have

$$\begin{split} &\sum_{k=-k_1+k_2+k_3} \frac{\Upsilon^p_{\mathrm{mid}}(T_1)(u,\tau)}{S(T_1)} (\Pi^{n,r}_{\mathrm{mid}}T_1)(\tau) = -i\sum_{k=-k_1+k_2+k_3} \frac{e^{-i\tau k^2}}{2} \tau \varphi_1(2i\tau k_1^2) \\ &\frac{1}{2} (e^{-i\tau k_1^2} \bar{u}_{k_1}^{\ell+1} + \bar{u}_{k_1}^{\ell}) \frac{1}{2} (e^{i\tau k_2^2} u_{k_2}^{\ell+1} + u_{k_2}^{\ell}) \frac{1}{2} (e^{i\tau k_3^2} u_{k_3}^{\ell+1} + u_{k_3}^{\ell}) \\ &+ \tau \frac{\varphi_1(-2i\tau k_1^2)}{2} \frac{1}{2} (\bar{u}_{k_1}^{\ell+1} + e^{i\tau k_1^2} \bar{u}_{k_1}^{\ell}) \frac{1}{2} (u_{k_2}^{\ell+1} + e^{-i\tau k_2^2} u_{k_2}^{\ell}) \frac{1}{2} (u_{k_3}^{\ell+1} + e^{-i\tau k_3^2} u_{k_3}^{\ell}). \end{split}$$

In physical space this leads to the first order symmetric low regularity integrator for the NLS equation (6.62).

Remark 6.4.9 The derivation of the scheme (6.62) from the general midpoint Duhamel iterations exhibits an interesting

recipe for resonance based schemes that are constructed for equations of the form

$$i\partial_t u(t,x) + \mathcal{L}(\nabla, \frac{1}{\varepsilon})u(t,x) = |\nabla|^{\alpha} p(u(t,x), \overline{u}(t,x)),$$
  
$$u(0,x) = v(x).$$

Indeed, suppose we have already obtained an explicit *first-order* resonance based scheme for the above equation (cf. [OS18, HS17b] etc.) of the general form

$$u^{n+1} = \Phi_{\tau}(u^n),$$

where  $\Phi_{\tau}$  is a general nonlinear map representing the time step, then this can be easily converted to a second order symmetric method simply by considering instead

$$u^{n+1} = \Phi_{\tau}\left(\frac{e^{-i\tau\mathcal{L}\left(\nabla,\frac{1}{\varepsilon}\right)}u^{n+1} + u^n}{2}\right).$$

**Remark 6.4.10** In similar vein to Proposition (6.4.8) we could derive the scheme (6.61) from a generalised tree series expansion. However, instead of using a midpoint iteration of Duhamel's formula as introduced in Section (6.3.3) we would have to iterate in the following way: By averaging (6.34) and (6.35) we find

$$u_{k}(t_{n}+s) = \frac{e^{-isk^{2}}u_{k}(t_{n}) + e^{-i(s-\tau)k^{2}}u_{k}(t_{n}+\tau)}{2}$$
$$-\frac{i}{2}e^{-isk^{2}}\sum_{k=-k_{1}+k_{2}+k_{3}}\underbrace{\int_{0}^{s}e^{i\tilde{s}k^{2}}\overline{u_{k_{1}}(t_{n}+\tilde{s})}u_{k_{2}}(t_{n}+\tilde{s})u_{k_{3}}(t_{n}+\tilde{s})d\tilde{s}}_{=:\mathcal{I}_{1}}$$
$$-\frac{i}{2}e^{-isk^{2}}\sum_{k=-k_{1}+k_{2}+k_{3}}\underbrace{\int_{\tau}^{s}e^{i\tilde{s}k^{2}}\overline{u_{k_{1}}(t_{n}+\tilde{s})}u_{k_{2}}(t_{n}+\tilde{s})u_{k_{3}}(t_{n}+\tilde{s})d\tilde{s}}_{=:\mathcal{I}_{2}}$$

Instead of iterating this midpoint expression throughout all appearances of  $u_j(t_n + \tilde{s})$  in the above expression (as we do for the Duhamel midpoint iterates) we could just as well choose to iterate the left endpoint Duhamel formula (6.34) in the terms from  $\mathcal{I}_1$  and the right endpoint Duhamel formula (6.35) in the terms from  $\mathcal{I}_2$ . Repeating this process can be captured with a decorated tree series in analogous manner to the midpoint iterations and truncating such an expansion again leads to symmetric low-regularity schemes, including (after one such iteration and truncation of all terms involving at least double integrals) the scheme (6.61).

## **Proposition 6.4.11** The schemes (6.59), (6.61) and (6.62) have a local error of order $\mathcal{O}(\tau^2 \nabla u)$ .

*Proof.* First, for the explicit scheme (6.59) the local error directly follows from Theorem (6.3.15) and can be computed using Definition (6.3.12) as it is performed in the proof of [BS22, Cor. 5.1]. In order to obtain the local error bounds of the implicit schemes (6.61), (6.62) one needs to apply Theorem (6.3.15) and to combine it with a fixed-point argument on the numerical flow. To go further, the first order convergence of these schemes follow by combining the local error bound with a stability argument, we refer to the works [MS22, AB23b] which perform this analysis in full detail.  $\Box$ 

**Remark 6.4.12** The symmetric scheme (6.61) was first rigorously analysed in [AB23b]. In particular it was shown in [AB23b] that the local error of the scheme is of order  $\mathcal{O}(\tau^2 \nabla u)$ , which is optimal in regard of the regularity assumptions. Indeed, the scheme (6.61) does not require more regularity on the solution than previously constructed asymmetric low regularity integrators such as (6.59) introduced in [OS18, BS22]. As the scheme (6.61) is symmetric it is naturally also of second order; however, not under optimal regularity assumptions (see also Remark (6.2.6)). More precisely, by exploiting the tools presented in [AB23b] one can show that the scheme (6.61) (as well as (6.62)) is of second order with a local error of order  $\mathcal{O}(\tau^3 \nabla \Delta u)$ . This error structure imposes more regularity on the solution than asymmetric low-regularity integrators such as the ones proposed in [BS22] which only require the boundedness of two additional derivatives instead of three due to the local error of the form  $\mathcal{O}(\tau^3 \Delta u)$ .

Our new symmetric midpoint rule framework (6.43) allows for a symmetric second order scheme which is optimal in the sense of regularity, i.e., has a local error structure of the form  $\mathcal{O}(\tau^3 \Delta u)$ , see the scheme (6.63) below.

**Proposition 6.4.13** The second order scheme coming from (6.43) is given by:

$$u^{n+1} = \varphi_{NLS,4}^{\tau}(u^n) = e^{i\tau\Delta}u^n$$
(6.63)

$$-i\frac{\tau}{8}e^{i\tau\Delta}\Big((u^{n}+e^{-i\tau\Delta}u^{n+1})^{2}(\varphi_{1}(-2i\tau\Delta)-\varphi_{2}(-2i\tau\Delta))\Big(\overline{u^{n}}+e^{i\tau\Delta}\overline{u^{n+1}}\Big)\Big)$$
$$-i\frac{\tau}{8}\Big((e^{i\tau\Delta}u^{n}+u^{n+1})^{2}\varphi_{2}(-2i\tau\Delta)\Big(e^{i\tau\Delta}\overline{u^{n}}+e^{2i\tau\Delta}\overline{u^{n+1}}\Big)\Big)$$

with a local error structure of the form  $\mathcal{O}(\tau^3 \Delta u)$  and  $\varphi_2(\sigma) = \frac{e^{\sigma} - \varphi_1(\sigma)}{\sigma}$ .

**Remark 6.4.14** In practice the computational effort required to compute  $u^{n+1}$  in (6.63) is not significantly larger than the solution of (6.15) or other previous symmetric low-regularity methods for the NLSE. In particular, a similar analysis to that presented in [AB23b, Section 3], [BMS22, Appendix A] and [MS23] shows that the implicit equations can be solved efficiently using fixed point iterations, and that the number of iterations required is independent of the number of spatial discretisation points.

**Remark 6.4.15** One can find the coefficients b for the scheme (6.63) such that it is of the form given by (6.45).

Proof of Proposition (6.4.13). For the scheme of order two (r = 1) and n = 2, one has to consider:

$$\begin{split} U_k^{n,1}(\tau,u) &= \sum_{T \in \hat{\tau}_0^{1,k}(R)} \frac{\Upsilon_{\text{mid}}^p(T)(u,\tau)}{S(T)} (\Pi_{\text{mid}}^{n,1}T_0)(\tau) \\ &= \frac{\Upsilon_{\text{mid}}^p(T_0)(u,\tau)}{S(T_0)} (\Pi_{\text{mid}}^{n,1}T_0)(\tau) + \sum_{k=-k_1+k_2+k_3} \frac{\Upsilon_{\text{mid}}^p(T_1)(u,\tau)}{S(T_1)} (\Pi_{\text{mid}}^{n,1}T_1)(\tau) \\ &+ \sum_{k=-k_1+k_2+k_3-k_4+k_5} \frac{\Upsilon_{\text{mid}}^p(T_2)(u,\tau)}{S(T_2)} (\Pi_{\text{mid}}^{n,1}T_2)(\tau) \\ &+ \sum_{k_1-k_2-k_3+k_4+k_5=k} \frac{\Upsilon_{\text{mid}}^p(T_3)(u,\tau)}{S(T_3)} (\Pi_{\text{mid}}^{n,1}T_3)(\tau). \end{split}$$

From the definition of the symmetry factor, we have

$$S(T_2) = 1 \times 2 = 2, \quad S(T_3) = 2 \times 2 = 4,$$

for  $S(T_2)$  the factor one corresponds to the fact that for the node on top of the first blue edges the symmetry factor is one. Indeed, the trees on top of the brown edges are different: a leaf decorated by  $k_5$  is different from a tree having three leaves. Moreover, we have:

$$\Upsilon^{p}_{\mathrm{mid}}(T_{j})(u,\tau) = \Upsilon^{p}(T_{j})(\frac{1}{2}(e^{i\tau k^{2}}u^{n+1}+u^{n})), \quad j \in \{2,3\},$$

where

$$\Upsilon^{p}(T_{2})(u) = 4\bar{u}_{k_{1}}u_{k_{2}}u_{k_{3}}\bar{u}_{k_{4}}u_{k_{5}}, \quad \Upsilon^{p}(T_{3})(u) = 4u_{k_{1}}\bar{u}_{k_{2}}\bar{u}_{k_{3}}u_{k_{4}}u_{k_{5}}.$$

The factor 4 in both expressions comes from the two brown edges that appear twice inside the decorated trees  $T_2$  and  $T_3$ . For the term  $(\prod_{\text{mid}}^{2,1} T_1)(\tau)$ , we proceed with interpolation at two nodes  $(a_0 = 0, a_1 = 1)$ :

$$p_2(s,\tau) = 1 + \frac{s}{\tau} \left( e^{is\mathcal{L}_{\text{low}}} - 1 \right),$$

where

$$\mathcal{L}_{\rm dom} = 2k_1^2, \quad \mathcal{L}_{\rm low} = k^2 - k_1^2 - k_2^2 - k_3^2$$

We obtain

$$\mathcal{K}_{o_2}^{k,1}((\Pi_{\text{mid}}^{2,0}\tilde{T}_1)(\cdot,\tau),2)(\tau) = -i\int_0^\tau e^{is\mathcal{L}_{\text{dom}}}ds - i\int_0^\tau se^{is\mathcal{L}_{\text{dom}}}ds \left(\frac{e^{i\tau\mathcal{L}_{\text{low}}}-1}{\tau}\right) - i\tau\varphi_1(2i\tau k_1^2) - i\tau\varphi_2(2i\tau k_1^2)\left(e^{ik^2 - k_1^2 - k_2^2 - k_3^2} - 1\right).$$

Therefore,

$$\sum_{k=-k_1+k_2+k_3} \frac{\Upsilon^p_{\mathrm{mid}}(T_1)(u,\tau)}{S(T_1)} (\Pi^{2,1}_{\mathrm{mid}}T_1)(\tau)$$

ł

$$= \left(-i\tau\varphi_1(2i\tau k_1^2) - i\tau\varphi_2(2i\tau k_1^2)\left(e^{ik^2 - k_1^2 - k_2^2 - k_3^2} - 1\right)\right) \\ \times \left(e^{-i\tau k_1^2}\bar{u}_{k_1}^{n+1} + \bar{u}_{k_1}^n\right) \frac{1}{2} \left(e^{i\tau k_2^2} u_{k_2}^{n+1} + u_{k_2}^n\right) \frac{1}{2} \left(e^{i\tau k_3^2} u_{k_3}^{n+1} + u_{k_3}^n\right)$$

For the decorated tree  $T_2$ , we have

$$\left(\Pi_{\rm mid}^{2,1} T_2\right)(\tau) = e^{-i\tau k^2} \mathcal{K}_{o_2}^{k,1}\left((\Pi_{\rm mid}^{2,0} F_2)(\cdot,\tau),2\right)(0,\tau)$$
(6.64)

where

$$F_2 = \mathcal{I}_{(\mathfrak{t}_1,1)}(\lambda_{k_4})\mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{k_5})T_1$$

Then,

$$(\Pi_{\text{mid}}^{2,0}\mathcal{I}_{(\mathfrak{t}_{1},1)}(\lambda_{k_{4}})\mathcal{I}_{(\mathfrak{t}_{1},0)}(\lambda_{k_{5}}))(s,\tau) = e^{is(k_{4}^{2}-k_{5}^{2})}$$

and for  $\tilde{k} = -k_1 + k_2 + k_3$ 

$$(\Pi_{\text{mid}}^{2,0}T_1)(s,\tau) = \frac{1}{2}e^{-is\tilde{k}^2} \Big( \mathcal{K}_{o_2}^{\tilde{k},0}((\Pi_{\text{mid}}^{2,-1}\tilde{T}_1)(\cdot,\tau),2)(s,\tau) \\ + \mathcal{K}_{o_2}^{\tilde{k},0}((\Pi_{\text{mid}}^{2,-1}\tilde{T}_1)(\cdot,\tau),2)(s,0) \Big),$$

Now, because of n = 2, we perform a direct interpolation of the full operator which gives

$$\mathcal{K}_{o_2}^{\tilde{k},0}((\Pi_{\mathrm{mid}}^{2,-1}\tilde{T}_1)(\cdot,\tau),2)(s,\tau) = -i\int_{\tau}^s ds \left(\frac{1+e^{i\tau(\tilde{k}^2+k_1^2-k_2^2-k_3^2)}}{2}\right).$$

We obtain

$$(\Pi_{\text{mid}}^{2,0}T_1)(s,\tau) = -i\frac{(2s-\tau)}{2}e^{-is\tilde{k}^2}\left(\frac{1+e^{i\tau(\tilde{k}^2+k_1^2-k_2^2-k_3^2)}}{2}\right)$$

Therefore, we find

$$(\Pi_{\rm mid}^{2,0}F_2)(s,\tau) = -i\frac{(2s-\tau)}{2}e^{is(k_4^2-k_5^2-\tilde{k}^2)}\left(\frac{1+e^{i\tau(\tilde{k}^2+k_1^2-k_2^2-k_3^2)}}{2}\right)$$

and by performing again an interpolation of the full operator:

$$\begin{split} \left(\Pi_{\text{mid}}^{2,1}T_2\right)(\tau) &= -\int_0^\tau \frac{(2s-\tau)}{2} ds \\ & e^{-i\tau k^2} \left(\frac{1+e^{i\tau (k^2+k_4^2-k_5^2-\tilde{k}^2)}}{2}\right) \left(\frac{1+e^{i\tau (\tilde{k}^2+k_1^2-k_2^2-k_3^2)}}{2}\right) \\ &= 0 \end{split}$$

because we have

$$\int_0^\tau \frac{(2s-\tau)}{2} ds = 0$$

A similar computation shows that

$$\left(\Pi_{\rm mid}^{2,1}T_3\right)(\tau) = 0$$

The local error analysis follows from the proof of [BS22, Cor. 5.3].

Example 26 To illustrate how the general formula can be used to express more general higher order resonance based

schemes, let us consider the following low-regularity integrator

$$u^{n+1} = e^{i\tau\Delta}u - i\frac{\tau}{2}e^{i\tau\Delta}\left((u^n)^2(\varphi_1(-i\tau\Delta) - \varphi_2(-i\tau\Delta))\overline{u^n}\right) - i\frac{\tau}{2}\left(\left(u^{n+1}\right)^2(\varphi_1(i\tau\Delta) - \varphi_2(i\tau\Delta))\overline{u^{n+1}}\right) - i\frac{\tau}{2}e^{i\frac{\tau}{2}\Delta}\left(e^{i\frac{\tau}{2}\Delta}u^n\right)^2\varphi_2(-i\tau\Delta)e^{i\frac{\tau}{2}\Delta}\overline{u^n} - i\frac{\tau}{2}e^{i\frac{\tau}{2}\Delta}\left(e^{-i\frac{\tau}{2}\Delta}u^{n+1}\right)^2\varphi_2(i\tau\Delta)e^{-i\frac{\tau}{2}\Delta}\overline{u^{n+1}} - \frac{\tau^2}{8}|u^n|^4e^{-i\tau\Delta}u^{n+1} + \frac{\tau^2}{8}|u^{n+1}|^4e^{i\tau\Delta}u^n$$
(6.65)

which is motivated from [BS22, (5.16)] and has a local error of the form  $\mathcal{O}(\tau^3 \Delta)$ . We will see that this scheme is in the form of the general formula (6.45) with r = 1. Indeed, we already saw at the beginning of this section that

$$\hat{\mathcal{T}}_{0}^{1,k}(R) = \left\{ T_{0}, T_{1}, T_{2}, T_{3}, \, k_{i} \in \mathbf{Z}^{d} \right\}$$

where  $T_i$ , i = 0, 1, 2, 3, were defined in (6.56) and (6.57). In the interest of brevity we will not derive all the coefficients b in the expression of (6.65) in the form (6.45). Instead let us focus on the coefficients arising from the contribution to (6.45) arising from  $T = \hat{T}_2$  where  $\hat{T}_2$  is obtained by removing the brown edge attached to the root. To understand this we first recall the forest splittings of this choice of T from Example (23):



Let us now consider the contribution from the second of these splittings,  $T_0 \cdot T_1 \cdot T_2$  with

$$\tilde{T}_0 = \mathbf{1}, \quad \tilde{T}_1 = \overset{(k_1)}{\longrightarrow} \overset{(k_2)}{\longrightarrow} , \tilde{T}_2 = \overset{(k_1)}{\longrightarrow} \overset{(k_2)}{\longrightarrow} \overset{(k_3)}{\longrightarrow}$$

where from Kirchhoff's law we have  $k = -k_4 + l + k_5$  and  $l = -k_1 + k_2 + k_3$ . The dominant and lower order operators arising in these splittings are given by

$$\begin{aligned} \mathscr{F}_{\text{dom}}(T_0) &= 0, \mathscr{F}_{\text{low}}(T_0) = 0\\ \mathscr{F}_{\text{dom}}(\tilde{T}_1) &= 2k_4^2, \mathscr{F}_{\text{low}}(\tilde{T}_1) = k^2 - k_4^2 - l^2 - k_5^2 = 2(-k_4l - k_4k_5 + lk_5)\\ \mathscr{F}_{\text{dom}}(\tilde{T}_2) &= 2k_1^2, \mathscr{F}_{\text{low}}(\tilde{T}_2) = l^2 - k_1^2 - k_2^2 - k_3^2 = 2(-k_1k_2 - k_1k_3 + k_2k_3), \end{aligned}$$

Moreover we have  $|\tilde{E}_T|$  (there are only two blue edges in T corresponding to time-integration) and  $|L_T| = 5$  (T has 5 leaves), and given the above splitting,  $|\tilde{E}_{\tilde{T}_0}| = 0$ ,  $|\tilde{E}_{\tilde{T}_1}| = |\tilde{E}_{\tilde{T}_2}| = 1$ . Thus the contribution from this term to the overall sum in (6.45) is of the form (recall that  $C_T = 1$  for all T in the NLSE case)

$$\sum_{\mathbf{a}\in[0,1]^2} \sum_{\chi\in\{0,1\}^5} b_{\mathbf{a},\chi,T,\tilde{T}_0\cdot\tilde{T}_1\cdot\tilde{T}_2}(\tau, i\tau\mathscr{F}_{\mathrm{dom}}(\tilde{T}_0), i\tau\mathscr{F}_{\mathrm{dom}}(\tilde{T}_1), i\tau\mathscr{F}_{\mathrm{dom}}(\tilde{T}_2))$$
$$e^{i\tau a_1\mathscr{F}_{\mathrm{low}}(\tilde{T}_1)}e^{i\tau a_2\mathscr{F}_{\mathrm{low}}(\tilde{T}_2)} \frac{\Upsilon^p_{\chi}(T)(u_{k_v}^{n+\chi_v}, v\in L_T, \tau)}{S(T)}$$

From the derivation in the proof of Proposition (6.4.13) and (6.44) we have that

$$\frac{\Upsilon_{\chi}^{p}(T)(u_{k_{v}}^{n+\chi_{v}}, v \in L_{T}, \tau)}{S(T)} = 2e^{-i\tau\chi_{1}k_{1}^{2}}\overline{u}_{k_{1}}^{\chi_{1}}e^{i\tau\chi_{2}k_{2}^{2}}u_{k_{2}}^{\chi_{2}}e^{i\tau\chi_{3}k_{3}^{2}}u_{k_{3}}^{\chi_{3}}e^{-i\tau\chi_{4}k_{4}^{2}}\overline{u}_{k_{4}}^{\chi_{4}}e^{i\tau\chi_{5}k_{5}^{2}}u_{k_{5}}^{\chi_{5}}$$

Thus the contribution to (6.45) equals

$$\begin{split} \sum_{\mathbf{a}\in[0,1]^2} \sum_{\chi\in\{0,1\}^5} b_{\mathbf{a},\chi,T,\tilde{T}_0\cdot\tilde{T}_1\cdot\tilde{T}_2}(\tau,0,2i\tau k_4^2,2i\tau k_1^2) \\ e^{i\tau a_1(k^2-k_4^2-(-k_1+k_2+k_3)^2-k_5^2)} e^{i\tau a_2((-k_1+k_2+k_3)^2-k_1^2-k_2^2-k_3^2)} \\ 2e^{-i\tau\chi_1k_1^2} \overline{u}_{k_1}^{\chi_1} e^{i\tau\chi_2k_2^2} u_{k_2}^{\chi_2} e^{i\tau\chi_3k_3^2} u_{k_3}^{\chi_3} e^{-i\tau\chi_4k_4^2} \overline{u}_{k_4}^{\chi_4} e^{i\tau\chi_5k_5^2} u_{k_5}^{\chi_5} \end{split}$$

Let us now show that the quintic terms in (6.65) arise precisely from these contributions: Indeed suppose we choose

$$b_{\mathbf{0},(0,0,0,1),T,\tilde{T}_{0}.\tilde{T}_{1}.\tilde{T}_{2}}(\tau,z_{0},z_{1},z_{2}) = -\frac{\tau^{2}}{16},$$
(6.66)

$$b_{(1,1),(1,0,0,0,0),T,\tilde{T}_{0},\tilde{T}_{1},\tilde{T}_{2}}(\tau,z_{0},z_{1},z_{2}) = -e^{z_{0}+z_{1}+z_{2}}\frac{\tau^{2}}{16}$$
(6.67)

and all other coefficients b in the above expression equal to zero then we arrive precisely at the contributions of the form

$$-\frac{\tau^2}{8}|u^n|^4e^{-i\tau\Delta}u^{n+1} + \frac{\tau^2}{8}|u^{n+1}|^4e^{i\tau\Delta}u^n$$

in the overall scheme, corresponding to the quintic terms in (6.65). The remaining terms in the scheme can be expressed similarly from contributions from lower rank trees  $T_0, T_1$ . Moreover, we note that the coefficients as given by (6.66) clearly satisfy (6.52) and that the same holds for the coefficients of lower order contributions, thus confirming that the scheme (6.65) is symmetric.

## The Korteweg-de Vries equation

The Korteweg-de Vries (KdV) equation is given by

$$\partial_t u + \partial_x^3 u = \frac{1}{2} \partial_x u^2 \tag{6.68}$$

It fits into the general framework with

$$\mathcal{L}\left(\nabla, \frac{1}{\varepsilon}\right) = i\partial_x^3, \quad \alpha = 1 \quad \text{and} \quad p(u, \overline{u}) = p(u) = i\frac{1}{2}u^2.$$

Here  $\mathcal{L} = \{\mathfrak{t}_1, \mathfrak{t}_2\}$ ,  $P_{\mathfrak{t}_1} = -\lambda^3$  and  $P_{\mathfrak{t}_2} = \lambda^3$ . Moreover, in this case the structure constant  $C_T$  reflects the presence of the Burger's nonlinearity in the iterations of Duhamel's formula, which means

$$C_T = \prod_{\substack{e=(v,u)\in \tilde{E}_T\\u\in N_T\setminus\{\varrho_T\}}} (-1)^{\mathfrak{p}(e)} i\mathfrak{o}(u)$$

where we recall  $\mathfrak{e}(e) = (\mathfrak{t}(e), \mathfrak{p}(e))$  is the edge decoration of e with  $\mathfrak{t}(e) \in \mathfrak{L}$  and  $\mathfrak{p}(e) \in \{0, 1\}$ . Note by Kirchhoff's law the above definition is invariant under the choice of node u or v for an edge e = (u, v) in the product, so long as the node is an interior one. Then, we denoted by | an edge decorated by  $(\mathfrak{t}_1, 0)$  and by | an edge decorated by  $(\mathfrak{t}_2, 0)$ . Following the formalism given in [BHZ19], we can provide the rules that generate the trees obtained by iterating Duhamel's formula:

$$R(|) = \{(|,|)\}, \quad R(|) = \{(|),()\}$$

The general framework (6.42) derived in Section (6.3.3) builds the foundation of the first- and second-order resonance based schemes presented below for the KdV equation (6.68). The structure of the schemes depends on the regularity of the solution.

**Corollary 6.4.16** For the KdV equation (6.68) the general midpoint scheme (6.42) takes at first order the form

$$u^{\ell+1} = e^{-\tau \partial_x^3} u^{\ell} + \frac{1}{24} \left( e^{-\tau \partial_x^3} \partial_x^{-1} u^{\ell} + \partial_x^{-1} u^{\ell+1} \right)^2 - \frac{1}{24} e^{-\tau \partial_x^3} \left( \partial_x^{-1} u^{\ell} + e^{\tau \partial_x^3} \partial_x^{-1} u^{\ell+1} \right)^2$$
(6.69)

with a local error of order  $\mathcal{O}\left(\tau^2 \partial_x^2 u\right)$  at first-order and with a local error of order  $\mathcal{O}\left(\tau^3 \partial_x^4 u\right)$  at second order.

**Remark 6.4.17** Note that this schemes has been obtained in [MS23]. It was shown that this scheme is of even order for higher regularity in  $H^4$  (see [MS22, Thm 5.2]). By embedding this scheme into our general framework, we know that it has the same local error analysis as the second-order scheme introduced in [BS22].

$$u^{\ell+1} = e^{-\tau \partial_x^3} u^{\ell} + \frac{1}{6} \left( e^{-\tau \partial_x^3} \partial_x^{-1} u^{\ell} \right)^2 - \frac{1}{6} e^{-\tau \partial_x^3} \left( \partial_x^{-1} u^{\ell} \right)^2 + \frac{\tau^2}{4} e^{-\tau \partial_x^3} \Psi \left( i \tau \partial_x^2 \right) \left( \partial_x \left( u^{\ell} \partial_x (u^{\ell} u^{\ell}) \right) \right)$$
(6.70)

with a local error of order  $\mathcal{O}\left(\tau^3\partial_x^4 u\right)$  and a suitable filter function  $\Psi$  satisfying

$$\Psi = \Psi(i\tau\partial_x^2), \quad \Psi(0) = 1, \quad \|\tau\Psi(i\tau\partial_x^2)\partial_x^2\|_r \le 1.$$

Proof. The proof follows the line of argumentation to the analysis for the Schrödinger equation. For the first-order scheme, we have

$$U_k^{n,0}(\tau, v) = \frac{\Upsilon_{\rm mid}^p(T_0)(\tau, v)}{S(T_0)} \Pi_{\rm mid}^{n,0}(T_0)(\tau) + \sum_{k=k_1+k_2} \frac{\Upsilon_{\rm mid}^p(T_1)(\tau, v)}{S(T_1)} \Pi_{\rm mid}^{n,0}(T_1)(\tau).$$
(6.71)

where the trees of interest are

$$\hat{\mathcal{T}}_{0}^{0,k}(R) = \{T_{0}, T_{1}, k_{i} \in \mathbf{Z}^{d}\}, \quad T_{0} = \overset{(k)}{\downarrow} \text{ and } T_{1} = \overset{(k)}{\downarrow}$$

and in symbolic notation takes the form

$$T_1 = \mathcal{I}_{(t_1,0)}(\mathcal{I}_{(t_2,0)}(\lambda_k F_1)) \quad F_1 = \mathcal{I}_{(t_1,0)}(\lambda_{k_1})\mathcal{I}_{(t_1,0)}(\lambda_{k_2}) \quad \text{with } k = k_1 + k_2.$$

For the first term we readily obtain that

$$\frac{\Upsilon^p_{\rm mid}(T_0)(\tau, v)}{S(T_0)} \Pi^{n,0}_{\rm mid}(T_0)(\tau) = e^{-i\tau k^3} \hat{v}_k$$

It remains to compute the second term. Note that thanks to (6.28) we have that

$$\Pi^{n,0}(T_1)(\tau) = e^{-i\tau k^3} \Pi^{n,0}(\mathcal{I}_{(t_2,0)}(\lambda_k F_1))(\tau)$$

$$= e^{-i\tau k^3} \mathcal{K}^{k,0}_{(t_2,0)}(\Pi^{n,-1}(F_1), n)(\tau)$$

$$= e^{-i\tau k^3} \mathcal{K}^{k,0}_{(t_2,0)} \left( e^{i\xi(-k_1^3 - k_2^3)}, n \right)(\tau).$$
(6.72)

where we have used for the third line

$$(\Pi^{n,-1}F_1)(\tau) = (\Pi^{n,-1}\mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{k_1}))(\tau)(\Pi^{n,-1}\mathcal{I}_{(\mathfrak{t}_1,0)}(\lambda_{k_2}))(\tau) = e^{-i\tau k_1^3} e^{-i\tau k_2^3}.$$

Next we observe that

$$P_{(t_2,0)}(k) - k_1^3 - k_2^3 = k^3 - k_1^3 - k_2^3 = 3k_1k_2(k_1 + k_2)$$

such that

$$\frac{1}{P_{(\mathfrak{t}_2,0)}(k) - k_1^3 - k_2^3}$$

can be mapped back to physical space. Therefore, we set

$$\mathcal{L}_{\text{dom}} = P_{(t_2,0)}(k) - k_1^3 - k_2^3 = 3k_1k_2(k_1 + k_2)$$

and integrate all frequencies exactly. This implies

$$\Pi^{n,0}(T_1)(\tau) = e^{-i\tau k^3} \frac{i(k_1 + k_2)}{3ik_1k_2(k_1 + k_2)} \left( e^{i\tau(k^3 - k_1^3 - k_2^3)} - 1 \right)$$
$$= \frac{1}{3k_1k_2} \left( e^{-i\tau(k_1^3 + k_2^3)} - e^{-i\tau k^3} \right).$$

Together with (6.71) this yields the scheme (6.69). For the second-order scheme, we first notice that

$$\Pi_{\rm mid}^{n,0}(T_0)(\tau) = \Pi_{\rm mid}^{n,1}(T_0)(\tau), \quad \Pi_{\rm mid}^{n,0}(T_1)(\tau) = \Pi_{\rm mid}^{n,1}(T_1)(\tau).$$

Indeed, for the tree  $T_1$ , we perform an exact integration without any discretisation. Then, we need to take into account the following trees

$$\hat{\mathcal{T}}_{0}^{1,k}(R) = \{T_{0}, T_{1}, T_{2}, k_{i} \in \mathbf{Z}^{d}\}, \quad T_{2} =$$

Then we can proceed as in the second-order schemes for the Schrödinger equation to show its contribution is zero that is

$$\Pi^{4,1}_{\mathrm{mid}}(T_2)(\tau)=0.$$

 $\bigcirc$ 

# 6.5 Numerical Experiments

We now test the practical performance of our new symmetric schemes in practical experiments evaluating both their low-regularity convergence properties and their ability to correctly preserve constants of motion in the relevant equations. In fitting with our above construction our spatial discretisation is a Fourier spectral method throughout with M modes. In order to understand the low-regularity convergence properties of our methods we follow [OS18] and consider the following types of initial data:

1. Smooth inital data,

$$u_0(x) = \frac{\cos(x)}{2 + \sin(x)}.$$
(6.73)

2. Low-regularity initial data  $u_0 \in H^\vartheta$ , for some  $\vartheta > 0$ , of the following form. Firstly we choose a vector sampled from a uniform distribution  $U_m \sim U([0,1] + i[0,1]), m = -M/2 + 1, \dots, M/2$  and then we define

$$u_0(x) := U_0 + \sum_{\substack{m=-M/2+1\\m\neq 0}}^{M/2} e^{imx} |m|^{-\vartheta} U_m.$$
(6.74)

Both choices of initial data are rescaled such that  $u_0 \mapsto u_0/||u_0||_{L^2}$ .

## 6.5.1 The Korteweg–de Vries equation

We consider the resonance based midpoint rule (6.69) introduced in [MS23] which our midpoint iterates (section (6.3.3)) are able to recover. This rule has excellent low-regularity convergence properties and at the same time is able to conserve momentum and energy of the KdV equation over long times even in the low-regularity regime. We refer to the numerical simulations made in [MS23] where the favourable properties of this method are displayed.

## 6.5.2 The Nonlinear Schrödinger equation

In the upcoming numerical experiments we compare the performance of our new symmetric integrators ((6.62) and (6.63)), to the following state-of-the-art reference schemes for the NLSE:

— The Strang splitting [MQ02], as an example of a classical symmetric numerical technique;

- The first and second order resonance based integrators introduced by Ostermann & Schratz [OS18] and Bruned & Schratz [BS22, Section 5.1.2] respectively, as examples of asymmetric low-regularity schemes;
- The symmetrised low-regularity integrator introduced by Alama Bronsard [AB23b], as an example of previous structure preserving low-regularity schemes.

In the following numerical experiments we focus on the 1d case, i.e. the NLSE formulated on  $\mathbb{T}$ , but our methods equally apply to higher dimensional settings where their favourable performance can also be observed.

In the first instance we consider the long-time structure preservation properties of our newly designed symmetric low-regularity integrators. For this we consider two first integrals of the cubic NLSE, the normalisation

$$I_0^{[NLSE]}[u] = \int_{\mathbb{T}} |u|^2 dx,$$

and the energy

$$I_{1}^{[NLSE]}[u] = \int_{\mathbb{T}} |\nabla u|^{2} + \frac{1}{2} |u|^{4} dx.$$

Symmetric numerical schemes are typically unable to preserve such conservation laws exactly, however it is known for the ODE case [HLW10, Chapter XI] (and also observed numerically for the PDE case, for example in [CCO08]) that symmetric methods can exhibit very good approximate long-time preservation of such first integrals. In the following numerical experiments we test this behaviour by looking at the error in these quantities for a fixed time step  $\tau = 0.02$  and highest frequency M = 1024, over a long time interval, much larger than  $\mathcal{O}(1/\tau)$ . Firstly, in figure (6.4) we observe that the normalisation appears to be preserved really well, in particular (the example is representative of a host of numerical experiments for various time steps which we performed) the preservation is much better than previous asymmetric resonance based schemes. We note that the Strang splitting conserves quadratic first integrals, i.e. the normalisation, to machine accuracy and thus undoubtedly outperforms our schemes on the level of normalisation preservation.

Our next numerical result for the NLSE, presented in Figure (6.6)&(6.5), shows the error in the NLSE energy over a long time interval, for a fixed time step  $\tau = 0.02$ . Albeit rigorous theory exists for the ODE case [HLW10], there is again no theoretical guarantee for the long-time preservation of the energy under symmetric methods. Indeed in practical experiments it can be seen that symmetric schemes are able to clearly outperform asymmetric integrators in the long-time approximate energy preservation. For the Strang splitting this behaviour was rigorously analysed in [Fao12] where a CFL condition was necessary to guarantee long-time approximate energy preservation beyond the realms of forward error analysis. This CFL condition is indeed observed even for smooth data in our experiments. Perhaps somewhat surprisingly our new symmetric resonance based scheme do not appear to suffer from comparable CFL conditions and, as expected, perform well for both smooth and low-regularity solutions.

Finally, we performed experiments to confirm that the low-regularity convergence properties of our symmetric schemes are at least as good as in prior asymmetric methods. In the following numerical experiments our reference solutions were computed with  $M = 2^{14}$  Fourier modes and a time step  $\tau = 10^{-6}$  with the symmetrised method from [AB23b]. In Figures (6.7) & (6.8) we choose to measure the error in  $H^1$ -norm, and observe the convergence properties of our methods for initial data of various levels of regularity. In all of these experiments the number of spatial discretisation modes was taken to be M = 1024 and the initial data chosen according to (6.74) & (6.73). In Figure (6.7) we observe that our new methods have exactly the predicted convergence properties at those levels of regularity: The integrator (6.62) is optimal for first order convergence in the sense of regularity, meaning it converges at first order in  $H^1$  with  $H^2$  data, while the integrator (6.63) is optimally convergent in the sense of regularity up to second order meaning it converges at  $\mathcal{O}(\tau)$  in  $H^1$ for data in  $H^2$  and at  $\mathcal{O}(\tau^2)$  in  $H^1$  for data in  $H^3$ . For smaller values of  $\tau$  the error forms a plateau around  $10^{-4}$  and  $10^{-7}$  in Figures 6.7a and 6.7b respectively. This is due to the error made by the pseudo-spectral space discretisation, which decreases as the regularity of the initial data is increased. The behaviour of the splitting methods observed in these experiments matches exactly with the convergence analysis given by [Lub08, BBD02] and suffers from significant order reduction in low-regularity regimes.



Figure 6.4 – Error in the normalisation  $||u^n||_{L^2}$ , for time step  $\tau = 0.02$ , M = 1024, and long-time interval  $t = n\tau \in [0, 4000]$ .



Figure 6.5 – Error in the Hamiltonian, for time step  $\tau = 0.02, M = 1024$  and  $C^{\infty}$  data.



Figure 6.6 – Error in the Hamiltonian, for time step  $\tau = 0.02$ , M = 1024, and  $H^2$  data.



Figure 6.7 –  $H^1$ -error at T = 1 as a function of the timestep  $\tau$  for low-regularity initial data.



Figure 6.8 –  $H^1$ -error at T = 1 as a function of the timestep  $\tau$  for more regular initial data.

[]

- [AB23a] Yvonne Alama Bronsard. Error analysis of a class of semi-discrete schemes for solving the Gross-Pitaevskii equation at low regularity. J. Comput. Appl. Math., 418:Paper No. 114632, 19, 2023.
- [AB23b] Yvonne Alama Bronsard. A symmetric low-regularity integrator for the nonlinear Schrödinger equation. IMA Journal of Numerical Analysis, page drad093, 2023.
- [ABBMS23] Yvonne Alama Bronsard, Yvain Bruned, Georg Maierhofer, and Katharina Schratz. Symmetric resonance based integrators and forest formulae. arXiv preprint arXiv:2305.16737, 2023.
- [ABBS22a] Yvonne Alama Bronsard, Yvain Bruned, and Katharina Schratz. Approximations of dispersive PDEs in the presence of low-regularity randomness. arXiv preprint arXiv:2205.02156, 2022.
- [ABBS22b] Yvonne Alama Bronsard, Yvain Bruned, and Katharina Schratz. Low regularity integrators via decorated trees. arXiv preprint arXiv:2202.01171, 2022.
- [ABB13] Xavier Antoine, Weizhu Bao, and Christophe Besse. Computational methods for the dynamics of the nonlinear Schrödinger/Gross-Pitaevskii equations. Comput. Phys. Commun., 184(12):2621–2633, 2013.
- [ACG21] Ioakeim Ampatzoglou, Charles Collot, and Pierre Germain. Derivation of the kinetic wave equation for quadratic dispersive problems in the inhomogeneous setting. *arXiv preprint arXiv:2107.11819*, 2021.
- [AF03] Robert A Adams and John JF Fournier. Sobolev spaces. Elsevier, 2003.
- [APR18] Harbir Antil, Johannes Pfefferer, and Sergejs Rogovs. Fractional operators with inhomogeneous boundary conditions: analysis, control, and discretization. *Commun. Math. Sci.*, 16(5):1395–1426, 2018.
- [BBD02] Christophe Besse, Brigitte Bidégaray, and Stéphane Descombes. Order estimates in time of splitting methods for the nonlinear Schrödinger equation. *SIAM Journal on Numerical Analysis*, 40(1):26–40, 2002.
- [BC23] Charles-Edouard Bréhier and David Cohen. Analysis of a splitting scheme for a class of nonlinear stochastic Schrödinger equations. *Applied Numerical Mathematics*, 186:57–83, 2023.
- [BCCH20] Yvain Bruned, Ajay Chandra, Ilya Chevyrev, and Martin Hairer. Renormalising SPDEs in regularity structures. *Journal of the European Mathematical Society*, 23(3):869–947, 2020.
- [BCD11] Hajer Bahouri, Jean-Yves Chemin, and Raphaël Danchin. Fourier analysis and nonlinear partial differential equations, volume 343 of Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer, Heidelberg, 2011.
- [BDDLV21] Christophe Besse, Stéphane Descombes, Guillaume Dujardin, and Ingrid Lacroix-Violet. Energy-preserving methods for nonlinear Schrödinger equations. *IMA J. Numer. Anal.*, 41(1):618–653, 2021.
- [BEF20] Yvain Bruned and Kurusch Ebrahimi-Fard. Bogoliubov type recursions for renormalisation in regularity structures. arXiv preprint arXiv:2006.05284, 2020.
- [Bes04] Christophe Besse. A relaxation scheme for the nonlinear Schrödinger equation. SIAM J. Numer. Anal., 42(3):934–952, 2004.
- [BG94] Giancarlo Benettin and Antonio Giorgilli. On the Hamiltonian interpolation of near-to-the-identity symplectic mappings with application to symplectic integration algorithms. J. Statist. Phys., 74(5-6):1117–1143, 1994.
- [BG22] Joackim Bernier and Benoît Grébert. Birkhoff normal forms for Hamiltonian PDEs in their energy space. J. Éc. polytech. Math., 9:681–745, 2022.
- [BGHS21] T. Buckmaster, P. Germain, Z. Hani, and J. Shatah. Onset of the wave turbulence description of the longtime behavior of the nonlinear Schrödinger equation. *Invent. Math.*, 225(3):787–855, 2021.
- [BGT04] Nicolas Burq, Patrick Gérard, and Nikolay Tzvetkov. Strichartz inequalities and the nonlinear Schrödinger equation on compact manifolds. *American Journal of Mathematics*, 126(3):569–605, 2004.
- [BHZ19] Yvain Bruned, Martin Hairer, and Lorenzo Zambotti. Algebraic renormalisation of regularity structures. Inventiones mathematicae, 215:1039–1156, 2019.

- [BHZ20] Yvain Bruned, Martin Hairer, and Lorenzo Zambotti. Renormalisation of stochastic partial differential equations. *EMS Newsl*, 115(3):7–11, 2020.
- [BLW23] Genming Bai, Buyang Li, and Yifei Wu. A constructive low-regularity integrator for the one-dimensional cubic nonlinear Schrödinger equation under Neumann boundary condition. IMA J. Numer. Anal., 43(6):3243–3281, 2023.
- [BM23] Yvain Bruned and Dominique Manchon. Algebraic deformation for (S)PDEs. Journal of the Mathematical Society of Japan, 75(2):485–526, 2023.
- [BMS22] Valeria Banica, Georg Maierhofer, and Katharina Schratz. Numerical integration of Schrödinger maps via the Hasimoto transform. *arXiv preprint arXiv:2211.01282*, 2022.
- [BMW23] Weizhu Bao, Ying Ma, and Chushan Wang. Optimal error bounds on time-splitting methods for the nonlinear Schrödinger equation with low regularity potential and nonlinearity. *arXiv preprint arXiv:2308.15089*, 2023.
- [BN22] Yvain Bruned and Usama Nadeem. Convergence of space-discretised gKPZ via regularity structures. arXiv preprint arXiv:2207.09946, 2022.
- [BOP19] Árpád Bényi, Tadahiro Oh, and Oana Pocovnicu. On the probabilistic Cauchy theory for nonlinear dispersive PDEs. Landscapes of time-frequency analysis, pages 1–32, 2019.
- [Bou93] J. Bourgain. Fourier transform restriction phenomena for certain lattice subsets and applications to nonlinear evolution equations. I. Schrödinger equations. *Geom. Funct. Anal.*, 3(2):107–156, 1993.
- [BP57] N. N. Bogoliubow and O. S. Parasiuk. Über die Multiplikation der Kausalfunktionen in der Quantentheorie der Felder. Acta Math., 97:227–266, 1957.
- [BPSS14] J. L. Bona, G. Ponce, J.-C. Saut, and C. Sparber. Dispersive blow-up for nonlinear Schrödinger equations revisited. J. Math. Pures Appl. (9), 102(4):782–811, 2014.
- [Bri20] Bjoern Bringmann. Almost Sure Local Well-Posedness for a Derivative Nonlinear Wave Equation. International Mathematics Research Notices, 2021(11):8657–8697, 01 2020.
- [Bro22] Eugen Bronasco. Exotic b-series and s-series: algebraic structures and order conditions for invariant measure sampling. *arXiv preprint arXiv:2209.11046*, 2022.
- [BS93] J. L. Bona and J.-C. Saut. Dispersive blowup of solutions of generalized Korteweg-de Vries equations. J. Differential Equations, 103(1):3–57, 1993.
- [BS94] P. B. Bochev and C. Scovel. On quadratic invariants and symplectic structure. *BIT*, 34(3):337–345, 1994.
- [BS10] Jerry L. Bona and Jean-Claude Saut. Dispersive blow-up II. Schrödinger-type equations, optical and oceanic rogue waves. *Chinese Ann. Math. Ser. B*, 31(6):793–818, 2010.
- [BS22] Yvain Bruned and Katharina Schratz. Resonance-based schemes for dispersive equations via decorated trees. In *Forum of Mathematics, Pi*, volume 10, page e2. Cambridge University Press, 2022.
- [BSV15] Matteo Bonforte, Yannick Sire, and Juan Luis Vázquez. Existence, uniqueness and asymptotic behaviour for fractional porous medium equations on bounded domains. Discrete Contin. Dyn. Syst., 35(12):5725–5767, 2015.
- [BT08] Nicolas Burq and Nikolay Tzvetkov. Random data cauchy theory for supercritical wave equations i: local theory. *Inventiones mathematicae*, 173:449–475, 2008.
- [BT13] Nicolas Burq and Nikolay Tzvetkov. Probabilistic well-posedness for the cubic wave equation. Journal of the European Mathematical Society, 16(1):1–30, 2013.
- [But72] John C Butcher. An algebraic theory of integration methods. *Mathematics of Computation*, 26(117):79–106, 1972.
- [But16] J. C. Butcher. Numerical methods for ordinary differential equations. John Wiley & Sons, Ltd., Chichester, third edition, 2016. With a foreword by J. M. Sanz-Serna.
- [BW23a] Weizhu Bao and Chushan Wang. An explicit and symmetric exponential wave integrator for the nonlinear Schrödinger equation with low regularity potential and nonlinearity. arXiv preprint arXiv:2310.20181, 2023.
- [BW23b] Weizhu Bao and Chushan Wang. Optimal error bounds on the exponential wave integrator for the nonlinear Schrödinger equation with low regularity potential and nonlinearity. arXiv preprint arXiv:2302.09262, 2023.
- [CCO08] Elena Celledoni, David Cohen, and Brynjulf Owren. Symmetric exponential integrators with an application to the cubic Schrödinger equation. *Foundations of Computational Mathematics*, 8:303–317, 2008.
- [CG12] David Cohen and Ludwig Gauckler. One-stage exponential integrators for nonlinear Schrödinger equations over long times. *BIT Numerical Mathematics*, 52:877–903, 2012.
- [CG19] Charles Collot and Pierre Germain. On the derivation of the homogeneous kinetic wave equation. arXiv preprint arXiv:1912.10368, 2019.

- [CGI22] Nicolas Camps, Louise Gassot, and Slim Ibrahim. Cubic Schrödinger half-wave equation and random initial data. Journées équations aux dérivées partielles, 2022.
   [CH16] Ajay Chandra and Martin Hairer. An analytic BPHZ theorem for regularity structures. arXiv preprint arXiv:1612.08138, 2016.
   [CHA11] A. Chabchoub, N. P. Hoffmann, and N. Akhmediev. Rogue wave observation in a water wave tank. Phys.
- *Rev. Lett.*, 106:204502, May 2011.[CHL08] David Cohen, Ernst Hairer, and Christian Lubich. Conservation of energy, momentum and actions in
- numerical discretizations of non-linear wave equations. *Numer. Math.*, 110(2):113–143, 2008.
- [Chr07] Michael Christ. Power series solution of a nonlinear Schrödinger equation. Mathematical aspects of nonlinear dispersive equations, Ann. of Math. Stud, 163:131–155, 2007.
- [CK99] Alain Connes and Dirk Kreimer. Hopf algebras, renormalization and noncommutative geometry. In *Quantum field theory: perspective and prospective*, pages 59–109. Springer, 1999.
- [CK00] Alain Connes and Dirk Kreimer. Renormalization in quantum field theory and the Riemann-Hilbert problem I: The hopf algebra structure of graphs and the main theorem. *Communications in Mathematical Physics*, 210(1):249–273, 2000.
- [CLL23] Jiachuan Cao, Buyang Li, and Yanping Lin. A new second-order low-regularity integrator for the cubic nonlinear Schrödinger equation. *IMA Journal of Numerical Analysis*, page drad017, 2023.
- [CLR20] Clémentine Courtès, Frédéric Lagoutière, and Frédéric Rousset. Error estimates of finite difference schemes for the Korteweg–de Vries equation. *IMA J. Numer. Anal.*, 40(1):628–685, 2020.
- [CM75] R. R. Coifman and Yves Meyer. On commutators of singular integrals and bilinear singular integrals. Trans. Amer. Math. Soc., 212:315–331, 1975.
- [CM07] Philippe Chartier and Ander Murua. Preserving first integrals and volume forms of additively split systems. IMA J. Numer. Anal., 27(2):381–405, 2007.
- [CO12] James Colliander and Tadahiro Oh. Almost sure well-posedness of the cubic nonlinear Schrödinger equation below  $L^2(\mathbb{T})$ . Duke Math. J., 161(3):367–414, 2012.
- [CS22a] María Cabrera Calvo and Katharina Schratz. Uniformly accurate low regularity integrators for the Klein– Gordon equation from the classical to nonrelativistic limit regime. *SIAM Journal on Numerical Analysis*, 60(2):888–912, 2022.
- [CS22b] Rémi Carles and Chunmei Su. Scattering and uniform in time error estimates for splitting method in nls. Foundations of Computational Mathematics, pages 1–40, 2022.
- [Deb11] Arnaud Debussche. Weak approximation of stochastic partial differential equations: the nonlinear case. Mathematics of Computation, 80(273):89–117, 2011.
- [DH21] Yu Deng and Zaher Hani. On the derivation of the wave kinetic equation for NLS. In *Forum of Mathematics*, *Pi*, volume 9, page e6. Cambridge University Press, 2021.
- [DH23] Yu Deng and Zaher Hani. Full derivation of the wave kinetic equation. *Inventiones mathematicae*, pages 1–182, 2023.
- [DNY22] Yu Deng, Andrea R Nahmod, and Haitian Yue. Random tensors, propagation of randomness, and nonlinear dispersive equations. *Inventiones mathematicae*, 228(2):539–686, 2022.
- [Duj09] Guillaume Dujardin. Exponential Runge-Kutta methods for the Schrödinger equation. Appl. Numer. Math., 59(8):1839–1857, 2009.
- [EH21] Dirk Erhard and Martin Hairer. A scaling limit of the parabolic anderson model with exclusion interaction. arXiv preprint arXiv:2103.13479, 2021.
- [ESS16] Johannes Eilinghoff, Roland Schnaubelt, and Katharina Schratz. Fractional error estimates of splitting schemes for the nonlinear Schrödinger equation. J. Math. Anal. Appl., 442(2):740–760, 2016.
- [Fao12] Erwan Faou. Geometric numerical integration and Schrödinger equations, volume 15. European Mathematical Society, 2012.
- [FG11] Erwan Faou and Benoît Grébert. Hamiltonian interpolation of splitting approximations for nonlinear PDEs. Found. Comput. Math., 11(4):381–415, 2011.
- [FGP10a] Erwan Faou, Benoît Grébert, and Eric Paturel. Birkhoff normal form for splitting methods applied to semilinear Hamiltonian PDEs. I. Finite-dimensional discretization. Numer. Math., 114(3):429–458, 2010.
- [FGP10b] Erwan Faou, Benoît Grébert, and Eric Paturel. Birkhoff normal form for splitting methods applied to semilinear Hamiltonian PDEs. II. Abstract splitting. Numer. Math., 114(3):459–490, 2010.

- [FMS23] Yue Feng, Georg Maierhofer, and Katharina Schratz. Long-time error bounds of low-regularity integrators for nonlinear Schrödinger equations. arXiv preprint arXiv:2302.00383, 2023.
- [Gau11] Ludwig Gauckler. Convergence of a split-step Hermite method for the Gross-Pitaevskii equation. IMA J. Numer. Anal., 31(2):396–415, 2011.
- [GKO13] Zihua Guo, Soonsik Kwon, and Tadahiro Oh. Poincaré-dulac normal form reduction for unconditional well-posedness of the periodic cubic nls. *Communications in Mathematical Physics*, 322:19–48, 2013.
- [GL10a] Ludwig Gauckler and Christian Lubich. Nonlinear Schrödinger equations and their spectral semidiscretizations over long times. Found. Comput. Math., 10(2):141–169, 2010.
- [GL10b] Ludwig Gauckler and Christian Lubich. Splitting integrators for nonlinear Schrödinger equations over long times. *Found. Comput. Math.*, 10(3):275–302, 2010.
- [GRT18] Stéphane Gaudreault, Greg Rainwater, and Mayya Tokman. KIOPS: a fast adaptive Krylov subspace solver for exponential integrators. J. Comput. Phys., 372:236–255, 2018.
- [Gub04] M. Gubinelli. Controlling rough paths. J. Funct. Anal., 216(1):86–140, 2004.
- [Gub10] Massimiliano Gubinelli. Ramification of rough paths. J. Differential Equations, 248(4):693–721, 2010.
- [Gub12] Massimiliano Gubinelli. Rough solutions for the periodic Korteweg–de Vries equation. Commun. Pure Appl. Anal., 11(2):709–733, 2012.
- [Hail4] Martin Hairer. A theory of regularity structures. Inventiones mathematicae, 198(2):269–504, 2014.
- [Hep69] Klaus Hepp. On the equivalence of additive and analytic renormalization. *Comm. Math. Phys.*, 14:67–69, 1969.
- [HKRT11] Helge Holden, Kenneth H. Karlsen, Nils Henrik Risebro, and Terence Tao. Operator splitting for the KdV equation. *Math. Comp.*, 80(274):821–846, 2011.
- [HL04] Ernst Hairer and Christian Lubich. Symmetric multistep methods over long times. *Numer. Math.*, 97(4):699–723, 2004.
- [HLO20] Marlis Hochbruck, Jan Leibold, and Alexander Ostermann. On the convergence of Lawson methods for semilinear stiff problems. *Numer. Math.*, 145(3):553–580, 2020.
- [HLR13] Helge Holden, Christian Lubich, and Nils Henrik Risebro. Operator splitting for partial differential equations with Burgers nonlinearity. *Math. Comp.*, 82(281):173–185, 2013.
- [HLW10] Ernst Hairer, Christian Lubich, and Gerhard Wanner. Geometric numerical integration, volume 31 of Springer Series in Computational Mathematics. Springer, Heidelberg, 2010. Structure-preserving algorithms for ordinary differential equations, Reprint of the second (2006) edition.
- [HNW93] E. Hairer, S. P. Norsett, and G. Wanner. Solving ordinary differential equations. I, volume 8 of Springer Series in Computational Mathematics. Springer-Verlag, Berlin, second edition, 1993. Nonstiff problems.
- [HO10] Marlis Hochbruck and Alexander Ostermann. Exponential integrators. Acta Numerica, 19:209–286, 2010.
- [Hol10] Helge Holden. Splitting methods for partial differential equations with rough solutions: Analysis and MATLAB programs, volume 11. European Mathematical Society, 2010.
- [Hör97] Lars Hörmander. Lectures on nonlinear hyperbolic differential equations, volume 26. Springer Science & Business Media, 1997.
- [HP17] Patrick Henning and Daniel Peterseim. Crank-Nicolson Galerkin approximations to nonlinear Schrödinger equations with rough potentials. *Math. Models Methods Appl. Sci.*, 27(11):2147–2184, 2017.
- [HRST22] Amirali Hannani, Matthew Rosenzweig, Gigliola Staffilani, and Minh-Binh Tran. On the wave turbulence theory for a stochastic kdv type equation–generalization for the inhomogeneous kinetic limit. *arXiv preprint* arXiv:2210.17445, 2022.
- [HS17a] Sebastian Herr and Katharina Schratz. Trigonometric time integrators for the Zakharov system. IMA J. Numer. Anal., 37(4):2042–2066, 2017.
- [HS17b] Martina Hofmanová and Katharina Schratz. An oscillatory integrator for the KdV equation. *Numer. Math*, 136:1117–1137, 2017.
- [HW74] E. Hairer and G. Wanner. On the Butcher group and general multi-value methods. *Computing (Arch. Elektron. Rechnen)*, 13(1):1–15, 1974.
- [IQT07] Arieh Iserles, G. R. W. Quispel, and P. S. P. Tse. B-series methods cannot be volume-preserving. BIT, 47(2):351–378, 2007.
- [IZ00] Arieh Iserles and Antonella Zanna. Preserving algebraic invariants with Runge-Kutta methods. J. Comput. Appl. Math., 125(1-2):69–81, 2000. Numerical analysis 2000, Vol. VI, Ordinary differential equations and integral equations.

- [IZ09] Liviu I Ignat and Enrique Zuazua. Numerical dispersive schemes for the nonlinear Schrödinger equation. SIAM journal on numerical analysis, 47(2):1366–1390, 2009.
   [JL00] Tobias Jahnke and Christian Lubich. Error bounds for exponential operator splittings. BIT, 40(4):735–744,
- 2000. [JO23] Lun Ji and Alexander Ostermann. Low regularity error estimates for high dimensional nonlinear Schrödinger
- equations. arXiv preprint arXiv:2312.11071, 2023.
- [JORS23] Lun Ji, Alexander Ostermann, Frédéric Rousset, and Katharina Schratz. Low regularity error estimates for the time integration of 2d nls. arXiv preprint arXiv:2301.10639, 2023.
- [KPS08] Christian Kharif, Efim Pelinovsky, and Alexey Slunyaev. *Rogue waves in the ocean*. Springer Science & Business Media, 2008.
- [Kul03] G. Yu. Kulikov. Symmetric Runge-Kutta methods and their stability. Russian J. Numer. Anal. Math. Modelling, 18(1):13–41, 2003.
- [Law67] J Douglas Lawson. Generalized Runge-Kutta processes for stable systems with large lipschitz constants. SIAM Journal on Numerical Analysis, 4(3):372–380, 1967.
- [LMS21] Buyang Li, Shu Ma, and Katharina Schratz. A semi-implicit low-regularity integrator for Navier-Stokes equations. arXiv preprint arXiv:2107.13427, 2021.
- [LO13] Vu Thai Luan and Alexander Ostermann. Exponential B-series: the stiff case. SIAM J. Numer. Anal., 51(6):3431–3445, 2013.
- [LPS14] Gabriel J Lord, Catherine E Powell, and Tony Shardlow. An introduction to computational stochastic PDEs, volume 50. Cambridge University Press, 2014.
- [LR04] Benedict Leimkuhler and Sebastian Reich. Simulating Hamiltonian dynamics, volume 14 of Cambridge Monographs on Applied and Computational Mathematics. Cambridge University Press, Cambridge, 2004.
- [LS11] Jani Lukkarinen and Herbert Spohn. Weakly nonlinear Schrödinger equation with random initial data. Invent. Math., 183(1):79–188, 2011.
- [LS23] Hang Li and Chunmei Su. Low regularity exponential-type integrators for the "good" Boussinesq equation. IMA Journal of Numerical Analysis, page drac081, 2023.
- [Lub08] Christian Lubich. On splitting methods for Schrödinger-poisson and cubic nonlinear Schrödinger equations. Mathematics of computation, 77(264):2141–2153, 2008.
- [LV20] Adrien Laurent and Gilles Vilmart. Exotic aromatic B-series for the study of long time integrators for a class of ergodic SDEs. *Mathematics of Computation*, 89(321):169–202, 2020.
- [LW21] Buyang Li and Yifei Wu. A fully discrete low-regularity integrator for the 1d periodic cubic nonlinear Schrödinger equation. *Numerische Mathematik*, 149(1):151–183, 2021.
- [LW22] Buyang Li and Yifei Wu. An unfiltered low-regularity integrator for the KdV equation with solutions below  $H^1$ . arXiv preprint arXiv:2206.09320, 2022.
- [MKL13] Hans Z Munthe-Kaas and Alexander Lundervold. On post-Lie algebras, Lie–Butcher series and moving frames. *Foundations of Computational Mathematics*, 13:583–613, 2013.
- [MQ02] Robert I McLachlan and G Reinout W Quispel. Splitting methods. Acta Numerica, 11:341–434, 2002.
- [MRRS22] Frank Merle, Pierre Raphaël, Igor Rodnianski, and Jeremie Szeftel. On blow up for the energy super critical defocusing nonlinear Schrödinger equations. *Invent. Math.*, 227(1):247–413, 2022.
- [MRS10] Frank Merle, Pierre Raphaël, and Jeremie Szeftel. Stable self-similar blow-up dynamics for slightly  $L^2$  super-critical NLS equations. *Geom. Funct. Anal.*, 20(4):1028–1071, 2010.
- [MS22] Georg Maierhofer and Katharina Schratz. Bridging the gap: symplecticity and low regularity on the example of the KdV equation. arXiv preprint arXiv:2205.05024, version 1, 2022.
- [MS23] Georg Maierhofer and Katharina Schratz. Bridging the gap: symplecticity and low regularity in runge-kutta resonance-based schemes. arXiv preprint arXiv:2205.05024, version 2, 2023.
- [MWZ24] Norbert J Mauser, Yifei Wu, and Xiaofei Zhao. The cubic nonlinear Schrödinger equation with rough potential. *arXiv preprint arXiv:2403.16772*, 2024.
- [NBP13] Branko Nikolić, Antun Balaž, and Axel Pelster. Dipolar Bose-Einstein condensates in weak anisotropic disorder. *Physical Review A*, 88(1):013624, 2013.
- [NPS13] Andrea R. Nahmod, Nataša Pavlović, and Gigliola Staffilani. Almost sure existence of global weak solutions for supercritical Navier-Stokes equations. SIAM J. Math. Anal., 45(6):3431–3452, 2013.

[NS15]	Andrea R Nahmod and Gigliola Staffilani. Almost sure well-posedness for the periodic 3d quintic nonlinear Schrödinger equation below the energy space. <i>Journal of the European Mathematical Society</i> , 17(7):1687–1759, 2015.
[NWZ22]	Cui Ning, Yifei Wu, and Xiaofei Zhao. An embedded exponential-type low-regularity integrator for mKdV equation. <i>SIAM Journal on Numerical Analysis</i> , 60(3):999–1025, 2022.
[ORS21]	Alexander Ostermann, Frédéric Rousset, and Katharina Schratz. Error estimates of a Fourier integrator for the cubic Schrödinger equation at low regularity. <i>Foundations of Computational Mathematics</i> , 21(3):725–765, 2021.
[ORS22a]	Alexander Ostermann, Frédéric Rousset, and Katharina Schratz. Error estimates at low regularity of splitting schemes for NLS. <i>Mathematics of Computation</i> , 91(333):169–182, 2022.
[ORS22b]	Alexander Ostermann, Frédéric Rousset, and Katharina Schratz. Fourier integrator for periodic NLS: low regularity estimates via discrete Bourgain spaces. <i>Journal of the European Mathematical Society</i> , 2022.
[OS18]	Alexander Ostermann and Katharina Schratz. Low regularity exponential-type integrators for semilinear Schrödinger equations. <i>Foundations of Computational Mathematics</i> , 18:731–755, 2018.
[OS20]	Alexander Ostermann and Chunmei Su. A Lawson-type exponential integrator for the Korteweg–de Vries equation. <i>IMA J. Numer. Anal.</i> , 40(4):2399–2414, 2020.
[OWY22]	Alexander Ostermann, Yifei Wu, and Fangyan Yao. A second-order low-regularity integrator for the nonlinear Schrödinger equation. <i>Adv. Contin. Discrete Models</i> , 23:14, 2022.
[OY22]	Alexander Ostermann and Fangyan Yao. A fully discrete low-regularity integrator for the nonlinear Schrödinger equation. Journal of Scientific Computing, $91(1)$ :9, 2022.
[RS80]	Michael Reed and Barry Simon. <i>Methods of modern mathematical physics. I.</i> Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York, second edition, 1980. Functional analysis.
[RS21]	Frédéric Rousset and Katharina Schratz. A general framework of low regularity integrators. SIAM Journal on Numerical Analysis, 59(3):1735–1768, 2021.
[RS22]	Frédéric Rousset and Katharina Schratz. Convergence error estimates at low regularity for time discretizations of KdV. <i>Pure and Applied Analysis</i> , 4(1):127–152, 2022.
[SS84]	JM Sanz-Serna. Methods for the numerical solution of the nonlinear Schrödinger equation. mathematics of computation, 43(167):21–27, 1984.
[SS88]	J. M. Sanz-Serna. Runge-Kutta schemes for Hamiltonian systems. BIT, 28(4):877–883, 1988.
[SSC18]	Jesus-Maria Sanz-Serna and Mari-Paz Calvo. <i>Numerical hamiltonian problems</i> , volume 7. Courier Dover Publications, 2018.
[ST21]	Gigliola Staffilani and Minh-Binh Tran. On the wave turbulence theory for stochastic and random multidi- mensional KdV type equations. arXiv preprint arXiv:2106.09819, 2021.
[SWZ21]	Katharina Schratz, Yan Wang, and Xiaofei Zhao. Low-regularity integrators for nonlinear dirac equations. <i>Mathematics of Computation</i> , 90(327):189–214, 2021.
[SX16]	Chenmin Sun and Bo Xia. Probabilistic well-posedness for supercritical wave equations with periodic boundary condition on dimension three. <i>Illinois Journal of Mathematics</i> , $60(2)$ :481–503, 2016.
[Tao06]	Terence Tao. Nonlinear dispersive equations, volume 106 of CBMS Regional Conference Series in Mathematics. Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 2006. Local and global analysis.
[Tat04]	Daniel Tataru. The wave maps equation. Bull. Amer. Math. Soc. (N.S.), 41(2):185–204, 2004.
[Tha12]	Mechthild Thalhammer. Convergence analysis of high-order time-splitting pseudospectral methods for nonlinear Schrödinger equations. <i>SIAM Journal on Numerical Analysis</i> , 50(6):3231–3258, 2012.
[Tho07]	Vidar Thomée. Galerkin finite element methods for parabolic problems, volume 25. Springer Science & Business Media, 2007.
[WWW <sup>+</sup> 98]	Jamie Williams, Reinhold Walser, Carl Wieman, John Cooper, and Murray Holland. Achieving steady-state bose-einstein condensation. <i>Physical Review A</i> , 57(3):2030, 1998.
[WY22]	Yifei Wu and Fangyan Yao. A first-order Fourier integrator for the nonlinear Schrödinger equation on $\mathbb{T}$ without loss of regularity. <i>Math. Comp.</i> , 91(335):1213–1235, 2022.
[WZ21]	Yifei Wu and Xiaofei Zhao. Optimal convergence of a second-order low-regularity integrator for the KdV equation. <i>IMA Journal of Numerical Analysis</i> , 42(4):3499–3528, 07 2021.
[WZ22a]	Yan Wang and Xiaofei Zhao. A symmetric low-regularity integrator for nonlinear klein-gordon equation. Mathematics of Computation, 91(337):2215–2245, 2022.

- [WZ22b] Yifei Wu and Xiaofei Zhao. Embedded exponential-type low-regularity integrators for KdV equation under rough data. *BIT Numerical Mathematics*, 62(3):1049–1090, 2022.
- [Yag09] Atsushi Yagi. Abstract parabolic evolution equations and their applications. Springer Science & Business Media, 2009.
- [ZF67] VE Zakharov and NN Filonenko. Weak turbulence of capillary waves. Journal of applied mechanics and technical physics, 8(5):37–40, 1967.
- [Zim69] W. Zimmermann. Convergence of Bogoliubov's method of renormalization in momentum space. Comm. Math. Phys., 15:208–234, 1969.